

COPYRIGHT AND USE OF THIS THESIS

This thesis must be used in accordance with the provisions of the Copyright Act 1968.

Reproduction of material protected by copyright may be an infringement of copyright and copyright owners may be entitled to take legal action against persons who infringe their copyright.

Section 51 (2) of the Copyright Act permits an authorized officer of a university library or archives to provide a copy (by communication or otherwise) of an unpublished thesis kept in the library or archives, to a person who satisfies the authorized officer that he or she requires the reproduction for the purposes of research or study.

The Copyright Act grants the creator of a work a number of moral rights, specifically the right of attribution, the right against false attribution and the right of integrity.

You may infringe the author's moral rights if you:

- fail to acknowledge the author of this thesis if you quote sections from the work
- attribute this thesis to another author
- subject this thesis to derogatory treatment which may prejudice the author's reputation

For further information contact the University's Director of Copyright Services

sydney.edu.au/copyright

Evaluation of pathology ordering by general practitioners in Australia

CLARE FRANCES BAYRAM

This thesis is submitted in fulfilment of the requirements for the degree of Doctor of Philosophy

Sydney School of Public Health Sydney Medical School University of Sydney

March, 2013

Abstract

This thesis describes and evaluates ordering of pathology tests by general practitioners (GPs) in Australia. Over the past decade the volume and cost of pathology services generated by GP orders have grown markedly, raising questions about reasons for, and appropriateness of, orders contributing to the increase. This study uses data collected between 2000 and 2010 in the Bettering the Evaluation and Care of Health (BEACH) program, from 9,842 randomly sampled GPs about

984,200 encounters and 1,490,194 pathology tests/batteries of tests.

Multiple factors contributed to the total increase in volume of GP pathology orders from 2000–02 to 2006–08: increased likelihood of GPs' deciding to order test(s), increased number of tests ordered by GPs per episode, increased number of problems managed at GP–patient encounters, and increased population attendance rates. Just 22 health problems accounted for 59% of the growth in testing.

Significant independent predictors of the volume of pathology ordered by GPs, included some GP and practice characteristics, but the principal explanatory variable was the type of problem being managed.

For six common problems with high pathology test ordering rates, appropriateness of ordering was assessed, by measuring alignment of GPs' ordering with guidance documents. Alignment was good for: hypertension, Type 2 diabetes, lipid disorders and weakness/tiredness; and poor for 'health checks' and overweight/obesity.

Of the six problems investigated, overall increases in pathology ordering were seen for both 'appropriate' and 'inappropriate' tests. However, only a small proportion of tests were deemed inappropriate. I found no evidence to support concerns raised in the literature about assumed widespread inappropriate ordering, or assertions that increases in ordering reflect disproportionate increases in inappropriate ordering For the ongoing management of chronic problems, pathology testing guidance was poor. Australia has an ageing population and therefore chronic problem management and the testing associated with it will inevitably increase. Improved guidance regarding pathology testing in chronic problem management could help support GPs' appropriate ordering in this high growth area.

Acknowledgements

I wish to acknowledge and thank the many people who have helped, supported and encouraged me throughout my PhD.

I would like to express my gratitude to my research supervisors, Professor Helena Britt and Professor Graeme Miller, for their dedication, guidance, encouragement and editorial assistance. Thank you for supporting and encouraging my professional development over the years, and for creating a nurturing and supportive workplace in which such development is possible.

Special thanks are extended to my colleagues at the Family Medicine Research Centre for their continued support and patience. In particular, I thank Lisa Valenti for her contribution in undertaking the majority of the data analysis for this thesis and Christopher Harrison for his analysis work. I am grateful to Joan Henderson for editing this thesis, and for her day-to-day support in our work environment.

I also thank Julie O'Halloran for providing endless encouragement (and empathy) and sharing her wisdom, as we undertook our PhD journeys simultaneously.

The research in this thesis was partially funded by a grant from the Quality Use of Pathology Program, Australian Government Department of Health and Ageing. The Postgraduate Research Support Scheme, and Research Student Grant Scheme from the Sydney School of Public Health at the University of Sydney provided financial support for conference attendance.

I would also like to thank the members of the pathology profession who have shared their time, knowledge and experience. In particular, I wish to acknowledge and thank Ed Wilson, Michael Harrison, and Bev Rowbotham.

I thank the organisations that have supported the BEACH program. During the period covered in this thesis, the BEACH program was supported by: Australian Government Department of Health and Ageing; AstraZeneca Pty Ltd (Australia); Merck, Sharp and Dohme (Australia) Pty Ltd; Pfizer Australia Pty Ltd; Sanofiaventis Australia Pty Ltd; Novartis Pharmaceuticals Australia Pty Ltd; Janssen-Cilag Pty Ltd; Abbott Australasia Pty Ltd; Wyeth Australia Pty Ltd; Roche Products Pty Ltd; Aventis Pharma Pty Ltd; National Prescribing Service Ltd; Australian Government Department of Veterans' Affairs; The Office of the Australian Safety and Compensation Council, Department of Employment and Workplace Relations. Sincere thanks to the GPs who have participated in BEACH, without whom this research would not have been possible.

I am grateful to my family for providing quiet reassurance and cheering me on from the sidelines. I thank my sister, Laura, for proof reading sections of this thesis. To my mum (Sharyn Sutton) and dad (Graeme Sutton), I am deeply grateful for your love, faith and support. Mum, thank you for teaching me to always consider another's perspective, to challenge what I am told, and to make up my own mind—these life skills have served me well in my research career.

Finally, I would like to thank my husband, Abdul, for his love, encouragement and mischievous sense of humour which helped maintain my equilibrium. This thesis would not have been possible without you.

Contents

Ał	Abstract	ii
Ac	Acknowledgements	iii
Li	List of tables	ix
Li	list of figures	xiii
Li	ist of boxes	xiv
Ał	Abbreviations	XV
Sv	Symbols	
Sj Gl	Hossarv	viv
1	Aims and candidate's contribution	1
1	1 1 Aims	1
	1.2 Candidate's contribution	1
2	Introduction	1
- 3	Method	20
5	3.1 An overview of the BEACH study	20
	3.2 The BEACH method	21
	3.2.1 Sample size	
	3.2.2 The sample	
	3.2.3 GP recruitment	
	3.2.4 Data collection	
	3.2.5 Data elements	
	3.2.6 SAND substudy data	
	3.2.7 Data entry and classification	
	3.2.8 Quality assurance	
	3.2.9 Validity and reliability	
	3.2.10 Representativeness	
	3.2.11 Statistical methods	
	3.2.12 Extrapolated national estimates	
4	Overview of pathology ordering	
	4.1 Background	
	4.2 Objectives	
	4.3 Method	
	4.3.1 Extrapolation	
	4.4 Results	
	4.4.1 Overview of data set	
	4.4.2 Problems generating high volumes of pathology ord	ers 55
	4.5 Discussion	

5	GP	patho	logy testing versus recommended testing	. 81
	5.1	Object	tives	. 81
	5.2	Backg	round	. 81
	5.3	Metho	vd	. 84
		5.3.1	General practice data	. 84
		5.3.2	Selecting the problems for investigation	. 84
		5.3.3	Identifying the guidelines and other sources of guidance	. 87
		5.3.4	Determining recommended pathology testing	. 88
	5.4	Hyper	tension	. 90
		5.4.1	Background	. 90
		5.4.2	Management rate in Australian general practice	. 90
		5.4.3	Pathology ordered for hypertension	. 91
		5.4.4	Guidance documents for hypertension	. 95
		5.4.5	Extent of alignment between GP testing and guidance documents	. 95
		5.4.6	Discussion	103
	5.5	Type 2	2 diabetes	111
		5.5.1	Background	111
		5.5.2	Management rate in Australian general practice	112
		5.5.3	Pathology ordered for T2D	112
		5.5.4	Guidance documents for T2D	117
		5.5.5	Extent of alignment between GP testing and guidance documents.	117
		5.5.6	Discussion	124
	5.6	Lipid	disorders	131
		5.6.1	Background	131
		5.6.2	Management rate in Australian general practice	131
		5.6.3	Pathology ordered for lipid disorders	132
		5.6.4	Guidance documents for lipid disorder	136
		5.6.5	Extent of alignment between GP testing and guidance documents.	137
		5.6.6	Discussion	143
	5.7	Health	i check	151
		5.7.1	Background	151
		5.7.2	Management rate in Australian general practice	151
		5.7.3	Pathology ordered for health check	152
		5.7.4	Guidance documents for health check	156
		5.7.5	Extent of alignment between GP testing and guidance documents.	156
	- 0	5.7.6	Discussion	163
	5.8	Weaki	ness / tiredness	169
		5.8.1	Background	169
		5.8.2	Management rate in Australian general practice	169
		5.8.3	Pathology ordering for weakness/tiredness	170
		5.8.4	Guidance for weakness/ tiredness	175
		5.8.5	Extent of alignment between GP testing and guidance documents.	176
		5.8.6	Discussion	183

	5.9 Overv	veight/obesity	188
	5.9.1	Background	188
	5.9.2	Management rate in Australian general practice	189
	5.9.3	Pathology ordering for overweight/obesity	190
	5.9.4	Guidance documents for overweight/obesity	194
	5.9.5	Extent of alignment between GP testing and guidance documents.	195
	5.9.6	Discussion	201
	5.10 Discu	ussion	206
6	Investigat	tion of FBC and lipid testing	219
	6.1 Backg	ground	219
	6.2 Objec	tives	222
	6.3 Metho	od	222
	6.4 Result	ts	225
	6.4.1	The sample	225
	6.4.2	FBC testing	226
	6.4.3	Lipid testing	234
	6.5 Discu	ssion	241
7	Predictor	s of GPs' pathology ordering	252
	7.1 Backg	ground	252
	7.2 Objec	tive	256
	7.3 Metho	od	256
	7.4 Result	ts	259
	7.4.1	The sample	259
	7.4.2	Univariate analysis	260
	7.4.3	Multivariate analysis	265
	7.5 Discu	ssion	268
8	Pathology	y generated by GP orders in the future	277
	8.1 Objec	tive	277
	8.2 Backg	ground	277
	8.3 Metho	od	279
	8.4 Result	ts	286
	8.5 Discu	ssion	295
9	Discussio	n	302
10	Conclusio	on	320
Re	ferences in	citation order	322
Re	ferences in	alphabetical order (citation order in parentheses)	351
Ap	pendices		380
•	Appendix	1: Publications and presentations from this thesis	380
	Appendix	2: Example of a 2007–08 BEACH recording form	382
	Appendix	3: GP characteristics questionnaire, 2007–08	384
	Appendix	4: Instructions for participating GPs	385
	Appendix	5: Patient information card	393

Appendix 6: ICPC-2 and ICPC-2 PLUS code groups used for analysis of	
problems managed	. 396
Appendix 7: ICPC-2 PLUS code groups used for analysis of pathology tests.	. 400

List of tables

Table 3.1:	Number of general practice professional services claimed from Medicare Australia each financial year, 2000–01 to 2009–10
Table 4.1:	Overview of data set and summary of pathology ordering, 2000–08, 2000–02 and 2006–08
Table 4.2:	Distribution of pathology orders across MBS pathology groups and most frequent individual test orders for all problems, 2000–08, 2000–02 and 2006–08
Table 4.3:	Overview of problems that accounted for at least 1% of total problem–pathology links, 2000–08, 2000–02 and 2006–08
Table 4.4:	Changes in pathology ordering for problems that accounted for at least 1% of problem–pathology links, 2000–02 compared with 2006–08 69
Table 4.5:	Extrapolated national estimated number (thousands) of management and pathology ordered for problems that accounted for at least 1% of problem–pathology links, changes from 2000–02 to 2006–08 ^(a)
Table 5.1:	Summary of hypertension data set, 2000-08, 2000-02 and 2006-08 91
Table 5.2:	Summary of pathology ordering for hypertension, 2000–08, 2000–02 and 2006–08
Table 5.3:	Rate of pathology test orders for hypertension by MBS pathology groups and most frequent individual test orders within each group, 2000–08
Table 5.4:	Rate of pathology test orders for hypertension by MBS pathology groups and most frequent individual test orders within each group, 2000–02 compared with 2006–08
Table 5.5:	Summary of guidance recommendations by most frequent individual test orders for hypertension, 2000–08
Table 5.6:	Summary of support for GP pathology ordering for the most frequent individual test orders for hypertension, 2000–08
Table 5.7:	Rate of pathology ordering for hypertension by level of support, 2000–08, 2000–02 and 2006–08
Table 5.8:	Summary of T2D data set, 2000–08, 2000–02 and 2006–08 112
Table 5.9:	Summary of pathology ordering for T2D, 2000–08, 2000–02 and 2006–08
Table 5.10:	Rate of pathology test orders for T2D by MBS pathology groups and most frequent individual test orders within each group, 2000–08 114
Table 5.11:	Rate of pathology test orders for T2D by MBS pathology groups and most frequent individual test orders within each group, 2000–02 compared with 2006–08
Table 5.12:	Prescribed medications for T2D by Anatomical Therapeutic Chemical classification levels 3 and 4, 2000–02 and 2006–08
Table 5.13:	Summary of guidance recommendations by most frequent individual test orders for T2D, 2000–08

Table 5.14:	Summary of support for GP pathology ordering for the most frequent individual test orders for T2D, 2000–08	122
Table 5.15:	Rate of pathology ordering for T2D by level of support, 2000–08, 2000–02 and 2006–08	123
Table 5.16:	Summary of lipid disorders data set, 2000–08, 2000–02 and 2006–08	132
Table 5.17:	Summary of pathology ordering for lipid disorder, 2000–08, 2000–02 and 2006–08	133
Table 5.18:	Rate of pathology test orders for lipid disorder by MBS pathology groups and most frequent individual test orders within each group, 2000–08	134
Table 5.19:	Rate of pathology test orders for lipid disorder by MBS pathology groups and most frequent individual test orders within each group, 2000–02 compared with 2006–08	135
Table 5.20:	Summary of guidance recommendations by most frequent individual test orders for lipid disorder, 2000–08	140
Table 5.21:	Summary of support for GP pathology ordering for the most frequent individual test orders for lipid disorder, 2000–08	141
Table 5.22:	Rate of pathology ordering for lipid disorder by level of support, 2000–08, 2000–02 and 2006–08	142
Table 5.23:	Summary of health check data set in patients aged 15+ years, 2000–08, 2000–02 and 2006–08	152
Table 5.24:	Summary of pathology ordering for health check (patients 15+ years). 2000–08, 2000–02 and 2006–08	, 153
Table 5.25:	Rate of pathology test orders for health check (patients 15+ years) by MBS groups and most frequent individual tests within each group, 2000–08	154
Table 5.26:	Rate of pathology test orders for health check (patients 15+ years) by MBS groups and most frequent individual tests within each group, 2000–02 compared with 2006–08	155
Table 5.27:	Summary of guidance recommendations by most frequent individual test orders for health check, 2000–08	160
Table 5.28:	Summary of support for GP pathology ordering for the most frequent individual test orders for health check (patients aged 15+ years), 2000–08	161
Table 5.29:	Rate of pathology ordering for health check by level of support, 2000–08, 2000–02 and 2006–08	162
Table 5.30:	Summary of weakness/tiredness data set, 2000–08, 2000–02 and 2006–08	170
Table 5.31:	Summary of pathology ordering for weakness/tiredness, 2000–08, 2000–02 and 2006–08	171
Table 5.32:	Rate of pathology test orders for weakness/tiredness by MBS groups and most frequent individual tests within each group, 2000–08	172

Table 5.33:	Rate of pathology test orders for weakness/tiredness by MBS groups and most frequent individual tests within each group, 2000–02 compared with 2006–08	174
Table 5.34:	Summary of guideline/guidance recommendations by most frequent individual test orders for weakness/ tiredness, 2000–08	180
Table 5.35:	Summary of support for GP pathology ordering for the most f requent individual test orders for weakness/tiredness, 2000–08	181
Table 5.36:	Rate of pathology ordering for weakness/tiredness by level of support, 2000–08, 2000–02 and 2006–08	182
Table 5.37:	Summary of overweight/obesity data set (adult patients), 2000–08, 2000–02 and 2006–08	189
Table 5.38:	Summary of pathology ordering for overweight/obesity (adult patients), 2000–08, 2000–02 and 2006–08	191
Table 5.39:	Rate of pathology test orders for overweight/obesity by MBS pathology groups and most frequent individual test orders within each group, 2000–08	192
Table 5.40:	Rate of pathology test orders for overweight/obesity by MBS pathology and most frequent individual test orders within each group, 2000–02 compared with 2006–08	ogy 193
Table 5.41:	Summary of guidance recommendations by most frequent individual test orders for overweight/obesity, 2000–08	198
Table 5.42:	Summary of support for GP pathology ordering for the most frequent individual test orders for overweight/obesity, 2000–08	t 199
Table 5.43:	Rate of pathology ordering for overweight/obesity by level of support, 2000–08, 2000–02 and 2006–08	200
Table 6.1:	Characteristics of patients in the FBC and lipid study sample	225
Table 6.2:	FBC testing in general practice patients	228
Table 6.3:	Time since previous FBC test when ordered for monitoring versus other purposes	229
Table 6.4:	Most common problems for which GPs ordered a FBC test	230
Table 6.5:	Most common differential diagnoses for investigative FBC tests	231
Table 6.6:	Purpose of ordering FBC tests for selected problems	233
Table 6.7:	Lipid testing in general practice patients	237
Table 6.8:	Time since previous lipid test when ordered for monitoring versus other purposes	238
Table 6.9:	Most common problems managed with a lipid test	239
Table 6.10:	Purpose of ordering lipid tests for selected problems	240
Table 7.1:	Description of pathology ordering per 100 encounters with GPs	260
Table 7.2:	Univariate analysis of GP pathology ordering	262
Table 7.3:	Final multivariate model of independent predictors of GP pathology ordering	267
Table 8.1:	Overview of pathology ordering data from the BEACH study	
	2005–06 and 2009–10	286

Table 8.2:	Total number of pathology tests/batteries of tests ordered by GPs for	
	the Australian population	. 287
Table 8.3:	Number of GP contacts involving the ordering of at least one pathology test/battery of tests in the Australian population	. 287

List of figures

Figure 3.1:	The BEACH relational database
Figure 3.2:	The structure of the International Classification of Primary Care, Version 2 (ICPC-2)
Figure 4.1:	Rate of pathology test/battery orders per 100 encounters and per 100 problems managed, 2000–01 to 2007–08 (95% confidence intervals) 47
Figure 4.2:	Distribution of volume of pathology ordered across patient age groups, 2000–08, 2000–02 and 2006–08 (95% confidence intervals)
Figure 4.3:	Patient age-specific rate of pathology test orders per 100 encounters, 2000–08, 2000–02 and 2006–08 (95% confidence intervals)
Figure 4.4:	Distribution of the number of pathology tests/batteries ordered per encounter when at least one test was ordered, 2000–08, 2000–02 and 2006–08 (95% confidence intervals)
Figure 5.1:	Age-specific rate of management of overweight/obesity (2000–08) and age-specific prevalence of overweight/obesity (2007–08) in adult
Figure 6 1.	Patient age specific rate of FBC orders (95% confidence intervals) 226
Figure 6.2	Patient age-specific rate of lipid orders (95% confidence intervals) 220
Figure 8.1:	Age-sex-specific average number of pathology tests/batteries of tests ordered by GPs per person in the Australian population in 2005–06 (95% confidence intervals)
Figure 8.2:	Age-sex–specific average number of pathology tests/batteries of tests ordered by GPs per person in the Australian population in 2009–10 (95% confidence intervals)
Figure 8.3:	Age-sex–specific average number of GP contacts involving at least one pathology test/battery of tests per person in the Australian population in 2005–06 (95% confidence intervals)
Figure 8.4:	Age-sex–specific average number of GP contacts involving at least one pathology test/battery of tests per person in the Australian population in 2009–10 (95% confidence intervals)
Figure 8.5:	Estimated number (millions) of pathology tests/batteries of tests ordered by GPs for the projected Australian population 2015 to 2050 based on per head GP service use and GPs' pathology ordering in 2005–06 and 2009–10 (95% confidence intervals)
Figure 8.6:	Estimated number (millions) of GP contacts involving at least one pathology test/battery of tests for the projected Australian population 2015 to 2050 based on per head GP service use and GPs' pathology ordering in 2005–06 and 2009–10 (95% confidence intervals)

List of boxes

Box 3.1:	Encounter data elements	26
Box 3.2:	Patient data elements	26
Box 3.3:	Data elements about the problems managed	27
Box 3.4:	Data elements about pathology tests and other management of problems at the encounter	28
Box 3.5:	Data elements about the participating GPs and their practice	29
Box 5.1:	Guidance documents reviewed for hypertension	95
Box 5.2:	Guidance documents reviewed for T2D	117
Box 5.3:	Guidance documents reviewed for lipid disorders	136
Box 5.4:	Guidance documents reviewed for health checks	156
Box 5.5:	Guidelines and guidance documents reviewed for weakness/	175
Dox 5 6.	Guidenes documents reviewed for overweight/obesity	104
DOX 3.0:	Guidance documents reviewed for overweight/obesity	194
Box 6.1:	Example of SAND form completed by participating GPs (with instructions)	224
Box 7.1:	Variables included in univariate and multivariate analysis	258
Box 8.1:	Summary of source data	279
Box 8.2:	Australian projected population 2015 to 2050 (millions)	280
Box 8.3:	Calculation example of GP ordered pathology (number of tests/bat and GP contacts involving testing) for females aged 40–44 years	tteries 285

Abbreviations

AACE	American Association of Clinical Endocrinologists
ABS	Australian Bureau of Statistics
ACE	angiotensin converting enzymes
ACRRM	Australian College of Rural and Remote Medicine
ADA	American Diabetes Association
AMTS	Australian Morbidity and Treatment Survey
ASGC	Australian Standard Geographical Classification
ATC	Anatomical Therapeutic Chemical [classification]
BEACH	Bettering the Evaluation And Care of Health
BMI	body mass index
bpac	Best practice advocacy centre
CDA	Canadian Diabetes Association
CHEP	Canadian Hypertension Education Program
CI	confidence interval [in this thesis 95% CI is used]
СК	creatine kinase
CRP	C reactive protein
CSANZ	Cardiac Society of Australia and New Zealand
CTFPHC	Canadian Task Force on Preventive Health Care
DA	Diabetes Australia
DAGDC	Diabetes Australia Guideline Development Consortium
DCGP	Dutch College of General Practitioners
DoHA	[Australian Government] Department of Health and Ageing
DVA	[Australian Government] Department of Veterans' Affairs
ESC	European Society of Cardiology
ESH	European Society of Hypertension
ESR	erythrocyte sedimentation rate
eTG	electronic Therapeutic Guidelines
EUC	electrolytes, urea and creatinine

FBC	full blood count
FMRC	Family Medicine Research Centre
FOBT	faecal occult blood test
FRACGP	Fellowship of the Royal Australian College of General Practitioners
FTE	full-time equivalent
GP	general practitioner
HbA1c	haemoglobin, type A1c
HDL	high density lipoprotein
HIV	human immunodeficiency virus
ICPC-2	International Classification of Primary Care (Version 2)
ICPC-2 PLUS	a clinical terminology classified according to ICPC-2
ICSI	Institute for Clinical Systems Improvement
IDF	International Diabetes Federation
IHD	ischaemic heart disease
ISH	International Society of Hypertension
JNC 7	Joint National Committee [seventh report of the committee]
LCL	lower confidence limit [in this thesis, this refers to the 95% LCL]
LDL	low density lipoprotein
LFT	liver function test
M,C&S	microscopy, culture and sensitivity
MBA	multibiochemical analysis
MBS	Medicare Benefits Schedule
MoH	[Singapore] Ministry of Health
MoU	Memorandum of Understanding
N/A	not applicable
NCEP	National Cholesterol Education Program
NEC	not elsewhere classified
NHF	National Heart Foundation
NHLBI	National Heart, Lung, and Blood Institute
NHMRC	National Health and Medical Research Council
NICE	National Institute for Health and Clinical Excellence

NICE NHS	NICE guideline designed for the National Health Service
NLASSA	National Lipid Association Statin Safety Assessment
NOF	National Obesity Forum
NOS	not otherwise specified
OCCPG	Obesity Canada Clinical Practice Guidelines
OECD	Organization for Economic Co-operation and Development
Pap smear	Papanicolaou smear
PBS	Pharmaceutical Benefits Scheme
PSA	prostate specific antigen
PST	Pathology Services Table [Category 6 of the MBS]
RACGP	Royal Australian College of General Practitioners
RCPA	Royal College of Pathologists of Australasia
SAND	Supplementary Analysis of Nominated Data
SAS	Statistical Analysis System
SIGN	Scottish Intercollegiate Guidelines Network
STI	sexually transmitted infection
T2D	Type 2 diabetes
TFT	thyroid function test
UCL	upper confidence limit [in this thesis, this refers to the 95% UCL]
UK	United Kingdom
US	United States [of America]
USPSTF	United States Preventive Services Task Force
USyd	University of Sydney
WHO	World Health Organization

Symbols

	indicates that the term is defined in the Glossary
^/₩	indicates a statistically significant linear change
\wedge/ψ	indicates a marginally significant linear change
	indicates no change
<	less than
>	more than
2	greater than or equal to
\leq	less than or equal to
n	number
	indicates that the table category/data item does not apply

Glossary

Throughout this thesis terms that are defined in the glossary are marked with the symbol '‡'. Terms are only marked the first time they are used in the text.

- Accreditation: Indicates that the practice is accredited against the Royal Australian College for General Practitioner's 'Standards for general practices'.
- Activity level: The number of MBS GP consultation service items claimed during the previous 3 months by a participating GP.
- **Analyte:** A chemical substance (in a fluid or other specimen from the body) that a laboratory test aims to detect. A single pathology test or battery of tests may include multiple analytes.
- **Battery of tests:** A standard group of related tests (sometimes called a 'test profile') that are ordered together and referred to as a group. For example, a 'full blood count' is a battery of tests that examine different parts of the blood, and often includes: white blood cell count, red blood cell count, haemoglobin, and platelets.
- **Diagnosis/problem:** A statement of the provider's understanding of a health problem presented by a patient, family or community. GPs are instructed to record at the most specific level possible from the information available at the time. It may be limited to the level of symptoms.
 - **New problem:** The first presentation of a problem, including the first presentation of a recurrence of a previously resolved problem, but excluding the presentation of a problem first assessed by another provider.
 - Old problem: A previously assessed problem that requires ongoing care, including follow-up for a problem or an initial presentation of a problem previously assessed by another provider.

Encounter: Any professional interchange between a patient and a GP.

• **Indirect:** Encounter where there is no face-to-face meeting between the patient and the GP but a service is provided (for example, prescription, referral).

- **Direct:** Encounter where there is a face-to-face meeting of the patient and the GP.
- **Episode coning rule:** The MBS payment rule that restricts (to a maximum of three per ordering episode) the number of MBS pathology items that can be claimed by the pathologist for pathology tests ordered by GPs for non-hospitalised patients. Payment is made for the three items with the highest rebate amount.
- **General practitioner (GP):** A medical practitioner who provides primary, comprehensive and continuing care to patients and their families within the community.
- **Health concession card:** Patients holding a Health care/benefit card and/or a Repatriation health card.
- Health care/benefit card: A card entitling the holder to a higher level of Government subsidy for health services (for example, reduced-cost medicines under the Pharmaceutical Benefits Scheme). Examples of patients who may be eligible include pensioners, unemployed, low-income earners.
- **ICPC-2 chapters:** The main divisions within ICPC-2. There are 17 chapters primarily representing the body systems.
- **Iso-resource group:** A group of pathology tests that each use a similar amount of resources. The reimbursement and structure of some MBS pathology items are based on iso-resource groups of tests.
- Medicare Benefits Schedule (MBS) item: Each item number identifies a service funded through Medicare. The MBS lists all the Medicare services subsidised by the Australian Government, their schedule fees and conditions for use.
 - MBS GP consultation service items: Includes GP services provided under the MBS professional services category including MBS items classed as A1, A2, A5, A6, A7, A14, A17, A18, A19, A20, A22 and selected items provided by GPs classified in A11, A15 and A27.
 - Level A items: MBS item numbers 3, 4, 13, 19, 20. A 'Level A' item will be used for obvious and straightforward cases and this should be reflected in the practitioner's records. In this context, the practitioner should undertake the necessary examination of the affected part if required, and note the action taken.

- Level B items: MBS item numbers 23, 24, 25, 33, 35. A 'Level B' item will be used for a consultation lasting less than 20 minutes for cases that are not obvious or straightforward in relation to one or more health related issues. The medical practitioner may undertake all or some of the following tasks (as clinically relevant): taking a detailed history, a clinical examination, arranging any necessary investigations, implementing a management plan, and providing appropriate care.
- Level C items: MBS item numbers 36, 37, 38, 40, 43. A 'Level C' item will be used for a consultation lasting at least 20 minutes for cases in relation to one or more health related issues. The medical practitioner may undertake all or some of the following tasks (as clinically relevant): taking a detailed history, a clinical examination, arranging any necessary investigations, implementing a management plan, and providing appropriate care.
- Level D items: MBS item numbers 44, 47, 48, 50, 51. A 'Level D' item will be used for a consultation lasting at least 40 minutes for cases in relation to one or more health related issues. The medical practitioner may undertake all or some of the following tasks (as clinically relevant): taking a detailed history, a clinical examination, arranging any necessary investigations, implementing a management plan and providing appropriate care.
- **MBS pathology item:** Each pathology item number identifies a pathology service that is funded through Medicare. The MBS pathology items are listed in the Pathology Services Table of the MBS.
- **Pathology ordering:** The ordering of pathology tests or batteries of tests by GPs or other clinicians. Primarily in this thesis it refers to GPs' ordering of pathology tests or batteries of tests.

Patient status: The status of the patient to the practice.

- New patient: The patient has not been seen before in the practice.
- **Old patient:** The patient has attended the practice before.

Problem managed: See Diagnosis/problem.

- **Problem–pathology link:** The link between the pathology test (or battery) and the problem under management at the encounter. Each test must be linked to at least one problem, and up to four problems per encounter. There are more problem–pathology links than numbers of tests recorded at encounters.
- **Provider:** A person to whom a patient has access when contacting the health care system.

Recognised GP: A medical practitioner who is:

- vocationally recognised under Section 3F of the Health Insurance Act, or
- a holder of the Fellowship of the Royal Australian College of General Practitioners who participates in, and meets the requirements for, quality improvement and continuing medical education as defined in the Royal Australian College of General Practitioner's Quality Improvement & Continuing Professional Development Program, *or*
- undertaking an approved placement in general practice as part of a training
 program for general practice leading to the award of the Fellowship of the
 Royal Australian College of General Practitioners, or undertaking an
 approved placement in general practice as part of some other training
 program recognised by the Royal Australian College of General
 Practitioners as being of equivalent standard.
- **Repatriation health card:** An entitlement card provided by the Department of Veterans' Affairs that entitles the holder to access a range of Repatriation health care benefits, including access to prescription and other medications under the Pharmaceutical Benefits Scheme.

Rubric: The title of an individual code in ICPC-2.

Significant: This term is used to refer to a statistically significant result. Statistical significance is measured at the 95% confidence level in this report.

1 Aims and candidate's contribution

1.1 Aims

This thesis aims to assess general practitioners' (GPs) pathology test ordering and the changes in their ordering patterns; evaluate the quality of this ordering in terms of the variance among GPs, and its alignment with guidelines; and predict future ordering for an increasing and ageing population. More specifically, the aims are:

- to describe GPs' pathology test ordering, and growth over time in this ordering, in terms of the types of tests ordered, and the types of problems for which tests were ordered.
- to identify the tests and problems that accounted for high growth in the volume of GPs' pathology ordering over time, and the factors that contributed to this growth.
- to determine the appropriateness of GPs' pathology test ordering for selected problems in terms of its alignment with recommendations for pathology testing made in guidelines and other sources of guidance.
- to determine the variance among GPs in their pathology test ordering rates and identify factors that may explain this variance.
- to estimate the growth in volume of GPs' pathology ordering to 2050, as a result of the projected growth and ageing of Australia's population.

1.2 Candidate's contribution

The candidate was fully involved in all aspects of the research reported in this thesis, including the conceptualisation, design, planning and conduct of the research. The candidate undertook an extensive literature review involving selection of databases, libraries and other sources of information including published guidelines and guidance documents.

In this thesis I present a series of studies, all of which utilise data collected in the BEACH (Bettering the Evaluation and Care of Health) program. The candidate's involvement in the BEACH program, at the Family Medicine Research Centre (FMRC), University of Sydney, is described below.

The study measuring the appropriateness of GPs' pathology test ordering (reported in Chapter 5) was funded through a competitive grant from the Quality Use of Pathology Program, Australian Government Department of Health and Ageing. The candidate assumed full responsibility for conduct of the study, research design (including selection of problems for investigation, identification and review of guidance documents) and preparation of reports. Professors Helena Britt and Graeme Miller prepared the initial funding application, nominating the candidate as the responsible researcher.

The candidate conceptualised, designed and initiated the SAND (Supplementary Analysis of Nominated Data) study (reported in Chapter 6). This included conceptualising and designing the questionnaire form and instructions, and obtaining ethics approval for the study from the Ethics Committee of the Australian Institute of Health and Welfare, and the Human Research Ethics Committee of the University of Sydney. The candidate prepared the specifications for the electronic data entry form which was implemented by the Centre's IT manager, Mr Tim Chambers. The candidate also oversaw the data entry undertaken by trained casual staff.

Data cleaning and checking were performed by the candidate and assisted by Ms Lisa Valenti, under instruction from the candidate. Statistical analyses for the results reported in Chapters 4 to 8 were specified by the candidate with the assistance of her supervisors Professors Helena Britt and Graeme Miller. The candidate independently performed the analyses for the extrapolations presented in Chapter 4. Under the candidate's close instruction, senior analysts in the research team at the FMRC undertook the remaining analyses: Ms Lisa Valenti conducted the analyses for Chapters 4 to 7; and Mr Christopher Harrison conducted the analysis for Chapter 8.

The preparation and creation of this manuscript, the literature review, reporting and interpretation of results, the discussion and conclusions made, are solely the work of the candidate.

A list of publications and presentations arising from this thesis is provided in Appendix 1. Two publications of which the candidate is lead author have emanated from this work. The research presented in Chapter 4 was published as a peerreviewed chapter in the report *General practice in Australia, health priorities and policies 1998–2008.*¹ The final report of the work conducted for the Quality Use of

2

Pathology Program grant (reported in Chapter 5) was published by the Australian Government Department of Health and Ageing online: *Evidence-practice gap in GP pathology test ordering: a comparison of BEACH pathology data and recommended testing.*² As a result of this work, the Australian Government Department of Health and Ageing invited the candidate to become a member of the (pathology) Demand Management Advisory Committee (established in late 2011). Her involvement with this committee is ongoing.

Candidate's involvement in the BEACH study

The research reported in this thesis uses data collected in the BEACH program, a continuous national study of general practice activity. BEACH commenced in April 1998 and in April 2012 started its fifteenth year of data collection.

The BEACH study was established two years prior to the candidate joining the research team at the FMRC. The candidate was initially employed at the FMRC in 2000 while an undergraduate student as a casual data entry staff member. She completed her Honours thesis using BEACH data (Male consultations in general practice³) in 2001, and has been employed on a full-time basis since this time. Over the years the candidate has been involved in every aspect of the BEACH study including recruitment, data entry, training data entry staff, data checking, form design, planning and overseeing analysis for specific topics, producing and editing reports, preparing papers and presenting results at conferences.

The candidate's current primary work role is to update and design the BEACH recording forms. This includes annual updates of the standard BEACH data collection forms as required, co-ordination and design of the SAND substudies (in collaboration with external stakeholders), submission of forms for ethics approval, and preparing the forms for printing. The candidate is also responsible for co-ordinating the publication of the General Practice Series BEACH books, involving timeline management, editing and liaising with the publisher.

2 Introduction

Pathology is defined as "the branch of medicine that studies the essential nature of disease, especially the structural, biochemical, and functional changes in the cells, tissues and organs of the body that cause or are caused by disease".⁴ Pathology tests are critical in modern medical practice, and are essential for disease prevention, diagnosis and monitoring.

In Australia, the use and cost of pathology testing has increased over past decades across multiple health sectors.^{5,6} A similar pattern has occurred in most developed countries, despite differences in how the various health systems are managed and pathology services funded.⁷⁻¹²

This growth has raised concerns, in Australia and internationally, about what is driving the increased use of pathology testing, the viability of the increased costs, and whether this expenditure represents appropriate health care spending.⁶⁻⁹ Despite these concerns being expressed and investigated over decades, the use and cost of pathology tests have continued to increase.

In Australia, over the decade 2000 to 2010, the cost of pathology services funded by the Australian Government through the national insurance scheme, Medicare, increased by 78.0%: from \$1.2 billion in the 2000–01 financial year to \$2.0 billion in $2009-10.^{5}$ Similarly the volume of pathology services increased, from 62.1 million claimed (3.2 per capita) in 2000–01 to 103.7 million (4.7 per capita) in 2009–10, a 67.0% increase in the number of claimed services and a 46.9% increase in the number of services per capita.^{5,13}

These pathology services represent those ordered by general practitioners[‡] (GPs) and other medical specialists for non-hospitalised patients, and 68–70% of Medicare pathology outlays over the decade (E Wilson, personal communication, March 2011) were generated by GPs' pathology ordering[‡]. The focus of this thesis is pathology ordering by GPs, as it accounts for the majority of the Government-funded pathology services in Australia.

The structure of the Australian health care system is complex, as are the arrangements for Government funding of pathology services within the system. An

understanding of both is required to understand the context of the research presented in this thesis.

A brief overview of the Australian health care system

Governance of the Australian health care system is shared between the Federal, and the State and Territory Governments. The health care system is funded through a mixture of public and private sources including governments, health insurers and individual Australians.

The Federal Government is responsible for national health policy, and controlling and managing the national health insurance scheme, Medicare. Through Medicare, the Australian population have access to free or subsidised treatment provided by a variety of health professionals. The Medicare Benefits Schedule (MBS) lists the health services for which Medicare benefits are available, the rebates for these services and the conditions for use. The health care system provides free treatment to Australian residents using public hospitals, and free or subsidised treatment by GPs and other medical specialists. Individual clinicians determine whether or not their services are charged directly to Government at the Medicare rebate level and are free to patients (referred to as bulk-billing). When not bulk-billed, the patient pays the fee set by the provider[‡] and seeks reimbursement of the appropriate Medicare rebate, with the patient covering any difference between the fee and rebate. The Federal Government is also responsible for the Pharmaceutical Benefits Scheme (PBS), which subsidises the cost of prescribed medications. Private health insurance is optional in Australia, and its cost to an individual is subsidised through taxation rebates provided by the Federal Government. Private health insurance does not cover services provided by GPs or the pathology tests ordered by GPs, when such services/tests are covered by Medicare.

As set out in the Australian Health Care Agreements, Federal and State/Territory Governments jointly fund public hospitals. The State and Territory Governments are responsible for the management of public hospitals and community care services. Australia's health expenditure in 2009–10 was \$121.4 billion, and accounted for 9.4% of gross domestic product. Governments funded 70% of this expenditure and other non-government sources including patients, funded the remaining 30%.¹⁴ In June 2010, the population of Australia was estimated to be 22.3 million people.¹⁵

General practitioners

GPs are usually the first point of contact in the Australian health care system. In 2009–10, about 83% of the Australian population had at least one GP visit funded by Medicare (Department of Health and Ageing [DoHA], personal communication, June 2010). Payment for GP visits is largely on a fee-for-service basis, there being no compulsory patient lists or registration. People are free to see multiple GPs and visit multiple practices of their choice.

GPs have a gatekeeper role in the health care system. GP 'referral' is needed in order for patients to access Medicare benefits for many health services including pathology tests, imaging tests and medical specialist care. GP referral is the most common way by which the population accesses Medicare-subsidised pathology services. Patients may also access these services through medical specialist referral. Similarly access to PBS-subsidised medications requires a prescription from a medical practitioner.

In Australia, there were 24,029 practising primary care practitioners (vocationally recognised GPs^{\ddagger} and other medical practitioners) in 2008, making up 23,188 full-time equivalents (based on a 40 hour week), or 107.9 per 100,000 people.¹⁶

The vast majority of GP services are funded through Medicare.¹⁷ In 2009–10 there were 116.6 million Medicare-funded GP encounters, an average of 5.2 per person in Australia, and the majority (79.5%) were bulk-billed.^{5,13} The Medicare cost for these GP encounters was about \$4.9 billion.⁵

The number of Medicare-funded GP services has increased over the decade 2000 to 2010, from 100.6 million encounters in 2000–01 to 116.6 million in 2009–10.⁵ The number of services per person and the proportion of services that were bulk-billed were similar in 2000 and 2010. However there was a decline in both during the mid 2000's: the number of GP visits per person fell to a low of 4.8 per capita in 2003–04 compared with 5.2 in 2000 and 2010;¹³ and the proportion of services bulk-billed fell to 68.2% in 2003–04 compared with 78–80% in 2000 and 2010.⁵ These falls prompted concerns about patient access to care and in response the Government introduced several 'Strengthening Medicare' initiatives in 2004¹⁸ which were successful in increasing access to GP services.

Pathology funding arrangements

The pathology sector comprises private and public laboratories. Public laboratories are primarily based in public hospitals and private laboratories are based in the

community. The private sector is dominated by a few large private pathology companies. The way pathology services are funded and the type of laboratory conducting the testing varies based on the health setting in which pathology tests are ordered and the type of clinician ordering the test.

Pathology ordered for patients of public hospitals is conducted in public laboratories and funded under the Australian Health Care Agreements. Pathology ordered by GPs and other medical specialists for non-hospitalised patients is funded through Medicare as set out in the MBS Pathology Services Table (PST), and mostly undertaken in laboratories owned by private pathology companies. The PST outlines the tests that are funded through MBS pathology items[‡], their rebates and conditions of use.

All MBS pathology services must be ordered (or 'referred') by GPs or other medical specialists. Pathologists claim for MBS pathology items for tests ordered on a feefor-service basis and can elect whether to bulk-bill patients for these services. A high proportion of pathology services are bulk-billed (i.e. free to the patient). Across the period 2000 to 2010, an average of 85.0% of MBS pathology services were bulk-billed.

While bulk-billing has remained high, the average rebate for MBS pathology items has decreased. In real terms the average rebate per MBS pathology item in 2008–09 was below what it was when Medicare began in 1984.¹⁹ This has been achieved through amalgamation and centralisation of pathology laboratories in Australia and increased automation of testing. Therefore, while the total number and cost of Medicare-funded tests continues to increase, the proportion of total Medicare benefits accounted for by pathology services is decreasing. In 2000–01 pathology services accounted for 15.8% of Medicare benefits and in 2009–10 they accounted for 13.0%.⁵

In funding MBS pathology services the Australian Government acts as the sole purchaser, setting the regulatory framework, and managing the PST.²⁰ These structures and rules ensure high quality pathology services, and facilitate management of outlays. To some extent they have evolved in response to allegations during the 1980's and 1990's of fraud (such as inducements and 'kickbacks') and overservicing in Australia.²¹

7

The vast majority of pathology tests ordered by GPs are included in the MBS and funded through Medicare. However, there are some exclusions. For example, point of care tests, while available to GPs, are not currently funded by Medicare. In part this is due to poor cost-effectiveness.²²

This thesis is concerned with tests ordered by GPs that are funded through Medicare. Theoretically the data held by the Federal Government of the MBS pathology items claimed should provide a picture of the ordering behaviour of GPs. However, there are MBS structure and payment rules that mean the Medicare data cannot provide an exact reflection of pathology requested by GPs. The two main factors are the episode coning rule[‡] and the structure of some MBS pathology items.

The episode coning rule restricts the number of MBS pathology item numbers that can be claimed by the pathologist per episode of ordering for pathology tests requested by GPs for non-hospitalised patients. This rule means that a maximum of three MBS pathology items can be claimed per episode of ordering, payment being made for the three items with the highest rebate amount (i.e. highest cost). Some MBS pathology item numbers are exempt from the coning rule, such as Papanicolaou (Pap) smear items.²³

Each MBS pathology item number can either represent a single pathology test or multiple pathology tests. Item numbers that include multiple tests usually reflect iso-resource[‡] groupings, and for some of these items, the amount paid for such items is based on the number of tests ordered from within the group (referred to as ladder items). For example, one test from the group, two tests from the group, and so on, to a maximum of five tests from the group.²³

In addition to the MBS payment rules outlined in the PST, outlays for pathology services were capped through a series of Memoranda of Understanding (MoU) (up to 2009) between the Australian Government and the pathology industry and profession. The first was introduced in 1996, and in 2008, at the commencement of the research for this thesis, the third MoU was in place. It covered 1 July 2004 to 30 June 2009, and aimed to contain growth in outlays to an average of 5.3% per annum.²⁴

At the completion of the third MoU, the MBS pathology funding arrangements were reviewed.²⁰ Following this review, the MBS funding arrangements remain largely unchanged, and a new 5-year funding agreement was introduced in July 2011.²⁵ The

review was partially prompted by the 2008 audit of the MoU, undertaken to investigate why the financial objectives of the MoU were not being met. Over the 2004–08 period, the annual growth in pathology outlays covered by the MoU was 7%, rather than the projected 5%, following the Government's introduction of national policy initiatives that produced an unanticipated increase in demand for pathology services. In addition to policy influences, the audit reported several factors that were thought to be associated with increased demand for pathology testing. The audit revealed that there was limited Government understanding of the drivers of demand and recommended that this be improved as it is essential for management of future pathology outlays.²⁶

Why is pathology testing increasing?

Over the years, the increases in pathology testing have been attributed to numerous factors, including: the ageing population;^{9,26,27} defensive testing due to practitioner concerns about medicolegal implications of not testing;^{9,26,28} increasing patient demand for testing;^{26,28,29} increased disease awareness (e.g. due to educational campaigns) and disease prevalence, leading to increased diagnostic and/or management rates;²⁷ changes in the management of established disease;^{9,27} increased availability of tests;^{9,26} improvements in testing technologies;²⁹ increased use of computers for ordering tests;^{29,30} and policy initiatives.^{26,27,29}

In Australia, policy initiatives that have been associated with increases include those that aim to improve patient access to medical services (such as the Strengthening Medicare initiatives), improve provision of preventive care, and facilitate better management of diseases (such as MBS items[‡] for health assessments and chronic disease management).^{26,27} Similar influence of policy initiatives on test ordering is seen internationally, for example the introduction of Quality and Outcomes Framework indicators that incorporated pathology test indicators in the United Kingdom (UK).²⁹

Another factor linked to increased use of pathology services is a lack of economic signal. In Australia, due to the structure of MBS pathology funding and the high rate of bulk-billing, the patient and the referring clinician do not pay for, and are unaware of, the costs of pathology tests. This funding situation is called the 'third party payer problem' and has been referred to as a "fundamental market flaw".³¹

These factors are used to explain increases in pathology ordering by all clinicians, including that by GPs. The increased volume of pathology tests generated by GPs can be attributed to these factors working on the mechanisms of increasing GP workload, and/or changes in GPs' pathology ordering behaviour. The latter can be characterised by changes in the GPs' decision to initiate ordering, and the type and/or number of pathology tests ordered when the decision to order is made.

The relationship between these factors and the increases in GPs' ordering of pathology tests is not always proven. For example, while patient demand for testing is high (14–39% expect pathology or imaging tests when they visit the GP³²⁻³⁵) and GPs report this as a reason for ordering pathology,^{36,37} there is no evidence that patient demand has increased. Similarly the increased use of computers by GPs was thought to increase the ease with which pathology tests could be ordered and therefore contribute to increased GP demand.³⁰ However, two studies found no independent effect of computer use on GPs' pathology ordering.^{30,38} In contrast, the association of other factors is much stronger. For example the effect of Australia's ageing population on increased demand for pathology tests can be established, as the use of pathology services increases with age.²⁷ Similarly some national policy initiatives were determined to have increased the demand for pathology testing, and this resulted in adjustments to the third MoU.²⁶

With the numerous factors linked to increased demand for pathology testing, and the likely overlap between them,³⁹ it is no wonder that understanding what is driving the increased demand for pathology testing has proved difficult. Further, it is reasonable to assume that the factors associated with increased demand would not equally affect all circumstances in which pathology tests were ordered.

Notably absent in the literature that discusses increases in pathology testing, is information about the tests and clinical problems for which GPs' pathology testing has increased.

Why are pathology tests ordered?

Two recent reviews investigated the reasons tests are ordered by clinicians.^{40,41} Both reviews took a very focussed approach: Whiting et al. concentrated on reasons tests were ordered for patients with undiagnosed symptoms,⁴⁰ and Sood et al. only on variables that were related to the physician and were 'non-evidence-based'.⁴¹ The focus taken by these reviews largely excluded the accepted clinical reasons for

pathology test ordering: to diagnose, monitor, assess prognosis, assess severity, or to screen for diseases.^{8,30,37,42}

Further, the extent to which results of the reviews can be generalised to describe the reasons that GPs' order pathology tests is limited due to their narrow focus and the heterogeneity of the reviewed studies. Differences between the studies include: the types of tests investigated (i.e. a mixture of pathology, imaging and other diagnostic tests); the methodology used (including qualitative, quantitative and mixed methods); and the context of the study (i.e. the health settings, the types of clinicians ordering tests and the specific disease or test investigated).^{40,41}

Whiting et al. identified more than 30 reasons for ordering tests for undiagnosed symptoms and categorised these into five interrelated groups.⁴⁰ Reasons classified as: diagnostic (e.g. to modify pretest probability of disease, rule disease 'in' or 'out'); and therapeutic and prognostic (e.g. decide on appropriate treatment) could be considered evidence-based reasons for testing. However, the same could not be said for many of the: patient-related (e.g. patient preference); doctor-related (e.g. clinical experience and confidence in clinical judgement, speciality); and policy and organisational factors (e.g. practice size, test availability).⁴⁰

In their review, Sood et al. divided 'non-evidence-based' physician variables associated with ordering into those that could be modified and those that could not. Non-modifiable factors included geographic location, practice setting, age and sex, and clinical setting. Modifiable factors included experience and knowledge, belief systems, medicolegal concerns, financial incentives/awareness of cost of testing, and provision of feedback/education.⁴¹ It is somewhat inflammatory to label all of these factors as non-evidence-based. In particular, the provision of feedback and education, which is cited as being very successful in reducing test ordering and the "best-studied modifiable variable", often involves the introduction of evidence through feedback and education. To label it as non-evidence-based is misleading.

Despite the limited applicability of the reviews to general practice, Van der Weijden et al.³⁶ found GPs in a small Dutch qualitative study reported similar reasons for ordering pathology testing in diagnostic uncertainty as those reported by Whiting et al.⁴⁰ for undiagnosed symptoms. Variables most often linked by others to GPs' pathology ordering are those related to the GP and the practice, and would be considered 'non-evidence-based' according to Sood's review. GP variables linked to

testing include age,⁴³⁻⁴⁵ sex,^{43,46-48} geographic location and number of GPs working in the practice.^{43,45-47,49-51}

The bulk of the literature discussing reasons for pathology ordering focuses on the non-clinical reasons for testing, excluding the crucial fact that pathology is ordered in a clinical interaction between a patient and a clinician primarily for a clinical purpose. I believe the focus on non-evidence-based factors has occurred because they can be considered inappropriate and are potentially a modifiable area to target in the context of increasing testing. However, while they exist, a small study by Enno et al. demonstrated that they were not the primary reasons that GPs in Australia ordered pathology tests. The main reasons for ordering 3,419 pathology tests were: to establish a diagnosis (31.6% of 3,840 reasons for ordering), to monitor illness (29.1%), for screening (15.1%), or to monitor drug therapy (11.5%). GPs nominated non-evidence-based patient-related (3.7%) and doctor-related reasons (3.2%) far less frequently.³⁷

Inappropriate testing and outcomes

The question of whether pathology tests are ordered appropriately has become an important issue due to the continuing increase in pathology ordering, and our inability to completely determine what is driving this increase.^{9,42,52-54}

Determining what testing is inappropriate and how much pathology testing is ordered inappropriately is difficult. In their review of 44 studies, Van Walraven and Naylor found that 5-95% of tests were reported as inappropriate, the majority of studies reporting that 10-50% of tests were inappropriate.⁵² They found that the level of evidence in studies evaluating the appropriateness was poor. The studies were heterogeneous in terms of the health setting investigated, the clinicians involved, the clinical problem investigated, and the tests investigated. In addition, the criteria used to judge appropriate test use is the dearth of high-level evidence demonstrating the clinical value of pathology tests and the optimal way in which they should be used.^{39,52,55-57}

Despite the poor evidence, and Van Walraven and Naylor's conclusion that "allusions to extensive inappropriate use should not be made without appropriate qualifiers",⁵² these statements are common within the literature.^{8,9,28,29,58,59} It is reasonable on the basis of the available evidence to say that some pathology ordered in selected circumstances is inappropriate and/or that some tests are ordered inappropriately. But to conclude that inappropriate test ordering is widespread, overstates the available evidence.

Another statement repeated in the literature is that the increase in pathology ordering is due (either totally or partly) to increases in inappropriate pathology ordering.^{12,58} However when you trace the origins of references used to support these statements the evidence is not there or is decades old. It appears that two true separate statements [(i) that pathology ordering is increasing, and (ii) that some of this testing is inappropriate] have merged to become a single unsubstantiated concept.

Some have proposed that the appropriateness of pathology testing can be evaluated by whether it improves the patient's or public's health outcomes.^{60,61} This has given rise to the argument that the increased use of pathology testing has not corresponded with improved health outcomes or health status,^{9,31,61} the inference being that the increased use of testing is unjustified.

This argument is extremely hard to prove or disprove as measuring outcomes associated with pathology ordering is notoriously difficult and studies that evaluate outcomes are rare.^{52,61,62} Partly this is because it is not the use of a pathology test in isolation that affects outcomes, it is the clinician's actions (or lack thereof) in response to test results that affect outcomes. Further, numerous outcome measures can be used.^{61,63} There are examples in the literature where: use of pathology tests has resulted in improved outcomes;^{64,65} reduction in the use of tests has not adversely affected outcomes;⁶⁶⁻⁶⁸ and use of tests have contributed to adverse patient outcomes.⁶⁹ Such studies are usually conducted in confined clinical settings (e.g. patients in hospital) where the association between testing and a specific outcome can be measured.

There is no conclusive evidence to support the broad assertion that the increased use of pathology tests has not improved patient outcomes. The lack of quality evidence to measure outcomes has resulted in the call for an outcomes research agenda for pathology testing.^{42,57} The development of methods to assess such outcomes is a current area of research.⁷⁰

While these issues are important, the extrapolation of flawed evidence to make broad statements about the appropriateness (or lack thereof) of all pathology testing, paints the whole issue in an unnecessarily negative light. These general perceptions are
applied to all pathology ordered in all settings. As GPs in Australia order the majority of Medicare-funded pathology tests, the negative perception is often applied to their ordering behaviour, even though many of the studies of appropriateness and outcomes were not conducted in general practice.

Criteria used to assess the quality of GPs' pathology ordering behaviour often include the appropriateness of their ordering in selected clinical situations (e.g. for a specific disease) and the variance among GPs in their use of pathology tests. The presence of inter-GP variance is well established. Studies have found this variance is only partially explained by GP, practice and patient characteristics.^{43-51,71-73} However, investigation of this variance often excludes consideration of the contribution of the clinical problem (or purpose) for ordering as these data are not routinely available. When included, the clinical problem being managed has been found to explain the largest proportion of variance.^{45,73}

Studies introducing interventions to improve appropriateness of GPs' pathology ordering are used in the literature to demonstrate the presence of inappropriate ordering. Change, usually a reduction, achieved following interventions is used to 'prove' that appropriateness of ordering can be improved. However, these studies rarely assess the appropriateness of these changes. Success is measured purely on test ordering rates without linked clinical data (such as the problem for which testing was ordered). Hence, while the intervention makes recommendations about appropriate pathology ordering in a given clinical situation using clinical data available to GPs at the time of ordering, such clinical data are not available in the assessment of the impact of the intervention. In the absence of such data, a measured change in the ordering rate of the related pathology tests is assumed to be due to improved appropriateness of ordering.

I found two studies that directly assessed appropriateness of change following interventions targeting a limited number of pathology and diagnostic tests.^{74,75} The authors reported that this assessment was laborious and that appropriateness was only improved for selected tests. Both studies were randomised control trials. In one study, Verstappen et al. found that improvement in appropriateness aligned with reduction in use of tests.⁷⁴ This was not the case for Winkens et al.⁷⁵ who found that for one group of tests, GPs' testing rates did not change but appropriateness of ordering improved, and for a second group of tests, GPs' testing decreased

significantly but there was no change in the proportion of tests that were appropriate.⁷⁵ It was hypothesised that this unusual finding was due to a high level of appropriate use of testing in the second group at baseline (prior to the intervention). Some authors acknowledge the limitation of being unable to assess appropriateness of change, but argue that such change can be assumed to be due to improved appropriateness because this is the basis of the intervention.⁷⁶⁻⁷⁹ I believe that this is a fair assumption in cases where testing is known (or likely) to be inappropriate at baseline. Most studies meet this requirement, interventions having been targeted to 'problem' areas of pathology ordering. This approach is logical from the perspective that limited funds are available for interventions, and targeted interventions are likely to be cost-effective. However, in relation to the broader discussion of appropriateness of GP ordering, it is problematic.

First, the need for cost effectiveness means that interventions are almost exclusively targeted to areas where a reduction in testing is anticipated. As summarised by Winkens and Dinant "ideal interventions would improve the rationality of ordering of investigations while at the same time leading to fewer requests being made."⁹ Consequently areas in which improved appropriateness of testing may lead to increases in testing are excluded.

Second, targeting problem areas means that most interventional studies are conducted in very focussed clinical situations targeting a limited number of pathology tests^{58,80-84} or pathology testing for a limited number of clinical problems.^{11,75,77,85-87} While we know that a minority of problems and tests account for the majority of pathology ordered by GPs,^{45,79,82,88,89} this is not the basis for the selection of tests and problems. Therefore these studies do not provide an overall impression of the appropriateness of GPs pathology ordering behaviour.

While these interventional studies can be used to demonstrate that there are areas of inappropriate ordering by GPs, they cannot be used to give an accurate evaluation of the extent of this inappropriate ordering.

Interventions in pathology ordering

Numerous interventions have been used in an attempt to modify clinicians' pathology test ordering behaviour. In her 1991 review, Leese categorised interventions as involving: education; feedback; participation (i.e. involving clinicians in the need for, and development of an intervention); peer review; financial incentives; and administrative changes (i.e. changes that control or direct choices regarding testing, such as policy change).⁹⁰ Others have used similar categories to describe interventions.⁹¹⁻⁹⁴

Reviews of interventional studies found that success of the interventions was limited and the methodological quality of the studies was poor.⁹¹⁻⁹⁴ For example, many studies lacked a control group or used a historical control group when evaluating interventions, and many lacked follow-up to determine whether the effect of interventions was sustained.⁹³ The most recent review was conducted by Solomon et al. in 1998. Of the 49 studies reviewed that involved interventions to improve clinicians use of diagnostic tests, 76% reported favourable change in the clinicians use of tests. Despite the poor methodology and diversity of the studies, Solomon et al. found that those targeting multiple behavioural factors were more successful than those that targeted a single behavioural factor.⁹¹

The extent to which results of these reviews can be applied to the general practice setting is limited due to the heterogeneity of the included studies. They included multiple types of tests (pathology and imaging), clinicians, health settings, and patient groups. Most of the reviewed studies were not conducted in general practice. For example of the 49 studies reviewed by Solomon et al. only 8 were conducted in general practice.

Several studies have reported interventions in general practice. Since 1990 results of 19 intervention strategies have been published.^{11,58,74-77,79-87,95-100} Each 'strategy' often incorporated more than one intervention and some strategies were used in multiple studies. Types of interventions used include: changes to the laboratory order form (either paper-based^{11,80,83,97,99} or computer-based order form^{79,81,100}), education (such as educational materials and meetings),^{11,58,74,75,77,82-85,95,100}

feedback,^{58,74,75,77,82,85,95,96} decision support,^{76,79,86,87} and changes to funding rules or arrangements.^{98,99} Guidelines were the basis for many of these interventions.^{74-77,83,85-87,95}

Almost all of the published studies reported success in achieving all or some of the desired change (most often a decrease) in GPs' pathology ordering rates, which may represent an element of publication bias. Only two reported no success in changing clinician behaviour.^{82,86}

In Australia in the past 15 years, no studies were published that reported the effect of interventions targeting GPs' pathology ordering behaviour. However, the Australian Government Quality Use of Pathology Program (established in 1999) supports quality ordering of pathology services and provision of quality services by funding projects that promote quality use of services.¹⁰¹ Other educational programs (such as the Common Sense Pathology series produced by the Royal College of Pathologists of Australasia [RCPA]¹⁰²) and administrative interventions have also been introduced. To date, most national interventions have been administrative and reflect efforts to contain pathology outlays. These were primarily supply-based controls acting on the pathology industry through the MoUs and the MBS structures and rules.

Few national initiatives have sought to modify demand for pathology services by referring GPs. Those that do have been largely incorporated into MBS rules, including indication and frequency restrictions for selected MBS pathology items.²³ For example, the MBS 'HbA1c test' item is only claimable for patients with established diabetes and can be claimed a maximum of four times within a 12 month period. Additionally, paper-based laboratory order forms are regulated to ensure that no 'tick box' lists of tests are provided, although the high level of GPs' use of clinical software¹⁷ which allows 'tick box' pathology ordering may negate this measure.

The Government's ability to introduce interventions targeting GPs' pathology ordering is hampered by the limitations of the Medicare data in describing the GPs' pathology 'referral' behaviour. The Medicare data do not accurately reflect of the types of tests ordered by GPs due to the structure of the MBS items and the episode coning rule. In 2010, it was estimated that approximately 30% of MBS pathology items generated by GP orders were not eligible for funding due to coning.¹⁹ Further the Medicare data do not include any information about the clinical problems for which pathology tests were ordered. Such data are required to evaluate the appropriateness of GPs' ordering and identify areas in which interventions may be needed. This evaluation step is missing in much of the international literature due to the lack of availability of clinical data (particularly the patient problem under management) related to GPs' pathology ordering. This has led to judgements of appropriateness based on limited evidence, and has hampered investigation of the clinical problems contributing to the increases in GPs' pathology ordering. In Australia, data describing GPs' pathology ordering behaviour are collected in the BEACH (Bettering the Evaluation and Care of Health) program. BEACH is a continuous cross-sectional national study of general practice activity, and is the only national source of data that allows analysis of the relationships between pathology ordered and problems managed[‡] by GPs. Other factors such as the characteristics of the patient and the GP are also available for such analyses. For these reasons the BEACH data are used in this thesis.

Two previous studies investigating GPs' pathology ordering have been conducted using BEACH data.^{45,89} The first, conducted on 9 months of data collected in 1998, provided the first comprehensive national snapshot of GPs' pathology ordering unaffected by the limitations of the Medicare data. It described GP characteristics associated with test ordering, the most common tests ordered, and the clinical problems for which pathology tests were ordered.⁸⁹ The second was conducted using data from 1998 to 2001 and investigated increases in GPs' pathology ordering with a focus on the GP characteristics associated with this increase.⁴⁵ Since publication of these reports, upper level BEACH data summarising GPs' pathology ordering (published annually) have demonstrated statistically significant increases in the likelihood and rate of GPs' pathology ordering in Australia.¹⁰³

The ongoing increases in GPs' pathology ordering evident in the BEACH and Medicare data are placing increasing pressure on Australia's health budget. Many assumptions are made in the literature about why these increases are occurring and the appropriateness of GPs' ordering. There is an underlying leaning that more testing represents poor quality, but with very little supporting evidence that measured increases in testing represent either 'good' or 'bad' change. This is partially due to a lack of clinical data describing the clinical context (e.g. clinical problems) of increases in pathology ordering. Such data are required to evaluate appropriateness of GPs' pathology ordering.

Framework for this thesis

In this thesis I seek to address some of these issues. The overall aims are presented in Chapter 1, and topic-specific aims are included in the chapters in which they are investigated. In Chapter 3, I describe the methods used in this thesis. In Chapter 4, I examine the changes in GPs' pathology ordering patterns over time, and the pathology tests and the clinical problems contributing to the increase. For each of the identified clinical problems, I investigate the contribution of changes in GPs' workload and GPs' pathology ordering behaviour to the growth in pathology ordering. In Chapter 5, I evaluate the appropriateness of GPs' pathology ordering for selected problems based on the extent to which GPs' ordering aligns with pathology test recommendations made in clinical guidance for these problems. In Chapter 6, I investigate the use of the two most commonly ordered pathology test types by Australian GPs, the full blood count and lipid tests. Chapter 7 examines the extent to which the variance in GPs pathology ordering rates can be explained by GP, practice, and patient characteristics, the clinical problem under management, and the type of encounter. Chapter 8 considers the impact of the expected growth and ageing of the Australian population on GPs' future pathology ordering. Discussion and conclusions are presented in Chapters 9 and 10.

3 Method

In this thesis I use data collected in an ongoing research program called the BEACH (Bettering the Evaluation and Care of Health) study. The methods of the BEACH study are described in detail below.

Throughout the thesis, additional methods have been designed and applied, and different data periods used for specific topics—these are described in detail in the method section of each chapter. Broadly the chapters rely on data that were collected at some point between April 2000 and March 2010.

The description of BEACH and its methods provided in this chapter are drawn from previously published BEACH reports. The candidate has been a co-author of these reports since 2002. Each section is referenced to the report(s) from which it was sourced.

3.1 An overview of the BEACH study

The BEACH study is a continuous national study of general practice activity in Australia that commenced in April 1998 and began its fifteenth year of continuous operation in April 2012. The study was conducted by the Australian General Practice Statistics and Classification Centre, a collaboration between the Family Medicine Research Centre (FMRC) of the University of Sydney and the Australian Institute of Health and Welfare. This collaboration ceased in 2011 (after the data used in this thesis were collected), and the study has since been conducted by the FMRC alone. The BEACH study is supported by multiple stakeholders, including the Australian Government DoHA, pharmaceutical companies, not-for-profit organisations and other organisations that require general practice data (recognised in Acknowledgements). The program is overseen by an Advisory Board that is made up of representatives of each stakeholder organisation and representatives of the Royal Australian College of General Practitioners (RACGP), Australian College of Rural and Remote Medicine (ACRRM), Australian Medical Association, Australian General Practice Network and the Consumer Health Forum.¹⁰⁴

Ethics approval

For each year, ethics approval for the BEACH study was obtained from the Human Research Ethics Committee of the University of Sydney. From 1998 to 2011 ethics approval was also obtained from the Ethics Committee of the Australian Institute of Health and Welfare.

Aims of the study

The BEACH study has three primary aims:

- to provide a reliable and valid data-collection process for general practice which is responsive to the ever-changing needs of information users,
- to establish an ongoing database of GP-patient encounter information,
- to assess patient risk factors and health states and the relationship these factors have with health service activity.¹⁰⁴

3.2 The BEACH method

The core BEACH methodology, and data elements collected, did not change over the data years used in this thesis (2000 to 2010). While there have been minor changes over time (such as changes to questions on the GP profile), these do not affect the data elements used in this thesis. The example forms and instructions included as Appendices 2 to 5 reflect those used in the 2007–08 recording period.

BEACH involves an ever-changing random sample of approximately 1,000 GPs per year. The study is paper-based. Each GP records details for 100 doctor–patient encounters[‡] and each GP also completes a questionnaire about themselves and their practice.¹⁰⁴

The BEACH methods were developed from those used in the 1990–91 Australian Morbidity and Treatment Survey (AMTS) undertaken by the Department of General Practice at the University of Sydney.¹⁰⁵

While the BEACH study is ongoing, its methods are described throughout this chapter in past tense to describe how the BEACH data used in this study were collected.

3.2.1 Sample size

Using data from the AMTS, Meza et al. developed sample size models for national general practice surveys.¹⁰⁶ They found that a sample size of 1,000 GPs, each recording data about 100 patient encounters, provided reliable estimates of the most common problems managed and medications prescribed.¹⁰⁶

3.2.2 The sample

In each year, the GP sample frame included all recognised GPs who had claimed at least 375 MBS general practice consultation service items[‡] in the most recently available Medicare data quarter. The use of this cut-off meant that the vast majority of full-time and part-time GPs who were currently practising were included, but those who were very part-time or were not currently practising (e.g. on maternity leave) were excluded.

Random samples for recruitment were drawn by DoHA. The samples included the GP's: name, contact details at their major practice (address and telephone number), age, sex, and activity level[‡] (that is, number of MBS general practice consultation service items[‡] claimed in the previous quarter and in the previous 12 months).¹⁰⁴

3.2.3 GP recruitment

The randomly sampled GPs were sent a letter describing the study, detailing what participation involved and inviting them to participate.¹⁰⁴ An average of 60 GPs were sent letters each week.

In order to retain their vocational registration, Australian GPs are required to participate in professional development programs conducted by the RACGP or the ACRRM. Registered GPs are required to participate in activities in the program to earn a specific number of points in each triennium (3 year period). BEACH is classified as a clinical audit, and participation earns GPs these points.

Approximately ten days after the approach letter was sent, the GP was contacted via telephone by a trained recruiter to ask whether they would like to participate. When a GP agreed to participate, they were entered into the GP participant database, assigned an identification number and a starting date was arranged. A participating sample of 25 GPs per week (with a target completion rate of 20 GPs per week) was recruited for 50 weeks each year. This created a rolling ever-changing sample.

3.2.4 Data collection

A research pack was posted to each GP 10 days before the start date. Each research pack contained:

- a pad of 105 (to allow for error) paper encounter forms. An example of the encounter form is included as Appendix 2.
- a one page questionnaire about the GP and their practice called the 'GP Profile'. An example is included as Appendix 3.
- a cover letter
- a set of instructions about how to complete the encounter form and a sample of a completed form (Appendix 4)
- two copies of the patient information card (Appendix 5).
- height and weight conversion charts (imperial to metric) used to record details of the patients height and weight for conversion to body mass index
- an alcoholic drinks chart providing information on 'standard' drinks.

The patient information card was intended to inform patients that their GP was participating in the study. GPs were instructed to ensure that all patients were given the card to read. The card instructed the patients to advise their GP if they did (or did not) want unidentified details of their encounter recorded. Consent was obtained as per the ethics requirements: from 1998–99 to 2004–05 only verbal consent was required; from 2005–06 onward, GPs were required to include a note in the patient's medical record that she/he had agreed to take part.

A member of the research team telephoned each GP on their agreed starting date to remind the GP to commence recording, and to answer any questions the GP might have.

GPs were instructed to record details for 102 patient encounters, in a consecutive manner (or as consecutive as possible where patient consent enabled this). The first 100 encounter forms that were completed were used in the study. Each GP returned their completed pad and their GP profile using the reply paid envelope provided.

3.2.5 Data elements

Three interrelated sets of data were collected in BEACH: encounter data; GP and practice data; and substudies of various topics. The substudies, referred to as SAND (Supplementary Analysis of Nominated Data), are described in Section 3.2.6. Figure 3.1 describes the relationships between the BEACH data elements diagrammatically. All variables can be directly related to GP and patient characteristics and to the encounter. All types of management (including pathology tests) are directly related to one or more problems being managed at the encounter. The data used in this thesis are highlighted in Figure 3.1, and defined below.

Encounter data

In BEACH, the encounter was defined as any professional interchange between a patient and a GP. Encounters can be 'direct'[‡] involving face-to-face meeting between the GP and patient or 'indirect'[‡] where there was no face-to-face contact but a service was provided (e.g. a prescription or referral arranged by telephone). Encounter data elements were collected on the encounter form (Appendix 2) and included details about the encounter itself, the patient, the problems that were managed and the management provided. The encounter data were cross-sectional. Therefore, the problems managed and the management provided reflect the encounter activity.



Encounter details

The only data element describing the encounter that is used in this thesis, is the MBS item number (Box 3.1). It is used to classify the encounter as level A, B, C or D^{\ddagger} (where applicable) based on the type of GP–patient consultation. These MBS levels are defined in the glossary.

Box 3.1: Encounter data elements										
Element	Definition and format									
Medicare Benefits Schedule item number [‡]	Item number recorded as claimable for the encounter. Up to 3 item numbers per encounter (Free text number)									
‡ Definition of term is included in glossary.										

The patient

Data elements about the patient used in this thesis are described in Box 3.2. The 'Health concession card'[‡] variable used in Chapter 7 is a binary variable that was created using the health care/benefits card[‡] and Repatriation health card[‡] data elements. A patient with either of these cards was classed as having a health concession card.

Box 3.2: Patient data elements	
Element	Definition and format
Age	Day (2 digit number), month (2 digit number) and year (4 digit number) of the patient's date of birth
Sex	Male, Female (Tick box)
New patient status [‡]	Indicates whether this is the patient's first visit to the practice (Yes/No tick box)
Health care/benefits card [‡]	Patient holds a card entitling them to a higher level of Government subsidy for health services. Examples of patients who may be eligible include pensioners, unemployed, low income earners (Yes/No tick box)
Repatriation health card [‡]	Patient holds a card from the Department of Veteran's Affairs entitling them to a range of Repatriation health care benefits (Yes/No tick box)
 Definition of term is included in glossary. 	

The problems managed

Data elements collected about a problem under management at the encounter that are used in this thesis, are described in Box 3.3.

Box 3.3: Data elements about the problems managed									
Element	Definition and format								
Diagnosis/problem [‡]	A statement of the provider's understanding of a health problem presented by a patient, family or community. GPs are instructed to record at the highest diagnostic level possible from the information available at the time. It can be recorded as a symptom, morbidity or process of care.								
	At least one and up to four problems can be recorded per encounter. Free text								
Problem status, either 'New' [‡] or 'Old' [‡]	New problem is defined as the first presentation of a problem, including the first presentation of a recurrence of a previously resolved problem, but excluding the presentation of a problem first assessed by another provider.								
	Old problem is defined as a previously assessed problem that requires ongoing care, including follow-up for a problem or an initial presentation of a problem previously assessed by another provider.								
	New/Old tick box for each problem.								
Definition of term is included in glossary									

Pathology tests ordered and other management provided

In BEACH, several types of management actions could be recorded, and all of these management activities were linked by the GP to the problem (or problems) for which they were given.

In this thesis, pathology data are the main type of management data used. GPs recorded each pathology test in free text as either a single test (such as fasting glucose test) or a battery of tests[‡] (such as a lipid profile) (Box 3.4). Up to five pathology tests or batteries of tests can be recorded per encounter. Each pathology test or battery must be linked to at least one problem but could be linked to up to four problems managed (the maximum recorded per encounter). This means that there can be a one-to-one, one-to-many or many-to-one relationship between pathology tests/batteries and problems managed.

The medication 'name' data element was also used in Chapter 5 (Box 3.4).

the encounter									
Element	Definition and format								
Pathology Single test (e.g. fasting glucose test)	Pathology tests/batteries of tests ordered by the GP at the encounter.								
Battery of tests [∓] (e.g. lipid profile)	recorded per encounter. Each test or batteries of tests could be recorded per encounter. Each test or battery of tests was linked by the GP to the related problem or problems managed at the encounter for which the test was ordered. Free text, and circle the problem number(s) to which the test related.								
Medication Details collected include: name, form	Includes medications prescribed, supplied by the GP and advised for over-the-counter purchase.								
(where required), strength, regimen.	Up to 4 medications could be recorded per problem, and a maximum of 16 medications could be recorded per encounter.								
	Tick boxes for GP-supply, and over-the-counter status. Free text for other medication data variables.								
Definition of term is included in glossary									

Box 3.4: Data elements about pathology tests and other management of problems at the encounter

GP and practice data

In BEACH, a single page questionnaire was used to gather data about each participating GP and their practice (Appendix 3). As it is possible for a GP to work at multiple practices, GPs were instructed to provide details about their major practice.

The GP and practice data elements used in this thesis are described in Box 3.5. In Chapter 7, the practice postcode was used to assign the relevant Australian State or Territory of the practice location, and to classify the rurality of the practice location using the Australian Standard Geographical Classification (ASGC) remoteness areas.¹⁰⁷

Box 3.5: Data elements about the participating GPs and their practice

Element	Definition and format
Age	GP's age. Free text (number).
Sex	GP's sex. Circle 'male' or 'female'.
Years in general practice	Number of years spent in general practice. Free text (number).
Fellowship of the Royal Australian College of General Practitioners (FRACGP)	Status of the GP as a fellow of the Royal Australian College of General Practitioner. Circle 'yes' or 'no'.
Country of graduation	Country where the GP's primary medical degree was obtained. Circle number to indicate one of the listed countries or record (in free text) an 'other' country.
Year of graduation	Year in which primary medical degree was obtained. Free text (number).
Sessions per week/Workload	Number of general practice sessions usually worked by the GP per week. A session is defined as approximately 4 hours e.g. a morning session. Free text (number).
Practice postcode	Postcode of the major practice address. Free text (number).
Size of practice (number of full-time equivalent [FTE] GPs)	Number of FTE GPs, including the participating GP, who work at the major practice. Free text (number).
Accreditation [‡] status	Whether the major practice is accredited according to the Royal Australian College of General Practitioner standards. Circle 'yes' or 'no'.
Status as a teaching practice	Whether the major practice provides training for undergraduate medical students and/or GP registrars. Circle a number(s) to indicate training provided.
‡ Definition of term is included in gloss	ary.

3.2.6 SAND substudy data

Each recording form had a section at the bottom that was used to investigate additional topics not covered in the encounter-based data. These additional investigations are referred to as SAND substudies.¹⁰⁸

In each GP's recording pad of 100 forms there were three 'sets' of SAND forms: one set of 40 forms and two sets of 30 forms. Each set represented a separate topic.

The set of 40 forms were the same in every pad and covered selected patient risk factors: alcohol intake, smoking status and self-reported height and weight (for calculation of body mass index [BMI] using the World Health Organization's BMI classification¹⁰⁹). Start and finish times were also recorded for calculation of the length of the encounter. The encounter form attached as Appendix 2 includes this set of SAND questions.

The questions on the two sets of 30 forms varied throughout each year. Each BEACH data year was divided into 10 five-week periods. In each five-week period

information was collected from about 100 GPs (20 recording per week), with a potential sample size of about 3,000 patient encounters on each topic. New studies were introduced at the beginning of each five-week period. It was possible to repeat topics over two or more different periods to increase the sample size.

The order of SAND sections in the GP recording pack was rotated, so that the set of 40 forms may have appeared first, second or third in the pad. Rotation of the forms minimised order effect on the data collected.

SAND data are used in two places in this thesis.

- The patient BMI data are used in Chapter 5, Section 5.9 to describe prevalence of overweight and obesity.
- A SAND study was designed for this thesis to investigate GP ordering of full blood counts and lipid tests (see Chapter 6).

3.2.7 Data entry and classification

Data from the BEACH study were entered into a Microsoft Access database designed for the study. The Access database was designed to enable efficient accurate data entry (for example picklists that automatically classified the selected term). Data were entered by trained secondary coders.

Most data elements collected in BEACH were classified according to the International Classification of Primary Care—Version 2 (ICPC-2), a product of the World Organization of Family Doctors (Wonca).¹¹⁰ ICPC-2 is accepted by the World Health Organization (WHO) in the WHO Family of International Classifications,¹¹¹ and is the recommended Australian standard for classification of data from general practice.¹¹²

In this thesis, the data elements classified to ICPC-2 are problems managed, and pathology tests ordered. These elements were coded using ICPC-2 PLUS (see below) and classified to ICPC-2.

International Classification of Primary Care

The ICPC-2 has a biaxial structure, with 17 chapters on one axis (each with an alphabetic code) and seven components on the other (numeric codes) (Figure 3.2). Chapters are based on body systems, with additional chapters for psychological and

social problems. Component 1 includes symptoms and complaints and component 7 covers diagnoses.

Component 2 (diagnostic, screening and prevention) is often applied in describing the problem managed (for example, check-up, immunisation). Components 3 to 6 cover other processes of care, including referrals, other treatments and orders for pathology and imaging. The components are standard and independent throughout all chapters.¹¹⁰

Components	Α	в	D	F	н	κ	L	Ν	Ρ	R	S	т	U	w	Х	Y	z
1. Symptoms, complaints																	
2. Diagnostic, screening, prevention																	
3. Treatment, procedures, medication																	
4. Test results																	
5. Administrative																	
6. Other																	
7. Diagnoses, disease																	
A General and unspecified L Musculoskeletal U Urological																	
B Blood and blood-formin organs	g N Neurological						W Pregnancy and family planning										
D Digestive			Ρ	Psychological							X Female genital						
F Eye			R	Respiratory						Y	Y Male genital						
H Ear	S			Skin							Z Social						
K Circulatory			т	End	docri	ine,	meta	aboli	ic ar	nd nu	utritio	onal					

ICPC-2 PLUS

Version 2 (ICPC-2)

In 1995, recognising a need for a coding and classification system for general practice electronic health records, the Family Medicine Research Centre (then Unit) developed a clinical terminology classified according to the ICPC-2, called ICPC-2 PLUS.¹¹³ It was based on the free text terms recorded by GPs in studies such as the AMTS.¹⁰⁵ Approximately 800,000 encounter records were used in the development of the terminology,¹⁰⁴ and it is regularly updated using the terms recorded in BEACH (approximately 1.2 million encounters from 1998 to 2010)¹⁷ and input from GPs using ICPC-2 PLUS in their electronic health records.

All free text data elements were coded using ICPC-2 PLUS. This ensured high coder reliability, and automatic classification of the concept. It also enabled analysis of the data at the classification level or (where required) at the more specific terminology level.

Presentation of data classified in ICPC-2

In this thesis, data coded in ICPC-2 PLUS are reported at the ICPC-2 classification level. However, there are some circumstances where this was not meaningful, and it was necessary to group concepts.

Problems managed

Where problems managed are grouped in this thesis they are marked with an asterisk in the tables and listed in Appendix 6 with the associated codes.

Concepts may need to be grouped 'above' the classification level. This involves grouping multiple ICPC-2 codes. For example, two ICPC-2 codes K86 (hypertension, uncomplicated) and K87 (hypertension, complicated) were grouped together and reported as 'Hypertension (non-gestational)*'.

Concepts are also grouped across the classification by grouping multiple ICPC-2 PLUS terms. For example, multiple PLUS terms that describe checks-ups were grouped together and reported as 'health check*'.

Grouping of pathology data

Pathology tests are classified in ICPC-2 in component 2, 'Diagnostic, screening and prevention'. There are seven rubrics[‡] within this component that relate to pathology tests. These rubrics can be applied in 16 of the 17 ICPC-2 chapters[‡]. They cannot be applied in the Social chapter of ICPC-2. The rubrics are:

- -32 Sensitivity
- -33 Microbiological/immunological test
- -34 Blood test
- -35 Urine test
- -36 Faeces test
- -37 Histological/exfoliative cytology
- -38 Other laboratory test not elsewhere classified.

This means there are 112 rubrics available to classify pathology tests in ICPC-2. While it is possible to analyse and report the pathology data using ICPC-2, it is too broad for meaningful interpretation. For example, a fasting glucose test is classified in T34—Blood test associated with the Metabolic, endocrine, nutritional system.

In Australia, pathology data are often reported using the groupings from the MBS PST in which pathology tests are grouped by pathology discipline. These groupings are more meaningful in the Australian setting so all pathology ICPC-2 PLUS codes are grouped to align with the MBS standard pathology groups. Some of the terms I have used to refer to these groups differ slightly from those used by the MBS, as I have used terminology that is common in general practice. The pathology groups analysed and reported in this thesis are:

- Haematology
- Chemistry (referred to as 'Chemical' in the MBS)
- Microbiology
- Immunology
- Histopathology (referred to as 'Tissue pathology' in the MBS)
- Cytopathology (referred to as 'Cytology' in the MBS)
- Infertility and pregnancy tests
- Simple basic tests
- Other tests not elsewhere classified (NEC). This group includes pathology tests not included in any of the above groups (such as, the 'Genetics' MBS group).

Each of these pathology groups with its associated ICPC-2 PLUS pathology codes is listed in Appendix 7.

Individual pathology tests and batteries of tests are also grouped together to form logical reporting entities (for example, the 'Glucose/glucose tolerance' test group includes all types of serum glucose tests). All pathology tests/batteries of tests that include multiple ICPC-2 PLUS codes are marked with an asterisk in the tables and listed in Appendix 7 with the associated PLUS codes.

Classification of pharmaceuticals

Pharmaceuticals that are prescribed, provided by the GP or advised for over-thecounter purchase are coded and classified according to the Anatomical Therapeutic Chemical (ATC)¹¹⁴ classification, which is the international standard for classifying medications for drug utilisation studies.¹¹⁵ The ATC has a hierarchical structure with five levels. For example:

- Level 1: C—Cardiovascular system
- Level 2: C10—Serum lipid reducing agents
- Level 3: C10A—Cholesterol and triglyceride reducers
- Level 4: C10AA—HMG CoA reductase inhibitors
- Level 5: C10AA01—Simvastatin (the generic drug).

3.2.8 Quality assurance

A quality assurance program was applied to ensure reliability of data entry. This included ongoing development and application of computer-aided error checks and 'locks' at the data entry stage (such as preventing female-specific problems being coded for male patients), and a series of logical data checks (e.g. medication dose outliers) to identify encounters where the coded data should be checked against the original recording form. All forms entered by new data entry staff were checked against the original form, and as staff became more experienced a sample of forms (e.g. one-in-ten) were checked by senior research staff. This ongoing process identified areas where further training was required.¹⁰⁸

3.2.9 Validity and reliability

In the development of a database such as BEACH, data gathering moves through specific stages: GP sample selection, cluster sampling around each GP, GP data recording, secondary coding and data entry. At each stage the data can be invalidated by the application of inappropriate methods. The methods adopted to ensure maximum reliability of coding and data entry have been described above. The statistical techniques adopted to ensure valid analysis and reporting of recorded data are described in Section 3.2.11. Previous work has demonstrated the extent to which a random sample of GPs recording information about a cluster of patients, represents all GPs and all patients attending GPs.¹¹⁶ Other studies have reported the degree to which GP-reported patient reasons for encounter and problems managed accurately reflect those recalled by the patient,¹¹⁷ and the reliability of secondary coding of reasons for encounter¹¹⁸ and problems managed.¹⁰⁵ The validity of ICPC as a tool with which to classify the data has also been investigated in earlier work.¹¹⁹

3.2.10 Representativeness

The extent to which data drawn from a sample can be generalised is a function of the ability of the sample to represent the population from which it is drawn. Random sampling of GPs improves the likelihood that a study will be representative, as each GP has equal probability of being selected. Even with random sampling, it is possible to end up with under-representation and/or over-representation of some groups within the final sample.¹⁰⁸

Sample weights can be calculated to improve the representativeness of the sample and adjust for any identified under-representation or over-representation within the sample. Weights were assigned by comparing the distribution of the sample against the distribution in the benchmark population for those characteristics that may influence the final results (e.g. GP age group and sex). Weights are calculated as the proportion of each subgroup in the population divided by the proportion in the sample. Over-representation results in a weight less than one, under-representation in a weight greater than one.¹⁰⁸

The BEACH study aims to gain a representative sample of GP–patient encounters. Representativeness of the GP sample is used to weight the encounters, based on the assumption that the characteristics of the patient encounter are related to the characteristics of the GP. Therefore to weight the encounters the representativeness of the GP sample needs to be determined. This is done by comparing the sample of participating GPs with GPs in the national sample frame (using MBS data supplied by DoHA). GP weights were calculated for the participants to match the age-sex distribution of all GPs in the total sample frame, correcting for any measured underrepresentation or over-representation. Weightings for GP age were stratified by GP sex.

The BEACH process requires that each GP provides details of 100 consecutive encounters. The assumption (based on earlier research) is that 100 encounters provide a reliable sample of the GP's patients and practice style.¹⁰⁶ However, there is considerable variation in the number of services provided by different GPs in a given year. This may have an impact on the reliability of any estimate due to the differences in the sampling fraction for each GP—a GP who provides 6,000 services in a given year should make a greater contribution to any national estimate of a selected activity than a GP who provides 3,000 services. Encounters were therefore

assigned an additional weight that was directly proportional to the activity level of the GP who recorded the encounter, based on the number of MBS general practice consultation service items claimed in the previous 12 months.

The final annual weighted encounter data set was created by multiplying raw rates by the GP age-sex weight and the GP activity weight. However, weights can only be applied to each single annual sample of BEACH data because the MBS data supplied by DoHA (on which weighting is based) are only provided for the specific BEACH 12 month data period (April to March).

In this thesis, weighted annual data are used in Chapter 8. In all other results chapters (Chapters 4–7), it is not methodologically sound to apply weighting for various reasons.

- In Chapters 4 and 5, multiple years of data are combined in the analysis, and it is not possible to apply weighting. However, combining multiple years of data increases the statistical power of the analysis.
- In Chapter 6, a SAND substudy is analysed, and as this uses a subsample of the total BEACH sample, weighting cannot be applied.
- In Chapter 7, it was not appropriate to weight the data because the variables used for weighting (GP age and GP sex) were adjusted for in the analysis.

The representativeness of the each of the annual GP samples and samples of encounter data (both unweighted and weighted) used in this thesis, 2000–01 to 2009–10, have been investigated and published elsewhere.^{17,120-128} This thesis centres on pathology testing ordered at encounters, therefore, the representativeness of the sampled encounters needs to be considered to determine whether the results can be generalised.

Briefly, the representativeness of each annual encounter sample was assessed by comparing the age-sex distribution of patients at BEACH encounters where MBS general practice consultation service items were recorded as claimable with the age-sex distribution of patients at all encounters claimed in Australia as MBS general practice consultation service items (data provided by DoHA). To aid this comparison, precision ratios were created by dividing the proportion of BEACH encounters by the proportion of Australian MBS encounters in each age-sex group.^{121,125} In determining whether any estimate is reliable, power calculations use a precision of 0.2 or 20% of the true proportion (or value). Therefore the BEACH

encounters are considered a reliable representation of MBS encounters when the range of precision is 0.8–1.2.

In each of the years of data used in this thesis (2000–01 to 2009–10), there was an excellent fit of the MBS and BEACH age and sex distribution both with and without weighting, with no age–sex category varying by more than 20% from the population distribution, with one exception. In 2008–09 one unweighted age–sex category (males aged 1–4 years) varied by 23% from the population distribution (based on raw data). After weighting, this ratio improved to within 20%.^{17,120-128} The fact that raw precision ratios (unweighted data) rarely varied by more than 20% indicates that each of the annual BEACH encounter samples is a good representation of Australian GP–patient encounters. Therefore the unweighted data used in Chapters 4–7 can be considered a reliable representation of GP–patient encounters in Australia.

3.2.11 Statistical methods

The analysis of BEACH data was conducted with Statistical Analysis System (SAS) version 9.1.3.¹²⁹ The primary unit of inference in the BEACH study is the encounter. Rates per 100 encounters are used when an event can occur more than once at the consultation (for example, problems managed or medications). Rates per 100 problems are also used when a management event (such as pathology tests ordered) can occur more than once per problem managed.^{45,108}

Proportions (%) are used when describing the distribution of an event that can arise only once per encounter (for example, patient age, sex) or once per problem (for example, new problem), or to describe the distribution of events within a class of events (for example, problem A as a percentage of total problems).^{45,108}

In general, the results presented include: the number of observations (*n*), the rate per 100 encounters or the rate per 100 problems managed, and the 95% confidence interval (CI).

It is possible to report the rate of pathology test ordering as a rate per 100 encounters and as a rate per 100 problems managed. When presented as a rate per 100 encounters, the number of pathology tests/batteries of tests at the encounter are analysed. When presented as a rate per 100 problems the analysis can be based on the number of pathology tests/batteries or the number of problem–pathology links.[‡] The problem–pathology links must be used in analyses investigating GPs' pathology ordering behaviour for specific problems. In most cases, data presented as a rate per 100 problems are based on the number of problem–pathology links. It is possible to link each test/battery to more than one problem; therefore there are more problem– pathology links than numbers of tests recorded at encounters. This is discussed further in Chapter 4.

BEACH is a single-stage cluster sample study design. The randomly sampled GPs each provide data about a cluster of 100 encounters. Studies with a cluster sample design violate the simple random sample (SRS) assumption because the probability of an encounter being included is a function of the probability of the GP being selected. There is also a secondary probability function of particular encounters being included in each GP's cluster (e.g. associated with characteristics of the location or type of the practice) and this increases the likelihood of sampling bias. In cluster samples, variance needs to be adjusted to account for the correlation between observations within clusters.^{45,108}

When a study design other than SRS is used, analytical techniques that consider the study design should be used. In this thesis, survey procedures in SAS version 9.1.3 were used to adjust the standard error used in calculating the 95% CIs, to accommodate the single-staged cluster sample study design.¹²⁹

Statistically significant differences

In this thesis, a statistically significant difference between two results was assessed by comparing the 95% CI around each result. Non-overlapping 95% CIs indicate a statistically significant difference between the results. The magnitude of this difference can be described as at least p < 0.05; however, non-overlapping CIs are known to be a conservative measure of significance.¹³⁰⁻¹³² By using this measure I am increasing the specificity while reducing the sensitivity, thus decreasing the chance of false positive and increasing the chance of false negative results. Due to the number of comparisons made in this thesis, and the large sample size of the BEACH study, I believe it is more appropriate to reduce the chance of false positive results.

Cumming and Finch stated that when comparing results from independent random samples, non-overlapping 95% CIs are more likely to represent a confidence level of approximately p < 0.01 than p < 0.05.¹³¹ Each annual sample of BEACH data is an

independent random sample. Therefore, in Chapters 4, 5 and 8, statistically significant changes over time measured using data from separate independent random samples, indicate a confidence level of p < 0.01.

The following gives an example of non-overlapping 95% CIs:

Result A: 11.5 per 100 problems (95% CI: 11.3–11.7) is significantly less than Result B: 11.9 per 100 problems (95% CI: 11.8–12.0).

When comparing results, if the two sets of CIs butt together the difference is regarded as marginal. For example:

Result A: 11.5 per 100 problems (95% CI: 11.3–11.7) is marginally lower than Result B: 11.9 per 100 (95% CI: 11.7–12.1).

If the CIs overlap, then no difference has been demonstrated.

In measuring changes in pathology ordering over time in Chapters 4 and 5, results from April 2000 to March 2002 (referred to as 2000–02), are compared with those from April 2006 to March 2008 (2006–08). Grey shading in tables indicates changes between 2000–02 and 2006–08, darker shading indicates a statistically significant change and lighter shading indicates a marginal change. The direction and type of change is indicated for each result in the far right column of the tables:

- \hbar/Ψ indicates a statistically significant linear change
- Λ/Ψ indicates a marginally significant linear change
- — indicates there was no change.

3.2.12 Extrapolated national estimates

Extrapolations can be used to estimate the number of occurrences of a selected event at GP encounters in Australia at a single time point or to estimate the total national effect of measured change. In this thesis, extrapolations are made to estimate the number of encounters in Australia involving the management of selected problems and the number involving pathology ordering.

The extrapolation method described in this section is the standard method used throughout this thesis. Extensions of this method are used in Chapters 4 and 8, and are described in the methods section of these chapters.

Extrapolations are calculated using the total number of MBS general practice consultation service items claimed in Australia in each financial year, rounded to the

nearest 100,000.⁵ These are listed for each year (2000–01 to 2009–10) in Table 3.1. Throughout this thesis, the number of MBS general practice consultation service items claimed in Australia is referred to as the number of national MBS GP–patient encounters.

Financial year	Number of GP MBS items	Rounded number of GP MBS items
2000–01	100,645,000	100,600,000
2001–02	99,921,000	99,900,000
2002–03	96,919,000	96,900,000
2003–04	96,330,000	96,300,000
2004–05	98,180,000	98,200,000
2005–06	101,095,000	101,100,000
2006–07	103,433,000	103,400,000
2007–08	109,518,000	109,500,000
2008–09	113,045,000	113,000,000
2009–10	116,646,000	116,600,000

 Table 3.1: Number of general practice professional services claimed from Medicare

 Australia each financial year, 2000–01 to 2009–10

Note: MBS - Medicare Benefits Schedule.

Source: Medicare statistics.5

When an extrapolation is based on data from multiple years, the average number of national MBS GP–patient encounters (rounded to the nearest 100,000) is used for the extrapolation and the extrapolation is interpreted as the average 'per annum' estimate in those years. Chapters 4 and 5 include extrapolations that are based on data from multiple years. These chapters use data collected over the period April 2000 to March 2008 (referred to as 2000–08); and changes over time are measured by comparing results from the first two years of the study period (2000–02) with those from the last two years (2006–08). For 2000–08 data, the number of national MBS GP–patient encounters used for extrapolation is 100.8 million. For 2000–02 data, the number of national MBS GP–patient encounters used is 100.3 million, and for 2006–08 data, it is 106.5 million.

The method used to calculate extrapolations is described below. The national number of encounters from 2000–02 is used in the example to describe the method, but it can be applied for other periods. To calculate the number of encounters involving the management of a specific problem:

 divide the BEACH management rate of each problem i.e. the 'rate per 100 encounters' for 2000–02 by 100, and then multiply by the total number of MBS general practice consultation service items in 2000–02, 100.3 million (rounded to the nearest 100,000) to give the estimated number of GP encounters at which the problem was managed nationally <u>per year</u> in 2000–02.

As an example the management rate of hypertension is extrapolated. In BEACH, hypertension was managed at a rate of 9.1 per 100 encounters in 2000–02. To extrapolate: $(9.1/100) \times 100.3$ million = an estimated 9.1 million encounters where hypertension was managed in general practice per year in 2000–02.

The above extrapolation is used most commonly throughout this thesis. All extrapolation estimates made are average annual estimates. For example, the number of encounters at which hypertension is managed by GPs was estimated to be 9.1 million encounters per annum in 2000–02.

Analysis of some problem data (and the related pathology testing data) are restricted by patient age. For example, overweight/obesity problems are investigated in patients aged 18 years and over. When data pertaining to these problems are extrapolated the number of GP–patient encounters is adjusted based on the proportion of encounters in the selected age group. For example, in 2000–02, 84.1% of encounters were with patients aged 18 years and over, therefore the number of GP–patient encounters used in extrapolations related to overweight/obesity data was 84.4 million (84.4% of 100.3 million).

Extrapolation estimates are rounded to the nearest 100,000 if more than a million, to the nearest 10,000 if between 100,000 and a million, and to the nearest 5,000 if less than 100,000.

4 Overview of pathology ordering

4.1 Background

As in most developed countries, the volume and cost of pathology tests ordered by GPs in Australia has increased over the past decades.^{5,6} In the literature, this increase is usually described in terms of the overall increase in and/or cost of pathology tests. Rarely are the clinical problems or pathology tests that contribute to this increase described.

It would be expected that data about tests contributing to the increase may be available in Australia from sources such as Medicare claims data or from laboratories that perform testing ordered by GPs. However, this is not the case because of limitations of these data sources. MBS payment structure and rules, in particular the episode coning rule and the structure of MBS pathology items (as described in Chapter 2), mean that Medicare data do not provide an accurate view of the tests ordered by GPs.

The private pathology laboratories that conduct the bulk of GP-ordered pathology testing have data on the tests ordered by GPs. As private companies own these laboratories, the data are not readily accessible and the extent of data recorded is unclear. Further, accessing a representative sample of GP ordering from laboratories is logistically difficult.

In order to describe the clinical problems contributing to the increase in GPs' pathology ordering, data about the problem must be linked to the pathology tests ordered by GPs. This type of data is rarely available. The BEACH study collects this data for all pathology tests recorded by participating GPs.

In the absence of knowledge about the causes of increases in pathology ordering many assumptions have been made in the literature. This chapter seeks to examine the influence of GP workload and pathology ordering behaviour on the national increase in pathology ordering.

4.2 Objectives

• To describe the extent to which general practice encounters involve pathology ordering, and the changes over time in GPs' pathology ordering behaviour.

- To investigate the types of pathology tests and batteries of tests ordered by GPs and what changes have occurred.
- To identify problems that generate the majority of GPs' pathology ordering, and to determine whether the management rate of these problems has changed (i.e. change in GP's clinical workload) and/or whether GPs' ordering behaviour in the management of these problems has changed.
- To identify problems for which GPs' pathology ordering behaviour has changed, and specifically those where assessment of appropriateness of GPs' pathology ordering may be indicated.

4.3 Method

Data used in this chapter were recorded in the BEACH study over eight years, from April 2000 to March 2008, inclusive (labelled as 2000–08). Changes in pathology ordering behaviour are measured using two 2-year data points: April 2000 – March 2002 and April 2006 – March 2008 inclusive (labelled respectively as 2000–02 and 2006–08).

The majority of analysis in this chapter uses unweighted data from multiple years. When data years are combined the data cannot be weighted (see Section 3.2.10). Weighted data is only used in this chapter for Figure 4.1 where single year data are presented.

Investigation of GP pathology ordering for all problems (Section 4.4.1) is based on the number of tests/batteries recorded at GP–patient encounters. As it is not based on problem–pathology links, each test/battery is counted only once per encounter. However, investigations of pathology ordering behaviour of GPs for specific problems and calculation of extrapolations (Section 4.4.2) are based on the number of problem–pathology links. It is possible for a test/battery to be linked by the GP to more than one problem; therefore there are more problem–pathology links than number of tests recorded (see Section 3.2.11).

The most common individual pathology tests/batteries, each accounting for at least 1% of pathology tests in either 2000–08, 2000–02 or 2006–08 are described and changes from 2000–02 to 2006–08 investigated.

Individual problems each generating at least 1% of total problem–pathology links in either 2000–08, 2000–02 or 2006–08 are described and changes from 2000–02 to 2006–08 investigated to determine whether there were changes in: the management rate of the problem; the likelihood of pathology being ordered in the management of the problem; the number of pathology tests/batteries ordered for that problem. Investigation of pathology ordering for these selected problems is based on the pathology linked by the GP to the selected problem.

4.3.1 Extrapolation

Three types of extrapolations are made in this chapter, the first using the method described in Section 3.2.12, and the second and third using an extension of this method. These extrapolations are used to demonstrate the magnitude of national change from 2000–02 to 2006–08. Extrapolation estimates made in this chapter are rounded to the nearest 10,000.

In this chapter extrapolations are based on data from multiple years; therefore, the average number of MBS GP-patient encounters over the period is used for each data point (2000–02 and 2006–08), and this is interpreted as the average 'per annum' estimate in those years. The number of national GP encounters used for extrapolation of 2000–02 data is 100.3 million and for 2006–08 data is 106.5 million.

The first type of extrapolation is used to estimate the number of encounters involving the management of a specific problem per year nationally (see Section 3.2.12).

The second and third types of extrapolations are based on the problem–pathology links for each problem. The second type of extrapolation estimates the number of contacts with the selected problem at which at least one pathology test/battery was ordered at GP encounters per year. The calculation was:

- for each problem divide the number of problems involving at least one pathology test by the number of BEACH encounters in 2000–02, and multiply by the total number of MBS general practice consultation service items in 2000–02
- repeat the process using data for 2006–08.

The difference between the two estimates gives the estimated national change between 2000–02 and 2006–08 in the number of problem contacts where at least one pathology test/battery was ordered.

The third type of extrapolation estimates the number of pathology tests/batteries ordered for each specific problem per year. The calculation was:

- for each problem divide the number of pathology tests/batteries of tests linked to the problem by the number of BEACH encounters in 2000–02, and multiply by the total number of MBS general practice consultation service items in 2000–02
- repeat the process using data for 2006–08.

The difference between the two estimates gives the estimated national change between 2000–02 and 2006–08 in the number of pathology tests/batteries ordered for each specific problem.

4.4 Results

4.4.1 Overview of data set

From April 2000 to March 2008, 7,843 GPs participated in BEACH, and provided data about 784,300 encounters involving the management of 1,174,893 problems. At these encounters GPs ordered 307,013 pathology tests/batteries of tests. In 2000–02, there were 198,200 encounters recorded and 64,389 pathology tests/batteries ordered by 1,982 GPs, and in 2006–08 there were 188,300 encounters recorded and 87,444 pathology tests/batteries ordered by 1,883 GPs (Table 4.1).

In BEACH, each pathology test/battery was linked to at least one and up to four problems per encounter. There were more problem–pathology links than tests/batteries: in 2000–08, there were 3.1% more links than tests (n = 316,572 problem–pathology links); in 2000–02, there were 3.2% more links than tests (n = 66,429 problem–pathology links); in 2006–08, there were 3.8% more links than tests (n = 90,753 problem–pathology links) (results not tabled).

Pathology ordered

In 2000–08, pathology tests/batteries were ordered at a rate of 39.1 tests/batteries per 100 encounters, and 26.1 per 100 problems. At least one pathology test/battery was ordered at 16.8% of encounters and for 12.8% of problems managed (Table 4.1). Figure 4.1 presents the annual rate of pathology ordered by GPs at encounters and in the management of problems over the eight years of the study. From 2000 to 2008 there was a significant[‡] linear increase in the rate of pathology ordered by GPs per

100 encounters, and per 100 problems managed (Figure 4.1), supporting the validity of measuring changes over time by comparing data from 2000–02 with that from 2006–08.

The rate of pathology tests/batteries ordered per 100 encounters increased significantly from 32.6 per 100 encounters in 2000–02 to 46.4 per 100 in 2006–08. This was due to significant increases in:

- the likelihood of at least one pathology test/battery being ordered at encounters (14.9% of encounters in 2000–02 and 18.7% in 2006–08)
- the number of pathology tests ordered per encounter once the decision to order was made (217.8 per 100 tested encounters in 2000–02 and 247.8 in 2006–08) (Table 4.1).

The number of problems managed per GP encounter also increased significantly from 147.3 per 100 encounters in 2000–02 to 153.3 per 100 in 2006–08. As this indicates an increase in the volume of clinical work undertaken per encounter, it is important to measure the rate of pathology per 100 problems managed. The rate of pathology tests/batteries ordered per 100 problems managed significantly increased from 22.2 per 100 in 2000–02 to 30.3 in 2006–08. This was due to significant increases in:

- the likelihood of at least one pathology test/battery being ordered in the management of problems (11.4% of problems in 2000–02 and 14.2% in 2006–08)
- the number of pathology tests ordered per problem once the decision to order was made (200.1 per 100 tested problems in 2000–02 and 221.3 in 2006–08) (Table 4.1).



Figure 4.1: Rate of pathology test/battery orders per 100 encounters and per 100 problems managed, 2000–01 to 2007–08 (95% confidence intervals)

		2000-0		2000-	02								
	Number	Rate/ Per cent	95% LCL	95% UCL	Number	Rate/ Per cent	95% LCL	95% UCL	Number	Rate/ Per cent	95% LCL	95% UCL	Change
General practitioners	7,843				1,982				1,883				
Number of encounters	784,300				198,200				188,300				
Problem management rate per 100 encounters	1,174,893	149.8	149.2	150.4	291,890	147.3	146.1	148.4	288,610	153.3	151.9	154.7	↑
Pathology order rate per 100 encounters	307,013	39.1	38.6	39.7	64,389	32.6	31.7	33.5	87,444	46.4	45.2	47.7	↑
At least one pathology order per encounter (Per cent of all encounters)	131,586	16.8	16.6	17.0	29,559	14.9	14.6	15.3	35,284	18.7	18.3	19.2	↑
Pathology order rate per 100 tested encounters		233.1	231.6	234.7		217.8	214.9	220.6		247.8	244.6	251.1	↑
Pathology order rate per 100 problems managed	307,013	26.1	25.8	26.5	64,389	22.2	21.6	22.7	87,444	30.3	29.6	31.0	↑
At least one pathology order per problem (Per cent of all problems managed)	150,187	12.8	12.6	12.9	33,196	11.4	11.1	11.6	41,019	14.2	13.9	14.5	↑
Pathology order rate per 100 tested problems		210.8	209.5	212.1		200.1	197.6	202.6		221.3	218.5	224.0	↑

Table 4.1: Overview of data set and summary of pathology ordering, 2000–08, 2000–02 and 2006–08

Note: Pathology data reported in this table are based on the number of pathology tests/batteries. The number of problem–pathology links is not used in the problem-based analyses reported in this table (see Section 3.2.11 and Section 4.4.1). Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change. LCL – lower confidence limit; UCL – upper confidence limit.

Figure 4.2 shows the distribution of the volume of pathology tests/batteries by patient age groups in 2000–08, 2000–02 and 2006–08. The patient age groups accounting for the highest volume of pathology tests/batteries ordered in 2000–08 were the 45–64 year age group (33.9% of tests/batteries) and the 25–44 year age group (27.4% of tests/batteries) (Figure 4.2).

The pattern of distribution was similar in 2000–02 and 2006–08. However, there were statistically significant changes over time in the proportion of testing accounted for by some age groups. Between 2000–02 and 2006–08, there was a significant decrease in the proportion of tests/batteries ordered for patients aged less than 45 years (41.0% in 2000–02 and 37.2% in 2006–08), and significant increases in the proportion ordered for patients aged 45–64 years (33.2% in 2000–02 and 34.9% in 2006–08) and for those aged 75 years and over (12.1% in 2000–02 and 14.0% in 2006–08) (Figure 4.2).

The distribution of likelihood of testing at encounters (i.e. at least one pathology test/battery ordered) across patient age groups is not presented in this thesis because it showed the same pattern of overall age distribution and changes over time as that presented in Figure 4.2 for volume of pathology.



Figure 4.2: Distribution of volume of pathology ordered across patient age groups, 2000–08, 2000–02 and 2006–08 (95% confidence intervals)
Figure 4.3 shows the age-specific rates of pathology ordering at encounters. In 2000–08, the rate of testing was highest at encounters with patients aged 45–64 years (48.8 pathology tests/batteries per 100 encounters with patients in this age group), followed by those with patients aged 65–74 years (44.2 per 100) and 25–44 years (43.6 per 100) (Figure 4.3).

Between 2000–02 and 2006–08, there were significant increases in the age-specific rates of pathology ordering at encounters with all patient age groups except at those with children aged less than 5 years. Patients aged 25 to 74 years had the highest age-specific rates of pathology testing, and these age groups also had the largest age-specific increases in pathology ordering rates from 2000–02 to 2006–08 (Figure 4.3).



The number of pathology tests ordered per encounter when at least one pathology test was ordered is presented in Figure 4.4. At encounters where pathology testing was ordered (tested encounters), GPs most commonly ordered one or two pathology tests/batteries (45.6% and 15.9% of tested encounters respectively in 2000–08). Over time there was a significant decrease in the proportion of tested encounters with one or two tests/batteries ordered: in 2000–02, 48.7% of tested encounters had one test and 17.0% had two, while in 2006–08, 43.1% had one test and 14.5% had two tests.

Simultaneously, there was a significant increase in the proportion with four or five tests/batteries ordered at tested encounters (Figure 4.4).

The distribution of number of tests per tested problem is not presented in this thesis as it showed the same pattern of overall distribution and changes over time as that presented in Figure 4.4 for number of tests per tested encounter.



when at least one test was ordered, 2000–08, 2000–02 and 2006–08 (95% confidence intervals)

Types of pathology tests ordered

Table 4.2 describes the pathology orders made for all problems by MBS pathology groups and the most common individual pathology tests, in the total data period 2000–08, and changes measured from 2000–02 to 2006–08. Pathology data are reported as rates per 100 problems as the number of problems managed per encounter increased over the study period.

At the MBS pathology group level, chemistry tests were the tests most commonly ordered by GPs in 2000–08 (14.3 per 100 problems managed), followed by those classified as haematology (4.9 per 100), microbiology (3.9) and cytopathology (1.5). Between 2000–02 and 2006–08, there were statistically significantly increases in the order rates of chemistry, haematology, microbiology and cytopathology tests; and

marginal significant increases in histopathology and immunology test order rates (Table 4.2).

The most common individual tests or batteries of tests ordered by GPs in 2000–08 were full blood counts (FBC) (3.5 per 100 problems), lipid tests (2.5 per 100), electrolytes, urea and creatinine (EUC) tests (1.8), liver function tests (LFT) (1.7), and glucose tests (1.6). The 22 most common tests/batteries accounted for 85.7% of all pathology tests recorded by GP participants in 2000–08 (Table 4.2). Between 2000–02 and 2006–08, there were significant increases in the ordering rate (per 100 total problems) of almost all frequently ordered individual tests. There were statistically significant increases in the order rate of FBC, lipid tests, EUC tests, LFT, glucose tests, thyroid function tests (TFT), multibiochemical analysis (MBA), ferritin, 'other chemistry' tests, HbA1c, 'other microbiology' tests, prostate specific antigen (PSA), histology skin tests, C reactive protein (CRP) and vitamin B12 tests. There were also marginal increases in the rates of Pap smears, urine microscopy, culture and sensitivity (M,C&S), coagulation tests, and vaginal swab M,C&S (Table 4.2).

		2000–08				2000–02	2006–08						
Pathology test ordered	Number (%)	Rate per 100 problems ^(a)	95% LCL	95% UCL	Number (%)	Rate per 100 problems ^(a)	95% LCL	95% UCL	Number (%)	Rate per 100 problems ^(a)	95% LCL	95% UCL	Change
MBS pathology groups													
Chemistry	167,676 (54.6)	14.3	14.0	14.5	33,843 (52.6)	11.6	11.2	12.0	49,681 (56.8)	17.2	16.7	17.7	↑
Haematology	57,594 (18.8)	4.9	4.8	5.0	12,636 (19.6)	4.3	4.2	4.5	15,743 (18.0)	5.5	5.3	5.6	↑
Microbiology	45,604 (14.9)	3.9	3.8	4.0	10,098 (15.7)	3.5	3.3	3.6	12,186 (13.9)	4.2	4.0	4.4	1
Cytopathology	17,152 (5.6)	1.5	1.5	1.5	3,931 (6.1)	1.4	1.3	1.4	4,534 (5.2)	1.6	1.5	1.7	↑
Other tests NEC	6,285 (2.0)	0.5	0.5	0.6	1,492 (2.3)	0.5	0.5	0.6	1,741 (2.0)	0.6	0.5	0.7	—
Histopathology	5,218 (1.7)	0.4	0.4	0.5	978 (1.5)	0.3	0.3	0.4	1,456 (1.7)	0.5	0.4	0.6	\uparrow
Immunology	4,540 (1.5)	0.4	0.4	0.4	970 (1.5)	0.3	0.3	0.4	1,369 (1.6)	0.5	0.4	0.5	\uparrow
Infertility and pregnancy tests	1,841 (0.6)	0.7	0.2	0.2	502 (0.8)	0.2	0.1	0.2	374 (0.4)	0.1	0.1	0.1	—
Simple basic tests	1,103 (0.4)	0.1	0.1	0.1	193 (0.3)	0.1	0.1	0.1	360 (0.4)	0.1	0.1	0.1	—
Individual pathology tests/batterie	es												
Full blood count	40,882 (13.3)	3.5	3.4	3.5	8,629 (13.4)	3.0	2.8	3.1	11,696 (13.4)	4.1	3.9	4.2	↑
Lipids*	29,578 (9.6)	2.5	2.5	2.6	6,627 (10.3)	2.3	2.2	2.4	8,410 (9.6)	2.9	2.8	3.0	↑
Electrolytes, urea and creatinine*	21,037 (6.9)	1.8	1.7	1.8	4,234 (6.6)	1.5	1.4	1.5	6,175 (7.1)	2.1	2.0	2.3	↑
Liver function*	20,183 (6.6)	1.7	1.7	1.8	4,201 (6.5)	1.4	1.4	1.5	6,067 (6.9)	2.1	2.0	2.2	1
Glucose/glucose tolerance*	18,615 (6.1)	1.6	1.5	1.6	4,215 (6.5)	1.4	1.4	1.5	5,170 (5.9)	1.8	1.7	1.9	1
Thyroid function*	17,225 (5.6)	1.5	1.4	1.5	3,335 (5.2)	1.1	1.1	1.2	5,034 (5.8)	1.7	1.7	1.8	↑
Pap smear*	16,818 (5.5)	1.4	1.4	1.5	3,844 (6.0)	1.3	1.2	1.4	4,449 (5.1)	1.5	1.4	1.6	\mathbf{T}
Urine M,C&S*	14,243 (4.6)	1.2	1.2	1.2	3,371 (5.2)	1.2	1.1	1.2	3,613 (4.1)	1.3	1.2	1.3	\mathbf{T}

Table 4.2: Distribution of pathology orders across MBS pathology groups and most frequent individual test orders for all problems, 2000–08, 2000–02 and 2006–08

(continued)

		2000–08				2000–02	2006–08						
Pathology test ordered	Number (%)	Rate per 100 problems ^(a)	95% LCL	95% UCL	Number (%)	Rate per 100 problems ^(a)	95% LCL	95% UCL	Number (%)	Rate per 100 problems ^(a)	95% LCL	95% UCL	Change
Individual pathology tests/batter	ies (continued)												
Multibiochemical analysis*	12,094 (3.9)	1.0	1.0	1.1	2,181 (3.4)	0.8	0.7	0.8	3,615 (4.1)	1.3	1.1	1.4	↑
Erythrocyte sedimentation rate	8,018 (2.6)	0.7	0.7	0.7	1,974 (3.1)	0.7	0.7	0.7	1,908 (2.2)	0.7	0.6	0.7	—
Ferritin*	7,780 (2.5)	0.7	0.6	0.7	1,463 (2.3)	0.5	0.5	0.5	2,540 (2.9)	0.9	0.8	0.9	↑
Other chemistry*	7,467 (2.4)	0.6	0.6	0.7	1,035 (1.6)	0.4	0.4	0.4	2,594 (3.0)	0.9	0.8	1.0	↑
Hormone assay*	7,118 (2.3)	0.6	0.6	0.6	1,663 (2.6)	0.6	0.5	0.6	1,739 (2.0)	0.6	0.5	0.7	_
HbA1c*	6,901 (2.2)	0.6	0.6	0.6	1,330 (2.1)	0.5	0.5	0.5	1,959 (2.2)	0.7	0.6	0.7	↑
Coagulation*	6,201 (2.0)	0.5	0.5	0.5	1,447 (2.2)	0.5	0.5	0.5	1,539 (1.8)	0.5	0.5	0.6	\uparrow
Other microbiology*	5,872 (1.9)	0.5	0.5	0.5	1,089 (1.7)	0.4	0.4	0.4	1,702 (1.9)	0.6	0.5	0.6	↑
Hepatitis serology*	4,697 (1.5)	0.4	0.4	0.4	1,191 (1.8)	0.4	0.4	0.4	1,116 (1.3)	0.4	0.3	0.4	_
Prostate specific antigen*	4,656 (1.5)	0.4	0.4	0.4	893 (1.4)	0.3	0.3	0.3	1,514 (1.7)	0.5	0.5	0.6	↑
Histology; skin	4,603 (1.5)	0.4	0.4	0.4	790 (1.2)	0.3	0.3	0.3	1,328 (1.5)	0.5	0.4	0.5	↑
C reactive protein	3,522 (1.1)	0.3	0.3	0.3	472 (0.7)	0.2	0.1	0.2	1,288 (1.5)	0.5	0.4	0.5	↑
Vaginal swab M,C&S	3,091 (1.0)	0.3	0.2	0.3	741 (1.2)	0.3	0.2	0.3	830 (0.9)	0.3	0.3	0.3	\uparrow
Vitamin B12*	2,482 (0.8)	0.2	0.2	0.2	400 (0.6)	0.1	0.1	0.2	847 (1.0)	0.3	0.3	0.3	↑
Subtotal	263,083 (85.7)				55, 125 (85.6)				75,133 (85.9)				
Total pathology tests	307,013 (100.0)	26.1	25.8	26.5	64,643 (100.0)	22.2	21.6	22.7	87,444 (100.0)	30.3	29.6	31.0	↑

Table 4.2 (continued): Distribution of pathology orders across MBS pathology groups and most frequent individual test orders for all problems, 2000–08, 2000–02 and 2006–08

(a) The rate at which the pathology test was ordered, expressed as a rate per 100 problems managed. Based on the number of pathology tests/batteries (not problem–pathology links) (see Section 3.2.11 and Section 4.4.1). For the number of problems in each data period see Table 4.1.

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the pathology tests/batteries accounting for more than 1% of all tests/batteries in any of the three data periods are included. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change (darker shading), ↑/♥ indicates a marginal change (lighter shading), and — indicates no change. LCL – lower confidence limit; UCL – upper confidence limit; NEC – not elsewhere classified; M,C & S – microscopy, culture and sensitivity.

4.4.2 Problems generating high volumes of pathology orders

There were 22 problems that each accounted for 1% or more of problem–pathology links recorded by GPs in BEACH. These problems accounted for more than half of problem–pathology links in each period (53.4% in 2000–08, 52.5% in 2000–02 and 54.8% in 2006–08) (Table 4.3).

Table 4.3 presents for each data period and for each problem: (i) the number of times it was recorded as a problem managed in the data set; (ii) the GP clinical workload it accounted for (management rate per 100 encounters); and (iii) the volume of pathology orders it generated (per cent of problem–pathology links).

Table 4.4 reports for 2000–02 and 2006–08 GP pathology ordering behaviour in the management of each problem, that is, the likelihood of testing (per cent of problem contacts involving at least one pathology test order) and the number of tests/batteries ordered (per 100 tested contacts with the problem).

Problems in Tables 4.3 and 4.4 are listed in decreasing order according to the proportion of total problem–pathology links accounted for by each problem in 2006–08 (Table 4.3).

For each problem, Table 4.5 reports the extrapolated estimates for 2000–02 and 2006–08 of: (i) the number of national encounters involving its management; (ii) the number of national encounters involving at least one pathology order; (iii) the number of pathology tests/batteries of tests ordered in its management in Australia. It also describes the amount of national change from 2000–02 to 2006–08 attributed to each problem. Problems in Table 4.5 are listed in decreasing order according to the proportion of national change accounted for by each problem.

Tables 4.3, 4.4 and 4.5, that are referred to throughout this section, are located on pages 67–72.

When the total rates of pathology ordering for 2000–02 and 2006–08 are extrapolated to the GP encounters claimed through Medicare (100.3 million per year in 2000–02 and 106.5 million per year in 2006–08), these data suggest that compared with 2000–02, in 2006–08 there were:

- 6.4 million additional problems per year for which the GP ordered at least one pathology test/battery of tests (23.2 million per year in 2006–08 compared with 16.8 million per year in 2000–02)
- 17.7 million additional tests/batteries of tests ordered per year by GPs (51.3 million per year in 2006–08 compared with 33.6 million per year in 2000–02) (Table 4.5).

GP pathology ordering behaviour did not change in the management of seven of the 22 problems investigated. These problems were sexually transmitted infections, urinary tract infection, pregnancy, arthritis, anaemia, hypothyroidism, and atrial fibrillation (Table 4.4). Pathology tests/batteries ordered for these seven problems accounted for a considerable proportion of total pathology ordering (12.1% of total problem–pathology links in 2000–02 and 11.7% in 2006–08) (Table 4.3). However there was no change in GPs' pathology ordering behaviour (i.e. decision of whether to test and number of tests per order) in the management of these problems. Any contribution to national change made by these problems was due to the increase in the total number of GP encounters claimed through Medicare with or without a simultaneous change in the management rate of the problem between 2000–02 and 2006–08 (Table 4.5). Therefore, these problems are not described in further detail in this chapter.

GP pathology ordering increased significantly in the management of 15 of the 22 problems investigated, from 2000–02 to 2006–08. These were: hypertension, Type 2 diabetes, lipid disorders, female genital check-ups, health checks (in patients aged 15 years and over), weakness/tiredness, 'blood test', 'abnormal test results', depression, menstrual problems, ischaemic heart disease, abdominal pain, overweight/obesity (in patients aged 18 years and over), menopause and viral illness (Table 4.4). The changes in GPs' pathology ordering in the management of these 15 problems are described in greater detail below.

Hypertension

The management rate of hypertension in general practice in Australia did not differ significantly between 2000–02 and 2006–08 (9.1 per 100 encounters in 2000–02 and 9.5 per 100 in 2006–08). Pathology ordered for hypertension problems accounted for

5.9% of all problem–pathology links recorded in 2000–02 and 6.3% in 2006–08 (Table 4.3).

The rate of pathology ordering per 100 hypertension contacts increased significantly, from 21.6 per 100 contracts in 2000–02 to 32.3 per 100 in 2006–08. This increase was due to significant increases in the likelihood of pathology being ordered in the management of hypertension (8.7% of hypertension contacts in 2000–02 compared with 11.9% in 2006–08), and increased number of pathology tests/batteries ordered per tested hypertension problem (248.2 per 100 tested contacts in 2000–02 compared with 270.4 per 100 in 2006–08) (Table 4.4).

When these data were extrapolated to the number of GP encounters claimed through Medicare nationally, I estimated there were about:

- 950,000 more encounters involving the management of hypertension problems in 2006–08 (10.1 million per year) than in 2000–02 (9.1 million per year)
- 410,000 additional hypertension contacts that involved the ordering of at least one pathology test/battery (tested contacts) in 2006–08 (1.2 million per year) compared with 2000–02 (790,000 per year)
- 1.3 million additional pathology tests/batteries ordered for hypertension in 2006–08 (3.2 million per year) than in 2000–02 (2.0 million per year). Pathology ordered in the management of hypertension accounted for 7.2% of the total national increase in pathology ordering that occurred between 2000–02 and 2006–08 (Table 4.5).

Type 2 diabetes

The management rate of Type 2 diabetes (T2D) increased significantly from 2.6 per 100 encounters in 2000–02 to 3.3 per 100 in 2006–08. Pathology ordered for T2D problems accounted for 5.0% of all problem–pathology links recorded in 2000–02 and 6.0% in 2006–08 (Table 4.3).

The rate of pathology ordering increased significantly from 63.6 tests/batteries ordered in 2000–02 per 100 contacts with T2D to 88.4 per 100 in 2006–08. This was due to significant increases in both the likelihood of pathology testing being ordered for T2D (27.3% in 2000–02 to 31.6% in 2006–08 of T2D problems), and the number of tests ordered once the decision to order tests was made (232.9 per 100 tested T2D contacts in 2000–02 and 280.2 in 2006–08) (Table 4.4).

When these data were extrapolated to the number of GP encounters claimed through Medicare nationally, I estimated there were about:

- 850,000 more encounters involving the management of T2D in 2006–08 (3.5 million per year) than in 2000–02 (2.6 million per year)
- 380,000 additional T2D contacts that involved the ordering of at least one pathology test/battery (tested contacts) in 2006–08 (1.1 million per year) than in 2000–02 (720,000 per year)
- 1.4 million additional pathology tests/batteries ordered for T2D in 2006–08
 (3.1 million per year) than in 2000–02 (1.7 million per year). T2D accounted for 8% of the national increase in pathology ordering that occurred between 2000–02 and 2006–08 (Table 4.5).

Lipid disorders

From 2000–02 to 2006–08, there was a significant increase in the management rate of lipid disorders, from 2.9 per 100 encounters to 3.5 per 100. Pathology ordered for lipid problems accounted for 5.1% of all problem–pathology links recorded in 2000–02 and 4.9% in 2006–08 (Table 4.3).

The rate of pathology ordering increased from 58.2 per 100 contacts with lipid disorders in 2000–02 to 66.5 per 100 in 2006–08. This was due to a significant increase in the number of tests ordered per tested lipid problem (191.4 tests/ batteries per 100 tested contacts in 2000–02 compared with 219.4 per 100 in 2006–08). There was no change in the likelihood of pathology tests being ordered in the management of lipid disorders (30.4% of lipid disorder contacts in 2000–02 and 30.3% in 2006–08) (Table 4.4).

When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there were about:

- 820,000 more encounters involving the management of lipid disorders
- 250,000 more lipid disorder problems for which pathology was ordered
- 790,000 more tests/batteries ordered for lipid disorders (accounting for 4.5% of the national increase in pathology tests/batteries) (Table 4.5).

Female genital check-ups/Pap smears

The management rate of female genital check-ups/Pap smears increased significantly from 2.0 per 100 encounters in 2000–02 to 2.4 in 2006–08. Pathology ordered for

female genital check-ups/Pap smears accounted for 4.7% of all problem–pathology links recorded in 2000–02 and 4.6% in 2006–08 (Table 4.3).

The rate of pathology ordering increased from 79.1 per 100 contacts with female genital check-ups/Pap smears in 2000–02 to 93.1 per 100 in 2006–08. This was due to an increased likelihood of pathology tests being ordered (70.1% of female genital check-up/Pap smear contacts in 2000–02 to 77.9% in 2006–08), and an increased number of tests being ordered once the decision to order had been made (112.9 per 100 tested contacts in 2000–02 to 119.5 in 2006–08) (Table 4.4).

When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there were about:

- 530,000 more encounters involving management of female genital checkups/Pap smears
- 580,000 more female genital check-up/Pap smear problems for which pathology was ordered
- 790,000 more tests/batteries ordered for female genital check-ups/Pap smears (accounting for 4.4% of the national increase in pathology tests/batteries) (Table 4.5).

Health checks

The management rate of health checks in patients aged 15 years and over increased significantly from 1.1 per 100 encounters in 2000–02 to 1.5 in 2006–08. Pathology ordered for health checks accounted for 3.4% of all problem–pathology links recorded in 2000–02 and 4.9% in 2006–08 (Table 4.3).

The rate of pathology ordering increased significantly from 122.0 tests/batteries in 2000–02 per 100 health check contacts to 178.9 per 100 in 2006–08. This was due to a significant increase in the number of tests for health checks ordered, once the decision to order tests was made (250.1 per 100 tested health check contacts in 2000–02 to 334.1 in 2006–08). There was no significant change in the likelihood that GPs would order pathology in the management of health checks (48.8% of contacts in 2000–02 and 53.5% in 2006–08) (Table 4.4).

When these data were extrapolated to the number of GP encounters claimed through Medicare nationally, I estimated there were about:

- 460,000 more encounters involving the management of health checks in 2006–08 (1.4 million per year) than in 2000–02 (930,000 per year)
- 290,000 additional health check contacts that involved the ordering of at least one pathology test/battery (tested contacts) in 2006–08 (750,000 per year) than in 2000–02 (460,000 per year)
- 1.4 million additional pathology tests/batteries ordered in the management of health checks in 2006–08 (2.5 million per year) than in 2000–02 (1.1 million per year). The proportion of the national increase in pathology tests/batteries ordered by GPs between 2000–02 and 2006–08 attributable to health checks was 7.6% (Table 4.5).

Weakness and tiredness

There was no change in the management rate of weakness/tiredness (0.8 per 100 encounters in 2000–02 and 0.7 in 2006–08) over the study period. Pathology ordered for weakness/tiredness problems accounted for 4.0% of all problem–pathology links recorded in 2000–02 and 3.5% in 2006–08 (Table 4.3).

The rate of pathology ordered for weakness/tiredness increased significantly (177.9 per 100 contacts in 2000–02 to 233.0 in 2006–08), due to an increased likelihood of pathology tests being ordered (50.3% of contacts in 2000–02 to 62.2 in 2006–08). The data also suggests a trend toward increased numbers of tests ordered per tested weakness/tiredness problem, but this did not reach statistical significance (Table 4.4). When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there were about:

- 10,000 more encounters involving management of weakness/tiredness
- 100,000 more weakness/tiredness problems for which pathology was ordered
- 450,000 more tests/batteries ordered for weakness/tiredness (accounting for 2.5% of the national increase in pathology tests/batteries) (Table 4.5).

'Blood test' problems

Problems labelled as 'blood tests' were managed at a significantly higher rate in 2006–08 (0.8 per 100 encounters) than in 2000–02 (0.6 per 100). Pathology ordered for 'blood test' problems accounted for 2.5% of all problem–pathology links recorded in 2000–02 and 3.3% in 2006–08 (Table 4.3).

The rate of pathology tests/batteries for 'blood test' problems increased significantly (from 147.6 to 199.1 per 100 'blood test' contacts), due to an increased likelihood of pathology tests being ordered (68.8% of blood test contacts in 2000–02 to 75.0% in 2006–08), and increased numbers of tests ordered per tested problem (214.7 tests/batteries per 100 tested contacts in 2000–02 to 265.4 in 2006–08) (Table 4.4). When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there were about:

- 290,000 more encounters involving management of 'blood test' problems
- 250,000 more 'blood test' problems for which pathology was ordered
- 870,000 more tests/batteries ordered for 'blood test' problems (accounting for 4.9% of the national increase in pathology tests/batteries) (Table 4.5).

'Abnormal test result' problems

The management rate of problems labelled as 'abnormal test results' increased significantly from 0.7 per 100 encounters in 2000–02 to 1.1 per 100 in 2006–08. Pathology ordered for 'abnormal test result' problems accounted for 1.4% of all problem–pathology links recorded in 2000–02 and 2.0% in 2006–08 (Table 4.3). The rate of pathology tests/batteries ordered for 'abnormal test result' problems increased from 68.6 per 100 'abnormal test result' contacts to 88.5 in 2006–08, due to an increased likelihood of pathology tests being ordered (42.3% of contacts in 2000–02 to 52.5% in 2006–08). There was no change in the number of tests ordered per 100 tested contacts (162.9 per 100 tested contacts in 2000–02 and 168.6 in 2006–08) (Table 4.4).

When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there were about:

- 430,000 more encounters involving management of 'abnormal test results'
- 300,000 more 'abnormal test result' problems for which pathology was ordered
- 530,000 more tests/batteries ordered for 'abnormal test result' problems (3.0% of the national increase in pathology tests/batteries) (Table 4.5).

Depression

There was a marginally significant increase in the management rate of depression between 2000–02 (3.9 per 100 encounters) and 2006–08 (4.2 per 100). Pathology

ordered for depression problems accounted for 1.1% of all problem–pathology links recorded in 2000–02 and 1.3% in 2006–08 (Table 4.3).

There was a significant increase in the ordering of pathology tests in the management of depression, from 9.8 per 100 contacts in 2000–02 to 14.7 per 100 in 2006–08. This was due to a significant increase in the likelihood of pathology tests/batteries being ordered for depression problems (from 3.3% of depression problems in 2000–02 to 4.6% in 2006–08). There was no change in the number of tests/batteries ordered per tested problem (299.6 per 100 tested depression problems in 2000–02 and 322.4 per 100 in 2006–08) (Table 4.4).

When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there were about:

- 590,000 more encounters involving the management of depression problems
- 80,000 more depression problems involving at least one pathology request
- 280,000 more tests/batteries ordered for depression problems (1.6% of the national increase in pathology tests) (Table 4.5).

Menstrual problems

The management rate of menstrual problems did not change, remaining at 0.8 per 100 encounters in 2000–02 and 2006–08. Pathology ordered for menstrual problems accounted for 1.4% of all problem–pathology links recorded in 2000–02 and 1.3% in 2006–08 (Table 4.3).

There was a significant increase in the rate of pathology ordered in the management of menstrual problems (60.6 per 100 contacts in 2000–02 to 80.2 in 2006–08), due to increased numbers of tests ordered per tested problem (209.3 per 100 tested menstrual problem contacts in 2000–02 compared with 252.3 in 2006–08). There was no change in the likelihood of pathology being ordered in the management of menstrual problems (28.9% in 2000–02 and 31.8% in 2006–08) (Table 4.4).

When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there were about:

- 50,000 more encounters involving the management of menstrual problems
- 40,000 more menstrual problems involving at least one pathology request
- 190,000 more tests/batteries ordered for menstrual problems (1.1% of the national increase in pathology tests) (Table 4.5).

Ischaemic heart disease

There was a marginal decrease in the management rate of ischaemic heart disease (IHD) from 1.4 per 100 encounters in 2000–02 to 1.2 per 100 in 2006–08. Pathology ordered for IHD problems accounted for 1.4% of all problem–pathology links recorded in 2000–02 and 1.2% in 2006–08 (Table 4.3).

The pathology test order rate per 100 contacts with IHD increased significantly, from 33.3 per 100 IHD contacts in 2000–02 to 46.7 in 2006–08. This was due to a significant increase in the number of tests/batteries ordered for IHD problems once the decision to order had been made (231.3 tests/batteries per 100 tested IHD problems in 2000–02 compared with 272.6 tests per 100 in 2006–08). However, there was no change in the likelihood of pathology ordering being involved in the management of IHD problems (Table 4.4).

When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there were about:

- 90,000 fewer encounters involving the management of IHD problems
- 20,000 more encounters with IHD problems for which pathology was requested
- 140,000 more tests/batteries ordered for IHD problems (0.8% of the total national increase in pathology tests) (Table 4.5).

Abdominal pain

There was no change in the management rate of abdominal pain over the study period, staying at 0.6 per 100 encounters. Pathology ordering in the management of abdominal pain accounted for 1.1% of all problem–pathology links recorded in 2000–02 and 1.0% in 2006–08 (Table 4.3).

The rate of pathology tests/batteries ordered for abdominal pain increased from 57.5 per 100 contacts in 2000–02 to 79.1 in 2006–08. This was due to a significant increase in the number of tests/batteries ordered for abdominal pain problems once the decision to order had been made (224.8 per 100 tested abdominal pain contacts in 2000–02 compared with 283.1 in 2006–08). There was no change in the likelihood that pathology would be ordered in the management of abdominal pain (25.6% of contacts in 2000–02 and 27.9 in 2006–08) (Table 4.4).

When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there were about:

- 60,000 more encounters involving the management of abdominal pain problems
- 30,000 more encounters with abdominal pain problems for which pathology was requested
- 180,000 more tests/batteries ordered for abdominal pain problems (1.0% of the total national increase in pathology tests) (Table 4.5).

Overweight/obesity

The overweight/obesity analysis in this section includes problems managed that were labelled by the GP as obesity or overweight at encounters with patients aged 18 years and over. From 2000–02 to 2006–08, there was no significant change in the management rate of overweight/obesity, remaining at 1.2 per 100 adult encounters. Pathology ordered for overweight/obesity problems accounted for 0.9% of all problem–pathology links recorded in 2000–02 and 1.0% in 2006–08 (Table 4.3). The rate of pathology tests ordered in the management of overweight/obesity among adult patients increased between 2000–02 and 2006–08 by more than 50% (30.6 to 47.1 per 100 overweight/obesity contacts). This increase was due to an increased likelihood that at least one test/battery was ordered (11.7% of contacts for overweight/obesity in 2000–02 and 16.5% in 2006–08). The number of tests ordered per tested problem did not change significantly (262.3 per 100 tested contacts in 2000–02 and 285.9 per 100 in 2006–08) (Table 4.4).

When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there were about:

- 100,000 more encounters involving the management of overweight/obesity (1 million per year in 2000–02 and 1.1 million per year in 2006–08)
- 60,000 additional tested overweight/obesity contacts (120,000 per year in 2000–02 and 180,000 per year in 2006–08)
- 210,000 additional tests/batteries requested for overweight/obesity (310,000 per year in 2000–02 and 520,000 per year in 2006–08). Of the total national increase in pathology tests/batteries, 1.2% was attributable to pathology ordered in the management of overweight/obesity (Table 4.5).

Menopausal complaint

The management rate of menopausal symptoms and complaints decreased significantly from 1.6 per 100 encounters in 2000–02 to 1.0 in 2006–08. Pathology ordered for menopausal complaint problems accounted for 1.5% of all problem– pathology links recorded in 2000–02 and 0.9% in 2006–08 (Table 4.3).

There was a significant increase in the order rate of pathology tests/batteries for menopausal complaints (33.0 per 100 contacts in 2000–02 and 44.5 in 2006–08), due to increased likelihood of pathology tests being ordered from 13.4% of menopausal complaint contacts in 2000–02 to 19.6 in 2006–08. The number of tests ordered per tested problem did not change significantly (247.0 per 100 tested contacts in 2000–02 and 227.7 per 100 in 2006–08) (Table 4.4).

When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there:

- were about 480,000 fewer encounters involving the management of menopausal complaint problems
- was no change in the number of menopausal contacts involving at least one pathology test
- were about 30,000 fewer tests/batteries requested for menopausal complaints. The proportion of the national change in pathology tests/batteries ordered by GPs in 2006–08, compared with 2000–02, attributable to menopausal complaints was –0.2% (Table 4.5). The decrease in the management rate outweighed the effect of

the increased pathology ordering of the GPs, creating a net decrease in pathology ordering attributable to menopausal complaints.

Viral illness

There was a significant decrease in the management rate of viral illness from 1.4 per 100 encounters in 2000–02 to 1.1 in 2006–08. Pathology ordered for viral illness problems accounted for 1.1% of all problem–pathology links recorded in 2000–02 and 0.9% in 2006–08 (Table 4.3).

There was a significant increase in the order rate of pathology tests/batteries for viral illness, increasing from 26.7 per 100 contacts in 2000–02 to 37.4 in 2006–08. There was a trend toward increased likelihood of tests being ordered for viral illness, but this did not reach statistical significance. There was also no change in the number of tests ordered per tested contact over the study period for viral illness (Table 4.4).

When these data were extrapolated to national GP Medicare encounters, I estimated that, compared with 2000–02, in 2006–08 there:

- were about 260,000 fewer encounters involving the management of viral illness
- was no change in the number of viral illness contacts involving at least one pathology test
- were about 60,000 more tests/batteries requested for viral illness. The proportion of the national change in pathology tests/batteries ordered by GPs in 2006–08, compared with 2000–02, attributable to viral illness was 0.3% (Table 4.5).

	Numb I	er of probl managed	lems	Manageme	nt rate per 100 enco (95% Cl)	▲ (b)	Number of pathology tests linked to the problem (% of total links) ^(c)			
Problem	2000–08	2000–02	2006–08	2000–08	2000–02	2006–08	¥	2000–08	2000–02	2006–08
Hypertension (non-gestational)*	72,171	18,007	17,793	9.2 (9.0–9.4)	9.1 (8.8–9.4)	9.5 (9.1–9.8)	_	18,889 (6.0)	3,884 (5.9)	5,744 (6.3)
Type 2 diabetes	22,938	5,211	6,172	2.9 (2.9–3.0)	2.6 (2.5–2.8)	3.3 (3.1–3.4)	↑	17,709 (5.6)	3,314 (5.0)	5,459 (6.0)
Lipid disorders	25,248	5,782	6,629	3.2 (3.2–3.3)	2.9 (2.8–3.0)	3.5 (3.4–3.7)	↑	15,777 (5.0)	3,363 (5.1)	4,410 (4.9)
Female genital check-up/Pap smear*	17,416	3,920	4,471	2.2 (2.1–2.3)	2.0 (1.8–2.1)	2.4 (2.2–2.6)	↑	14,778 (4.7)	3,100 (4.7)	4,163 (4.6)
Health check (15+ years)*	8,120	1,846	2,464	1.2 (1.1–1.2)	1.1 (1.0–1.2)	1.5 (1.4–1.6)	↑	12,008 (3.8)	2,252 (3.4)	4,407 (4.9)
Weakness/tiredness	5,627	1,509	1,373	0.7 (0.7–0.8)	0.8 (0.7–0.8)	0.7 (0.7–0.8)	_	11,559 (3.7)	2,684 (4.0)	3,199 (3.5)
Blood test – all*	5,222	1,121	1,516	0.7 (0.6–07)	0.6 (0.5–0.6)	0.8 (0.7–0.9)	↑	9,444 (3.0)	1,655 (2.5)	3,018 (3.3)
Sexually transmitted infection*	5,999	1,141	1,830	0.8 (0.7–0.8)	0.6 (0.5–0.6)	1.0 (0.8–1.1)	↑	7,128 (2.3)	1,198 (1.8)	2,292 (2.5)
Urinary tract infection*	13,283	3,253	3,152	1.7 (1.7–1.7)	1.6 (1.6–1.7)	1.7 (1.6–1.7)	_	8,201 (2.6)	2,003 (3.0)	2,026 (2.2)
Pregnancy*	8,077	1,799	2,730	1.0 (1.0–1.1)	0.9 (0.8–1.0)	1.5 (1.3–1.6)	↑	6,030 (1.9)	1,197 (1.8)	2,022 (2.2)
Abnormal test result*	6,955	1,389	2,022	0.9 (0.9–0.9)	0.7 (0.7–0.8)	1.1 (1.0–1.1)	↑	5,462 (1.7)	953 (1.4)	1,790 (2.0)
Arthritis – all*	29,755	7,788	6,703	3.8 (3.7–3.9)	3.9 (3.8–4.1)	3.6 (3.4–3.7)	$\mathbf{\Psi}$	5,577 (1.8)	1,277 (1.9)	1,345 (1.5)
Depression*	31,309	7,658	7,898	4.0 (3.9–4.1)	3.9 (3.7–4.0)	4.2 (4.0-4.4)	\uparrow	3,943 (1.2)	749 (1.1)	1,164 (1.3)
Menstrual problems*	5,961	1,489	1,419	0.8 (0.7–0.8)	0.8 (0.7–0.8)	0.8 (0.7–0.8)	_	4,247 (1.3)	902 (1.4)	1,138 (1.3)

Table 4.3: Overview of problems that accounted for at least 1% of total problem–pathology links, 2000–08, 2000–02 and 2006–08

(continued)

	Numb	er of prob managed	lems	Manager	nent rate per 100 en (95% Cl)	▲ (b)	Number of pathology tests linked to the problem (% of total links) ^(c)			
Problem	2000–08	2000–02	2006–08	2000–08	2000–02	2006–0	- 17 B ↓	2000–08	2000–02	2006–08
Anaemia*	5,129	1,271	1,188	0.7 (0.6–0.7)	0.6 (0.6–0.7)	0.6 (0.6–0.7) —	4,363 (1.4)	1,007 (1.5)	1,077 (1.2)
Ischaemic heart disease*	10,048	2,707	2,261	1.3 (1.2–1.3)	1.4 (1.3–1.5)	1.2 (1.1–1.3) ↓	3,959 (1.3)	902 (1.4)	1,055 (1.2)
Hypothyroidism	4,990	1,057	1,397	0.6 (0.6–0.7)	0.5 (0.5–0.6)	0.7 (0.7–0.8) 🛧	3,541 (1.1)	728 (1.1)	1,036 (1.1)
Abdominal pain*	4,803	1,215	1,185	0.6 (0.6–0.6)	0.6 (0.6–0.7)	0.6 (0.6–0.7) —	3,389 (1.1)	699 (1.1)	937 (1.0)
Overweight/obesity (adults)*	7,797	1,975	1,935	1.2 (1.1–1.2)	1.2 (1.1–1.3)	1.2 (1.1–1.3) —	2,916 (0.9)	605 (0.9)	912 (1.0)
Atrial fibrillation	6,334	1,323	1,813	0.8 (0.8–0.8)	0.7 (0.6–0.7)	1.0 (0.9–1.0) 🛧	3,251 (1.0)	642 (1.0)	900 (1.0)
Menopausal complaint	10,044	3,093	1,923	1.3 (1.2–1.3)	1.6 (1.5–1.7)	1.0 (0.9–1.1) 🗸	3,687 (1.2)	1,020 (1.5)	856 (0.9)
Viral illness	9,814	2,847	2,086	1.3 (1.2–1.3)	1.4 (1.3–1.5)	1.1 (1.0–1.2) 🗸	3,174 (1.0)	760 (1.1)	780 (0.9)
Subtotal	317,040	77,401	79,960		- 			169,033 (53.4)	34,894 (52.5)	49,730 (54.8)
Total	1,174,893	291,890	288,610	149.8 (149.2–150.4)	147.3 (146.1–148.4)	153.3 (151.9–154.7) 🛧	316,549 (100.0)	66,429 (100.0)	90,753 (100.0)

Table 4.3 (continued): Overview of problems that accounted for at least 1% of total problem-pathology links, 2000-08, 2000-02 and 2006-08

(a) Management rate of the problem, expressed as a rate per 100 encounters. For the number of problems in each data period see Table 4.1. Shading indicates a change between 2000–02 and 2006–08, darker shading indicates a statistically significant change and lighter shading indicates a marginal change.

(b) The direction and type of statistically significant change is indicated for each measure between 2000–02 and 2006–08: ↑/↓ indicates a statistically significant change, ↑/↓ indicates a marginal change, and — indicates no change.

(c) The number and proportion of pathology tests/batteries of tests linked to each problem. It is possible for a single pathology test/battery to be linked to more than one problem (see Section 3.2.11 and Section 4.4.1).

* Includes multiple ICPC-2 PLUS codes (see Appendix 6).

Note: Problems that account for at least 1% of problem-pathology links in any of the three data periods are included in this table. CI - confidence interval.

	Pathology per 100 problem contacts ^(a) (95% Cl)			Per cent of problem contacts with at least 1 test ^(c) (95% CI)			Pathology per 100 tested problems ^(d) (95% Cl)			
Problem	2000–02	2006–08	¥	2000–02	2006–08	¥	2000–02	2006–08	. ↓	
Hypertension (non-gestational)*	21.6 (20.0–23.2)	32.3 (30.3–34.2)	↑	8.7 (8.1–9.3)	11.9 (11.3–12.6)	↑	248.2 (239.5–257.0)	270.4 (262.5–278.4)	↑	
Type 2 diabetes	63.6 (59.6–67.6)	88.4 (83.7–93.2)	↑	27.3 (25.8–28.8)	31.6 (30.1–33.0)	1	232.9 (224.8–241.0)	280.2 (272.4–288.1)	1	
Lipid disorders	58.2 (54.7–61.6)	66.5 (62.5–70.6)	↑	30.4 (28.9–31.9)	30.3 (28.9–31.8)	-	191.4 (184.6–198.2)	219.4 (211.6–227.3)	1	
Female genital check-up/Pap smear*	79.1 (76.4–81.8)	93.1 (90.2–96.0)	↑	70.1 (62.2–72.0)	77.9 (76.2–79.7)	1	112.9 (110.7–115.1)	119.5 (116.7–122.2)	↑	
Health check (15+ years)*	122.0 (110.7–133.3)	178.9 (167.3–190.4)	↑	48.8 (45.1–52.5)	53.5 (50.7–56.3)	-	250.1 (236.1–264.2)	334.1 (322.8–345.4)	↑	
Weakness/tiredness	177.9 (164.0–191.8)	233.0 (217.3–248.7)	↑	50.3 (46.7–53.9)	62.2 (58.5–65.9)	↑	353.6 (343.2–364.0)	374.6 (363.5–385.7)		
Blood test – all*	147.6 (136.3–159.0)	199.1 (186.4–211.8)	↑	68.8 (65.7–71.8)	75.0 (72.2–77.8)	^	214.7 (201.6–227.8)	265.4 (253.1–277.8)	1	
Sexually transmitted infection*	105.0 (95.3–115.7)	125.2 (115.3–135.2)	_	42.6 (39.3–45.9)	46.6 (43.9–49.3)	_	246.5 (231.4–261.6)	268.7 (251.2–286.2)	_	
Urinary tract infection*	61.6 (58.9–64.3)	64.3 (61.4–67.1)	—	53.4 (51.4–55.4)	55.7 (53.7–57.7)	—	115.3 (112.5–118.0)	115.4 (112.6–118.2)		
Pregnancy*	66.5 (60.5–72.6)	74.1 (68.6–79.6)	—	34.5 (31.9–37.1)	35.0 (33.0–37.0)	_	192.8 (180.6–240.9)	211.7 (200.8–222.6)		
Abnormal test result*	68.6 (62.7–74.5)	88.5 (83.2–93.9)	↑	42.3 (39.4–45.1)	52.5 (50.2–54.9)	↑	162.9 (154.1–171.7)	168.6 (161.6–175.6)		
Arthritis – all*	16.4 (14.6–18.2)	20.1 (18.0–22.1)	_	5.5 (5.0–6.1)	6.5 (5.9–7.1)	_	296.3 (283.7–308.9)	308.5 (294.2–322.8)		
Depression*	9.8 (8.4–11.2)	14.7 (12.8–16.7)	↑	3.3 (2.8–3.7)	4.6 (3.9–5.2)	↑	299.6 (280.2–319.0)	322.4 (300.6–344.3)		
Menstrual problems*	60.6 (54.3–66.9)	80.2 (72.2–88.2)	↑	28.9 (26.5–31.4)	31.8 (29.1–34.5)	-	209.3 (196.1–222.4)	252.3 (238.4–266.2)	↑	

Table 4.4: Changes in pathology ordering for problems that accounted for at least 1% of problem–pathology links, 2000–02 compared with 2006–08

(continued)

	Pathology per 100 problem contacts ^(a) (95% Cl)			Per cent of problem contacts with at least 1 test ^(c) (95% CI)			Pathology per 100 tested problems ^(d) (95% Cl)			
Problem	2000–02	2006–08	¥	2000–02	2006–08	₩	2000–02	2006–08	Ψ	
Anaemia*	79.2 (71.8–86.6)	90.7 (82.3–99.0)	_	34.9 (32.2–37.7)	37.3 (34.4–40.2)	_	226.8 (215.4–238.2)	243.1 (230.4–255.8)	_	
Ischaemic heart disease*	33.3 (29.2–37.5)	46.7 (41.3–52.1)	↑	14.4 (12.9–15.9)	17.1 (15.4–18.8)	-	231.3 (216.3–246.3)	272.6 (257.5–287.8)	↑	
Hypothyroidism	68.9 (62.3–75.5)	74.2 (68.2–80.1)	_	43.0 (39.7–46.2)	44.4 (41.5–47.2)	_	160.4 (150.7–170.1)	167.1 (158.2–176.0)	—	
Abdominal pain*	57.5 (50.8–64.3)	79.1 (70.4–87.8)	↑	25.6 (23.0–28.2)	27.9 (25.3–30.6)	-	224.8 (210.0–239.6)	283.1 (265.7–300.5)	↑	
Overweight/obesity (adults)*	30.6 (24.6–36.7)	47.1 (39.9–54.4)	↑	11.7 (9.7–13.7)	16.5 (14.1–18.9)	1	262.3 (241.4–283.3)	285.9 (268.2–303.6)	_	
Atrial fibrillation	48.5 (43.9–53.1)	49.6 (45.3–54.0)	_	37.4 (34.4–40.5)	37.4 (34.6–40.2)	_	129.7 (121.8–137.6)	132.7 (125.6–139.9)	_	
Menopausal complaint	33.0 (28.9–37.0)	44.5 (37.9–51.1)	↑	13.4 (12.0–14.7)	19.6 (15.2–23.9)	↑	247.0 (231.4–262.5)	227.7 (191.1–264.2)	_	
Viral illness	26.7 (22.8–30.6)	37.4 (32.0–42.7)	↑	10.6 (9.3–12.0)	13.4 (11.7–15.1)		250.8 (232.9–268.8)	279.6 (261.2–298.0)	_	
Total problems	22.8 (22.2–23.4)	31.4 (30.7–32.2)	↑	11.4 (11.1–11.6)	14.2 (13.9–14.5)	▲	200.1 (197.6–202.6)	221.3 (218.5–224.0)	↑	

Table 4.4 (continued): Changes in pathology ordering for problems that accounted for at least 1% of problem–pathology links, 2000–02 compared with 2006–08

(a) The rate at which pathology tests/batteries of tests were ordered for each problem (based on the number of problem–pathology links), expressed as a rate per 100 specified problems. For details on the number of cases and pathology tests for each problem see Table 4.3.

(b) The direction and type of change is indicated for each measure between 2000–02 and 2006–08: ↑/ indicates a statistically significant change, ↑/ indicates a marginal change, and — indicates no change.

(c) The proportion of contacts with each problem where at least one pathology test/battery was ordered.

(d) The number of pathology tests/batteries per 100 tested contacts with each problem.

* Includes multiple ICPC-2 PLUS codes (see Appendix 6).

Note: Problems that account for at least 1% of problem–pathology links in either 2000–08, 2000–02 or 2006–08 are included in this table. Shading indicates a statistically significant change between 2000–02 and 2006–08. CI – confidence interval.

	Number of e	encounters problem	with each	Number o involv	of problem co /ing patholog	ontacts IV ^(b)	Number of p	Per cent of the change due to		
Problem	2000–02	2006–08	Change ^(c)	2000–02	2006–08	Change ^(c)	2000–02	2006–08	Change ^(c)	each problem ^(d)
Type 2 diabetes	2,640	3,490	850	720	1,100	380	1,680	3,090	1,410	8.0
Health check (15+ years)*	930	1,390	460	460	750	290	1,140	2,490	1,350	7.6
Hypertension (non-gestational)*	9,110	10,060	950	790	1,200	410	1,970	3,250	1,280	7.2
Blood test – all*	560	850	290	390	640	250	840	1,710	870	4.9
Lipid disorders	2,920	3,750	820	890	1,140	250	1,700	2,490	790	4.5
Female genital check-up/ Pap smear*	1,860	2,390	530	1,390	1,970	580	1,570	2,350	790	4.4
Sexually transmitted infection*	570	1,000	430	250	480	240	610	1,300	690	3.9
Pregnancy*	910	1,540	630	310	540	230	610	1,140	540	3.0
Abnormal test result*	690	1,120	430	300	600	300	480	1,010	530	3.0
Weakness/tiredness	760	780	10	380	480	100	1,360	1,810	450	2.5
Depression*	3,870	4,470	590	130	200	80	380	660	280	1.6
Hypothyroidism	530	790	260	230	350	120	370	590	220	1.2
Overweight/obesity (adults)*	1,000	1,090	100	120	180	60	310	520	210	1.2
Menstrual problems*	750	800	50	220	260	40	460	640	190	1.1
Atrial fibrillation	670	1,030	360	250	380	130	320	510	180	1.0

Table 4.5: Extrapolated national estimated number (thousands) of management and pathology ordered for problems that accounted for at least 1% of problem–pathology links, changes from 2000–02 to 2006–08^(a)

(continued)

	Number of e	encounters v problem	with each	Number o involv	of problem co ving patholog	ontacts IV ^(b)	Number of p	Per cent of the		
Problem	2000–02	2006–08	Change ^(c)	2000–02	2006–08	Change ^(c)	2000–02	2006–08	Change ^(c)	each problem ^(d)
Abdominal pain*	610	670	60	160	190	30	350	530	180	1.0
Ischaemic heart disease*	1,360	1,280	-90	200	220	20	460	600	140	0.8
Urinary tract infection*	1,640	1,780	140	880	990	110	1,010	1,150	130	0.7
Arthritis – all*	3,910	3,770	-140	220	250	30	650	760	110	0.6
Anaemia*	640	670	30	220	250	30	510	610	100	0.6
Viral illness	1,440	1,180	-260	150	160	0	380	440	60	0.3
Menopausal complaint	1,560	1,080	-480	210	210	0	520	480	-30	-0.2
Total	100,300	106,500	6,200	16,800	23,200	6,400	33,620	51,330	17,710	100.0

Table 4.5 (continued): Extrapolated national estimated number (thousands) of management and pathology ordered for problems that accounted for at least 1% of problem–pathology links, changes from 2000–02 to 2006–08^(a)

(a) Extrapolations are presented in thousands and are rounded to the nearest 10,000.

(b) Calculations are based on the number of problem-pathology linkages. There are more problem-pathology linkages than pathology tests recorded as each pathology test can be linked to more than one problem (see Section 3.2.11 and Section 4.4.1).

(c) Calculations of change were made prior to rounding to the nearest 10,000. Due to the rounding, calculating change using the 2000–02 and 2006–08 data presented in the table above may produce a different result to that reported in the 'Change' column.

(d) The proportion of the total extrapolated national increase in pathology tests/batteries (*n* = 17.7 million) that was attributable to each problem.

* Includes multiple ICPC-2 PLUS codes (see Appendix 6).

Note: Problems that account for at least 1% of problem-pathology links in either 2000-08, 2000-02 or 2006-08 are included in this table.

4.5 Discussion

This chapter has provided a comprehensive overview of GPs' pathology ordering in Australia in 2000–08. It described the total increase in GPs' pathology ordering that occurred between 2000–02 and 2006–08, and identified the most common individual tests and clinical problems contributing to the measured increase in pathology ordering.

Three factors (each of which can operate independently) contributed to the total increase in the volume of pathology tests/batteries of tests ordered by GPs between 2000–02 and 2006–08: an increase in the number of national GP encounters in the country (reflecting an increase in total GP workload); a statistically significant increase in the number of problems managed per encounter (reflecting a change in GPs' encounter workload); and a statistically significant increase in GPs' pathology ordering in the management of these problems (reflecting change in GPs' pathology ordering behaviour). The latter was measured as an increase in the rate of pathology ordering due to increases in the likelihood of GPs' initiating pathology ordering in the management of problems, and the number of tests ordered per occasion of testing.

These findings differ considerably from those of the earlier BEACH study that investigated change in GPs' pathology ordering between 1998 and 2001, which found that GPs' ordering increased due to an increase in the number of tests ordered per tested problem, without any change in likelihood of testing.⁴⁵ The other two factors did not contribute to the national increase in total volume of pathology ordered in general practice from 1998 to 2001. This suggests that the behaviour of increased likelihood of ordering pathology tests has occurred since that time.

From the perspective of managing financial outlays associated with pathology testing, the increase in occasions of testing has a greater impact than the number of tests ordered on each occasion of testing.¹³³ This is because the majority of the costs associated with laboratory testing are labour costs. Each additional test ordered on a single occasion contributes a relatively small additional cost burden due to the high use of automated machinery in the pathology sector.¹³⁴

The extrapolations used in this chapter incorporate the increases in the three factors to estimate the national effect of change. When extrapolated, the BEACH data

suggest that in 2006–08, compared with 2000–02, there was a national increase of about 53% in the volume of tests/batteries ordered by GPs. In contrast, the volume of Medicare-funded pathology items increased by 54% from 2000–01 to 2007–08.⁵ However, this does not accurately reflect changes in GP pathology ordering behaviour, as only 70% of the MBS pathology items are generated by GPs (E Wilson, personal communication, March 2011). Further, only three pathology item numbers can be claimed per episode of ordering (due to episode coning), and multiple tests may be included in each MBS item.²³ While BEACH data reflect actual GP orders for pathology tests, it is likely that it under-estimates the true number of pathology tests/batteries ordered, as there is only space for up to five tests/batteries to be recorded per encounter (see 'strengths and limitations' below).

Individual tests ordered by GPs

The majority of pathology ordered by GPs' was generated by a relatively small number of tests-the 22 most common tests accounted for 86% of all GP-ordered tests in 2000–08. This finding aligns with that reported in earlier studies: the 20 most common tests accounted for 80% of all pathology ordered by GPs in Australia in 1998⁸⁹ and in the Netherlands in 1996–97;⁷⁹ and, similarly, in the UK in 1997–98 the 28 most common test types accounted for 95% of GPs' pathology ordering activity.⁸² In the current study, there were significant increases between 2000–02 and 2006–08 in GPs' ordering rates for the majority of the commonly ordered individual tests; only three individual tests showing no change. The size of the increase varied considerably between the individual tests, from a marginal increase of 8% in the ordering rate of urine M,C&S tests to an increase of 200% in the order rate of vitamin B12 tests. GPs' ordering rate of the five most commonly ordered tests increased by between 26% and 55%. Despite these increases, the types of tests ordered by GPs stayed relatively constant between 2000–02 and 2006–08 and they accounted for a similar proportion of total pathology ordered over time, suggesting that the majority of the increase in GPs' pathology ordering reflected an increase in the ordering of common tests rather than the emergence of new tests.

The problems generating the highest volume of tests

GPs' management of a relatively small group of clinical problems accounted for the majority of pathology ordered by GPs. The 22 problems investigated in this study accounted for 53.4% of all problem–pathology links.

The most common problems for which GPs ordered pathology tests stayed relatively constant from 2000–02 to 2006–08. However, GPs' pathology ordering for these 22 problems increased. Extrapolation suggests they accounted for 59% of the national increase in pathology ordered in general practice. This national increase was due to: increases in GPs' pathology ordering in the management of the problems; changes in the GP workload accounted for by these problems; and the increase in number of GP encounters nationally.

It is important to consider the influence of GP workload in combination with GPs' pathology ordering behaviour. The example of 'menopausal complaint' demonstrates why these aspects must be considered together. GPs' pathology ordering in the management of menopausal problems increased significantly. Specifically GPs were more likely to initiate pathology testing for 'menopausal complaint' in 2006–08 than in 2000–02. In isolation this result would suggest that there would be an increase in the volume of pathology associated with the management of menopausal complaints. However, due to a significant decrease in the management rate of menopausal complaint problems, the opposite was true.

The change in management rate was likely triggered by the 2002 publication of results from the Women's Health Initiative study that indicated adverse effects were associated with use of hormone replacement therapy (HRT).¹³⁵ Following this finding, use of HRT declined. As HRT was the principal method of managing menopausal complaints, this led to the decrease in management rate of menopausal complaints at GP–patient encounters. The total decrease in the management rate of menopausal complaints was a greater influence than the increase in likelihood of GPs' ordering pathology. This is a good example of how a change in evidence influences volume of pathology ordered in general practice.

Menopausal complaint was the only problem for which a decrease in pathology ordering was found. Generally speaking in the absence of a change in GPs' ordering behaviour, there was an increase in volume of tests generated by each problem due to the national increase in GP workload (as demonstrated by the increased number of encounters over time) with or without a contributing increase in GPs management rate of the individual problem. Hence for each of the seven problems for which GPs' pathology ordering behaviour did not change, there was still a national increase in total number of tests ordered. I have described the increasing volume of pathology tests/batteries ordered by GPs as being related to changes within three factors (total GP workload, management rates of problems and GPs' pathology ordering behaviour in the management of these problems). Many potential causes for these changes are discussed in the literature, and the influence of some of these causes may be seen in the results of this chapter. Some causes of change will be beyond the control of GPs, while others may contribute to changes in GPs' pathology ordering behaviour. The quality or appropriateness of a change also varies, it may reflect 'good' clinical practice (for example, in response to new evidence) or 'bad' (for example, defensive testing). Examples of causes of change in GPs' pathology ordering are given below.

The publication of new guidelines or changes to existing guideline recommendations relating to pathology testing have the potential to change GPs' pathology ordering. Such changes are often made in response to emergence of new evidence.

The increase in the number of national GP encounters is largely beyond the control of GPs. Australia's ageing population will have contributed to the past increase in number of GP encounters, as rates of attendance increase with age.¹³⁶ Over the period of 2000 to 2008 the proportion of the population aged 65 years or more increased from 12.4% to 13.2%,¹⁵ and this is expected to continue. By 2050 it is projected to be 22-24%.¹³⁷ The influence of population ageing on future growth in pathology ordering is investigated in Chapter 8.

Health policy initiatives have been associated with an increased volume of pathology tests.^{26,27,29} Depending on their content and design, these initiatives have the potential to influence pathology ordering by changing GP workload and/or GPs' pathology ordering behaviour. How such initiatives are targeted (e.g. to the GP, or to the patient) will also influence whether GPs have any control over the changes in pathology that may be generated by the initiatives. For example, the 'Strengthening Medicare' package introduced in 2004–05 is a population-based initiative that aimed (among other things) to improve access to GP services.¹⁸ It was regarded as successful in achieving this aim and contributed to the increase in GP workload over the period investigated in this study. The content of the initiative was unlikely to have caused any change in GPs' pathology ordering behaviour, but the growth in workload it generated was accepted by the Government as contributing to growth in Medicare pathology outlays between 2004 and 2009.²⁶

The Practice Incentives Program (PIP) is a national initiative that provides a series of financial incentives for specific activity in general practice. PIP incentives were introduced for diabetes (in late 2001) and cervical screening (in late 2000). These incentive payments were made for specific outcomes which required pathology testing and recall systems within general practice to meet these targets.¹³⁸ These may have contributed to both the increased management rate and the increased pathology ordering for these problems that were reported in this study.

The broad range of problems for which GPs ordered pathology tests, illustrates the different clinical purposes for which GPs order pathology tests. For example, the majority of pathology ordered in the management of chronic problems such as hypertension and T2D is likely to be for monitoring purposes, whereas testing in the management of weakness/tiredness problems is likely to be for diagnostic purposes, and that ordered in the management of health checks and female genital check-ups for screening purposes.

Most of the problems I have investigated in this chapter reflect diseases or symptoms. Two notable exceptions are problems labelled as 'blood test' and 'abnormal test results'. Increases in workload and pathology ordering related to management of these problems may represent a change in GPs' process of care when ordering tests and giving results at GP–patient encounters. This study spans a period of workforce crisis and financial stress in general practice.¹³⁹ Discontent with level of Medicare rebates led to falls in bulk-billing rates in the early 2000's.⁵ It is possible that GPs changed their process of managing blood tests and results to increase income from this management by always seeing the patient to prepare the blood test order and provide results.

Also occurring at around the same time was the December 2001 introduction of privacy legislation that covered, for the first time, patient information collected, used and communicated by GPs.¹⁴⁰ The legislation stated the need to ensure the identity of the patient prior to communicating any sensitive information. Telephone communication may make this difficult. It is now not uncommon for general practices to have policies precluding communication of test results over the telephone. Almost half of Australians surveyed in the 2008–09 Patient Experience Survey who had had a pathology test, received the test results at a follow-up appointment.¹⁴¹

The increase in the management rate of abnormal test results may reflect an increase in the number of false positive test results. False positive results are a function of the reference range used to determine whether a result is 'normal'. When a 95% reference range is used to test a normal sample, there is a 95% chance that the result will be normal and a 5% chance that the result will be inappropriately reported as abnormal.¹⁴²⁻¹⁴⁴ The current study demonstrates that the number of tests ordered on each occasion of testing is increasing. As the number of tests ordered increases, so too does the probability that at least one false positive result will be reported. Most GPs are aware of the likelihood of false positive results and manage these intuitively. However, it is possible that medicolegal concerns may be contributing to the change in GPs' pathology ordering behaviour in the management of abnormal test results. GPs' awareness of medicolegal issues was high throughout the period of this study due to the political and media coverage of the 'medical indemnity crisis' which followed the collapse of the general insurer HIH in 2001 and the 2002 voluntary liquidation of Australia's largest medical defence organisation. The causes of the crisis were complex, and involved poor regulation of defence organisations, insufficiencies in their financial arrangements, and an increased number of medicolegal claims.^{145,146} National data on medicolegal claims involving GPs are only readily available from 2007–08 onward. Of the new claims initiated in 2007–08, general practice was the speciality that generated the highest proportion (21%). This is due to the high number of services provided by GPs. The most common type of claim made regarding GPs was related to 'diagnosis' (26.4% of all claims against GPs) which includes missed, delayed or incorrect diagnosis. As pathology testing is frequently involved in the diagnostic process, it is conceivable that some of the change observed in GPs management of abnormal test results may be due to defensive behaviour. Further, medicolegal concerns have been frequently described as a reason for ordering pathology tests.9,26,28,36,147

The management rate and GPs' pathology ordering increased for both 'blood test' and 'abnormal test results' problems over the study period. Some of the many external factors that may have contributed to these changes have been discussed above.

Strengths and limitations

The strength of the investigation of pathology ordering associated with individual problems, is that the analytical approach used enabled separation of the influence of (a) the GP workload associated with the management of each problem and (b) the GPs' pathology ordering behaviour in the management of each problem. These data, together with national Medicare claims data for GP encounters have been used in the extrapolations to estimate the national increase in pathology ordering attributed to each problem.

There are limitations that should be considered when interpreting the extrapolations. Extrapolations are only estimates and are likely to provide an underestimate of the true 'GP workload' because they are made to Medicare-claimed GP services, not to the total number of GP encounters per year (which include those not charged for, and those paid by sources other than Medicare, for example, state governments, WorkCover, employers). Approximately 5% of GP encounters are funded by sources other than Medicare.¹²¹

Further the extrapolated estimates of pathology ordering are based on the problem– pathology links. Each test may be linked (by the GP) to more than one problem being managed at the encounter. Therefore, it is possible for a single pathology test/battery to be linked to more than one problem. In each of the data periods used in this chapter there were 3–4% more links than tests. Therefore the extrapolations may overestimate the number of tests attributed to the problem.

In BEACH there was also an increase over time in the number of encounters where the maximum number of pathology tests/batteries (five) were recorded, suggesting that the number of tests/batteries missed may be increasing due to lack of space on the recording form. Therefore, extrapolations may under-represent the number of tests ordered by GPs and the amount of change that has occurred.

Despite these limitations, the extrapolations provide an estimate of the contribution of each problem to the increase in the national volume of pathology generated from GP orders. This is the first time that such analyses have been attempted.

Conclusion

This chapter highlights the influence of increases in GP workload and change in GPs' pathology ordering behaviour on the total volume of pathology ordering. This study provides, for the first time, information on the problems under management

that contribute to the growth in GPs' pathology ordering. It also demonstrates that change in GPs' pathology ordering behaviour is not always a contributing factor to the total volume of pathology associated with an individual problem. Further, even when present, changes in GPs' pathology ordering behaviour are not the sole factor contributing to national increases.

GPs' ordering rates of 22 pathology tests, and testing in the management of 22 problems, generated the majority of the volume and growth in GPs' pathology ordering between 2000–02 and 2006–08.

The findings of this chapter are used as the basis for selection of problems for the evaluation of the appropriateness of GPs' pathology ordering in the next chapter. The criteria used to select problems consuming increasing amounts of pathology resources were: (a) contribution to a high volume of total pathology ordering; (b) GPs' management has been characterised by changes in their pathology ordering behaviour. Hence, of the 22 problems investigated in this chapter, 10 were identified priorities for investigation of appropriateness. These were: hypertension, T2D, lipid disorders, health checks, weakness/tiredness, overweight/obesity, depression, menstrual problems, IHD, and abdominal pain (see Chapter 5).

The increases in GPs' pathology ordering described in this chapter do not provide any indication of the appropriateness of this ordering, and are likely to reflect both 'good' and 'bad' behaviours. The investigation of appropriateness of GPs' pathology ordering is the focus of Chapter 5.

5 GP pathology testing versus recommended testing

The results of the previous chapter confirmed that GPs' pathology ordering in the management of all problems increased significantly over time. However, investigation of the problems generating the highest volume of pathology ordering, demonstrated that GPs pathology ordering behaviour did not change for all problems. Further, for the problems where GPs' ordering increased, the type of ordering behaviour that changed differed across problems. While discerning whether GPs' pathology ordering behaviour changed, the research described in Chapter 4 did not investigate the types of tests nor the appropriateness of tests ordered in the management of each of these problems.

5.1 Objectives

In this chapter I will investigate the types of tests ordered by GPs for selected problems and evaluate the extent to which GPs' pathology ordering for these selected problems aligns with recommendations for pathology testing made in national and international guidelines and other sources of guidance for the management of selected problems. Secondary objectives are:

- to identify whether any changes over time that occurred in pathology ordered for selected problems reflect a change to be 'more' or 'less' in line with recommendations.
- to evaluate the quality of guidance for pathology testing available and identify areas in which guidance needs to be improved (such as areas where guidance is lacking or inconsistent).

5.2 Background

The increased rate of pathology testing and associated cost increases has stimulated much research and debate into the reasons for the increased demand. Numerous factors have been described in the literature as contributing to the increase.^{9,26-30} One of the aspects of the debate is whether these factors are contributing to increases in appropriate or inappropriate pathology ordering.

Several authors have implied that inappropriate pathology testing is widespread, and that the continued increase in pathology testing reflects inappropriate ordering.^{8,9,12,28,29,58,59} The Carter report (investigating pathology ordering in the UK) noted that several 'witnesses' reported widespread unnecessary testing but found there were "no robust studies which verify this statement."⁸ However, it is relatively safe to conclude from the published literature that there is a degree of inappropriate or unnecessary pathology ordering by GPs and other clinicians.^{11,52,76,85,96}

Concern about appropriate ordering is not new, having been discussed for several decades.^{28,42,148-150} What is unclear is the proportion of pathology testing that is inappropriate, and whether inappropriate testing is increasing.

Defining and measuring appropriateness of pathology ordering is difficult. In their review, van Walraven and Naylor⁵² found huge variability in the volume (5–95%) of tests ordered that were deemed 'inappropriate' in studies. This variation was due to the different clinical scenarios in the studies reviewed, and the diverse and often flawed methods used to evaluate appropriateness. Although most of the reviewed studies were not conducted in general practice, the review highlights the difficulty of defining appropriateness. This is echoed by Smellie, who noted that the appropriateness of pathology tests "depends on the clinical context of the patient, the severity of the disorder, administrative necessities and many other factors."⁵⁴

To date there have been no large studies attempting to define the amount of inappropriate pathology testing in general practice. Much of the evidence used to support the presence of inappropriate pathology ordering by GPs is drawn from studies reporting results of interventions that aim to improve appropriateness of their pathology ordering.^{11,58,74-77,79-87,95-97} Guidelines are usually the basis of these interventions. However, as discussed in Chapter 2, these studies are often targeted to 'problem' areas of pathology ordering where the intervention is most likely to be cost-effective by achieving a reduction in use of pathology testing. As such they are conducted in very focussed clinical situations targeting a limited number of pathology tests^{58,80-84} or pathology testing for a limited number of clinical problems.^{11,75,77,85-87} Only a handful of interventions target groups of tests and/or problems that reflect the breadth of clinical situations in which GPs' order pathology tests.^{76,79,82,88,95}

Success of these intervention studies is usually judged on the basis of a reduction in pathology testing following the intervention. The assumption is made that the reduction is a reduction in inappropriate tests. However most studies do not assess the appropriateness of the reduction in testing. A reduction (even when only inappropriate tests are reduced) does not enumerate the proportion of tests that are inappropriate as there may be situations where the test was used inappropriately but the intervention was not successful. This is illustrated by van Wijk et al. who introduced a decision support tool to support appropriate pathology ordering by GPs,⁷⁹ and then assessed the appropriateness of GPs' ordering after the intervention.⁸⁸ The intervention reduced ordering,⁷⁹ but the subsequent assessment demonstrated that only 40% of pathology ordering sets for 12 indications were compliant with guidelines.⁸⁸ Van Wijk et al. did not measure whether the reduction in ordering represented improved appropriateness, but the subsequent assessment demonstrates that there is potential for further improvement in GPs' ordering for the 12 indications investigated. Thus a 'successful' intervention is not a reflection of the underlying appropriateness of GPs' ordering.

The extent to which results of these studies can be generalised is limited by the diversity of the circumstances being investigated. Other authors have misinterpreted findings of such studies extrapolating them beyond the context of the clinical situation in which the study was designed, resulting in statements such as "between 25 and 40% of all tests sent to the laboratory are unnecessary".²⁸

There is currently insufficient evidence to determine the proportion of GP-ordered pathology tests that could be considered inappropriate in Australia. However, it is important to establish whether inappropriate pathology ordering is an issue, and if it is, the extent of this inappropriate ordering, before investing resources in interventions to improve pathology ordering. Whether such interventions should be targeted to specific clinical problems, types of tests or other areas of testing (such as, purpose of testing e.g. diagnostic versus monitoring) also needs to be determined. This chapter aims to evaluate the proportion of the tests ordered by GPs that are appropriate, for each of a subset of clinical problems that are generating a high volume of pathology ordering in general practice.

5.3 Method

5.3.1 General practice data

The data used in this chapter were collected in the BEACH study over 8 years from April 2000 to March 2008 (referred to as 2000–08). Changes over time were measured by comparing results from the first two years of the study period (April 2000 to March 2002) with those from the last two years (April 2006 to March 2008) (referred to as 2000–02 and 2006–08 respectively).

In BEACH, all pathology tests recorded are linked (by the GP) to the specific problem(s) for which they were ordered. The analysis of pathology ordering in this chapter is based on these problem–pathology links (see Section 3.2.11). I used unweighted data in the analysis because multiple data years were combined (see Section 3.2.10).

5.3.2 Selecting the problems for investigation

A problem was considered for investigation if GPs' pathology ordering had increased for that problem between 2000–02 and 2006–08, and if:

- it was considered a National Health Priority Area, and/or
- the problem generated a high volume of total pathology ordering, and/or
- pathology ordering was commonly used by GPs in the management of the problem.

The eight National Health Priority Areas in Australia (in 2008) were: arthritis and musculoskeletal conditions, asthma, cancer control, cardiovascular health, diabetes mellitus, injury prevention and control, mental health, and obesity.¹⁵¹

A total of 22 problems generating high volumes of pathology ordering by GPs were identified in Chapter 4. Of these, I identified 10 as priorities for investigation based on the criteria above. These were: hypertension, Type 2 diabetes, lipid disorders, health checks, weakness/tiredness, overweight/obesity, depression, menstrual problems, ischaemic heart disease, and abdominal pain. The first six problems from this list were investigated because they generated the highest volume of pathology tests ordered by GPs and/or were areas of current policy interest.

Defining the problems

The six problems investigated are defined below. The list of ICPC-2 or ICPC-2 PLUS codes included for each problem is provided in Appendix 6.

Hypertension

Hypertension includes problems recorded by GPs that were classified as uncomplicated or complicated hypertension in ICPC-2 (codes K86 and K87). It excludes hypertension in pregnancy (pre-eclampsia) because this condition is not managed in the same way.

Type 2 diabetes

Type 2 diabetes includes problems recorded by GPs that were classified as 'noninsulin dependent diabetes' (ICPC-2 code T90). In line with ICPC-2 inclusion criteria, this includes diabetes mellitus not specified by the GP as Type 1 or Type 2.

Lipid disorder

Lipid disorder includes problems recorded by GPs that were classified as 'lipid disorder' in ICPC-2 code T93.

Health check

'Health check' was investigated when recorded as a clinical problem at encounters with patients aged 15 years or more. 'Health check' problems are a group of ICPC-2 PLUS (terminology) codes classified as check-ups in the general and unspecified chapter of ICPC-2 (codes A30 and A31). The 'health check' problems can be considered preventive in nature but whether each is for primary, secondary or tertiary prevention is not recorded in the BEACH data set. The age limit (of 15 years and over) was selected because check-ups for children rarely involve pathology testing. Check-ups that were likely to be privately funded (such as employment check-ups) were excluded because any pathology generated in the management of these checks would also be privately funded. This is one of the few areas in general practice that generates a substantial amount of privately funded pathology. The majority of pathology ordered by GPs is publicly funded through the MBS and this is the focus of this thesis.

Check-ups related to a specific health condition were also excluded because presence of this condition would affect pathology test selection and the resources GPs use to guide their test selection. In addition, the specific condition usually indicated that the check-up was not preventive.
Weakness/tiredness

Weakness/tiredness was investigated when recorded by the GP as the clinical problem under management. It includes problems classified as 'weakness/tiredness' in ICPC-2 (code A04). This does not include all patient presentations of weakness/tiredness. In many cases GPs were able to assign a presentation of weakness/tiredness to a more specific diagnosis based on other information (e.g. patient history). The problem label of weakness/tiredness indicates that a more precise diagnostic label could not be assigned to the problem at the time of the encounter.

Overweight/obesity

Overweight/obesity includes problems labelled by the GP as 'obesity' or 'overweight' (ICPC-2 codes T82 and T83) at encounters with adult patients (aged 18 years and over). This does not represent all encounters with overweight/obese adult patients, only those at which overweight or obesity was managed as a specific problem at the encounter. It also does not include GP management of overweight/obesity as part of the management of other morbidity (e.g. weight/diet advice in the management of hypertension).

The method(s) used by the GP to define the problem as obesity or overweight is not known. It may be clinical opinion, calculation of BMI, waist measurement, weight measurement, or a combination of the above indications.

Overweight and obesity were combined to provide a larger sample with greater statistical power than obesity alone. In addition, many of the available guidance documents were for overweight/obesity. If a patient is overweight they are at considerable risk of progressing to obesity. Hence the WHO regards overweight as preobesity.¹⁰⁹

The analysis of pathology ordering for overweight and obesity was limited to adults because guidance provided for management of overweight/obesity in children and adolescents differed from that for adults. Further, most encounters (95.1%) involving the management of overweight/obesity were with adult patients.

5.3.3 Identifying the guidelines and other sources of guidance

National and international guidelines, and other published guidance documents for the management of each selected problem were identified using literature searches in Medline and internet-based search engines and databases (e.g. Google, National Guideline Clearinghouse, TRIP Database). The most recently available international and national guidelines and other Australian sources of GP guidance for each selected problem published prior to May 2009 were reviewed.

Search terms used included *disease management* and *practice guidelines* and terms related to the specific problem. For example, for overweight/obesity search terms included *overweight*, *obesity*, and *body weight*. Additional websites and databases were found by consulting 'related links' on websites and 'related articles' links within electronic databases.

Peak bodies that develop and disseminate guidelines were identified, and their published guidelines examined for any relating to the selected problems. Reference lists in documents were also reviewed to find other relevant publications.

Any guidelines or guidance documents that were largely based on another guidance document were excluded, and the source guidance document reviewed. Only guidance published in English or with an English translation available was included. The guidance documents reviewed in this study included guidelines and other types of published guidance documents (such as the pathology manual produced by the RCPA). All types of guidance documents were reviewed because GPs use multiple sources when looking for clinical information, including guidelines and other published guidance documents.¹⁵² The review also sought to evaluate the extent to which pathology test recommendations were consistent among guidance documents. Therefore it was considered appropriate to review all published sources of guidance available to GPs.

Some guideline authors have also produced abbreviated GP guides and fact sheets in an effort to overcome identified barriers to guideline use by GPs (e.g. length of guidelines).¹⁵³ These guides were compared with the matching guidelines to identify mismatches (e.g. differences in level of recommendations, and omissions).

87

Throughout the remainder of this chapter the term 'guidance documents' is used to refer to both guidelines and other sources of guidance.

5.3.4 Determining recommended pathology testing

Guidance documents were reviewed to identify pathology tests recommended in the management of the specific problems. Level of support was determined for each pathology test/battery of tests that accounted for at least 1% of pathology tests ordered for the selected problem.

When there were differences among guidance document recommendations for an individual test, the level of support for the test was determined by agreement in the majority of guidance documents.

Tests were classified as:

- **supported**: those that the guidance documents indicated have a role in any phase of management of the problem. For example, tests only recommended for diagnosis were considered supported. In addition tests that were recommended for a large specific group of patients were considered supported (e.g. all female patients).
- having **conditional support**: guidance documents indicated the test has a role in the management of the problem in certain circumstances (such as the presence of a risk factor).
- **unsupported**: tests that the guidance documents specifically stated should not be performed or tests not mentioned by guidance documents as having a role in the management of the problem.
- support unable to be determined: level of support in guidance documents was unclear for two groups of tests—multibiochemical analysis (MBA) and 'other chemistry' tests.

The MBA group includes two tests: the MBA test and the electrolytes and liver function test (E&LFT) (see Appendix 7). The MBA group potentially includes a large number of biochemical analytes[‡]. The MBS chemical analysis pathology item 66500 gives an example of the range of biochemical tests that could be included in a MBA.²³ The specific analytes measured in response to a GP's order for a MBA or E&LFT test varied between laboratories, therefore it was not possible to determine the exact tests included in the MBA group. When a ⁸⁸

guidance document recommended one or more biochemical tests that could be considered part of the MBA group (but did not mention the MBA or E&LFT tests) this is discussed in the relevant section.

The 'other chemistry' group includes a large number of individual chemistry tests that are not frequently ordered in general practice. The included tests are listed in Appendix 7. Where an individual test within the group accounted for more than 1% of pathology tests for an investigated problem it was evaluated and discussed in the relevant section.

5.4 Hypertension

5.4.1 Background

Hypertension is the most common cardiovascular condition in the population.¹⁵⁴ Cardiovascular disease was made a National Health Priority Area in 1996 due its' high burden of morbidity and mortality.¹⁵⁵ High blood pressure accounted for 7.6% of the burden of disease in Australia (in 2003).¹⁵⁶ In 1999–00, the prevalence of hypertension was 29% of Australians aged 25 years and over, and prevalence increased with age.¹⁵⁷

A 2005 SAND substudy of 9,156 patients estimated the prevalence of hypertension to be 23.3% of patients at encounters in general practice. After adjusting for frequency of attendance, the prevalence was 17.6% of the general practice patient population (i.e. patients who attend general practice at least once).¹⁵⁸

Hypertension is identified and managed because of the cardiovascular risk associated with high blood pressure.¹⁵⁹⁻¹⁶² Pathology testing does not have a role in the diagnosis of hypertension, but does have a role in identifying causes of secondary hypertension and detecting cardiovascular risk factors and disease.

5.4.2 Management rate in Australian general practice

In BEACH, hypertension was managed at 72,169 encounters by 7,489 GPs in 2000–08, at a rate of 9.2 per 100 encounters (Table 5.1). This equates to an estimated 9.3 million GP encounters per year where hypertension was managed in Australia. The vast majority (99.9%) of hypertension managed was uncomplicated hypertension (results not tabled), and there was no change in the management rate of hypertension between 2000–02 and 2006–08 (Chapter 4). In contrast, the management rate of 'new' hypertension problems increased significantly (by 24%), from 0.48 per 100 encounters in 2000–02 to 0.60 in 2006–08 (Table 5.1), indicating an increase in its diagnosis or detection rate.

	20	00008	20	000–02	20		
Variable	Number	Rate per 100 encounters (95% Cl) (<i>n</i> =784,300)	Number	Rate per 100 encounters (95% Cl) (<i>n</i> = 198,200)	Number	Rate per 100 encounters (95% CI) (n = 188,300)	Change
General practitioners	7,489		1,900		1,810		
Hypertension encounters	72,169		18,007		17,792		
Hypertension problems managed	72,171	9.2 (9.0–9.4)	18,007	9.1 (8.8–9.4)	17,793	9.5 (9.1–9.8)	_
New hypertension problems	4,237	0.54 (0.52–0.56)	958	0.48 (0.44–0.52)	1,131	0.60 (0.56–0.64)	↑

Table 5.1: Summary of hypertension data set, 2000–08, 2000–02 and 2006–08

Note: Data about hypertension problems managed are drawn from Chapter 4, Table 4.3. CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change, and — indicates no change.

5.4.3 Pathology ordered for hypertension

Pathology was ordered at a rate of 26.2 tests/batteries of tests per 100 hypertension contacts in 2000–08. One in ten hypertension problem contacts (10.2%) resulted in at least one pathology order. Once the decision to order pathology was made GPs ordered on average 2.56 tests/batteries per tested problem (Table 5.2).

As described in Chapter 4, the rate of pathology ordering per 100 hypertension contacts increased significantly between 2000–02 and 2006–08. This was due to significant increases in both the likelihood of pathology being ordered in the management of hypertension and the number of tests ordered per tested hypertension problem (Table 5.2).

	20	000-08	20	00–02	20	06-08	
Variable	Number	Per cent / Rate of hypertension problems (95% Cl) (n=22,938)	Number	Per cent / Rate of hypertension problems (95% Cl) (n=18,007)	Number	Per cent / Rate of hypertension problems (95% Cl) (n=17,793)	Change
Pathology (Rate per 100 hypertension problems)) 18,889	26.2 (25.3–27.1)	3,884	21.6 (20.0–23.2)	5,744	32.3 (30.3–34.2)	↑
At least one pathology order (Per cent of hypertension problems)	7,377	10.2 (9.9–10.6)	1,565	8.7 (8.1–9.3)	2,124	11.9 (11.3–12.6)	↑
Number of tests/ batteries per 100 tested hypertension problems		256.1 (251.8–260.4)		248.2 (239.5–257.0)		270.4 (262.5–278.4)	↑

Table 5.2: Summary of pathology ordering for hypertension, 2000–08, 2000–02 and2006–08

Note: Pathology ordering data from 2000–02 and 2006–08 are drawn from Chapter 4, Table 4.4. CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/ indicates a statistically significant change.

Types of pathology tests/batteries ordered

Table 5.3 shows the rate of pathology tests/batteries ordered for hypertension in 2000–08 by MBS groups and the most common individual types of tests ordered.

Chemistry tests were the group most often ordered (21.0 per 100 hypertension

contacts) and the most common were:

- lipid tests (5.8 per 100 hypertension contacts)
- EUC tests (5.3 per 100 contacts)
- glucose/glucose tolerance tests (2.9 per 100)
- LFTs (2.3) (Table 5.3).

Haematology tests (4.0 per 100 contacts), in particular FBC tests (3.6), were also commonly ordered in the management of hypertension (Table 5.3).

One-eighth (16.4%) of pathology tests were ordered for 'new' cases of hypertension. While the rate of pathology ordering for new cases was significantly higher than the average rate, the majority of tests ordered by GPs were for the ongoing management or monitoring of hypertension (Table 5.3).

	All hypert	ension problems	New	hypertens	ion problems
Pathology test	Number	Rate per 100 hypertension problems (95% Cl) (<i>n</i> =22,938)	Number	Per cent of test	Rate per 100 new hypertension problems (95% Cl) (n=1,421)
Chemistry	15,148	21.0 (20.2–21.7)	2,387	15.8	56.3 (52.6–60.0)
Lipids*	4,203	5.8 (5.6–6.1)	644	15.3	15.2 (13.9–16.5)
EUC*	3,836	5.3 (5.1–5.6)	528	13.8	12.5 (11.4–13.6)
Glucose/glucose tolerance*	2,119	2.9 (2.8–3.1)	348	16.4	8.2 (7.3–9.1)
LFT*	1,624	2.3 (2.1–2.4)	281	17.3	6.6 (5.8–7.4)
MBA*	1,237	1.7 (1.6–1.9)	207	16.7	4.9 (4.2–5.6)
TFT*	768	1.1 (1.0–1.1)	179	23.3	4.2 (3.6–4.8)
Other chemistry*	456	0.6 (0.5–0.7)	78	17.1	1.8 (1.4–2.3)
PSA*	270	0.4 (0.3–0.4)	45	16.7	1.1 (0.8–1.4)
HbA1c*	182	0.3 (0.2–0.3)	14	7.7	0.3 (0.2–0.5)
Haematology	2,917	4.0 (3.8–4.2)	528	18.1	12.5 (11.3–13.6)
FBC	2,564	3.6 (3.4–3.7)	480	18.7	11.3 (10.3–12.3)
ESR	261	0.4 (0.3–0.4)	40	15.3	0.9 (0.7–1.2)
Other tests NEC	462	0.6 (0.6–0.7)	74	16.0	1.8 (1.3–2.2)
Blood test	197	0.3 (0.2–0.3)	30	15.2	0.7 (0.4–1.0)
Microbiology	293	0.4 (0.4–0.5)	107	36.5	2.5 (2.0–3.0)
Urine M,C&S*	274	0.4 (0.3–0.4)	100	36.5	2.4 (1.9–2.9)
Other pathology groups	69		7	10.1	
Total pathology tests	18,889	26.2 (25.3–27.1)	3,103	16.4	73.2 (68.6–77.9)

Table 5.3: Rate of pathology test orders for hypertension by MBS pathology groups and most frequent individual test orders within each group, 2000–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. CI – confidence interval; EUC – Electrolytes, urea and creatinine; MBA – multibiochemical analysis; TFT – thyroid function tests; PSA – Prostate specific antigen; FBC – full blood count; ESR – erythrocyte sedimentation rate; M,C&S – microscopy, culture and sensitivity; NEC – not elsewhere classified.

Changes in the types of pathology tests/batteries ordered

Table 5.4 shows the most common pathology tests/batteries ordered for hypertension in 2000–02 and 2006–08. Listed below are the tests for which significant change in GPs' order rate occurred between 2000–02 and 2006–08 (listed in decreasing test rate order). There were significant increases in the order rate of:

- lipid tests—28% increase
- EUC tests—36% increase
- FBCs—64% increase
- glucose/glucose tolerance tests—35% increase

- LFTs—71% increase
- MBA tests—109% increase
- TFTs—75% increase
- PSA tests—200% increase
- 'other chemistry' tests—400% increase (Table 5.4). This was mostly due to increases in the rate of urinary albumin and albumin:creatinine ratio tests that are grouped within the 'other chemistry' group.

	:	2000–02		2006–08	
Pathology test	Number	Rate per 100 hypertension problems (95% Cl) (n=18,007)	Number	Rate per 100 hypertension problems (95% Cl) (<i>n</i> = 17,793)	Change
Chemistry	3,105	17.2 (15.9–18.6)	4,615	25.9 (24.3–27.6)	↑
Lipids*	970	5.4 (4.8–6.0)	1,220	6.9 (6.4–7.3)	♠
EUC*	816	4.5 (4.1–5.0)	1,084	6.1 (5.5–6.6)	♠
Glucose/glucose tolerance*	462	2.6 (2.3–2.9)	616	3.5 (3.1–3.8)	↑
LFT*	306	1.7 (1.5–1.9)	524	2.9 (2.6–3.3)	↑
MBA*	199	1.1 (0.9–1.3)	407	2.3 (2.0–2.6)	↑
TFT*	141	0.8 (0.6–0.9)	254	1.4 (1.2–1.6)	↑
Other chemistry*	39	0.2 (0.1–0.3)	174	1.0 (0.8–1.2)	↑
PSA*	39	0.2 (0.1–0.3)	110	0.6 (0.5–0.7)	↑
HbA1c*	36	0.2 (0.1–0.3)	55	0.3 (0.2–0.4)	—
Haematology	588	3.3 (2.9–3.6)	900	5.1 (4.6–5.5)	↑
FBC	501	2.8 (2.5–3.1)	811	4.6 (4.2–5.0)	↑
ESR	65	0.4 (0.3–0.5)	71	0.4 (0.3–0.5)	—
Other tests NEC	97	0.5 (0.4–0.7)	135	0.8 (0.6–0.9)	—
Blood test	39	0.2 (0.1–0.3)	64	0.4 (0.2–0.5)	—
Microbiology	74	0.4 (0.3–0.5)	78	0.4 (0.3–0.6)	_
Urine M,C&S*	67	0.4 (0.3–0.5)	72	0.4 (0.3–0.5)	_
Other pathology groups	20		16		
Total pathology tests	3,884	21.6 (20.0–23.2)	5,744	32.3 (30.3–34.2)	♠

Table 5.4: Rate of pathology test orders for hypertension by MBS pathology groups and most frequent individual test orders within each group, 2000–02 compared with 2006–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change, and— indicates no change

5.4.4 Guidance documents for hypertension

Г

Eleven guidance documents were reviewed for hypertension (Box 5.1).

Box 5.1: Guidance documents reviewed for hypertension										
Title	Year	Author	Abbreviated to							
Guide to management of hypertension 2008: assessing and managing raised blood pressure in adults ¹⁶³	2008	National Heart Foundation (NHF) of Australia	NHF							
Hypertension: management of hypertension in adults in primary care ¹⁶⁴	2006	National Collaborating Centre for Chronic Conditions and the British Hypertension Society. <i>National Institute for Health and</i> <i>Clinical Excellence (NICE) guideline</i> [United Kingdom]	NICE							
Seventh report of the Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood pressure (JNC 7) ¹⁶⁵	2004	United States Department of Health and Human Services, National Institutes of Health	JNC 7							
Hypertension in older people ¹⁶⁶	2001	Scottish Intercollegiate Guidelines Network (SIGN)	SIGN							
Statement on management of hypertension ¹⁶⁷	2003	World Health Organization (WHO) and International Society of Hypertension (ISH)	WHO & ISH							
Guidelines for the management of arterial hypertension ¹⁶⁸	2007	European Society of Hypertension (ESH) and of the European Society of Cardiology (ESC)	ESH & ESC							
Health care guideline: hypertension diagnosis and treatment ¹⁶⁹	2006	Schwartz G, Canzanello V, Woolley A, Miller T, O'Connor P, Klein D et al. <i>Institute for Clinical Systems Improvement</i> (ICSI) guideline [United States of America]	ICSI							
Canadian Hypertension Education Program recommendations for the management of hypertension: parts 1 ¹⁷⁰ and 2 ¹⁷¹	2007	Canadian Hypertension Education Program (CHEP)	CHEP							
RCPA manual, hypertension section ¹⁷²	2004	The Royal College of Pathologists of Australasia (RCPA)	RCPA							
Murtagh's general practice, hypertension chapter ¹⁷³	2007	Murtagh J [Australia]	Murtagh							
Patient presentations in general practice: blood pressure check up and hypertension review sections ¹⁷⁴	1999	Steven I [Australia]	Steven							

5.4.5 Extent of alignment between GP testing and guidance documents

In Table 5.5, the frequently ordered pathology tests/batteries for hypertension are categorised by level of support in the guidance documents listed in Box 5.1. The key explaining the colours used in the table is provided before Table 5.5.

Supported tests

The four types of tests that were recommended in the majority of guidance documents were lipids, EUC, FBC and glucose/glucose tolerance (Table 5.5). Together these four tests accounted for 67.4% of tests/batteries ordered by GPs for hypertension (Table 5.6). The order rate of these tests increased significantly between 2000–02 and 2006–08 (Table 5.4).

Lipid and glucose testing were almost unanimously recommended (Table 5.5), most often as part of the initial assessment of newly diagnosed hypertension. The reason for testing lipids was to determine patient's cardiovascular risk profile, and for glucose testing it was to detect undiagnosed diabetes. Ongoing testing was recommended in three guidance documents (JNC 7, ¹⁶⁵ Steven, ¹⁷⁴ CHEP^{170,171}) and in one other it was referred to in a footnote of a flowchart (NHF¹⁶³).

There was unanimous agreement among guidance documents for EUC testing (predominately creatinine, potassium and sodium testing), primarily to assess kidney function both as end (or target) organ damage (in ongoing management) and kidney disease as a cause of secondary hypertension (in the initial assessment). EUC was also recommended in the monitoring of response to medications, specifically potassium monitoring in the management of diuretics, and sodium and creatinine in the monitoring of angiotensin converting enzyme (ACE) inhibitors and angiotensin receptor blockers/angiotensin II receptor antagonists (see medication monitoring discussion in Section 5.4.6).

The FBC (also referred to as a 'complete blood picture' or 'full blood evaluation') is a battery of tests. GPs routinely order a FBC rather than the individual analytes or components within it. Most of the hypertension guidance documents reviewed recommended that one or more of the analytes in a FBC should be tested (Table 5.5). Haematocrit and/or haemoglobulin tests (two analytes within a FBC) were commonly recommended, but the rationale for ordering these tests was not given.^{163,165,168,169,173} In contrast, the CHEP guideline specifically recommended against the use of these tests as they did not aid in investigation or monitoring of hypertension.¹⁷⁰

The SIGN guideline recommended FBC testing because the mean cell volume (one part of a FBC) may indicate excess alcohol consumption, and such consumption is associated with resistance to antihypertensive therapy and to risk of stroke.¹⁶⁶

Alcoholism/excess alcohol consumption was also mentioned as a potential cause of secondary hypertension in other guidance documents.^{164,165,169-172} However, with the exception of SIGN, no other guidance documents recommended pathology tests to identify excess alcohol consumption.

In all guidance documents, when FBC testing was recommended it was only as part of the initial assessment.

Tests with conditional support

There were four types of tests that had conditional support or mixed levels of support: LFT, TFT, urine M,C&S, and albumin (urinary albumin or albumin:creatinine ratio) tests (Table 5.5) and together they accounted for 15.3% of tests ordered by GPs for hypertension (Table 5.6). Between 2000–02 and 2006–08 there were significant increases in the order rate of LFT, TFT and 'other chemistry' tests (which includes albumin tests) and no change in the rate of urine M,C&S (Table 5.4)

LFT was recommended in two guidance documents (Table 5.5). The NHF guideline recommended LFT as part of the investigations of a newly diagnosed patient, but the reason for testing was not given.¹⁶³ The SIGN guideline recommended testing of Gamma glutamyl transpeptidase (one part of LFT) as a possible indicator of alcoholism/excess alcohol intake.¹⁶⁶ As discussed above, many guidelines discussed alcoholism/excess alcohol intake as a cause of secondary hypertension.^{164,165,169-172}

Thyroid disease was mentioned as a possible cause of secondary hypertension in six guidance documents (Table 5.5). Testing was only recommended if thyroid disease was suspected (e.g. clinical suspicion or abnormal physical examination) as part of the initial assessment.

Two guidance documents recommended urine M,C&S in initial investigation of newly diagnosed hypertension to identify possible urinary tract infection.^{168,174} NHF recommended it as a follow-up test if urinalysis was abnormal.¹⁶³ RCPA recommended it as part of 'further investigation' if necessary.¹⁷²

Testing for microalbuminuria (using urine albumin or albumin:creatinine ratio tests) was recommended in the majority of guidance documents but for different reasons. Some recommended testing as part of the routine initial assessment, ^{164,166,167,174} while others recommended it if the patient had an abnormal urinalysis test^{163,168,169} or the

patient had diabetes.^{170,171,173} JNC 7 stated that urine albumin or albumin:creatinine ratio tests were optional unless diabetes or kidney disease is present, in which case it was recommended annually.¹⁶⁵

Support unable to be determined

It was not possible to determine whether MBA and 'other chemistry' tests (excluding urine albumin tests) were recommended in the guidance documents (see Section 5.3.4). These two tests accounted for 7.8% of pathology ordered for hypertension (Table 5.6).

With the exception of urinary albumin and albumin:creatinine (split from the 'other chemistry' group and discussed above), the remainder of the 'other chemistry' group was classified as having unclear guidance.

The MBA test was not recommended in any guidance documents. However, some tests that may be considered part of the MBA, such as EUC and LFT, were partially or completely supported in the management of hypertension (as discussed above).

Unsupported tests

The guidance documents did not recommend testing of PSA, erythrocyte sedimentation rate (ESR), HbA1c and unspecified 'blood tests' (Table 5.5). These four tests together accounted for 4.8% of pathology ordered by GPs in the management of hypertension (Table 5.6). PSA was the only test within this group to increase significantly between 2000–02 and 2006–08 (Table 5.4).

Other tests mentioned in the guidance documents

Other tests mentioned in guidance documents were urinalysis, uric acid, calcium and CRP. These were ordered rarely, each accounting for less than 1% of pathology ordered for hypertension.

Urinalysis was recommended in nine of the eleven guidance documents as part of the initial investigations for hypertension to assess kidney function.^{163,165,166,168-174}

Uric acid and calcium testing were rarely ordered as individual tests. However it is possible that they would be tested as part of a MBA. Testing urate/uric acid was recommended in four guidance sources. It was recommended as a measure of kidney function,^{168,173} as part of the initial assessment¹⁶³ or as a baseline when initiating a diuretic.¹⁶⁶ Calcium testing was recommended as part of the initial review in three guidance documents as an indicator for hyperparathyroidism.^{165,166,169}

The CRP, a non-specific measure of inflammation¹⁷⁵ was recommended in Murtagh.¹⁷³ The rationale for ordering this test was not provided. While CRP was rarely ordered, GPs did order the ESR (an alternative test to the CRP). ESR was not recommended in guidance documents (see above) (Table 5.5).

Key to Table 5.5

Colour	Description
	The document specifically recommended this test. Any notes within the cell indicate further detail. For example, a specific disease to test for within subset of patients; a specific test within a group.
	The document stated that this test should be considered. Any notes within the cell indicate further detail (e.g. a specific test to consider)
	Unable to determine guidance (see Section 5.3.4).
	The document specifically stated not to do this test. Additional information is supplied if certain conditions apply (e.g. specific clinical situations).
	Guidance document does not mention this test

Pathology test	Per cent of hypertension pathology (<i>n</i> = 18,889)	NHF ¹⁶³	NICE ¹⁶⁴	JNC 7 ¹⁶⁵	SIGN ¹⁶⁶	WHO & ISH ¹⁶⁷	ESH & ESC ¹⁶⁸	ICSI ¹⁶⁹	CHEP ^{170,171}	RCPA ¹⁷²	Murtagh ¹⁷³ S	Steven ¹⁷⁴
Lipids*	22.3											
EUC*	20.3											
FBC	13.6	Hb		Haematocrit			Hb & haematocrit	Haematocrit			Hb & haematocrit	
Glucose/glucose tolera	ance* 11.2											
LFT*	8.6				Gamma GT							
MBA*	6.6											
TFT*	4.1											
Urine M,C&S*	1.5	If abnormal urinalysis										
PSA*	1.4											
ESR	1.4											
Other chemistry (excluurinary albumin/ albumin:creatinine rati	ıding 1.3 o)											
Urinary albumin/ albumin:creatinine rati	o 1.1	lf abnormal urinalysis					lf abnormal urinalysis	lf abnormal urinalysis	Diabetes		Diabetes	
HbA1c*	1.0											
Blood test	1.0											

Table 5.5: Summary of guidance recommendations by most frequent individual test orders for hypertension, 2000–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Hb - haemoglobulin; also see Abbreviations. Any notes within the coloured cells are described in detail in Section 5.4.5.

Evaluation of GP pathology ordering against guidance documents

Table 5.6 provides a summary of the individual tests recorded in BEACH and the level of support provided in the guidance documents for each. Of the tests/batteries ordered by GPs for hypertension in 2000–08:

- 67.4% were supported
- 24.2% had conditional support or support could not be determined
- 4.8% were not supported by the guidance documents.

The individual tests/batteries listed in Table 5.6 accounted for 96.3% of pathology ordered for hypertension because only the most commonly ordered tests were evaluated.

Pathology test supported by guidance	Number	% of all pathology for hypertension
Supported	12,722	67.4
Lipids*	4,203	22.3
EUC*	3,836	20.3
FBC	2,564	13.6
Glucose/glucose tolerance*	2,119	11.2
Conditional/unclear support	4,567	24.2
LFT*	1,624	8.6
MBA*	1,237	6.5
TFT*	768	4.1
Other chemistry* (excluding urine albumin/albumin:creatinine ratio)	248	1.3
Urinary albumin/Albumin:creatinine ratio	208	1.1
Urine M,C&S*	274	1.5
Unsupported	910	4.8
PSA*	270	1.4
HbA1c*	182	1.0
ESR	261	1.4
Blood test	197	1.0
Subtotal (n, % of total tests)	18,199	96.3
Total pathology tests	18,889	100.0

Table 5.6: Summary of support for GP pathology ordering for the most frequentindividual test orders for hypertension, 2000–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. See Abbreviations.

In 2000–08, GPs ordered supported pathology tests at a rate of 17.6 tests/batteries per 100 hypertension problems, followed by tests with conditional support (6.0 per 100),

and unsupported tests (1.3). Tests ordered for hypertension but not evaluated were ordered at a rate of 1.3 per 100 hypertension problems. The rate at which GPs ordered tests in all 'level of support' groups increased significantly over time (Table 5.7).

	2000–08	2000–02	2006–08	
Level of support	Rate per 100 hypertension problems (95% CI) (<i>n</i> = 72,171)	Rate per 100 hypertension problems (95% CI) (<i>n</i> = 18,007)	Rate per 100 hypertension problems (95% Cl) (<i>n</i> = 17,793)	Change
Supported	17.6 (17.0–18.3)	15.3 (14.0–16.5)	21.0 (19.6–22.3)	↑
Conditional/unclear support	6.0 (5.7–6.3)	4.2 (3.8–4.6)	8.0 (7.4–8.6)	↑
Unsupported	1.3 (1.2–1.4)	1.0 (0.8–1.2)	1.7 (1.4–1.9)	↑
Not evaluated	1.3 (1.2–1.4)	1.1 (0.9–1.3)	1.6 (1.4–1.9)	↑

Table 5.7: Rate of pathology ordering for hypertension by level of support, 2000–08,2000–02 and 2006–08

Note: CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change.

GPs ordered only pathology tests that were supported or conditionally supported at 8.7% (95% CI: 8.4–9.0) of hypertension problems in 2000–08. A further 1.2% (95% CI: 1.1–1.3) of hypertension problems involved at least one unsupported test, and 90.1% (95% CI: 89.7–90.4) of problems involved either no pathology tests (89.8%) or tests that were not evaluated (0.3%) (results not tabled).

Between 2000–02 and 2006–08 there were significant increases in the proportion of hypertension problems that involved GP orders for:

- only completely or partially supported tests, 7.5% (95% CI: 6.9–8.0) in 2000–02 and 10.1% (95% CI: 9.4–10.7) in 2006–08.
- at least one unsupported test 1.0% (95% CI: 0.8–1.1) in 2000–02 and 1.6% (95% CI: 1.4–1.9) in 2006–08 (results not tabled).

When GPs ordered unsupported tests, they were usually accompanied by one or more supported/partially-supported tests (75% of occasions). Unsupported tests were ordered alone at the remaining 25%. GPs ordered only unsupported pathology tests at 0.3% of hypertension problems in 2000–08. This did not change significantly over time, 0.2% (95% CI: 0.1–0.3) in 2000–02 and 0.4% (95% CI: 0.2–0.5) in 2006–08 (results not tabled).

5.4.6 Discussion

Pathology testing was recommended in the management of hypertension in all guidance documents, and the majority (> 90%) of tests ordered by GPs were recommended (with either partial or complete support) in this guidance.

The statistically significant increase in the rate of GPs' total pathology ordering for hypertension between 2000–02 and 2006–08 was reflected in significant increases in the rate of tests that were supported, partially supported, and unsupported in guidance documents. The largest increase was in the rate of partially supported tests. Due to the increase in total rate of pathology ordering, the correlation between GPs' ordering and level of support, as a proportion of total tests should be considered. Supported tests decreased as a proportion of total tests between 2000–02 and 2006–08, counteracted by an increase in partially supported tests. The proportion of total pathology tests that were unsupported stayed stable. Overall this suggests that the increase in GPs ordering did not result in testing being 'more' or 'less' in line with recommendations. However this disguises a shift in GPs' ordering, from supported to partially supported tests that warrants investigation.

Pathology testing in the initial assessment

The majority of pathology recommendations made in guidance documents were for the initial assessment of newly diagnosed hypertension. An initial assessment was recommended for all patients to detect cardiovascular risk and end organ damage (usually kidney damage).

Guidance documents also recommended investigation of secondary causes of hypertension (as part of the assessment of hypertension) if initial testing was abnormal or other clinical indicators suggested that secondary hypertension was likely. The causes of secondary hypertension that involve pathology ordering in their diagnosis are listed with the relevant recommended test(s) (as defined by the guidance):

- kidney disease—EUC, albumin, albumin:creatinine ratio
- aldosteronism—aldosterone, renin
- Cushing's syndrome—cortisol
- phaeochromocytoma-catecholamines, methylated amines
- parathyroid disease—parathyroid hormone, calcium

• thyroid disease—thyroid-stimulating hormone, T4 (i.e. TFT)

Only kidney disease and thyroid disease were tested for by GPs in the management of hypertension. The tests for the rare diseases that cause secondary hypertension (aldosteronism, Cushing's syndrome, phaeochromocytoma) were ordered very infrequently, accounting for less than 1% of pathology tests for hypertension. This reflects appropriate pathology ordering by GPs based on the low probability of these diseases.

It is possible that GPs ordered pathology tests/batteries for the initial assessment of hypertension at the time of its 'diagnosis' as a new clinical problem, explaining the higher pathology order rate at these new cases compared with the average for hypertension problems. There was a significant increase in the management rate of new cases of hypertension (between 2000–02 and 2006–08), suggesting an increase in the number of new cases, which contributed to the increase in the total rate of pathology ordered for hypertension. Despite this increase, new cases accounted for a small proportion of hypertension contacts and total pathology ordered for hypertension (6% and 16% respectively).

There are limitations to linking pathology testing to new cases as a way of defining the initial assessment. A new case indicates that the patient was diagnosed at that consultation. But GPs may order pathology tests for the initial assessment after the diagnosis (e.g. at the second or third consultation). While this may occur, the chronic nature of the disease suggests that a large proportion of tests recommended as part of the initial investigation were ordered in ongoing management.

As tests were considered 'supported' if recommended at any phase of management, it is important to establish whether tests recommended only for the initial assessment have a role in the ongoing management of hypertension.

Pathology testing in long-term management

Hypertension usually requires life long management. Guidance documents highlighted this fact and often included a section on long-term management, with recommendations for frequency of review and intensifying therapy. However, most did not provide guidance about the need for pathology testing in long-term management. This is alarming because the majority of contacts (94%) and pathology tests (84%) ordered by GPs were for the long-term management of hypertension. The guidance documents that did discuss pathology tests for monitoring did so in relation to: reassessing cardiovascular risk; monitoring (or detecting) end organ kidney damage; and monitoring medications.

Reassessing cardiovascular risk (involving lipid and glucose testing) was recommended in four guidance documents. Steven¹⁷⁴ recommended annual testing and was the only one to specify interval between monitoring. CHEP referred to a diabetes guideline for assessment of 'incident' diabetes (i.e. the development of diabetes) and recommended lipid testing 'with a frequency reflecting the clinical situation'.^{170,171} JNC 7 recommended that cardiovascular risk should be monitored.¹⁶⁵ Similarly, NHF recommended reassessing cardiovascular risk 'regularly' but this was only mentioned in a footnote to a management flowchart.¹⁶³

Presence of end organ kidney damage is also considered a cardiovascular risk in patients with hypertension. JNC 7¹⁶⁵ and CHEP^{170,171} specifically recommended monitoring of electrolytes and creatinine for all patients; only JNC 7¹⁶⁵ specified an interval (1–2 times per year). The SIGN guideline recommended annual testing for proteinuria.

When guidance was given, specific information about the interval between testing was often omitted. Information from other sources would need to be used by GPs if they wanted to know about interval between monitoring. For example, the cardiovascular risk guideline recommended reassessment based on risk level (every 2 years in low risk, 6–12 monthly in medium, and based on clinical situation in high risk);¹⁷⁶ and the RACGP recommended annual testing for diabetes in patients aged 45 years and over with hypertension.¹⁷⁷

It should not be necessary for GPs to reference multiple sources to find this information. Producers of hypertension guidance should improve recommendations on reassessment of cardiovascular risk in long-term care, including the specific pathology testing required and interval between testing.

Medication monitoring

Pathology tests related to medication use were discussed in most guidance documents, but specific recommendations about testing and frequency of tests were often not provided. Some guidance documents listed the common side effects of medications (including hypo/hyperkalaemia, hyponatraemia, hyperglycaemia, worsening renal function) without recommending the testing needed to identify side effects.^{165,167-171,173,178} Others did not specifically mention the side effects.^{164,172,174} It is possible that some of the guidance documents may have considered the monitoring of adverse medication effects outside the scope of the guidance.

When guidance was provided for medication monitoring, potassium and/or creatinine testing was recommended (dependent on the medication used) but no recommendation for frequency of testing was given. Some examples are given below.

- Testing potassium in use of thiazide diuretics. The SIGN guideline recommended testing within 4–6 weeks of initiation but monitoring was not discussed.¹⁶⁶ Steven recommended annual testing of sodium and potassium for patients taking diuretics without discussing the side effects of its use.¹⁷⁴ The NHF guideline recommended action if potassium levels were below the reference range in patients taking thiazide but the recommendation to test and the interval to monitor was not given.¹⁶³ The NHF also listed hyperglycaemia, gout, and hyponatraemia as possible adverse effects of thiazide use 'to be considered' but whether these should be tested for was not clear.¹⁶³
- Monitoring serum creatinine and potassium in use of ACE inhibitor or angiotensin receptor blocker. SIGN recommended testing creatinine and potassium within 1–2 weeks of therapy initiation.¹⁶⁶ NHF and NICE discussed the possibility of an initial rise in creatinine at initiation of these medications without specifically mentioning the need to test creatinine.^{163,164} NHF recommended 'monitoring of kidney function' when combination therapy of ACE inhibitor and angiotensin receptor blocker was used but no information on the monitoring interval or the specific test was given.¹⁶³

In monitoring medications, GPs may have used other sources such as the electronic Therapeutic Guidelines (eTG) (an Australian resource for medication guidance). It recommended testing electrolytes (i.e. potassium) and creatinine prior to initiation of ACE inhibitors or angiotensin receptor blockers and one to two weeks after initiation or dose adjustments. However clear guidance on adverse effect monitoring was not provided for thiazide diuretics or loop diuretics—the common adverse effects were listed but testing was not mentioned.¹⁷⁹

Information about the role of pathology testing in monitoring adverse effects of medication was poor. Guidance documents need to provide clearer recommendations

about whether testing for adverse effects of hypertension medications is required, the interval between tests and the duration that monitoring needs to persist. Alternatively, if not considered within the scope of the guidance, this should be clearly stated and appropriate resources referenced. Further, the completeness of recommendations for pathology testing in these alternative resources should be checked (as demonstrated by the example of diuretics in the eTG).

The discussion in this section demonstrates that guidance documents either lacked guidance or provided incomplete guidance about pathology testing in the long-term management of hypertension. Despite this, the guidance available did suggest that lipid, glucose, electrolyte and kidney function tests had a role in its ongoing management. These tests accounted for about half of pathology ordered for hypertension, and their order rate increased significantly between 2000–02 and 2006–08. This suggests that GPs are acting in accordance with the limited guidance available.

GPs' use of these tests is further supported by research that shows that hypertension rarely occurs in isolation. In 80% of people it is clustered with metabolic conditions (dyslipidaemia, insulin resistance, glucose intolerance and obesity).¹⁶² While test recommendations in guidance documents for the initial assessment of hypertension reflect these comorbidities, they are not necessarily present at the time of diagnosis. The underlying prevalence of these diseases and the fact that they may develop over the longer term in patients with hypertension further supports GPs' continued ordering of these tests.

The lack of guidance about the required interval between monitoring is concerning. However, even if guidance was provided, I would not have been able to assess this using the cross-sectional BEACH data. This assessment would require quality longitudinal data (which may be available from the pathology industry) but such investigation was outside the scope of this thesis.

Tests for which guidance was lacking

There was a significant discord between guidance documents' recommendations and GPs' ordering for hypertension for four tests: FBC, LFT, TFT and MBA. For three of these (FBC, LFT and TFT), guidance documents only recommended their use in the initial assessment of hypertension, but GPs usually ordered them in ongoing management (more than three-quarters of occasions), suggesting their level of

support may be over-estimated. The fourth test, the MBA, was not recommended in any guidance documents. Despite this lack of guidance, GPs' ordering of these tests increased significantly between 2000–02 and 2006–08, and they accounted for one-third of total tests ordered for hypertension in 2000–08.

The rationale for recommending FBC and LFT in the initial assessment was provided by only one guideline (to identify excess alcohol consumption).¹⁶⁶ However, no recommendations were made for ongoing management.

The lack of rationale is more concerning for FBC (than for LFT) because it was frequently recommended (6 out of 11 documents) and commonly ordered by GPs. It raises questions about what clinical purpose the FBC fulfils for GPs in hypertension management, and on what guidance authors based their recommendations. For example, was evidence found that supported FBC use? Were recommendations consensus-based or based on the guidance authors' clinical experience? GPs' have reported in qualitative research that clinical experience influences their pathology ordering.³⁶ However, the fact that one guideline specifically recommended against FBC use (because it did not aid either initial investigation or monitoring of hypertension) adds weight to the argument that guidance authors should revisit the FBC recommendation. Future guidance should include the rationale for the test (to aid in its clinical interpretation) and its role (if any) in monitoring, and if applicable, the frequency of testing required.

GPs' ordering of LFTs may be associated with the management of lipid levels, in particular, monitoring side effects of lipid-lowering medications such as statins (see Section 5.6.6). While guidance documents did not recommend monitoring of LFT for this purpose, they did recommend assessment and management of lipid levels.

Thyroid disease was discussed as a cause of secondary hypertension, suggesting that most TFTs would be ordered in initial testing, but GPs usually ordered them in ongoing care. Information on whether there is a need to reassess TFT when initial results are clinically insignificant is needed in guidance documents. For example, are there certain patient groups for whom reassessing thyroid function is valid (such as, older patients, if incidence increases with age) and what is the interval at which patients should be reassessed if it is valid to do so? The RACGP's guideline on preventive activities in general practice states that there is unproven benefit in screening for thyroid disease in adults.¹⁷⁷ Although this guidance is not specific to

hypertension, GPs' clinical experience may suggest that there is value in reassessing TFT when managing hypertension. However, the question about the validity of reassessing TFT should be investigated further and the findings incorporated into future hypertension guidance.

The increase in the proportion of GP-ordered tests classified as 'partially supported' (mentioned earlier in this discussion) was mainly due to increases in the rate of LFT, TFT and MBA tests. These tests were classified as partially supported because either very few guidance documents recommended the test (in the case of MBA and LFT) or guidance suggested their use should be confined to a specific clinical situation (for TFT).

The four tests for which guidance was unclear are likely to represent the greatest opportunity to decrease the volume of pathology ordered by GPs for hypertension. Reducing their use would require investigation of whether the tests are clinically useful in the management of hypertension at any stage (including for ongoing monitoring). The results of this investigation should be used to improve guidance documents, and this guidance promoted to GPs to improve test ordering (if necessary).

Summary

GPs' selection of appropriate pathology tests in the management of hypertension was excellent, the vast majority being recommended in guidance documents. However, the lack of recommendations provided in guidance documents about pathology testing in long-term management of hypertension is alarming given the majority (84%) of pathology ordered by GPs was for ongoing care. Guidance needs to be improved in this area, specifying the tests required and the recommended interval between monitoring tests. Future research is needed to investigate whether deficiencies identified in guidance also represent areas in which GP pathology ordering could be improved.

The lack of guidance about the testing interval required for monitoring tests may provide an opportunity to decrease the number of testing occasions (where at least one pathology test was ordered) for hypertension, which increased over the study period. If further investigation reveals that the recommended testing interval is less frequent than current GP practice there would be potential for decrease. However, the reverse may be found, that current practice is less frequent than recommended testing, which would suggest an increase in the number of testing occasions is needed.

Targeting a reduction in unsupported tests will achieve modest reductions in volume of pathology (as they accounted for only 5% of pathology ordered for hypertension), and would be unlikely to reduce the number of testing occasions, because GPs usually ordered unsupported tests with supported tests.

Hypertension was the problem that generated the highest volume of pathology tests in Australian general practice. Further, the ageing population is expected to lead to an increase in the prevalence of hypertension, which in turn would generate more management of hypertension in general practice and increased associated pathology ordering. Therefore, achieving a reduction (even a modest reduction) in GPs' pathology test (volumes or occasions of testing) ordering by reducing unnecessary tests would be worthwhile.

5.5 Type 2 diabetes

5.5.1 Background

The prevalence of diabetes in Australia has doubled over the last 20 to 30 years,^{180,181} and is expected to continue to increase,¹⁸⁰⁻¹⁸² a situation that is occurring in many countries.^{182,183} The expected increase is linked to increases in population, ageing of population, and increasing obesity.¹⁸⁰⁻¹⁸³

Diabetes was made a National Health Priority Area in 1996 due to the burden it places on the health system.¹⁸⁴ In 2003, diabetes was responsible for 5.5% of the total burden of disease and injury in Australia.¹⁵⁶

The majority of diabetes in Australia is Type 2 diabetes (T2D).^{154,181,185} The selfreported prevalence of diagnosed T2D in the population was 3.5% in 2007–08.¹⁵⁴ However, measured prevalence is far higher—7.1% of adults aged 25 years and over in 1999–00—because it includes diagnosed and undiagnosed T2D (approximately half was undiagnosed).¹⁸¹

There have been numerous national and state-based policies and initiatives introduced to identify undiagnosed diabetes, improve the care of people with diabetes, and prevent diabetes in those at risk in Australia. Examples of initiatives implemented in general practice include, the National Integrated Diabetes Program¹⁸⁶ (which included the MBS diabetes annual cycle of care items), the National Chronic Disease Strategy,¹⁸⁷ the MBS Chronic Disease Management items,¹⁸⁸ the Prevention of Type 2 Diabetes Program,¹⁸⁹ and the Australian Primary Care Collaboratives Program.¹⁹⁰

General practice has been the focus of many initiatives because GPs hold the primary role in the diagnosis and management of T2D. For the majority of patients they are the health professional most often seen and they coordinate multidisciplinary care, through referrals and through the MBS multidisciplinary care plans. Diabetes is one of the most commonly managed problems in general practice.¹⁷ A 2009 SAND substudy of 3,021 patients estimated the prevalence of T2D to be 9.0% of patients at encounters in general practice.¹⁹¹ Pathology testing is used by GPs to diagnose T2D, to monitor glycaemic control and to identify morbidities associated with diabetes.

5.5.2 Management rate in Australian general practice

In BEACH, T2D was managed at 22,935 patient encounters by 6,451 GPs in 2000–08, at a rate of 2.9 per 100 encounters (Table 5.8). This equates to about 3.2 million encounters per year where T2D was managed by GPs in Australia. As described in Chapter 4, the management rate of T2D increased significantly between 2000–02 and 2006–08. There was no change in the diagnosis or detection rate of new cases of T2D between 2000–02 and 2006–08 (Table 5.8). This indicates that the increased management rate largely reflected an increase in monitoring encounters for T2D rather than an increase in its detection or diagnosis rate.

	20	00008	20	00002	20		
Variable	Number	Rate per 100 encounters (95% Cl) (<i>n</i> =784,300)	Number	Rate per 100 encounters (95% Cl) (<i>n</i> = 198,200)	Number	Rate per 100 encounters (95% CI) (n = 188,300)	Change
General practitioners	6,451		1,573		864		•••
T2D encounters	22,935		5,211		6,171		
T2D problems managed	22,938	2.9 (2.9–3.0)	5,211	2.6 (2.5–2.8)	6,172	3.3 (3.1–3.4)	↑
New T2D problems	1,421	0.2 (0.2–0.2)	325	0.2 (0.1–0.2)	369	0.2 (0.2–0.2)	_

Table 5.8: Summary of T2D data set,	2000-08, 2000-02 and 2006-08
-------------------------------------	------------------------------

Note: Data about T2D problems managed are drawn from Chapter 4, Table 4.3. T2D – Type 2 diabetes; CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/Ψ indicates a statistically significant change, and — indicates no change.

5.5.3 Pathology ordered for T2D

Pathology was ordered at a rate of 77.2 per 100 T2D problems in 2000–08. At almost one-third of T2D problem contacts (29.7%) at least one pathology test/battery was ordered by the GPs. Once the decision to order pathology was made, GPs ordered an average 2.59 tests/batteries per tested T2D problem (Table 5.9).

As reported in Chapter 4, the rate of pathology ordering for T2D increased significantly between 2000–02 and 2006–08. This was due to significant increases in both the likelihood of pathology being ordered for T2D, and the number of tests ordered once the decision to order pathology was made (Table 5.9).

	20	00–08	20	2000–02		2006–08		
Variable	Number	Per cent / Rate of T2D problems (95% Cl) (<i>n</i> = 22,938)	Number	Per cent / Rate of T2D problems (95% Cl) (<i>n</i> =5,211)	Number	Per cent / Rate of T2D problems (95% CI) (n = 6,172)	Change	
Pathology (Rate per 100 T2D problems)	17,709	77.2 (75.0–79.5)	3,314	63.6 (59.6–67.6)	5,459	88.4 (83.7–93.2)	↑	
At least one pathology order (Per cent of T2D problems)	6,818	29.7 (29.0–30.5)	1,423	27.3 (25.8–28.8)	1,948	31.6 (30.1–33.0)	↑	
Number of tests/ batteries per 100 tested T2D problems		259.8 (255.7–263.8)		232.9 (224.8–241.0)		280.2 (272.4–288.1)	↑	

Table 5.9: Summary of pathology ordering for T2D, 2000-08, 2000-02 and 2006-08

Note: Pathology ordering data from 2000–02 and 2006–08 are drawn from Chapter 4, Table 4.4. CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/ indicates a statistically significant change.

Types of pathology tests/batteries ordered

Table 5.10 shows the rate of pathology tests/batteries ordered for T2D in 2000–08 by MBS groups and the most common individual types of tests ordered.

Chemistry tests were the group most often ordered (68.5 per 100 T2D contacts) and the most common were:

- HbA1c tests (23.0 per 100 T2D contacts)
- lipid tests (11.7 per 100 contacts)
- glucose/glucose tolerance tests (10.0)
- EUC tests (7.2)
- 'other chemistry' tests (6.2)—90% of this group were urine albumin tests
- LFTs (4.5) (Table 5.10).

Haematology tests (6.1 per 100 contacts), in particular FBCs (5.5 per 100), were also commonly ordered in the management of T2D (Table 5.10).

Only 7% of pathology tests/batteries were ordered in the management of 'new' cases of T2D, the vast majority being ordered for ongoing management (Table 5.10).

	All T	2D problems	New T2D problems				
Pathology test	Rate per 100 T2D problems (95% Cl) Number (<i>n</i> = 22,938)		Number	Per cent of test	Rate per 100 new T2D problems (95% Cl) (<i>n</i> =1,421)		
Chemistry	15,718	68.5 (66.5–70.6)	1,051	6.7	74.0 (67.6–80.3)		
HbA1c*	5,271	23 (22.3–23.7)	257	4.9	18.1 (16.0–20.1)		
Lipids*	2,681	11.7 (11.2–12.2)	124	4.6	8.7 (7.2–10.2)		
Glucose/glucose tolerance*	2,299	10.0 (9.5–10.5)	287	12.5	20.2 (18.1–22.3)		
EUC*	1,657	7.2 (6.8–7.6)	89	5.4	6.3 (4.9–7.6)		
Other chemistry*	1,418	6.2 (5.8–6.6)	110	7.8	7.7 (6.3–9.2)		
LFT*	1,040	4.5 (4.2–4.9)	60	5.8	4.2 (3.2–5.3)		
MBA*	803	3.5 (3.2–3.8)	46	5.7	3.2 (2.3–4.2)		
TFT*	235	1.0 (0.9–1.2)	47	20.0	3.3 (2.3–4.3)		
Haematology	1,402	6.1 (5.7–6.5)	116	8.3	8.2 (6.6–9.7)		
FBC	1,266	5.5 (5.2–5.9)	106	8.4	7.5 (6.1–8.9)		
Other tests NEC	401	1.8 (1.5–2.0)	31	7.7	2.2 (1.4–3.0)		
Microbiology	164	0.7 (0.6–0.8)	23	14.0	1.6 (0.9–2.3)		
Other pathology groups	24		3	12.5			
Total pathology tests	17,710	77.2 (75.0–79.5)	1,224	6.9	86.1 (78.7–93.6)		

Table 5.10: Rate of pathology test orders for T2D by MBS pathology groups and most frequent individual test orders within each group, 2000–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations.

Changes in the types of pathology tests/batteries ordered

The pathology tests/batteries ordered for T2D problems in 2000–02 and 2006–08 are shown in Table 5.11. Listed below are the tests for which significant change in GPs' order rate occurred between 2000–02 and 2006–08 (listed in decreasing test rate order). There were significant increases in the order rate of:

- HbA1c tests—33% increase
- lipid tests—53% increase
- EUC tests—48% increase
- 'other chemistry' tests—135% increase (due to a 125% rise in urine albumin tests)
- LFTs—79% increase
- MBA tests—83% increase
- FBCs—89% increase (Table 5.11).

There was also a significant 28% decrease in the order rate of glucose/glucose tolerance tests between 2000–02 and 2006–08 (Table 5.11).

		2000–02	:		
Pathology test ordered	Number	Rate per 100 T2D problems (95% Cl) (<i>n</i> = 5,211)	Number	Rate per 100 T2D problems (95% Cl) (<i>n</i> = 6,172)	Change
Chemistry	2,951	56.6 (53.0–60.2)	4,828	78.2 (73.9–82.5)	↑
HbA1c*	994	19.1 (17.7–20.4)	1,566	25.4 (23.9–26.8)	↑
Glucose/glucose tolerance*	641	12.3 (11.1–13.5)	550	8.9 (7.9–9.9)	₽
Lipids*	478	9.2 (8.2–10.1)	869	14.1 (13.0–15.2)	♠
EUC*	289	5.6 (4.8–6.3)	510	8.3 (7.4–9.1)	♠
Other chemistry*	176	3.4 (2.8–4.0)	492	8.0 (7.1–8.8)	↑
LFT*	170	3.3 (2.7–3.8)	363	5.9 (5.1–6.6)	↑
MBA*	121	2.3 (1.8–2.9)	258	4.2 (3.5–4.9)	↑
TFT*	43	0.8 (0.6–1.1)	69	1.1 (0.8–1.4)	—
Haematology	226	4.3 (3.7–5.0)	462	7.5 (6.6–8.4)	↑
FBC	190	3.7 (3.1–4.2)	433	7.0 (6.2–7.8)	♠
Other tests NEC	94	1.8 (1.3–2.3)	123	2.0 (1.5–2.5)	—
Other test NEC*	34	0.7 (0.4–0.9)	51	0.8 (0.5–1.1)	—
Microbiology	35	0.7 (0.4–0.9)	41	0.7 (0.5–0.9)	—
Other pathology groups	8		5		
Total pathology tests	3,314	63.6 (59.6–67.6)	5,459	88.4 (83.7–93.2)	♠

Table 5.11: Rate of pathology test orders for T2D by MBS pathology groups and most frequent individual test orders within each group, 2000–02 compared with 2006–08

Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change, and — indicates no change.

Medications prescribed for T2D

As pathology test recommendations often relate to the medications prescribed in the management of T2D, the most common prescribed medications for T2D are listed in Table 5.12.

Between 2000–02 and 2006–08, there were significant increases in prescribing of the hypoglycaemic agents: thiazolinediones (glitazones), combination oral blood glucose lowering drugs, and long-acting insulin. There were simultaneous decreases in the prescribing rates of sulfonamides and fast acting insulins.

The prescribing rates of statins, ACE inhibitors and aspirin also increased significantly between 2000–02 and 2006–08 (Table 5.12).

-	:	2000–02			
ATC levels 3 and 4	Number	Rate per 100 T2D problems (95% Cl) (<i>n</i> =5,211)	Number	Rate per 100 T2D problems (95% Cl) (n=6,172)	Change
Blood glucose lowering drugs, excluding insulins	3,058	56.7 (55.7–61.6)	3,207	52.0 (49.4–54.5)	¥
Biguanides (e.g. metformin)	1,528	29.3 (27.6–31.0)	1,779	28.8 (27.3–30.3)	
Sulfonamides, urea derivatives	1,480	28.4 (26.6–30.2)	1,088	17.6 (16.4–18.9)	₩
Thiazolidinediones (i.e. glitazones)	8	0.2 (0.0–0.3)	245	4.0 (3.4–4.5)	↑
Combination oral blood glucose lowering drugs	0 [†]	0 [†]	75	1.2 (0.8–1.6)	↑
Insulins and analogues	297	5.7 (4.7–6.7)	395	6.4 (5.6–7.2)	—
Intermediate combined with fast-acting	186	3.6 (2.9–4.2)	269	4.4 (3.8–5.0)	—
Long-acting	19	0.4 (0.2–0.5)	71	1.2 (0.8–1.5)	↑
Fast-acting	92	1.8 (1.3–2.2)	55	0.9 (0.6–1.1)	¥
Lipid modifying agents, plain	59	1.1 (0.8–1.5)	260	4.2 (3.6–4.8)	♠
HMG CoA reductase inhibitors (i.e. statins)	54	1.0 (0.7–1.3)	242	3.9 (3.4–4.5)	↑
ACE inhibitors, plain	51	1.0 (0.7–1.3)	110	1.8 (1.4–2.1)	↑
Other analgesics and antipyretics (e.g. aspirin)	27	0.5 (0.3–0.7)	111	1.8 (1.3–2.3)	↑
Total prescribed medications	3,749	71.9 (68.7–75.2)	4,440	71.9 (68.7–75.2)	_

Table 5.12: Prescribed medications for T2D by Anatomical Therapeutic Chemical classification levels 3 and 4, 2000–02 and 2006–08

† Medication was not available in 2000–02.

Note: Only the medications accounting for > 1% of prescribed medications in either of the data periods are included. ATC – Anatomical Therapeutic Chemical classification; CI – confidence interval; ACE – angiotensin converting enzyme. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/↓ indicates a statistically significant change, and — indicates no change.

5.5.4 Guidance documents for T2D

Г

Twelve guidance documents were reviewed for T2D (Box 5.2).

Box 5.2: Guidance documents reviewed for T2D									
Title	Year	Author	Abbreviated to						
Diabetes management in general practice: guidelines for Type 2 diabetes 2008–09 ¹⁹²	2008	Diabetes Australia (DA) and Royal Australian College of General Practitioners (RACGP)	DA & RACGP						
National evidence based guidelines for the management of Type 2 diabetes mellitus ¹⁹³	2005	Diabetes Australia Guideline Development Consortium (DAGDC)	DAGDC						
Clinical practice guidelines for the prevention and management of diabetes in Canada ¹⁹⁴	2008	Canadian Diabetes Association (CDA)	CDA						
Type 2 diabetes: national clinical guideline for management in primary and secondary care ¹⁹⁵	2008	National Collaborating Centre for Chronic Conditions, Royal College of Physicians. <i>National Institute for Health and Clinical</i> <i>Excellence (NICE) guideline</i> [United Kingdom]	NICE						
Management of diabetes: a national clinical guideline ¹⁹⁶	2001	Scottish Intercollegiate Guidelines Network (SIGN)	SIGN						
Standards of medical care in Diabetes ¹⁹⁷	2008	American Diabetes Association (ADA)	ADA						
Clinical practice guidelines: Diabetes mellitus ¹⁹⁸	2006	Ministry of Health (MoH) [Singapore]	МоН						
Medical guidelines for clinical practice for the management of diabetes mellitus ¹⁹⁹	2007	American Association of Clinical Endocrinologists (AACE)	AACE						
Global guideline for Type 2 diabetes ²⁰⁰	2005	International Diabetes Federation (IDF)	IDF						
RCPA manual, diabetes mellitus section ¹⁷²	2004	The Royal College of Pathologists of Australasia (RCPA)	RCPA						
Murtagh's general practice, diabetes mellitus diagnosis and management chapters ¹⁷³	2007	Murtagh J [Australia]	Murtagh						
Patient presentations in general practice, 'diabetes mellitus' and 'diabetes check-up' ¹⁷⁴	1999	Steven I [Australia]	Steven						

5.5.5 Extent of alignment between GP testing and guidance documents

In Table 5.13, the frequently ordered pathology tests/batteries for T2D are categorised by level of support in the guidance documents listed in Box 5.2.

Supported tests

The five tests recommended in the majority of guidance documents were HbA1c, lipids, glucose/glucose tolerance, EUC, and urinary albumin or albumin:creatinine

(Table 5.13). Together these five tests accounted for 74.4% of tests/batteries ordered by GPs for T2D (Table 5.14). The rate at which supported tests were ordered for T2D increased significantly between 2000–02 and 2006–08 (Table 5.15). However individually, the rate of glucose/glucose tolerance tests decreased significantly while the other tests increased (Table 5.11).

There was strong agreement among guidance documents for monitoring glycaemic control using the HbA1c test. Frequency of recommended testing was specified in the majority of guidance documents. However, three guidelines did not explicitly recommend testing of HbA1c.^{193,196,199}

Fasting glucose tests and oral glucose tolerance tests were recommended for the diagnosis of T2D by all guidance documents that provided diagnostic guidance. Only three documents^{194,195,200} recommended annual fasting plasma glucose testing to check the accuracy of the patient's self-monitoring blood glucose machine.

There was strong agreement among guidance documents for assessment of lipid levels. Frequency of testing and targets were provided in the majority of guidance documents. However, three guidance documents^{172,196,199} did not provide clear guidance on lipid testing.

Annual assessment of kidney function was recommended in all documents with the exception of the DAGDC guideline. Serum creatinine (for calculation of estimated glomerular filtration rate) and urine albumin (albumin:creatinine ratio) were the tests specifically recommended. Annual testing of urea was also recommended in two documents.^{173,174}

Renal function was also often discussed in the context of medications. However, the DAGDC guideline only discussed renal function in the context of medications and blood pressure target—it did not discuss monitoring of creatinine or urine albumin to detect diabetic nephropathy.¹⁹³

Tests with conditional support

LFT was classified as having conditional support (Table 5.13). LFT accounted for 5.9% of tests ordered for T2D (Table 5.14) and their ordering increased significantly between 2000–02 and 2006–08 (Table 5.11).

When LFT was mentioned in guidance documents it was most often in the context of medications, specifically in regard to monitoring glitazone use.^{192,198,199} In two

guidance documents liver function was discussed as a consideration in prescribing metformin but specific testing was not recommended.^{194,195} In contrast, NICE recommended against LFT testing when statins were prescribed.¹⁹⁵

Two documents recommended LFT as part of the initial assessment of $T2D^{174,197}$ and one of these also recommended annual testing.¹⁷⁴

Support unable to be determined

It was not possible to determine whether MBA and 'other chemistry' tests (excluding urine albumin tests) were recommended in the guidance documents (see Section 5.3.4). Together these two tests accounted for 5.3% of pathology ordered for T2D (Table 5.14).

With the exception of urinary albumin and albumin:creatinine (split from the 'other chemistry' group and discussed above), the remainder of the 'other chemistry' group was classified as having unclear guidance.

The MBA test was not recommended in any guidance documents. However, some tests that may be considered part of the MBA, such as EUC and LFT, were partially or completely supported in the management of T2D (as discussed above).

Unsupported tests

The majority of guidance documents did not recommend testing of FBC and TFT (Table 5.13). These tests accounted for 8.5% of pathology ordered for T2D (Table 5.14). The order rate of FBC increased significantly between 2000–02 and 2006–08 (Table 5.11).

FBC monitoring was mentioned in two documents as a check for anaemia when chronic kidney disease was present.^{199,200}

TFT was not mentioned in the majority of guidance documents. Two documents did recommend TFT: DA & RACGP¹⁹² recommended it as part of the initial assessment if there was a family history or clinical suspicion of thyroid disease; and ADA¹⁹⁷ recommended TFT as part of the initial assessment if dyslipidaemia was present or the patient was female and over 50 years of age.

Other tests mentioned in the guidance documents

Other tests were mentioned in the guidance documents but each of these accounted for less than 1% of tests recorded by GPs for T2D.

Urinalysis was recommended as part of the initial investigations in two sources.^{174,192}

Urine M,C&S was recommended by DA & RACGP if risk of urine infection was high¹⁹² and was mentioned in RCPA in relation to diabetic nephropathy but the specific purpose of testing (whether to identify or monitor nephropathy) was not stated.¹⁷²

Parathyroid function tests were recommended in the presence of kidney disease by AACE.¹⁹⁹

Two guidance documents specifically recommended that routine monitoring of blood ketones¹⁹⁸ and creatine kinase (CK)¹⁹⁵ should not be done.

Key to Table 5.13

Colour	Description
	The document specifically recommended this test. Any notes within the cell indicate further detail. For example, a specific disease to test for within subset of patients; a specific test within a group.
	The document stated that this test should be considered. Any notes within the cell indicate further detail (e.g. a specific test to consider)
	Unable to determine guidance (see Section 5.3.4).
	The document specifically stated not to do this test. Additional information is supplied if certain conditions apply (e.g. specific clinical situations).
	Guidance document does not mention this test

Pathology test	Per cent of T2D path (<i>n</i> = 17,710)	DA & RACGP ¹⁹²	DAGDC ¹⁹³	CDA ¹⁹⁴	NICE ¹⁹⁵	SIGN ¹⁹⁶	ADA ¹⁹⁷	MoH ¹⁹⁸	AACE ¹⁹⁹	IDF ²⁰⁰	RCPA ¹⁷²	Murtagh ¹⁷³	Steven ¹⁷⁴
HbA1c*	29.8		Implied			Implied			Implied				
Lipids*	15.1					Implied			Implied		Implied		
Glucose (excluding tolerance tests)	12.1	Diagnosis	Diagnosis	Diagnosis & annual	Annual	Diagnosis	Diagnosis	Diagnosis	Diagnosis	Diagnosis & annual	Diagnosis	Diagnosis	N/A
EUC*	9.4		Implied										
Urine albumin/ albumin:creatinine	ratio 7.2		Implied										
FBC	7.2								Kidney disease	Kidney disease			
LFT*	5.9	Meds		Meds	<mark>Statins</mark> Meds		Diagnosis	Meds	Meds				
MBA*	4.5												
TFT*	1.3	Diagnosis (Family Hx or symptoms)					Diagnosis (dyslipidaemia or F > 50 years)						
Glucose tolerance	0.9	Diagnosis	Diagnosis	Diagnosis	N/A	Diagnosis	Diagnosis	Diagnosis	Diagnosis	Diagnosis	Diagnosis	Diagnosis	N/A
Other chemistry* (excluding urine alb albumin:creatinine	oumin/ 0.8 ratio)												

Table 5.13: Summary of guidance recommendations by most frequent individual test orders for T2D, 2000–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: T2D – Type 2 diabetes; path – pathology; N/A – not applicable; Meds – medications; Hx – history; F – female; also see Abbreviations. Any notes within the coloured cells are described in detail in Section 5.5.5, briefly: implied – indicates the use of the test is implied; Diagnosis – Test recommended at diagnosis; Meds – test recommended when specific medications are taken.
Evaluation of GP pathology ordering against guidance documents

Table 5.14 provides a summary of the individual tests recorded in BEACH and the level of support provided in the guidance documents for each. Of the tests/batteries ordered for T2D in 2000–08:

- 74.4% were supported
- 11.2% had conditional support or support could not be determined
- 8.5% were not supported (Table 5.14).

The tests/batteries listed in Table 5.14 account for 94.1% of pathology ordered for T2D because only the most commonly ordered tests for T2D were evaluated.

Table 5.14: Summary of support for GP pathology ordering for the most frequent individual test orders for T2D, 2000–08

Pathology test	Number	Per cent of all pathology for T2D
Supported	13,184	74.4
HbA1c*	5,271	29.8
Lipids*	2,681	15.1
Glucose/glucose tolerance*	2,299	13.0
EUC*	1,657	9.4
Urine albumin/albumin:creatinine ratio	1,276	7.2
Conditional/unclear support	1,985	11.2
LFT*	1,040	5.9
MBA*	803	4.5
Other chemistry* ^(a) (excluding urine albumin/albumin:creatinine ratio)	142	0.8
Unsupported	1,501	8.5
FBC	1,266	7.1
TFT*	235	1.3
Subtotal (n, % of total tests)	16,671	94.1
Total pathology tests	17,710	100.0

(a) 'Other chemistry' after excluding albumin tests accounts for < 1% of total tests in 2000–08 but in 2006–08 it accounted for 1.6% of tests. It is included because as a whole group it accounted for ≥ 1% of tests for T2D.</p>

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. See Abbreviations.

In 2000–08, GPs ordered supported pathology tests at a rate of 57.5 tests/batteries per 100 T2D problems, followed by tests with conditional support (8.6 per 100), and unsupported tests (6.5). Tests ordered for T2D but not evaluated were ordered at a rate of 4.6 per 100 T2D problems. The rate at which GPs ordered tests in all 'level of support' groups increased significantly over time (Table 5.15).

	2000–08	2000–02	2006–08	
Level of support	Rate per 100 T2D problems (95% Cl) (<i>n</i> = 22,938)	Rate per 100 T2D problems (95% Cl) (<i>n</i> =5,211)	Rate per 100 T2D problems (95% Cl) (<i>n</i> =6,172)	Change
Supported	57.5 (55.7–59.2)	49.1 (45.9–52.3)	63.7 (60.1–67.3)	↑
Conditional/unclear support	8.6 (8.2–9.1)	6.0 (5.2–6.7)	11.0 (10.0–12.0)	↑
Unsupported	6.5 (6.1–7.0)	4.5 (3.8–5.1)	8.1 (7.2–9.1)	↑
Not evaluated	4.6 (4.2–4.9)	4.1 (3.4–4.7)	5.7 (4.9–6.4)	↑

Table 5.15: Rate of pathology ordering for T2D by level of support, 2000–08, 2000–02and 2006–08

Note: CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change.

GPs ordered only pathology tests that were supported or conditionally supported at 22.6% (95% CI: 21.9–23.3) of T2D problems in 2000–08. A further 6.0% (95% CI: 5.6–6.3) of T2D problems involved at least one unsupported test, and 71.4% (95% CI: 70.7–72.2) of problems involved either no pathology tests (70.3%) or tests that were not evaluated (1.1%) (results not tabled).

There was no change in the proportion of T2D problems that involved supported testing between 2000–02 (22.2%, 95% CI: 20.7–23.6) and 2006–08 (22.7%, 95% CI: 21.4–24.1). However the proportion involving at least one unsupported test increased significantly over this time, from 4.1% (95% CI: 3.5–4.7) to 7.4% (95% CI: 6.6–8.2) (results not tabled).

When GPs ordered unsupported tests, the vast majority (98%) were accompanied by one or more supported/partially-supported tests. Unsupported tests were ordered alone at the remaining 2%. GPs ordered only tests that were unsupported at 0.1% of T2D problems in 2000–08, and this did not change over time (0.1% in 2000–02 and 2006–08) (results not tabled).

5.5.6 Discussion

More than 85% of pathology tests ordered by GPs in the management of T2D were recommended (with either partial or complete support) by guidance documents. The statistically significant increase in the rate of GPs total pathology ordering for T2D between 2000–02 and 2006–08 was reflected in significant increases in the rate of tests that were supported, partially supported, and unsupported in guidance documents. The largest increases were in the rate of partially supported and unsupported tests.

Due to the increase in total rate of pathology ordering, the correlation between GPs' ordering and level of support, as a proportion of total tests should be considered. Supported tests accounted for the largest proportion of total tests, but they accounted for a smaller proportion of total tests in 2006–08 than in 2000–02. Concomitantly there were small increases in the proportion of both the partially supported and unsupported tests. Overall the increase in total tests ordered by GPs for T2D suggested a small shift (approximately 2% of total pathology) to be 'less' in line with guidance recommendations.

Comments on selected guidance documents

The DAGDC, SIGN and AACE guidelines lacked clear recommendations for testing in (at least one of) three crucial aspects of T2D management: glycaemic control, lipid testing and renal function.

Regular monitoring of glycaemic control (using the HbA1c test) was not recommended in any of these guidelines. SIGN and AACE both discussed the HbA1c target but specific recommendations for monitoring HbA1c were not given.^{196,199} The DAGDC guideline was published as seven stand-alone sections (each section focused on an aspect of T2D management), but there was no section on glycaemic control. It was only mentioned in the lipid section of the guideline in the context of the beneficial effect of glycaemic control on lipids, and specific test recommendations for HbA1c were not given.¹⁹³

Lipid test recommendations were lacking in the SIGN and AACE guidelines. SIGN discussed dyslipidaemia as a cardiovascular risk factor,¹⁹⁶ and AACE provided the target for lipid levels,¹⁹⁹ but neither included testing or monitoring recommendations.

The DAGDC guideline did not include a section on diabetic nephropathy. Renal function was only discussed as a consideration in choice of therapy and monitoring of medication side effects.¹⁹³

Since work for this chapter was completed, new SIGN and DAGDC guidelines have been published that supersede those reviewed in this study, and an additional AACE guideline about developing a diabetes care plan has been published (but this does not supersede the existing guidance). The new SIGN guideline does not correct the omissions noted above. In contrast, the new DAGDC guideline corrects the noted omissions, but it is worth mentioning that the structure of separate sections persists in the new version. Publishing the guideline in multiple sections makes it difficult to search for information and creates a very long guideline (the length of the 2005 guideline reviewed in this chapter was more than 750 pages, easily the longest of the reviewed documents) making it difficult to use as a guidance document.

Monitoring of T2D

As T2D is a chronic condition, the majority of management provided and pathology ordered by GPs, was for its ongoing management. In most guidance documents there were clear recommendations on the use and frequency of tests to monitor glycaemic control, lipid levels and renal function in the ongoing management of T2D. However the role of fasting glucose tests, FBC and LFT in long-term management was not clear.

The fasting glucose test was unanimously recommended to diagnose T2D, suggesting the majority would be associated with 'new' cases of T2D. However, BEACH data showed that most (87%) were ordered after the diagnosis of T2D. Glucose testing accounted for 12% of testing for T2D. While the order rate of fasting glucose tests did decrease between 2000–02 and 2006–08, the high proportion ordered for ongoing care suggests that GPs routinely use this test for ongoing management. Only three guidance documents recommended glucose testing in ongoing management, to annually assess the accuracy of the patient's home glucose monitor. GPs' rationale for ongoing use may reflect this recommendation or may be due to other reasons. Considering the volume of glucose testing, guidance on the role of fasting glucose testing in ongoing care of T2D should be clarified in future guidance documents.

FBC testing was classified as unsupported in this study because only two guidance documents suggested it had a role in the management of T2D. However, GPs commonly ordered FBC for T2D (7% of total tests in 2000–08), and the order rate almost doubled between 2000–02 and 2006–08. This test was responsible for the small shift in GPs ordering being 'less' in line with guidance recommendations. Two guidance documents recommended FBC testing to check for anaemia when chronic kidney disease was present. The presence of chronic kidney disease as a comorbidity in T2D was not recorded in BEACH. Therefore whether the rate of FBC testing ordered by GPs reflects monitoring of kidney disease cannot be determined. However, the lack of guidance for the test, its increased ordering rate and the high proportion of tests it accounted for, suggest that the role of FBC in ongoing management of T2D needs to be clarified.

Assessment of liver function was most often mentioned in regard to medications in guidance documents, either to identify liver dysfunction or to monitor potential adverse effects of medications. However, frequency and duration of monitoring were not specified, with one exception—Steven¹⁷⁴ recommended annual monitoring. In the management of T2D, the order rate of LFTs increased significantly between 2000–02 and 2006–08, as did the prescribing rate for medications for which LFT monitoring was recommended (statins and glitazones). Given these increases, the required frequency and duration of LFT for monitoring adverse effects of medications needs to be clarified (see 'medication monitoring' discussion below). For fasting glucose tests, FBC and LFT, in addition to the issues discussed above, there was a lack of guidance about interpretation of test results in the context of monitoring, even among tests with an established role in monitoring. While the target and testing interval were discussed, guidance was not provided on what represents a true change in repeated testing results. This is known as the 'critical difference' and incorporates the expected analytical and intra-individual variation of a test.²⁰¹ An understanding of critical difference is necessary when making treatment decisions (e.g. titrating medications) as it determines the level at which a change in results represents a true change versus a chance finding.

Previous studies have demonstrated that there is large variation in GPs estimation of critical difference^{201,202} and that there is limited guidance available to users of pathology testing about interpreting repeated results.²⁰¹⁻²⁰³ As providing guidance

about critical difference is complex (due to the number of factors involved), diseasespecific guidelines may not be the best place to disseminate this information for monitoring tests. Nevertheless, authors of guidance should consider including at least some information about interpretation of the primary monitoring tests on which major treatment decisions for T2D are based (such as HbA1c).

Medication monitoring

The amount of information about pathology tests required when selecting medication (e.g. presence of impaired renal function) and identifying adverse effects, varied considerably among guidance documents. Some documents stated that medication information was outside the scope of the guidance and referred the reader to the relevant product information.^{194,195} In contrast, the AACE guideline provided a summary table for oral hypoglycaemic medications that described the monitoring required to determine medication response and presence of side effects, with information on time interval to test.¹⁹⁹

Recommendations regarding testing to measure response to therapy were clear in the guidance. However, the testing required to identify contraindications and to monitor adverse effects was not clear and these are discussed below.

For metformin, renal impairment and liver disease were considerations in the appropriateness of the medication.^{179,192,194,198,199} Despite this, most guidance did not recommend testing prior to initiation of metformin. Based on pathology test recommendations in other parts of the guidance documents, GPs should have been aware of renal impairment as it would be tested in routine assessment of T2D, but this is not the case for liver function. As metformin is the first line medication recommended for T2D and the most commonly prescribed medication for T2D in this study, the high rate of LFTs ordered by GPs may be partially due to the need to assess presence of liver disease in these patients. Guidance about the need for testing prior to initiation of metformin should be clarified.

In the use of glitazones, liver dysfunction/disease was stated as a consideration and monitoring of liver enzymes was recommended.^{192,198,199} The frequency of monitoring of liver enzymes was often not specified. In contrast, the Australian medication guideline, eTG, recommended LFT prior to initiation of glitazone, followed by 2-monthly monitoring for the first year of therapy and periodically thereafter.¹⁷⁹ The prescribing rate of glitazones increased significantly between

2000–02 and 2006–08 and this may have contributed to the significant increase in LFT testing.

Most T2D guidance documents included recommendations about the management of lipid levels and blood pressure. For the blood pressure medications—ACE inhibitors, angiotensin receptor blockers and diuretics—testing of electrolytes and creatinine were recommended.^{192,196-199}

Statins were the lipid lowering medication most often discussed in T2D guidance. The NICE guideline and a paper by the American Physicians Association specifically stated monitoring of liver function in statin use for T2D was not necessary.^{195,204} In contrast, the majority of guidance for management of lipid disorders reviewed in Section 5.6 recommended monitoring LFT in statin use. The eTG also recommended monitoring liver function after 4 weeks of statin use, but information on ongoing monitoring was not provided.¹⁷⁹ In the management of T2D the prescribing rate of statins and ACE inhibitors increased significantly over the duration of this study (2000–02 to 2006–08) and this may have contributed to the significant increase in LFT and kidney function testing.

In summary, pathology testing related to medication use may have contributed to increases in the rates of LFT, electrolyte and kidney function testing. Guidance documents need to provide clearer recommendations about when testing is required, whether monitoring of adverse effects is required, the interval between tests and the duration of monitoring needed.

Other factors contributing to the increased management rate of T2D

The recommended management of T2D has changed as our understanding of cardiovascular disease risk has improved. Guidance documents' recommendations are based on evidence about the combined influence of multiple risk factors (such as lipid levels and blood pressure) on cardiovascular disease risk. GPs' management of T2D recorded in BEACH suggests they manage the multiple risk factors (such as blood pressure and lipids) as part of T2D rather than as separate clinical problems, probably because the threshold at which they become a cardiovascular risk is lower than the level required for clinical diagnosis.

As awareness of the threshold of risk (the level at which a factor contributes to risk) improved, the targets for a number of observable findings were changed in guidance documents (for example, lower HbA1c, low density lipoprotein [LDL] cholesterol

and blood pressure, and increased high density lipoprotein [HDL] cholesterol) for certain 'at risk' patients. Changes in these targets may have influenced GPs' management rate of T2D and pathology ordering rates because these targets are potentially harder and take longer to achieve. Pathology testing to measure response is more frequent while actively trying to achieve a target (titrating medications, monitoring adverse-effects of medications). While the guidance documents acknowledge that targets should be adjusted to the individual patient, it is likely that a change in targets that requires intensifying management will result in increased testing rates.

Basing treatment recommendations on multiple factors is likely to increase the number of patients requiring active treatment. For example, using conventional targets, a single risk factor in a patient with T2D may not need treatment—such as, a LDL level of 2.4 mmol/L, where the target is < 2.5mmol/L. However, it may require treatment when other risk factors are considered—such as, presence of coronary heart disease which lowers the recommended LDL target to < 2.0 mmol/L,¹⁹² presence of other risk factors (age > 60 years, microalbuminuria) which require lipid-lowering treatment regardless of lipid levels.¹⁷⁶

Management of cardiovascular disease risk factors in patients with T2D may have contributed to the increased management rate and pathology rate seen in this study. While these changes may increase the resources required to manage T2D they represent an increase in high quality evidence-based care.

Past health initiatives aiming to improve detection of diabetes and care of patients with diabetes are also likely to have contributed to the increased management rate and pathology ordering rate for T2D in general practice. These initiatives provide funding for evidence-based clinical activities. For example the diabetes cycle of care (introduced as part of the National Integrated Diabetes Program¹⁸⁶ in 2006) required patients to see the GP at least twice (in a 11–13 month period) and have HbA1c, lipids and microalbuminuria assessed at least once in the cycle.¹⁸⁶ The success of these initiatives may have had an impact on the management rate and pathology ordering rate for T2D in a positive way – as they are evidence-based, the recommended activities reflect best practice in the management of T2D.

Summary

The majority of tests ordered by GPs in the management of T2D were recommended in guidance documents. It can be concluded that much of the increase in the management rate and testing rate seen in BEACH was related to best practice management of T2D and has been stimulated through numerous evidence-based initiatives. The prevalence of T2D is expected to increase in the future due to the ageing population and increasing obesity prevalence.¹⁸⁰⁻¹⁸² This would cause a concomitant increase in the management rate of T2D in general practice, and a corresponding increase in pathology ordering based on the current pattern of GPs' pathology ordering for T2D. This study suggests that most of this ordering would be supported by guidance documents.

These finding suggest that the main opportunity to reduce unsupported testing ordered for T2D lies in clarification of whether FBC has a role in T2D management. Monitoring recommendations should also be improved to clarify: the role (if any) of fasting glucose and LFT in long-term management of T2D; and testing required to monitor adverse effects of medications. The majority of T2D management and pathology ordered by GPs for T2D is for monitoring, reinforcing the need for clarity in recommendations regarding pathology testing in long-term management.

5.6 Lipid disorders

5.6.1 Background

Lipid disorder is national priority because it is a risk factor for cardiovascular disease.¹⁵¹ In 2003, 'high blood cholesterol' was responsible for 6.2% of the total burden of disease and injury in Australia.¹⁵⁶

The prevalence of measured elevated total cholesterol (> 5.5 mmol/L) was 51.2% of adults aged 25 years and over in 1999–00.¹⁵⁷ The National Health Survey estimated the prevalence of self-reported high cholesterol to be 6% of the Australian population in 2007-08.¹⁵⁴

The link between lipid levels and cardiovascular disease has been known for many decades.^{205,206} However the management strategy has changed over time. Lipid levels are now assessed and managed in the context of the patient's absolute cardiovascular risk.²⁰⁷ This incorporates multiple risk factors and allows identification of the patients who will benefit most from treatment rather than basing treatment decisions on lipid levels alone.¹⁷⁸ In this context, the criteria for subsidised lipid-lowering medications (through the PBS) in Australia were broadened in October 2006 to allow access on the basis of cardiovascular risk rather than measured lipid levels alone.²⁰⁸ Lipid disorder is one of the most commonly managed problems in general practice.¹⁷ A SAND substudy of 2,960 patients estimated that 22% of patients at general practice encounters had diagnosed dyslipidaemia, and a further 5% had their lipid levels managed for other reasons (such as cardiovascular risk).²⁰⁹ Pathology tests are required for the diagnosis and monitoring of lipid disorders, and for the assessment of cardiovascular risk.

5.6.2 Management rate in Australian general practice

In BEACH, lipid disorder was managed at 25,231 patient encounters by 6,480 GPs, at a rate of 3.2 per 100 encounters in 2000–08 (Table 5.16). This equates to approximately 3.2 million encounters per year where lipid disorder was managed by GPs in Australia.

As discussed in Chapter 4, the management rate of lipid disorders increased significantly between 2000–02 and 2006–08 (Table 5.16). There was a significant

increase in the diagnosis or detection rate of new cases of lipid disorder, from 0.35 per 100 encounters in 2000–02 to 0.48 in 2006–08 (Table 5.16). This suggests that the increased management rate reflects increases in both detection and monitoring encounters for lipid disorders.

	20	00008	20	2000–02		2006–08		
Variable	Number	Rate per 100 encounters (95% Cl) (<i>n</i> =784,300)	Number	Rate per 100 encounters (95% CI) (<i>n</i> = 198,200)	Number	Rate per 100 encounters (95% CI) (<i>n</i> = 188,300)	Change	
General practitioners	6,480		1,629		1,602			
Lipid encounters	25,231		5,780		6,624			
Lipid problems managed	25,248	3.2 (3.2–3.3)	5,782	2.9 (2.8–3.0)	6,629	3.5 (3.4–3.7)	↑	
New lipid problems	3,169	0.40 (0.39–0.42)	699	0.35 (0.32–0.38)	902	0.48 (0.44–0.52)	↑	

Table 5.16: Summary of lipid disorders data set, 2000–08, 2000–02 and 2006–08

Note: Data about lipid disorder problems managed are drawn from Chapter 4, Table 4.3. Cl – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change.

5.6.3 Pathology ordered for lipid disorders

Pathology was ordered at a rate of 62.5 per 100 lipid disorder problems managed in 2000–08. Almost one-third of lipid disorder problem contacts (30.5%) resulted in at least one test/battery being ordered by GPs. Once the decision to order pathology was made, GPs ordered on average 2.05 tests/batteries per tested lipid problem (Table 5.17).

As reported in Chapter 4, the rate of pathology ordering for lipid disorders increased significantly between 2000–02 and 2006–08. This was due to a significant increase in the number of tests ordered once the decision to order pathology was made. There was no change in the likelihood of pathology being ordered in the management of lipid disorders (Table 5.17).

	2000–08			00–02	2		
Variable	Number	Per cent / Rate of lipid problems (95%Cl) (<i>n</i> = 25,248)	Number	Per cent / Rate of lipid problems (95%Cl) (n=5,782)	Number	Per cent / Rate of lipid problems (95%Cl) (n=6,629)	Change
Pathology (Rate per 100 lipid problems)	15,777	62.5 (60.6–64.4)	3,363	58.2 (54.7–61.6)	4,410	66.5 (62.5–70.6)	↑
At least one pathology order (% of lipid problems)	7,704	30.5 (29.8–31.3)	1,758	30.4 (28.9–31.9)	2,010	30.3 (28.9–31.8)	_
Number of tests/ batteries per 100 tested lipid problems		204.8 (201.2–208.4)		191.4 (184.6–198.2)		219.4 (211.6–227.3)	↑

Table 5.17: Summary of pathology ordering for lipid disorder, 2000–08, 2000–02 and2006–08

Note: Pathology ordering data from 2000–02 and 2006–08 are drawn from Chapter 4, Table 4.4. CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ♠/♥ indicates a statistically significant change, and — indicates no change.

Types of pathology tests/batteries ordered

Table 5.18 shows the rate of pathology ordered for lipid disorders in 2000–08 by

MBS groups and the most common individual types of tests ordered.

Chemistry tests were the group most often ordered (57.1 per 100 lipid disorder

contacts) and the most common were:

- lipid tests (31.4 per 100 lipid disorder contacts)
- LFTs (7.8)
- glucose/glucose tolerance tests (6.0)
- EUC tests (3.2).

Haematology tests (4.3 per 100 contacts), in particular FBCs (3.8), were also commonly ordered in the management of lipid disorders.

Only 7.6% of pathology tests were ordered in the management of 'new' cases of lipid disorders, the vast majority being ordered for ongoing management (Table 5.18).

	All lipid d	isorder problems	New I	ipid disord	id disorder problems			
- Pathology test	Number	Rate per 100 lipid problems (95% Cl) (<i>n</i> = 22,938)	Number	Per cent of test	Rate per 100 new lipid problems (95% Cl) (<i>n</i> = 1,421)			
Chemistry	14,414	57.1 (55.4–58.8)	1,104	7.7	34.8 (31.9–37.8)			
Lipids*	7,919	31.4 (30.5–32.3)	704	8.9	22.2 (20.4–24.0)			
LFT*	1,961	7.8 (7.3–8.2)	106	5.4	3.3 (2.7–4.0)			
Glucose/glucose tolerance	* 1,520	6.0 (5.7–6.4)	107	7.0	3.4 (2.7–4.0)			
EUC*	801	3.2 (2.9–3.5)	42	5.2	1.3 (0.9–1.7)			
MBA*	683	2.7 (2.4–3.0)	37	5.4	1.2 (0.8–1.6)			
СК	671	2.7 (2.4–2.9)	44	6.6	1.4 (1.0–1.8)			
TFT*	318	1.3 (1.1–1.4)	27	8.5	0.9 (0.5–1.2)			
PSA*	157	0.6 (0.5–0.7)	4	2.5	0.1 (0.0–0.2)			
Haematology	1,076	4.3 (3.9–4.6)	55	5.1	1.7 (1.2–2.2)			
FBC	949	3.8 (3.5–4.1)	48	5.1	1.5 (1.1–1.9)			
Other tests NEC	210	0.8 (0.7–1.0)	16	7.6	0.5 (0.2–0.8)			
Other pathology groups	77		18	23.4				
Total pathology tests	15,777	62.5 (60.6–64.4)	1,193	7.6	37.7 (34.5–40.8)			

Table 5.18: Rate of pathology test orders for lipid disorder by MBS pathology groups and most frequent individual test orders within each group, 2000–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations.

Changes in the types of pathology tests/batteries ordered

Table 5.19 shows the most common tests/batteries ordered for lipid disorders in 2000–02 and 2006–08. Listed below are the tests for which significant change in GPs' order rate occurred between 2000–02 and 2006–08 (listed in decreasing test rate order). There were significant increases in the order rate of:

- FBCs—85% increase
- EUC tests—110% increase
- MBA tests—68% increase
- CK tests—94% increase
- TFTs—70% increase
- PSA tests (a marginal increase)—60% increase (Table 5.19).

In contrast, there was a significant 12% decrease in the order rate of lipid tests (Table 5.19).

	2	2000–02	:	2006–08	
Pathology test	Number	Rate per 100 lipid problems (95% Cl) (n=5,782)	Number	Rate per 100 lipid problems (95% Cl) (<i>n</i> = 6,629)	Change
Chemistry	3,124	54.0 (50.8–57.3)	3,954	59.7 (56.1–63.2)	_
Lipids*	1,932	33.4 (31.5–35.4)	1,957	29.5 (27.9–31.1)	$\mathbf{+}$
LFT*	408	7.1 (6.1–8.0)	550	8.3 (7.4–9.2)	_
Glucose/glucose tolerance*	305	5.3 (4.6–6.0)	417	6.3 (5.5–7.0)	_
EUC*	121	2.1 (1.6–2.5)	294	4.4 (3.8–5.1)	↑
MBA*	112	1.9 (1.4–2.4)	211	3.2 (2.6–3.8)	↑
СК	101	1.8 (1.2–2.3)	232	3.5 (2.9–4.1)	↑
TFT*	56	1.0 (0.7–1.2)	112	1.7 (1.3–2.1)	↑
PSA*	26	0.5 (0.3–0.6)	55	0.8 (0.6–1.1)	\uparrow
Haematology	185	3.2 (2.6–3.8)	361	5.5 (4.7–6.2)	↑
FBC	156	2.7 (2.2–3.2)	333	5.0 (4.3–5.8)	↑
Other tests NEC	35	0.6 (0.4–0.8)	71	1.1 (0.7–1.4)	—
Other pathology groups	19		24		
Total pathology tests	3,363	58.2 (54.7–61.6)	4,410	66.5 (62.5–70.6)	↑

Table 5.19: Rate of pathology test orders for lipid disorder by MBS pathology groups and most frequent individual test orders within each group, 2000–02 compared with 2006–08

Note: CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change (darker shading), ↑/♥ indicates a marginal change (lighter shading), and — indicates no change.

Medications prescribed for lipid disorders

Lipid lowering agents (plain and combination) accounted for more than 97% of medications prescribed in the management of lipid disorders, in 2000–08. Most of these were plain statins (91% of all prescribed medications for lipid disorders). There was a marginal increase in the rate of prescribed medications between 2000–02 and 2006–08, from 63.0 per 100 contacts with lipid disorders (95% CI: 61.1–64.8) to 66.5 per 100 (95% CI: 64.8–68.2). This increase was due GPs' prescribing of ezetimibe and statin/ezetimibe medications, which were not available for purchase in 2000–02 and were prescribed at a rate of 4.0 per 100 lipid disorder contacts in 2006–08. There was no change between 2000–02 and 2006–08 in the prescribing rate of: statins; fibrates; and other lipid-lowering medications (results not tabled).

5.6.4 Guidance documents for lipid disorder

Guidance documents for the management of lipid disorder and the lipid section of cardiovascular disease prevention guidelines were considered in this study. Twelve guidance documents were reviewed for lipid disorder (Box 5.3).

Box 5.3: Guidance documents reviewed for lipid disorders

Title	Year	Author	Abbreviated to
Position Statement on Lipid Management ¹⁷⁸	2005	National Heart Foundation (NHF) of Australia and the Cardiac Society of Australia and New Zealand (CSANZ)	NHF & CSANZ
Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults - Adult Treatment Panel ²¹⁰ and the 2004 update: Implications of recent clinical trials ²¹¹	2002 (2004)	National Cholesterol Education Program (NCEP) Expert Panel [United States of America]	NCEP
Clinical Guidelines and Evidence Review for Lipid Modification: cardiovascular risk assessment and the primary and secondary prevention of cardiovascular disease ²¹²	2008	National Collaborating Centre for Primary Care and Royal College of General Practitioners. <i>National Institute</i> <i>for Health and Clinical Excellence</i> <i>(NICE) guideline</i> [United Kingdom]	NICE
Risk estimation and the prevention of cardiovascular disease ²¹³	2007	Scottish Intercollegiate Guidelines Network (SIGN)	SIGN
Medical Guidelines for Clinical Practice for the Diagnosis and Treatment of Dyslipidemia and Prevention of Atherogenesis ²¹⁴	2002	American Association of Clinical Endocrinologists (AACE)	AACE
European guidelines on cardiovascular disease prevention in clinical practice ²¹⁵	2007	European Society of Cardiology (ESC)	ESC
Screening and Management of Lipids ²¹⁶	2009	Barrie WE, Harrison RV, Khanderia UB, Kiningham RB, Rosenson RS University of Michigan [United States of America]	Barrie et al.
Recommendations for the management of dyslipidemia and the prevention of cardiovascular disease ^{217,218}	2003	Genest J, Frohlich J, Fodor G, McPherson R. Working Group on Hypercholesterolemia and Other Dyslipidemias [Canada]	Genest et al.
Clinical practice guidelines: Lipids ²¹⁹	2006	Ministry of Health (MoH) [Singapore]	МоН
Health Care Guideline: Lipid Management in Adults ²²⁰	2007	Woolley A, Kopecky S, Kottke T, O'Connor P, Hanson S, Conroy W et al. <i>Institute for Clinical Systems</i> <i>Improvement (ICSI) guideline</i> [United States of America]	ICSI
RCPA manual, hyperlipidaemia section ¹⁷²	2004	The Royal College of Pathologists of Australasia (RCPA)	RCPA
Murtagh's general practice, dyslipidaemia chapter ¹⁷³	2007	Murtagh J [Australia]	Murtagh

5.6.5 Extent of alignment between GP testing and guidance documents

In Table 5.20, the frequently ordered tests/batteries for lipid disorder are categorised by level of support in the guidance documents listed in Box 5.3.

Supported tests

The five types of tests recommended in the majority of guidance documents were Lipid tests, LFT, glucose/glucose tolerance, EUC and TFT (Table 5.20). Together these tests accounted for 79.3% of pathology ordered by GPs for lipid disorder (Table 5.21). Between 2000–02 and 2006–08 the order rate of EUC and TFT increased significantly while the order rate of lipid tests decreased (Table 5.19). The latter represents a change in how GPs record the lipid test rather than a change in lipid ordering behaviour. GPs were more likely to record the specific lipid subfractions in 2000–02 (counted as multiple tests) whereas in 2006–08 they were more likely to record the lipid profile test (counted as a single test) (results not tabled).

Lipid tests were unanimously recommended, including the need to test lipid subfractions (i.e. total cholesterol, LDL cholesterol, HDL cholesterol and triglycerides).

Liver function testing was also recommended in all guidance documents to: determine presence of liver dysfunction as a cause of secondary lipid disorders,^{172,210-212,214,216,219,220} as a consideration in medication selection and safety,^{173,178,210-213,215,216,219,220} and/or in the monitoring of statin medications and selected other medications (e.g. combination therapy).^{173,210-214,216-220}

Glucose testing was recommended in most guidance documents to determine the presence of diabetes as a cause of secondary lipid disorder.^{172,173,210-214,216,219,220} Diabetes was also discussed in regard to cardiovascular risk as its presence increases the patient's risk and affects which lipid target is appropriate for the patient.^{178,210,211,213-220}

Assessment of renal function (EUC test) was recommended in most guidance documents to determine the presence of renal impairment as a cause of secondary lipid disorder.^{172,173,210-212,214,216,219,220} Renal function was also discussed as a consideration in medication selection,^{178,210-213,216-220} and in regard to cardiovascular

risk^{178,210,211,213,215,217,218} as this affects which lipid target is appropriate for the patient.

Assessment of thyroid function was recommended in most guidance documents to determine the presence of hypothyroidism as a cause of secondary lipid disorder, ^{172,173,210-212,214,216,219,220} and prior to initiating a statin.²²⁰

Tests with conditional support

The CK test was classified as having conditional support (Table 5.20). CK accounted for 4.3% of tests ordered for lipid disorders (Table 5.21) and increased significantly between 2000–02 and 2006–08 (Table 5.19).

CK testing was discussed in regard to medication (primarily statin) use to detect myopathy. Most guidance stated that routine CK monitoring was not necessary, but CK testing was recommended in patients with muscle symptoms^{178,210-213,216,219,220} and some guidance documents recommended taking a baseline measure prior to starting statins (or combination statin therapy) for future comparison in all patients^{173,178,210,211,220} or for high risk patients.²¹³

A few guidance documents recommended routine monitoring in certain high risk patients (e.g. renal disease, high dose statins, statin combination therapy).^{173,178,217-219} In documents where guidance about CK testing was not given, the potential for myopathy as an adverse effect of statin use was discussed.^{214,215}

Support unable to be determined

It was not possible to determine whether MBA were recommended in the guidance documents (see Section 5.3.4). MBA accounted for 4.3% of pathology ordered for lipid disorder (Table 5.21) and increased significantly over time (Table 5.19).

The MBA test itself was not recommended in any guidance documents. However, some tests that may be considered part of the MBA, such as EUC and LFT, were partially or completely supported in the management of lipid disorder (as discussed above).

Unsupported tests

The guidance documents did not mention FBC and PSA testing (Table 5.20). These tests accounted for 7.0% of pathology ordered for lipid disorders (Table 5.21) and the order rate of both tests increased significantly between 2000–02 and 2006–08 (Table 5.19). It may be possible that GPs ordered the FBC to assess the presence of

systemic lupus erythematosus,²²¹ a possible cause of secondary lipid disorders (mentioned in two guidance documents^{214,220}).

Key to Table 5.20

Colour	Description
	The document specifically recommended this test. Any notes within the cell indicate further detail. For example, a specific disease to test for within subset of patients; a specific test within a group.
	The document stated that this test should be considered. Any notes within the cell indicate further detail (e.g. a specific test to consider)
	Unable to determine guidance (see Section 5.3.4).
	The document specifically stated not to do this test. Additional information is supplied if certain conditions apply (e.g. specific clinical situations).
	Guidance document does not mention this test

	Per cent of lipid disorders											
Pathology test	pathology (<i>n</i> = 15,777)	NHF & CSANZ ¹⁷⁸	NCEP ^{210,211}	NICE ²¹²	SIGN ²¹³	AACE ²¹⁴	ESC ²¹⁵	Barrie et al. ²¹⁶	Genest et al. ^{217,218}	MoH ²¹⁹	ICSI ²²⁰	RCPA ¹⁷² Murtagh ¹⁷
Lipids*	50.2											
LFT*	12.4	Medication safety					Medication safety		Medication			Medication
Glucose/glucose/ tolerance*	e 9.6	CV risk			CV risk				CV risk			
FBC	6.0					SLE					SLE	
EUC*	5.1	CV risk and medication							CV risk and meds			
MBA*	4.3											
СК	4.3	Baseline & muscle sx	Baseline & muscle sx	Muscle sx	Muscle sx & high risk pts	Implied	Medication safety	Muscle sx	High risk medication	High risk medication	Baseline & muscle sx	Monitor in statin use
TFT*	2.0											
PSA*	1.0											

Table 5.20: Summary of guidance recommendations by most frequent individual test orders for lipid disorder, 2000–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: CV risk - cardiovascular risk; SLE - systemic lupus erythematosus; sx - symptom; pts - patients. Also see Abbreviations. Any notes within the coloured cells are described in detail in Section 5.6.5.

Evaluation of GP pathology ordering against guidance documents

Table 5.21 provides a summary of the individual tests recorded in BEACH and the level of support provided in the guidance documents for each. Of the tests/batteries ordered for lipid disorder in 2000–08:

- 79.3% were supported
- 8.6% had conditional support or support could not be determined
- 7.0% were not supported by the guidance documents.

The individual tests/batteries listed in Table 5.21 account for 94.9% of pathology ordered for lipid disorders because only the most commonly ordered tests for lipid disorder were evaluated.

Pathology test	Number	Per cent of all pathology for lipid disorders
Supported	12,519	79.3
Lipids*	7,919	50.2
LFT*	1,961	12.4
Glucose/glucose tolerance*	1,520	9.6
EUC*	801	5.1
TFT*	318	2.0
Conditional/unclear support	1,354	8.6
MBA*	683	4.3
СК	671	4.3
Unsupported	1,106	7.0
FBC	949	6.0
PSA*	157	1.0
Subtotal (n, % of total tests)	14,980	94.9
Total pathology tests	15,777	100.0

 Table 5.21: Summary of support for GP pathology ordering for the most frequent individual test orders for lipid disorder, 2000–08

Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. See Abbreviations.

In 2000–08, GPs ordered supported pathology tests at a rate of 49.6 tests/batteries per 100 lipid disorder problems, followed by tests with conditional support (5.4 per 100), and unsupported tests (4.4). Tests ordered for lipid disorder but not evaluated were ordered at a rate of 3.2 per 100 problems. The rate at which GPs ordered tests that were conditionally supported, unsupported and not evaluated increased significantly over time (Table 5.22).

	2000–08	2000–02	2006–08	
Level of support	Rate per 100 lipid disorder problems (95% Cl) (<i>n</i> = 25,248)	Rate per 100 lipid disorder problems (95% Cl) (n=5,782)	Rate per 100 lipid disorder problems (95% Cl) (n=6,629)	Change
Supported	49.6 (48.1–51.1)	48.8 (45.9–51.7)	50.3 (47.3–53.3)	_
Conditional/unclear support	5.4 (5.0–5.8)	3.7 (3.0–4.4)	6.7 (5.8–7.6)	↑
Unsupported	4.4 (4.0–4.7)	3.1 (2.6–3.7)	5.9 (5.0–6.7)	↑
Not evaluated	3.2 (2.9–3.4)	2.5 (2.0–3.0)	3.8 (3.1–4.4)	↑

Table 5.22: Rate of pathology ordering for lipid disorder by level of support, 2000–08,2000–02 and 2006–08

Note: CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/↓ indicates a statistically significant change, and — indicates no change.

GPs ordered only pathology tests that were supported or conditionally supported at 25.6% (95% CI: 24.9–26.3) of lipid disorder problems in 2000–08. A further 4.1% (95% CI: 3.8–4.4) of problems involved at least one unsupported test, and 70.3% (95% CI: 69.5–71.0) involved either no pathology tests (69.5%) or tests that were not evaluated (0.8%) (results not tabled).

There was no significant change in the proportion of lipid disorder problems that involved supported testing between 2000–02 (26.8%, 95% CI: 25.3–28.2) and 2006–08 (24.1%, 95% CI: 22.7–25.4). However the proportion of problems involving at least one unsupported test increased significantly over this time, from 3.0% (95% CI: 2.5–3.5) to 5.4% (95% CI: 4.6–6.2) (results not tabled).

When GPs ordered unsupported tests, the vast majority (98%) were accompanied by one or more supported/partially-supported tests. Unsupported tests were ordered alone at the remaining 2%. GPs ordered only tests that were unsupported at 0.1% of lipid disorder problems managed in 2000–08. This did not change over time (0.1% in 2000–02 and 2006–08) (results not tabled).

5.6.6 Discussion

GPs' pathology testing aligned well with that recommended in guidance documents, with 88% of tests ordered by GPs in the management of lipid disorders being recommended (with either partial or complete support). The statistically significant increase in the rate of GPs total pathology ordering for lipid disorders between 2000–02 and 2006–08 was reflected in the increased rate of tests that were partially supported, and unsupported in guidance documents. There was no change in the rate of supported tests.

Due to the increase in total rate of pathology ordering, the correlation between GPs' ordering and level of support, as a proportion of total tests should be considered. Supported tests decreased as a proportion of total tests between 2000–02 and 2006–08, and was counteracted by increases in partially supported and unsupported tests. Overall this suggests that the increase in GPs ordering resulted in testing being 'less' in line with recommendations.

Part of the decrease in the proportion of supported tests was due to a change in the way GPs' recorded lipid tests on the BEACH encounter forms. In 2000–02, GPs were more likely to record the specific lipid subfractions (counted as multiple tests) whereas in 2006–08 they were more likely to record the lipid profile test (counted as a single test). This may reflect the recommendation to test the entire lipid profile so that treatment decisions are based on all lipid subfractions. In the past, treatment recommendations were based on the total cholesterol level alone.^{178,222}

Criteria for funding may also contribute to changes in the pattern of lipid components being recorded by GPs in BEACH. Prior to November 2001, the MBS only funded HDL tests if the total cholesterol was found to be abnormal,²²³ suggesting that GPs ordered a HDL test separately after obtaining an abnormal total cholesterol result. Further, the limit of five pathology tests per BEACH recording form (see Section 5.10) may have an influence on recording. For example, as the number of tests recorded per problem has increased (see Chapter 4), GPs may record a lipid profile (one test) rather than specifying all the lipid subfractions (multiple tests) in order to have space available to record other tests.

Monitoring lipid levels

The treat-to-target approach to managing lipid levels was recommended (usually with LDL cholesterol targets) in most guidance documents.^{173,178,210,211,214-220}

Recommendations for monitoring lipid levels were based on the phase of management: an 'active' phase of management while trying to achieve targets and a monitoring phase.

In the active phase of managing lipid levels (response to diet and exercise and/or medications), most guidance documents recommended an interval of between 4 and 12 weeks for measuring response (i.e. retesting lipid levels) until the recommended target lipid levels were achieved.^{210,211,214,216-218,220}

The NICE and SIGN guidelines did not adopt the usual treat-to-target approach to managing lipids. The NICE guideline recommended that patients taking statins for primary prevention of cardiovascular disease should not have their lipid levels measured in response to statin use unless clinical judgement or patient preference indicate the need to review the lipid profile. For secondary prevention of cardiovascular disease, the authors provided 'desirable' total cholesterol and LDL levels to guide the intensification of treatment rather than as targets.²¹²

The SIGN guideline authors found there was insufficient evidence to support recommending lipid targets. Instead they recommended "intensive lipid lowering therapy" for high-risk patients taking lipid-lowering therapy for secondary prevention and "lifetime treatment with simvastatin 40mg" for those taking it for primary prevention. The authors advocated the same total cholesterol target as recommended by the National Health Service for Scotland as a minimum standard of care.²¹³

In their review of available evidence, Smellie et al. recommended an interval of 8 weeks (+/- 4 weeks) for monitoring lipid levels in the active phase of management, concluding that differences between guideline recommendations were unlikely to influence long-term outcomes.²²⁴ However, Bell et al. questioned the need to monitor patients taking lipid-lowering medications where randomised control trial data are available to give an indication of whether target will be achieved based on initial lipid levels.²²⁵ This approach assumes high levels of medication adherence and similarities between patients in everyday clinical settings and trial participants. There is concern that clinical trial data may not reflect the 'real world' clinical setting as trial participants are highly selected and levels of medication adherence are higher than in the general patient population. Smellie et al. concluded that further evidence

is needed before recommending changes in monitoring practices on the basis of clinical trial evidence.²²⁴

The recommended interval for testing in the long-term monitoring phase varied between the guidance documents. For patients whose lipid levels were being managed (i.e. those on lipid-lowering medications or at high cardiovascular risk), the recommended monitoring interval ranged from 3 to 12 monthly, with individual guidance documents often suggesting a range for retesting such as 4–6 monthly, 6–12 monthly.^{178,210,211,213,214,216-220}

Smellie et al., after reviewing available guidelines, recommended annual lipid tests for long-term monitoring. They also noted that most recommendations on testing intervals were consensus-based.²²⁴ In the guidance documents reviewed the level of evidence behind the recommended testing interval (for either the active or monitoring phase of management) was not stated.

Although there were differences among the guidance documents, the most commonly recommended interval for long-term monitoring of lipid levels was 6–12 months. However, recent evidence suggests that in patients for whom lipid levels are stable (within 0.5 mmol/L of target) the interval for monitoring should be every 3–5 years because more frequent testing is more likely to reflect measurement error than true change.²²⁶

This measurement error is a product of the analytical and intra-individual variation in lipid testing. Most of the guidance documents reviewed did not discuss intraindividual and analytical variation of lipid testing. Only two guidance documents^{212,216} discussed the level of expected variance in lipid measurements and the need for at least two tests before starting therapy. The NICE guideline highlighted that repeated measurement improves precision of the lipid testing, and was the only guideline to mention this in the context of monitoring, stating that monitoring is often based on one measurement, multiple testing being impractical in practice. Murtagh, AACE and NCEP also stated the need for at least two tests before starting therapy but the reason for this was not provided.^{173,210,211,214}

The only time authors of guidance documents appear to consider precision is at the time of initial assessment of lipid levels before therapy is started. Although it is important to establish an accurate baseline measurement of lipid levels (because lipid lowering therapy is life-long), the recommendation to initiate lipid-lowering therapy

in guidance documents was usually based on a patient's cardiovascular disease risk which incorporates multiple factors, lipid levels being just one of these factors. This puts less importance on the precision of the lipid measurement at initiation. However, in monitoring the response to therapy, lipid levels are usually the sole indicator.

Further, the majority of management of lipid disorders in general practice, and lipid tests ordered in their management, are for monitoring. Therefore, information on variation in lipid levels is arguably of most importance in monitoring. The amount of intra-individual and analytical variation in monitoring lipid levels has been discussed by other authors,^{224,226-229} but not incorporated into guidance documents. More information on the degree of variation is needed in guidance documents to inform GPs of the likelihood of measurement error when monitoring lipid levels should be 3–5 yearly²²⁶ is too recent to have been incorporated into the reviewed guidance documents or to be reflected in the lipid test order rate observed in this study. This evidence may be incorporated in future guidance. However, further investigation of the impact of a longer monitoring interval may be needed – particularly the impact on patient adherence as some guideline authors discussed frequent monitoring of lipid levels as a tool to aid improvement in patient adherence to therapy.^{178,220}

Monitoring statin use

The majority of lipid-lowering medications prescribed in the management of lipid disorders in the current study were plain statins (91% of medications for lipid disorders). The guidance documents reviewed primarily referred to monitoring in relation to statin therapy.

Liver function

There was reasonable agreement among guidance documents on the need to test LFT before initiating statins, after commencing and after increasing dose. However, guidance varied on the need for ongoing monitoring.

Other authors have published recommendations about testing in the context of statin use. The United States (US) National Lipid Association Statin Safety Assessment (NLASSA) taskforce recommended ongoing monitoring but noted there was little evidence to support it.²³⁰ Recognition of the lack of evidence for long-term monitoring was echoed by others.^{212,224,227} The NICE guideline provided a consensus recommendation that LFT testing was needed pre-treatment, within 3 months of starting medication, and a year after that. Further monitoring was not recommended unless clinically indicated.

The majority of guidance documents recommended monitoring of LFT but offered no comment on frequency or duration of monitoring.^{173,178,214,220} This was also the case in the Australian medication guideline 'eTG'.¹⁷⁹

The order rate of LFT did not change over the period of this study, suggesting GPs have not changed their behaviour in regard to monitoring LFTs in the management of lipid disorders.

Creatine kinase

CK testing was not recommended in routine monitoring of statin use in most guidance documents. The exception was Murtagh, who recommended ongoing monitoring of CK in statin use.¹⁷³

Other authors provided more detail. Smellie et al recommended a baseline CK test prior to initiating a statin for two reasons: if baseline CK is elevated statin should not be started, and if CK testing is indicated in the future (e.g. muscle symptoms develop) results can be compared with baseline.^{224,227} The NLASSA taskforce stated that baseline testing in patients at high risk for muscle toxicity may be considered but routine baseline testing for all patients commencing a statin was not recommended.²³⁰

In the reviewed guidance documents, CK testing was commonly recommended in patients who develop muscle symptoms. The NLASSA taskforce stated that muscle symptoms or increased CK were likely to be caused by other aetiologies and should be investigated by health professionals.²³⁰ If rhabdomyolysis is suspected serum creatinine should also be measured.

GPs ordering of CK increased significantly between 2000–02 and 2006–08, even though the vast majority of guidance recommended against routine monitoring of CK.

Causes of secondary dyslipidaemia

Testing for causes of secondary dyslipidaemia was recommended in most guidance documents prior to starting lipid-lowering therapy. The exceptions were: the Australian NHF & CSANZ guideline;¹⁷⁸ and the cardiovascular prevention

guidelines from ESC²¹⁵ and SIGN.²¹³ The SIGN guideline recommended that clinicians 'consider secondary causes of dyslipidaemia' but details of these secondary causes were not listed.²¹³ For the two cardiovascular prevention guidelines^{213,215} this omission is possibly because they did not solely provide guidance on the management of lipid disorder, but included all aspects of cardiovascular prevention (e.g. blood pressure, antiplatelet therapy).

The conditions commonly listed as secondary causes that involved pathology tests were hypothyroidism, renal disease, liver disease and diabetes. Testing to identify causes of secondary dyslipidaemia were mentioned as part of the initial evaluation of the patient. In contrast, the majority (> 90%) of tests used to identify these conditions were ordered by GPs for ongoing management.

Of the tests related to secondary causes of lipid disorders, the rate of thyroid and kidney function testing (TFT and EUC) increased significantly. The proportion of lipid problems that were newly diagnosed increased between 2000–02 and 2006–08 and this may have contributed to some of the increased rates of EUC, and TFT as these were recommended as part of the initial investigations. However, this is unlikely to account for the entire increase in these tests.

No guidance was provided on whether there is a need to periodically retest for these secondary causes of lipid disorders in the future. However, liver function testing was recommended in monitoring of medication use; and ongoing testing for diabetes or kidney disease may be undertaken as part of reassessment of cardiovascular risk (as these conditions increase cardiovascular risk) although the need to test glucose and kidney function was not explicitly stated. Kidney function testing, specifically creatinine levels were mentioned in guidance documents for diagnosis of rhabdomyolysis as a rare adverse effect of statin therapy. But the NLASSA taskforce stated that it is not necessary to monitor serum creatinine or proteinuria routinely during statin therapy.²³⁰

Most guidance documents recommended thyroid testing in all patients regardless of initial lipid levels. In contrast, Smellie et al. recommended thyroid testing only if the initial total cholesterol level was > 8.0 mmol/L.²³¹ The need to reassess thyroid function was not mentioned in any of the guidance documents.

Authors of guidance documents should consider incorporating information on whether these conditions, particularly thyroid disease, are likely to occur in the future (e.g. increasing prevalence with age) and whether subsequent diagnosis of the condition is likely to affect management of lipid disorders, to inform whether repeated testing is needed for patients without thyroid disease at initial assessment.

Tests for which guidance was lacking

The significant increases in GPs' ordering rates of FBC and MBA tests and marginal increase in PSA tests are not directly related to guidance provided in the management of lipid disorders. It is unclear why the rates of these tests increased over the period of this study. GPs' increased use of FBC and MBA tests are discussed further in Section 5.10.

Other factors contributing to the increased management rate of lipid disorder

The guideline documents recommended initiating lipid therapy and management of lipid levels in the context of the patient's absolute cardiovascular risk.

Lipid targets have become lower over time as knowledge about cardiovascular risk improved. The majority of guidance documents recommended LDL targets, with lower targets recommended for patients with higher cardiovascular risk. Lower targets may have influenced GPs' management rate of lipid disorder problems and pathology ordering rates because they are potentially harder, and may take longer, to achieve. More frequent visits and pathology testing were recommended while actively trying to achieve a target (titrating medications). While the guidance documents acknowledged that targets may not be achievable in all patients and should be adjusted to the individual patient, it is likely that a change in target that requires intensifying treatment will result in increased management and testing rates. Changes in policies are also likely to have contributed to the increased management rate of lipid disorders seen in this study. In particular, the decision to broaden the PBS criteria for subsidised lipid-lowering medications in October 2006 to allow access on the basis of cardiovascular risk rather than measured lipid levels alone,²⁰⁸ and the introduction of the MBS item for health checks in patients aged 45-49 years in February 2006 to prevent or delay the onset of chronic disease²³² may have contributed to the increased management rate of lipid disorders.

The increased focus on total cardiovascular risk does not appear to have altered GPs' pathology ordering behaviour in the management of lipid disorders. The data suggest that GPs have not changed the rate at which they monitor lipids in response to

therapy. GPs have also not changed the rate of glucose testing (e.g. assessing presence of diabetes/impaired glucose tolerance) or liver function testing (i.e. presence of liver disorder/monitoring side effect of statin) when managing lipid disorder. Of the individual tests that increased (FBC, EUC, MBA, CK, TFT and PSA) for lipid disorder problems, only EUC (kidney function) may be associated with assessment or management of cardiovascular risk.

Summary

The majority of tests ordered by GPs in the management of lipid disorders were recommended in guidance documents. However, the increased rate of pathology tests ordered for lipid disorders reflected GPs' increased use of tests that were partially supported and unsupported in guidance documents. This highlights the importance of targeting a reduction in unsupported tests, even if the absolute reduction in testing would be modest.

5.7 Health check

5.7.1 Background

Prevention is one of the cornerstones of Australian general practice. Promoting the prevention of disease was one of the founding objectives of the Australian College of General Practitioners (now the RACGP) in 1958.²³³

Primary care is the health sector with the most contact with the population and the most opportunity to provide preventive care. In 2009–10, 83% of the population visited a GP at least once (DoHA, personal communication, June 2010) and per head of population there were an average 5.2 GP visits funded by Medicare Australia.¹³ Through activities such as vaccination, health promotion, and risk management (identification and reduction) GPs have the opportunity to prevent disease, and to detect disease in its early stages. Therefore they are involved in all types of preventive care.

In Australia, the recent focus of preventive care has been the prevention of chronic diseases because our ageing population is expected to increase the burden on the health system.²³⁴ Recent initiatives include the National Chronic Disease Strategy¹⁸⁷ (2005) and the National Preventative Health Strategy²³⁵ (2009), both acknowledged the role of primary health care in preventive care and provided funding to support GPs in preventing, detecting and managing chronic disease.^{187,235}

MBS item numbers for health assessments in general practice have been introduced to encourage preventive action in target population groups. Examples include: the annual health assessment in patients aged 75 years and over (1999);²³⁶ the health assessment for people aged 45–49 years who are at risk of developing chronic disease (2006);²³² the assessment of people aged 40–49 years with a high risk of T2D (2008).¹⁸ Pathology tests are used by GPs in preventive activities to identify risk factors, and to diagnose and monitor disease.

5.7.2 Management rate in Australian general practice

In this section 'health check' problems include those labelled as unspecified health check-ups in patients aged 15 years and over.

In BEACH, 'heath check' was managed at 8,113 encounters by 3,707 GPs in 2000–08, at a rate of 1.2 per 100 encounters (Table 5.23). This equates to approximately 1.0 million encounters per year with patients aged 15 years or more where health checks were managed by GPs in Australia.

As reported in Chapter 4, there was a significant increase in the management rate of health checks, between 2000–02 and 2006–08 (Table 5.23).

	20	000–08	20	000–02	200		
Variable	Number	Rate per 100 encounters (95% Cl) (n=682,932)	Number	Rate per 100 encounters (95% Cl) (n=171,136)	R Number (ate per 100 encounters (95% CI) <i>n</i> = 165,439)	Change
General practitioners	3,707		872		1,028		
Health check encounters	8,113		1,845		2,463		
Health check problems managed	8,120	1.2 (1.1–1.2)	1,846	1.1 (1.0–1.2)	2,464	1.5 (1.4–1.6)	↑

Table 5.23: Summary of	health check data	set in patients	aged 15+ year	s, 2000–08,
2000–02 and 2006–08				

Note: Data about health check problems managed are drawn from Chapter 4, Table 4.3. Cl – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change.

5.7.3 Pathology ordered for health check

Pathology was ordered at a rate of 147.9 tests/batteries per 100 'health check' contacts with patients aged 15 years and over in 2000–08. Almost half of the contacts (49.5%) resulted in at least one pathology order. Once the decision to order pathology was made the GP ordered on average 2.99 tests/batteries per tested 'health check' contact (Table 5.24).

The increase in the rate of pathology ordering between 2000–02 and 2006–08 was due to a significant increase in the number of tests ordered once the decision to order tests was made (as reported in Chapter 4). There was no change in the likelihood that at least one pathology test would be ordered in the management of health check problems (Table 5.24).

	2000–08		2000–02		2006–08		
Variable	Number	Per cent / Rate of health check problems (95%Cl) (n=8,120)	Number	Per cent / Rate of health check problems (95%Cl) (n=1,846)	Number	Per cent / Rate of health check problems (95%Cl) (n=2,464)	Change
Pathology (Rate per 100 health check problems)	12,008	147.9 (142.2–153.6)	2,252	122.0 (110.7–133.3)	4,407	178.9 (167.3–190.4)	↑
At least one pathology order (% health check problems)	4,023	49.5 (48.0–51.1)	900	48.8 (45.1–52.5)	1,319	53.5 (50.7–56.3)	_
Number of tests/ batteries per 100 tested health check problems		298.5 (291.6–305.3)		250.1 (236.1–264.2)		334.1 (322.8–345.4)	↑

Table 5.24: Summary of pathology ordering for health check (patients 15+ years),2000–08, 2000–02 and 2006–08

Note: Pathology ordering data from 2000–02 and 2006–08 are drawn from Chapter 4, Table 4.4. CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/ indicates a statistically significant change.

Types of pathology tests/batteries ordered

Table 5.25 shows the rate of pathology tests/batteries ordered for 'health check'

problems in patients aged 15 years and over in 2000-08 by MBS groups and the

most common individual types of tests ordered.

Chemistry tests were the group most often ordered (102.2 per 100 health check

contacts) and the most common were:

- lipid tests (29.7 per 100 'health check' contacts)
- glucose/glucose tolerance tests (18.5)
- LFTs (11.7)
- EUC tests (10.3)
- MBA tests (8.7).

Haematology tests (25.3 per 100 contacts), in particular FBC (22.7), and

cytopathology tests (9.2), in particular Pap smear (9.1), were commonly ordered as part of the management of 'health check' (Table 5.25).

Pathology test	Number	Rate per 100 health check problems (95% Cl) (<i>n</i> = 8,120)
Chemistry	8,301	102.2 (97.8–106.6)
Lipids*	2,413	29.7 (28.4–31.1)
Glucose/glucose tolerance*	1,500	18.5 (17.3–19.6)
LFT*	952	11.7 (10.7–12.7)
EUC*	839	10.3 (9.5–11.2)
MBA*	705	8.7 (7.8–9.6)
PSA*	663	8.2 (7.5–8.9)
TFT*	583	7.2 (6.5–7.9)
Ferritin*	203	2.5 (2.1–2.9)
Other chemistry*	144	1.8 (1.2–2.3)
Haematology	2,055	25.3 (23.9–26.7)
FBC	1,839	22.7 (21.4–23.9)
ESR	153	1.9 (1.5–2.2)
Cytopathology	745	9.2 (8.0–10.4)
Pap smear*	740	9.1 (7.9–10.3)
Microbiology	574	7.1 (5.8–8.4)
Hepatitis serology*	161	2.0 (1.5–2.5)
Other tests NEC	199	2.5 (2.0–2.9)
Other pathology groups	134	
Total pathology tests	12,008	147.9 (142.2–153.6)

Table 5.25: Rate of pathology test orders for health check (patients 15+ years) by MBS groups and most frequent individual tests within each group, 2000–08

^{*} Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations.

Changes in types of pathology tests/batteries ordered

Table 5.26 shows the most common pathology tests/batteries ordered for health check in 2000–02 and in 2006–08. Listed below are the tests for which significant change in GPs' order rate occurred between 2000–02 and 2006–08 (listed in decreasing test rate order). There were significant increases in the order rate of:

- lipid tests—36% increase
- FBCs—72% increase
- glucose/glucose tolerance—40% increase
- LFTs—100% increase
- EUC tests—128% increase
- MBA tests—60% increase

• TFTs—121% increase

Other chemistry*

Hepatitis serology*

Haematology

Microbiology

Cytopathology

Pap smear*

Other tests NEC

Simple basic tests

Occult blood test

Total pathology tests

Other pathology groups

FBC

ESR

- PSA tests—95% increase
- 'other chemistry' tests—233% increase
- occult blood tests—280% increase (Table 5.26).

There was also a significant 58% decrease in the order rate of Pap smears as part of health checks between 2000–02 and 2006–08 (Table 5.26).

2000-02 2006-08 Rate per 100 Rate per 100 health check health check problems (95% CI) problems (95% CI) Pathology test Number (n = 1,846)Number (n = 2,464) Change Chemistry 1,423 77.1 (68.9-85.3) 3,182 129.1 (120.3-138) ↑ 26.2 (23.4-29.1) Lipids* 484 877 35.6 (33.1-38.1) Υ Glucose/glucose tolerance* 278 15.1 (12.6-17.5) 519 21.1 (18.8-23.3) LFT* 8.1 (6.1-10.0) 16.2 (14.0-18.4) 149 399 EUC* 107 5.8 (4.5-7.1) 324 13.2 (11.4-14.9) ተ 6.7 (5.0-8.4) 10.7 (8.9-12.5) MBA* 124 264 PSA* 101 5.5 (4.1-6.8) 263 10.7 (9.2-12.1) TFT* 4.7 (3.2-6.1) 10.4 (8.8-11.9) 86 255 Υ Ferritin* 33 1.8(1.0-2.5)82 3.3 (2.4-4.3)

0.9(0.4 - 1.4)

2.2 (1.5-2.9)

7.0 (4.5-9.6)

2.0 (1.1-2.9)

2.3 (1.4-3.2)

0.5 (0.2-0.9)

0.5 (0.2-0.9)

14.6 (11.4-17.8)

14.6 (11.4-17.8)

20.0 (17.1-22.9)

16.9 (14.4-19.4)

74

779

716

42

161

49

149

149

78

46

46

58

3.0 (1.5-4.5)

1.7 (1.0-2.4)

6.5 (4.4-8.6)

2.0 (0.9-3.1)

6.1 (4.3-7.8)

6.1 (4.3–7.8)

3.2 (2.1-4.2)

1.9 (1.2-2.5)

1.9 (1.2-2.5)

4,407 178.9 (167.3-190.4)

31.6 (28.8-34.4)

29.1 (26.4–31.7)

Υ

 $\mathbf{1}$

↓

17

369

312

41

130 37

270

270

42

10

10

18

Table 5.26: Rate of pathology test orders for health check (patients 15+ years) by MBS
groups and most frequent individual tests within each group, 2000–02 compared with
2006–08

Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations. Shading indicates a statistically significant change. The direction and type of change is indicated for each measure between 2000–02 and 2006–08: ↑/♥ indicates a statistically significant change, and — indicates no change.

2,252 122.0 (110.7-133.3)

5.7.4 Guidance documents for health check

The five preventive health guidance documents reviewed for health check are listed in Box 5.4.

Box 5.4: Guidance documents reviewed for health checks					
Title	Year	Author	Abbreviated to		
Guidelines for preventive activities in general practice (The Red Book) ¹⁷⁷	2009	Royal Australian College of General Practitioners (RACGP)	RACGP		
US Preventive Services Task Force recommendations ²³⁷	2008	United States Preventive Services Task Force (USPSTF)	USPSTF		
Health Care Guidelines: Preventive services for adults ²³⁸	2008	Grenz K, Mortinsen R, Pine D, Solberg L, Wilkinson JM, Harvey L et al. <i>Institute for</i> <i>Clinical Systems Improvement (ICSI)</i> <i>guideline</i> [United States of America]	ICSI		
Canadian Task Force on Preventive Health Care recommendations ²³⁹	1994– 2005	Canadian Task Force on Preventive Health Care (CTFPHC)	CTFPHC		
Health Screening ²⁴⁰	2004	Health Promotion Board, Ministry of Health (MoH) [Singapore]	МоН		

5.7.5 Extent of alignment between GP testing and guidance documents

In Table 5.27, the frequently ordered pathology tests/batteries for health check are categorised by level of support in the guidance documents listed in Box 5.4.

Supported tests

The three tests that were recommended in the majority of guidance documents were lipids, Pap smears and faecal occult blood tests (Table 5.27). Together these tests accounted for 26.3% of tests/batteries ordered by GPs for health checks (Table 5.28). Between 2000–02 and 2006–08, the order rate of lipid and faecal occult tests increased significantly, but the order rate of Pap smears for health check problems decreased (Table 5.26). While the rate of Pap smears decreased for health checks, the total rate of Pap smears (for all problems) increased between 2000–02 and 2006–08 (see Chapter 4).

There was strong agreement among the guidance documents that lipids should be measured in selected patients. Recommendations were made on the basis of identifying patients who are at increased risk of cardiovascular disease and are therefore most likely to benefit from testing. The evidence for screening in male patients is stronger than for females.^{177,237} Recommendations for screening intervals

were made in three guidance documents.^{177,237,240} The RACGP and USPSTF recommended intervals based on risk, 5 yearly for the lower risk group and 1–2 yearly for the higher risk groups.^{177,237} The MoH guideline recommended an interval of 3 years if lipid levels were within the 'desirable' range.²⁴⁰

The CTFPHC²³⁹ was the only document to conclude that there was insufficient evidence to recommend lipid testing. However, this guideline was released in 1994 and considering the new evidence and new pharmacological treatments now available, can be considered out of date.

There was unanimous agreement among guidance documents that cervical cancer screening should be routinely undertaken in all sexually active females until the age of 65 or 69 years. Testing older women was recommended if patients were not previously screened^{177,237} or they had a new sexual partner.²³⁸ Frequency of recommended Pap smear testing varied based on risk. For average risk patients testing every 2 years¹⁷⁷ or 3 years²³⁷⁻²⁴⁰ was recommended.

Guidance documents also unanimously recommended that colorectal cancer screening should be routinely undertaken in patients aged 50 years and over and earlier for those at increased risk. The faecal occult blood test (FOBT) was the only pathology test that GPs could order for colorectal screening. Other screening options given in guidance documents were sigmoidoscopy or colonoscopy. The type, frequency, and age of onset of screening depended on the patient's risk of colorectal cancer. For patients at average risk, all guidance documents recommended FOBT from the age of 50 years either annually²³⁷⁻²⁴⁰ or biannually.^{177,237,239} Alternatively sigmoidoscopy or colonoscopy were also recommended.²³⁷⁻²³⁹ For patients at higher risk (e.g. family history of bowel cancer) the primary screening tools were colonoscopy or sigmoidoscopy.^{177,237,239} In high risk patients the potential need for genetic testing was discussed in two guidance documents.^{177,237}

Tests with conditional support

There were three types of tests that had conditional support or mixed levels of support in the guidance documents: glucose/glucose tolerance, hepatitis tests and tests for other sexually transmitted infections (STI) (Table 5.27). Together these tests accounted for 15.4% of the tests ordered for health checks (Table 5.28). Over the period of this study, the order rate of glucose/glucose tolerance tests increased
significantly, and the rate of hepatitis testing (Table 5.26) and other STI tests did not change significantly (results not tabled).

Testing for diabetes mellitus (using glucose testing) was recommended for selected high risk patients in three guidance documents.^{177,237,239} MoH provided stronger support, recommending screening for all patients from the age of 40 years and for patients with known risk factors from 30 years.²⁴⁰ In contrast, ICSI stated that testing was not recommended in asymptomatic patients but did not provide further guidance.²³⁸

Screening for STIs (including chlamydia, HIV, and hepatitis) was recommended in most guidance documents for high risk patients.^{177,237-240} Most also recommended chlamydia testing for all females aged less than 25 years in addition to other high risk groups.^{177,237-239} The USPSTF recommended against routine screening for Hepatitis B or C. For Hepatitis B this was because screening strategies have poor predictive value due to difficulty in accurately identify at risk individuals, and for Hepatitis C there was insufficient evidence to recommend screening.²³⁸

Support unable to be determined

It was not possible to determine whether MBA and 'other chemistry' tests were recommended in the guidance documents (see Section 5.3.4). Together these two tests accounted for 7.1% of GP pathology ordered for health checks (Table 5.28). Both increased significantly between 2000–02 and 2006–08 (Table 5.26).

The ICSI guideline specifically recommended against use of routine laboratory testing (including blood chemistry panels) without clinical suspicion of disease.²³⁸

Unsupported tests

The guidance documents did not recommend testing of FBC, LFT, EUC, PSA, TFT, ferritin and ESR (Table 5.27). These tests accounted for 43.6% of pathology ordered by GPs in the management of health check (Table 5.28). Between 2000–02 and 2006–08, the order rate of most of these tests increased significantly (with the exception of ferritin and ESR tests).

All guidance documents recommended against routine testing of PSA to detect prostate cancer.^{177,237-240} RACGP recommended against the test but stated that men should be informed of risks and benefits to make an informed choice.¹⁷⁷ Both USPSTF and RACGP highlighted that PSA testing was not recommended in men

aged 75 years and over,^{177,237} and USPSTF stated that there was insufficient evidence to make a recommendation in men aged less than 75 years.²³⁷ MoH recommended screening in high risk men aged more than 50 years who have had close relatives diagnosed with prostate cancer when aged less than 60 years.²⁴⁰

EUC tests were supported in only one document, the RACGP recommending annual screening using the estimated glomerular filtration rate for high risk patients. High risk patients were defined as those with hypertension, obesity, diabetes, family history of kidney disease, and Indigenous patients aged more than 35 years.¹⁷⁷ The other guidelines did not discuss EUC testing nor kidney disease with one exception. The 1993 CTFPHC recommended against screening asymptomatic adults using a dipstick to identify chronic renal failure because there was no efficacious, non harmful treatment available for early stages of the disease course.²³⁹

Screening for thyroid disease was not supported in most guidance. RACGP specifically recommended against the use of TFT as a screening test.¹⁷⁷ USPSTF, CTFPHC and ICSI stated that there was insufficient evidence to make a recommendation for or against screening in asymptomatic adults.²³⁷⁻²³⁹

LFT, ferritin and ESR were not mentioned in any of the guidance documents. FBC was also not directly mentioned in any document. However, ICSI recommended against use of routine laboratory testing (including "haemoglobulin and haematocrit screening") without clinical suspicion of disease.²³⁸ These would be considered part of the FBC in Australia, suggesting that ICSI did not support routine use of the FBC.

Colour	Description
	The document specifically recommended this test. Any notes within the cell indicate further detail. For example, a specific disease to test for within subset of patients; a specific test within a group.
	The document stated that this test should be considered. Any notes within the cell indicate further detail (e.g. a specific test to consider)
	Unable to determine guidance (see Section 5.3.4).
	The document specifically stated not to do this test. Additional information is supplied if certain conditions apply (e.g. specific clinical situations).
	Guidance document does not mention this test

Kev	to	Table	5.27
IXUY	w	Lanc	3.41

Pathology test	Per cent of path $(n = 12,008)$	RACGP ¹⁷⁷	USPSTF ²³⁷	ICSI ²³⁸	CTFPHC ²³⁹	MoH ²⁴⁰
Lipids*	20.1	> 45 yrs or high risk	Not in F without CHD risk	M > 34 yrs & F > 45 yrs	Insufficient evidence (1994)	>40 yrs or high risk
FBC	15.3			Not as screening		
Glucose/glucose tolerance*	12.5	Patients at increased risk	Yes in HT. Consider for CVD and dyslipidaemia Not in asymptomatic pts	Not recommended in asymptomatic pts	Yes in HT and dyslipidaemia (2005)	> 40 yrs or > 30 yrs and risk factors
LFT*	7.9					
EUC*	7.0	High risk pts				
Pap smear*	6.2	Sexually active until 69 yrs	Sexually active until 69 yrs	21 to 64 yrs & 65+ if new sexual partner	Sexually active until 65 yrs (1994)	F > 25 yrs
MBA*	5.9			Not as screening		
PSA*	5.5	Not recommended. Pt choice (50– 75 yrs) informed of risk & benefit.	Insufficient evidence in M<75 yrs and NO in M>75 yrs	Not recommended	Not recommended (1994)	FHx onset < 60 yrs
TFT*	4.9	Low prevalence even with FHx	Insufficient evidence	Insufficient evidence	Insufficient evidence (1994)	
Ferritin*	1.7					
STI tests (exclud hepatitis) ^(à)	ing 1.6	High risk pts and Chlamydia in F < 25 yrs	High risk pts and Chlamydia in F < 25 yrs	High risk pts and Chlamydia in F < 25 yrs	High risk pts (1994) and Chlamydia in F < 25 yrs (1996)	High risk pts
Hepatitis serolog	y* 1.3	High risk / request STI check	Only in pregnant women	High risk pts		
ESR	1.3					
Other chemistry*	1.2					
Occult blood test	^(b) 0.9	> 50 to 75 yrs or > 25 yrs in high risk pts	> 50 to 75 yrs	> 50 yrs or > 45 yrs in high risk pts	> 50 yrs or earlier in high risk pts (2001)	> 50 yrs

Table 5.27: Summary of guidance recommendations by most frequent individual test orders for health check, 2000–08

(a)

STI tests include tests for chlamydia, HIV, and 'STI screen'. Hepatitis is excluded because it is listed separately. Occult blood tests accounted for < 1% of pathology in 2000–08. It is included because the order rate increased significantly between 2000–02 and 2006–08. In 2006–08 it accounted for 1.0% of pathology. (b) * Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Path - pathology; M - male; F - female; yrs - years; pts - patients; HT - hypertension; CVD - cardiovascular disease; FHx - family history; STI - sexually transmitted infection; also see Abbreviations. Any notes within the coloured cells are described in detail in Section 5.7.5.

Evaluation of GP pathology ordering against guidance documents

Table 5.28 provides a summary of the individual tests and the level of support provided in the guidance for each. Of the tests/batteries ordered by GPs for health check in 2000–08:

- 26.3% were supported
- 22.5% had conditional support or support could not be determined
- 43.6% were not supported by the guidance documents (Table 5.28).

The individual tests/batteries listed in Table 5.28 account for 93.2% of pathology ordered by GPs for health checks because only the most common tests ordered for health checks were evaluated.

Pathology test	Number	Per cent of pathology for health check
Supported	3,152	26.3
Lipids*	2,413	20.1
Pap smear*	740	6.2
Occult blood test ^(a)	104	0.9
Conditional/unclear support	2,707	22.5
Glucose/glucose tolerance*	1,500	12.5
MBA*	705	5.9
Other STI testing (including Chlamydia, HIV, STI screen)	197	1.6
Hepatitis serology*	161	1.3
Other chemistry*	144	1.2
Unsupported	5,232	43.6
FBC	1,839	15.3
LFT*	952	7.9
EUC*	839	7.0
PSA*	663	5.5
TFT*	583	4.9
Ferritin*	203	1.7
ESR	153	1.3
Subtotal (n, % of total tests)	11,196	93.2
Total pathology tests	12,008	100.0

Table 5.28: Summary of support for GP pathology ordering for the most frequent individual test orders for health check (patients aged 15+ years), 2000–08

(a) Occult blood tests accounted for < 1% of pathology in 2000–08. It is included because the order rate increased significantly between 2000–02 and 2006–08. In 2006–08 it accounted for 1.0% of pathology.

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. See Abbreviations.

In 2000–08, GPs ordered unsupported pathology tests at a rate of 64.5 tests/batteries per 100 health check problems, followed by supported tests (40.1 per 100), and tests with conditional support (33.4). Tests ordered for health check but not evaluated were ordered at a rate of 10.1 per 100 problems. The rate at which GPs ordered unsupported and conditionally supported tests increased significantly over time (Table 5.29).

	2000–08	2000–02	2006–08	
Level of support	Rate per 100 health check problems (95% CI) (<i>n</i> = 8,120)	Rate per 100 health check problems (95% Cl) (<i>n</i> =1,846)	Rate per 100 health check problems (95% Cl) (n=2,464)	Change
Supported	40.1 (38.3–41.9)	41.4 (37.1–45.6)	43.5 (40.4–46.6)	_
Conditional/unclear support	33.4 (31.5–35.2)	26.4 (23.3–29.6)	39.1 (35.7–42.6)	↑
Unsupported	64.5 (61.1–67.9)	44.9 (38.4–51.4)	84.5 (77.4–91.6)	↑
Not evaluated	10.1 (9.0–11.1)	9.3 (7.0–11.6)	11.8 (9.8–13.9)	_

Table 5.29: Rate of pathology ordering for health check by level of support, 2000–08,2000–02 and 2006–08

Note: CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/↓ indicates a statistically significant change, and — indicates no change.

GPs ordered only pathology tests that were supported or conditionally supported at 17.1% (95% CI: 15.8–18.3) of health check problems in 2000–08. A further 30.2% (95% CI: 28.8–31.5) of health check problems involved at least one unsupported test, and 52.8% (95% CI: 51.2–54.4) involved either no pathology tests (50.4%) or tests that were not evaluated (2.4%) (results not tabled).

The proportion of health check problems that involved supported testing decreased significantly from 23.9% (95% CI: 20.9–26.9) in 2000–02 to 12.6% (95% CI: 10.8–14.5) in 2006–08. In parallel, the proportion of problems involving at least one unsupported test increased from 22.3 % (95% CI: 19.6–25.0) to 38.5% (95% CI: 35.7–41.2) (results not tabled).

When GPs ordered unsupported tests, the majority (89%) were accompanied by one or more supported/partially supported tests. Unsupported tests were ordered alone at the remaining 11%. GPs ordered only tests that were unsupported at 3.3% of health check problems in 2000–08. This did not change significantly over time 2.7% (95% CI: 1.9–3.5) in 2000–02 and 3.4% (95% CI: 2.6–4.2) in 2006–08 (results not tabled).

5.7.6 Discussion

GPs' pathology testing did not align well with that recommended in guidance documents, only about half of tests ordered by GPs for health checks (in patients aged 15 years and over) being partially or completely supported. The statistically significant increase in the rate of GPs' total pathology ordering for health checks between 2000–02 and 2006–08 was reflected in the increased rate of tests that were partially supported, and those that were unsupported in guidance documents. There was no change in the order rate of supported tests.

Due to the increase in total rate of pathology ordering, the relationship between GPs' ordering and level of support, as a proportion of total tests should be considered. Supported tests decreased as a proportion of total tests between 2000–02 and 2006–08, and was counteracted by an increase in unsupported tests. Overall this suggests that the increase in GPs' ordering resulted in testing being 'less' in line with recommendations.

Comments on guidance documents

There were relatively few guidance documents for preventive health care available for review. Further, of the five guidance documents reviewed, two^{239,240} would be considered out of date.

The CTFPHC²³⁹ guideline recommendations dated from 1994, 1996, 2001 and 2005. The year of each recommendation is provided in Table 5.27, and those from the earlier years were out of date (although not formally withdrawn). Despite this, the CTFPHC recommendations were extensively referenced throughout the literature. MoH guidelines are withdrawn after 5 years (unless updated), as the MoH guideline²⁴⁰ (published in June 2004) was not updated it was withdrawn in June 2009.

Unsupported pathology tests in guidance documents

Prostate cancer screening

Screening for prostate cancer (involving the PSA test) in the general male population was not recommended in any of the reviewed guidance documents (including the Australian RACGP guideline¹⁷⁷). Since these documents were released, interim results from two clinical trials on prostate cancer screening outcomes have been

published (in 2009 and 2012). The US and European studies provided inconclusive evidence of whether screening for prostate cancer was worthwhile.

In 2009, the European study reported after an average 9-year follow-up that screening reduced deaths from cancer by 20% (2.94 deaths per 1,000 in the screened men compared with 3.65 per 1,000 in the control group). However, there was a high rate of over-diagnosis: 1,410 patients would need to be screened and 48 patients treated, in order to save one life.²⁴¹ In contrast, the US study reported no difference in mortality from prostate cancer was seen in the screened group compared with the control group after a 7–9 year follow-up.²⁴² The differences in the results of the two studies are partially explained by differences in the screening protocol used (PSA was tested annually in the US study and every 4 years in the European study), and the proportion of the control group who received PSA testing as part of 'usual care', which was higher in the US study.

In early 2012, results from both the above European and US studies were published using data after a longer follow-up period—both reported no significant change from the results reported in 2009.^{243,244} Independent reviews of the 2009 and 2012 publications concluded that the results were unlikely to change recommendations for prostate cancer screening, nor current practice.²⁴⁵⁻²⁴⁷

Authors of the guidance documents reviewed in the current study have not changed their recommendations in response to these publications. Similarly other organisations (e.g. Cancer Council of Australia,²⁴⁸ international governments) still recommend against routine screening for prostate cancer.²⁴⁹ Despite this, the media coverage of prostate screening often incorrectly reports the evidence²⁵⁰ and encourages screening.²⁵¹

Further, there are organisations that support prostate cancer screening. For example, the Prostate Cancer Foundation of Australia recommends annual screening (PSA test and digital rectal exam) in all men commencing at 50 years of age for those with no family history or at 40 years of age with a family history.²⁵² The RCPA and others have recommended use of the PSA test to predict long-term risk in men, first testing around the age of 45 years^{253,254} and using the result to guide the frequency of future PSA monitoring. This approach has been criticised.²⁵⁵ However, the purpose of the RCPA recommendation was to provide guidance on the 'optimal use of PSA testing', accepting that there is existing widespread use of the PSA test.^{254,256}

The current study confirms this, demonstrating that GPs commonly order PSA testing during health checks. Further, use of the test increased significantly between 2000–02 and 2006–08. In fact, total PSA testing (for all problems) increased significantly over the duration of this study (see Chapter 4, Table 4.2). This increase occurred without change in the guidance documents' recommendations. Contributing factors may include: increased patient requests for testing; 'pro-screening' media coverage;²⁵¹ and increased number of men in the population aged 50–75 years.¹⁵ Further, increased incidence of prostate cancer may also play a role. As incidence in the population increases,²⁵⁷ the number of men with a family history of the disease also increases. As family history increases the risk of prostate cancer,^{177,240} it may influence the patient's decision to request PSA testing. However, the increased number of men being tested has contributed to increased incidence of prostate cancer,²⁴⁹ creating the potential for perpetuating the 'screening–diagnosis' cycle.

Thyroid function test

The majority of guidelines reviewed in this study stated that there was insufficient evidence to make a recommendation regarding screening of thyroid function in asymptomatic patients. The RACGP guideline recommended against the testing of thyroid function in asymptomatic adults (regardless of family history) due to low prevalence and lack of evidence of benefit.¹⁷⁷

Other reviews/authors have recommended that opportunistic thyroid testing is not supported in the healthy population.²⁵⁸⁻²⁶² However, as thyroid disease is most prevalent in older women²⁶³ some guidance suggested that screening in menopausal women may be cost-effective.^{258,261,262,264} The American Thyroid Association guidelines for detection of thyroid dysfunction was the only guideline to recommend routine screening (every 5 years, using the thyroid-stimulating hormone test) in all adults from the age of 35 years.²⁶⁵

Despite this lack of support or evidence the BEACH data demonstrated a significant increase in the rate of TFTs. Between 2000–02 and 2006–08, the order rate increased significantly both in the management of health checks and in the management of all problems (see Chapter 4, Table 4.2).

Pathology tests not referenced in guidance documents

Most guidance documents did not mention the use of FBC, MBA, LFT, EUC, ferritin and ESR tests for health checks or screening, which together accounted for 39% of tests. Despite this lack of guidance, GPs' order rate of FBC, MBA, LFT, and EUC increased significantly in the management of health checks.

It is important to note that GPs may use the 'health check' label to refer to preventive checks done in a diverse range of clinical situations. While health checks ordered for a specific disease were excluded in this study, it is possible that some tests were appropriate for the patient's individual clinical circumstance but were not recommended in guidance documents. This is discussed in more detail in Section 5.10 (see 'Prevention'), and may explain why GPs order tests not referenced in guidance documents.

The FBC was the second most commonly ordered test for health checks (15% of tests). A FBC is often requested as a routine screening blood test in general practice.²²¹ However, the guidance documents did not mention FBC testing. Screening for kidney disease was only recommended in the Australian RACGP guideline. Annual screening of glomerular filtration rate estimated from serum creatinine was recommended for high risk patients. The prevalence among high risk patients may have increased, as the prevalence of conditions associated with high risk have increased (e.g. diabetes,^{180,181} obesity¹⁵⁴). This in turn may have contributed to increased rates of EUC testing.

The ICSI guideline provided a consensus recommendation against ordering routine testing in preventive health care, particularly the use of blood chemistry panels, haemoglobulin/haematocrit screening and urinalysis without clinical suspicion of an underlying condition.²³⁸ This suggests that routine ordering of MBA and FBC tests were not supported by the ICSI.

The high proportion of tests ordered by GPs for health checks but not mentioned in guidance documents is concerning. Investigation of the clinical usefulness of these pathology tests for health checks is needed. The results of this investigation should be used to improve guidance documents, and this guidance promoted to GPs to improve test ordering (if necessary).

Changes in order rate of tests that were supported

The reduction in the rate of Pap smears ordered for health checks reflects a change in the problem label recorded by GPs (from 'check-up' to 'well woman check-up', the latter was not included in the analysis for this chapter) rather than a reduction in cervical cancer screening. The pathology ordering data for all problems (Chapter 4) showed that the total rate of Pap smear testing increased significantly over the duration of this study (2000–02 to 2006–08).

In addition to being supported for health checks, lipid and glucose testing were supported in guidelines for the assessment of cardiovascular risk^{176,213,215} (see Section 5.6). GPs might consider assessment of cardiovascular risk as part of the health check. Contributing factors to the increased ordering rate of these tests when doing health checks include the increasing number of patients in the target age group for screening due to Australia's ageing population,¹⁵ and the increase in the proportion of patients with known risk factors for lipid disorders or diabetes (e.g. hypertension,¹⁵⁷ metabolic syndrome,²⁶⁶ chronic kidney disease²⁶⁷) particularly if the diagnosed prevalence of risk factors increases with age.

The increase in the rate of occult blood tests between 2000–02 and 2006–08 may be partially attributed to the introduction of the National Bowel Cancer Screening program in August 2006.²⁶⁸ Tests conducted in the screening program do not require GPs to order FOBTs (as test packs are supplied directly to screening participants). However, the program has been introduced slowly and currently provides a 'one-off' test per person turning either 50, 55 or 65 years of age, far less often than the biannual screening recommendation in the Australian guideline for people aged 50 to 75 years.¹⁷⁷ The increased rate of FOBTs may reflect GPs' biannual testing (as per the guideline recommendation) for patients not targeted in the screening program. Another contributor may be the increased number of people in the target age group for screening due to Australia's ageing population.¹⁵

Other factors contributing to the increased management rate of health checks Check-ups are an important preventive activity and the community is encouraged to have a regular check-up with a GP.²⁶⁹ To a large extent, the increased management rate of health checks was due to the introduction and subsequent uptake of MBS items funding specific age or risk-based health assessments.^{18,232,236}

Australia's ageing population may have contributed to the increased rate of health checks provided by GPs. Over the decade 2000 to 2010, the proportion of the population aged 40 years and above increased.¹⁵ These older adults are the age-group in whom routine disease screening (e.g. colorectal cancer, diabetes, lipid disorders) was recommended (commencing at 40–50 years) and for whom many of the MBS

health assessments were designed.^{18,232,236} It can be assumed that the increased number of older Australians may have contributed to the past growth in the rate of health checks provided by GPs. It may also contribute to future growth in rates of health checks, as the growth in the older population is expected to continue over the coming decades.¹³⁷ A corresponding increase in volume of GPs' pathology ordering (based on the current pattern of pathology test ordering) would be likely.

Summary

Less than half the pathology tests ordered by GPs for health checks were supported by the guidance documents. Further, the increased rate of pathology tests ordered for health checks primarily reflected GPs' increased use of tests that were unsupported in guidance documents. Of particular importance is the need to determine the role of tests commonly ordered by GPs but not referenced in guidance documents. There is significant discord between guidance recommendations and GPs' ordering practice for health checks that should be investigated further.

5.8 Weakness / tiredness

5.8.1 Background

Tiredness or fatigue is a common presentation in general practice. Studies have reported its prevalence as a presenting symptom in 6–14% of general practice patients,²⁷⁰⁻²⁷² and as a secondary symptom in 19% of patients.²⁷⁰

In BEACH, patients presented with weakness/tiredness as a reason for encounter at a rate of 1.4 per 100 encounters in 2006–08,²⁷³ and it was managed as a clinical problem at a rate of 0.7 per 100 encounters (Chapter 4, Table 4.3). The difference between the presentation and management rates indicates that at the encounter GPs were able to apply a more specific diagnostic label to about half of the weakness/tiredness presentations.

Tiredness or fatigue has been described as the most common medically unexplained condition managed in general practice.²⁷⁴ It represents a diagnostic challenge for GPs due to the broad range of conditions for which it may be a symptom, including psychological disorders, sleep problems, chronic disorders and serious disease.^{271,272,275-279} Pathology testing is often ordered by GPs in an attempt to identify the underlying cause (and/or exclude serious disease as the cause) of tiredness.

5.8.2 Management rate in Australian general practice

In BEACH, weakness/tiredness problems were managed at 5,624 patient encounters with 3,279 GPs in 2000–08, at a rate of 0.7 per 100 encounters (Table 5.30). This equates to approximately 710,000 encounters per year where weakness/tiredness problems are managed by GPs in Australia.

As described in Chapter 4, there was no change in the management rate of weakness/tiredness between 2000–02 and 2006–08. There was also no significant change in the rate of new cases per 100 encounters (Table 5.30).

	20	00–08	20	000–02	2006–08		
Variable	Number	Rate per 100 encounters (<i>n</i> = 784,300)	Number	Rate per 100 encounters (<i>n</i> = 198,200)	Number	Rate per 100 encounters (<i>n</i> = 188,300)	
General practitioners	3,279		853		788		
Weakness/tiredness encounters	5,624		1,507		1,372		
Weakness/tiredness problems managed	5,627	0.7 (0.7–0.8)	1,509	0.8 (0.7–0.8)	1,373	0.7 (0.7–0.8)	
New weakness/ tiredness problems	2,456	0.31 (0.30–0.33)	597	0.30 (0.27–0.33)	657	0.35 (0.32–0.38)	

Table 5.30: Summary of weakness/tiredness data set, 2000-08, 2000-02 and 2006-08

Note: Data about the weakness/tiredness problems managed are drawn from Chapter 4, Table 4.3. CI – confidence interval.

5.8.3 Pathology ordering for weakness/tiredness

Pathology was ordered at a rate of 205.4 tests/batteries per 100 weakness/tiredness contacts in 2000–08, and more than half of contacts (56.6%) resulted in at least one pathology order. Once the decision to order pathology was made, the GP ordered on average 3.63 tests/batteries per tested contact (Table 5.31).

As described in Chapter 4, the rate of pathology ordering increased significantly between 2000–02 and 2006–08. This was due to a significant increase in the likelihood of pathology being ordered in the management of weakness/tiredness problems. There was no statistically significant change in the number of tests ordered once the decision to order tests was made (Table 5.31).

	20	000–08	2	000–02	2		
Variable	Number	Per cent / Rate of weak/tired problems (95%Cl) (n=5,627)	Number	Per cent / Rate of weak/tired problems (95%Cl) (n=1,509)	Number	Per cent / Rate of weak/tired problems (95%Cl) (n=1,373)	Change
Pathology (Rate per 100 weak/tired problems)	11,559	205.4 (197.2–213.6)	2,684	177.9 (164.0–191.8)	3,199	233.0 (217.3–248.7)	↑
At least one pathology order (% of weak/tired problems)	3,187	56.6 (54.7–58.6)	759	50.3 (46.7–53.9)	854	62.2 (58.5–65.9)	↑
Number of tests/ batteries per 100 teste weak/tired problems	d	362.7 (357.0–368.3)		353.6 (343.2–364.0)		374.6 (363.5–385.7)	_

Table 5.31: Summary of pathology ordering for weakness/tiredness, 2000–08, 2000–02and 2006–08

Note: Pathology ordering data from 2000–02 and 2006–08 are drawn from Chapter 4, Table 4.4. Weak/tired – weakness/ tiredness; CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change, and — indicates no change.

Types of pathology tests/batteries ordered

Table 5.32 shows the distribution of pathology tests/batteries ordered for

weakness/tiredness in 2000–08 by MBS groups and the most common individual

types of tests ordered.

Chemistry tests were the group most often ordered (135.7 per 100 weakness/

tiredness contacts) and the most common were:

- TFTs (32.3 per 100 contacts)
- ferritin tests (21.0)
- LFTs (17.4)
- EUC tests (15.5)
- MBA tests (11.6)
- glucose/glucose tolerance tests (11.5) (Table 5.32).

Haematology tests (53.9 per 100 contacts), in particular FBCs (45.7), and microbiology tests (9.7) were also commonly ordered in the management of weakness/tiredness (Table 5.32).

Almost 60% of pathology tests were ordered by GPs for 'new' cases of weakness/ tiredness and the order rate for 'new' problems was significantly higher than average (Table 5.32). The likelihood of pathology being ordered at new weakness/tiredness contacts was high, with at least one pathology test ordered for 74.3% (95% CI: 72.4–76.1) of new cases (results not tabled).

	All weakness/tiredness			eakness/t	kness/tiredness problems		
Pathology test	Number	Rate per 100 weak/tired problems (95%Cl) (<i>n</i> = 5,624)	Number	Per cent of test	Rate per 100 new weak/tired problems (95%Cl) (<i>n</i> = 2,456)		
Chemistry	7,636	135.7 (130.0–141.5)	4,558	59.7	185.6 (178.9–192.3)		
TFT*	1,819	32.3 (30.5–34.1)	1,113	61.2	45.3 (43.1–47.5)		
Ferritin*	1,183	21.0 (19.6–22.5)	727	61.5	29.6 (27.6–31.6)		
LFT*	979	17.4 (16.1–18.7)	589	60.2	24.0 (22.1–25.8)		
EUC*	872	15.5 (14.3–16.7)	510	58.5	20.8 (18.9–22.6)		
MBA*	650	11.6 (10.5–12.6)	405	62.3	16.5 (14.8–18.2)		
Glucose/glucose tolerance*	647	11.5 (10.5–12.5)	391	60.4	15.9 (14.3–17.6)		
Lipids*	329	5.9 (5.1–6.6)	184	55.9	7.5 (6.4–8.6)		
Vitamin B12*	311	5.5 (4.8–6.2)	171	55.0	7.0 (5.9–8.1)		
Other chemistry*	213	3.8 (3.0–4.6)	99	46.5	4.0 (3.0–5.0)		
Hormone assay*	197	3.5 (2.7–4.3)	117	59.4	4.8 (3.4–6.1)		
CRP	147	2.6 (2.1–3.1)	91	61.9	3.7 (2.9–4.5)		
Haematology	3,034	53.9 (51.5–56.3)	1,832	60.4	74.6 (71.9–77.3)		
FBC	2,572	45.7 (43.7–47.7)	1,561	60.7	63.6 (61.5–65.7)		
ESR	430	7.6 (6.9–8.4)	250	58.1	10.2 (8.9–11.4)		
Microbiology	543	9.7 (8.6–10.7)	311	57.3	12.7 (11–14.3)		
Monospot*	153	2.7 (2.2–3.2)	95	62.1	3.9 (3.1–4.7)		
Urine M,C&S*	127	2.3 (1.8–2.7)	79	62.2	3.2 (2.5–3.9)		
Other tests NEC	181	3.2 (2.6–3.8)	100	55.2	4.1 (3.1–5.1)		
Immunology	147	2.6 (2.1–3.1)	74	50.3	3.0 (2.2–3.8)		
Other pathology groups	18		10	55.6			
Total pathology tests	11,559	205.4 (197.2–213.6)	6,885	59.6	280.3 (271.6–289.0)		

 Table 5.32: Rate of pathology test orders for weakness/tiredness by MBS groups and most frequent individual tests within each group, 2000–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations.

Changes in types of pathology tests/batteries ordered

The most common pathology tests/batteries ordered for weakness/tiredness in 2000–02 and 2006–08 are shown in Table 5.33. Listed below are the tests for which significant change in GPs' order rate occurred between 2000–02 and 2006–08 (listed in decreasing test rate order). There were significant increases in the order rate of:

- FBCs—20% increase
- TFTs—35% increase
- ferritin tests—64% increase
- LFTs—33% increase
- vitamin B12 tests—105% increase
- CRP tests—281% increase (there was trend for a corresponding decrease in the order rate of ESR tests but this trend did not reach statistical significance)
- 'other immunology' tests—600% increase, due to an increased number of immunoglobulin and anti-endomysial antibody tests (Table 5.33).

		2000–02		2006–08	
– Pathology test	Number	Rate per 100 weak/tired problems (95% CI) (n=1,509)	Number	Rate per 100 weak/tired problems (95% CI) (<i>n</i> = 1,373)	Change
Chemistry	1,696	112.4 (102.9–121.9)	2,163	157.5 (146.1–168.9)	↑
TFT*	406	26.9 (24.1–29.7)	498	36.3 (32.9–39.7)	↑
Ferritin*	245	16.2 (13.9–18.5)	365	26.6 (23.6–29.5)	♠
LFT*	232	15.4 (13.2–17.6)	282	20.5 (17.9–23.2)	↑
EUC*	199	13.2 (10.9–15.5)	221	16.1 (13.8–18.4)	
MBA*	154	10.2 (8.3–12.1)	167	12.2 (10.0–14.3)	_
Glucose/glucose tolerance	e* 174	11.5 (9.6–13.4)	154	11.2 (9.1–13.3)	_
Vitamin B12*	59	3.9 (2.8–5.0)	110	8.0 (6.3–9.7)	↑
Lipids*	65	4.3 (3.2–5.4)	91	6.6 (5.1–8.2)	_
Other chemistry*	47	3.1 (2.0–4.2)	76	5.5 (3.7–7.4)	_
CRP	17	1.1 (0.6–1.7)	58	4.2 (2.9–5.5)	↑
Hormone assay*	45	3.0 (1.8–4.2)	46	3.4 (2.0–4.7)	_
Haematology	778	51.6 (47.3–55.9)	782	57.0 (52.5–61.4)	_
FBC	629	41.7 (38.3–45.1)	689	50.2 (46.4–53.9)	♠
ESR	138	9.2 (7.5–10.8)	85	6.2 (4.8–7.6)	_
Microbiology	135	9.0 (6.9–11.0)	131	9.5 (7.5–11.6)	—
Monospot*	37	2.5 (1.6–3.4)	44	3.2 (2.2–4.2)	—
Urine M,C&S*	33	2.2 (1.4–3.0)	30	2.2 (1.4–3.0)	—
Immunology	18	1.2 (0.6–1.8)	69	5.0 (3.5–6.6)	↑
Other immunology*	6	0.4 (0.1–0.7)	38	2.8 (1.6–3.9)	♠
Other tests NEC	56	3.7 (2.4–5.0)	47	3.4 (2.1–4.7)	_
Blood test	29	1.9 (1.0–2.8)	29	2.1 (1.1–3.2)	—
Other pathology groups	1		7		
Total pathology tests	2,684	177.9 (164.0–191.8)	3,199	233.0 (217.3–248.7)	1

Table 5.33: Rate of pathology test orders for weakness/tiredness by MBS groups and most frequent individual tests within each group, 2000–02 compared with 2006–08

Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Weak/tired – weakness/tiredness; CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations. Shading indicates a statistically significant change, the direction and type of change is indicated for each measure between 2000–02 and 2006–08: ↑/↓ indicates a statistically significant change, and — indicates no change.

5.8.4 Guidance for weakness/ tiredness

Guidance documents for the management of tiredness and fatigue were reviewed in this study, but guidance for chronic fatigue syndrome was excluded.

The Dutch College of General Practitioners (DCGP) guideline for blood testing in 'medically unexplained' complaints was also reviewed because fatigue is the most common unexplained complaint managed in general practice.²⁷⁴ Medically unexplained complaints were defined as 'those complaints for which a GP, after clarifying the reason for the encounter, taking the patient's history, and performing a physical examination, is unable to establish a diagnosis.²⁸⁰ There were no guidance documents that specifically addressed management of 'weakness', but it was covered in the DCGP guideline.

In total eight guidance documents were reviewed (Box 5.5). However there was some duplication as two authors produced two guidance documents each. These documents (written by the same author) were independent of each other and were therefore reviewed.

Title	Year	Author	Abbreviated to
Investigating fatigue of less than 6 months' duration: Guidelines for family physicians ²⁷⁶	1999	Godwin M, Delva D, Miller K, Molson J, Hobbs N, MacDonald S et al. [Canada]	Godwin et al.
Dutch College of General Practitioners guideline for blood testing in medically unexplained complaints ²⁸¹	1994 (2009)	Dutch College of General Practitioners (DCGP), cited in Koch et al. because the original guideline is published in Dutch [The Netherlands]	DCGP
Fatigue – a general diagnostic approach ²⁸²	2003	Murtagh J [Australia]	Murtagh 2003
Murtagh's general practice, tiredness chapter ¹⁷³	2007	Murtagh J [Australia]	Murtagh
Patient presentations in general practice, 'I feel tired' ¹⁷⁴	1999	Steven I [Australia]	Steven
Laboratory investigation of tiredness ²⁸³	2006	bpac (Best practice advocacy centre) [New Zealand]	bpac
ABC of psychological medicine: Fatigue ²⁸⁴	2002	Sharpe M and Wilks D [United Kingdom]	Sharpe & Wilks
RCPA manual, chronic fatigue section ¹⁷²	2004	The Royal College of Pathologists of Australasia (RCPA)	RCPA
There must be something wrong'— primary presentation in the scenario is fatigue of a chronic duration > 6 months ²⁸⁵ (RCPA case scenario)	2003	The Royal College of Pathologists of Australasia (RCPA) and The University of Sydney (USyd) Department of Medical Education	RCPA & USyd

Box 5.5: Guidelines and guidance documents reviewed for weakness/tiredness

5.8.5 Extent of alignment between GP testing and guidance documents

In Table 5.34, the frequently ordered pathology tests/batteries for weakness/tiredness are categorised by level of support in the guidance documents listed in Box 5.5.

Overview

Some guidance documents consistently referred to the same clinical situation or conditions when recommending pathology testing. These conditions are summarised here and referred to in the next sections.

Godwin et al.,²⁷⁶ Steven¹⁷⁴ and DCGP²⁸¹ only provided conditional support for the use of any pathology tests in the management or investigation of tiredness. Each document consistently recommended testing in the following circumstances:

- if indicated by the physical examination (Godwin et al.)²⁷⁶
- 'if felt to be required' by the GP (Steven)¹⁷⁴
- tests should be postponed until 4 weeks after the first presentation of the unexplained complaint (of which fatigue is the most common) (DCGP).²⁸¹

In the bpac guidance, test recommendations were based on patient age, sex, presence of risk factors and/or duration of tiredness. After relevant clinical review, specific tests were recommended:

- for all patients (after relevant clinical review)
- if the duration of tiredness was longer than 1 month or the patient was aged more than 50 years, or
- for those patients aged less than 50 years who had additional risk factors for a specific disease.²⁸³

In the following text when Godwin et al.,²⁷⁶ Steven¹⁷⁴ DCGP²⁸¹ and bpac²⁸³ are mentioned as giving conditional support, the conditions refer to those listed above.

Supported tests

The tests that were recommended in the majority of guidance documents were FBC, TFT, ferritin, LFT, EUC, glucose/glucose tolerance and ESR (Table 5.34). Together these tests accounted for 73.6% of tests/batteries ordered by GPs for weakness/tiredness (Table 5.35). The order rate of FBC, TFT, ferritin, and LFT increased significantly between 2000–02 and 2006–08 (Table 5.33).

FBC testing was recommended in all guidance documents. Godwin et al.,²⁷⁶ DCGP²⁸¹ and Steven¹⁷⁴ provided conditional support (as described above), but the DCGP²⁸¹ only recommended the haemoglobin test, a component of the FBC battery in Australia, rather than the full battery.

TFT was also supported in all guidance documents. Though four documents provided conditional support—as described above for Godwin et al.,²⁷⁶ DCGP,²⁸¹ Steven,¹⁷⁴ and in bpac for patients at risk of thyroid problems.²⁸³

Ferritin testing was recommended to identify iron deficiency anaemia in five of the guidance documents and not mentioned in the other four documents—of these four, two discussed anaemia as a common cause of tiredness/fatigue^{276,284} but did not recommend ferritin testing.

LFT was supported in the majority of guidance documents. Conditional support was provided by Steven¹⁷⁴ and in bpac for patients at risk of liver dysfunction.²⁸³

EUC was recommended in seven guidance documents. Three of these provided conditional support: Godwin et al.,²⁷⁶ Steven,¹⁷⁴ and in bpac for patients at risk of renal problems.²⁸³

Glucose testing was recommended in most guidance documents. Conditional support was provided in Godwin et al.²⁷⁶ and DCGP,²⁸¹ and in bpac for patients at risk of diabetes.²⁸³

ESR testing was recommended in the majority of guidance documents. Godwin et al.,²⁷⁶ DCGP²⁸¹ and Steven¹⁷⁴ gave conditional support.

Tests with conditional support

There were two tests that had conditional support or mixed levels of support in the guidance documents: Monospot and CRP (Table 5.34). Together these tests accounted for 2.6% of tests ordered for weakness/tiredness (Table 5.35). Between 2000–02 and 2006–08 the rate of CRP increased significantly but the order rate for Monospot tests did not change (Table 5.33)

CRP was recommended in four guidance documents,^{172,173,283,284} two of which recommended use of either CRP or ESR.^{173,284} The bpac guidance recommended CRP testing if the patient was over 50 years of age or tiredness had lasted longer than a month.²⁸³ Epstein Barr Virus (Monospot testing) was mentioned in most guidance documents as a possible cause of tiredness. The limitations of the test in diagnosing the condition were also discussed. RCPA and USyd recommended testing only if there was recent infection because approximately 90% of patients will have positive results due to previous infection,²⁸⁵ and bpac specifically recommended against testing.²⁸³

Support unable to be determined

It was not possible to determine whether MBA and 'other chemistry' tests were recommended in the guidance documents (see Section 5.3.4). Together these two tests accounted for 7.4% of pathology ordered for weakness/tiredness (Table 5.35). Their order rate did not change between 2000–02 and 2006–08 (Table 5.33).

The MBA test itself was not recommended in any guidance documents. However, some tests that may be considered part of the MBA, such as EUC and LFT, were partially or completely supported in the management of weakness/tiredness (as discussed above).

Unsupported tests

Most guidance documents did not recommend testing of lipids, vitamin B12, hormone assay and Urine M,C&S (Table 5.34). These four tests accounted for 8.3% of pathology ordered for weakness/tiredness (Table 5.35). Vitamin B12 was the only test within this group to increase significantly between 2000–02 and 2006–08 (Table 5.33).

Lipid tests were not mentioned by any of the guidance documents in the management of weakness/tiredness.

Vitamin B12 testing was only recommended by Steven¹⁷⁴ if anaemia was suspected. Other documents discussed anaemia as a cause of tiredness/fatigue, but did not mention vitamin B12 testing even though vitamin B12 deficiency is one possible cause of anaemia.

The 'hormone assay' test group in BEACH represented orders for sex hormones in > 90% of cases. Testing sex hormones was not mentioned in any of the guidance documents, but testing the cortisol hormone was mentioned in four documents if Cushing's syndrome was suspected (a rare disease that may cause tiredness). However, because cortisol was rarely ordered and 'hormone assay' largely reflected sex hormone testing, this test group was classified as unsupported.

Urine M,C & S test was recommended in the two Murtagh documents; however, it was not mentioned in any of the other guidance documents.

Other tests mentioned in the guidance documents

Other tests were mentioned in the guidance documents but each of these accounted for less than 1% of tests recorded by GPs for weakness/tiredness.

Urinalysis was recommended in six documents,^{174,276,282-285} three^{174,276,283} of which gave conditional support.

Calcium testing was commonly recommended^{173,174,282,283} often with magnesium^{173,282} or phosphate testing.²⁸³

Other recommendations included testing for autoimmune diseases (e.g. antinuclear antibody test),^{173,283,285} testing for chronic infections (e.g. HIV, hepatitis),^{173,174,283} muscle enzymes (e.g. CK)^{173,284,285} and cancer antigens.¹⁷³

Key to Table 5.34

Colour	Description
	The document specifically recommended this test. Any notes within the cell indicate further detail. For example, a specific disease to test for within subset of patients; a specific test within a group.
	The document stated that this test should be considered. Any notes within the cell indicate further detail (e.g. a specific test to consider)
	Unable to determine guidance (see Section 5.3.4).
	The document specifically stated not to do this test. Additional information is supplied if certain conditions apply (e.g. specific clinical situations).
	Guidance document does not mention this test

	Per cent of	Fatigue/tiredness						Chronic fatigue (> 6 months)		
Pathology test	pathology (<i>n</i> = 11,559)	Godwin et al. ²⁷⁶	DCGP ²⁸¹	Murtagh 2003 ²⁸²	Murtagh ¹⁷³	Steven ¹⁷⁴	bpac ²⁸³	Sharpe & Wilks ²⁸⁴	RCPA ¹⁷²	RCPA & USyd ²⁸⁵
FBC	22.3		Wait 4 weeks Hb only							
TFT*	15.7		Wait 4 weeks				At risk/ > 50 yrs/ 1+month			
Ferritin*	10.2					Anaemia				
LFT*	8.5						At risk/ > 50 yrs/ 1+month			
EUC*	7.5						At risk/ > 50 yrs/ 1+month			
MBA*	5.6									
Glucose/glucose tolerance*	5.6		Wait 4 weeks				At risk/ > 50 yrs/ 1+month			
ESR	3.7		Wait 4 weeks		or CRP			or CRP		
Lipids*	2.9									
Vitamin B12*	2.7					Anaemia				
Other chemistry*	1.8									
Hormone assay*	1.7			Cortisol	Cortisol	Cortisol				Cortisol
Monospot*	1.3						Not in diagnosis			Recent infection
CRP	1.3				or ESR		> 50 yrs/ 1+month	or ESR		
Urine M,C&S*	1.1									

Table 5.34: Summary of guideline/guidance recommendations by most frequent individual test orders for weakness/ tiredness, 2000–08

Includes multiple ICPC-2 PLUS terms (see Appendix 7)

*

Note: Weak/tired – weakness/tiredness; Hb – haemoglobulin; > 50 yrs – aged more than 50 years; 1+ month – duration of one month or more; also see Abbreviations. Any notes within the coloured cells are described in detail in Section 5.8.5. Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included.

Evaluation of GP pathology ordering against guidelines/guidance

Table 5.35 provides a summary of the individual tests and the level of support provided in the guidance for each. Of the tests/batteries ordered by GPs for weakness/tiredness in 2000–08:

- 73.6% were supported
- 10.1% had conditional support or support could not be determined
- 8.3% were not supported by the guidance documents (Table 5.35).

The individual tests/batteries listed in Table 5.35 account for 92.0% of pathology tests/batteries ordered for weakness/tiredness because only the most commonly ordered were evaluated.

Pathology test	Number	Per cent of all pathology for weakness/tiredness
Supported	8,502	73.6
FBC	2,572	22.3
TFT*	1,819	15.7
Ferritin*	1,183	10.2
LFT*	979	8.5
EUC*	872	7.5
Glucose/glucose tolerance*	647	5.6
ESR	430	3.7
Conditional/unclear support	1,163	10.1
MBA*	650	5.6
Other chemistry*	213	1.8
Monospot*	153	1.3
CRP	147	1.3
Unsupported	964	8.3
Lipids*	329	2.9
Vitamin B12*	311	2.7
Hormone assay*	197	1.7
Urine M,C&S*	127	1.1
Subtotal (n, % of total tests)	10,629	92.0
Total pathology tests	11,559	100.0

Table 5.35: Summary of support for GP pathology or	dering for the most frequent
individual test orders for weakness/tiredness, 2000-08	3

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. See Abbreviations.

In 2000–08, GPs ordered supported pathology tests at a rate of 151.0 tests/batteries per 100 weakness/tiredness problems, followed by tests with conditional support (20.5 per 100), and unsupported tests (17.0). Tests ordered for weakness/tiredness but not evaluated were ordered at a rate of 16.9 per 100 problems. The rate at which GPs ordered tests in all 'level of support' groups increased significantly over time (Table 5.36).

	2000–08	2000–02	2006–08	
Level of support	Rate per 100 weak/tired problems (95%Cl) (<i>n</i> = 5,627)	Rate per 100 weak/tired problems (95% CI) (<i>n</i> = 1,509)	Rate per 100 weak/tired problems (95% Cl) (n=1,373)	Change
Supported	151.0 (144.5–157.5)	134.1 (122.7–145.6)	166.8 (153.9–179.7)	↑
Conditional/unclear support	20.5 (19.2–21.9)	16.9 (14.3–19.4)	24.9 (21.8–28.0)	↑
Unsupported	17.0 (15.6–18.5)	13.3 (11.1–15.6)	20.0 (17.0–23.0)	↑
Not evaluated	16.9 (15.3–18.4)	13.5 (11.2–15.7)	21.4 (17.9–24.8)	↑

Table 5.36: Rate of pathology ordering for weakness/tiredness by level of support, 2000–08, 2000–02 and 2006–08

Note: CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change.

GPs ordered only pathology tests that were supported or conditionally supported at 38.5% (95% CI: 37.0–40.1) of weakness/tiredness problems in 2000–08. A further 15.4% (95% CI: 14.1–16.6) of problems involved at least one unsupported test, and 46.1% (95% CI: 44.2–48.0) involved either no pathology tests (43.3%) or tests that were not evaluated (2.8%) (results not tabled).

There was no change in the proportion of weakness/tiredness problems that involved supported testing, 36.2% (95% CI: 33.1–39.4) in 2000–02 and 40.5% (95% CI: 37.4–43.7) in 2006–08. However the proportion of problems involving at least one unsupported test increased significantly over this time, from 11.8% (95% CI: 9.9–13.6) to 18.3% (95% CI: 15.6–21.0) (results not tabled).

When GPs ordered unsupported tests, the vast majority (95%) were accompanied by one or more supported/partially supported tests, unsupported tests being ordered alone at the remaining 5%. GPs ordered only tests that were unsupported for 0.8% of weakness/tiredness problems in 2000–08. This did not change significantly over time 0.7% (95% CI: 0.3–1.1) in 2000–02 and 1.2% (95% CI: 0.2–2.1) in 2006–08 (results not tabled).

5.8.6 Discussion

GPs pathology testing aligned well with that recommended in guidance documents, with approximately 85% of tests ordered by GPs in the management of weakness/tiredness being recommended (with either partial or complete support). The statistically significant increase in the rate of GPs total pathology ordering for weakness/tiredness between 2000–02 and 2006–08 was reflected in significant increases in the rate of tests that were supported, partially supported, and unsupported in guidance documents.

As a proportion of total tests ordered there was very little change in the proportion that were supported (either completely or partially) or unsupported for weakness/tiredness, indicating that the increase in GPs' testing did not reflect a change to be 'more' or 'less' in line with guidance recommendations. This study includes weakness/tiredness when recorded as a problem under management by GPs. It does not reflect all patient presentations of weakness/tiredness, as GPs at encounters were able to apply a more specific diagnostic label to about half of these presentations.²⁷³ In BEACH, GPs were instructed to record the problem at the highest diagnostic level possible with the information available at the encounter. Therefore the management of weakness/tiredness represents either: a diagnostic process (i.e. a symptom under investigation), or an undifferentiated problem (i.e. following investigation a 'diagnosis' is unable to be determined).

Pathology tests would be considered part of the diagnostic process. Guidance documents discussed them in the context of the entire diagnostic process, that is, in combination with an appropriate medical history and physical examination. This was also emphasized in the literature, Cornuz et al. stated that 'in the absence of positive history or physical examination laboratory tests are rarely helpful'.²⁸⁶ This approach was recommended to focus testing because of the broad range of conditions linked to tiredness.

In this study, only guidance documents for fatigue and tiredness were reviewed for weakness/tiredness because no guidance documents were found for weakness. This is unlikely to impact significantly on the results of the study because the majority of problems classified to the ICPC-2 'weakness/tiredness' group were recorded by GPs as tiredness or related synonyms such as fatigue or malaise.

In the guidance documents reviewed, pathology recommendations were consensusbased rather than evidence-based because there is very little evidence available. Recommendations were based on the conditions in general practice that commonly involve fatigue/tiredness as a symptom, and on the likelihood of underlying disease.

Underlying disease

Prevalence of underlying disease in general practice patients with fatigue is low. Three prospective studies investigating outcomes of fatigue in general practice patients reported diagnosis of a somatic disease as the (known or likely) cause of fatigue in 8% of patients.^{275,278,281} Two of these studies specifically linked diagnosis of somatic diseases to pathology tests.^{275,281} An Australian retrospective study examined medical records where tiredness was recorded as a symptom or diagnosis, and identified a 'significant diagnosis' as the likely cause of tiredness in 9% of patients, but only 4% of patients were diagnosed as a result of pathology testing.²⁸⁷ While three of the studies linked the diagnosis of somatic diseases to pathology tests,^{275,281,287} only two recommended the most useful tests in making these diagnoses.^{281,287} Koch et al.²⁸¹ compared the tests recommended in the DCGP's guideline with a separate panel of tests developed by expert consensus, and concluded that the four blood tests in the DCGP's guideline (FBC, TFT, glucose and ESR) were most useful. In contrast Gialamas et al. in their retrospective study concluded that the most useful tests were FBC, glucose and EUC.²⁸⁷ The five tests (from these two studies) were evaluated as supported in this study, but only represent a subset of the supported tests (74% of the supported tests ordered by GPs, 55% of total tests ordered). The results of these studies suggest authors of future guidance documents should consider a more focussed approach to pathology testing than is currently recommended.

Abnormal results and false positive results

The low prior probability of underlying disease in patients with weakness/tiredness and high rates of pathology testing ordered by GPs are likely to generate high rates of abnormal test results, many of which will be false positive test results. This is demonstrated in results of the three studies that investigated the role of pathology tests in diagnosing somatic diseases in general practice.^{275,281,287}

In the retrospective study, 14% of 1,046 tests had abnormal results, but 90% of these were not clinically significant.²⁸⁷ In the British prospective study, 33% of patients

had an abnormal test result, but for 72% of these patients the abnormal result was not clinically relevant.²⁷⁵ In the Dutch prospective study, at least one false positive result was recorded for 23% of 192 patients with medically unexplained fatigue with the limited DCGP set of tests and for 56% of patients with the expanded test panel.²⁸¹ The issue of abnormal results and false positive results is discussed in greater depth in Section 5.10.

Delaying testing

The desire to reduce false positive test results, the low probability of underlying disease and the self-limiting nature of most medically unexplained complaints such as fatigue led the DCGP to recommend a limited set of tests and to delay testing until 4 weeks after first presentation of the complaint. The study by Koch et al. designed to test the DCGP recommendation, could not assess the impact of delaying pathology testing in patients with medically unexplained fatigue, because most patients randomised to wait 4 weeks did not return for a follow-up consultation – 78% (of 111 patients) did not return. An additional 27 patients were excluded because GPs were unable to delay testing. Whether this was due to patient or GP factors was not investigated.²⁸¹ Even though there was insufficient follow-up data, Koch et al. concluded there was value in delaying pathology testing, on the basis of similar rates of somatic diseases in the delayed versus immediate testing groups.²⁸¹

It is possible that the low level of follow-up consultations in Koch et al.'s study could indicate that these patients experienced a self-limiting episode of fatigue that resolved within 4 weeks, making a follow-up visit and testing unnecessary. The likelihood of patients requiring multiple visits for management of weakness/tiredness may be relatively low regardless of the pathology testing strategy used. The retrospective study of medical records conducted by Gialamas et al. reported that the majority of patients (55%) required only one GP visit to manage tiredness.²⁸⁷ The current study also supports this: the high proportion of new weakness/tiredness contacts (44%) suggests that the rate of follow-up contacts was relatively low (when compared with the chronic diseases investigated in this chapter). However, because this study used cross-sectional data, I was not able to assess the number of GP consultations patients required to manage weakness/tiredness.

Even though each episode of tiredness may not require many GP visits, pathology tests are frequently initiated at the first visit. In this study, three-quarters of new

weakness/tiredness problems had at least one pathology test ordered. Similarly, in a small study of medically unexplained complaints, 83% of 59 patients with fatigue had tests initiated at the first GP visit,²⁸⁸ and in the study by Gialamas et al. likelihood of ordering at the first contact was not reported but 67% of test orders were initiated at the first contact.²⁸⁷ This suggests that application of a recommendation to 'delay testing' would require a significant change in GPs' current pathology ordering behaviour for weakness/tiredness.

The DCGP was the only guideline to recommend delaying testing at the first GP visit. However, duration of tiredness was mentioned as a component of medical history to be considered prior to ordering pathology testing in many guidance documents.^{172,173,276,281,283-285} It may be that GPs incorporate patients' duration of tiredness in their decision to order pathology testing, and patients' duration of symptoms at their first GP visit was sufficient to warrant pathology testing. Given that over the period of this study the likelihood of ordering pathology increased and likelihood of ordering was high for new cases of weakness/tiredness, delaying pathology testing as per the DCGP's recommendation may achieve a reduction in the likelihood of testing. Delaying testing has been found not to have an impact on patient satisfaction and anxiety levels when compared with immediate testing for medically unexplained complaints.²⁸⁹ However, further research needs to be conducted to determine the clinical outcomes of patients in whom testing is delayed. Such research would also need to compare cost effectiveness of delaying testing versus immediate testing, for example, the cost of pathology tests, GP visits required, and the time taken to explain the reasons for delaying testing.

Summary

The majority of tests ordered by GPs for weakness/tiredness were recommended in guidance documents. The measured increase in pathology testing was due to an increased likelihood that GPs would initiate testing in the management of weakness/tiredness. There was no contributing increase in the management rate of weakness/tiredness and no clear cause for the increase in likelihood of testing. But importantly the extent to which GPs' testing aligned with recommendations did not change.

Results from some small studies have suggested that limiting the number of pathology tests ordered, at least initially, for weakness/tiredness may have a role in

reducing clinically insignificant abnormal test results. It has also been recommended that GPs should delay testing in patients with tiredness. Both of these suggestions may reduce pathology testing for weakness/tiredness, but the level of evidence is weak. Further research is needed to increase the evidence base and to ensure recommendations to delay testing would not adversely affect patient outcomes, and that any associated change in clinical practise is cost-effective.

The evidence-base available to support recommendations for pathology ordering in the management of weakness/tiredness is weak. Consequently many of the recommendations in guidance documents were consensus-based. Increasing the research in this area could improve the robustness of future guidance documents for weakness/tiredness.

5.9 Overweight/obesity

5.9.1 Background

In most developed countries, including Australia, the high and increasing prevalence of overweight and obesity has a significant economic and health impact.¹⁰⁹ Recognising the burden of chronic disease caused by obesity, it was named a National Health Priority Area in April 2008²⁹⁰ and is targeted in the current National Preventative Health Strategy.²³⁵

The 2007–08 National Health Survey, estimated that 61.3% of Australian adults were overweight/obese (measured BMI \ge 25). Prevalence was higher among males (68%) than females (55%), and had increased from 56.3% in 1995.¹⁵⁴ In a SAND substudy of adult patients (aged 18 years and over) attending general practice, prevalence of overweight/obesity (calculated BMI from self-reported height and weight) increased from 54.3% in 2000–01 to 59.3% in 2007–08.²⁷³

In 2003, high body mass was responsible for 7.5% of total burden of disease and injury in Australia.¹⁵⁶ In 2005, the direct cost (health care and non-health care costs) of overweight and obesity (in adults aged 30 years and over) was \$21 billion, \$6.5 billion for overweight and \$14.5 billion for obesity.²⁹¹ Three-quarters of these direct costs were health care costs.

Obesity and overweight are associated with many diseases including T2D, cancer and cardiovascular diseases.²⁹² Research suggests that about two-thirds of overweight/obese patients seen in general practice have at least one obesity-related cardiovascular comorbidity,²⁹³ and likelihood of comorbidities increases with increasing BMI.^{293,294}

Pathology testing has a role in the diagnosis, assessment and management of many of these associated diseases and is one small component of the health cost burden attributed to overweight/obesity. GPs' management of overweight/obesity generated the smallest amount of pathology of the six problems investigated in this study (see Chapter 4). It was prioritised for investigation because it is the focus of much current public health policy. There have been several national and state-based programs that aim to reduce the prevalence of overweight and obesity. Recent national examples include the 'Measure Up' campaign²⁹⁵ and the 'Swap it' campaign.²⁹⁶ With

increasing public awareness it is likely that the management rate of overweight and obesity in general practice will increase. If this occurs it would create a corresponding increase in pathology orders.

5.9.2 Management rate in Australian general practice

In BEACH, obesity or overweight was managed at 7,797 encounters with adult patients (aged 18 years and over) by 3,677 GPs in 2000–08, at a rate of 1.2 per 100 adult encounters (Table 5.37). This equates to approximately 1.0 million adult encounters nationally per year where overweight/obesity was managed by GPs. Obesity accounted for 71.8% of overweight/obesity problems managed and overweight for the remaining 28.2% (Table 5.37).

As discussed in Chapter 4, there was no significant change in the management rate of overweight/obesity between 2000–02 and 2006–08. There was also no change in the diagnosis or detection rate of new cases of overweight/obesity (Table 5.37).

		2000–08		2000–02	2006–08	
Variable	Number	Rate per 100 adult encounters (95% CI) (<i>n</i> = 666,135)	Number	Rate per 100 adult encounters (95% CI) (<i>n</i> = 166,770)	Number	Rate per 100 adult encounters (95% Cl) (<i>n</i> = 161,571)
General practitioners	3,677		954		864	
Overweight/obesity encounters	7,797		1,975		1,935	
Overweight/obesity problems managed	7,797	1.2 (1.1–1.2)	1,975	1.2 (1.1–1.3)	1,935	1.2 (1.1–1.3)
Obesity	5,598	0.8 (0.8–0.9)	1,458	0.9 (0.8–1.0)	1,418	0.9 (0.8–1.0)
Overweight	2,199	0.3 (0.3–0.4)	517	0.3 (0.3–0.4)	517	0.3 (0.3–0.4)
New overweight/ obesity problems	1,141	0.17 (0.16–0.18)	292	0.18 (0.15–0.20)	242	0.15 (0.12–0.18)

Table 5.37: Summary of overweight/obesity data set (adult patients), 2000–08, 2000–02and 2006–08

Note: Data about overweight/obesity problems managed are drawn from Chapter 4, Table 4.3. Cl - confidence interval.

Figure 5.1 compares the age-specific rates of management of overweight/obesity among adult patients and the age-specific prevalence of overweight/obesity among patients encountered in general practice. The latter uses calculated BMI from self-reported height and weight drawn from a subsample of 31,062 patients at BEACH encounters in 2007–08 (see Section 3.2.6).

Figure 5.1 should be interpreted as follows—for patients aged 45–64 years, 1.6% had obesity or overweight managed as a clinical problem at the encounter, whereas 68.1% of patients attending in this age group were obese or overweight.

There is a huge gap between the proportion of patients who are overweight/obese and the management rate of overweight/obesity. The combination of prevalence and encounter data suggests there were 51 overweight/obese adult patients seen by GPs per 1 management encounter for overweight/obesity. This illustrates that there is huge scope for an increase in the management rate of overweight/obesity.



5.9.3 Pathology ordering for overweight/obesity

Pathology was ordered at a rate of 37.4 per 100 overweight/obesity problems in 2000–08 for adult patients. More than one in ten overweight/obesity problem contacts (13.6%) resulted in at least one pathology order. Once the decision to order was made the GP ordered on average 2.75 tests per tested problem (Table 5.38). As reported in Chapter 4, the rate of pathology ordering for overweight/obesity increased significantly between 2000–02 and 2006–08, due to an increase in the

likelihood of at least one test/battery being ordered. There was no significant change in the number of tests ordered per tested overweight/obesity problem (Table 5.38).

	2000–08		2000–02		2006–08		
	Number	Per cent / Rate of ov/ob problems (95% Cl) (n=7,797)	Number	Per cent / Rate of ov/ob problems (95% Cl) (n = 1,975)	Number	Per cent / Rate of ov/ob problems (95% Cl) (<i>n</i> = 1,935)	Change
Pathology (Rate per 100 ov/ob problems)	2,916	37.4 (34.4–40.4)	605	30.6 (24.6–36.7)	912	47.1 (39.9–54.4)	↑
At least one pathology order (% of ov/ob problems)	1,062	13.6 (12.6–14.6)	231	11.7 (9.7–13.7)	319	16.5 (14.1–18.9)	↑
Number of tests/ batteries per 100 test ov/ob problems	ed ··	274.7 (265.5–283.8)		262.3 (241.4–283.3)		285.9 (268.2–303.6)	

Table 5.38: Summary of pathology ordering for overweight/obesity (adult patients),2000–08, 2000–02 and 2006–08

Note: Pathology ordering data from 2000–02 and 2006–08 are drawn from Chapter 4, Table 4.3. Ov/ob – overweight/ obesity; CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/♥ indicates a statistically significant change, and — indicates no change.

Types of pathology tests/batteries ordered

Table 5.39 shows the rate of pathology tests/batteries ordered for overweight/obesity in 2000–08 by MBS groups and the most common individual types of pathology tests ordered. Chemistry tests were the group most often ordered (30.9 per 100 overweight/obesity management contacts) and the most common were:

- lipid tests (8.7 per 100)
- glucose/glucose tolerance tests (7.1 per 100)
- TFTs (5.0) (Table 5.39).

Haematology tests (5.2 per 100 contacts), in particular FBCs (4.7), were also commonly ordered in the management of overweight/obesity (Table 5.39).

One-quarter of pathology tests (26.5%) were ordered for 'new' cases of overweight/obesity. Tests were ordered at a higher rate for 'new' overweight/obesity problems (67.7 per 100 new problems) than average for all overweight/obesity problems (37.4 per 100 problems). However, the majority of pathology tests ordered were for ongoing management or monitoring of overweight/obesity (Table 5.39).

	All ove	rweight/obesity problems	New overweight/obesity problems			
– Pathology test	Number	Rate per 100 overweight/obesity problems (95% Cl) (<i>n</i> =7,797)	Number	Per cent of test	Rate per 100 new overweight/obesity problems (95% Cl) (n=1,141)	
Chemistry	2,411	30.9 (28.4–33.5)	643	26.7	56.4 (48.5–64.2)	
Lipids*	676	8.7 (7.8–9.5)	178	26.3	15.6 (13.1–18.1)	
Glucose/glucose tolerance*	550	7.1 (6.3–7.8)	144	26.2	12.6 (10.1–15.1)	
TFT*	392	5.0 (4.5–5.6)	119	30.4	10.4 (8.5–12.3)	
LFT*	248	3.2 (2.7–3.6)	70	28.2	6.1 (4.5–7.7)	
EUC*	197	2.5 (2.1–2.9)	47	23.9	4.1 (2.9–5.4)	
MBA*	149	1.9 (1.6–2.3)	47	31.5	4.1 (2.9–5.4)	
Hormone assay*	53	0.7 (0.4–0.9)	16	30.2	1.4 (0.6–2.2)	
Other chemistry*	46	0.6 (0.4–0.8)	8	17.4	0.7 (0.2–1.2)	
Ferritin*	30	0.4 (0.2–0.6)	3	10.0	0.3 (0.0–0.6)	
Haematology	408	5.2 (4.6–5.8)	112	27.5	9.8 (7.8–11.8)	
FBC	369	4.7 (4.2–5.3)	101	27.4	8.9 (7.1–10.6)	
ESR	30	0.4 (0.2–0.5)	9	30.0	0.8 (0.3–1.3)	
Other tests NEC	53	0.7 (0.5–0.9)	10	18.9	0.9 (0.3–1.4)	
Other pathology groups	44		7	15.9		
Total pathology tests	2,916	37.4 (34.4–40.4)	772	26.5	67.7 (58.6–76.7)	

Table 5.39: Rate of pathology test orders for overweight/obesity by MBS pathology groups and most frequent individual test orders within each group, 2000–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations. Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included.

Changes in the types of pathology tests/batteries ordered

Table 5.40 shows the most common pathology tests/batteries ordered for overweight/obesity in 2000–02 and 2006–08. Listed below are the tests for which significant change in GPs' order rate occurred between 2000–02 and 2006–08 (listed in decreasing test rate order). There were significant increases in the order rate of:

- TFTs—70% increase
- FBCs—68% increase
- 'other chemistry' tests—450% increase.

There was also a marginal increase in the order rate of MBA (Table 5.40).

	2	2000–02	2		
Pathology test	Number	Rate per 100 ov/ob problems (95% Cl) (<i>n</i> = 1,975)	Number	Rate per 100 ov/ob problems (95% Cl) (<i>n</i> = 1,935)	Change
Chemistry	498	25.2 (20.2–30.2)	754	39.0 (32.7–45.2)	1
Lipids*	153	7.8 (6.0–9.5)	208	10.8 (8.8–12.7)	
Glucose/glucose tolerance*	127	6.4 (4.8–8.0)	164	8.5 (6.7–10.3)	_
TFT*	72	3.7 (2.7–4.6)	122	6.3 (5.0–7.6)	↑
LFT*	45	2.3 (1.3–3.2)	78	4.0 (2.9–5.1)	
EUC*	38	1.9 (1.3–2.6)	63	3.3 (2.3–4.2)	_
MBA*	25	1.3 (0.7–1.8)	50	2.6 (1.8–3.4)	\mathbf{T}
Other chemistry*	4	0.2 (0.0-0.4)	21	1.1 (0.6–1.6)	↑
Hormone assay*	13	0.7 (0.2–1.1)	14	0.7 (0.3–1.2)	
Ferritin*	8	0.4 (0.1–0.7)	12	0.6 (0.1–1.1)	_
PSA*	6	0.3 (0.0–0.6)	9	0.5 (0.2–0.8)	_
Haematology	83	4.2 (3.0–5.4)	135	7.0 (5.6–8.4)	↑
FBC	75	3.8 (2.7–4.9)	123	6.4 (5.1–7.6)	↑
ESR	6	0.3 (0.1–0.5)	11	0.6 (0.2–0.9)	
Other tests NEC	17	0.9 (0.3–1.4)	17	0.9 (0.5–1.3)	_
Other test NEC*	5	0.3 (0.0–0.5)	12	0.6 (0.3–1.0)	_
Blood test	10	0.5 (0.1–0.9)	4	0.2 (0.0–0.4)	_
Other pathology groups	7		6		
Total pathology tests	605	30.6 (24.6–36.7)	912	47.1 (39.9–54.4)	↑

Table 5.40: Rate of pathology test orders for overweight/obesity by MBS pathology groups and most frequent individual test orders within each group, 2000–02 compared with 2006–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Ov/ob – overweight/obesity; CI – confidence interval; NEC – not elsewhere classified; also see Abbreviations. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/↓ indicates a statistically significant change (darker shading), ↑/↓ indicates a marginal change (lighter shading), and — indicates no change.
5.9.4 Guidance documents for overweight/obesity

Eleven guidance documents were reviewed for overweight/obesity (Box 5.6). Only

documents for overweight and/or obesity in adults were included.

Box 5.6: Guidance documents reviewed for overweight/obesity										
Title	Year	Author	Abbreviated to							
Clinical practice guidelines for the management of overweight and obesity in adults ²⁹⁷	2004	National Health and Medical Research Council (NHMRC) [Australia]	NHMRC							
Overweight and obesity in adults: a guide for general practitioners (GP) ²⁹⁸	2004	NHMRC [Australia]	NHMRC GP							
Canadian clinical practice guidelines on the management and prevention of obesity in adults and children ²⁹⁹	2006	Obesity Canada Clinical Practice Guidelines (OCCPG) Expert Panel	OCCPG							
Guidelines on management of adult obesity and overweight in primary care ³⁰⁰	2006	National obesity forum (NOF) [United Kingdom]	NOF							
In depth assessment resource for health professionals ³⁰¹	2004	NOF [United Kingdom]	NOF assess							
Obesity: guidance on the prevention, identification, assessment and management of overweight and obesity in adults and children ³⁰²	2006	National Collaborating Centre for Primary Care & the Centre for Public Health Excellence at National Institute for Health and Clinical Excellence (NICE) [United Kingdom]	NICE							
Quick reference guide 2 for the National Health Service (NHS) ³⁰³	2006	National Collaborating Centre for Primary Care & the Centre for Public Health Excellence at NICE [United Kingdom]	NICE NHS							
The practical guide to the identification, evaluation, and treatment of overweight and obesity in adults ³⁰⁴	2000	National Heart, Lung, and Blood Institute (NHLBI) Obesity Education Initiative Expert Panel on the Identification, Evaluation, and Treatment of Overweight and Obesity in Adults [United States of America]	NHLBI							
RCPA manual, obesity section ¹⁷²	2004	The Royal College of Pathologists of Australasia (RCPA)	RCPA							
Murtagh's general practice, weight gain chapter ¹⁷³	2007	Murtagh J [Australia]	Murtagh							
Patient presentations in general practice, 'management of overweight' ¹⁷⁴	1999	Steven I [Australia]	Steven							

5.9.5 Extent of alignment between GP testing and guidance documents

In Table 5.41, the frequently ordered pathology tests/batteries for overweight/obesity are categorised by level of support in the guidance documents listed in Box 5.6.

Supported tests

The four types of tests that were recommended in the majority of guidance documents were lipids, glucose/glucose tolerance, LFT and hormone assay (Table 5.41). Together these four tests accounted for 52.4% of tests/batteries ordered by GPs for overweight/obesity (Table 5.42). The order rate of these tests did not change between 2000–02 and 2006–08 (Table 5.40).

Guidance documents unanimously recommended lipid testing to identify lipid disorders and glucose testing to identify diabetes or impaired glucose tolerance.

Liver function testing was recommended in six guidance documents, four documents^{173,297-299} recommended testing all patients and two indicated it could be considered.^{300,304} When provided, the rationale for LFT was to identify non-alcoholic fatty liver disease and other liver disorders.^{297-299,304}

Of the 'hormone assay' test group for overweight/obesity, sex hormones accounted for about 80% and cortisol tests for the remaining 20%. Three documents recommended testing sex hormones for all female patients,²⁹⁷⁻²⁹⁹ and two listed it as a test to consider.^{173,300} The rationale for testing was to identify polycystic ovary syndrome and infertility problems. In five documents, cortisol testing was recommended if symptoms of Cushing's syndrome were present.^{172-174,300,301}

Tests with conditional support

There was mixed support for TFT in the management of overweight/obesity (Table 5.41). TFT accounted for 13.4% of tests ordered (Table 5.42) and the ordering rate increased significantly between 2000–02 and 2006–08 (Table 5.40).

Thyroid disease was discussed in guidance as a possible cause of overweight/obesity in seven guidance documents.^{172-174,298,300,301,304} Only two guidance documents recommended routine thyroid testing in the assessment of overweight/obesity, one of which recommended ordering TFT,¹⁷⁴ and the other recommended the thyroid-stimulating hormone test.³⁰¹ There were two occasions where guidance documents produced by the same authors (the NHMRC guideline and GP guide,^{297,298} and the

NOF guideline and assessment document^{300,301}) provided inconsistent recommendations for thyroid function assessment. This is discussed in more detail in Section 5.9.6.

Support unable to be determined

It was not possible to determine whether MBA and 'other chemistry' tests were recommended in the guidance documents (see Section 5.3.4). Together these two tests accounted for 6.7% of pathology ordered for overweight/obesity (Table 5.42). Between 2000–02 and 2006–08, the rate of MBA increased marginally and the rate of 'other chemistry' tests increased significantly (Table 5.40).

The MBA test itself was not recommended in any guidance documents. However, some tests that may be considered part of the MBA, such as LFT, were partially or completely supported in the management of overweight/obesity (as discussed above).

Unsupported tests

The guidance documents did not recommend testing of FBC, EUC, ferritin and ESR (Table 5.41). Together these tests accounted for 21.4% of all pathology ordered by GPs in the management of overweight/obesity (Table 5.42). FBC was the only test within this group to increase significantly between 2000–02 and 2006–08 (Table 5.40).

EUC testing or assessment of kidney function was not mentioned in most guidance documents. Murtagh¹⁷³ recommended testing kidney function in all patients and NOF³⁰⁰ and NHLBI³⁰⁴ listed it as a consideration.

The FBC, ferritin and ESR tests were not mentioned in most of the guidance documents, with the exception of the NHLBI guideline, that stated that FBC should be considered.³⁰⁴

Other tests mentioned in the guidance documents

Other tests were mentioned in the guidance documents but each of these accounted for less than 1% of tests recorded by GPs for overweight/obesity.

Urinalysis was recommended in two sources,^{299,300} and two documents recommended testing for gout/hyperuricaemia.²⁹⁷⁻²⁹⁹ The most common test for gout is urate/uric acid. While it was not among the most common individual tests, it may have been a part of the MBA test.

Key to Table 5.41

Colour	Description
	The document specifically recommended this test. Any notes within the cell indicate further detail. For example, a specific disease to test for within subset of patients; a specific test within a group.
	The document stated that this test should be considered. Any notes within the cell indicate further detail (e.g. a specific test to consider)
	Unable to determine guidance (see Section 5.3.4).
	The document specifically stated not to do this test. Additional information is supplied if certain conditions apply (e.g. specific clinical situations).
	Guidance document does not mention this test

Pathology test	Per cent of ov/ob path (<i>n</i> = 2,917)	NHMRC ²⁹⁷	NHMRC GP ²⁹⁸	OCCPG ²⁹⁹	NOF ³⁰⁰	NOF assess ³⁰¹	NICE ³⁰²	NICE NHS ³⁰³	NHLBI ³⁰⁴	RCPA ¹⁷²	Murtagh ¹⁷³	Steven ¹⁷⁴
Lipids*	23.2											
Glucose/glucose tolerance*	18.9											
TFT*	13.4			Only if clinically indicated		Thyroid- stimulating hormone						
FBC	12.7											
LFT*	8.5											
EUC*	6.8											
MBA*	5.1											
Hormone assay*	1.8	Polycystic ovary and fertility	Polycystic ovary and fertility	Polycystic ovary	Cortisol and sex hormones	Cortisol				Cortisol	Cortisol and sex hormones	Cortisol
Other chemistry*	1.6											
Ferritin*	1.0											
ESR	1.0											

 Table 5.41: Summary of guidance recommendations by most frequent individual test orders for overweight/obesity, 2000–08

* Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Ov/ob – overweight/obesity; also see Abbreviations. Any notes within the coloured cells are described in detail in Section 5.9.5. Only the groups of tests/individual tests accounting for > 1% of all pathology tests for the selected problem are included.

Evaluation of GP pathology ordering against guidance documents

Table 5.42 provides a summary of the individual tests ordered for overweight/ obesity and the level of support provided in the guidance for each. Of the tests ordered by GPs in 2000–08:

- 52.4% were supported
- 20.1% had conditional support or support could not be determined
- 21.4 were not supported by the guidance documents (Table 5.42).

The individual tests/batteries listed in Table 5.42 account for 93.9% of pathology tests/batteries ordered for overweight/obesity because only the most commonly ordered tests for overweight/obesity were evaluated.

		Per cent of all pathology tests
Pathology test supported by guidance	Number	for overweight/obesity
Supported	1,527	52.4
Lipids*	676	23.2
Glucose/glucose tolerance*	550	18.9
LFT*	248	8.5
Hormone assay*	53	1.8
Conditional/unclear support	587	20.1
TFT*	392	13.4
MBA*	149	5.1
Other chemistry*	46	1.6
Unsupported	625	21.4
FBC	369	12.7
EUC*	196	6.8
Ferritin*	30	1.0
ESR	30	1.0
Subtotal (n, % of total tests)	2,739	93.9
Total pathology tests	2,916	100.0

Table 5.42: Summary of support for GP pathology ordering for the most frequent individual test orders for overweight/obesity, 2000–08

Includes multiple ICPC-2 PLUS codes (see Appendix 7).

Note: Only the groups of tests/individual tests accounting for ≥ 1% of all pathology tests for the selected problem are included. See Abbreviations.

In 2000–08, GPs ordered supported pathology tests at a rate of 19.6 tests/batteries per 100 overweight/obesity problems, followed by unsupported tests (8.0) and those with conditional support (7.5). Tests ordered for overweight/obesity but not evaluated

were ordered at a rate of 2.3 per 100 problems. The rate at which GPs ordered unsupported and conditionally supported tests increased significantly over time (Table 5.43).

	2000–08	2000–02	2006–08	
Level of support	Rate per 100 overweight/obesity problems (95%CI) (<i>n</i> =7,797)	Rate per 100 overweight/obesity problems (95%Cl) (<i>n</i> = 1,975)	Rate per 100 overweight/obesity problems (95%Cl) (n=1,935)	Change
Supported	19.6 (17.7–21.4)	17.1 (13.2–21.0)	24.0 (19.4–28.6)	_
Conditional/unclear support	7.5 (6.7–8.3)	5.1 (3.9–6.4)	9.8 (8.0–11.7)	↑
Unsupported	8.0 (7.1–8.9)	6.4 (4.7–8.2)	10.8 (8.6–13.0)	↑
Not evaluated	2.3 (2.0–2.7)	2.0 (1.3–2.7)	2.5 (1.8–3.3)	

Table 5.43: Rate of pathology ordering for overweight/obesity by level of support,2000–08, 2000–02 and 2006–08

Note: CI – confidence interval. Shading indicates a statistically significant change between 2000–02 and 2006–08. The direction and type of change is indicated for each measure: ↑/↓ indicates a statistically significant change, and — indicates no change.

GPs ordered only pathology tests that were supported or conditionally supported at 7.1% (95% CI: 6.4–7.9) of overweight/obesity problems in 2000–08. A further 5.9% (95% CI: 5.2–6.5) of problems involved at least one unsupported test, and 87.0% (95% CI: 86.1–88.0) of problems involved either no pathology tests (86.4%) or tests that were not evaluated (0.6%) (results not tabled).

Between 2000–02 and 2006–08 there was no change in the proportion of overweight/obesity problems that involved supported testing, 6.1% (95% CI: 4.8–7.4) in 2000–02 and 8.1% (95% CI: 6.2–9.9) in 2006–08. However the proportion of problems involving at least one unsupported test increased significantly over this time, from 4.8% (95% CI: 3.5–6.0) to 7.7% (95% CI: 6.3–9.2) (results not tabled).

When GPs ordered unsupported tests, the vast majority were accompanied by one or more supported/partially supported tests (96.6% of occasions). Unsupported tests were ordered alone at the remaining 3.4%. GPs ordered only tests that were unsupported at 0.2% of overweight/obesity problems in 2000–08, and this did not change over time (0.1% in both 2000–02 and 2006–08) (results not tabled).

5.9.6 Discussion

GPs' pathology test ordering for overweight/obesity problems (in adult patients) aligned moderately well with that recommended in guidance documents, with 72% of tests partially or completely supported. The statistically significant increase in the rate of GPs total pathology ordering for overweight/obesity problems between 2000–02 and 2006–08 was reflected in the increased rate of tests that were partially supported, and in those unsupported in guidance documents. There was no change in the rate of supported tests.

Due to the increase in total rate of pathology ordering, the correlation between GPs' ordering and level of support, as a proportion of total tests should be considered. Supported tests accounted for the largest proportion of total tests, but they accounted for a smaller proportion of total tests in 2006–08 than in 2000–02. Concomitantly there was an increase in the proportion of partially supported tests and a small increase in unsupported tests. Overall the increase in total tests ordered by GPs for overweight/obesity suggested a small shift (less than 2% of total pathology) towards being 'less' in line with guidance recommendations. However, there was a more pronounced shift in GPs' ordering, from supported to partially supported tests that warrants further investigation.

Comments on guidance documents

Guidance documents for management of overweight/obesity logically followed the 'normal' management pathway from identification, to assessment, to management, to monitoring.

Most documents had no clear section for recommended pathology tests. Mixed terminology was used to refer to testing e.g. 'diagnostic testing', 'laboratory investigations', 'diagnostic investigations', 'assessment', the specific test name, or the disease to be tested for—making identification of recommended testing within the lengthy documents (often over 200 pages) difficult. The Australian NHMRC guideline (and related GP guide) did not specifically recommend any pathology tests, conditions associated with obesity being listed by relative risk of developing the condition with the statement 'standard procedures are used to test for these'.^{297,298}

Contradictory statements

Recognising that size, complexity and structure of guidance documents was a barrier to them being utilised, some authors created an abbreviated guidance document. However, there were inconsistencies between the full and abbreviated guidance. There were also conflicting statements made within the same guidance document.

The Australian NHMRC guideline and the related GP guide provided contradictory advice for thyroid disease. In the full guideline thyroid disease was not mentioned as a disease to consider in assessment. However, hypothyroidism was mentioned in two other places in the guideline: as a condition that places people at risk of obesity with the statement that it generally shows up earlier in life; and, in the alternative treatment section it was noted that it caused "very few cases" of obesity.²⁹⁷ These comments suggest that the authors of the full guideline do not consider thyroid disease testing to be a routine part of the assessment of overweight/obesity. In contrast, the GP guide stated "Medical conditions: Certain medical conditions, for example, hypothyroidism, are known causes of overweight" and goes on to recommend that these medical conditions should be assessed by clinicians.²⁹⁸ This suggests support for thyroid testing in the GP guide, though not in the full guideline. The recommended pathology tests in the NOF guideline and separate 'full assessment' did not align. In the NOF guideline³⁰⁰ a number of blood tests were listed 'if appropriate', suggesting they would not be necessary in all patients. In contrast, the assessment document recommended a smaller number of tests for all

patients.³⁰¹

Authors of two guidance documents made inconsistent recommendations within the document. The NHLBI guideline listed the lipids and glucose tests as recommended tests in one section of the guideline and as tests to be considered in another section.³⁰⁴ The NHRMC GP guide listed conditions to test for in the assessment phase using "standard testing", but also included as an appendix a Weight Management Plan, which prompted for specific test results. These were: triglycerides, cholesterol, insulin, glucose, LFT, and endocrine tests. However, insulin is not a recommended standard pathology test to use for testing any of the conditions.

Testing for phase of management

Pathology tests were, in the main, only recommended in the assessment phase of management (i.e. prior to starting treatment) to identify the presence/absence of other morbidities associated with overweight/obesity, and/or possible medical causes for overweight/obesity. It is likely that GPs ordered pathology tests/batteries for the assessment of overweight/obesity at the time of its 'diagnosis' as a new clinical problem, which explains the higher pathology order rate at these new cases than the average rate for overweight/obesity problems.

Multiple guidance documents acknowledged the chronic nature of overweight/obesity and the need for long-term management.^{297-299,302-304} The majority of GP-patient contacts for overweight/obesity management (85%) were for its ongoing management, as were the majority of pathology tests/batteries ordered (74%). However, only one guidance document made a recommendation about pathology testing in ongoing management: the Canadian guideline made a consensus recommendation to retest glucose and lipid levels 'at regular intervals'.²⁹⁹

There was a lack of guidance about whether there is a need for pathology testing in ongoing management to reassess the absence or presence of associated morbidities. As the majority of contacts and pathology testing were provided by GPs for ongoing management of overweight/obesity this is an area authors of future guidance documents should improve.

Tests for which guidance was lacking

There was a significant discord between guidance documents' recommendations and GPs' ordering for overweight/obesity for three tests: FBC, TFT and MBA. FBC was only mentioned in one guidance document, as a possible inclusion in initial assessment, and the MBA was not recommended in any guidance documents. Only TFT was commonly mentioned in guidance documents. However, due to the variation between recommendations in guidance documents TFT was judged as 'partially supported' in this study. However, when it was recommended it was only in the initial assessment of overweight/obesity, but GPs usually ordered it in ongoing management (70% of occasions). The increase in the proportion of GP-ordered tests classified as 'partially supported' (mentioned earlier in this discussion) was mainly due to increases in the rate of TFT and MBA tests. The three tests (FBC, TFT and MBA) accounted for one-third of total tests ordered for overweight/obesity in 2000–08. GPs' ordering of these tests increased significantly between 2000–02 and 2006–08, and the increases in order rates of TFT and FBC were the main contributors to the total increase in GPs' pathology ordering for overweight/obesity.

Investigation of whether the FBC and TFT tests are clinically useful in the management of overweight/obesity and at which stage of management (including for ongoing monitoring) is needed. The results of this investigation should be used to improve guidance documents, and this guidance promoted to GPs to improve test ordering.

Evidence base for overweight

A lot of the available evidence, on which recommendations in the documents were based, was for obesity only. For example, the studies referenced in guidance documents that identified links to associated morbidities and causative conditions/diseases were among obese patients, not in the overweight. However, recommendations were made for overweight and obese patients based on this evidence though the causative links may be less clear in overweight patients.

Factors influencing the future management rate of overweight /obesity

The prevalence of overweight/obesity is high and increased over the period of this study (between 2000 and 2008) in the Australian population,¹⁵⁴ and in the Australian general practice patient population.²⁷³ However, there was no change in the management rate of overweight/obesity as a clinical problem in general practice. It remained very low. In 2000–08, on average, GPs saw 51 overweight/obese patients for every one management occasion.

Multiple studies have similarly noted the discord between prevalence of obesity and its management in primary care. The proportion of obese patients who recall receiving weight management or have it documented in the record varies from 19% to 49%.³⁰⁵⁻³⁰⁸ GPs acknowledge the importance of management of overweight and obesity and most report feeling well prepared to manage it.³⁰⁹ However, they also perceive that they are unlikely to be very effective in influencing the factors driving obesity in patients,^{310,311} despite evidence that patients who receive weight loss counselling are more likely to attempt weight loss.³⁰⁵

As the current management rate is low, the amount of total pathology accounted for by overweight/obesity is also small. However, with public awareness campaigns underway to address overweight/obesity at a population level it is likely that its management rate in general practice will increase as it is the usual first point of contact Australians have with the health care system. The potential for increased management is huge and would create a corresponding increase in pathology, which would occur without any change in GPs' current pathology ordering behaviour. However, GP's pathology ordering for overweight/obesity has increased and may continue to do so, which would further increase future pathology test rates for overweight/obesity.

The inclusion of obesity as a National Health Priority Area²⁹⁰ and as part of the National Preventative Health Strategy²³⁵ reflects the urgency and scale of the overweight/obesity problem. The Strategy's Obesity Working Group recommendation to 'strengthen, skill and support primary healthcare and public health workforce to support people in making healthy choices'²³⁵ is timely.

The Australian NHMRC guideline and GP guide were created seven years ago and are overdue for revision, providing an opportunity to improve the overweight/obesity guidance available to GPs addressing the deficiencies highlighted in this study. It is clear from the Working Group' recommendation that primary care workers will continue to have a role key in the management of obesity in the future.

Summary

More than 70% of pathology ordered by GPs in the management of overweight/obesity were recommended in guidance documents. Based on the current prevalence-management pattern and increases in GP's pathology ordering for overweight/obesity there is potential for future growth in pathology ordering associated with overweight/obesity. Future guidance developed for GPs should address the gaps and limitations highlighted in this study, particularly in long-term management. Addressing the appropriateness of GP test ordering in management of overweight/obesity may decrease some future costs associated with its management.

5.10 Discussion

GP adherence to guidance documents

Pathology tests/batteries ordered by GPs in the management of hypertension, T2D, lipid disorders and weakness/tiredness aligned well with those recommended in guidance documents, with at least 85% of tests ordered for these problems being completely or partially supported. This was not the case for 'health check' and overweight/obesity problems, with 50% and 70% of tests, respectively, being completely or partially supported in guidance documents.

Although only six problems were investigated in this chapter, they accounted for one-quarter (25%) of the total pathology tests/batteries recorded by GPs in 2000–08. When viewed as one group, 66% of the pathology ordered for the six conditions were supported in guidance documents, 15% were partially supported, 13% were unsupported and 6% were not evaluated. When described in terms of the total volume of pathology ordered 17% of tests were supported for the six problems, 4% were partially supported, and 3% were unsupported.

Over the period of this study, GPs' pathology ordering for the six problems increased, and this represented a shift to be 'less' in line with recommendations in guidance documents. The proportion of tests ordered by GPs that were supported decreased (from 70% in 2000–02 to 61% in 2006–08), counteracted by increases in the proportion that were partially supported (13% to 17%) and unsupported (11% to 16%). The proportion of tests that were not evaluated in this study remained constant (6% in 2000–02 and 2006–08). Although the proportion of unsupported tests increased, it is important to note that the increase of 5% was relatively modest over the 8 years of the study. Inappropriate pathology ordering was not the major contributor to the increased rates of GPs' pathology ordering, as increases occurred in all 'level of support' groups.

The results in Chapter 4 suggest that 13.6 million tests/batteries were ordered for the six problems in 2006–08 in Australia. This is the extrapolated number of pathology tests ordered across the country and incorporates GPs pathology ordering for the problems, the management rate of the problems and the number of national GP encounters. Of these tests, 10.7 million (78.2%) were completely or partially

supported in guidance documents, 2.1 million (16.6%) were not supported and 850,000 (6.2%) were not evaluated in this study.

It is interesting to note that most of the pathology tests evaluated as unsupported in the current study were not mentioned in the guidance documents, rather than being specifically recommended against. In other words, guidance documents were prescriptive in encouraging the necessary pathology tests and rarely proscriptive in recommending against use of pathology tests.

It may be that some of the tests not mentioned in guidance documents, and therefore evaluated as unsupported, have a role in the management of the clinical problem. Prior to implementing interventions to curb these 'unsupported' tests, further investigation of the clinical value of such tests may be needed.

When GPs ordered unsupported tests this was primarily in addition to ordering supported tests. It was rare that an order for pathology for any of the six problems included only unsupported tests. This aligns with the study by van Wijk et al.⁷⁹ that found the most common cause of 'non-compliant' pathology ordering by GPs was the addition of tests. However, it was also found that half of the tests considered 'non-compliant' were subsequently recommended in the revised guidelines, leading van Wijk et al. to conclude that GPs were early adopters of new evidence.⁷⁹ This does not appear to be the case in the current study. The eight years of data evaluated and the range of publication dates of the reviewed guidance documents suggest that if GPs' ordering of unsupported tests was pre-empting changes in guidance this would have been noted.

The fact that unsupported tests were rarely ordered alone has significant implications in the context of the current MBS funding structure of pathology tests ordered by GPs in Australia. In studies evaluating the cost-effectiveness of implementing pathology interventions in other countries, cost-effectiveness was favourable on the basis of the saved testing costs.^{76,78,312} However, because of the coning structure in the current funding arrangement, only the three most expensive MBS pathology item numbers per testing episode are funded by the Australian Government. This means that reducing the ordering of inappropriate tests may not achieve a reduction in cost of the tests at an episode, unless they fall in the three most expensive pathology items ordered on that occasion.

Limitations

Essentially this study assesses the appropriateness of GPs' pathology test selection in the management of the six investigated problems, with appropriateness judged on recommendations for testing provided in guidance documents. It was not the purpose of this study to determine whether GPs performed all the recommended tests for each problem. There are circumstances where GPs' use of appropriate tests may be suboptimal.

There were a few occasions where guidance documents recommended pathology tests that GPs did not order. These tests were not strongly recommended, that is, there was not consensus among guidance documents for these tests. The only exception was the urinalysis test, which was supported in guidance documents for hypertension and weakness/tiredness. It is likely that urinalysis was not recorded by GPs participating in BEACH because GPs are instructed not to record dipstick tests. Urinalysis includes a dipstick and microbiology test. Therefore, low rates of urinalysis may reflect a limitation of the BEACH pathology data. However, in general, GPs' failure to order tests that were not strongly supported in guidance document is not likely to be a concern.

There are limitations related to the type of data available in this study. The crosssectional and encounter-based nature of the data meant some clinical information (e.g. presence of clinical signs, comorbidities, test results, interval between repeated tests) was not known. When a guidance document recommended a test in a selected clinical circumstance, GPs' ordering of the test was evaluated as 'supported' in the current study. This is a somewhat lenient approach to assessing the appropriateness of testing. It is possible that a test that was evaluated as supported in this study may not be appropriate in the circumstances of the individual patient. It is also possible that a test not recommended in guidance may be appropriate in the circumstances of the individual patient.

The measure used to assess 'appropriateness' in this study, is just one possible measure. Others have used different measures (such as unnecessary repetition of tests³¹³) although most of these have been applied in settings other than general practice.⁵²

There are also limitations to the pathology data recorded in the BEACH study. As discussed in Chapter 3, there are a maximum of five pathology tests/batteries of tests that can be recorded per encounter. Each test may relate to more than one problem being managed and the analysis in this chapter is based on these problem–pathology linkages. It was possible for a single pathology test/battery to be linked to more than one problem, and therefore evaluated in this study for more than one problem, but this was rare.

Some of the guidance documents reviewed in this study were published subsequent to the data being recorded. However, the relative consistency of recommendations between guidance documents and consistency in the types of tests ordered by GPs for each problem suggest that differences in publication dates of the guidance documents would not have affected the conclusions of this study.

Guidance documents

Guidance documents, in particular guidelines, have been shown to be successful in improving care when produced, disseminated and implemented successfully.^{11,85,314} However uptake of guidance is often poor for a variety of reasons. Barriers to guideline adherence may relate to the individual patient, clinicians attitudes to guidelines, the characteristics of the guidance (e.g. the source, evidence base and practical features such as length), and characteristics of the individual recommendations.^{153,315-323} In this study, while acknowledging that these barriers exist, guidance documents are viewed as summaries of the best available evidence and reflect the guidance available to GPs for pathology testing.

As a whole, the quality of the guidance documents reviewed in this study was highly variable. For example, it is likely that some of the guidelines reviewed would not meet the indicators for high quality using the Appraisal of Guidelines for Research and Evaluation (AGREE) instrument.³²⁴ However, it was not the purpose of this study to assess the quality of the guidance documents in this respect. Nor was it within the scope of this thesis to determine the accuracy or appropriateness of the pathology recommendations made by authors of guidance documents (e.g. that the evidence was correctly interpreted). Instead guidance documents were viewed to determine what is available to GPs in terms of guidance about pathology testing.

I may not have identified and reviewed all the relevant guidance documents available for the six problems investigated in this study. Also as the study was limited to guidance documents available in English, most of the Dutch College of General Practitioners guidelines could not be evaluated, despite being widely referenced in the literature.^{76,79,86,88,95,281}

Locating relevant Australian guidelines and other guidance documents was not straightforward as several organisations produce guidance documents.³²⁵ At the time this study was conducted there was no central listing of Australian guidelines or other guidance documents. In contrast, other countries had organisations responsible for guideline creation and dissemination (such as NICE in the UK) and/or central listings of guidelines (e.g. the US National guideline clearinghouse). Since this study was completed, a web-based 'National clinical practice guidelines portal' has been launched³²⁶ which should facilitate locating guidelines in the future.

The guidance documents reviewed, commonly guidelines, were usually not designed for GPs. Many of the documents were lengthy (frequently 200+ pages) in order to include all the evidence. The size and number of guidelines applicable to diseases commonly managed in general practice has been acknowledged as a barrier to their use by GPs.^{153,319,322,323}

In addition, locating information about recommended pathology testing in the guidance documents was often difficult. There was mixed terminology used to refer to testing e.g. 'diagnostic testing', 'laboratory investigations', 'diagnostic investigations', 'assessment', and the specific test name, or the disease that testing aims to identify or exclude. Implementing common terminology to refer to pathology tests in guidance documents would be one easy way to help clinicians identify relevant recommendations in these lengthy documents.

This study identified quite a few inconsistencies in pathology recommendations both within and between guidance documents for equivalent clinical situations. Incomplete pathology test recommendations were also identified, particularly in the context of long-term management (this is discussed in more detail below in 'Chronic conditions').

I found one other study that investigated the quality of blood test recommendations in guidelines. Van Wijk et al.³²⁷ reported only a few instances of incomplete blood

test recommendations in the guidelines of the Dutch College of General Practitioners. Van Wijk's study suggests a better quality of recommendations in the Dutch guidelines than found in the current study. This is probably because the guidelines reviewed were all from one source. Additionally these guidelines have been extensively used in studies to improve GPs' pathology ordering in the Netherlands^{76,79,86,88,95,281} over several decades and it is likely that any inconsistencies have been corrected over time.

The problems investigated

The six problems included in this study can be considered to represent three broad groups: chronic conditions, diagnostic problems and preventive care. Lipid disorders, T2D, hypertension, and overweight/obesity are chronic conditions. Health check, as a problem under management, represents a preventive process of care and the management of weakness/tiredness problems represents a diagnostic process. There were different issues raised in this study that were common to each of these three broad groups.

Chronic conditions

The study highlights that the majority of GP management and pathology tests/batteries ordered (70–90% of pathology) for the four chronic conditions were for the long-term management of the conditions. Despite this, recommendations regarding pathology testing in long-term management of conditions were frequently omitted or incomplete in guidance documents.

Guidance documents often included recommendations about monitoring the key indicator(s) for the disease (e.g. lipid levels for lipid disorders, HbA1c for diabetes). However, recommendations regarding the monitoring of other factors including medication side effects were frequently omitted or incomplete. For example, guidance documents recommended monitoring liver function in the use of statin medications but gave no advice on the frequency and duration of monitoring.

For hypertension, lipid disorders and overweight/obesity a number of pathology tests were only recommended as part of the initial assessment of new cases of disease (e.g. those recommended to identify possible underlying causes of the disease). In this study, the level of support may be over-estimated for such tests. GPs continued to order these pathology tests in the ongoing management of the problems, that is, beyond the initial assessment. No guidance was provided on the need (or not) to periodically reorder tests recommended in the initial assessment. For example, guidance is needed on whether the secondary causes of disease (recommended for testing in the initial assessment) are likely to occur in the future (e.g. increasing prevalence with age), and therefore whether these secondary causes need to be reassessed. Such guidance should inform GPs whether subsequent diagnosis of the condition is likely to affect management of the disease and whether repeated testing is needed.

In addition to incomplete guidance about pathology tests needed for monitoring chronic conditions, there was also a lack of information on interpretation of repeated testing. Even among tests with an established role in monitoring chronic conditions, the expected variation of test results (i.e. the expected analytical and intra-individual variation in repeated test results) was not discussed in guidance documents. Knowledge of this variance is essential for GPs to accurately interpret results, to determine the level at which clinical intervention (e.g. increase medication dose) is required to maintain 'control' and avoid progression of disease.³²⁸

Lipid testing is a good example to highlight the issues around recommendations for monitoring. In this study lipid testing was supported for the four chronic conditions, but the level of detail provided in guidance documents about reassessing or monitoring lipid levels varied. No guidance documents discussed interpretation of repeated lipid results (i.e. expected variation of results), although its variability is well studied.^{224,226-229} Recommendations to monitor lipid levels and the interval between repeated testing were usually complete in guidance documents for lipid disorders and T2D. However this was not the case for hypertension and overweight/obesity. On the few occasions where recommendations for repeated lipid testing were made, they were vague, such as reassess 'regularly' or 'at regular intervals'. GPs' ordering of lipid tests is further investigated in a substudy in Chapter 6.

The current study suggests that there is scope for improvement in pathology ordering guidance in the long-term management of chronic diseases. Complete guidance is needed on tests required in ongoing management particularly beyond the primary monitoring test(s) for conditions. These improvements should include whether other tests have an ongoing role in the long-term management of the condition (such as those recommended in the initial management) and monitoring of medication side effects. Specifically these recommendations should be complete and include the recommended interval between tests and the duration that testing should persist. It has been said that monitoring practices are understudied,³²⁸ and that quality evidence in monitoring protocols in health care is lacking.^{56,329} This may explain why the monitoring recommendations in guidance documents are not ideal. Improving recommendations about the testing interval for monitoring is not straightforward because frequency may depend on the phase of management.^{328,329} Never-the-less making improvements in recommendations for monitoring tests (including interval and interpretation) is worthwhile considering the volume of testing generated by monitoring chronic disease in general practice.

Multiple morbidity

Managing patients with multiple chronic problems has become part of everyday practice for GPs.^{330,331} Prevalence of multimorbidity increases with age^{330,331} and as the Australian population ages the proportion of people with multiple chronic morbidities will increase.

Both guidance documents and this study are disease-specific. This is necessary because much of the evidence is disease-specific. However disease-specific guidance documents do not reflect the real world experience of managing patients with multiple diseases in general practice.^{153,332,333} A similar limitation applies in this study when comparing disease-specific recommended testing with actual clinical practice.

There is a need for improved guidance for pathology testing in patients with multiple morbidities. As many individual chronic diseases require pathology testing in their ongoing management, producing guidance relating to appropriate pathology in multiple morbidity situations may benefit GPs. However, since the number of possible combinations of chronic diseases is almost innumerable, realistically such guidelines should concentrate on the most prevalent combinations of disease.

Testing for a diagnostic purpose

Pathology tests are usually ordered by GPs for diagnostic purposes in the management of weakness/tiredness. The low prior probability of disease in these patients and the high rates of pathology ordering mean that the likelihood of receiving false positive or false negative tests results is common.

Even though GPs' test selection aligned well with that recommended in guidance documents, most of these recommendations were based on consensus because there is little available evidence. The few studies conducted suggest that limiting the number of tests (so that fewer than that currently recommended in guidance documents are ordered) may reduce the likelihood of false positive results with no change in the diagnostic yield of pathology testing for fatigue.^{281,287} However, the study by Koch et al.²⁸¹ suggested that even with a limited set of pathology tests, the chances of false positive results are relatively high. Increasing the number of tests ordered per testing occasion and the characteristics of tests ordered (such as sensitivity) contribute to the likelihood of false positive results.

In the current study, the odds of false positive test results are not likely to have increased over the study period, as the number of tests ordered per testing occasion and the types of tests ordered for weakness/tiredness did not change significantly between 2000–02 and 2006–08. Therefore, the odds of false positive results are unlikely to have contributed to the increased rate of pathology testing for weakness/tiredness. The increase rate of GPs' pathology ordering in the management weakness/tiredness was due to an increased likelihood that GPs would initiate pathology testing (i.e. an increased proportion of patients with weakness/tiredness problems were being tested).

Increasing the number of patients being tested may create a cascade of effects that contribute to more pathology testing. For example, assuming that the odds of receiving false positive results stay the same and GPs react to these abnormal results in the same way (with the same rate of follow-up pathology tests), an increase in the volume of patients being tested will flow on to an increased volume of patients experiencing false positive results and receiving follow-up testing. This is not to suggest that this is clinically inappropriate testing – but simply demonstrating the cascade that may be triggered by increasing GP likelihood of testing.

Based on the low probability of disease and high rates of testing (as described above), it can be assumed that receiving false positive results or clinically insignificant abnormal results is a common occurrence in general practice for weakness/tiredness. It may be that GPs' management of abnormal results in the management of weakness/tiredness changed and this contributed to increased rates of pathology testing. Certainly the management rate of abnormal tests results (as a clinical problem) in Australian general practice has increased, so too has GPs' pathology ordering in managing these abnormal results (as described in Chapter 4). A high probability of clinically insignificant abnormal test results in diagnostic situations is not unique to weakness/tiredness—GPs deal with similar situations on a regular basis. Authors have discussed the potential harms of clinically insignificant abnormal test results, such as cascades of investigations and testing, unnecessary procedures and patient anxiety.^{9,53,144,334} However, Houben et al. found that in diagnostic situations where probability of disease is low (as is the case for tiredness) GPs interpreted most abnormal tests results as 'normal' or clinically insignificant,³³⁵ and cascades of unnecessary investigations, referrals and treatments were uncommon.³³⁶ The small sample of GPs (n = < 90) in these studies may limit the generalisability of the results, particularly as other studies have reported considerable variation in GPs' pathology ordering behaviour. 43-51,71-73,337

Patient experiences of abnormal results are not well understood, but it is likely that they will also be highly variable. An interesting contradiction is apparent when you consider that one of the motivations cited by GPs^{36,37,338} for ordering pathology tests (and cited by patients for wanting to be tested^{33,339}) in diagnostic situations is to reassure, but abnormal results may achieve the opposite effect, leading to patient anxiety—a conundrum considering that clinically insignificant abnormal results are a common occurrence in these diagnostic situations.³³⁸ Future investigation of the flow-on effects of abnormal pathology results in Australian general practice for weakness/tiredness, or similar diagnostic problems, would identify the frequency of the discussed 'harms' and whether interventions (such as education) either for GPs, patients or both are needed.

Prevention

Health checks are considered a valuable tool in providing preventive care. Their use has been shown to improve use of cholesterol tests, Pap smears and faecal occult blood testing.³⁴⁰ However, there is a significant amount of literature that shows performance of preventive health activities in primary care is suboptimal.^{92,341-343} The Australian Government has introduced several MBS items for health assessments^{18,232,236} among other initiatives to "ensure that we encourage GPs to deliver quality preventative health."³⁴⁴

Health checks had the poorest alignment between GPs' pathology ordering and guidance recommendations of the six problems investigated. This may be partially due to the diverse range of clinical situations that GPs may face when using the 'health check' label. For example, MBS items were introduced to encourage preventive care in a range of selected high risk patient groups, such as older patients aged 75 years and over,²³⁶ patients at risk of chronic disease,²³² and patients at risk of T2D.¹⁸ While the definition of health checks used in this study excludes those related to established disease, it is possible that the patient's individual clinical circumstances call for different pathology tests than those recommended in guidance documents. As discussed in the limitations section above, the cross-sectional encounter-based nature of this study meant that some data about the patient's individual clinical circumstance was not available. Therefore a conservative approach was used; all pathology tests recommended in guidance documents for preventive care were included as supported tests. However, as discussed below, the difference between tests required for an individual patient's clinical situation and tests recommended in preventive care guidance documents may be more pronounced than for the other 5 problems evaluated in this study.

In considering appropriateness of pathology ordering Lundberg states that we need to consider the effect of the test in terms of whether "it was useful for the patient or for the public's health".⁴² Of course a test may be useful to both the patient and the public. However, the concept of individual benefit versus public benefit may partially explain the discord between pathology tests ordered by GP's for health checks and those recommended in preventive care guidelines.

Prevention is a core population health strategy, and as such guidelines for preventive care have their basis in public health benefit. By contrast the interaction between a

GP and patient is primarily about the patient's individual benefit. Beaulieu et al. found that both physicians and patients were as concerned with usefulness of a preventive intervention to the individual as to the general population.³⁴¹ The lack of applicability of guidelines to the individual patient has been identified as a barrier to guideline adherence particularly for GPs.^{153,316,317,319-321} Potentially this may be more pronounced for guidelines based on population benefit (such as preventive care) than those created for a specific morbidity.

Over the study period, there was an increase in the number of tests ordered by GPs per ordering occasion for health checks. This may have increased the likelihood of false positive results, particularly among well patients.¹⁴²⁻¹⁴⁴

One of the limitations of the cross-sectional data is that we don't know the nature of the 'health check' being provided (e.g. presence of other risk factors or morbidities) and cannot determine the type of preventive care likely to be represented.

The RACGP guideline states that health checks must "involve preventive interventions for which there is clear evidence of their effectiveness."¹⁷⁷ It appears that the Government's initiatives to increase the number of health checks were successful but this has also increased the rates of both supported and unsupported pathology tests for preventive care.

Only a few guidance documents were available for preventive care and the tests that were unsupported for health checks were usually not mentioned in guidance documents rather than specifically recommended against. This suggests that further research is needed to determine whether there is evidence to support the pathology tests that were unsupported in this study but commonly ordered by GPs for health checks.

Specific tests of interest

There were a few tests identified in this study which were commonly ordered in management of the six problems, but for which GPs ordering and recommendations in guidance documents frequently did not align. These tests were FBC, TFT and MBA.

These tests are highlighted because the GP ordering rate increased significantly, they account for a significant proportion of pathology ordered for these six conditions but guidance documents frequently did not support use of the tests.

Further investigation is needed to determine the clinical indications for ordering FBC, TFT and MBA tests, particularly to investigate their role in the long-term monitoring of the chronic conditions. GPs' ordering of FBCs are investigated in a separate study in Chapter 6.

Conclusion

The majority of pathology ordered for the six problems were supported in guidance documents. The significant increases in GPs' rates of pathology ordering for these problems were reflected in significant increases in tests at all levels of support. Thus GPs' pathology ordering in these six problems does not reflect significant amounts of inappropriate tests, nor does it support the assertion that increases in pathology ordering are caused by disproportionate increases in unsupported tests.

As a proportion of total tests, there was an increase in the proportion of unsupported tests over time, suggesting a shift toward GPs' pathology ordering being less in line with recommendations over time. However, this was a modest change in the context of the increases in total pathology rates.

There is scope for improvements in GPs' pathology ordering for the six problems investigated. However there are considerable differences between the problems. GPs' pathology test ordering aligned well with recommendations for pathology testing in guidance documents for hypertension, T2D, lipid disorders and weakness/tiredness. However, due to the lack of guidance on pathology testing in ongoing management, the level of support for some tests may be over-estimated, particularly for hypertension and lipid disorders. Pathology tests/batteries ordered by GPs for 'health checks' and overweight/obesity problems did not align well with recommended testing.

Pathology recommendations in guidance documents could be improved in multiple ways. Removing inconsistent pathology test recommendations, and standardising terminology used to refer to pathology testing would help GPs locate relevant information within the lengthy guidance documents. The most important improvements needed are for guidance in the long-term monitoring of chronic problems. Guidance documents need to include detail about: which tests need to be monitored, the frequency and duration of monitoring required; and the interpretation of results of repeated tests.

6 Investigation of FBC and lipid testing

6.1 Background

In previous chapters of this thesis, the focus of the research has been the problems for which GPs order high volumes of pathology tests and, for a selection of these problems, the extent to which GPs' test selection represents that recommended in evidence-based guidelines and guidance documents. In contrast, the focus in this chapter is the pathology test. Specifically the study investigated the use of the two most commonly ordered pathology test types in Australian general practice as previously identified in Chapter 4: the FBC and the lipid test.

The FBC test (also known as a 'complete blood count') and lipid tests are 'low cost, high volume' pathology tests. Despite the high volume at which they are ordered,³⁷ they are not the most commonly funded MBS pathology items.³⁴⁵ This is due to the effect of the MBS payment structure: the iso-resource item numbers, and the coning rule which limits MBS funding to the three most expensive MBS items from each episode of GP-ordered pathology tests (as described in Chapter 2).

The extent of 'coned out' or missing MBS data was investigated in 2010, using data from pathology laboratories. Of the FBCs ordered by GPs only 45–55% were charged to the MBS, the remainder being 'coned out'. For lipid tests, HDL was the only lipid component that could be investigated (because it has its own MBS item number) and just 20–30% of these were charged to the MBS.^{19,346} It was not possible to investigate the effect of coning on the remaining lipid components because these are in MBS items that are structured as iso-resource chemistry groups, and the specific tests ordered cannot be identified. This study demonstrated that MBS data do not reflect GPs actual pathology ordering patterns for FBC and lipid tests.

GPs' rate of FBC and lipid test ordering increased significantly between 2000–02 and 2006–08 (see Chapter 4). These tests were among the most commonly ordered tests for the six problems investigated in Chapter 5, although the extent to which their use was supported varied. While lipid tests were generally supported for the investigated problems, this was not the case for FBC tests.

The purpose of this chapter is to investigate issues raised in the literature review and during research undertaken in previous chapters that could not be investigated using the cross-sectional BEACH encounter data. These issues and the objectives for this study are summarised below.

Who initiates the tests

The literature suggests that general practice patients frequently wish to have their blood tested.³³⁻³⁵ Not all patients communicate a request for testing,³⁴⁷ but when they do, GPs have described this as a contributing factor in their decision to order pathology tests.^{36,37} However, few studies have quantified the proportion of tests perceived by GPs to be generated by patient request. A SAND substudy conducted in 2002, found that 15.1% of pathology tests ordered at general practice encounters were initiated by patient requests for testing.³⁴⁸

Purpose of tests

While the clinical problem for which a pathology test was ordered was recorded in the BEACH data set, the specific clinical purpose of the test (whether for monitoring, diagnostic or preventive purposes) was not requested. However, as discussed in Chapter 5, this has a bearing on the appropriateness of ordering specific tests and the interpretation of the result.

Interval between repeat testing

The cross-sectional nature of the BEACH encounter data meant that the interval between testing was not known. The review of guidance documents undertaken in Chapter 5 revealed that recommendations on the ideal interval between tests were often lacking, particularly when tests were used for monitoring purposes and, the majority of lipid and FBC tests were ordered by GPs for the long-term management of the chronic conditions. For monitoring of lipid levels, there has also been debate as to whether the commonly recommended monitoring interval of 6–12 months should be lengthened or whether there is a need for monitoring at all.^{224-226,349}

Status of FBC testing

The clinical purpose for GP ordering of FBCs was unclear for most of the six problems investigated in Chapter 5. They appeared to be ordered opportunistically or routinely when GPs ordered pathology tests in the management of the problems investigated. Others have described clinician's use of the FBC as 'indiscriminate', 'reflexive' or 'routine' in various inpatient and outpatient health settings.³⁵⁰⁻³⁵⁴ Even though the FBC was the most commonly ordered pathology test in general practice, it has not been the subject of much 'independent' discussion in the literature. In contrast, there is a wealth of literature about lipid testing in primary care.^{83,226-228,355,356}

The use of FBC tests has been discussed in the outpatient³⁵²⁻³⁵⁴ (including the emergency department^{357,358}) and inpatient health settings.⁶⁷ Outcomes of these studies suggest that FBC tests either were not supported in the specific setting studied^{67,352,353} or that improved appropriateness of ordering would reduce the number of FBCs.^{357,358} Only one study recommended the routine use of FBCs in ambulatory care because they 'contributed to the wellbeing of the population'.³⁵⁴ I could find only one small study discussing use of FBCs in general practice and it was conducted more than 20 years ago.³⁵⁰ This study introduced an intervention to improve appropriateness of ordering that reduced use of FBCs and improved appropriateness of their use.³⁵⁰ Other studies in general practice would have included evaluation of GPs' use of FBC tests in the context of a broader intervention, ^{76,82,86,88,281} but these publications did not report information specifically about FBC ordering.

While GPs' use of FBC tests is rarely discussed in the literature, there is no shortage of articles recommending GPs use the test, most frequently for diagnostic purposes for a range of problems (such as, fatigue,^{276,282,284} unexplained complaints,^{281,359} nausea and vomiting³⁶⁰) and for the diagnosis and management of haematological diseases (such as anaemia^{173,361}).

Medication status and morbidity profile of patients with lipid tests

The cardiovascular risk associated with lipid levels means that testing lipids and management of lipid levels is recommended for prevention and secondary prevention of cardiovascular disease in the management of related chronic diseases (see Chapter 5). The BEACH encounter data set provides a snapshot of the characteristics of patients and the details of the clinical problems for which lipid tests are ordered. However, this snapshot does not provide a complete profile of the morbidities that the patient has nor information about whether the patient currently takes lipidlowering medication. Both of these factors may influence the decision to order lipid tests. In this chapter I aim to fill the gaps in the available information highlighted above.

6.2 Objectives

The objectives of this study were:

- to measure the extent to which FBC and lipid tests are initiated by patients
- to investigate the purpose of ordering FBC and lipid tests for all problems, and specifically for the six problems investigated in Chapter 5
- to describe the interval between repeated FBC and lipid tests, and specifically the interval between these tests when used for monitoring purposes.

Additional objectives that are unique to either FBC or lipid tests are:

- to describe the differential diagnosis or diagnoses investigated by GPs when FBC tests were ordered for diagnostic purposes
- to investigate the morbidity profile, and lipid lowering medication use among patients for whom lipid tests were ordered.

6.3 Method

This study used the SAND method as described in Section 3.2.6. Ethics approval for this SAND study was obtained from the Ethics Committee of the Australian Institute of Health and Welfare, and the Human Research Ethics Committee of the University of Sydney. Data in this study were collected over two periods: July–August 2008 (Sample A), and September–October 2009 (Sample B).

At each encounter where the GP had ordered a FBC or lipid test they were asked a series of questions. I designed these questions to elicit information for each test on: who initiated the test; the clinical purpose of the test; whether the patient had been tested in the past, and (where applicable) the time since previous testing. Box 6.1 shows the SAND form designed for this study. These questions were attached to the bottom of the BEACH encounter form (see Appendix 2) and the SAND data collected can be linked to data (such as patient age and sex) recorded in the

encounter form. Additionally the linkages between the FBC and lipid test and the problem(s) for which they were ordered were drawn from the encounter form.

The 'purpose of test' categories were based on a previous study conducted in 2002 that defined tests as investigative, monitoring and/or preventive.³⁴⁸ The preventive category was further broken down into tests for primary and secondary prevention. The purpose of tests were defined as:

- investigative or diagnostic for a new condition
- monitoring of an existing diagnosed condition
- primary prevention, i.e. screening in an otherwise healthy patient (e.g. following RACGP red book guidelines¹⁷⁷ for preventive activities)
- secondary prevention, i.e. screening in a patient with established risk factors for a condition but without the condition (Box 6.1).

GPs were also asked whether ordering the FBC or lipid test was 'opportunistic', i.e. the test was added to the pathology order once the decision to order another test had already been made.

In circumstances where FBC tests were ordered for investigative purposes, GPs were asked to specify the differential diagnosis or diagnoses for which the FBC was ordered. Selecting a differential diagnosis was defined as 'the process of weighing the probability of one disease versus that of other diseases possibly accounting for a patient's illness'.³⁶² These differential diagnoses were recorded by the GPs as free text and coded using ICPC-2 PLUS.¹¹³

When lipid tests were ordered, GPs were asked whether the patient was currently taking lipid lowering medication, and to indicate which conditions the patient had from a short list of conditions or risk factors (tick boxes). The listed conditions/risk factors were: dyslipidaemia, family history of dyslipidaemia, obesity, diabetes, hypertension and 'other cardiovascular disease' (Box 6.1).

ox 6.1: Exampl	e of SAND form c	ompleted by participating	g GPs (with	inst	tructions)				
PLEASE R The shaded sec You may tear	EAD CAREFU tion of the following out this page as a	LLY forms asks questions about guide to completing the fo	SELECTEI	D P/ tion	ATHOLOGY TEST of forms.	S.			
INSTRUCTIO Ask ALL of the p in which the p Please DO NOT	NS next 30 PATIENTS the f atients are seen. select patients to suit t	ollowing questions in the order ne topic being investigated.	Please ens encounter in the shac	ure in th led s	you complete details ne main section of th section below.	s of all ne form	pathology te before comp nvestigative F	sts ordered at bleting the que BC test	today's stions
 Selected pathology tests Please advise whether you ordered a lipid test and/or a full blood count (FBC) for this patient <u>at this encounter</u>. Lipid test includes total cholesterol, HDL, LDL and triglycerides. If you did not order a lipid test or full blood count at this encounter please tick the box labelled 'neither of the above' and end the questions here for this patient. Initiation of tests Under each heading (lipid or FBC) please use the tick boxes to indicate who suggested each test—you, the patient or another heatth professional (e.g. specialist). Please tick only one option per test using the tick boxes in the appropriate column 'Lipid' or 'FBC'. 			Previous testing If a FBC was ordered today for inverse or inverse or indicate whether this patient has had a lipid/FBC test on any previous occasion. If a FBC was ordered today for inverse or inverse or indicate whether this patient has had a lipid/FBC test on any previous occasion. If you do not know if the patient has previously been tested please tick the box labelled 'don't know'. If a FBC was ordered today for inverse or inverse o				vestigative purposes losis(es) in the space ferential diagnoses, least probable. gnosis please write s 'the process of ase versus that of for a patient's illness'. 508		
			nt (e.g. < guidelines screening in risk factors the condition was added e the est was ach test.		Time since last test If 'yes' to either test ple advise how long it has been since the patien last had a lipid/FBC to If you do not know whe the last test was ordered please tick the box labelled 'don't know'.	st ease s nt test. en ed		Lipid medie other cond For patients w test ordered indicate wheth indicate wheth is current lowering has any of condition	cation and itions who have a <u>lipid</u> today please her the patient: ly taking a lipid medication f the listed s or risk factors.
↓ 	<u> </u>			\downarrow	<u> </u>	For FD(↓ 		
At today's encounter did you order a: □ Lipid test □ Full blood count □ Neither of the above → End questions BL116B	Who suggested the test(s)? (Please tick one option per test) Lipid FBC GP	Ioday's test(s) is/are: (Incentified and that apple Lipid FBC Investigative Image: Comparison of the comparison of	Has this patie had this test before? Lipid Yes No Don't know	FBC	The test was last ordered: Lipid FBC <3 mnth ago	If <u>inver</u> is/are y diagno	<u>stigative</u> ' what rour differential sis(es)?	Is this patient currently taking lipid lowering medication? Ves No Don't know	 Dyslipidaemia Dyslipidaemia Family Hx dyslipidaemia Obesity Diabetes Hypertension Other CVD None of the above

6.4 Results

6.4.1 The sample

Table 6.1 shows the age and sex distributions of patients in: each of the two samples (A and B) included in this study; the combined sample (A + B); and, all encounters recorded in BEACH in 2008–09.¹²²

There was no statistically significant difference in the patient age and sex distributions in samples A and B did not differ significantly, indicating that the samples were homogenous (Table 6.1). Therefore, the samples were combined for the remaining analysis. In the combined sample, 193 GPs recorded details of FBC and lipid testing for 5,629 patients who visited them when participating in this study. The patients in this study were a subsample of all patients in the BEACH study. The age-sex distribution of the combined SAND sample was compared with that of all patients in the BEACH study (Table 6.1). There was no difference in the sex distribution, but there were significant differences in the age distribution. Significantly more patients were aged 1–4 years, and fewer were aged 75 years and over, in the SAND sample than in the total BEACH sample (Table 6.1).

	Sample A	Sample B	Combined sample	BEACH 2008-09 ¹²²
Variable	Per cent ^(a) (95% Cl) (<i>n</i> = 2,940)	Per cent ^(a) (95% Cl) (<i>n</i> = 2,689)	Per cent ^(a) (95% Cl) (<i>n</i> = 5,629)	Per cent ^(a) (95% Cl) (<i>n</i> = 101,000)
Sex (missing <i>n</i>)	(14)	(16)	(30)	(781)
Male	41.8 (38.6–45.0)	41.2 (38.5–43.9)	41.5 (39.4–43.6)	41.5 (40.7–42.3)
Female	58.2 (55.0–61.4)	58.8 (56.1–61.5)	58.5 (56.4–60.6)	58.5 (57.7–59.4)
Age (missing <i>n</i>)	(19)	(15)	(34)	(626)
< 1 years	2.1 (1.6–2.7)	2.6 (1.8–3.4)	2.3 (1.9–2.8)	1.9 (1.8–2.0)
1–4 years	4.8 (3.7–5.8)	6.4 (5.1–7.8)	5.6 (4.7–6.4)	4.0 (3.8–4.2)
5-14 years	5.8 (4.6–7.0)	4.8 (3.9–5.7)	5.3 (4.6–6.1)	5.1 (4.8–5.3)
15–24 years	8.0 (6.7–9.4)	9.6 (8.1–11.0)	8.8 (7.8–9.8)	8.1 (7.7–8.4)
25–44 years	22.2 (19.9–24.5)	23.8 (21.0–26.6)	23.0 (21.2–24.7)	20.7 (20.1–21.3)
45-64 years	29.0 (27.0–30.9)	28.0 (25.6–30.5)	28.5 (27.0–30.1)	29.1 (28.5–29.6)
65–74 years	13.3 (11.5–15.1)	11.1 (9.2–13.0)	12.3 (11.0–13.6)	13.8 (13.4–14.2)
75+ years	14.8 (12.4–17.1)	13.6 (10.8–16.5)	14.2 (12.4–16.1)	17.5 (16.7–18.3)

Table 6.1: Characteristics of	f patients in the	e FBC and lipid	study sample
-------------------------------	-------------------	-----------------	--------------

(a) Missing data removed.

Note: FBC - full blood count; CI - confidence interval.

6.4.2 FBC testing

GPs ordered a FBC test for 9.4% (95% CI: 8.4–10.5) of sampled patients. This equates to an estimated 10.6 million FBC tests ordered by GPs at the 113 million MBS paid GP encounters that occurred in Australia in 2008–09. There was no statistical difference in the order rate of FBC tests by patient sex: 9.6% (95% CI: 8.1–11.0) of males and 9.3% (95% CI: 8.0–10.5) of female patients (results not tabled).

The age-specific rate of FBC testing is shown in Figure 6.1. The likelihood of GPs ordering a FBC was constant across age groups for patients aged 15 years and over, approximately 10% of patients in each adult age group receiving an order for a FBC test. Ordering FBC tests for children aged less than 15 years was uncommon (Figure 6.1).



Test initiation

Of the 530 patients for whom GPs ordered a FBC test, responses to who initiated or suggested the test were recorded for 486. In most cases the GP initiated the FBC test (92.2%), but the patient initiated 6.0% and an 'other health professional' 1.9% (Table 6.2).

Purpose of test

It was possible to record multiple reasons for ordering FBC tests and at least one reason was recorded for 500 patients. In most cases (94.4% of patients) GPs listed only one reason. Two reasons were recorded for 4.8% of patients, and three reasons for 0.8% of patients (results not tabled).

An investigative or diagnostic purpose was the most common reason for ordering FBC tests, indicated for 48.0% of patients for whom a FBC was ordered. Monitoring was a recorded reason for 35.0% of patients, opportunistic testing (i.e. adding the test to a pathology order once the decision to order was already made) for 10.6%, primary prevention (i.e. screening in an otherwise healthy patient) for 8.0%, secondary prevention (i.e. screening in a patient with known risk factors) for 4.8% (Table 6.2).

Previous testing and interval between repeat testing

Most patients with a FBC test ordered at the recorded encounter had previously had a FBC test (80.6%), 12.4% were tested for the first time at the recorded encounter, and for 7.0% of patients previous testing status was not known (Table 6.2).

The time since the last test was ordered was recorded for 396 of the 404 patients who had a previous FBC test. The last test was ordered > 12 months ago for 39.7% of patients, 7–12 months ago for 22.2%, 3–6 months ago for 19.7%, < 3 months ago for 16.4%, and was not known for 2.0% of patients (Table 6.2).

Variable	Number	Per cent ^(a) (<i>n</i> = 530)	95% LCL	95% UCL
Initiation of testing (missing)	(44)			
GP	448	92.2	89.8	94.6
Patient	29	6.0	3.9	8.0
Other health professional	9	1.9	0.7	3.0
Purpose of ordering ^(b) (missing)	(30)			
Investigative	240	48.0	42.8	53.2
Monitoring	175	35.0	29.4	40.6
Primary prevention	40	8.0	5.3	10.7
Secondary prevention	24	4.8	2.6	7.0
Opportunistic	53	10.6	6.7	14.5
Prior testing status (missing)	(29)			
Previously tested	404	80.6	76.6	84.7
Never tested	62	12.4	8.9	15.8
Unknown	35	7.0	4.6	9.3
Time since previous test ^(c) (missing)	(8)			
< 3 months	65	16.4	12.0	20.9
3–6 months	78	19.7	14.8	24.6
7–12 months	88	22.2	18.1	26.4
> 12 months	157	39.7	34.2	45.1
Unknown	8	2.0	0.5	3.5

Table 6.2: FBC testing in general practice patients

(a) Missing data removed.

(b) Multiple responses allowed.

(c) Proportion of patients who had been tested previously.

Note: FBC - full blood count; LCL - lower confidence limit; UCL - upper confidence limit.

The interval since the previous FBC test was recorded for 170 of the 175 patients who had a FBC ordered for monitoring purposes. Almost half (48.3%) of those patients with a monitoring FBC had been tested in the previous six months. When compared with FBC tests ordered for non-monitoring purposes, monitoring FBC tests were significantly more likely to have been ordered within the previous 12 months (Table 6.3).

	Мо	nitoring FBC	test	FBC test for other purposes				
Time since previous test	Number	Per cent ^(a) (<i>n</i> = 170)	95% LCL	95% UCL	Number	Per cent ^(a) (<i>n</i> = 218)	95% LCL	95% UCL
< 3 months	37	21.8	14.8	28.8	25	11.5	6.3	16.6
3–6 months	45	26.5	18.6	34.4	33	15.1	9.5	20.8
7–12 months	47	27.6	20.4	34.9	40	18.3	13.4	23.3
> 12 months	41	24.1	17.4	30.8	112	51.4	44.0	58.8
Unknown	0				8	3.7	1.0	6.4

 Table 6.3: Time since previous FBC test when ordered for monitoring versus other purposes

(a) Missing data removed. Of the 404 patients with a previous FBC test data were missing for 16 patients. *Note:* FBC – full blood count; LCL – lower confidence limit; UCL – upper confidence limit.

Problems for which FBC tests were ordered

The SAND data were linked to the data recorded on the encounter form to investigate the problems for which GPs ordered FBC tests. The problem or problems for which the test was ordered were linked for 466 of the 530 patients for whom a FBC was ordered. There were 64 patients for whom the FBC test was not linked to a problem on the encounter form. Where the FBC test and problem(s) was linked, GPs indicated that the 466 tests were ordered in the management of 492 problems (Table 6.4).

There was a broad range of problems for which FBC tests were ordered by GPs. The most common problems were hypertension (8.5%), weakness/tiredness (6.1%), health checks (4.9%), Type 2 diabetes (4.9%), pregnancy (3.5%), problems described as blood tests (3.0%), and lipid disorders (2.8%) (Table 6.4). The 23 problems listed in Table 6.4 accounted for 57% of all problems linked to FBC and the three most common problems accounted for less than 20%.
		Per cent of problem–FBC test links	95%	95%
Problem managed	Number	(<i>n</i> = 492)	LCL	UCL
Hypertension (non-gestational)*	42	8.5	6.1	11.0
Weakness/tiredness	30	6.1	3.7	8.5
Health check (15+ years)*	24	4.9	2.8	6.9
Type 2 diabetes	24	4.9	2.8	6.9
Pregnancy*	17	3.5	1.8	5.1
Blood test – all*	15	3.0	1.2	4.9
Lipid disorders	14	2.8	1.4	4.3
Anaemia*	13	2.6	1.3	4.0
Vitamin/nutritional deficiency	10	2.0	0.7	3.4
Menstrual problems*	9	1.8	0.5	3.1
Anxiety*	8	1.6	0.4	2.9
HIV infection/AIDS	7	1.4	0.0	4.2
Abdominal pain*	7	1.4	0.4	2.5
Depression*	7	1.4	0.3	2.6
Rheumatoid arthritis	7	1.4	0.4	2.5
Upper respiratory tract infection	7	1.4	0.1	2.7
Abnormal test results*	6	1.2	0.3	2.2
Ischaemic heart disease*	6	1.2	0.3	2.2
Cough	6	1.2	0.1	2.3
Urinary disease, other	6	1.2	0.3	2.2
Back complaint*	5	1.0	0.2	1.9
Vertigo/dizziness	5	1.0	0.1	1.9
Acute bronchitis/bronchiolitis	5	1.0	0.1	1.9
Subtotal	280	56.9		
Total	492	100.0		

Table 6.4: Most common problems for which GPs ordered a FBC test

* Includes multiple ICPC-2 PLUS codes (see Appendix 6).

Note: Only problems accounting for > 1% of problem–FBC test links. FBC – full blood count; LCL – lower confidence limit; UCL – upper confidence limit; HIV – human immunodeficiency virus; AIDS – acquired immune deficiency syndrome.

Differential diagnoses of investigative FBC tests

When FBC tests were ordered for investigative purposes, GPs were asked to describe the differential diagnosis/diagnoses. Of the 240 FBC test ordered for investigative purposes, GPs recorded differential diagnoses for 230 patients. More than one differential diagnosis could be recorded per patient and a total of 349 diagnoses were recorded: 60% of patients had one differential diagnosis recorded, 28.3% had two and 11.7% had three recorded (results not tabled).

There was a wide range of problems listed as differential diagnoses. The most common were:

- anaemia (29.2% of diagnoses)
- infectious disease (8.0%), which was primarily expressed by GPs as 'infection'
- vitamin/nutritional deficiency (4.0%), primarily expressed as 'iron deficiency'
- viral disease (3.7%), primarily expressed as 'viral illness'
- infectious mononucleosis (2.9%)
- other disease of the digestive system, recorded by GPs as coeliac disease, colitis or pancreatitis (2.3%) (Table 6.5).

Table 6.5: Most common differential diagnoses for investigative FBC tests

Differential diagnoses	Number	Per cent (<i>n</i> = 349)
Anaemia*	102	29.2
Infectious disease, other/NOS	28	8.0
Vitamin/nutritional deficiency	14	4.0
Viral disease, other/NOS	13	3.7
Infectious mononucleosis	10	2.9
Other disease of the digestive system	8	2.3
Pneumonia	7	2.0
Purpura/coagulation defect	7	2.0
Abnormal test results*	6	1.7
Acute bronchitis/bronchiolitis	6	1.7
Subtotal	201	57.6
Total	349	100.0

* Includes multiple ICPC-2 PLUS codes (see Appendix 6).

Note: FBC - full blood count; NOS - not otherwise specified.

Purpose of FBC test when ordered for selected problems

Table 6.6 reports the purpose of FBC tests ordered for each of the six problems investigated in Chapter 5. It was possible for multiple purposes to be recorded for each problem.

The majority of FBCs ordered for hypertension, Type 2 diabetes and lipid disorders were ordered for monitoring (Table 6.6). These three chronic conditions accounted for almost one-third (31.4%) of the monitoring FBCs ordered (results not tabled). There were insufficient numbers of cases where GPs ordered FBCs for overweight/obesity to draw conclusions about the purpose of testing (Table 6.6). FBCs for weakness/tiredness were almost exclusively ordered for investigative purposes (Table 6.6). It was the clinical problem for which GPs most often ordered investigative FBCs (11.7% of investigative FBC tests) (results not tabled). For the 28 patients with an investigative FBC ordered for weakness/tiredness, GPs listed 45 differential diagnoses for weakness/tiredness (12.9% of all differential diagnoses). The most common were anaemia (n = 22), infectious diseases (n = 6), and vitamin/nutritional deficiency (n = 4) (results not tabled).

When FBCs were ordered as part of the health check, the three main purposes were monitoring (27.3%), primary prevention (27.3%) and investigative (22.7%). There was no primary purpose for which GPs ordered FBCs for health checks (Table 6.6).

Problem managed ^(a)	Purpose of FBC	Number	Per cent of problem-specific FBC test links	95% LCL	95% UCL
Hypertension (non-gestational)*	Investigative	7	17.1	4.5	29.6
	Monitoring	24	58.5	42.4	74.6
	Primary prevention	5	12.2	0.0	24.4
	Secondary prevention	4	9.8	0.3	19.2
	Opportunistic	6	14.6	3.6	25.7
Weakness/tiredness	Investigative	28	96.6	89.3	100.0
	Monitoring	1	3.4	0.0	10.7
	Primary prevention	0			
	Secondary prevention	0			
	Opportunistic	0			
Type 2 diabetes	Investigative	2	8.3	0.0	20.5
	Monitoring	20	83.3	67.5	99.2
	Primary prevention	1	4.2	0.0	12.9
	Secondary prevention	4	16.7	0.0	36.2
	Opportunistic	1	4.2	0.0	12.9
Health check (15+ years)*	Investigative	5	22.7	3.0	42.4
	Monitoring	6	27.3	6.2	48.3
	Primary prevention	6	27.3	7.5	47.1
	Secondary prevention	2	9.1	0.0	21.7
	Opportunistic	3	13.6	0.0	27.8
Lipid disorders	Investigative	0			
	Monitoring	11	91.7	72.9	100.0
	Primary prevention	0			
	Secondary prevention	0			
	Opportunistic	1	8.3	0.0	27.1
Overweight/obesity (adults)*	Investigative	2	66.7	0.0	100.0
	Monitoring	1	33.3	0.0	100.0
	Primary prevention	0			
	Secondary prevention	0			
	Opportunistic	0			

Table 6.6: Purpose of ordering FBC tests for selected problems

(a) Multiple purposes for ordering FBC tests could be ordered per problem. There were cases where the purpose of the FBC was missing for the problem. The purpose of FBC was known for: 41 hypertension problems (missing n = 1); 29 weakness/tiredness problems (missing n = 1); 24 Type 2 diabetes problems (missing n = 0); 22 health check problems (missing n = 2); 12 lipid disorder problems (missing n = 2); 3 overweight/obesity problems (missing n = 0).

* Includes multiple ICPC-2 PLUS codes (see Appendix 6).

Note: FBC - full blood count; LCL - lower confidence limit; UCL - upper confidence limit.

6.4.3 Lipid testing

GPs ordered a lipid test for 8.3% (95% CI: 7.3–9.2) of sampled patients. This equates to an estimated 9.4 million lipid tests ordered at GP encounters in Australia in 2008–09. GPs were more likely to order lipid tests for male patients (9.9%, 95% CI: 8.4–11.4) than for female patients (7.0%, 95% CI: 5.9–8.1) (results not tabled). The age-specific rate of lipid testing is shown in Figure 6.2. Lipid tests were most likely to be ordered for patients aged 45–64 years (14.2%), followed by patients aged 65–74 years (10.5%), and those aged 75 years and over (8.2%). GPs rarely ordered lipid tests for patients aged less than 25 years of age (Figure 6.2).



Test initiation

Of the 465 patients for whom GPs ordered a lipid test, the person who initiated or suggested the test was recorded for 456 patients. In most cases the GP initiated the lipid test (87.1%), followed by the patient (11.2%), and an 'other health professional' (1.8%) (Table 6.7).

Purpose of test

The purpose(s) of the test was provided for 455 patients. It was possible to record multiple reasons for ordering lipid tests. However, in most cases (83.7% of patients) GPs recorded only one reason for ordering lipid tests. Two reasons were recorded for 13.4% of patients, and three reasons for 2.9% of patients.

The most common purpose was monitoring (58.5% of patients). Prevention was recorded as the purpose for 35.8% of patients: 18.0% being for primary prevention (i.e. screening in an otherwise healthy patient), and 17.8% for secondary prevention (i.e. screening in a patient with known risk factors). An investigative or diagnostic purpose was recorded for 18.2% of patients, and the lipid test was opportunistic (i.e. adding the test to a pathology order once the decision to order was already made) for 7.3% (Table 6.7).

The most common combinations of indications were: monitoring and secondary prevention (5.9%), monitoring and primary prevention (2.9%), and investigative and primary prevention (1.3%) (results not tabled).

Previous testing and interval between repeat testing

Most patients with a lipid test ordered at the recorded encounter had previously had their lipid levels assessed (86.0% of 458 respondents), 10.5% were tested for the first time today, and for 3.5%, previous testing status was not known (Table 6.7).

Of the 394 patients who had previously had their lipid levels assessed, the time since the last test was ordered was recorded for 391 patients. The last test was ordered > 12 months ago for 42.2% of patients, 7–12 months ago for 26.6% of patients,

3-6 months ago for 22.0%, < 3 months ago for 7.4%, and was not known for 1.8% of patients (Table 6.7).

Table 6.8 shows the interval since the previous lipid test, recorded for 259 of the 266 patients who had a lipid test ordered for monitoring purposes. The last test was ordered more than 12 months ago for 31.3% of patients, 7–12 months ago for 30.1% of patients, 3–6 months ago for 29.3%, and < 3 months ago for 8.5% of patients. When compared with lipid tests ordered for non-monitoring purposes, monitoring lipid tests were significantly more likely to have been ordered within the previous 12 months (Table 6.8).

Medication status and morbidity profile

Current lipid lowering medication status was specified for 447 patients for whom lipid tests were ordered and almost 40% were currently taking a lipid lowering medication (Table 6.7). Of the 174 patients taking a lipid lowering medication, 80.4% had dyslipidaemia (with or without any of the other listed conditions). The remaining 19.6% of patients had at least one of the other listed cardiovascular diseases or risk factors (results not tabled).

Details on the presence/absence of the listed morbidities and risk factors were provided for 452 patients: 43.8% had dyslipidaemia, 43.1% had hypertension, 25.7% were obese, 18.8% had diabetes, 18.8% had a family history of dyslipidaemia, and 12.2% had an 'other cardiovascular disease' (Table 6.7).

At least one of these morbidities/risk factors was present in 84.3% (95% CI: 80.7–87.9) of patients who had a lipid test ordered and multiple morbidities/risk factors were common (44.5% of patients with lipid tests). Of the 381 patients with at least one of the listed morbidities/risk factors, 47.2% had only one, 24.9% had two, 18.6% had three morbidities and 9.2% had four or more. The most common morbidities/risk factors and combinations were: dyslipidaemia alone (13.6%, 95% CI: 9.6–17.7 of patients with morbidities/risk factors), hypertension alone (10.0%, 95% CI: 6.7–13.3), family history of dyslipidaemia alone (8.9%, 95% CI: 5.9–12.0), dyslipidaemia + hypertension (7.3%, 95% CI: 4.4–10.3), obesity only (7.1%, 95% CI: 4.5–9.7), dyslipidaemia + diabetes + hypertension (5.0%, 95% CI: 2.8–7.2), and diabetes alone (4.7%, 95% CI: 2.2–7.1) (results not tabled).

Of the 198 patients with dyslipidaemia who had a lipid test ordered, 73.7% (95% CI: 66.5–81.0) had at least one of the other listed morbidities/risk factors, and 71.4% were taking a lipid lowering medication. Of patients with dyslipidaemia: 55.1% (95% CI: 47.0–63.1) had hypertension; 28.3% (95% CI: 20.9–35.6) obesity; 24.2% (95% CI: 17.7–30.8) diabetes; 17.7% (95% CI: 10.6–24.8) family history of dyslipidaemia; 16.2% (95% CI: 11.3–21.0) other cardiovascular disease, and 26.3% had none of the other listed morbidities/risk factors (results not tabled).

There were 183 patients (40.5% of those with a lipid test ordered at the current encounter) with at least one of the listed morbidities or risk factors who did not have dyslipidaemia. Of these patients: 47.0% (95% CI: 39.0–55.0) had hypertension;

32.8% (95% CI: 24.6–41.0) obesity; 27.3% (95% CI: 19.6–35.0) family history of dyslipidaemia; 20.2% (95% CI: 13.9–26.6) diabetes; and 12.6% (95% CI: 7.5–17.6) had an other cardiovascular disease (results not tabled).

Variable	Number	Per cent ^(a) (<i>n</i> = 465)	95% LCL	95% UCL
Initiation of testing (missing)	(9)			
GP	397	87.1	83.6	90.6
Patient	51	11.2	7.9	14.4
Other health professional	8	1.8	0.6	3.0
Purpose of ordering ^(b) (missing)	(10)			
Monitoring	266	58.5	53.5	63.5
Investigative	83	18.2	13.9	22.6
Primary prevention	82	18.0	13.5	22.5
Secondary prevention	81	17.8	13.0	22.6
Opportunistic	33	7.3	4.3	10.2
Prior testing status (missing)	(7)			
Previously tested	394	86.0	82.6	89.4
Never tested	48	10.5	7.4	13.6
Unknown	16	3.5	1.9	5.1
Time since previous test ^(c) (missing)	(3)			
< 3 months	29	7.4	4.4	10.5
3–6 months	86	22.0	17.1	26.9
7–12 months	104	26.6	21.9	31.3
> 12 months	165	42.2	36.6	47.8
Unknown	7	1.8	0.5	3.1
Medication status (missing)	(18)			
Taking lipid lowering medication	174	38.9	33.9	43.9
Presence of associated morbidities ^(b) (missing)	(13)			
Dyslipidaemia	198	43.8	38.9	48.8
Hypertension	195	43.1	37.6	48.7
Obesity	116	25.7	21.0	30.4
Diabetes	85	18.8	15.0	22.6
Family history of dyslipidaemia	85	18.8	14.0	23.6
Other cardiovascular disease	55	12.2	9.0	15.3

Table 6.7: Lipid testing in general practice patients

(a) Missing data removed.

(b) Multiple responses allowed.

(c) Proportion of patients who had been tested previously.

Note: LCL - lower confidence limit; UCL - upper confidence limit.

	Monitoring lipid test				Lipid to	est for other	purpos	ses
Time since previous test	Number	Per cent ^(a) (<i>n</i> = 259)	95% LCL	95% UCL	Number	Per cent ^(a) (<i>n</i> = 128)	95% LCL	95% UCL
< 3 months	22	8.5	4.7	12.3	6	4.7	1.0	8.3
3–6 months	76	29.3	22.9	35.8	9	7.0	2.4	11.7
7–12 months	78	30.1	24.1	36.2	26	20.3	12.8	27.8
> 12 months	81	31.3	24.8	37.7	82	64.1	54.9	73.2
Unknown	2	0.8	0.0	1.8	5	3.9	0.6	7.3

 Table 6.8: Time since previous lipid test when ordered for monitoring versus other purposes

(a) Missing data removed. Of the 394 patients with a previous lipid test data were missing for 7 patients. *Note:* LCL – lower confidence limit; UCL – upper confidence limit.

Problems for which lipid tests were ordered

The SAND data were linked to the data recorded on the encounter form to investigate the problems for which GPs ordered lipid tests. Of the 465 patients for whom a lipid test was ordered, the problem or problems for which the test was ordered were linked for 401 patients. There were 64 patients where the lipid test was not linked to a problem on the encounter form. Where the lipid test and problem(s) was linked, GPs indicated that the 401 tests were ordered in the management of 427 problems (Table 6.9).

The most common problems for which GPs ordered lipid tests were lipid disorders (20.8%), hypertension (15.9%), Type 2 diabetes (11.9%), health checks (6.9%) and problems labelled as blood tests (5.9%). Approximately half of all lipid test-problem linkages were accounted for in the three most common problems: lipid disorders, hypertension and diabetes (Table 6.9).

		Per cent of	95%	95%
Problem managed	Number	(<i>n</i> = 427)	LCL	UCL
Lipid disorders	89	20.8	16.7	25.0
Hypertension (non-gestational)*	68	15.9	12.2	19.7
Type 2 diabetes	51	11.9	8.6	15.3
Health check (15+ years)*	29	6.9	4.1	9.7
Blood test – all*	25	5.9	3.1	8.6
Ischaemic heart disease*	13	3.0	1.4	4.7
Weakness/tiredness	11	2.6	0.8	4.4
Overweight/obesity (adults)*	11	2.6	1.0	4.2
Risk factor NOS	8	1.9	0.6	3.1
Prescription – all*	7	1.6	0.4	2.8
Abnormal test results*	7	1.6	0.4	2.8
Cardiovascular check-up*	6	1.4	0.3	2.5
HIV infection/AIDS	5	1.2	0.0	3.5
Subtotal	330	77.3		
Total	427	100.0		

Table 6.9: Most common problems managed with a lipid test

Includes multiple ICPC-2 PLUS codes (see Appendix 6).

Note: Only problems accounting for > 1% of problem–lipid test links. LCL – lower confidence limit; UCL – upper confidence limit; NOS – not otherwise specified; HIV – human immunodeficiency virus; AIDS – acquired immune deficiency syndrome.

Purpose of lipid test for selected problems

Table 6.10 reports the purpose of lipid tests ordered for the six problems investigated in Chapter 5. It was possible for multiple purposes to be recorded for each problem. The majority of the lipid tests ordered for hypertension, Type 2 diabetes and lipid disorders were ordered for monitoring (Table 6.10). These three chronic conditions accounted for 60.5% of the monitoring lipids ordered (results not tabled).

Secondary prevention was also a common reason for ordering lipid tests for hypertension, Type 2 diabetes and overweight/obesity. When lipid tests were ordered as part of the health check, the main purposes were primary prevention (42.9%), and monitoring (35.7%) (Table 6.10).

Problem managed ^(a)	Purpose of lipid test	Number	Per cent of problem-specific lipid test links	95% LCL	95% UCL
Lipid disorders	Monitoring	78	87.6	79.1	96.2
	Investigative	3	3.4	0.0	7.3
	Primary prevention	10	11.2	3.2	19.3
	Secondary prevention	12	13.5	4.8	22.2
	Opportunistic	3	3.4	0.0	7.3
Hypertension (non-gestational)*	Monitoring	40	60.6	47.4	73.8
	Investigative	9	13.6	2.8	24.5
	Primary prevention	7	10.6	2.0	19.2
	Secondary prevention	17	25.8	14.1	37.4
	Opportunistic	2	3.0	0.0	7.3
Type 2 diabetes	Monitoring	43	89.6	80.5	98.6
	Investigative	4	8.3	0.4	16.3
	Primary prevention	3	6.3	0.0	15.5
	Secondary prevention	15	31.3	15.3	47.2
	Opportunistic	0			
Health check (15+ years)*	Monitoring	10	35.7	16.8	54.6
	Investigative	5	17.9	2.0	33.7
	Primary prevention	12	42.9	20.6	65.1
	Secondary prevention	2	7.1	0.0	17.2
	Opportunistic	2	7.1	0.0	16.8
Weakness/tiredness	Monitoring	4	36.4	1.7	71.0
	Investigative	6	54.5	20.6	88.5
	Primary prevention	0			
	Secondary prevention	0			
	Opportunistic	1	9.1	0.0	30.9
Overweight/obesity (adults)*	Monitoring	3	27.3	0.0	60.4
	Investigative	4	36.4	0.0	72.7
	Primary prevention	3	7.3	0.0	60.4
	Secondary prevention	5	45.5	5.9	85.0
	Opportunistic	1	9.1	0.0	27.8

Table 6.10: Purpose of ordering lipid tests for selected problems

(a) Multiple purposes for ordering lipid tests could be ordered per problem. There were cases where the purpose of the lipid was missing for the problem. The purpose of lipid testing was known for: 89 lipid disorder problems (missing n = 0); 66 hypertension problems (missing n = 2); 48 Type 2 diabetes problems (missing n = 3); 28 health check problems (missing n = 1); 11 weakness/tiredness problems (missing n = 0); 11 overweight/obesity problems (missing n = 0).

* Includes multiple ICPC-2 PLUS codes (see Appendix 6).

6.5 Discussion

FBC tests were widely ordered by GPs for a broad range of problems, most often for investigative or diagnostic purposes. The results suggest GPs generate an estimated 10.6 million FBC tests in Australia annually. Lipid tests were predominately ordered for a small number of chronic problems, usually for the purposes of monitoring lipid levels, and in total GPs' orders for lipid tests generate an estimated 9.4 million tests per annum in Australia.

Initiation of testing

FBC and lipid testing was primarily initiated by the GP. While patient desire to have pathology tests has been described as a factor influencing GPs' pathology ordering,^{36,37} GPs perceived patients as initiating FBC and/or lipid testing at relatively few encounters (5% of FBC orders and 11% of lipids). There was no statistically significant difference between the proportion of FBC and lipid tests that were patient-initiated. However, when compared with an earlier SAND study conducted in 2002, the proportion of patient-initiated FBC tests was significantly smaller than the average for all pathology tests (15.1%).³⁴⁸ This suggests that the decision to order FBC tests was more likely to be initiated by the GP than average for all tests, but this was not the case for lipid tests.

If an intervention or policy regarding FBC and/or lipid testing was to be introduced, then GPs should be the primary focus of the intervention or policy as the majority of testing is initiated by them.

Patient demand for testing has been described as one of the possible factors contributing to the increased use of pathology testing.^{28,29,363} Comparison of the results of the current study with the 2002 BEACH study suggests that there has been no increase over time in the proportion of testing that was initiated by patient requests. However, the current study measures 'successful' communicated requests for testing, not the total patient demand for testing.

Demand for testing has been assessed as expectations of testing³²⁻³⁵ and communicated requests to be tested.^{347,364,365} Other studies report that 14–39% of patients in general practice have an expectation of pathology or imaging testing being ordered at their visit.³²⁻³⁵ But only 7.5–9.4% of patients in outpatient settings

actually requested tests,^{364,365} a smaller proportion than those expecting to be tested.³⁴⁷ Both the proportions of patients expecting and requesting tests are larger than those whose request for testing is successful (as measured here).

The fact that a low proportion of patients in this study initiated pathology testing, together with the finding from other studies that fulfilment of expected or requested tests does not affect patient satisfaction,^{32,35,347} suggests that patient demand for testing is managed successfully by GPs.

Full blood count

Pattern of use of FBC tests

GPs' use of the FBC test was widespread, with GPs ordering FBCs for one-in-ten contacts with all patients aged 15 years and over, regardless of patient sex. There was a wide range of problems for which FBCs were ordered, reflecting the distribution of ordering across patient age groups.

Other authors have described clinician's ordering of FBCs as 'indiscriminate', 'reflexive' or 'routine' on the basis of patterns of ordering.³⁵⁰⁻³⁵⁴ The increased rate of GPs' FBC ordering over time for all problems (see Chapter 4), and specifically for the six problems investigated in Chapter 5, also raised concerns about GPs patterns of use of FBC tests.

The breadth and frequency of FBC testing in this study may be suggestive of routine ordering, but the purposes recorded by GPs for ordering FBC tests do not support this conclusion. GPs were able to describe a clinical purpose for 90% of FBCs ordered, the remainder being ordered 'opportunistically'. A FBC was described as 'opportunistic' if it was added to the pathology order after the decision to order was made. It is possible these opportunistic tests represent occasions of 'routine' or 'reflexive' ordering.

Investigative/diagnostic FBC tests

GPs indicated that the most common purpose for ordering FBCs was investigative/diagnostic, accounting for half of FBC testing in this study. GPs recorded the differential diagnoses being assessed with these investigative FBCs. The differential diagnoses were most commonly haematological disorders (e.g. anaemia) and infections, which aligns with FBC test ordering recommendations in the literature.^{173,276,281,282,284,359-361,366} Weakness/tiredness was the most common problem investigated with an FBC, but it accounted for only 12% of the investigative tests, reflecting the breadth of problems for which investigative FBC was ordered. The purpose of ordering FBC for weakness/tiredness was almost exclusively investigative, and the differential diagnoses recorded by GPs reflected those for which FBC testing was recommended in guidance documents (see Chapter 5).

Monitoring FBC tests

Monitoring was the second most common reason GPs ordered FBC tests, 35% of them being ordered for this purpose. The need to monitor the FBC in management of diseases where the result of the test directly reflects the disease status (e.g. anaemia, infections^{231,366}) is logical. In contrast, the reason for monitoring FBC in the management of hypertension, Type 2 diabetes and lipid disorders is unclear because the FBC result is unlikely to monitor variables directly related to these diseases. GPs indicated the main purpose of ordering FBCs for these conditions was monitoring, confirming the hypothesis I made in Chapter 5, that GPs' use of FBC tests for these conditions was for monitoring.

Monitoring tests for these three problems accounted for one-third of all monitoring FBCs. As discussed in Chapter 5, guidelines for these conditions do not recommend FBC monitoring. It is possible that for patients with chronic conditions, GPs' ordered FBC tests opportunistically. For example, if patients have comorbidities (such as anaemia) which require FBC testing, that were not managed at the current encounter, GPs may have ordered the test opportunistically when ordering other pathology tests for the condition under management. However, this may well not be the case. Further investigation is needed to gain a better understanding of the reasons GPs order FBC tests.

FBCs were ordered frequently when ordered for monitoring, patients being monitored were significantly more likely to have been tested within the previous 12 months (76%) than those ordered for other purposes (45%). It is unclear whether or not this represents an ideal monitoring interval because no general information is available to Australian GPs about use of FBC tests for monitoring.

I could find only one document giving GPs guidance on the use of the FBC test and the follow-up of abnormal FBC results.³⁶⁶ This New Zealand document provided consensus-based guidance on interpretation and follow-up of minor abnormal results

of the components of the FBC, because this was an area in which GPs indicated they needed guidance. As such, the guidance it contains is more relevant to FBCs ordered for investigative/diagnostic purposes than those ordered for monitoring.

It may be too difficult to provide general recommendations about FBC monitoring because the ideal frequency of monitoring will be influenced by the characteristics of the individual patient, the problem being monitored, and the expected variation of the test.³⁶⁶ Data on ideal monitoring intervals and interpretation of repeated FBC testing may be included in the guidance documents for the individual problems for which monitoring FBCs were ordered rather than in the general literature. However, as discussed above, this was not the case for hypertension, Type 2 diabetes and lipid disorders.

Given the volume of FBC tests ordered by GPs for monitoring it is of concern that there is no published guidance available to GPs on what constitutes the best monitoring strategy. There is also limited guidance available on the interpretation of monitoring FBC tests. As such, GPs would be guided by their clinical experience when ordering and interpreting FBCs used for monitoring. There is a need for guidance on the use of FBCs for monitoring, particularly for chronic disease.

Preventive FBC tests

GPs reported a purpose of primary or secondary prevention for 13% of FBCs ordered in this study, a proportion small enough to challenge the assumption that FBCs are usually used as a screening test.^{221,352,353} The use of FBCs for primary or secondary prevention was not supported in guidelines for preventive care (see Section 5.7). Similarly, studies investigating routine use of the FBC for case finding (i.e. primary prevention) in outpatient care recommended against the use of the FBC for this purpose because it generates a very low diagnostic yield (less than 1% of tests identified new cases of disease).^{352,353} Given this lack of support for FBC as a preventive test it is reassuring that it was not the main reason for GPs' use of the FBC test. However it represents an area in which FBC ordering may be inappropriate.

The 'health check' in general practice is considered a preventive care activity. Therefore, it was expected that FBCs ordered for 'health check' patients would be ordered for preventive purposes. While this was the case for 36% of FBC tests (primary prevention 27%, secondary prevention 9%), GPs also frequently reported monitoring (27%) and investigation (23%) as the purpose(s) of ordering FBCs for health checks.

The variety of purposes given for ordering FBCs for health check problems may indicate that the patients receiving health checks were not 'well' patients. They may have had symptoms requiring investigation with a FBC test or a disease (or history of disease) for which monitoring with a FBC was indicated. Caution is needed in interpreting the purpose of FBCs ordered for health checks as there were small numbers of health checks recorded in this study. However, the results demonstrate that the reason for ordering a test for a problem that is considered preventive cannot be assumed to be for prevention. If an intervention was implemented to discourage GPs from using FBC tests for preventive purposes, then targeting preventive health problems (such as the health check) to deliver this intervention would not be an effective strategy.

Due to the high volume of FBC tests ordered by GPs, the 13% ordered for prevention equated to approximately 1.4 million FBC tests ordered in Australia in 2008–09, a considerable volume of tests nationally that were not supported by current evidence-based guidelines.

Reducing the ordering of FBCs for prevention has the potential to substantially decrease the volume of FBC tests. But this will not necessarily translate directly into reduced cost of FBC tests funded through Medicare because of the coning funding rule (approximately 45–55% of FBCs ordered by GPs were 'coned out' and therefore were not Medicare-funded^{19,346}). It is unclear whether a similar proportion of FBCs ordered for prevention would be coned out. The argument about effectiveness of any future intervention aiming to reduce inappropriate FBC ordering must be based on the ability of the intervention to improve quality of ordering rather than cost-effectiveness. The cost-effectiveness argument does not hold in this instance due to the coning effect. However, reduction of FBC ordering for prevention represents an opportunity to improve the quality of pathology ordering and has the potential to generate savings in terms of resource use and manpower required for this testing.

245

Lipid testing

Pattern of use of lipid tests

GPs ordered lipid tests most frequently at encounters with adult patients aged 45–64 years, followed by those aged 65–74 years and 75 years and over. The peak in testing in middle age reflects the age group in which lipid levels in the population are at their highest;³⁶⁷ and the age from which lipid screening is recommended for cardiovascular risk assessment.^{176,177} It also reflects the age group targeted by MBS items for health assessments in patients at risk of chronic disease.^{18,232}

GPs ordered lipid testing more often at encounters with male patients than at those with females. This may be due to the increased risk of cardiovascular disease^{177,237} and mortality in males in the population compared with females.³⁶⁸ However, there were no corresponding sex-specific differences in the prevalence of dyslipidaemia or other conditions requiring management of lipid levels in patients attending general practice.²⁰⁹ This suggests that the difference seen in this study was not due to differences in the sex-specific prevalence of disease requiring lipid management. However, differences in sex-specific attendance patterns in general practice may explain some of the difference in rates of lipid testing. Men visit GPs less frequently than women,³⁶⁹ and so to maintain a comparable level (by sex) of lipid screening in the patient population, GPs would have to test males at a higher proportion of their attendances than females. The data imply that GPs are being proactive in ensuring male patients have their lipid levels tested rather than suggesting that females are being tested at a lower rate, and it is therefore unlikely that the difference is of concern.

GPs ordered lipid tests for a relatively focussed group of problems compared with the breadth seen for FBC tests. Lipid disorders, hypertension, Type 2 diabetes, and health checks accounted for 55% of lipid testing, and the majority of these tests were ordered for monitoring. This is not surprising as guidelines recommended that the best practice management of each of these problems requires assessment of lipid levels and, if necessary, their subsequent management and monitoring (see Sections 5.4–5.7).

The most common reason for testing lipid levels was for monitoring of an existing disease, recorded for almost 60% of lipid tests. Other purposes for ordering lipid tests

were also commonly recorded: an investigative/diagnostic purpose was recorded for 18% of lipid tests; primary prevention for 18%, and secondary prevention 18%. It was possible for GPs to record multiple purposes, and this occurred most often when testing for secondary prevention, with half also having 'monitoring' recorded as a reason for ordering.

Monitoring of lipids

GPs ordered lipid tests for monitoring more frequently than for other purposes, and generated approximately 5.6 million orders for monitoring lipid tests in 2008–09. The majority (59%) of lipid monitoring by GPs in the current study was repeated at an interval of 3–6 or 7–12 months, indicating that GPs are following current guideline recommendations that advocate 6–12 monthly monitoring (see Section 5.6).

Recent evidence from two studies suggests that monitoring lipid levels at an interval of less than 3 years is likely to represent measurement error, rather than true change, because the signal-to-noise ratio in cholesterol testing is weak.^{226,349} In other words, the amount of short-term biological and analytical variation (noise) makes it difficult to detect small changes in cholesterol levels (signal) when tests are repeated too frequently.

Only one of the above studies was published prior to the conduct of the current study. Glasziou et al. used data from the 'LIPID' clinical trial, and recommended that among patients in whom lipid levels are stable (within 0.5 mmol/L of target) the interval for monitoring should be every 3–5 years.²²⁶ A move toward such a testing interval would represent a major change to GPs' current lipid monitoring practices. The fact that this represents a change to GPs' existing behaviour is of itself a barrier to the adoption of the new evidence.³²¹

One of the commonly stated barriers to adoption of guideline recommendations is disagreement with the evidence or lack of applicability of the evidence to the patient population.^{153,316-319,321} This includes the perception that clinical trial evidence does not reflect the real world clinical setting. This 'barrier' can be applied to the LIPID trial: it included patients had a history of acute coronary syndrome, who were taking pravastatin and had high total cholesterol.²²⁶ Using the first criteria alone—acute coronary syndrome—would represent (at most) 12% of general practice patients in

the current study, assuming that all patients who had an 'other cardiovascular disease' had an acute coronary syndrome. GPs may be hesitant to adopt this new evidence, perceiving that the study findings would be relevant to a small minority of their patients.

The second study that recommended an interval of at least 3 years between repeated lipid measures was conducted in Japan and published in 2010. Takahashi et al.³⁴⁹ investigated the signal-to-noise ratio of lipid measures taken in adults receiving an annual health check-up (excluding those who were taking a lipid lowering medication), and provided further evidence supporting the recommendation for a longer interval between repeated lipid measures in a broader group of patients.³⁴⁹ GPs' may accept this 'real world' evidence more readily than the clinical trial data from the earlier study. Takahashi's study was published after my research was completed, therefore I could not measure its subsequent impact (if any) on GPs' ordering behaviour. GPs have been shown to adopt new evidence in their pathology ordering behaviour quickly (before it is incorporated in guidelines).⁸⁸ However, to do so they need to trust the evidence itself, and its applicability to their patients.^{153,316,317,319-321}

Another factor is that GPs may believe that monitoring lipid levels has an effect on patient behaviour. The outcomes of testing in terms of patients' emotional, social, cognitive and behavioural responses to testing are not well studied.⁶³ It is possible that these outcomes could be beneficial to the patient (such as, improved adherence to medication, reassurance) or detrimental (such as increased anxiety).⁶³ Numerous studies have shown that patient adherence to lipid lowering medications, particularly statins, is poor and decreases over time.³⁷⁰⁻³⁷² For example, non-adherence of statins in patients aged 65 years and over were 38%, 42% and 56% after 1 year, 2 years and 5 years respectively.³⁷¹ Patient adherence is important to ensure the health benefits of lipid lowering medication are realised and cost-effectiveness is maintained.^{370,373,374}

Thus patient adherence to medication is a valid concern for GPs, but it is unclear whether monitoring lipids will improve adherence. A few studies have shown an association between lipid testing (monitoring) and improved adherence to lipid medication, but the causality of this relationship is unproven.³⁷⁵ The results of

monitoring lipid tests have been found to have only limited ability to detect patients' lipid medication adherence.³⁵⁵ Hence it appears that the action of monitoring may have an effect on patient adherence even though the results of these tests may not accurately measure adherence.

Despite the lack of robust evidence, guidelines for the management of lipid disorders recommend frequent monitoring of lipid levels as a tool to improve patient adherence to therapy.^{178,220} GPs may be guided by these recommendations and their clinical experience of whether or not testing has a desirable impact on patient adherence. The extent to which a change in the lipid testing interval would influence the volume of lipid tests analysed in laboratories can be seen in a UK study. Doll et al. found that if lipids were tested only once every 3 years then 79% of the lipid tests ordered in primary and secondary care in 2005–07 would not have been needed. Alternatively if lipid levels were tested once per year, as recommended for high risk patients in guidelines, then 42% of lipid tests would not have been needed.³⁶⁷ The current study shows that most patients (84%) for whom GPs ordered lipid tests had cardiovascular conditions or risk factors for which current guidelines recommend annual testing. Therefore the lower estimate is likely to be a closer reflection of the amount of lipid testing that may be surplus in general practice based on current guideline recommendations.

The study by Doll et al. is likely to overestimate the amount of lipid testing that was 'unnecessary' by applying a simple measure of repeat testing to all patients. Further details about these patients (such as their diseases, risk factors or the reason for ordering a lipid test) were not available, yet may have an influence on the ideal interval between repeated testing. The longer interval of 3-5 yearly testing was not recommended for all patients.^{226,349} It would only apply to patients being monitored, and the current study shows that only 60% of lipid tests were ordered for monitoring. Recommendations for a long interval between repeated testing would be applicable to only a subset of patients. Further research is needed to identify the subset of patients for whom GPs could implement a longer monitoring interval.

Lipid tests ordered by GPs generate a substantial volume of testing—an estimated 9.4 million lipid tests were ordered at GP encounters in Australia in 2008–09. Of these, 5.5 million were ordered for monitoring purposes. A change in monitoring

practices has the potential to substantially reduce the volume of lipid testing ordered by GPs. Further research is needed to determine: whether a lengthening in monitoring interval is acceptable to GPs and patients, for which subset patients this would be applicable, and whether it would adversely affect patient outcomes. While a reduction in volume of lipid testing generated by GP orders would generate savings in terms of resources required for testing, the extent of cost savings to the MBS is not known because of the current iso-resource items and coning funding model.

Limitations

The lipid and FBC testing activity for patients aged 75 years and over may not be accurately assessed in the study due to an under-representation of patients in this age group.

In this study, the definition of secondary prevention provided to participating GPs was 'screening in a patient with established risk factors for a condition but without the condition' and was based on the definition of the US Preventive Task Force that states "*secondary preventive measures* identify and treat *asymptomatic* persons who have already developed risk factors or preclinical disease but in whom the condition has not become clinically apparent."¹⁰¹ The definition used in this study should have stated that the patient had not already been diagnosed with disease rather than stating that the patient did not have the disease. There is some contention around the definition of secondary prevention throughout the literature and confusion about the application of these definitions.^{376,377} Therefore, readers may wish to combine primary and secondary prevention and view results as being indicative of FBC and lipid tests ordered for preventive purposes.

The interval between FBC and lipid tests is a single measure of the time between the patient's current and most recent FBC or lipid test. When the purpose of the test is monitoring this interval between FBC or lipid tests has been interpreted as the usual monitoring interval, but this may not reflect the patient's regular pattern of testing. The FBC is a battery of tests which includes multiple components that are ordered together for convenience but have different clinical uses. The components of interest will vary based on the clinical situation. For example, when investigating anaemia the red cell count and red cell morphology are the components of interest. As such

the GPs' purpose for ordering the FBC test may reflect only part of the included tests.

Conclusion

FBC is the most commonly ordered test in general practice, yet there was a surprising lack of data and guidance available about the use of the test. This study partially fills this gap by providing data about GPs' use of FBC tests. GPs regularly ordered FBC tests for adult patients, for a broad range of problems, and most tests were ordered for investigative/diagnostic or monitoring purposes. Patients initiated only a small proportion of FBC tests, most of this testing being generated by the GPs' judgement that testing was needed.

There is a need for further research on the use of the FBC test for monitoring, including the situations in which it needs to be monitored, and the appropriate monitoring strategy. In contrast, FBC tests ordered for prevention represent a clinical use that is not supported in evidence-based guidelines. Prevention was not a major reason that GPs ordered FBCs, but due to the large volume of FBC tests ordered it represents a considerable number of tests that are potentially unnecessary, and is an area in which interventions could be targeted. Further, given the breadth of problems for which GPs' ordered the FBC, it would be better to target an intervention at the test and purpose (e.g. FBC tests used for prevention) than at the use of the test for a particular problem.

This chapter also describes GPs' ordering of lipid tests, the second most frequently ordered test in general practice. Most lipid testing was generated by the GPs' judgement that testing was needed. Lipid tests were ordered primarily for chronic problems, and most tests were ordered for monitoring purposes. New evidence suggests GPs may order monitoring tests more frequently than is needed. Introducing a longer interval between monitoring tests has the potential to significantly reduce the volume of lipid tests ordered by GPs. However, further research is needed to determine for which general practice patients this change in monitoring strategy could be applied, and to ensure it would not result in adverse outcomes for patients.

7 Predictors of GPs' pathology ordering

7.1 Background

The increasing use and cost of pathology services in general practice in Australia,⁵ and internationally,⁸⁻¹⁰ have prompted concern about the quality of GPs' pathology ordering. In Chapter 5, I assessed quality of ordering by measuring the appropriateness of GP pathology ordering for selected morbidities. In this chapter I investigate factors that contribute to variance in GPs' pathology ordering rates because the presence of variance among GPs is often regarded as an indicator of poor quality.

The presence of considerable variance among GPs in their use of pathology testing services has been established in Australia and internationally.^{43-51,71-73,337} There is a huge range of reasons for which clinicians (including GPs) order pathology tests, including: diagnostic factors; therapeutic and prognostic factors; to monitor illness and drug therapy; for screening; patient-related factors; doctor-related factors; and policy and organisational factors.^{36,37,40,41} All of these factors have the potential to account for variance among GPs' in terms of their pathology ordering and multiple factors may simultaneously influence this variance. Therefore studies seeking to investigate the variance among GPs' ordering of pathology tests aim to determine the independent predictors of the variance, using multivariate analysis.^{43-51,71-73}

The most common factors investigated to identify the independent causes of variation are doctor-related factors or policy and organisational factors, followed by factors related to the patients. Other studies have found behavioural factors (such as GP attitude to risk and malpractice claims,^{378,379} and communication style³⁸⁰) independently influenced GPs' pathology ordering behaviour, but these cannot be addressed through methods adopted in this thesis. The non-behavioural variables found to be independently associated with GPs' pathology ordering are summarised below.

Studies that found an association between GP sex and pathology testing reported that female GPs ordered more tests.^{43,46-48} In other studies there was a univariate

association between GP sex and pathology testing^{45,49,51} but with modelling this relationship disappeared^{44,45,49} or was unclear.⁵¹ Other studies did not test the influence of GP sex.^{50,71}

Increased GP age or more years of experience were found to be independent predictors of lower rates of pathology ordering in some studies.⁴³⁻⁴⁵ Others found a non-linear relationship between years of experience and testing in young GPs.^{46,48} In other cases no effect of GPs' age or experience has been shown^{47,49} or it was not tested.^{50,71}

Country of graduation was found be a factor in some studies^{43,46,47} but not others.⁴⁵ An effect related to education was found in one study in which variation was related to the training hospital attended by young GPs,⁴⁸ but 'medical school attended' did not have an effect in a separate study.⁴⁹ This suggests a possible effect of training, but the consistency of this result is affected by the health system of the country being studied.

Geographic location of the practice appeared to have an independent relationship with pathology ordering but the results were inconsistent as were the variables assessed. Rural areas were associated with lower^{47,49} and higher^{43,45} rates of testing. A relationship between increased distance to laboratory and lower rates of testing was found in one study⁴⁶ but not another.⁵⁰ Associations with size of the local population,⁴⁶ geographic region⁵⁰ and country⁵¹ have also been found.

The number of GPs working at a practice was associated with higher⁴⁵ and lower⁵⁰ rates of pathology ordering. A number of other GP variables were found to be independently associated with testing. Examples include: involvement with guideline development,⁵⁰ number of auxiliary staff,⁴⁶ and use of problem orientated order forms.⁵⁰ These were not frequently investigated in studies and may reflect associations unique to the study population.

In some studies, the type of work or workload associated with pathology ordering was assessed at an aggregated level and related to the GP. Factors associated with increased testing were high frequency of visits per patient,⁴⁴ high proportion of (age-adjusted) patients,⁴⁷ higher proportion of services performed in the office (versus the hospital),⁴⁹ lower rates of psychotherapy services,⁴⁹ and higher rates of

long (Level C) Medicare consultations.⁴⁵ These variables may reflect differences in pathology testing based on the types of patients seen by GPs.

Most authors incorporated patient age and sex in their studies but only a small number measured the effect of this on variation in GPs' pathology ordering. Patient age was incorporated in most,^{43-47,49,51,72,73} but not all^{50,71} studies. Older patients were independently associated with more testing,^{43,46} as was management of patients aged 15–64 years.⁴⁵ Hartley et al. found age a predictor of pathology testing but did not describe the nature of the relationship.⁷³

Similarly, patient sex was incorporated in most,^{43-46,49,51,72,73} but not all^{47,50,71} studies. Management of female patients was independently associated with higher testing levels in some studies^{43,46} and had no effect in others.^{45,73}

Ferrier et al. incorporated the proportion of female patients aged 15–49 years in their study and found this group of patients were independently associated with higher rates of testing. Two studies adjusted the outcome measure of the study (rate of pathology testing) by patient age and sex.^{44,51}

Adjustment for patient morbidity was only made in three studies.^{45,72,73} In one of these studies the amount of variation explained by morbidity was not stated.⁷² However, in the other two, it was the variable that accounted for the greatest proportion of variation in GPs' pathology ordering.^{45,73} In many studies it is possible that data about patient morbidity was not available. Despite this lack of data, some authors have assumed that casemix has little effect and variability is due to the individual GPs' clinical requesting practices.⁵⁰

Some of the studies investigating variation in GPs' pathology ordering did not specify the variables that were significant predictors or did not describe the nature of the relationship, instead stating that a relationship existed.^{44,51,72,73} In another study the multivariate model did not identify any statistically significant predictors of GPs' pathology ordering.⁷¹

There is some inconsistency between studies in the variables found to be independently associated with GPs' pathology ordering. There are several possible causes for this inconsistency. The outcome variable used in the studies varied. Outcomes used include: total pathology ordering per GP; ^{43-46,71,73} rates of pathology adjusted by patient age-sex;⁵¹ ordering rates of a selected group of the most common individual tests;^{47,50} a mixture of pathology and imaging tests;^{46,50,72} ratio of pathology services to all services claimed by the GP;⁴⁸ quintiles of GPs' testing from low to high;⁴⁷ number and cost of tests per patient;⁴⁹ and presence or absence of testing (i.e. a binary variable).⁷² To some extent these outcome measures were due to differences in the data sources used, such as laboratory data,^{47,50,71} billing data,^{43,48,49} and data collected at the GP encounter.^{44-46,51,72,73}

Further, the number of variables included in multivariate analyses differed considerably and this also may be due, in part, to the source data. In particular, the lack of inclusion of morbidity or casemix variables in the studies may reflect data limitations.

Authors often had a narrow objective or specific hypothesis that the study was designed to address (e.g. to assess the effect of attending medical schools of different universities⁴⁹), in terms of determining whether a specific variable was an independent predictor of pathology ordering.^{46,48-50} In some cases this led to authors choosing a limited number of variables to include in the multivariate models, but the basis for selection was not provided.^{49,50} Consequently variables were omitted that others had found were independent predictors of variation (such as GPs' sex⁵⁰) even when these variables were available in the data set.⁵⁰ This calls into question the quality of some of the models used.

Several studies involved a local sample^{44,46,48,49,71-73} and/or were not representative of the GP population within the respective country.^{46,48-51,72} In addition, the majority of studies were either published more than 10 years ago, or, though published more recently, used data more than 10 years old,^{43,44,46,48-51,71-73} and can be considered out of date. The limitations of previous studies, including inconsistent methodology, poor quality of the multivariate models, lack of representativeness and age of the data severely limit their reliability.

Only two studies included data that were collected since 2000 from nationally representative samples.^{45,47} The most recent of these was conducted by Vinker et al. in Israel in 2003 using laboratory data for 16 common pathology tests ordered by primary care physicians.⁴⁷ The limited number of tests, lack of data on morbidity and differences in health systems between Israel and Australia meant that this study

could not provide a reliable indication of factors likely to account for variation in GPs' pathology ordering in Australia.

The second study used data collected in the BEACH study (the same study as used in this thesis) from a representative Australian sample in 1998–2001 to investigate variation in GPs' pathology ordering.⁴⁵ This previous BEACH study is relevant and relatively contemporary. However, there has been significant growth in the rate of GPs' pathology ordering since the study was done—it increased by 31%, from 29.7 tests/batteries of tests per 100 encounters in 2000–01 to 43.2 in 2007–08.²⁷³ Due to this growth, and the limitations of previous research, I have investigated predictors of GP pathology ordering in this thesis. Identifying factors that are independent predictors of pathology ordering has the potential to inform future patterns of pathology ordering, particularly if these predictors relate to expected changes in the characteristics of the GP workforce and population characteristics.

7.2 Objective

This chapter investigates the extent to which the variance in GPs' pathology ordering rates can be explained by a number of factors including GP, practice, and patient characteristics, the clinical problem under management, and the type of encounter.

7.3 Method

The simple and multiple regression analyses were performed on data collected in BEACH from April 2007 to March 2008. In these analyses the GP was the unit of analysis and the outcome variable was total number of pathology tests/batteries of tests ordered per GP—referred to as rate of GP pathology ordering throughout this chapter. The statistical software package SAS 9.1.3¹²⁹ was used for univariate and multivariate analyses.

The data were unweighted because the variables used for weighting (GP age and GP sex) were adjusted for in the analyses (see Section 3.2.10).

In BEACH, problems under management and pathology tests ordered are coded at the GP terminology level using ICPC-2 PLUS and classified to the ICPC-2 (see Section 3.2.7). In this chapter problems managed were analysed at the classification chapter level using the 17 ICPC-2 chapters. As the analysis conduced in this chapter investigates GPs' total pathology ordering, the types of pathology tests ordered by GPs were not investigated.

Simple linear regression was used to determine the proportion of variance in GP pathology ordering rates explained by each individual variable. Explanatory variables investigated are listed by 'family' in Box 7.1.

Variables listed in Box 7.1 were included in both the univariate and multivariate analysis, except where variables were highly correlated. This was the case for GP age and years in general practice, hence only GP age was fitted in the multivariate analysis.

In the analysis of variance, most variables were assigned a reference group. This group (or category) is clearly labelled for each variable in the results, and results are interpreted in relation to the reference group. However, a static reference group was not assigned for three variables: proportion of encounters by patient age, rate of problems managed in each ICPC-2 chapter, and the proportion of encounters with MBS item consultations classified as level A, B, C and D. Instead, for these categorical variables, each category was compared with the remainder of the categories in that variable. In effect the 'remainder of the categories' becomes the reference group to the category being examined. For example, when investigating the variance explained by MBS consultation levels: level A MBS consultations are compared with, as one group, level B, C and D consultations; level B MBS consultations are compared with level A, C and D consultations, and so on. Multiple regression using stepwise (backward) elimination was used to find the independent predictors of GP pathology ordering rates. Predictor variables were fitted into the model in 'families' in the following order: GP demographics, practice characteristics, consultation type, patient demographics, and problems managed. The model was reduced by each family in turn, starting with problems managed, adjusting for all other families. Variables within problems managed were kept if significant (alpha=0.05) or improved the fit of the model. The next family (patient demographics) was then reduced, and so on, until all retained variables are (to some degree) significant.³⁸¹

Box 7.1: Variables included in univariate and multivariate analysis

- GP characteristics
 - Sex
 - Age
 - Years in general practice
 - Fellowship of the Royal Australian College of General Practitioners (FRACGP) (yes/no)^(a)
 - Workload (number of sessions worked per week)^(a)
 - Country of graduation (international medical graduates)^(a)
- Practice characteristics
 - Size of practice (number of full-time [FTE] equivalent GPs)^(a)
 - State or Territory of practice location
 - Rurality of practice location using the Australian Standard Geographical Classification (ASGC) remoteness areas^(a)
 - Accreditation[‡] status (yes/no)
 - Teaching status (undergraduate or registrar training) (yes/no)^(a)
- Patient characteristics
 - Proportion of encounters by patient sex
 - Proportion of encounters by patient age
 - Proportion of encounters with new patients[‡]
 - Proportion of encounters with patients with a health concession card[‡] (Health care card[‡] or Repatriation health card[‡])
- Problems managed
 - Rate of problems managed in each ICPC-2 chapter[‡]
- Consultation type
 - Proportion of encounters with MBS GP consultation service items classified as level A, B, C and D.[‡]
- (a) These variables are described in the data elements section of the method (Section 3.2.5).
- Definition of term is included in glossary.

7.4 Results

7.4.1 The sample

A total of 953 randomly sampled GPs participated in BEACH in 2007–08 and recorded data about 95,300 encounters.

The representativeness of the GPs and encounters sampled in 2007–08 has been demonstrated and published elsewhere.¹²¹ The assessment of representativeness of all annual BEACH samples included in this thesis are discussed in detail in Section 3.2.10. In summary, in 2007–08, the sampled GPs were representative of Australian GPs in the sample frame in terms of their sex, age, and distribution across states. However, participants who graduated in a country other than Australia were slightly under-represented (26.5%) when compared with the total sample (30.2%) ($\chi^2 = 5.72$, p = 0.017).¹²¹

The age-sex distribution of patients at BEACH encounters where MBS general practice consultation service items were recorded as claimable was compared with that of patients at all Australian encounters claimed as MBS general practice consultation service items in the 2007–08 study period. There is an excellent fit of the MBS and BEACH age-sex distribution, with no age-sex category varying by more than 15% from the population distribution. The range of precision ratios (0.91–1.14) indicates that the BEACH sample of encounters is a good representation of Australian GP–patient encounters.¹²¹

GPs who participated in BEACH in 2007–08 ordered 45,597 pathology tests/batteries. An average of 47.8 pathology tests/batteries were ordered per GP per 100 encounters (standard deviation: 29.6), and 95% of GPs ordered between 9 and 97 pathology tests/batteries per 100 encounters (Table 7.1). The outcome variable (rate of GP pathology ordering) showed a normal distribution (results not shown).

	Number of pathology tests per 100 encounters from each GP
Variable	(n = 45,597 tests/batteries of tests)
Mean (standard deviation)	47.8 (29.6)
Range	0–263
Percentile of distribution	
5%	9
25%–1st quartile	27
50%–Median	43
75%–3rd quartile	64
95%	97

Table 7.1: Description of pathology ordering per 100 encounters with GPs

7.4.2 Univariate analysis

The results of the univariate analysis are shown in Table 7.2. The number of GPs analysed for each variable differs due to the removal of missing data, and the number used in the analysis of each variable is specified. Of the 17 variables investigated, 15 were found to be significant univariate predictors of GP pathology ordering rates when fitted alone.

GP characteristics

Female GPs had significantly higher pathology ordering rates than male GPs. Similarly those with Fellowship of the Royal Australian College of General Practitioners (FRACGP) had higher pathology ordering rates than their non-FRACGP colleagues (Table 7.2).

The rate of ordering decreased significantly with increasing age, GPs aged 55 years and over having the lowest ordering rates. Correspondingly, GPs with more than 20 years experience in general practice had the lowest ordering rates. Lower rates of ordering were also generated by GPs with a high workload (more than 10 sessions per week) compared with GPs who worked 1–5 sessions per week, and those who obtained their primary medical degree in Asia compared with Australian graduates (Table 7.2).

Practice characteristics

Compared with GPs in small practices with less than 2.0 full-time equivalent (FTE) GPs, those in moderately sized practices (with 2.0–4.9 and 5.0–9.9 FTE GPs) had

higher rates of pathology ordering. GPs in practices that were accredited and in those that provided undergraduate and registrar training also had higher rates of ordering than their counterparts in non-accredited, and in non-teaching practices (Table 7.2). Location of practice by rurality was significantly associated with ordering rates. In particular, GPs in major cities had lower rates of ordering than GPs in inner and outer regional areas. The state or territory in which the GPs' practice was located showed a statistically significant association with pathology ordering rates (p = 0.045), but this association was not reflected in any individual state or territory (Table 7.2).

Patient characteristics

Higher ordering rates were associated with management of female patients, and of patients aged 45–64 years compared with patients in all other age groups. In contrast, management of children (aged less than 15 years) was associated with lower ordering rates of pathology (Table 7.2).

Problems under management

Higher rates of ordering were associated with management of problems classified in the following ICPC-2 chapters (in order of predictive value): female genital; endocrine, metabolic and nutritional; general and unspecified; urological; blood and blood-forming organs; pregnancy and family planning; psychological; circulatory; digestive; male genital; and social. Management of problems classified as respiratory was associated with lower pathology ordering rates (Table 7.2).

Types of consultations

General practice consultations claimable as standard (level B) were associated with lower ordering rates, whereas those claimable as long (level C) or prolonged (level D) were associated with higher ordering rates (Table 7.2).

Variable	Regression Coefficient	Effect size (standard Beta)	Per cent of variance explained (R ² *100)	<i>P</i> value
GP characteristics			. ,	
Sex (<i>n</i> = 945) (reference = Male)			10.86	<0.001
Female	20.18	0.33		
Age (<i>n</i> = 953) (reference = < 35 years)			5.44	<0.001
35–44 years	-8.07	-0.11		0.039
45–54 years	-9.83	-0.16		0.008
55+ years	-21.72	-0.35		<0.001
Years in general practice (<i>n</i> = 946) (reference = 1–4 years)			2.59	<0.001
5–9 years	-1.65	-0.02		0.675
10–19 years	-2.48	-0.03		0.490
20+ years	-11.14	-0.19		0.001
FRACGP (<i>n</i> = 948) (reference = FRACGP)			0.81	0.006
Not FRACGP	-5.32	-0.09		
Workload (<i>n</i> = 944) (reference = 1–5 sessions per week)			1.65	<0.001
6–10 sessions per week	-6.55	-0.10		0.015
11+ sessions per week	-15.07	-0.16		<0.001
International medical graduate (<i>n</i> = 950) (reference = Australian graduates)			2.49	<0.001
UK / Ireland / New Zealand	-5.21	-0.05		0.136
Asia	-15.52	-0.16		<0.001
Other	-1.91	-0.02		0.577
Practice characteristics				
Size of practice (number FTE GPs) (<i>n</i> = 930 (reference < 2 FTE GPs)	0)		3.54	<0.001
2.0–4.9	9.36	0.16		0.001
5.0–9.9	14.87	0.23		<0.001
10+	0.93	0.01		0.811

Table 7.2: Univariate analysis of GP pathology ordering^(a)

(continued)

			Per cent of variance	
Variable	Regression Coefficient	Effect size (standard Beta)	explained (R ² *100)	P value
Location by state/territory ($n = 952$) (reference = Australian Capital Territory)			1.51	0.045
New South Wales	8.25	0.13		0.323
Victoria	10.55	0.16		0.209
Queensland	16.72	0.22		0.049
South Australia	10.27	0.10		0.239
Western Australia	16.52	0.15		0.063
Tasmania	11.22	0.07		0.254
Northern Territory	18.60	0.08		0.101
Rurality of practice by ASGC ($n = 952$) (reference = Major cities)			1.04	0.007
Inner/outer regional	6.70	0.10		0.002
Remote/very remote	6.90	0.03		0.340
Accreditation ($n = 948$) (reference = accredited practice)			1.27	0.001
Not accredited	-9.71	-0.11		
Teaching practice ($n = 948$) (reference = teaching practice)			1.80	<0.001
Not a teaching practice	-7.99	-0.13		
Patient characteristics				
Sex (<i>n</i> = 953) (reference = Female)			11.87	<0.001
Male patients	-0.75	-0.34		
Age (<i>n</i> = 953)				
Patients < 5 years	-0.49	-0.09	0.85	0.004
Patients 5–14 years	-1.46	-0.20	4.07	<0.001
Patients 15–24 years	-0.23	-0.05	0.25	0.123
Patients 25–44 years	0.06	0.02	0.04	0.524
Patients 45–64 years	0.40	0.11	1.25	0.001
Patients 65+ years	0.05	0.03	0.09	0.352
Health concession card $(n = 940)$ (reference = no card)			0.39	0.054
Health concession card holders	-0.08	-0.06		
New patient (<i>n</i> = 953) (reference = not new patient)			0.16	0.222
New patients	-0.10	-0.04		

Table 7.2 (continued): Univariate analysis of GP pathology $ordering^{(a)}$

(continued)

	Regression	Effect size	Per cent of variance explained	
Variable	Coefficient	(standard Beta)	(R ⁻ *100)	P value
Rate of problems managed (ICPC-2 chapter) (n = 953)				
General and unspecified (A chapter)	0.90	0.31	9.56	<0.001
Blood and blood-forming organs (B chapter)	3.11	0.23	5.13	<0.001
Digestive (D chapter)	1.07	0.16	2.68	<0.001
Eye (F chapter)	0.47	0.03	0.08	0.385
Hearing (H chapter)	-0.35	-0.03	0.08	0.372
Circulatory (K chapter)	0.45	0.17	2.77	<0.001
Musculoskeletal (L chapter)	0.06	0.02	0.04	0.562
Neurological (N chapter)	1.27	0.10	1.01	0.002
Psychological (P chapter)	0.62	0.19	3.43	<0.001
Respiratory (R chapter)	-0.57	-0.17	2.98	<0.001
Skin (S chapter)	0.11	0.04	0.18	0.185
Endocrine, metabolic and nutritional (T chapter)	1.42	0.41	16.6	<0.001
Urological (U chapter)	3.13	0.23	5.35	<0.001
Pregnancy and family planning (W chapter)	1.18	0.20	4.00	<0.001
Female genital (X chapter)	1.57	0.44	18.96	<0.001
Male genital (Y chapter)	2.55	0.15	2.20	<0.001
Social (Z chapter)	2.33	0.11	1.28	0.001
MBS item ^(b) ($n = 942$)				
Level A – short	-0.02	-0.00	0.00	0.959
Level B – standard	-0.25	-0.17	2.85	<0.001
Level C – long	0.93	0.36	12.78	<0.001
Level D – prolonged	2.76	0.25	6.42	<0.001

Table 7.2 (continued): Univariate analysis of GP pathology ordering^(a)

(a) Missing data were removed from the analysis. There were 953 GPs in the total sample and the numbers available for each variable are specified next to the variable label.

(b) The MBS items are content-based GP consultation descriptors ranging from Level A (attendance for an obvious problem, no time specifications) to D (exhaustive consultation, minimum of 40 minutes) see glossary.

Note: FRACGP – Fellowship of the Royal Australian College of General Practitioners; ASGC – Australian Standard Geographical Classification; ICPC-2 – International Classification of Primary Care, Version 2; MBS – Medicare Benefits Schedule. Also see glossary.

7.4.3 Multivariate analysis

The independent predictors of pathology ordering rates are shown in Table 7.3. The final model explained 50.4% ($F_{21,880}$ =42.65, *p* <0.001) of the variance in GPs' pathology ordering. This model included data from 902 GPs (94.6% of the original sample), for whom all data variables were complete. Of the 16 variables investigated, nine were found to be significant multivariate predictors of GP pathology ordering rates, results for which are described below.

GP characteristics

The association with GP sex and age seen in the univariate analysis persisted after adjusting for all other variables in the model. After adjustment, female GPs had significantly higher pathology ordering rates than their male counterparts, whereas GPs aged 45 years and over (particularly those aged 55 years and over) had lower ordering rates relative to GPs aged less than 35 years. In contrast, the association with workload and country of graduation found in the univariate analysis did not persist after adjustment (Table 7.3).

Practice characteristics

The effect of practice size and rurality persisted after adjustment. GPs in moderately sized practices with 5 to 10 FTE GPs had higher rates of ordering than those in practices with less than 2 FTE GPs. GPs in practices located in inner and outer regional areas had higher order rates than those in major city practices. However, the effect of state/territory of practice location, and medium practice size (2.0–4.9 FTE GPs) were no longer significant after adjustment (Table 7.3).

Patient characteristics

Although not significant in the univariate analysis, after adjustment, management of new patients was associated with higher ordering; and management of patients with a health concession card was associated with lower ordering. This suggests the effect of each of these variables on pathology ordering was associated with other variables in the model (Table 7.3).
Both before and after adjustment, management of patients aged less than 5 years was associated with lower ordering rates. The effect of the proportion of female patients and patients aged 45–64 years did not persist in the multivariate analysis (Table 7.3).

Problems under management

In the multivariate analysis, higher rates of ordering were associated with management of problems classified in the following ICPC-2 chapters: (in order of predictive value) endocrine, metabolic and nutritional; male genital; general and unspecified; female genital; urological; and blood and blood-forming organs (Table 7.3).

The effects of the management of problems classified as pregnancy and family planning, circulatory, respiratory, psychological and social, identified in the univariate analysis were not significant after adjustment (Table 7.3).

Types of consultations

General practice consultations claimable as long (Medicare level C) and to a lesser extent those claimable as standard (level B) were independently associated with higher rates of pathology ordering. The effect of the proportion of prolonged (level D) consultations did not persist after adjustment (Table 7.3).

-	Regression	T-Value		Effect size (standard	Unique variance
Predictor variables	coefficient	(F-partial)	P value	Beta)	(per cent)
GP characteristics					
GP sex (reference = Male)					
Female	5.86	2.92	0.004	0.10	0.48
GP age (reference = < 35 years)					
35–44 years	-4.43	-1.52	0.129	-0.06	0.13
45–54 years	-6.22	-2.23	0.026	-0.10	0.28
55+ years	-13.37	-4.49	<0.001	-0.22	1.14
Practice characteristics					
Practice size (reference = < 2 FTE G	Ps)				
2.0–4.9	3.79	1.86	0.064	0.06	0.19
5.0–9.9	7.90	3.68	<0.001	0.13	0.76
10+	1.54	0.53	0.594	0.02	0.02
Rurality of practice by ASGC (reference = Major cities)					
Inner/outer regional	6.22	3.73	<0.001	0.09	0.78
Remote/very remote	5.16	0.91	0.363	0.02	0.05
Patient characteristics					
Patients < 5 years	-0.40	-2.88	0.004	-0.08	0.47
Health concession card holders	-0.09	-2.81	0.005	-0.08	0.44
New patients	0.19	3.02	0.003	0.08	0.52
Rate of problems managed (ICPC-	2 chapter)				
General and unspecified (A chapter)	0.53	6.96	<0.001	0.18	2.73
Blood and blood-forming organs (B chapter)	1.18	3.54	<0.001	0.09	0.71
Endocrine, metabolic and nutritional (T chapter)	1.10	11.23	<0.001	0.32	7.10
Urological (U chapter)	1.21	3.63	<0.001	0.09	0.74
Female genital (X chapter)	0.72	6.36	<0.001	0.20	2.28
Male genital (Y chapter)	3.11	7.09	<0.001	0.19	2.83
MBS item ^(b)					
Medicare level B – standard	0.11	2.56	0.011	0.07	0.37
Medicare level C – long	0.35	4.18	<0.001	0.13	0.98

Table 7.3: Final multivariate model of independent predictors of GP pathology ordering^(a)

(a) Missing data were removed from the analysis. 902 GPs had all data variables available and were included in the final multivariate data.

(b) The MBS items are content-based GP consultation descriptors ranging from Level A (attendance for an obvious problem, no time specifications) to D (exhaustive consultation, minimum of 40 minutes) see glossary.

Note: ASGC – Australian Standard Geographical Classification; ICPC-2 – International Classification of Primary Care, Version 2; MBS – Medicare Benefits Schedule. Also see glossary.

7.5 Discussion

This study demonstrated that there was considerable variation among GPs in their use of pathology tests, with 95% of GPs ordering between 9 and 97 tests/batteries per 100 encounters. This variation is consistent with that reported in other studies.^{44,46,47,50,51,72,82,337} The multivariate model used in this study explained just over half of the variance in the rate of GPs' pathology ordering. A number of variables were found to be independent explanatory predictors of pathology ordering. The strongest was the type of problem under management, followed by variables related to the GP, the practice, the patients and the types of consultations. This study investigated variables that were independent predictors of the volume of

GPs' pathology ordering in Australia. While these variables do not explain the increase in GPs' pathology ordering seen in recent years,⁵ comparison of results with those of previous studies can inform whether these independent predictors have changed over time or new predictors emerged that may have contributed to the increase. In addition, the significant predictors identified in the current study can be viewed in the context of expected changes in the characteristics of the Australian GP workforce and population to predict likely future pressures on the utilisation of health resources for pathology testing.

GP characteristics

GP sex and age were the only GP characteristics associated with pathology ordering after adjustment in this study. Female GPs were found to order significantly more pathology tests than their male counterparts. Previous studies have either found a similar result,^{43,46-48} or no independent association with GP sex.^{44,45,49} Reasons for this discrepancy are unclear, but differences in the models used in each of the studies may have contributed—particularly the extent to which patient characteristics and problems managed by GPs were included. For example, compared with male GPs, Australian female GPs manage significantly more encounters with female patients and consequently manage more female-specific problems, such as female genital checks for Pap smears.³⁸² These factors are not independent of each other. Therefore, interpreting the effect of GP sex on pathology ordering reported in these studies after adjustment requires consideration of which variables were included in the multivariate model.

Of the four studies that found female GPs ordered more tests than male GPs, only two included adjustment for patient sex,^{43,46} and none adjusted for morbidity.^{43,46-48} Three earlier studies found no association of pathology ordering with GP sex, all three included patient sex,^{44,45,49} and two also included morbidity in the multivariate models.^{44,45} Consequently, the only two earlier studies that adjusted for patient sex and morbidity (one of which was the previous study using BEACH data from 1998–2001), as I did in this study, did not support the finding that female GPs ordered more pathology tests. However, a recent study investigating sex-specific differences in GPs' practising style, using BEACH data from 2009–10, found that female GPs ordered more pathology tests after adjustment for patient sex and morbidity.³⁸² This supports the findings of the current study, and suggests that a relationship between GP sex and ordering may have developed over time.

GP age and years of experience are commonly related to GPs' pathology ordering rates. Increasing age could be assumed to be synonymous with increasing experience. In this study age and experience were highly correlated, therefore only GP age was analysed—and found to be inversely associated, as age increased the rate of testing decreased. This supports the findings of three previous studies,^{43,45} two of which were representative Australian studies.^{43,45} In contrast, one representative international study found no effect of GP age or years since graduation on pathology testing.⁴⁷ Other small studies either reported a non-linear independent effect^{46,48} or no effect⁴⁹ of GP age on pathology testing after adjustment.

In the context of the Australian GP workforce the relationship of GP sex and age to pathology ordering raises potential concerns. The GP workforce has become increasingly feminised and is ageing.¹⁶ In 2008, one-third of GPs were aged 55 years and over and 38.4% were female.¹⁶ There have been considerable efforts to increase the number of medical graduates entering general practice³⁸³ to replace the ageing workforce, and the majority (62% in 2008) of GP registrars (i.e. medical graduates currently training to become GPs) are female.³⁸³ The results of this study suggest this group of young female GPs could be expected to order disproportionately more pathology tests. Therefore, the focus on identifying gaps in GP registrars' education about quality use of pathology services³⁸⁴ is timely.

Practice variables

This study showed that GPs practising in rural locations classified as inner and outer regional areas ordered more tests than those in urban areas. This finding is consistent with two earlier Australian studies: one used BEACH data;⁴⁵ and the other used MBS pathology billing data.⁴³

However, this finding was not echoed in international studies. In two of these, the opposite effect was found, with pathology ordering being higher in urban areas;^{47,49} and in another, no association between rurality of practice and pathology ordering was found.⁵⁰ The small geographic size of the country or region(s) investigated in these studies may explain differences in the effect of rurality on pathology ordering between the studies and compared with the current study.^{47,49,50} Some international studies assessed whether geographic location of the practice had an independent effect of pathology ordering using variables other than rurality (e.g. distance to laboratory from practice location), but a consistent effect did not emerge.^{46,50,51} The effect of rural location in the current and past Australian studies^{43,45} is likely to be related to the frequency of GP-patient attendance in these areas. In 2006–07, fewer GP services per person were claimed through Medicare with increasing geographic remoteness,³⁸⁵ possibly related to lack of access in the more rural areas. The three Australian studies that reported higher ordering in rural areas (after adjustment) all used encounter-based data about GPs' pathology ordering, and were therefore affected by attendance patterns (e.g. in encounter-based studies, the lower the rate of attendance the lower the likelihood of being sampled). A population group which attends less frequently (such as those living in rural areas), assuming a similar need for pathology, would be more likely to have pathology ordered because they have fewer GP visits in which the opportunity to order can arise. Therefore, it is unlikely the higher rate of pathology of GPs in rural locations seen in this study is a concern. However, this could be tested in future studies by adjusting for attendance patterns of patients in different rural areas or by using data not affected by attendance patterns.

Australian GPs' practising in larger practices had higher rates of pathology ordering than those in small practices in the current and past studies using BEACH data.⁴⁵ In contrast, Verstappen et al.⁵⁰ found that GPs in the Netherlands who worked in group

practices (> 2 GPs) ordered fewer tests than those in solo practice. No other studies investigated practice size as a predictor of pathology ordering.

Verstappen et al.⁵⁰ hypothesized that lower ordering in group practices was due to clinical support provided by the opportunity to discuss a case with colleagues. It is conceivable that the behaviour of peers could influence GPs' pathology ordering behaviour. Several studies have used feedback comparing GPs' test utilisation with their peers as part of interventions to improve quality of pathology ordering.^{82,92,96,386} Often these interventions achieve a reduction in pathology ordering.^{92,96,386} However, it is possible that peer influence may be multi-directional, depending on the behaviour of peers. In the current study, GPs practicing in larger practices ordered more tests than those in smaller practices, and this may reflect exposure to colleagues with higher rates of pathology ordering behaviour.

Another possible explanation for the effect of practice size is corporate ownership of general practices within Australia. General practices owned by corporations tend to be large and incorporate other health services (such as pathology and imaging services) which are also commonly owned by the corporation, in one medical centre.^{387,388} Since the mid 1990's corporate ownership of general practices in Australia has increased, raising concerns about the potential for influence on GPs' use of health services in which corporations have a shared interest.³⁸⁷⁻³⁹¹ Catchlove acknowledged this risk, but argued there was no evidence that GPs' service use had been unduly influenced by corporation status.³⁸⁸ Instead he believed that simply by locating health services together, the proximity would provide the opportunity for the corporations to profit from GP referrals without the need for inducements.³⁸⁸ Studdert et al.³⁹² sought to investigate whether proximity to the pathology industry as indicated by presence of pathology collection centres in or adjacent to the practice influenced GPs pathology ordering in Australia. During the data period for this study (2000–09) the number and distribution of these collection centres was regulated.³⁹³ Due to this restriction they were commonly located in larger practices due to the high volume of patients. Studdert et al. found no independent effect of practice proximity to collection centres (in Sydney and Melbourne) on GP pathology ordering.³⁹² While no evidence has emerged of inducements in regard to GPs' pathology ordering, the concern about the potential^{390,391} for such led the Australian Government to

strengthen the legislation on activities that are (and are not) acceptable between requesters (such as GPs) and providers of pathology services to ensure no inducements are made in regard to pathology services.³⁹⁴

Defining 'corporate ownership' in Australian general practice is difficult as there is a wide range of possible ownership models, and so the extent to which practice size represents corporate ownership is uncertain. If the higher pathology rate at larger practices represents higher rates of testing at practices that are corporate owned, this may be a concern. It must be kept in mind that this higher testing does not provide any measure of the quality of the pathology testing. It may be that larger practices, whether corporately owned or not, have more resources available to them to ensure that patients are tested appropriately (e.g. through patient recall for monitoring or preventive care). It is unclear why GPs in larger practices order more pathology tests. Further investigation is needed to determine what causes this effect and whether it represents a cause for concern.

Characteristics of GP workload

Variables related to the patient, the types of problems managed and types of consultations are all indicators of the GPs clinical workload.

In the current study, after adjustment, GPs' pathology ordering was not associated with management of patients of a particular sex, and only management of young patients (aged less than 5 years) was associated with lower rates of testing. This suggests that the significant effect of patient age and sex reported in the univariate analysis was counteracted by other variables in the multivariate model. In particular, there is likely to be an interaction between patient age, sex and problems that generate high volumes of pathology testing. For example, the management of sexspecific problems classified to the female genital (such as Pap smear) and male genital (such as prostate problems) ICPC-2 chapters, together explained 5% of the variance in this study. The inclusion of sex-specific morbidities is likely to have contributed to negating any independent effect of patient sex on GPs pathology ordering. Therefore the fact that many studies did not incorporate patient and casemix characteristics, other than age and sex into multivariate models used to investigate independent predictors of pathology ordering should be borne in mind when comparing results of studies.

The results of the current study, when compared with those of past studies using a similar method and multivariate model (incorporating patient age, sex and morbidity variables, among others), support earlier findings that patient sex is not an independent predictor of GPs' pathology ordering.^{45,73} However, there were not similar findings for patient age. The current study found management of young patients resulted in fewer tests and no age groups were associated with higher testing. In contrast, Britt et al. reported higher ordering associated with management of patients aged 15–64 years.⁴⁵ Hartley et al.⁷³ stated that patient age explained 3.3% of variation in British GPs' pathology ordering but not which age groups explained the variation.

Adjustment for patient age and/or sex was made in several other studies,^{43,44,46,47,49,51,72} but the effect of these variables was only reported in two of these.^{43,46} Kristiansen et al.⁴⁶ and Calcino⁴³ both reported that management of older patients and female patients were each independently associated with increased rates of pathology testing, but neither study incorporated morbidity in their analysis. In the current study the types of problems managed by GPs were the strongest predictor of GPs pathology ordering. Very few studies adjusted for patient morbidity in their investigation of independent predictors of pathology ordering. Yet the two past studies that did, also found that it was the strongest independent explanatory predictor of GPs' rates of pathology ordering.^{45,73}

Management of problems related to the endocrine system, those of a general and unspecified nature, and those related to the male and female genital systems were the strongest problem predictors of higher pathology ordering rates. Management of problems classified to these four chapters explained almost 15% of the variance in GPs' testing rates.

The individual conditions in Chapter 4 that contributed to the majority of pathology tests ordered by GPs may provide some insight into the conditions within these chapters which have the greatest potential to explain the variation. For example, the conditions classified in the general and unspecified chapter include health check-ups, weakness/tiredness, abnormal test results and viral illness. The endocrine and metabolic conditions include diabetes, lipid disorders, overweight/obesity, and thyroid problems. The chronic endocrine and metabolic conditions are perhaps an

area of concern in the context of Australia's ageing population³⁹⁵ as many of these conditions are more prevalent in older patients.¹⁵⁴

The higher rate of pathology associated with level C Medicare consultations may reflect increased complexity and length of the consultation. In order to qualify for level C reimbursement the consultation must be a minimum of 20 minutes and may involve taking a detailed history, a clinical examination, arranging any necessary investigations and/or implementing a management plan in relation to one or more problems.²³ The number of problems managed at general practice encounters and the proportion of encounters with older patients is increasing.¹⁰³ In the context of an ageing population,³⁹⁵ it might be expected that GP workload will become more complex involving more older patients with multiple chronic conditions—the management of which is likely to require pathology testing. The future impact of Australia's ageing population on the volume of pathology ordered by GPs is investigated in Chapter 8 of this thesis.

New patients were significantly more likely to receive pathology testing in the current study. Kravitz et al.³⁶⁵ found that new patients were independently more likely to directly request imaging and pathology tests from their physician and also to receive tests. Establishing a thorough medical history for a new patient is accepted practice and this often involves pathology tests. Lack of access to previous pathology results may also be a factor. In our current health system, GPs would usually only have access to previous records and test results if a new patient brought a copy to their first encounter.

Lower rates of pathology testing were associated with management of patients holding health concession cards. Concession cards incorporate: health care cards, which give access to health care through a higher Government subsidy level for those on lower incomes; and Repatriation health cards for returned servicemen and women and their families. The lower rate of GPs' ordering could be due to higher attendance patterns in this group, or to physicians' or patients' perception of affordability. Patients with a lower socio-economic index attend general practice more often³⁹⁶ and therefore (as discussed above) in an encounter sample they would have a higher chance of being included in the sample and a lower chance of having pathology recorded at the sampled encounter (because they have more opportunities to be

tested). Lower rates of pathology in health concession card holders may also indicate an access problem to pathology services. If patients or physicians perceive barrier(s) to ordering pathology (e.g. cost) at an encounter this may influence the decision to order testing. However the high rates of bulk-billing of Medicare pathology services (86.1% of pathology services were bulk-billed in 2007–08⁵) mean that cost is unlikely to be a barrier to the majority of patients.

Amount of variance explained

The final predictive model explained 50% of the variance in GPs' pathology ordering. This is a larger proportion of variance than that explained by previous studies with a similar method (i.e. models including GP and patient characteristics, and morbidity data collected at patient encounters).^{45,72} One of these studies used BEACH data and was able to explain 33% of the variance in pathology ordering in 1998–2001.⁴⁵ Davis et al.⁷² were able to explain 21% of the variance in New Zealand GPs' pathology and imaging ordering, but did not describe the specific variables contributing to this variation.

Another three studies that primarily investigated GP and practice variables reported markedly different amounts of explained variation in multivariate analysis: 10%,⁴⁶ 30%⁵⁰ and 49%.⁵¹ The first two studies investigated GPs' ordering of both pathology and imaging and neither had representative GP samples.^{46,50} The study that was able to explain 49% of the variance, investigated blood tests ordered by GPs in eight European countries, and the main contributor was the country variable, accounting for 45% of the variation.⁵¹ Several authors did not describe the proportion of variance explained in the final multivariate models used in their studies. Of the studies that did, only the current and previous study using BEACH data had a representative sample.

Factors found elsewhere to independently GPs' pathology ordering behaviour (e.g. GP attitude to risk and malpractice claims, ^{378,379} and communication style³⁸⁰) were not measured in this study. The extent to which they are applicable in the Australian general practice setting is not known, and further research in this area could determine the effect of these factors in the Australian setting.

Conclusion

This study has demonstrated that much of the variation in GPs' pathology ordering is due to the GPs' workload, particularly the types of problems managed. The observation by Smellie et al. that "clinical practice is often assumed to account for differences in the use of pathology tests, but there is little published evidence to support this" is pertinent.³³⁷ This study provides this evidence, and demonstrates that the key factor within 'clinical practice' is differences in the types of clinical problems dealt with by individual GPs. The problems that are prevalent in older patients are likely to place the greatest pressure on future use of pathology services as the Australian population ages.

There were some GP and practice characteristics independently associated with use of pathology services. These accounted for a much smaller amount of the variation, but the finding that young GPs, and female GPs, had higher pathology rates has implications for future pathology utilisation, due to expected increases in the number of trainee GPs (registrars) and continued feminisation of the workforce.

8 Pathology generated by GP orders in the future

The previous chapters of this thesis have investigated the historical growth in GPs' pathology ordering, the morbidities that contributed to the increase and some of the explanatory factors contributing to variance in their pathology ordering. This chapter considers likely future growth in GPs' pathology ordering based on the expected growth in, and ageing of, the Australian population.

8.1 Objective

The objective of this work is to assess the effect of the projected population growth and ageing on GP's future pathology ordering in Australia.

8.2 Background

In the past, growth in the volume of pathology tests ordered in Australia has been associated with the ageing population (i.e. growth in the proportion of the population in older age groups).^{26,27} Population projections indicate continued increases in the number of people in older age groups.¹³⁷ As the use of pathology services increases with age,²⁷ it is likely this will continue to be a contributing factor to future growth in pathology ordering.

The Australian Government has described the ageing population as a major challenge facing the future of the country because of its associated fiscal pressure.²³⁴ Health care costs are described as the major contributor to this fiscal pressure, contributing two-thirds of the expected increase in Government spending from 2010 to 2050.²³⁴ The population projections published by the Australian Bureau of Statistics (ABS) demonstrate both the expected ageing of the population and population growth.¹³⁷ In June 2010, Australia's population was 22.3 million and 13.6% of people were aged 65 years and over.¹⁵ The ABS have made three population projections based on different assumptions about fertility rates, longevity and migration rates. In 2050 the population is projected to be between 30.2 million and 39.6 million, with 22.2% to 24.3% aged 65 years and over. In all three projections the proportion of the

population accounted for by older people aged 65 years and over from 2010 to 2050 grows faster than does the total population.¹³⁷

Australia is not the only country facing fiscal pressure from an ageing population. A number of developed countries face similar pressures and the associated burden of expected increases in public health expenditure. Per capita health service use (including pathology²⁷ and GP service¹³⁶ use) and cost increase with age.^{234,397} Growth will be driven by demographic trends and technological advances, including, new drugs, tests and medical interventions,^{234,397} leading the Organization for Economic Co-operation and Development (OECD) to describe the pressure for increasing health spending in the future to be "unrelenting, fuelled by technological changes, population expectations and ageing."³⁹⁷

Factors contributing to ageing populations include declining fertility rates and increased life expectancy. Many countries will experience pronounced increases in the proportion of the population that is aged 65 years and over due to the baby boom following World War II. Much of this increase is expected to occur over the next 3 decades as the 'baby boomers' age.³⁹⁸ In Australia, the resulting population age structure is expected to persist to (at least) 2050 (assuming fertility rates and longevity do not change significantly),³⁹⁵ carrying with it the ongoing fiscal challenge of an increased dependency ratio (i.e. fewer working people per older people in the population). The long-term nature of the change in population age structure and dependency ratio translates to ongoing health expenditure pressures as outlined by the Australian Government and other OECD nations.^{234,397}

All areas of Australia's health budget will face scrutiny as the Government prepares to meet the fiscal challenges caused by ageing to ensure that "spending on health is sustainable, affordable and provides maximum benefit to the greatest number of people."²⁰ Public expenditure on pathology services is one important aspect of the public health budget. In 2009–10, Medicare outlays on pathology services were \$2 billion, 13% of total Medicare outlays.⁵ At the time the research for this chapter was conceptualised and analysis undertaken, the Australian Government was in the process of reviewing Medicare funding arrangements for MBS pathology services. One of the primary tasks of the review, which commenced in mid 2009 (and was completed in 2011), was to identify suitable alternate funding models.^{20,399}

Regardless of the outcome of this review, demographic changes in the population will contribute to growth in the volume and cost of pathology services. This chapter attempts to estimate the amount of pathology that will be ordered by GPs for the Australian population in the future, based on recent rates of GP pathology ordering and recent patterns of GP service use, in combination with projected estimates of the future population.

8.3 Method

Patient age-sex patterns of GP pathology ordering and GP service use are used to estimate the future number of GP contacts involving pathology testing, and volume of pathology testing generated by GP orders for the Australian population. National data from three sources are used to calculate projected estimates for 2015 to 2050, using data from 2005–06 and 2009–10.

Data sources

The three types of national data used are: (i) the population of Australia; (ii) the number of GP services in Australia; and (iii) GPs' pathology ordering rates (summarised in Box 8.1).

Box	x 8.1: Summary of source data		
		2005–06	2009–10
Nur	nber of people in the Australian population ^(a)	20.4 million	22.0 million
Nur	nber of GP services ^(b)	107.7 million	120.0 million
BEA	ACH ^(c)		
Ν	lumber of participating GPs	1,017	988
Ν	lumber of GP-patient encounters recorded	101,933	101,349
Ν	lumber of pathology tests/batteries of tests recorded by GPs	39,358	45,594
Sour	ces:		
(a)	Australian Bureau of Statistics published estimated resident population for 2005	and 2009. ¹⁵	
(b)	General practice services funded by Medicare Benefits Schedule and Department the Australian Government Department of Health and Ageing, and the Australian Affairs.	nt of Veterans' Affairs. I Government Departm	Data supplied by ent of Veterans'

(c) The BEACH study annual data set published in General practice activity in Australia 2005–06¹²³ and General practice activity in Australia 2009–10.¹⁷

Population data

Australian population data were sourced from the ABS and are based on census data. Population estimates are formally updated each year,¹⁵ and population projections to 2101 are also produced.¹³⁷ Data on the number of people by age (single year) and sex are published by the ABS for population estimates and projections.^{15,137}

The estimated resident population from 30th June 2005^{15} , 30th June 2009^{15} , and the projected estimated population for 30th June for each 5th year from 2015 to 2050^{137} are used.

The estimated number of residents in the Australian population was 20.4 million in 2005 and 22.0 million in 2009 (Box 8.1). The 2005 estimate was based on the 2001 census data, adjusted for births, deaths and migration in subsequent years, and further adjusted following the 2006 census.¹⁵ The 2009 population estimate was based on 2006 census data, adjusted for births, deaths and migration in subsequent years.¹⁵

Three series of future projections of the Australian population are produced by the ABS: Series A, B and C.¹³⁷ These series are based on past trends and on different assumptions about future fertility rates, life expectancy and migration.¹³⁷ The projected estimates used in this chapter are summarised in Box 8.2. Series A projections predict the greatest growth in the population and Series C the lowest growth.

Box 8.2: Australian projected population 2015 to 2050 (millions)								
	2015	2020	2025	2030	2035	2040	2045	2050
Series A	24.0	26.1	28.3	30.5	32.7	35.0	37.3	36.6
Series B	23.6	25.3	26.9	28.5	30.0	31.3	32.7	34.0
Series C	23.3	24.5	25.7	26.9	27.8	28.7	29.5	30.2
Source: Australian Bureau of Statistics population projections. ¹³⁷								

General practice services data

The numbers of GP services provided to the Australian population were estimated using claims data. These data were provided by DoHA and the Department of Veterans' Affairs (DVA); the two Australian Government organisations that fund the majority (95%¹⁷) of GP services in Australia. Each GP service is defined as a single GP-patient-date combination in the claims data. There were approximately 107.7 million GP services in 2005–06 and 120.0 million in 2009–10 (Box 8.1). GP

services data were supplied by patient age (single year) and sex. These data are not publicly available with the required patient age and sex specificity and were provided directly by DoHA (DoHA, personal communication, June 2010) and DVA (DVA, personal communication, September 2009).

DoHA provided the number of Medicare-funded GP services funded for two 12 month time periods—April 2005 to March 2006, and April 2009 to March 2010. The majority of Australia's GP services were funded by Medicare (100.6 million in 2005–06 and 112.8 million in 2009–10).

DVA provided the number of GP services funded via the Veterans' Entitlements Act for Repatriation health card holders (including veterans and eligible dependents) for the 12 month period, April 2005 to March 2006. However, the April 2009 to March 2010 data were not available from DVA. Therefore the 2005–06 DVA data (7.1 million GP services) were applied in the calculations that use 2009–10 data.

GPs' pathology ordering

GPs' pathology ordering data were drawn from the BEACH study. Annual weighted data are used for two 12 month time periods—April 2005 to March 2006, and April 2009 to March 2010. As discussed in Section 3.2.10, the representativeness of these two annual data sets has been assessed, both before and after weighting, and the application of weighting marginally improved the precision of the encounter data set.^{17,123} Weighting is applied to ensure the estimates are based on nationally representative data.

In the BEACH study, there were a total of 101,993 encounters recorded by 1,017 participating GPs in 2005–06, and 101,349 encounters recorded by 988 GPs in 2009–10. Details about 39,358 pathology tests/batteries were recorded at these encounters in 2005–06 and 45,594 tests/batteries in 2009–10 (Box 8.1).

Two measures of GP pathology ordering are used—the rate of pathology tests/batteries ordered at GP–patient encounters, and the proportion of encounters that involve at least one pathology test/battery. For each measure the point estimate and 95% confidence limits were calculated. These measures can be analysed by patient age (single year) and sex groups as needed.

Calculation of national estimates

This section describes how data from the three national sources described above are used to calculate:

- the number of pathology tests/batteries of tests that were, and in the future are estimated to be, ordered by GPs for the total Australian population.
- the number of GP contacts that involved, and in the future are estimated to involve, at least one pathology test/battery of tests for the total population.

To ensure calculations were robust, patient or population data from each source were grouped:

- 5-year age-sex groups were created for patients/people aged 0 to 84 years
 (e.g. males aged 0–4 years, females aged 0–4 years, males 5–9 years, females aged 5–9 years)
- two age-sex groups were created for patients/people aged 85 years and over (males aged ≥ 85 years, females aged ≥ 85 years).

Data where age or sex was missing were excluded from the analysis.

The method used to calculate the number of tests/batteries in the population is described in detail below. The same method was used to calculate the number of GP contacts that involve pathology testing—by using the proportion of encounters with at least one pathology test/battery in steps 2 to 4 instead of the rate of pathology ordering.

Box 8.3 provides a worked example of the method applied to both measures of GP pathology ordering for one age-sex group, female patients aged 40–44 years.

Step 1: The average number of GP contacts per person in the population

The average number of GP contacts per person is calculated by dividing the number of GP services in each age-sex cohort by the number of people in the population in the corresponding age-sex group. This is repeated using data from 2005–06 and 2009–10. Referring to the example in Box 8.3 the average number of GP encounters per female head of population aged 40–44 years in 2005–06 was 5.0, and in 2009–10 it was 5.4.

Step 2: The average number of pathology tests/batteries ordered by GPs per person in the population

The average number of pathology tests/batteries per person is calculated by multiplying the rate of GPs' pathology tests/batteries ordered per encounter (the rate per 100 encounters divided by 100) by the number of GP contacts per person (as calculated in step 1) in each age-sex cohort. This calculation was repeated using data from 2005–06 and 2009–10, and repeated for each time point using the lower and upper 95% confidence limits of the rate of pathology tests/batteries per encounter. For example, per woman aged 40–44 years, GPs ordered (on average) 2.4 tests/ batteries (95% CI: 2.2–2.7) in 2005–06, and 3.2 (95% CI: 2.9–3.5) in 2009–10 (Box 8.3).

Step 3: The total number of pathology tests/batteries ordered by GPs for the population The total number of pathology tests/batteries ordered by GPs for the population is calculated by multiplying the number of people in the population by the average number of tests/batteries per person (as calculated in step 2) in each age-sex cohort. This calculation was repeated using data from 2005–06 and 2009–10, and repeated using the lower and upper 95% confidence limits (calculated in step 2). Referring to the example in Box 8.3, there were approximately 1.88 million (95% CI: 1.70 million–2.06 million) tests/batteries ordered by GPs for females aged 40–44 years in 2005–06, and 2.46 million (95% CI: 2.25 million–2.68 million) in 2009–10.

Adding the estimates for all age-sex cohorts provides the total number of pathology tests/batteries ordered by GPs for the total Australian population in 2005–06 and 2009–10.

Step 4: The total number of pathology tests/batteries ordered by GPs for the projected population 2015 to 2050

The total number of pathology tests/batteries ordered by GPs for the projected population is calculated by multiplying the expected number of people in the population in each age-sex cohort by the average number of tests/batteries per person (as calculated in Step 2).

The number of pathology tests/batteries ordered by GPs was calculated for the projected population at 5 yearly intervals from 2015 to 2050 (that is, 2015, 2020, 2025... etc). For each of the projected estimates of the annual population used (2015

to 2050) there are two estimates of the average number of tests/batteries per person (one based on 2005–06 data and a second based on 2009–10 data) applied to each series of population projections at each time point. Applying the three series of population projections (A, B and C) to the 2005–06 and 2009–10 data gives six estimates (each with 95% CI) of the projected total number of pathology tests/batteries ordered by GPs for each age-sex cohort, and (when the age-sex cohorts are added) for the total Australian population in each year used in the calculations. The projections based on 2005–06 data assume that in the future the population continues to have the same average number of GP contacts (per person in the 5-year age-sex cohorts) as they did in 2005–06, and that GPs continue to order pathology tests/batteries at the same rate at GP encounters (for patients in the 5-year age-sex cohorts) as they did in 2005–06. Similarly the projections based on 2009–10 data repeat these two assumptions, on the basis of 2009–10 GP contact and ordering rates. The example in Box 8.3 uses the estimated number of females aged 40–44 years in the population in 2020 to illustrate the method for Step 4. There are three 2005–06based estimates of the number of tests/batteries ordered by GPs for females aged 40-44 years in 2020:

- Series A: 2.02 million (95% CI: 1.83 million–2.22 million)
- Series B: 1.97 million (95% CI: 1.78 million–2.16 million)
- Series C: 1.92 million (95% CI: 1.73 million–2.11 million).

Similarly there are three 2009–10-based estimates:

- Series A: 2.68 million (95% CI: 2.44 million–2.91 million)
- Series B: 2.61 million (95% CI: 2.38 million–2.84 million)
- Series C: 2.54 million (95% CI: 2.32 million–2.76 million).

Source data (females aged	40–44 years)		2005–06		2009–10	
Number of people in the pop	ulation	776,908		769,345		
Number of GP services			3,881,819		4,127,491	
Rate of pathology tests/batte	ries ordered by G	Ps per 100 encounte	ers (95% CI) 48.39 (43.68–53.10)	59.66 (54	1.44–64.87)	
Per cent of encounters with a	at least 1 patholog	gy test/battery ordere	ed by GPs (95% Cl) 21.75 (19.87–23.62)	23.73 (22	2.01–25.46)	
	Step 1:		2005–06 2009–10			
	Average num	ber of GP contacts	per person 5.00 5.36			
	►					
Number of pathology tests ordered by	GPs in the pop	ulation	Number of GP contacts with at least the population	ast one pa	thology test	ordered in
Step 2:	2005–06	2009–10	Step 2:		2005–00	6 2009–10
Average number of pathology tests/ba	tteries 2.42	3.20	Average number of GP contacts v	vith at leas	t one 1.09	9 1.27
Average number of pathology tests/ba per person (95% Cl)	tteries 2.42 (2.18–2.65)	3.20 (2.92–3.48)	Average number of GP contacts v pathology test/battery per person	vith at leas (95% Cl)	t one 1.09 (0.99–1.18	9 1.27) (1.18–1.37)
Average number of pathology tests/ba per person (95% Cl) ▼	tteries 2.42 (2.18–2.65)	3.20 (2.92–3.48)	Average number of GP contacts v pathology test/battery per person	vith at leas (95% CI) ▼	t one 1.09 (0.99–1.18	9 1.27) (1.18–1.37)
Average number of pathology tests/ba per person (95% Cl) ▼ Step 3:	tteries 2.42 (2.18–2.65) 2005–06	3.20 (2.92–3.48) 2009–10	Average number of GP contacts v pathology test/battery per person Step 3:	vith at leas (95% Cl) ▼	t one 1.09 (0.99–1.18 2005–06	9 1.27) (1.18–1.37) 2009–10
Average number of pathology tests/ba per person (95% Cl) ▼ Step 3: Number (millions) of pathology tests/b	tteries 2.42 (2.18–2.65) 2005–06 atteries 1.88	3.20 (2.92–3.48) 2009–10 2.46	Average number of GP contacts v pathology test/battery per person Step 3: Number ('000's) of GP contacts w	vith at leas (95% Cl) ▼ ith at	t one 1.09 (0.99–1.18 2005–06 844	9 1.27) (1.18–1.37) 2009–10 980
Average number of pathology tests/ba per person (95% Cl) Step 3: Number (millions) of pathology tests/b for the population (95% Cl)	tteries 2.42 (2.18–2.65) 2005–06 atteries 1.88 (1.70–2.06)	3.20 (2.92–3.48) 2009–10 2.46 (2.25–2.68)	Average number of GP contacts w pathology test/battery per person Step 3: Number ('000's) of GP contacts w least one pathology test/battery for the population (95% Cl)	vith at leas (95% Cl) ▼ ith at or	t one 1.09 (0.99–1.18 2005–06 844 (771–917)	9 1.27) (1.18–1.37) 2009–10 980 (909–1,051)
Average number of pathology tests/ba per person (95% Cl) Step 3: Number (millions) of pathology tests/b for the population (95% Cl)	tteries 2.42 (2.18–2.65) 2005–06 atteries 1.88 (1.70–2.06)	3.20 (2.92–3.48) 2009–10 2.46 (2.25–2.68)	Average number of GP contacts w pathology test/battery per person Step 3: Number ('000's) of GP contacts w least one pathology test/battery for the population (95% CI)	vith at leas (95% Cl) ▼ ith at or	t one 1.09 (0.99–1.18 2005–06 844 (771–917)	9 1.27) (1.18–1.37) 2009–10 980 (909–1,051)
Average number of pathology tests/ba per person (95% CI) Step 3: Number (millions) of pathology tests/b for the population (95% CI) Step 4: Se	tteries 2.42 (2.18–2.65) 2005–06 atteries 1.88 (1.70–2.06) ries 2005–06	3.20 (2.92–3.48) 2009–10 2.46 (2.25–2.68) 2009–10	Average number of GP contacts w pathology test/battery per person Step 3: Number ('000's) of GP contacts w least one pathology test/battery for the population (95% CI) Step 4:	vith at leas (95% Cl) ▼ ith at or ▼ Series	t one 1.09 (0.99–1.18 2005–06 844 (771–917) 2005–06	9 1.27) (1.18–1.37) 2009–10 980 (909–1,051) 2009–10
Average number of pathology tests/ba per person (95% CI) Step 3: Number (millions) of pathology tests/b for the population (95% CI) Step 4: Se Number (millions) of pathology	tteries 2.42 (2.18–2.65) 2005–06 atteries 1.88 (1.70–2.06) ries 2005–06 A: 2.02	3.20 (2.92–3.48) 2009–10 2.46 (2.25–2.68) 2009–10 2.68	Average number of GP contacts w pathology test/battery per person Step 3: Number ('000's) of GP contacts w least one pathology test/battery for the population (95% Cl) Step 4: Number ('000's) of GP contacts	vith at leas (95% Cl) ▼ ith at or Series A:	t one 1.09 (0.99–1.18 2005–06 844 (771–917) 2005–06 909	9 1.27) (1.18–1.37) 2009–10 980 (909–1,051) 2009–10 1,065
Average number of pathology tests/ba per person (95% Cl) Step 3: Number (millions) of pathology tests/b for the population (95% Cl) Step 4: Se Number (millions) of pathology tests/batteries of tests ordered by	tteries 2.42 (2.18–2.65) 2005–06 atteries 1.88 (1.70–2.06) ries 2005–06 A: 2.02 (1.83–2.22)	3.20 (2.92–3.48) 2009–10 2.46 (2.25–2.68) 2009–10 2.68 (2.44–2.91)	Average number of GP contacts v pathology test/battery per person Step 3: Number ('000's) of GP contacts w least one pathology test/battery for the population (95% Cl) Step 4: Number ('000's) of GP contacts with at least one pathology	vith at leas (95% Cl) ▼ ith at or Series A:	t one 1.09 (0.99–1.18 2005–06 844 (771–917) 2005–06 909 (831–987)	9 1.27) (1.18–1.37) 2009–10 980 (909–1,051) 2009–10 1,065 (988–1,142)
Average number of pathology tests/ba per person (95% Cl) Step 3: Number (millions) of pathology tests/b for the population (95% Cl) Step 4: Se Number (millions) of pathology tests/batteries of tests ordered by GPs for the population in 2020	tteries 2.42 (2.18–2.65) 2005–06 atteries 1.88 (1.70–2.06) ries 2005–06 A: 2.02 (1.83–2.22) 3: 1.97	3.20 (2.92–3.48) 2009–10 2.46 (2.25–2.68) 2009–10 2.68 (2.44–2.91) 2.61	Average number of GP contacts w pathology test/battery per person Step 3: Number ('000's) of GP contacts w least one pathology test/battery for the population (95% Cl) Step 4: Number ('000's) of GP contacts with at least one pathology test/battery for the population in	vith at leas (95% Cl) ▼ ith at or ▼ Series A: B:	t one 1.09 (0.99–1.18 2005–06 844 (771–917) 2005–06 909 (831–987) 886	9 1.27) (1.18–1.37) 2009–10 980 (909–1,051) 2009–1 (988–1,065 (988–1,142) 1,038
Average number of pathology tests/ba per person (95% Cl) Step 3: Number (millions) of pathology tests/b for the population (95% Cl) Step 4: Se Number (millions) of pathology tests/batteries of tests ordered by GPs for the population in 2020 (95% Cl)	tteries 2.42 (2.18–2.65) 2005–06 atteries 1.88 (1.70–2.06) ries 2005–06 A: 2.02 (1.83–2.22) 3: 1.97 (1.78–2.16)	3.20 (2.92–3.48) 2009–10 2.46 (2.25–2.68) 2009–10 2.68 (2.44–2.91) 2.61 (2.38–2.84)	Average number of GP contacts w pathology test/battery per person Step 3: Number ('000's) of GP contacts w least one pathology test/battery for the population (95% Cl) Step 4: Number ('000's) of GP contacts with at least one pathology test/battery for the population in 2020 (95% Cl)	vith at leas (95% Cl) ▼ ith at or ▼ Series A: B:	t one 1.09 (0.99–1.18 2005–06 844 (771–917) 2005–06 909 (831–987) 886 (810–962)	9 1.27 9 1.27 2009–10 980 (909–1,051) 2009–10 1,065 (988–1,142) 1,038 (963–1,113)
Average number of pathology tests/ba per person (95% Cl) Step 3: Number (millions) of pathology tests/b for the population (95% Cl) Step 4: Se Number (millions) of pathology tests/batteries of tests ordered by GPs for the population in 2020 (95% Cl)	tteries 2.42 (2.18–2.65) 2005–06 atteries 1.88 (1.70–2.06) ries 2005–06 A: 2.02 (1.83–2.22) 3: 1.97 (1.78–2.16) C: 1.92	3.20 (2.92–3.48) 2009–10 2.46 (2.25–2.68) 2009–10 2.68 (2.44–2.91) 2.61 (2.38–2.84) 2.54	Average number of GP contacts w pathology test/battery per person Step 3: Number ('000's) of GP contacts w least one pathology test/battery for the population (95% Cl) Step 4: Number ('000's) of GP contacts with at least one pathology test/battery for the population in 2020 (95% Cl)	vith at leas (95% Cl) ▼ ith at or V Series A: B: C:	t one 1.09 (0.99–1.18 2005–06 844 (771–917) 2005–06 909 (831–987) 886 (810–962) 863	9 1.27) (1.18–1.37 2009–10 980 (909–1,051) 2009–10 1,065 (988–1,142) 1,038 (963–1,131) 1,011

Note: Figures are rounded to two decimal points for presentation (Calculations were made with the total number). CI – confidence interval.

8.4 Results

In the BEACH data set there were 100,566 GP–patient encounters recorded in 2005–06 and 99,843 encounters in 2009–10 for which patient's age and/or sex were known (Table 8.1). The total rate of pathology ordered by GPs increased significantly from 36.6 tests/batteries per 100 encounters in 2005–06 to 45.1 per 100 in 2009–10. Similarly the proportion of encounters that involved at least one pathology test/battery increased from 16.4% in 2005–06 to 17.8% in 2009–10 (Table 8.1).

Table 8.1: Overview of pathology ordering data from the BEACH study 2005–06 and2009–10

Data element	2005–06	2009–10
Number of GP-patient encounters ^(a)	100,566	99,843
Number of pathology tests/batteries of tests ^(a)	38,855	45,062
Rate of pathology tests/batteries of tests ordered per 100 encounters (95% confidence interval)	36.6 (36.9–40.4)	45.1 (43.3–47.0)
Per cent of encounters with at least 1 pathology test/battery of tests ordered (95% confidence interval)	16.4 (15.8–17.0)	17.8 (17.2–18.4)

(a) Data for which patient's age and/or sex were not known were excluded. Therefore data do not match that reported in Box 8.1.

In 2005–06, there were an estimated 41.0 million pathology tests/batteries ordered by GPs for the Australian population. This increased significantly to 53.3 million tests/batteries in 2009–10 (Table 8.2). The growth between 2005–06 and 2009–10 in the number of pathology tests/batteries ordered by GPs for the population was due to increases in both the rate at which GPs order pathology tests (as described above) and the number of GP contacts per head of population (4.9 per person in 2005–06 and 5.2 in $2009-10^{13}$).

At least one pathology test/battery was ordered by GPs at 17.5 million GP contacts in 2005–06, and at 21.1 million GP contacts in 2009–10 (Table 8.3). This was not a statistically significant change. Although there was a statistically significant increase in the proportion of encounters with at least one pathology test/battery recorded in BEACH, when the age-sex specific rates of GP service use were included in the calculation no population-based change was apparent between 2005–06 and 2009–10.

Time period	Number pathology tests/batteries	95% LCL	95% UCL
2005–06	40,990,000	35,950,000	46,030,000
2009–10	53,310,000	47,160,000	59,460,000

 Table 8.2: Total number of pathology tests/batteries of tests ordered by GPs for the Australian population

Note: LCL - lower confidence limit; UCL - upper confidence limit.

 Table 8.3: Number of GP contacts involving the ordering of at least one pathology test/battery of tests in the Australian population

Time period	Number of GP contacts with at least one pathology test/battery	95% LCL	95% UCL
2005–06	17,460,000	15,650,000	19,270,000
2009–10	21,100,000	19,040,000	23,160,000

Note: LCL - lower confidence limit; UCL - upper confidence limit.

The average number of tests/batteries ordered by GPs per person in each 5-year age-sex cohort in 2005–06 is shown in Figure 8.1, and in 2009–10 in Figure 8.2. The age-sex specific average number of GP contacts per person involving at least one pathology test/battery in 2005–06 and 2009–10 are shown in Figures 8.3 and 8.4, respectively.

The average number of tests/batteries ordered by GPs per head of population increased with age in 2005–06 and 2009–10. Australians aged less than 15 years had an average of < 1 test/battery per person, and those aged 75 years and over had an average of 5.5 to 7.2 tests/batteries per person. There were also significant differences between males and females in the number of tests/batteries per person. GPs ordered significantly more tests for females aged between 15 and 54 years than for males, in both 2005–06 and 2009–10 (Figures 8.1 and 8.2).

The statistically significant increase in number of tests ordered for the entire population from 2005–06 to 2009–10 was not reflected in all age-sex groups. For people aged 75 years and over, the average number of tests/batteries ordered by GPs increased significantly, from 2005–06 to 2009–10: for females from 5.5 (95% CI 5.0–6.0) to 6.5 (95% CI: 6.1–7.1); and for males from 6.0 (95% CI: 5.5–6.5) to 7.2 (95% CI: 6.6–7.8). There was also a marginal increase for people aged 50–54 years: for females from 2.9 (95% CI: 2.7–3.2) to 3.6 (95% CI: 3.2–3.9); and for males from 2.1 (95% CI: 1.9–2.4) to 2.7 (95% CI: 2.4–3.0). Further, for females aged 40–44 and 45–49 years the number of tests per head increased from 2.4 (95% CI: 2.2–2.7) in

both age groups in 2005–06 to: 3.2 (95% CI: 2.9–3.5) for those aged 40–44 years; and 3.1 (95% CI: 2.8–3.3) for those aged 45–49 years in 2009–10. For females aged 5–9 years the number of tests per head increased marginally from 0.3 (95% CI: 0.2–0.4) to 0.6 (95% CI: 0.4–0.7) (results not tabled).

The average number of GP contacts per person involving at least one pathology test/battery also increased with age in 2005–06 and 2009–10 (Figures 8.3 and 8.4). Australians aged less than 20 years had an average of < 1 GP contact that involved pathology testing, and those aged 75 years and over had an average of 2.4 to 2.9 GP contacts per person that involved at least one pathology tests/battery. Between the ages of 15 and 50 years, females had more GP contacts per person that involved pathology testing than did males, in both 2005–06 and 2009–10 (Figures 8.3 and 8.4).

As reported earlier, there was no statistically significant increase in the number of GP contacts that involved at least one pathology order for the total population from 2005–06 to 2009–10. However there was an increase (between these years) in the number of GP contacts per person that involved pathology for females aged 5–9 years (0.2, 95% CI: 0.2–0.2 compared with 0.3, 95% CI: 0.3–0.4), and a marginal increase for females aged 40–44 years (1.1, 95% CI: 1.0–1.2 in 2005–06 compared with 1.3 95% CI: 1.2–1.4 in 2009–10) (results not tabled).











pathology test/battery of tests per person in the Australian population in 2009-10 (95% confidence intervals)

The projected estimated number of pathology tests/batteries that will be ordered by GPs and the estimated number of GP contacts that will involve at least one pathology order for the Australian population in the coming decades are shown in Figures 8.5 and 8.6. In each figure there are six estimates for each projected population time point.

The first two column blocks in Figure 8.5 show estimated numbers of pathology tests/batteries based on the Series A population projections. The first column block is based on 2005–06 data (referred to as 2005-based) about GP pathology ordering per person and the second column block is based on 2009–10 data (referred to as 2009-based). Series A estimates have the highest number of people in the Australian population, therefore they produce the highest estimates of future pathology ordering. From 2015 to 2050 the 2005-based Series A calculations suggest an increase of 91.0%—from 50.3 million tests/batteries ordered by GPs in 2015 to 96.2 million in 2050. Over the same period, the 2009-based Series A estimate increased by 89.0% from 59.9 million tests/batteries ordered by GPs in 2015 to 113.2 million in 2050 (Figure 8.5).

The middle two column blocks are based on the Series B population projections, the intermediate estimates of population growth. From 2015 to 2050 the 2005-based Series B calculations suggest an increase of 63.4%—from 49.9 million tests/batteries ordered by GPs in 2015 to 81.5 million in 2050. Over the same period, the 2009-based Series B estimate increased by 62.6%, from 59.3 million tests/batteries ordered by GPs in 2015 to 96.5 million in 2050 (Figure 8.5).

The last two column blocks are based on the Series C population projections, the lowest estimates of population growth. From 2015 to 2050 the 2005-based Series C calculations suggest an increase of 53.6%—from 49.5 million tests/batteries ordered by GPs in 2015 to 76.0 million in 2050. Over the same period, the 2009-based Series C estimate increased by 52.6% from 58.9 million tests/batteries ordered by GPs in 2015 to 89.8 million in 2050 (Figure 8.5).

The estimated numbers of GP contacts that involve at least one pathology test/battery in the future are shown in Figure 8.6. From 2015 to 2050 the 2005-based Series A calculations suggest an increase of 91.7%—from 21.4 million GP contacts involving at least one pathology test/battery in 2015 to 41.1 million in 2050. Over the same

period, the 2009-based Series A estimate increased by 90.6% from 23.7 million GP contacts involving at least one pathology test/battery in 2015 to 45.2 million in 2050 (Figure 8.6).

From 2015 to 2050 the 2005-based Series B calculations suggest an increase of 63.7%—from 21.2 million GP contacts involving at least one pathology test/battery in 2015 to 34.7 million in 2050. Over the same period, the 2009-based Series B estimate increased by 63.2%, from 23.5 million GP contacts involving at least one pathology test/battery in 2015 to 38.3 million in 2050 (Figure 8.6).

From 2015 to 2050 the 2005-based Series C calculations suggest an increase of 53.7%—from 21.0 million GP contacts involving at least one pathology test/battery in 2015 to 32.3 million in 2050. Over the same period, the 2009-based Series C estimate increased by 53.1% from 23.3 million GP contacts involving at least one pathology test/battery in 2015 to 35.6 million in 2050 (Figure 8.6).

When comparing the 2005-based and 2009-based results presented in Figures 8.5 and 8.6, there were no statistically significant differences between projections when comparing equivalent time points. For example, the projections for 2015 of the estimated number of pathology tests/batteries that will be ordered and the estimated number of GP contacts that will involve at least one pathology order, based on 2005–06 data were not significantly different to those based on 2009–10 data.





Figure 8.6: Estimated number (millions) of GP contacts involving at least one pathology test/battery of tests for the projected Australian population 2015 to 2050 based on per head GP service use and GPs' pathology ordering in 2005–06 and 2009–10 (95% confidence intervals)

8.5 Discussion

The results presented in this chapter demonstrate that per capita, pathology generated by GP orders increases with age in the Australian population. This echoes the pattern of increased per capita health resource use, and total pathology service use associated with increased age.^{27,234,397}

The volume of pathology generated by GP orders and the number of occasions of testing for the population are projected to grow faster than would be expected with population growth alone. The same result is found for all series of projections and regardless of whether the calculations are based on 2005–06 or 2009–10 rates of GP pathology ordering and GP utilisation. This is because the proportion of older people in the population is projected to increase¹³⁷ and, as age increases, so too does per capita GP pathology ordering. Therefore the excess growth (above that generated by population growth) is due to the effect of the ageing population.

There are limitations that need to be kept in mind when considering the long-term projections made in this chapter. The projections of GPs' pathology ordering are reliant on the robustness of the three types of national data (population projections, GPs' pathology ordering and GP services) used.

There is inherent uncertainty in estimating population projections. Projections were made by the ABS based on assumptions about life expectancy, fertility rates and migration rates. An unanticipated change in any of these factors would render the population projections unreliable. This uncertainty is why three series of population projections are published and why I have used these three series to project GPs' pathology ordering.

The number of pathology tests/batteries recorded in the BEACH study is limited by the space available on the recording form. A maximum of five tests/batteries can be recorded per encounter. This cap may lead to an underestimation in the projections of the number of pathology tests/batteries that will be ordered by GPs. This limitation does not affect the projected number of GP contacts that involve at least one pathology order because it is based on the proportion of encounters that involve at least one pathology test/battery, not on a count of tests/batteries.

Pathology ordering recorded by GPs (in BEACH) reflects their intention that the patient will be tested. Inevitably a proportion of pathology ordered by GPs will never

be conducted, because the patient chooses not to be tested. Therefore the projections made in this chapter may overestimate the pathology testing that will be performed in the population in response to GPs' ordering.

The number of GP services provided to the community in 2005–06 and 2009–10 were based on services funded through Medicare and the DVA. As the 2009–10 DVA data were not available, the 2005–06 DVA data and 2009–10 Medicare data were combined to calculate the number of GP services per person in 2009–10. The number of GP services funded through Medicare (per capita) increased over the period of this investigation (2005–06 to 2009–10).¹³ As the total per capita use of DVA funded medical services (including GP, specialist and dental services) increased over this period,^{400,401} it is likely that DVA funded GP services (when viewed alone) also increased. Given these increases, the lack of 2009–10 DVA data may cause an underestimation of the number of GP services per person in 2009–10, particularly as most DVA funded GP services are for veterans' aged 80 years and over and GP service use increases with age.¹³⁶ In turn this means that the pathology projections based on 2009–10 data may be underestimated.

The projections created in this chapter were based on 2005–06 and 2009–10 data because over this time there were significant increases in: the number of GP services per capita;¹³ the rate of GPs' pathology ordering at encounters; and the proportion of GP encounters that involved at least one pathology test. Despite these increases, when the projections were calculated there were no statistically significant differences in the pathology projections based on 2005–06 data versus those based on 2009–10 data (when comparing equivalent projection time points). Theoretically it would be more reliable to use the 2005–06 based estimates because of the lack of 2009–10 DVA data. However the pathology projections based on 2009–10 were consistently above those based on 2005–06 data and (while not statistically significant) this has economic ramifications for funding pathology services. Therefore the 2009–10 projections are the focus of the discussion below.

The estimated projected number of pathology tests/batteries that will be ordered by GPs for the Australian population (based on 2009–10 data) was 58.9–59.9 million tests/batteries in 2015 and 89.8–113.2 million in 2050—an increase of 53–89% from 2015 to 2050. The estimated number of GP contacts that will involve at least

one pathology order for the Australian population (based on 2009–10 data) was 23.3–23.7 million in 2015 and 35.6–45.2 million in 2050—an increase of 53–91% over this period.

These projections assume that per head in each age-sex group the number of GP visits stays constant and the rate and likelihood of GPs' ordering pathology stays constant. For example (referring to Box 8.3), these assumptions mean that an average 40-44 year old female would have the same number of GP visits (5.4 per head), pathology tests/batteries (3.2) and GP visits involving pathology testing (1.3) in 2009–10, as a 40–44 year old female will have in 2015, 2020...and so on to 2050. The current age-sex-specific pattern of pathology ordered by GPs for the population has potential to change considerably over the coming decades. In order for pathology rates to stay constant in the age-sex cohorts of the population in the future and therefore for the projections of future pathology ordering to be correct, the use of GP services, and GPs' pathology ordering within the age-sex groups would need to stay constant. At a minimum, this would require prevalence of diseases and patterns of multiple morbidities in any one age-sex cohort of the population to remain constant in the future. Knowledge, guidance and rules about the role of pathology testing in the management of these morbidities would also need to be constant. However, in reality, changes within these variables will be inevitable over the coming decades because there are many factors that could create change. Some examples are given below.

The prevalence of overweight/obesity may change and consequently generate changes in GP service use and GPs' pathology ordering. The proportion of Australian adults who are overweight/obese has increased from 54% in 2004–05 to 62% in 2007–08.^{154,402} High body mass is a risk factor for a number of chronic conditions including (but not limited to) dyslipidaemia and diabetes mellitus. Both of these conditions are usually managed by GPs and require pathology for diagnosis and monitoring. If the proportion of the population who are overweight/obese were to increase (as suggested by the historical trend) it is likely that this would generate a subsequent increase in pathology ordered by GPs. Alternatively if the ongoing public health initiatives^{295,296} that aim to reduce the prevalence of overweight/obesity are successful it is likely that there would be a subsequent reduction in pathology testing.

Significant changes in the extent to which GPs used pathology tests in the management of selected conditions were demonstrated from 2000–02 to 2006–08 in Chapter 4. Changes in what we know about a condition, and the best way to treat it (such as, new scientific knowledge, technological developments including new tests, and new medications), contribute to changes in the role of pathology in its management. These changes have the potential to change the management of conditions and are considered by both the OECD and the Australian Government to be a likely source of future growth in health costs.^{234,397} For example, genetic testing has been highlighted as a potential area of growth during the Government's recent review of pathology funding, due to the rapid developments occurring in the genetics discipline.³⁹⁹

Alternatively technological advances may reduce the need for pathology testing in the future. An example is the development of a vaccine for the human papillomavirus (HPV) and the subsequent vaccination program that was implemented in Australia in 2007.⁴⁰³ The vaccine prevents the types of HPV that cause 70% of cervical cancer⁴⁰³ and therefore it is likely to reduce the incidence of cervical cancer. Australia's comprehensive national program for cervical cancer screening currently recommends screening with a Pap smear every 2 years for women aged 18 to 70 years, but this screening protocol is under review due to the introduction of the HPV vaccine, availability of new tests for screening, and emergence of evidence about the screening age and interval.⁴⁰⁴ It is expected that the screening interval will be increased and the age-groups covered in the program reduced.⁴⁰⁴ Consequently, the ordering of Pap smears, one of the most common tests ordered by GPs (see Chapter 4), may decrease.

If the likelihood and rate of GPs' pathology ordering stay constant, projected population growth and ageing will produce substantial future growth in the number of pathology tests/batteries and number of GP contacts involving pathology in the population. However, if the historic growth in the likelihood and rate of GPs' pathology ordering continued in the future, then any projected estimates (based on 2009–10 data) of GP-ordered pathology in the population would be too low. Governments internationally face uncertainties in estimating future health costs as changes in population, the prevalence of conditions and technological advances are

difficult to accurately predict. While it has limitations, basing projections on past patterns of health resource use is the best tool available to plan for future requirements. The projections of pathology in this chapter were made to 2050 as the growth in the proportion of the population in older age groups is expected to continue over the next four decades.³⁹⁵ In addition, the sustainability of health spending to 2050 is the Australian Government's current focus.²³⁴

The Australian Government has estimated that public health spending will increase by 77.5%, from 4.0% of gross domestic product in 2009–10 to 7.0% in 2049–50.²³⁴ Forty per cent of this expected growth is due to population growth and ageing (i.e. demographic changes).²³⁴ The population projection used by the Government to predict this growth lies between the Series A and Series B population projections published by the ABS. In this chapter, the estimated growth in the volume of pathology tests/batteries ordered by GPs and the GP contacts that involve pathology from 2009 to 2050 is approximately 80% based on Series B population projections and 110% based on Series A projections. These pathology projections are based only on demographic changes (i.e. population growth and ageing). In contrast, the Government's estimate of future health spending also incorporates growth in nondemographic health spending (such as increased use of doctors, tests and medicines).²³⁴

In this chapter, even without attempting to estimate the affect of non-demographic factors, growth in GPs' projected pathology ordering is considerably larger that the Government's expected growth in total health spending over the same timeframe. Comparing these estimates suggests that pathology generated by GP orders is likely to grow faster than total health care spending. Whether this means that the cost of pathology (ordered by GPs) will grow at the same rate as the projected pathology ordering is another matter. As the majority of GPs' pathology ordering is funded through Medicare, the magnitude of cost growth will be affected by the Medicare funding rules, in particular the episode coning rule which restricts payment to the three most expensive pathology items (as described in Chapter 2). Therefore the Medicare cost of funding pathology tests ordered by GPs only partially reflects the tests ordered by GPs. It is not possible to replicate the effect of the episode cone using the data presented in this chapter. However, it is reasonable to conclude that projected growth in GPs pathology ordering is likely to produce growth in Medicare

299

pathology outlays, but under the current Medicare funding structure the growth in cost and growth in GPs' ordering will not be equivalent.

The review of Medicare funding arrangements for MBS pathology services, undertaken by the Australian Government (underway when the study in this chapter was conceptualised and data analysed), was completed in early 2011. The outcome was that the funding arrangements remained largely unchanged. A new funding agreement between the pathology industry and the Australian Government covering July 2011 to June 2016 was introduced. This covered Medicare outlays and growth in these outlays from pathology services ordered by all requestors (i.e. GPs and other specialist doctors),²⁵ in a similar arrangement to the three previous MoUs that were in operation from 1996 to 2009 (the Government chose not to renew the previous MoU pending the outcome of this review). The new funding agreement includes a fixed annual range of funding growth, between a minimum of 4.4% and maximum of 5.2% each year (subject to review).²⁵

During the period of the Government's review, a pathology industry group, the Australian Association of Pathology Practices (AAPP) commissioned an analysis of the expected future growth in outlays for MBS pathology services. This analysis estimated an average annual growth rate in outlays of 6.2% (a range of 3.1% to 7.0%) over the five years 2010–11 to 2014–15, based on historical growth in per person pathology usage, costs of services, and projected growth and ageing of the population.²⁷ The annual growth in outlays attributed to the population was 1.4–1.5%. In comparison, I estimated a growth of between 2.1% and 2.5% per annum in the volume of pathology tests/batteries generated by GPs' orders, and number of GP contacts that involve pathology testing over the 5 years from 2009–10 to 2015 based on population changes. The AAPP-commissioned report differs from the current study as it focussed on the growth in cost of pathology services generated by orders from GP and other medical specialists whereas I estimated the growth in volume and occasions of pathology testing generated from GP orders. However, my study may indicate a greater influence of population changes than suggested in the AAPP-commissioned report.

It is likely that due to demographic changes alone there will be upward pressure on Medicare outlays for pathology services generated by GPs' pathology ordering, suggesting that the Australian Government's concern about future growth in the cost of pathology services is justified. Based only on the expected demographic changes in the population, future growth in GPs' pathology ordering is inevitable. However if the historical trends of increased per capita use of GP services and increased number and likelihood of GP pathology ordering continue, they will compound the impact of these demographic changes. Together these factors will put significant upward pressure on future costs of pathology services.
9 Discussion

Major findings

This research provides a detailed picture of GPs' pathology ordering and how it has changed from 2000 to 2008. The total increase in volume of GPs' pathology ordering between 2000–02 and 2006–08 was influenced by increases in three factors, each of which can operate independently: the rate of pathology ordered in the management of clinical problems (due to increased likelihood of ordering and increased number of tests ordered per testing occasion), the number of problems managed at GP–patient encounters, and the total number of encounters provided to the Australian population. The majority of the volume and growth in GPs' pathology ordering was attributable to a relatively small group of pathology tests and clinical problems. Just 22 problems accounted for 59% of the growth in testing between 2000–02 and 2006–08.

Variance among GPs' in the volume of pathology ordered is often used as an indicator of poor quality. In investigating factors that contribute to variance, I was able to explain 50% of the variance in GPs' testing rates. Significant independent predictors of volume included some GP and practice characteristics, but the strongest determinate was the type of problem being managed.

In this thesis, appropriateness of GPs' pathology ordering for six selected problems was assessed by measuring alignment with recommendations in guidance documents. Overall GPs' ordering aligned well, and increases in ordering over time for the six problems reflected increases in both 'appropriate' and 'inappropriate' ordering. I found no evidence to support concerns raised in the literature about widespread inappropriate ordering, or assertions that increases in ordering reflect disproportionate increases in inappropriate ordering.

Based on the growth and ageing of the population alone, future growth in the volume of GPs' pathology ordering is inevitable. The ageing of the population will increase the prevalence of diagnosed chronic disease and therefore increase its management and associated pathology testing. During my review of guidance documents I found that pathology test recommendations for long-term monitoring of chronic diseases were poor. Improving this guidance would support GPs in the appropriate ordering of pathology tests in this high growth area.

Limitations

As with all observational studies there are limitations in the studies described here. The BEACH data used in this thesis are cross-sectional and encounter-based. As such, they are affected by the frequency with which patients see their GP. I have no way of knowing how often individual patients at the sampled encounters return to see a GP. Attendance patterns vary according to patient age and sex.^{136,369} Distance from care and the GP:population ratio which affects availability of GP appointments also have a bearing, as the management provided by GPs may be affected by the patient's access to care. For example, the likelihood of a GP ordering pathology at an encounter may be higher for a patient who has not seen a GP for some time than for one who is seen regularly.

The encounter data presented in this thesis will be affected by attendance patterns but the extent of this affect varies. In Chapters 4–6 there is minimal impact on the conclusions drawn as the encounter-based data are used to describe the grouped clinical activity of GPs. However, attendance patterns have an influence when activity is compared for individual GPs as in the investigation of variance in GPs' pathology ordering (Chapter 7). For example, the finding that GPs in rural areas ordered more pathology than their urban counterparts is likely to reflect differences in attendance patterns and a need to go further down the diagnostic pathway before referring the patient to specialist care as access to care is more limited.

The study predicting future growth in GPs' pathology ordering (Chapter 8) is the only work in this thesis that incorporates the affect of attendance patterns. However, this was done using population age-sex-specific average attendance rates, as data about individual attendance patterns were not available.

There are many factors for which I lacked information that may have a bearing on a GP's decision to order pathology testing. These include: presence and duration of symptoms, presence of comorbidities not managed at the encounter, past medical history, family history, social circumstances, previous test results, and risk factors. As discussed in Chapter 5, the presence of these factors may have an influence on the accuracy of my assessment of appropriateness of GPs' pathology ordering. My assessment was based on the pathology test linked to the problem under management at the encounter for only six selected problems. Even if more detailed data were available, it would be very difficult to design an exact measure of appropriateness

that reflects the complexity of GPs' decision-making processes. The data set used in this thesis is one of few that links GPs' test selection to the clinical problem for which it was ordered. So while there are some limitations to the method I have adopted to assess the appropriateness of GPs' pathology ordering, my study is one of the few that incorporates the clinical problem.

The variables available in the study also have a bearing on the study of inter-GP variance in rates of pathology ordering (Chapter 7). I adjusted for the available GP, patient and encounter variables, but there may be other factors that have a bearing on this variance for which I have no measure.

As discussed in Chapter 4, the maximum of five pathology tests/batteries of tests that can be recorded per encounter has implications for the results presented in this thesis. Over time the proportion of encounters where the maximum number of pathology tests was recorded increased, suggesting that the proportion being missed may have increased over the years investigated.

Further, the maximum of five tests/batteries may have a greater bearing on some types of clinical problems managed at encounters. For example, a problem for which many tests are usually required on an occasion of testing is more likely to have missing data than a problem for which only a single test is required. This selection bias has implications for some of my results. In the study assessing the appropriateness of GPs' ordering, I could only make a judgement based on the tests recorded for each problem.

The large number of comparisons made in this thesis (in particular when investigating changes over time) provides the potential for reporting a difference between groups when none exist (i.e. a Type 1 error). There may be occasions where a statistically significant difference has been detected by chance rather than because it exists in reality. However, I used non-overlapping 95% CIs as the primary measure of significance (which is considered a conservative approach¹³⁰⁻¹³²) to reduce the likelihood of such error.

Appropriateness of GPs' pathology ordering

There is a negative narrative present throughout the literature regarding clinicians' pathology testing, whereby authors link (implicitly or explicitly) the increasing rates of pathology testing with inappropriate ordering, leading to a prevailing opinion that

the quality of clinicians ordering is poor or could be improved considerably. There is limited evidence to support this assumption. The landmark review of appropriateness of pathology ordering by van Walraven and Naylor found a wide range of 'inappropriate' ordering but suggested that the results were not generalisable (most studies having been conducted in inpatient settings) and that "allusions to extensive inappropriate ordering should not be made without appropriate qualifiers."⁵² While there has been some evaluation of clinician's pathology ordering in other health settings, there has been virtually none in general practice, and yet the same negative perceptions exist.

In the research reported in this thesis, I evaluated the appropriateness of GPs' pathology ordering for six problems: hypertension, Type 2 diabetes, lipid disorders, weakness/tiredness and health checks. Together these problems generated a quarter of pathology ordered at GP-patient encounters, and of the tests ordered 81% were appropriate, 13% were inappropriate and 6% could not be evaluated. GPs' ordering for these problems increased over time, and this was reflected in increases in tests in all levels of appropriateness (i.e. appropriate, inappropriate and unevaluated tests). These six problems were used as indicator conditions for the quality of GPs' pathology ordering, and suggest that GPs in Australia are not ordering a large proportion of pathology tests inappropriately, and that the increases were not driven by increases in inappropriate ordering, despite conjecture throughout the literature. When making reference to GPs' inappropriate ordering, other authors often refer to studies that report interventions aiming to improve appropriateness of GPs' pathology ordering. As I discussed in Chapter 2, the use of interventional studies to describe appropriateness is flawed because such studies demonstrate the success (or failure) of the interventions, rather than the appropriateness of GPs' ordering. These studies are designed to improve the appropriateness of ordering in a specific area in which GPs' pathology ordering is known or likely to be inappropriate. It is implied that preceding the intervention, an evaluation step of some kind identified the area to be targeted on the basis of the need for improvement. Unfortunately such evaluation is rarely reported.

Further, the success of an intervention (usually measured as a reduction in testing) is frequently assumed to reflect an improvement in the appropriateness of GPs'

ordering but this is rarely assessed. In many cases a lack of clinical data linked to testing data prevents assessment of appropriateness. I therefore conclude that interventional studies cannot act as a proxy for evaluation of appropriateness of GPs' pathology ordering.

The approach I took in this thesis to evaluate appropriateness of GPs pathology, is fundamentally different. The six problems for which GPs' test ordering was evaluated were selected on the basis of past increases in GPs' ordering in their management, and the high volume of pathology testing generated by the management of these problems in general practice. I made no prior assumption about the need for improvement in the ordering of pathology. This approach enabled me to conclude that (in the majority of cases) GPs' selection of tests in the investigated problems reflected appropriate pathology ordering. This is a substantially different conclusion to that reached by other authors. To the best of my knowledge, this study is the first to provide an assessment not influenced by the presence of a concomitant intervention, of the appropriateness of GPs' pathology test selection in the management of the selected problems.

While overall the majority of ordering was appropriate, there were differences among the six problems investigated. For four problems alignment was very high, with at least 85% of tests deemed appropriate, for overweight/obesity problems (in adult patients) alignment was moderate (72% of tests supported), but for 'health check' problems (in patients aged 15 years and above) alignment was poor, with only half the tests ordered being supported.

GPs' ordering for 'health checks' was assessed against guidance for preventive care. As discussed in Chapter 5, some of the discord between GPs' activity and recommended testing may be due to GPs' responding to the needs of the individual patient. Recommendations in preventive care guidance may reflect population health benefit, whereas GPs' provision of health checks will be adapted to the individual patient's benefit. Further research is needed to determine whether the lower level of alignment in GPs' management of 'health check' reflects appropriate individualised care or inappropriate ordering. This is particularly important if the Government continues to use incentives to encourage GPs' to provide specific health assessments, as it has done in the past.^{18,232,236} These types of assessments (often labelled by GPs as health checks) are one way in which the Government aimed to "encourage GPs to deliver quality preventative health"³⁴⁴ and in particular were targeted to prevent and detect chronic disease. Their use continues to be advocated by some peak bodies and researchers.^{405,406} My work suggests that GPs' may need further support regarding the selection of appropriate pathology tests when performing health checks.

Pathology recommendations in guidance documents

My evaluation of the appropriateness of GPs' pathology ordering was based on recommendations for pathology testing in guidance documents. In reviewing these documents I found that the accessibility of pathology recommendations and the content of recommendations need considerable improvement.

Accessibility issues related to accessing pathology recommendations within documents, and included: the sheer size of documents, many of which were not designed for GPs; lack of a central body in Australia responsible for creating and structuring guidelines in a consistent format; and mixed terminology used to refer to testing (e.g. 'diagnostic testing', 'laboratory investigations', 'assessment', the specific test name, the disease that testing aims to identify or exclude). I believe the international adoption of standardised terminology to refer to pathology testing in guidelines could have considerable impact on accessibility. It could be achieved with minimal effort and is worthwhile promoting.

Content issues identified in guidelines pertain to missing or incomplete guidance. Primarily these deficiencies related to recommendations for testing in long-term management of chronic disease, including incomplete guidance about what tests were needed for monitoring and the recommended interval between repeated testing, and lack of guidance about the interpretation of repeated test results. The latter refers to the critical difference between results, and describes the expected analytical and intra-individual variance in test results.²⁰¹ Such knowledge is essential for GPs to correctly interpret results to determine the level at which clinical intervention is required to maintain 'control' and avoid progression of disease.³²⁸

The majority of pathology recommendations in guidance documents refer to the decision to order testing and the types of tests ordered and so support GPs' decisions in the pre-analytical phase of testing. In contrast, guidance regarding test results in the post-analytical phase of testing relates primarily to the diagnosis of differentiated

disease. There is considerable scope for improvements in guidance for this phase of testing in regard to interpretation of abnormal results in undifferentiated disease and interpretation of monitoring tests. As I discussed in Chapter 5, providing guidance about interpretation of monitoring results is complex as critical difference may vary between patients. This is an area in which pathologists give ongoing high quality support to GPs and other referring clinicians for individual patients, but it could be supplemented by producing guidance for interpretation of common monitoring tests. As discussed in Chapter 5, the poor quality of recommendations in guidance documents for monitoring may also be due to the lack of evidence about ideal monitoring protocols in health care.^{56,329} Despite this, the volume of monitoring undertaken by GPs for chronic disease, and the expected increase in prevalence and management of these diseases due to our ageing population, leads me to recommend that monitoring guidance be improved. A pragmatic approach incorporating consensus views may be needed where evidence is not available.

Developing guidance about monitoring (as I have recommended) and supporting GPs' in applying recommended monitoring protocols may, or may not, change GPs' current monitoring practices. There are limited data about these monitoring practices. The study reported in Chapter 6 gives some preliminary information about GPs' monitoring of lipid tests, but further research is needed to understand GPs' current monitoring intervals and interpretation of repeated results. Such research should evaluate whether monitoring influences patients' social, cognitive or behavioural responses⁶³ (for example, whether monitoring affects medication adherence) as these factors will contribute to a GP's decision to order testing. Even if no change is needed to current practise, supporting GPs through improved guidance is still valuable as it supports quality of care.

Two recent national programs have the potential to address the accessibility and content issues I identified in the existing guidance. The first of these is the National Prescribing Service medical test program announced in the 2009–10 budget to promote evidence-based ordering of imaging and pathology tests. At the time this thesis was submitted (March 2013), programs had been initiated for only a small number of topic areas involving pathology testing. This program will primarily

involve education-based interventions, and at the time of the submission of this thesis there were no results available from the outcomes of the program.

The second national program is the development of pathology guidelines for GPs to be implemented as decision support tools within clinical software (announced as part of the new funding agreement between Government and the pathology industry, July 2011²⁵). There are potential benefits and pitfalls of this approach that bear discussion in regard to the results presented in this thesis.

It appears that this approach will result in the development of 'stand alone' pathology guidelines. GPs have previously indicated that they do not want pathology guidance to be produced in isolation as this does not assist the integration of pathology ordering and interpretation of results in their clinical management of problems.⁴⁰⁷ For example, the diagnostic process of undifferentiated disease involves more than the selection of appropriate pathology tests. Gialamas et al. suggested that a decision support tool for the management of presentations of tiredness would need to incorporate: patient age, sex, a psychological evaluation (to assess presence of depression), a physical examination (e.g. to assess presence of anaemia) as well as advice about appropriate tests.⁴⁰⁸ Other authors have suggested additional factors (such as the patient's medical history, duration of symptoms) that may also have a bearing on the decision.^{281,283,285} Therefore, the role of pathology testing in relation to other factors associated with the diagnostic process needs to be considered. The way in which the guidelines are implemented into a decision support tool and the associated level of integration may address this concern.

It is unclear how the proposed Australian decision support tools will be designed and integrated, but results of the few international examples of similar decision support tools,^{76,79,86} suggest these factors will have an impact on the efficacy of the tool. For example, existing pathology decision support tools are 'launched' once the GP opens the pathology ordering section of the electronic medical record, and provide guidance on test selection at this point. Such an approach is unlikely to affect the decision to (or not to) order because GPs' have already decided to order before any advice is presented. I have demonstrated that the increase in occasions of testing was a feature of past growth in GPs' pathology ordering (see discussion below). This type of decision support design would only support appropriate test selection in GPs'

clinical management—only one aspect of GPs' pathology testing behaviour. Further my results show the appropriateness of GPs' ordering is already high. Therefore, there may be little or no reduction in volume or cost of pathology testing achieved through introduction of this type of decision support tool.

Volume and growth of GPs' pathology ordering

The data presented in this thesis provides the first in-depth picture of GPs' pathology ordering in Australia in more than a decade. Other national pathology data (such as that collected by laboratories and Medicare Australia) cannot provide the same level of detail about GPs' ordering as that collected in the BEACH study. In particular, the complex Medicare funding structure (including the episode coning rule that restricts payment to the three most expensive MBS pathology items per episode of GP ordering, and the iso-resource grouping of some items) limits the ability of Medicare data to accurately reflect GPs' pathology ordering.

My investigation of growth in the total volume of GPs' pathology ordering from 2000–02 to 2006–08 demonstrated that three factors contributed to the growth, each of which can operate as a separate mechanism and all of which increased over the study period. These were: increased number of national GP encounters provided to the population; increased number of problems managed at GP–patient encounters; and increased rate of GPs' pathology ordering in the management of problems. The latter was due to significant increases in the likelihood of at least one pathology test being ordered (i.e. more occasions of testing), and the number of pathology tests ordered per problem once the decision to order was made (i.e. more tests per order) —these increases reflect change in the GPs' pathology ordering behaviour.

As I discussed in Chapter 4, there are many potential drivers of change in these three factors and each driver can act on one or more of the factors. Examples include: changes in the population demographics (primarily the ageing population); introduction of population health policies and initiatives; publication of new guidelines or changes to existing recommendations in guidelines; change in disease prevalence; and emergence of new evidence. In addition to their potential to influence one or more factors, they also interact among themselves. For example, new evidence may have an affect on guidelines and health policy, and the ageing

population may influence health policy and changing disease prevalence. This illustrates the complexity underlying growth, as discussed by several authors.^{9,26-30} Describing the growth in GPs' ordering in terms of the three factors highlighted in my research takes a somewhat reductionist approach. However its strength is that it demonstrates that change in GPs' ordering behaviour is only partially responsible for the total growth in GPs' pathology ordering—a finding that is crucial in the context of much of the conjecture and discussion in the literature. There is a perception that the increase in pathology ordering reflects a deliberate management action by GPs to order more pathology testing. In reality the 'choice' to order as represented by change in ordering behaviour is just one factor.

Similarly, in the context of an individual clinical problem, change in GPs' pathology ordering behaviour is only one possible factor generating growth in the volume of pathology ordered for its management. Therefore, growth can still occur without change in GPs' ordering behaviour. Further, where change occurs, it may be an increase in only one of the two measures of behaviour change—occasions of testing or tests per order. It is important to distinguish these as there may be different factors contributing to growth in each type of ordering behaviour. Also an increase in occasions of testing represents a higher cost burden than a decision to order additional tests.^{133,134}

While the discussion above largely focuses on factors contributing to growth over time, the same construct can be applied to demonstrate that GPs' pathology ordering at a single time point is only partially determined by GPs' pathology ordering behaviour. This holds true whether investigating the total volume of pathology generated by GPs' orders or the pathology generated in GPs' management of an individual clinical problem.

Demonstrating the separate contribution of GPs' ordering behaviour to the overall volume and growth of pathology testing has a bearing on the discussion of the appropriateness of ordering. This is because the judgement of whether a clinician's ordering is 'appropriate' or 'inappropriate' can only be a reflection of the appropriateness of their behaviour in terms of making the 'correct' decisions in initiating testing and/or ordering certain types of tests. Therefore, when inappropriate

ordering is present and there is potential to 'correct' this ordering it will only have an impact on the proportion of ordering that is driven by GPs' ordering behaviour. I found that the majority of the volume of GPs' pathology testing is confined to a relatively small group of tests and clinical problems. Other authors have previously noted this concentration of ordering.^{79,82,89} In investigating growth in GPs' ordering over time, I also demonstrated that these same tests and problems contribute to most of the growth that has occurred. In theory, targeting these tests and/or problems with interventions has the potential to have a significant impact on the volume of pathology tests ordered by GPs. However, my research suggests that there are not many situations where the quality of GPs' ordering is measurably poor, and therefore not many situations that could be targeted. One of the few examples is FBC testing in preventive care.

FBC tests are not recommended in preventive care guidelines (Chapter 5), but GPs' indicated that 8% of those ordered were for primary prevention (Chapter 6). The introduction of a successful intervention aiming to reduce this testing has the potential to have a significant impact on the volume of testing. Even if only a minor reduction in the ordering rate is achieved the impact on test ordering would still be significant because FBC tests account for the highest volume of tests ordered by GPs.

Hypothetically this type of intervention would be worthwhile, however the potential benefits in terms of quality of care and cost-benefit also need to be considered. For example, determining whether reducing FBC testing for prevention will improve quality of care is difficult. There is no evidence for its use in primary prevention, but FBC testing is unlikely to generate harm to patients and the benefit of testing to an individual patient's care is not known. Therefore, GPs' may not perceive a reduction of testing as of benefit to patient care, yet this perception is an important element in achieving change in testing behaviour.⁴⁰⁹

Further, the current Medicare rules (in particular the coning rule) for funding GPordered tests mean that a reduction in volume of testing may not translate into a reduction in the cost to Medicare. Due to coning, only 45–55% of GP-ordered FBCs performed by pathologists were funded by Medicare,^{19,346} so reducing the volume of GPs' FBC ordering (as discussed in the example above) may not achieve any cost reduction. For tests that are 'coned out' and therefore do not attract Medicare reimbursement, reducing testing will generate economic savings in terms of cost of reagents, labour etc., although these savings will not be reflected in public spending. It therefore seems that any interventions that aim to improve quality of GPs' pathology ordering and at the same time to reduce cost are unlikely to be successful under the current funding structure.

The growth in volume and cost of pathology testing has meant that the focus of governments and researchers has been on understanding the growth and managing the demand. As a result, the quality of GPs' pathology ordering as a whole has not been evaluated, including areas of overuse and underuse of pathology testing. A review of pathology services in England suggested that a significant proportion of testing (estimated as up to 25%) was inappropriate or unnecessary but that there was probably a similar amount of under-requesting.⁴¹⁰ This statement was based on the opinions of participants in the review about testing across the entire health sector rather than on quantitative evidence, yet it does highlight the possibility of significant amounts of under-utilisation. In my assessment of appropriateness of ordering, there were few recommended tests that were not ordered by GPs. However, this reflects a grouped measure of test appropriateness. In many cases under-utilisation may relate to patient-based measures such as inadequate monitoring of HbA1c over time in patients with diabetes. Further research to evaluate the extent of under-utilisation of GP pathology testing in individual patient care is needed.

The need to contain the increasing cost of pathology testing is understandable. Health budgets are limited and there is pressure to ensure maximum benefit for each dollar spent (or as the Australian Government describes it) ensuring that "spending on health is sustainable, affordable and provides maximum benefit to the greatest number of people."²⁰ This reflects the need for efficient health spending and implies the need to consider health care spending as a whole, but there is a tendency (due to 'siloing' of health budgets) to consider the pathology budget in isolation. While quality use of pathology services by ordering physicians has been and remains a core objective of the Australian Government (as reflected in the maintenance of the Quality Use of Pathology Program, and recently funded programs to promote evidence-based ordering and development of decision support tools), national policies have not attempted to reduce demand by ordering clinicians. Historically, the

Government has contained the pathology budget administratively through funding mechanisms, which has resulted in the current complex funding system.

There is an inherent conflict between population health strategies introduced by the Government and the objective of containing pathology spending. Many public health initiatives represent efficient health spending, as they have a net benefit to the community. However they can have a cost in terms of pathology testing. For example, the finding of the population-based AusDiab study that one in two Australians with diabetes were undiagnosed,¹⁵⁷ prompted many health initiatives to improve detection of diabetes, and therefore increased testing costs associated with its diagnosis and monitoring. It is thought that the health benefits of such initiatives (e.g. early detection of chronic disease, prevention of complications) more than outweigh the cost.⁴⁰⁵ While there are situations where logical links between pathology testing 'costs' and outcomes can be made, there are many more where evidence between pathology testing and outcomes is lacking. This has prompted the call for an outcomes-based research agenda.^{42,57} However, producing evidence linking testing to outcomes is complex and the development of methodology to do so is an area of current research.⁷⁰ While linking testing to outcomes would be ideal, in reality such evidence is many years away. It will be an ongoing challenge to find the optimal level of testing because while not enough testing will result in more 'flow on' costs in other sectors of the health system, there is a point at which too much testing will have no additional benefit.

Variance in GPs' ordering

The presence of considerable variance among GPs' in their use of pathology testing has been noted by several authors.^{43-51,71-73,82,337} This variance is discussed as an indicator of poor quality, the rationale being that in the same clinical situation, clinicians should be homogeneous in their use of pathology testing, and this would result in low levels of variation. I similarly found variance among GPs in their pathology ordering at encounters. However, my investigation of this variance suggests that it is not a good indicator of quality of GPs' pathology ordering. The amount of variance I was able to explain in the analysis (50%) was much higher than that explained by other researchers (10–30%). This is partially because (in most cases) others did not have information about the clinical problems for which GPs

ordered pathology tests and therefore were limited in their ability to assess the influence of the clinical situation on the variance. The GP's clinical workload, in particular the type of clinical problem under management at GP encounters, was the strongest determinate of pathology testing. However, I may have underestimated the influence of the clinical problem on variance as I used the ICPC-2 chapter to represent the clinical problem in the modelling. This chapter represents the body system that the clinical problem is classified to; for example, diabetes and lipid disorders are both classified to the 'endocrine, metabolic and nutritional' chapter. If the multivariate analysis had included the individual clinical problems (not just its ICPC-2 chapter), more variance may have been explained.

A number of other GP, practice and patient variables were found to be independent predictors of pathology ordering. These accounted for far less of the variance than did clinical problems. Some are not surprising such as lower testing for young patients aged < 5 years, and higher testing for new patients. However, others such as the higher ordering rates of young GPs and female GPs raise concerns (as discussed later) and may need to be further investigated.

Variation in response to patient-centred care represents higher quality care.⁴¹¹⁻⁴¹⁴ Therefore, the finding that the clinical problem under management was the primary factor driving variance can be considered an indicator of good quality care. The variables available in my analysis (i.e. those recorded in the BEACH study) only 'scratch the surface' in terms of describing the clinical situation that may generate 'appropriate' variance. GPs provide holistic care to their patients and their decision to order pathology in the management of the patient will incorporate consideration of biomedical and psychosocial factors. These types of drivers of patient-centred care are difficult to measure.

As discussed in Chapter 7, the inability of other researchers to incorporate the clinical problem in their models limits their ability to adequately assess 'appropriate variance'. Without adequate data to differentiate, researchers are effectively assuming that GPs' are homogeneous in the type of patients they see and care they provide to patients—we know this is not the case.³⁸² However, there may be specific clinical circumstances in which variance among GPs' pathology ordering reflects poor quality. Assessment of variance in such circumstances should reflect a specific

subset of pathology data; rather than variation in GPs' total ordering—the main measure used to date. I believe it is reasonable to conclude that a large proportion of the variance in GPs' total pathology ordering reflects patient-centred care and therefore quality care. It is misleading to assess variance of GPs' ordering without adjusting for the clinical workload of the GP, particularly the problem under management. I conclude that variance in total volume of pathology ordering is too blunt to be considered a useful measure of the quality of GPs' pathology ordering.

Predicted future growth in GPs' ordering

My research confirms that the growth and ageing of Australia's population will contribute to ongoing growth in volume of pathology tests ordered by GPs to at least 2050. Over the 5 years from 2009–10 to 2015, I estimated that the volume of GPs' pathology ordering will grow between 2.1% and 2.5% per annum purely because of these expected demographic changes.

The three factors (discussed earlier) that contributed to past growth in GPs' ordering behaviour were held steady in these projections. Therefore, future growth can be expected even if no change occurs in GPs' ordering behaviour, the number of problems managed by GPs' per encounter or the number of encounters per person. However, it is inevitable that there will be further changes in these three factors. In addition to demographic changes, governments cite technological advancements (including new tests) as an expected driver of increased future health spending.^{234,397} Other potential contributors include the many drivers linked to past growth, such as changes in disease prevalence, emergence of new evidence and new technologies, changes in health policy etc., and there are likely to be many more that emerge over time.

My research suggests that workforce factors may also influence the future growth of GPs' pathology ordering. For example, the finding that young GPs and female GPs ordered more pathology tests after adjustment than their older and male counterparts (as discussed in Chapter 7), may contribute to growth as the number of trainee GPs and the proportion of female GPs in the workforce are increasing.³⁸³

As I discussed in Chapter 8, under the current Medicare funding structure, the future increases in volume of GPs' pathology testing will only be partially reflected in increases in Medicare outlays. Nevertheless, the projected increases in volume of

GP-ordered tests will place ongoing pressure on the pathology budget, one of many areas of the health budget that will be under pressure as a result of the ageing population.

The ageing of the population is expected to increase the prevalence of diagnosed chronic disease and therefore its management in general practice. Pathology testing is integral to this management, and to the initiatives introduced by the Government to support GPs' prevention, detection and management of chronic disease.^{187,235} As discussed earlier, some of these initiatives have encouraged GPs' use of health checks to prevent (through identification of risk factors) and detect chronic disease. Early detection is encouraged, as the sooner chronic disease is diagnosed the greater the potential for preventing disease progression and complications.

Once diagnosed, chronic disease management frequently requires long-term (often life-long) monitoring to determine disease progress, presence of complications, effectiveness of drug therapy and/or presence of side effects related to drug therapy. Pathology ordered for this monitoring already accounts for the majority of pathology testing ordered by GPs.⁴¹⁵ If diagnosed chronic disease prevalence increases as expected there will be significant growth in the associated pathology testing. Early detection may compound this further, particularly if life expectancy in the population continues to increase, because if people are living longer and chronic disease is diagnosed earlier, there will be a longer duration of monitoring required. Therefore, the associated growth in pathology ordering may be exponential. The expected increases in this area underscore the importance of improving the poor monitoring guidance in guidelines, and supporting GPs in the quality use of pathology testing in long-term chronic disease management.

Suggestions for future research

Further investigation of the appropriateness of GPs' pathology ordering for clinical presentations beyond the six problems investigated in this thesis, would provide a more comprehensive evaluation of the appropriateness of GPs' pathology ordering. Such evaluation could be undertaken with BEACH data, using similar methods to those used in the study described in Chapter 5. This research would provide an assessment of the availability and quality of guidance documents for other clinical

presentations and highlight areas for future interventions where it was found that GPs' ordering and/or guidance could be improved.

Future research will be needed to evaluate the impact of recent national programs that aim to introduce decision support tools for GPs, and to promote evidence-based testing. These programs are likely to implement educational interventions targeting GPs' pathology ordering behaviour in regard to specific tests or clinical problems. Such interventions will need to be evaluated. The results of this thesis indicate that any future evaluation will need to measure the impact of the intervention on GPs' ordering behaviour (including the likelihood of ordering, and the number and types of tests ordered), as well as changes in GP workload related to management of the targeted clinical problem or area. It is also important to collect sufficient variables to assess the appropriateness of pathology ordering in the targeted clinical area, at baseline and after the intervention. A simple measure of the number of tests ordered will not provide sufficient detail to determine success of an intervention or the appropriateness of any measured change. Ideally such evaluations would be conducted as randomised controlled trials.

In this thesis, the need for further research has been highlighted in several other areas, not covered in the research proposed above. Examples include: the need to describe and assess the appropriateness of GPs' pathology ordering in selected clinical situations (that could not assessed in this cross-sectional study); the need for research to establish the role of selected pathology tests in the investigation or monitoring of a condition/presentation; and the need for evidence about outcomes associated with different testing practices.

Establishing a patient-based longitudinal study has the potential to fill these gaps. Experience from the research presented in this thesis suggests that the variables required for future pathology research include the clinical presentation, details about patients and their medical history (e.g. test results, presence of diagnosed morbidities, risk factors, use of selected medications, family history, and social circumstances), pathology tests ordered, and outcome measures (e.g. hospitalisations, measures of disease severity, and patients' responses to testing such as psychosocial, cognitive and behavioural responses).

318

There are numerous ways in which data collected in this type of longitudinal patientbased study could be utilised. For example, it would be possible to measure patient health outcomes associated with GPs' adherence to pathology monitoring protocols in the management of specific diseases. It would also be possible to use the additional variables collected in this research (as listed above) to identify factors associated with the unexplained variance among GPs in their pathology ordering behaviour. Establishing a general practice patient-based longitudinal research project would create a data set that contains pathology test ordering and results data in the broader clinical context of the patient.

10 Conclusion

Considerable growth in pathology services generated by GP orders has led to concerns about appropriateness of this ordering. In this thesis I investigated three aspects of GPs' pathology: the reasons for GPs' pathology ordering; how these reasons contribute to growth in pathology ordering; and the appropriateness of this ordering. This body of work provides the first clear picture in over a decade of GPs' pathology ordering and how it has changed over time.

Factors contributing to the increase in pathology ordering include: increasing likelihood of ordering pathology tests; increasing number of tests ordered per episode; increasing number of problems managed at GP–patient encounters; increasing total volume of GP services provided to the Australian population. A small number of tests and problems account for the majority of pathology tests ordered by Australian GPs, and the majority of growth. Therefore, targeting interventions to these tests and/or problems may have a significant impact on the volume of pathology ordered by GPs.

The volume of pathology tests related to an individual problem is only partially determined by GPs' pathology ordering behaviour in the management of the problem. This has implications for interventions that target GPs' pathology ordering behaviour because if successful such interventions will only partially affect the associated volume of pathology tests ordered. With recent national programs being funded to develop interventions to support GPs' in the area of pathology ordering, an evaluation strategy that measures the impact of these interventions is needed. Such evaluation needs to determine the impact of the intervention on the volume of pathology testing that can be attributed to GP ordering behaviour, and the appropriateness of any change in ordering behaviour.

Variance among GPs in their volume of pathology ordering is used in the literature as an indicator of poor quality of GPs' ordering. I found that the foremost factor in explaining variance is the clinical problem being managed by the GP, which suggests patient-centred care is the main driver for variance. As such variance is appropriate, I conclude that variation in total rates of pathology ordering is not a reasonable measure of the quality of GPs' ordering.

The good alignment with guidance documents of GPs' test selection in the management of six problems, and the fact that the increased ordering for these problems did not reflect a change in appropriateness of ordering, lead me to conclude that concerns about widespread inappropriateness of GPs pathology ordering are not substantiated, and that growth in testing does not equate with an increase in inappropriate ordering.

The expected ageing and growth in Australia's population will generate future growth in volume of pathology ordering irrespective of changes to GPs' clinical workload or GPs' pathology ordering behaviour. In particular, the ageing population will generate increased management of chronic diseases and along with it pathology testing associated with its management. The majority of this testing will be ordered for monitoring, however, recommendations for monitoring in guidelines were poor. Guidance needs to incorporate monitoring protocols detailing the pathology tests that have a role in ongoing monitoring, and the intervals at which these tests are required. Further research is needed to investigate and evaluate the monitoring protocols used by GPs. Ideally such research would be undertaken in a patient-based longitudinal research project that allows evaluation of patient health outcomes associated with GPs' adherence to pathology monitoring protocols. Improving the evidence base and guidance for pathology testing in long-term management of chronic disease will support GPs' in their quality use of pathology testing in this high growth area.

References in citation order

- Bayram C & Valenti L. GP pathology ordering. In: Britt H & Miller GC (eds). General practice in Australia, health priorities and policies 1998-2008. General practice series no. 24. Cat. no. GEP 24. Canberra: Australian Institute of Health and Welfare, 2009;57-86.
- Bayram C, Britt H, Miller G, Valenti L. Evidence-practice gap in GP pathology test ordering: a comparison of BEACH pathology data and recommended testing. Sydney: The University of Sydney, 2009. Viewed 7 March 2013, www.health.gov.au/internet/main/Publishing.nsf/Content/ 9B6B7B4EAAC0383DCA256F180048CCBD/\$File/BEACH_QUPP%20repo rt_FINAL_050609.pdf.
- 3. Bayram C, Britt H, Kelly Z, Valenti L. Male consultations in general practice in Australia 1999-00. General practice series no. 11. Cat. no. GEP 11. Canberra: Australian Institute of Health and Welfare, 2003.
- 4. Dorland's illustrated medical dictionary. 32nd Edition. Philadelphia: Elsevier Saunders, 2012.
- 5. Australian Government Department of Health and Ageing. Medicare statistics, March quarter 2012, Group BA tables. 2012. Viewed 25 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/medstatmar12-tables-ba.
- 6. Deeble J & Lewis-Hughes P. Directions for pathology. National Health Strategy Unit. National Health Strategy Background Papers 1991;6;1-88.
- 7. PriceWaterhouseCoopers. International review of pathology funding arrangements. 2010. Viewed 27 February 2013, www.hirc.health.gov.au/ internet/main/publishing.nsf/Content/20CBB30C457BEB1DCA2576D30082 2677/\$File/PwC_DoHAPathologyReview_180310_FINAL.pdf.
- 8. Pathology Review Advisory Group. Report of the review of NHS pathology services in England (The Carter report). 2005.
- 9. Winkens R & Dinant GJ. Evidence base of clinical diagnosis: Rational, cost effective use of investigations in clinical practice. BMJ 2002;324(7340):783.
- 10. The Lewin Group. Laboratory medicine: a national status report. Centers for Disease Control and Prevention's Laboratory Science, Policy and Practice Program Office, 2008.
- van Walraven C, Goel V, Chan B. Effect of population-based interventions on laboratory utilization: a time-series analysis. JAMA 1998;280(23):2028-33.
- 12. Trinick TR, Archbold GPR, Crooks GM, Damani NN, Hughes DFC, Jones FGC et al. Appropriateness of laboratory investigations: report of a CREST working group. Belfast: Clinical Resource Efficiency Support Team (CREST), 2000.
- 13. Australian Government Department of Health and Ageing. Medicare statistics, March quarter 2012, Group BB tables. 2012. Viewed 25 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/medstat-mar12-tables-bb.

- Australian Institute of Health and Welfare. Health expenditure Australia 2009-10. Health and Welfare Expenditure Series no 46. Cat. no. HWE 55. Canberra: AIHW, 2011.
- 15. Australian Bureau of Statistics. Population by Age and Sex, Australian States and Territories, Jun 2010. Cat. no. 3201.0. Canberra: Australian Bureau of Statistics, 2010.
- 16. Australian Institute of Health and Welfare. Medical labour force 2008. Cat. no. AUS 131. Canberra: AIHW, 2010.
- 17. Britt H, Miller GC, Charles J, Henderson J, Bayram C, Pan Y et al. General practice activity in Australia 2009-10. General practice series no. 27. Cat. no. GEP 27. Canberra: Australian Institute of Health and Welfare, 2010.
- Australian Government Department of Health and Ageing. MBS primary care items: History of key MBS primary care initiatives 1999-2010. 2011. Viewed 25 February 2013, www.health.gov.au/internet/main/publishing.nsf/ Content/mbsprimarycare-History.
- Australian Association of Pathology Practices Inc. AAPP submission to Department of Health and Ageing's Review of funding arrangements of pathology services. 2010. Viewed 8 February 2013, www.aapp.asn.au/images/document/AAPP%20Funding%20Submission%20-%20May%202010.pdf.
- 20. Australian Government Department of Health and Ageing (Medical Benefits Reviews Task Group). First Discussion Paper for Review of Funding of Pathology. 2010. Viewed 14 February 2013, www.health.gov.au/internet/ main/publishing.nsf/Content/A7F7F7F944757607CA2576A2001C1725/\$File /Pathology%20Review%20Discussion%20Paper.pdf.
- 21. Flynn K. Medical fraud and inappropriate practice in Medibank and Medicare, Australia 1975-1995. PhD thesis. University of Wollongong, 2004.
- 22. Laurence C, Gialamas A, Yelland L, Bubner T, Glastonbury B, Beilby J, Shephard M, Gill J on behalf of the Trial Management and Evaluation Group, PoCT Device Group, and RCPA Quality Assurance Programs Pty Ltd. Point of care testing in general practice trial. 2009. Viewed 14 February 2013, www.appn.net.au/Data/Sites/1/SharedFiles/Publications/200901poctfinalreport27jan09amended5feb09.pdf.
- 23. Australian Government Department of Health and Ageing. Medicare Benefits Schedule Book. Canberra: DoHA, 2007.
- 24. Australian Government Department of Health and Ageing. Pathology Quality and Outlays Memorandum of Understanding (MoU). 2005. Viewed 27 February 2013, http://pandora.nla.gov.au/pan/13326/20080318-0005/www.health.gov.au/internet/main/publishing.nsf/content/5762a7197be4 25efca256f1800469fa1/\$file/pathmou.pdf.
- 25. Australian Government Department of Health and Ageing. Pathology funding agreement between the Australian Government and the Australian Association of Pathology Practices and the Royal College of Pathologists of Australasia and the National Coalition of Public Pathology. 2011. Viewed 25 February 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/98BD1271A55D88 63CA257642001C643C/\$File/Pathology%20Funding%20Agreement.pdf.

- 26. Australian National Audit Office. Administration of the Pathology Quality and Outlays Memorandum of Understanding: Department of Health and Ageing. 2008. Viewed 27 February 2013, www.anao.gov.au/Publications/Audit-Reports/2007-2008/Administration-ofthe-Pathology-Quality-and-Outlays-Memorandum-of-Understanding.
- 27. KPMG. Review of the Intergenerational Report and Pathology Funding Arrangements in Australia. 2011. Viewed 21 February 2013, www.aapp.asn.au/images/document/KPMG%20Econotech%20Pathology%2 0Funding%20Study%20-%20Jan%202011.pdf.
- 28. Rao GG, Crook M, Tillyer ML. Pathology tests: is the time for demand management ripe at last? J Clin Pathol 2003;56(4):243-8.
- 29. Fryer AA & Hanna FW. Managing demand for pathology tests: financial imperative or duty of care? Ann Clin Biochem 2009;46(Pt 6):435-7.
- 30. Michael Legg & Associates, IRIS Research, University of Wollongong, Cheong I. A study of the impact of the use of general practice computer systems on the ordering of pathology. Quality Use of Pathology Program, Department of Health and Ageing, 2004.
- 31. Wong C & Chu S. Optimising clinical laboratory service delivery with electronic decision support: An analysis of market structure and impacts of regional clinical information strategy Part 1. Health care and informatics review online 2004;8(3).
- 32. Williams S, Weinman J, Dale J, Newman S. Patient expectations: what do primary care patients want from the GP and how far does meeting expectations affect patient satisfaction? Fam Pract 1995;12(2):193-201.
- 33. van Bokhoven MA, Pleunis-van Empel MC, Koch H, Grol RP, Dinant GJ, van der Weijden T. Why do patients want to have their blood tested? A qualitative study of patient expectations in general practice. BMC Fam Pract 2006;7:75.
- van der Weijden T, van Velsen M, Dinant GJ, van Hasselt CM, Grol R. Unexplained complaints in general practice: prevalence, patients' expectations, and professionals' test-ordering behavior. Med Decis Making 2003;23(3):226-31.
- 35. Peck BM, Ubel PA, Roter DL, Goold SD, Asch DA, Jeffreys AS et al. Do unmet expectations for specific tests, referrals, and new medications reduce patients' satisfaction? J Gen Intern Med 2004;19(11):1080-7.
- 36. van der Weijden T, van Bokhoven MA, Dinant GJ, van Hasselt CM, Grol RP. Understanding laboratory testing in diagnostic uncertainty: a qualitative study in general practice. Br J Gen Pract 2002;52(485):974-80.
- 37. Enno A, Foster M, Hines P, Ireland M, McGrath K, O'Connell D et al. Quality use of pathology by general practitioners. Final report by the Hunter Urban Division of General Practice and Hunter Area Pathology Service. Hunter Urban Division of General Practice, 2004.
- 38. Henderson J, Miller G, Britt H, Pan Y. Effect of computerisation on Australian general practice: does it improve the quality of care? Qual Prim Care 2010;18(1):33-47.

- 39. Smellie WS. Demand management and test request rationalization. Ann Clin Biochem 2012;49(Pt 4):323-36.
- 40. Whiting P, Toerien M, de Salis I, Sterne JA, Dieppe P, Egger M et al. A review identifies and classifies reasons for ordering diagnostic tests. J Clin Epidemiol 2007;60(10):981-9.
- 41. Sood R, Sood A, Ghosh AK. Non-evidence-based variables affecting physicians' test-ordering tendencies: a systematic review. Neth J Med 2007;65(5):167-77.
- 42. Lundberg GD. The need for an outcomes research agenda for clinical laboratory testing. JAMA 1998;280(6):565-6.
- 43. Calcino G. Variability in ordering of pathology by general practitioners and factors related to the variability. Graduate Diploma in Medical Statistics thesis. The University of Newcastle, 1997.
- 44. Hartley RM, Charlton JR, Harris CM, Jarman B. Patterns of physicians' use of medical resources in ambulatory settings. Am J Public Health 1987;77(5):565-7.
- 45. Britt H, Knox S, Miller GC. Changes in pathology ordering by general practitioners in Australia 1998-2001. General practice series no. 13. Cat. no. GEP 13. Canberra: Australian Institute of Health and Welfare, 2003.
- 46. Kristiansen IS & Hjortdahl P. The general practitioner and laboratory utilization: why does it vary? Fam Pract 1992;9(1):22-7.
- 47. Vinker S, Kvint I, Erez R, Elhayany A, Kahan E. Effect of the characteristics of family physicians on their utilisation of laboratory tests. Br J Gen Pract 2007;57(538):377-82.
- 48. Taylor PT. Some fundamental influences on the pathology requesting behaviour of Victorian general practitioners--preliminary study results. Pathology 1994;26(1):33-5.
- 49. Ferrier BM, Woodward CA, Cohen M, Goldsmith CH. Laboratory Tests: Which Physicians Order More?: How women's practices differ from men's. Can Fam Physician 1991;37:349-52.
- 50. Verstappen WH, ter Riet G, Dubois WI, Winkens R, Grol RP, van der Weijden T. Variation in test ordering behaviour of GPs: professional or context-related factors? Fam Pract 2004;21(4):387-95.
- 51. Leurquin P, van Casteren V, De Maeseneer J. Use of blood tests in general practice: a collaborative study in eight European countries. Eurosentinel Study Group. Br J Gen Pract 1995;45(390):21-5.
- 52. van Walraven C & Naylor CD. Do we know what inappropriate laboratory utilization is? A systematic review of laboratory clinical audits. JAMA 1998;280(6):550-8.
- 53. Schattner A. The unbearable lightness of diagnostic testing: time to contain inappropriate test ordering. Postgrad Med J 2008;84(998):618-21.
- 54. Smellie WS. Appropriateness of test use in pathology: a new era or reinventing the wheel? Ann Clin Biochem 2003;40(Pt 6):585-92.
- 55. Irwig L, Bossuyt P, Glasziou P, Gatsonis C, Lijmer J. Designing studies to ensure that estimates of test accuracy are transferable. BMJ 2002;324(7338):669-71.

- 56. Smellie WS, Finnigan DI, Wilson D, Freedman D, McNulty CA, Clark G. Methodology for constructing guidance. J Clin Pathol 2005;58(3):249-53.
- 57. Fryer AA & Smellie WS. Managing demand for laboratory tests: a laboratory toolkit. J Clin Pathol 2013;66(1):62-72.
- Thomas RE, Croal BL, Ramsay C, Eccles M, Grimshaw J. Effect of enhanced feedback and brief educational reminder messages on laboratory test requesting in primary care: a cluster randomised trial. Lancet 2006;367(9527):1990-6.
- 59. Bindels R, De Clercq PA, Winkens RA, Hasman A. A test ordering system with automated reminders for primary care based on practice guidelines. Int J Med Inform 2000;58-59:219-33.
- 60. Price CP. Evidence-based laboratory medicine: supporting decision-making. Clin Chem 2000;46(8 Pt 1):1041-50.
- 61. Price CP. Evidence-based laboratory medicine: linking tests to outcomes. Ned Tijdschr Kin Chem 2001;26:236-42.
- 62. Ferrante di Ruffano L, Davenport C, Eisinga A, Hyde C, Deeks JJ. A capturerecapture analysis demonstrated that randomized controlled trials evaluating the impact of diagnostic tests on patient outcomes are rare. J Clin Epidemiol 2012;65(3):282-7.
- 63. Bossuyt PM & McCaffery K. Additional patient outcomes and pathways in evaluations of testing. Med Decis Making 2009;29(5):E30-E38.
- 64. Cagliero E, Levina EV, Nathan DM. Immediate feedback of HbA1c levels improves glycemic control in type 1 and insulin-treated type 2 diabetic patients. Diabetes Care 1999;22(11):1785-9.
- 65. Troughton RW, Frampton CM, Yandle TG, Espiner EA, Nicholls MG, Richards AM. Treatment of heart failure guided by plasma aminoterminal brain natriuretic peptide (N-BNP) concentrations. Lancet 2000;355(9210):1126-30.
- 66. Stuart PJ, Crooks S, Porton M. An interventional program for diagnostic testing in the emergency department. Med J Aust 2002;177(3):131-4.
- 67. Johnson RK & Mortimer AJ. Routine pre-operative blood testing: is it necessary? Anaesthesia 2002;57(9):914-7.
- Schein OD, Katz J, Bass EB, Tielsch JM, Lubomski LH, Feldman MA et al. The value of routine preoperative medical testing before cataract surgery. Study of Medical Testing for Cataract Surgery. N Engl J Med 2000;342(3):168-75.
- 69. Moynihan R, Doust J, Henry D. Preventing overdiagnosis: how to stop harming the healthy. BMJ 2012;344:e3502.
- 70. Ferrante di Ruffano L, Hyde CJ, McCaffery KJ, Bossuyt PM, Deeks JJ. Assessing the value of diagnostic tests: a framework for designing and evaluating trials. BMJ 2012;344:e686.
- 71. Bugter-Maessen AM, Winkens RA, Grol RP, Knottnerus JA, Kester AD, Beusmans GH et al. Factors predicting differences among general practitioners in test ordering behaviour and in the response to feedback on test requests. Fam Pract 1996;13(3):254-8.

- 72. Davis P, Gribben B, Lay-Yee R, Scott A. How much variation in clinical activity is there between general practitioners? A multi-level analysis of decision-making in primary care. Journal of Health Services Research and Policy 2002;7(4):202-8.
- 73. Hartley RM, Charlton JR, Harris CM, Jarman B. Influence of patient characteristics on test ordering in general practice. Br Med J (Clin Res Ed) 1984;289(6447):735-8.
- 74. Winkens RA, Pop P, Bugter-Maessen AM, Grol RP, Kester AD, Beusmans GH et al. Randomised controlled trial of routine individual feedback to improve rationality and reduce numbers of test requests. Lancet 1995;345(8948):498-502.
- 75. Verstappen WH, van der Weijden T, Sijbrandij J, Smeele I, Hermsen J, Grimshaw J et al. Effect of a practice-based strategy on test ordering performance of primary care physicians: a randomized trial. JAMA 2003;289(18):2407-12.
- Poley MJ, Edelenbos KI, Mosseveld M, van Wijk MA, de Bakker DH, van der Lei J et al. Cost consequences of implementing an electronic decision support system for ordering laboratory tests in primary care: evidence from a controlled prospective study in the Netherlands. Clin Chem 2007;53(2):213-9.
- 77. Verstappen WH, van der Weijden T, Dubois WI, Smeele I, Hermsen J, Tan FE et al. Improving test ordering in primary care: the added value of a smallgroup quality improvement strategy compared with classic feedback only. Ann Fam Med 2004;2(6):569-75.
- Verstappen WH, van Merode F, Grimshaw J, Dubois WI, Grol RP, van der Weijden T. Comparing cost effects of two quality strategies to improve test ordering in primary care: a randomized trial. Int J Qual Health Care 2004;16(5):391-8.
- 79. van Wijk MA, van der Lei J, Mosseveld M, Bohnen AM, van Bemmel JH. Assessment of decision support for blood test ordering in primary care. a randomized trial. Ann Intern Med 2001;134(4):274-81.
- 80. Bailey J, Jennings A, Parapia L. Change of pathology request forms can reduce unwanted requests and tests. J Clin Pathol 2005;58(8):853-5.
- 81. Shalev V, Chodick G, Heymann AD. Format change of a laboratory test order form affects physician behavior. Int J Med Inform 2009;78(10):639-44.
- Smellie WS, Galloway MJ, Chinn D. Benchmarking general practice use of pathology services: a model for monitoring change. J Clin Pathol 2000;53(6):476-80.
- Smellie WS, Lowrie R, Wilkinson E. A laboratory based intervention to improve appropriateness of lipid tests and audit cholesterol lowering in primary care. BMJ 2001;323(7323):1224-7.
- 84. Mindemark M & Larsson A. Long-term effects of an education programme on the optimal use of clinical chemistry testing in primary health care. Scand J Clin Lab Invest 2009;69(4):481-6.
- 85. Tomlin A, Dovey S, Gauld R, Tilyard M. Better use of primary care laboratory services following interventions to 'market' clinical guidelines in

New Zealand: a controlled before-and-after study. BMJ Qual Saf 2011;20(3):282-90.

- 86. Bindels R, Hasman A, van Wersch JW, Talmon J, Winkens RA. Evaluation of an automated test ordering and feedback system for general practitioners in daily practice. Int J Med Inform 2004;73(9-10):705-12.
- Smith BJ & McNeely MD. The influence of an expert system for test ordering and interpretation on laboratory investigations. Clin Chem 1999;45(8 Pt 1):1168-75.
- 88. van Wijk MA, van der Lei J, Mosseveld M, Bohnen AM, van Bemmel JH. Compliance of general practitioners with a guideline-based decision support system for ordering blood tests. Clin Chem 2002;48(1):55-60.
- 89. Britt H, Miller GC, McGeechan K, Sayer GP. Pathology ordering by general practitioners in Australia 1998. Cat. no. GEP 4. Canberra: Department of Health and Aged Care, 1999. Viewed 27 February 2013, http://pandora.nla.gov.au/nphwb/20001220130000/http://www.health.gov.au/haf/docs/pathorder.htm.
- 90. Leese B. Is there too much laboratory testing? CHE discussion paper 79. York, UK: University of York, 1991.
- 91. Solomon DH, Hashimoto H, Daltroy L, Liang MH. Techniques to improve physicians' use of diagnostic tests: a new conceptual framework. JAMA 1998;280(23):2020-7.
- 92. Buntinx F, Winkens R, Grol R, Knottnerus JA. Influencing diagnostic and preventive performance in ambulatory care by feedback and reminders. A review. Fam Pract 1993;10(2):219-28.
- 93. Axt-Adam P, van der Wouden JC, van der Does E. Influencing behavior of physicians ordering laboratory tests: a literature study. Med Care 1993;31(9):784-94.
- 94. Young DW. Improving laboratory usage: a review. Postgrad Med J 1988;64(750):283-9.
- Winkens RA, Pop P, Grol RP, Kester AD, Knottnerus JA. Effect of feedback on test ordering behaviour of general practitioners. BMJ 1992;304(6834):1093-6.
- Bunting PS & van Walraven C. Effect of a controlled feedback intervention on laboratory test ordering by community physicians. Clin Chem 2004;50(2):321-6.
- Zaat JO, van Eijk JT, Bonte HA. Laboratory test form design influences test ordering by general practitioners in The Netherlands. Med Care 1992;30(3):189-98.
- Burnett L & Sugerman DA. Restrictions on 'multiple biochemical analyses' profiling decreases usage of biochemistry test requests. Pathology 1991;23(2):103-6.
- Kerr D, Malcolm L, Schousboe J, Pimm F. Successful implementation of laboratory budget holding by Pegasus Medical Group. N Z Med J 1996;109(1029):334-7.

- 100. Vardy DA, Simon T, Limoni Y, Kuperman O, Rabzon I, Cohen A et al. The impact of structured laboratory routines in computerized medical records in a primary care service setting. J Med Syst 2005;29(6):619-26.
- 101. Australian Government Department of Health and Ageing. Quality Use of Pathology Program. 2013. Viewed 1 October 2013, www.health.gov.au/internet/main/publishing.nsf/Content/pathology-quppindex.
- The Royal College of Pathologists of Australasia. Common Sense Pathology. 2012. Viewed 27 February 2013, www.rcpa.edu.au/Publications/CommonSensePathology.htm.
- 103. Britt H, Miller GC, Charles J, Henderson J, Bayram C, Valenti L et al. General practice activity in Australia 2000-01 to 2009-10: 10 year data tables. General practice series no. 28. Cat. no. GEP 28. Canberra: Australian Institute of Health and Welfare, 2010.
- 104. Britt H, Sayer GP, Miller GC, Charles J, Scahill S, Horn F et al. BEACH Bettering the Evaluation and Care of Health: A study of general practice activity, six-month interim report. General practice series no. 1. Cat. no. GEP 1. Canberra: Australian Institute of Health and Welfare, 1999.
- 105. Bridges-Webb C, Britt H, Miles DA, Neary S, Charles J, Traynor V. Morbidity and treatment in general practice in Australia 1990-1991. Med J Aust 1992;157(19 Oct Spec Sup):S1-S56.
- 106. Meza RA, Angelis M, Britt H, Miles DA, Seneta E, Bridges-Webb C. Development of sample size models for national general practice surveys. Aust J Public Health 1995;19(1):34-40.
- 107. Australian Bureau of Statistics. Australian Standard Geographical Classification. Cat. no. 1216.0. Canberra: ABS, 2008.
- 108. Britt H, Sayer GP, Miller GC, Charles J, Scahill S, Horn F et al. General practice activity in Australia 1998-99. General practice series no. 2. Cat. no. GEP 2. Canberra: Australian Institute of Health and Welfare, 1999.
- World Health Organization. Obesity: preventing and managing the global epidemic: report of a WHO consultation. WHO technical report series: 894. 2000. Viewed 27 February 2013, http://whqlibdoc.who.int/trs/WHO_TRS_894.pdf.
- Classification Committee of the World Organization of Family Doctors (WICC). ICPC-2: International Classification of Primary Care. 2 ed. Oxford: Oxford University Press, 1998.
- World Health Organization. World Health Organization family of international classifications: definition, scope and purpose. Geneva: WHO, 2004. Viewed 27 February 2013, www.who.int/entity/classifications/en/FamilyDocument2007.pdf.
- 112. Australian Institute of Health and Welfare. Australian family of health and related classifications matrix. Canberra: AIHW, 2003. Viewed 27 February 2013, www.aihw.gov.au/WorkArea/DownloadAsset.aspx?id=6442475388&libID=6

442475369.

- 113. Britt H. A new coding tool for computerised clinical systems in primary care--ICPC plus. Aust Fam Physician 1997;26(Suppl 2):S79-S82.
- 114. World Health Organization Collaborating Centre for Drug Statistics Methodology. Anatomical Therapeutic Chemical (ATC) classification index with Defined Daily Doses (DDDs). January 1998 ed. Oslo: WHO, 1997.
- 115. World Health Organization Collaborating Centre for Drug Statistics Methodology. International language for drug utilization research. 2012. Viewed 27 February 2013, http://www.whocc.no/.
- 116. Driver B, Britt H, O'Toole B, Harris M, Bridges-Webb C, Neary S. How representative are patients in general practice morbidity surveys? Fam Pract 1991;8(3):261-8.
- 117. Britt H, Harris M, Driver B, Bridges-Webb C, O'Toole B, Neary S. Reasons for encounter and diagnosed health problems: convergence between doctors and patients. Fam Pract 1992;9(2):191-4.
- 118. Britt H. Reliability of central coding of patient reasons for encounter in general practice, using the International Classification of Primary Care. Journ Informatics in Prim Care 1998;May:3-7.
- 119. Britt H. A measure of the validity of the ICPC in the classification of reasons for encounter. Journ Informatics in Prim Care 1997;Nov:8-12.
- 120. Britt H, Miller GC, Charles J, Bayram C, Pan Y, Henderson J et al. General practice activity in Australia 2006-07. General practice series no. 21. Cat. no. GEP 21. Canberra: Australian Institute of Health and Welfare, 2008.
- 121. Britt H, Miller GC, Charles J, Henderson J, Bayram C, Harrison C et al. General practice activity in Australia 2007-08. General practice series no. 22. Cat. no. GEP 22. Canberra: Australian Institute of Health and Welfare, 2008.
- 122. Britt H, Miller G, Charles J, Henderson J, Bayram C, Pan Y et al. General practice activity in Australia 2008-09. General practice series no. 25. Cat. no. GEP 25. Canberra: Australian Institute of Health and Welfare, 2009.
- 123. Britt H, Miller GC, Charles J, Pan Y, Valenti L, Henderson J et al. General practice activity in Australia 2005-06. General practice series no. 19. Cat. no. GEP 19. Canberra: Australian Institute of Health and Welfare, 2007.
- 124. Britt H, Miller GC, Knox S, Charles J, Pan Y, Henderson J et al. General practice activity in Australia 2004-05. General practice series no. 18. Cat. no. GEP 18. Canberra: Australian Institute of Health and Welfare, 2005.
- 125. Britt H, Miller GC, Knox S, Charles J, Valenti L, Henderson J et al. General practice activity in Australia 2000-01. General practice series no. 8. Cat. no. GEP 8. Canberra: Australian Institute of Health and Welfare, 2001.
- 126. Britt H, Miller GC, Knox S, Charles J, Valenti L, Henderson J et al. General practice activity in Australia 2001-02. General practice series no. 10. Cat. no. GEP 10. Canberra: Australian Institute of Health and Welfare, 2002.
- 127. Britt H, Miller GC, Knox S, Charles J, Valenti L, Henderson J et al. General practice activity in Australia 2002-03. General practice series no. 14. Cat. no. GEP 14. Canberra: Australian Institute of Health and Welfare, 2003.
- 128. Britt H, Miller GC, Knox S, Charles J, Valenti L, Pan Y et al. General practice activity in Australia 2003-04. General practice series no. 16. Cat. no. GEP 16. Canberra: Australian Institute of Health and Welfare, 2004.

- 129. SAS proprietary software release 9.1. Cary: SAS Institute Inc, 2003.
- 130. Wolfe R & Hanley J. If we're so different, why do we keep overlapping? When 1 plus 1 doesn't make 2. CMAJ 2002;166(1):65-6.
- 131. Cumming G & Finch S. Inference by eye: confidence intervals and how to read pictures of data. Am Psychol 2005;60(2):170-80.
- Austin PC & Hux JE. A brief note on overlapping confidence intervals. J Vasc Surg 2002;36(1):194-5.
- Winkelman JW. Less utilization of the clinical laboratory produces disproportionately small true cost reductions. Hum Pathol 1984;15(6):499-501.
- 134. PriceWaterhouseCoopers. Capital expenditure in the pathology sector. 2010. Viewed 27 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/20CBB30C457BEB 1DCA2576D300822677/\$File/PWC%20Final%20Report%2021%20Septemb er%202010.pdf.
- 135. Rossouw JE, Anderson GL, Prentice RL, LaCroix AZ, Kooperberg C, Stefanick ML et al. Risks and benefits of estrogen plus progestin in healthy postmenopausal women: principal results From the Women's Health Initiative randomized controlled trial. JAMA 2002;288(3):321-33.
- 136. Charles J, Britt H, Harrison C. General practice workforce and workload. In: Britt H & Miller GC (eds). General practice in Australia, health priorities and policies 1998-2008. General practice series no. 24. Cat. no. GEP 24. Canberra: Australian Institute of Health and Welfare, 2009;22-37.
- 137. Australian Bureau of Statistics. Population Projections, Australia, 2006 to 2101. Cat. no. 3222.0. Canberra: Australian Bureau of Statistics, 2008.
- Australian National Audit Office. Practice Incentives Program. 2010. Viewed 27 February 2013, www.anao.gov.au/Publications/Audit-Reports/2010-2011/Practice-Incentives-Program.
- 139. Australian Medical Workforce Advisory Committee. The general practice workforce in Australia: supply and requirements to 2013. Number AMWAC Report 2005.2. Sydney: AMWAC, 2005.
- 140. Australian Government, Office of the Australian Information Commissioner. Privacy in the Private Health Sector. 2001. Viewed 27 February 2013, www.privacy.gov.au/materials/types/download/8675/6517.
- 141. Australian Bureau of Statistics. Health services: patient experiences in Australia, 2009. Cat. no. 4839.0.55.001. Canberra: Australian Bureau of Statistics, 2010. Viewed 27 February 2013, www.ausstats.abs.gov.au/Ausstats/subscriber.nsf/0/4DE1EBF2E29AAC71C A2577730022B837/\$File/4839055001_2009.pdf.
- 142. Jorgensen LG, Brandslund I, Hyltoft PP. Should we maintain the 95 percent reference intervals in the era of wellness testing? A concept paper. Clin Chem Lab Med 2004;42(7):747-51.
- 143. Phillips P. The trouble with testing: screening. Aust Fam Physician 2003;32(11):925-6.
- 144. Smellie WS. When is "abnormal" abnormal? Dealing with the slightly out of range laboratory result. J Clin Pathol 2006;59(10):1005-7.

- 145. Wheatland FT. Medical indemnity reform in Australia: "first do no harm". J Law Med Ethics 2005;33(3):429-43.
- 146. Burdon J. Medical Indemnity Insurance 2006. Presented at the Australian College of Legal Medicine's Annual Scientific Meeting, Melbourne 14-15 October 2006; Australian College of Legal Medicine; 2006.
- 147. Nash LM, Walton MM, Daly MG, Kelly PJ, Walter G, van Ekert EH et al. Perceived practice change in Australian doctors as a result of medicolegal concerns. Med J Aust 2010;193(10):579-83.
- 148. Fowkes FG. Containing the use of diagnostic tests. Br Med J (Clin Res Ed) 1985;290(6467):488-90.
- 149. Hammett RJ & Harris RD. Halting the growth in diagnostic testing. Med J Aust 2002;177(3):124-5.
- 150. Lundberg GD. The modern clinical laboratory; Justification, scope, and directions. JAMA 1975;232(5):528-9.
- 151. Australian Institute of Health and Welfare. National Health Priority Areas. 2012. Viewed 27 February 2013, www.aihw.gov.au/health-priority-areas/.
- 152. Coumou HC & Meijman FJ. How do primary care physicians seek answers to clinical questions? A literature review. J Med Libr Assoc 2006;94(1):55-60.
- 153. Carlsen B, Glenton C, Pope C. Thou shalt versus thou shalt not: a metasynthesis of GPs' attitudes to clinical practice guidelines. Br J Gen Pract 2007;57(545):971-8.
- 154. Australian Bureau of Statistics. National Health Survey: Summary of Results, 2007-08. Cat. no. 4364.0. Canberra: Australian Bureau of Statistics, 2009. Viewed 25 February 2013, www.abs.gov.au/ausstats/abs@.nsf/ Latestproducts/4364.0Main%20Features42007-2008%20(Reissue)? opendocument&tabname=Summary&prodno=4364.0&issue=2007-2008%20(Reissue)&num=&view=.
- 155. Australian Government Department of Health and Ageing. Cardiovascular disease. 2012. Viewed 27 February 2013, http://health.gov.au/internet/main/publishing.nsf/Content/chronic-cardio.
- 156. Begg S, Vos T, Barker B, Stevenson C, Stanley L, Lopez AD. The burden of disease and injury in Australia 2003. Cat. no. PHE 82. Canberra: Australian Institute of Health and Welfare, 2007.
- 157. Dunstan D, Zimmet P, Welborn T, Sicree R, Armstrong T, Atkins R et al. Diabesity and associated disorders in Australia - 2000: The accelerating epidemic. Melbourne: International Diabetes Institute, 2001.
- 158. Knox SA, Harrison CM, Britt HC, Henderson JV. Estimating prevalence of common chronic morbidities in Australia. Med J Aust 2008;189(2):66-70.
- 159. Lewington S, Clarke R, Qizilbash N, Peto R, Collins R. Age-specific relevance of usual blood pressure to vascular mortality: a meta-analysis of individual data for one million adults in 61 prospective studies. Lancet 2002;360(9349):1903-13.
- 160. Levy D, Larson MG, Vasan RS, Kannel WB, Ho KK. The progression from hypertension to congestive heart failure. JAMA 1996;275(20):1557-62.

- Stamler J, Stamler R, Neaton JD. Blood pressure, systolic and diastolic, and cardiovascular risks. US population data. Arch Intern Med 1993;153(5):598-615.
- 162. Kannel WB. Blood pressure as a cardiovascular risk factor: prevention and treatment. JAMA 1996;275(20):1571-6.
- 163. National Heart Foundation of Australia (National blood pressure and vascular disease advisory committee). Guide to management of hypertension 2008: assessing and managing raised blood pressure in adults. National Heart Foundation of Australia, 2008.
- 164. National Collaborating Centre for Chronic Conditions & British Hypertension Society. Hypertension: management of hypertension in adults in primary care (partial update of NICE clinical guideline 18). NICE clinical guideline 34. London: National Institute for Health and Clinical Excellence, 2006.
- 165. National Heart, Lung and Blood Institute. Seventh report of the Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood pressure (JNC 7). National Heart, Lung, and Blood Institute. US Department of Health and Human Services, 2004.
- 166. Scottish Intercollegiate Guidelines Network (SIGN). Hypertension in older people: a national clinical guideline. SIGN publication no. 49. Edinburgh: SIGN, 2001.
- Whitworth JA. 2003 World Health Organization (WHO)/International Society of Hypertension (ISH) statement on management of hypertension. J Hypertens 2003;21(11):1983-92.
- 168. Mancia G, De Backer G, Dominiczak A, Cifkova R, Fagard R, Germano G et al. 2007 Guidelines for the Management of Arterial Hypertension: The Task Force for the Management of Arterial Hypertension of the European Society of Hypertension (ESH) and of the European Society of Cardiology (ESC). J Hypertens 2007;25(6):1105-87.
- 169. Schwartz G, Canzanello V, Woolley A, Miller T, O'Connor P, Klein D et al. Hypertension diagnosis and treatment. ICSI health care guideline. Institute for Clinical Systems Improvement, 2006.
- 170. Padwal RS, Hemmelgarn BR, McAlister FA, McKay DW, Grover S, Wilson T et al. The 2007 Canadian Hypertension Education Program recommendations for the management of hypertension: part 1- blood pressure measurement, diagnosis and assessment of risk. Can J Cardiol 2007;23(7):529-38.
- 171. Khan NA, Hemmelgarn B, Padwal R, Larochelle P, Mahon JL, Lewanczuk RZ et al. The 2007 Canadian Hypertension Education Program recommendations for the management of hypertension: part 2 therapy. Can J Cardiol 2007;23(7):539-50.
- 172. The Royal College of Pathologists of Australasia. RCPA Manual. RCPA, 2004.
- 173. Murtagh J. Murtagh's general practice. Sydney: McGraw-Hill Australia Pty Ltd, 2007.
- 174. Steven I. Patient presentations in general practice. Sydney: McGraw-Hill Book Company Australia Pty Ltd, 1999.

- 175. Lab Tests Online AU partners. C-reactive protein. 2013. Viewed 27 February 2013, www.labtestsonline.org.au/understanding/analytes/crp/glance.html.
- 176. National Vascular Disease Prevention Alliance. Guidelines for the assessment of absolute cardiovascular disease risk. National Vascular Disease Prevention Alliance, 2009.
- 177. The Royal Australian College of General Practitioners. Guidelines for preventive activities in general practice. 7th ed. South Melbourne: The Royal Australian College of General Practitioners, 2009.
- 178. Tonkin A, Barter P, Best J, Boyden A, Furler J, Hossack K et al. National Heart Foundation of Australia and the Cardiac Society of Australia and New Zealand: position statement on lipid management - 2005. Heart Lung Circ 2005;14(4):275-91.
- 179. Therapeutic Guidelines Ltd. Therapeutic Gudelines. eTG complete [CD-ROM]. Melbourne: Therapeutic Guidelines Limited, 2008.
- 180. Dunstan DW, Zimmet PZ, Welborn TA, De Court, Cameron AJ, Sicree RA et al. The rising prevalence of diabetes and impaired glucose tolerance: the Australian Diabetes, Obesity and Lifestyle Study. Diabetes Care 2002;25(5):829-34.
- Australian Institute of Health and Welfare. Diabetes: Australian facts 2008. Cat. no. CVD 40. Canberra: AIHW, 2008.
- 182. Shaw JE, Sicree RA, Zimmet PZ. Global estimates of the prevalence of diabetes for 2010 and 2030. Diabetes Res Clin Pract 2010;87(1):4-14.
- 183. Wild S, Roglic G, Green A, Sicree R, King H. Global prevalence of diabetes: estimates for the year 2000 and projections for 2030. Diabetes Care 2004;27(5):1047-53.
- 184. Australian Institute of Health and Welfare & Commonwealth Department of Health and Family Services. First report on National Health Priority Areas 1996. Cat. no. PHE 1. Canberra: AIHW & DHFS, 1997.
- 185. National Health and Medical Research Council (NHMRC). Diabetes. 2008. Viewed 27 February 2013, www.nhmrc.gov.au/your_health/facts/diabetes.htm.
- 186. Australian Government Department of Health and Ageing. COAG reducing the risk of type 2 diabetes. 2007. Viewed 27 February 2013, www.health.gov.au/internet/budget/publishing.nsf/Content/budget2007hfact31.htm.
- 187. Australian Government Department of Health and Ageing. National chronic disease strategy. 2005. Viewed 27 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/pq-ncds-strat.
- 188. Australian Government Department of Health and Ageing. Work of the Australian Government on diabetes. 2011. Viewed 27 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/pq-diabetes-gov.
- 189. Australian Government Department of Health and Ageing. Prevention of Type 2 Diabetes Program. 2010. Viewed 27 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/chronic-diab-prevmbs.

- 190. Australian Primary Care Collaboratives. Australian Primary Care Collaboratives. 2008. Viewed 27 February 2013, www.apcc.org.au/about_the_APCC/.
- 191. Australian General Practice Statistics and Classification Centre. SAND abstract No. 148 from the BEACH program: Type 2 diabetes, and blood glucose, lipid and blood pressure medication management. 2010. Viewed 27 February 2013, http://sydney.edu.au/medicine/fmrc/publications/sandabstracts/148-Type_2_diabetes.pdf.
- 192. Diabetes Australia & The Royal Australian College of General Practitioners. Diabetes management in general practice 2008/9. Diabetes Australia, 2008.
- 193. Diabetes Australia Guideline Development Consortium. National Evidence Based Guidelines for the Management of Type 2 Diabetes Mellitus [updated 2005]. National Health and Medical Research Council (NHMRC), 2001. Viewed 27 February 2013, www.nhune.com/www.ii/actions/www.nhune.com/di7te.di12.com/html

www.nhmrc.gov.au/publications/synopses/di7todi13syn.htm.

- 194. Canadian Diabetes Association Clinical Practice Guidelines Expert Committee CDA. Canadian Diabetes Association 2008 Clinical Practice Guidelines for the Prevention and Management of Diabetes in Canada. Can J Diabetes 2008;32(Supp 1):s1-s201.
- 195. National Collaborating Centre for Chronic Conditions. Type 2 diabetes: national clinical guideline for management in primary and secondary care (update). NICE clinical guideline 66. London: National Institute for Health and Clinical Excellence, 2008.
- 196. Scottish Intercollegiate Guidelines Network (SIGN). Management of diabetes: a national clinical guideline. SIGN publication no. 55. Edinburgh: SIGN, 2001.
- American Diabetes Association. Standards of medical care in diabetes--2008. Diabetes Care 2008;31 Suppl 1:S12-S54.
- 198. Singapore Ministry of Health. Clinical practice guidelines: diabetes mellitus. Singapore: Ministry of Health, 2006.
- 199. Rodbard HW, Blonde L, Braithwaite SS, Brett EM, Cobin RH, Handelsman Y et al. American Association of Clinical Endocrinologists medical guidelines for clinical practice for the management of diabetes mellitus. Endocr Pract 2007;13 Suppl 1:1-68.
- 200. International Diabetes Federation Clinical Guidelines Task Force. Global guideline for Type 2 Diabetes. Brussels: International Diabetes Federation, 2005.
- 201. Skeie S, Perich C, Ricos C, Araczki A, Horvath AR, Oosterhuis WP et al. Postanalytical external quality assessment of blood glucose and hemoglobin A1c: an international survey. Clin Chem 2005;51(7):1145-53.
- 202. Aakre KM, Thue G, Subramaniam-Haavik S, Bukve T, Morris H, Muller M et al. Postanalytical external quality assessment of urine albumin in primary health care: an international survey. Clin Chem 2008;54(10):1630-6.
- 203. Smellie WS. What is a significant difference between sequential laboratory results? J Clin Pathol 2008;61(4):419-25.

- 204. Snow V, Aronson MD, Hornbake ER, Mottur-Pilson C, Weiss KB. Lipid control in the management of type 2 diabetes mellitus: a clinical practice guideline from the American College of Physicians. Ann Intern Med 2004;140(8):644-9.
- 205. Kannel WB, Castelli WP, Gordon T, McNamara PM. Serum cholesterol, lipoproteins, and the risk of coronary heart disease. The Framingham study. Ann Intern Med 1971;74(1):1-12.
- 206. Stamler J, Wentworth D, Neaton JD. Is relationship between serum cholesterol and risk of premature death from coronary heart disease continuous and graded? Findings in 356,222 primary screenees of the Multiple Risk Factor Intervention Trial (MRFIT). JAMA 1986;256(20):2823-8.
- 207. Kannel WB, D'Agostino RB, Sullivan L, Wilson PW. Concept and usefulness of cardiovascular risk profiles. Am Heart J 2004;148(1):16-26.
- 208. Australian Government Department of Health and Ageing. PBS-eligibility criteria for cholesterol lowering medicines. 2006. Viewed 27 February 2013, www.health.gov.au/internet/main/publishing.nsf/content/lipid_eligibilitycriter ia.htm.
- 209. Australian General Practice Statistics and Classification Centre. SAND abstract No. 149 from the BEACH program: Dyslipidaemia and lipid management. 2010. Viewed 27 February 2013, http://sydney.edu.au/medicine/fmrc/publications/sand-abstracts/149-Dyslipidaemia.pdf.
- 210. National Cholesterol Education Program Expert Panel. The Third Report of the Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). National Heart, Lung, and Blood Institute, National Institutes of Health, 2002.
- 211. Grundy SM, Cleeman JI, Merz CN, Brewer HB, Jr., Clark LT, Hunninghake DB et al. Implications of recent clinical trials for the National Cholesterol Education Program Adult Treatment Panel III guidelines. Circulation 2004;110(2):227-39.
- 212. Cooper A, Nherera L, Calvert N, O'Flynn N, Turnbull N, Robson J et al. Clinical guidelines and evidence review for lipid modification: cardiovascular risk assessment and the primary and secondary prevention of cardiovascular disease. NICE clinical guideline 67. London: National Collaborating Centre for Primary Care and Royal College of General Practitioners [National Institute for Health and Clinical Excellence], 2008.
- 213. Scottish Intercollegiate Guidelines Network (SIGN). Risk estimation and the prevention of cardiovascular disease: a national clinical guideline. SIGN publication no. 97. Edinburgh: SIGN, 2007.
- 214. Jellinger PS, Dickey RA, Ganda OP, Mehta AE, Nguyen TT, Rodbard HW et al. AACE medical guidelines for clinical practice for the diagnosis and treatment of dyslipidemia and prevention of atherogenesis. Endocr Pract 2000;6(2):162-213.
- 215. Graham I, Atar D, Borch-Johnsen K, Boysen G, Burell G, Cifkova R et al. European guidelines on cardiovascular disease prevention in clinical practice: executive summary. Atherosclerosis 2007;194(1):1-45.

216. Barrie WE, Harrison RV, Khanderia UB, Kiningham RB, Rosenson RS. Screening and Management of Lipids. University of Michigan, 2009. Viewed 27 February 2013, www.med.umich.edu/linfo/fbn/meaticeguides/lipids

www.med.umich.edu/linfo/fhp/practiceguides/lipids/lipidsupdate.pdf.

- 217. Genest J, Frohlich J, Fodor G, McPherson R. Recommendations for the management of dyslipidemia and the prevention of cardiovascular disease: summary of the 2003 update. CMAJ 2003;169(9):921-4.
- 218. Genest J, Frohlich J, Fodor G, McPherson R. Recommendations for the management of dyslipidemia and the prevention of cardiovascular disease: 2003 update. Appendix to CMAJ 2003;168(9):921-4. CMAJ., 2003. Viewed 27 February 2013, www.cmaj.ca/cgi/content/full/169/9/921/DC1.
- 219. Singapore Ministry of Health. Clinical practice guidelines: lipids. Singapore: Ministry of Health, 2006.
- 220. Woolley A, Kopecky S, Kottke T, O'Connor P, Hanson S, Conroy W et al. Lipid management in adults. ICSI health care guideline. Institute for Clinical Systems Improvement, 2007.
- 221. Lab Tests Online AU partners. Full blood count. 2009. Viewed 27 February 2013, http://labtestsonline.org.au/understanding/analytes/fbc/tab/test.
- 222. Commonwealth of Australia. Budget measure on cholesterol-lowering drugs. Canberra: Commonwealth of Australia, 2001. Viewed 27 February 2013, http://health.gov.au/internet/main/publishing.nsf/Content/health-budget2001fact-hfact7a.htm.
- 223. Commonwealth Department of Health and Aged Care. Medicare Benefits Schedule book. Canberra: DHAC, 2000.
- 224. Smellie WS, Wilson D, McNulty CA, Galloway MJ, Spickett GA, Finnigan DI et al. Best practice in primary care pathology: review 1. J Clin Pathol 2005;58(10):1016-24.
- 225. Bell KJ, Irwig L, Craig JC, Macaskill P. Use of randomised trials to decide when to monitor response to new treatment. BMJ 2008;336(7640):361-5.
- 226. Glasziou PP, Irwig L, Heritier S, Simes RJ, Tonkin A. Monitoring cholesterol levels: measurement error or true change? Ann Intern Med 2008;148(9):656-61.
- 227. Smellie WS. Testing pitfalls and summary of guidance in lipid management. BMJ 2006;333(7558):83-6.
- 228. Criqui MH & Golomb BA. Lipid lowering: what and when to monitor. Lancet 2008;372(9638):516-7.
- 229. Phillips PJ & Phillipov G. Cholesterol frequently asked questions. Aust Fam Physician 2006;35(8):595-6.
- 230. McKenney JM, Davidson MH, Jacobson TA, Guyton JR. Final conclusions and recommendations of the National Lipid Association Statin Safety Assessment Task Force. Am J Cardiol 2006;97(8A):89C-94C.
- 231. Smellie WS, Forth J, Bareford D, Twomey P, Galloway MJ, Logan EC et al. Best practice in primary care pathology: review 3. J Clin Pathol 2006;59(8):781-9.
- 232. Australian Government Department of Health and Ageing. Health assessment for people aged 45 to 49 years who are at risk of developing chronic disease.
2012. Viewed 14 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/mbsprimarycare_mbsitem701_703_705_707.

- 233. Fisher E. Australian general practice a celebration. 2008. Viewed 27 February 2013, www.racgp.org.au/historyofthecollege.
- 234. Treasury. The Intergenerational report. Australia to 2050: future challenges. Canberra: Commonwealth of Australia, 2010. Viewed 25 February 2013, http://archive.treasury.gov.au/igr/igr2010/report/pdf/IGR_2010.pdf.
- 235. National Preventative Health Taskforce. National Preventative Health Strategy. 2009. Viewed 27 February 2013, www.preventativehealth.org.au/internet/preventativehealth/publishing.nsf/Co ntent/national-preventative-health-strategy-11p.
- 236. Australian Government Department of Health and Ageing. Health assessment for people aged 75 years and older. 2012. Viewed 14 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/mbsprimarycare_m bsitem_75andolder.
- 237. U.S. Preventive Services Task Force (USPSTF). Preventive Services. 2008. Viewed 19 May 2009, www.ahrq.gov/clinic/prevenix.htm.
- 238. Grenz K, Mortinsen R, Pine D, Solberg L, Wilkinson JM, Harvey L et al. Preventive services for adults. ICSI health care guideline. Institute for Clinical Systems Improvement, 2008.
- 239. Canadian Task Force on Preventive Health Care. Canadian Task Force on Preventive Health Care recommendations. 2005. Viewed 27 February 2009, http://www.ctfphc.org/.
- 240. Singapore Health Promotion Board. Health screening. Singapore: Ministry of Health, 2004.
- 241. Schroder FH, Hugosson J, Roobol MJ, Tammela TL, Ciatto S, Nelen V et al. Screening and prostate-cancer mortality in a randomized European study. N Engl J Med 2009;360(13):1320-8.
- 242. Andriole GL, Crawford ED, Grubb RL, III, Buys SS, Chia D, Church TR et al. Mortality results from a randomized prostate-cancer screening trial. N Engl J Med 2009;360(13):1310-9.
- 243. Andriole GL, Crawford ED, Grubb RL, III, Buys SS, Chia D, Church TR et al. Prostate cancer screening in the randomized Prostate, Lung, Colorectal, and Ovarian Cancer Screening Trial: mortality results after 13 years of follow-up. J Natl Cancer Inst 2012;104(2):125-32.
- 244. Schroder FH, Hugosson J, Roobol MJ, Tammela TL, Ciatto S, Nelen V et al. Prostate-cancer mortality at 11 years of follow-up. N Engl J Med 2012;366(11):981-90.
- 245. Woods K. PSA predicament. Medical Observer 2009 May 8:19-20.
- 246. Miller AB. New data on prostate-cancer mortality after PSA screening. N Engl J Med 2012;366(11):1047-8.
- 247. Pogson J. PSA study results positive but the debate rolls on. Australian Doctor 2012 Mar 23:2.
- 248. Cancer Council Australia. Position statement: prostate cancer screening. 2010. Viewed 25 February 2013, www.cancer.org.au/content/pdf/CancerControlPolicy/PositionStatements/PS-

Prostate_Cancer_Screening_Joint_key_messages_%20published_May2010.p df.

- 249. Chapman S, Barratt A, Stockler M. Let sleeping dogs lie? : what men should know before getting tested for prostate cancer. Sydney: Sydney University Press, 2010.
- 250. MacKenzie R, Chapman S, Barratt A, Holding S. "The news is [not] all good": misrepresentations and inaccuracies in Australian news media reports on prostate cancer screening. Med J Aust 2007;187(9):507-10.
- 251. MacKenzie R, Chapman S, Holding S, McGeechan K. 'A matter of faith, not science': analysis of media coverage of prostate cancer screening in Australian news media 2003-2006. J R Soc Med 2007;100(11):513-21.
- 252. Prostate Cancer Foundation of Australia. PCFA Policy Statement: PSA/DRE testing for early detection of Prostate Cancer. 2006. Viewed 2 March 2013, www.prostate.org.au/articleLive/attachments/1/PCFA%20PSA%20%20DRE %20Policy%20Statement%20270706.pdf.
- 253. Costello AJ & Murphy DG. Has PSA testing truly been a "public health disaster"? Med J Aust 2010;193(1):4-5.
- 254. The Royal College of Pathologists of Australasia. RCPA announces position on PSA testing for prostate cancer [Media release]. 2011. Viewed 27 February 2013, www.rcpa.edu.au//static/File/Asset%20library/ public%20documents/Media%20Releases/2011/RCPA_announces_position_ on_PSA_testing_for_prostate_cancer.pdf.
- 255. Chapman S & Barratt A. Has PSA testing truly been a "public health disaster"? Med J Aust 2010;193(7):431.
- 256. The Royal College of Pathologists of Australasia. Organised PSA testing could protect Australian men. 2011. Viewed 27 February 2013, www.rcpa.edu.au//static/File/Asset%20library/public%20documents/Media%20 Releases/2011/Organised_PSA_testing_could_protect_Australian_men.pdf.
- 257. Australian Institute of Health and Welfare. Australian Cancer Incidence and Mortality (ACIM) books: Prostate cancer for Australia. 2012. Viewed 2 March 2013, www.aihw.gov.au/acim-books/.
- 258. Smellie WS, Vanderpump MP, Fraser WD, Bowley R, Shaw N. Best practice in primary care pathology: review 11. J Clin Pathol 2008;61(4):410-8.
- 259. O'Reilly DS. Thyroid function tests-time for a reassessment. BMJ 2000;320(7245):1332-4.
- 260. British Columbia Ministry of Health. Thyroid Disease -Thyroid Function Tests in the Diagnosis and Monitoring of Adults. 2010. Viewed 2 March 2013, www.bcguidelines.ca/gpac/pdf/thyroid.pdf.
- Ontario Association of Medical Laboratories. Guideline for the Use of Laboratory Tests to Detect Thyroid Dysfunction. 2007. Viewed 25 February 2013,

www.oaml.com/PDF/FINALTSH%20Guideline%20July%2018,%2007.pdf.

262. Association for Clinical Biochemistry, British Thyroid Association, British Thyroid Foundation. UK Guidelines for the Use of Thyroid Function Tests. 2006. Viewed 25 February 2013, www.british-thyroid-association.org/infofor-patients/Docs/TFT_guideline_final_version_July_2006.pdf.

- 263. Helfand M. Screening for subclinical thyroid dysfunction in nonpregnant adults: a summary of the evidence for the U.S. Preventive Services Task Force. Ann Intern Med 2004;140(2):128-41.
- 264. Danese MD, Powe NR, Sawin CT, Ladenson PW. Screening for mild thyroid failure at the periodic health examination: a decision and cost-effectiveness analysis. JAMA 1996;276(4):285-92.
- 265. Ladenson PW, Singer PA, Ain KB, Bagchi N, Bigos ST, Levy EG et al. American Thyroid Association guidelines for detection of thyroid dysfunction. Arch Intern Med 2000;160(11):1573-5.
- 266. Cameron AJ, Magliano DJ, Zimmet PZ, Welborn T, Shaw JE. The metabolic syndrome in Australia: prevalence using four definitions. Diabetes Res Clin Pract 2007;77(3):471-8.
- 267. Chadban SJ, Briganti EM, Kerr PG, Dunstan DW, Welborn TA, Zimmet PZ et al. Prevalence of kidney damage in Australian adults: The AusDiab kidney study. J Am Soc Nephrol 2003;14(7 Suppl 2):S131-S138.
- 268. Cancer Council Australia. Bowel cancer screening. 2012. Viewed 25 February 2013, www.cancer.org.au/cancer-control-policy/early-detectionpolicy/bowel-cancer-screening.html.
- 269. National Prescribing Service Ltd. Are you as healthy as your car. Medicines talk. No. 35. 2010. Viewed 25 February 2013, www.nps.org.au/__data/assets/pdf_file/0018/116514/MT_Spring_10_press_2 41210.pdf.
- 270. Cullen W, Kearney Y, Bury G. Prevalence of fatigue in general practice. Ir J Med Sci 2002;171(1):10-2.
- 271. David A, Pelosi A, McDonald E, Stephens D, Ledger D, Rathbone R et al. Tired, weak, or in need of rest: fatigue among general practice attenders. BMJ 1990;301(6762):1199-202.
- 272. Cathebras PJ, Robbins JM, Kirmayer LJ, Hayton BC. Fatigue in primary care: prevalence, psychiatric comorbidity, illness behavior, and outcome. J Gen Intern Med 1992;7(3):276-86.
- 273. Britt H, Miller GC, Charles J, Henderson J, Bayram C, Harrison C et al. General practice activity in Australia 1998-99 to 2007-08: 10 year data tables. General practice series no. 23. Cat. no. GEP 23. Canberra: Australian Institute of Health and Welfare, 2008.
- 274. Koch H, van Bokhoven MA, ter Riet G, van der Weijden T, Dinant GJ, Bindels PJ. Demographic characteristics and quality of life of patients with unexplained complaints: a descriptive study in general practice. Qual Life Res 2007;16(9):1483-9.
- 275. Ridsdale L, Evans A, Jerrett W, Mandalia S, Osler K, Vora H. Patients with fatigue in general practice: a prospective study. BMJ 1993;307(6896):103-6.
- 276. Godwin M, Delva D, Miller K, Molson J, Hobbs N, MacDonald S et al. Investigating fatigue of less than 6 months' duration. Guidelines for family physicians. Can Fam Physician 1999;45:373-9.
- 277. Ridsdale L, Mandalia S, Evans A, Jerrett W, Osler K. Tiredness as a ticket of entry. The role of patients' beliefs and psychological symptoms in explaining

frequent attendance. General Practice Fatigue Group. Scand J Prim Health Care 1999;17(2):72-4.

- 278. Nijrolder I, van der Windt D, deVries H, van der Horst H. Diagnoses during follow-up of patients presenting with fatigue in primary care. CMAJ 2009;181(10):683-7.
- 279. Dick ML & Sundin J. Psychological and psychiatric causes of fatigue. Assessment and management. Aust Fam Physician 2003;32(11):877-81.
- 280. van Bokhoven MA, Koch H, van der Weijden T, Grol RP, Bindels PJ, Dinant GJ. Blood test ordering for unexplained complaints in general practice: the VAMPIRE randomised clinical trial protocol. [ISRCTN55755886]. BMC Fam Pract 2006;7:20.
- 281. Koch H, van Bokhoven MA, ter Riet G, van Alphen-Jager JT, van der Weijden T, Dinant GJ et al. Ordering blood tests for patients with unexplained fatigue in general practice: what does it yield? Results of the VAMPIRE trial. Br J Gen Pract 2009;59(561):e93-100.
- 282. Murtagh J. Fatigue--a general diagnostic approach. Aust Fam Physician 2003;32(11):873-6.
- 283. bpac. The laboratory investigation of tiredness. 2006. Viewed 25 February 2013, www.bpac.org.nz/resources/campaign/tiredness/tiredness_poem.asp.
- 284. Sharpe M & Wilks D. Fatigue. BMJ 2002;325(7362):480-3.
- 285. The Royal College of Pathologists of Australasia & University of Sydney Department of Medical Education. RCPA case scenario - 'There must be something wrong'. 2003. Viewed 25 February 2013, www.rcpamanual.edu.au/index.php?option=com_content&view=category&la yout=blog&id=35&Itemid=60.
- 286. Cornuz J, Guessous I, Favrat B. Fatigue: a practical approach to diagnosis in primary care. CMAJ 2006;174(6):765-7.
- 287. Gialamas A, Beilby JJ, Pratt NL, Henning R, Marley JE, Roddick JF. Investigating tiredness in Australian general practice. Do pathology tests help in diagnosis? Aust Fam Physician 2003;32(8):663-6.
- 288. Koch H, van Bokhoven MA, ter Riet G, Hessels KM, van der Weijden T, Dinant GJ et al. What makes general practitioners order blood tests for patients with unexplained complaints? A cross-sectional study. Eur J Gen Pract 2009;15(1):22-8.
- 289. van Bokhoven MA, Koch H, van der Weijden T, Grol RP, Kester AD, Rinkens PE et al. Influence of watchful waiting on satisfaction and anxiety among patients seeking care for unexplained complaints. Ann Fam Med 2009;7(2):112-20.
- 290. Australian Government Department of Health and Ageing. Australian Health Ministers' Conference. 2008. Viewed 13 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/mr-yr08-deptdept180408.htm.
- 291. Colagiuri S, Lee CM, Colagiuri R, Magliano D, Shaw JE, Zimmet PZ et al. The cost of overweight and obesity in Australia. Med J Aust 2010;192(5):260-4.

- 292. Guh DP, Zhang W, Bansback N, Amarsi Z, Birmingham CL, Anis AH. The incidence of co-morbidities related to obesity and overweight: a systematic review and meta-analysis. BMC Public Health 2009;9:88.
- 293. Tham M & Young D. The role of the General Practitioner in weight management in primary care--a cross sectional study in General Practice. BMC Fam Pract 2008;9:66.
- 294. Waring ME, Roberts MB, Parker DR, Eaton CB. Documentation and management of overweight and obesity in primary care. J Am Board Fam Med 2009;22(5):544-52.
- 295. Australian Better Health Initiative. The Measure Up campaign. 2008. Viewed 13 February 2013, www.measureup.gov.au/internet/abhi/publishing.nsf/Content/Home.
- 296. Australian Government Department of Health and Ageing. Swap it. 2011. Viewed 13 February 2013, http://swapit.gov.au/.
- 297. National Health and Medical Research Council (NHMRC). Clinical practice guidelines for the management of overweight and obesity in adults. Canberra: NHMRC, 2004. Viewed 27 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/obesityguidelines-guidelines-adults.htm.
- 298. National Health and Medical Research Council (NHMRC). Overweight and obesity in adults: a guide for general practitioners. Canberra: NHMRC, 2004. Viewed 27 February 2013, www.health.gov.au/internet/main/publishing.nsf/ Content/obesityguidelines-guidelines-gp_guide.htm.
- 299. Lau DC, Douketis JD, Morrison KM, Hramiak IM, Sharma AM, Ur E. 2006 Canadian clinical practice guidelines on the management and prevention of obesity in adults and children [summary]. CMAJ 2007;176(8):S1-13.
- 300. National Obesity Forum. Guidelines on management of adult obesity and overweight in primary care. United Kingdom: National Obesity Forum, 2006.
- National Obesity Forum. Training resource for health care professionals: In depth assessment: obesity and overweight. United Kingdom: National Obesity Forum, 2004.
- 302. National Collaborating Centre for Primary Care & Centre for Public Health Excellence at NICE. Obesity: guidance on the prevention, identification, assessment and management of overweight and obesity in adults and children. NICE clinical guideline 43. London: National Institute for Health and Clinical Excellence, 2006.
- 303. National Collaborating Centre for Primary Care & Centre for Public Health Excellence at NICE. Obesity: guidance on the prevention, identification, assessment and management of overweight and obesity in adults and children: Quick reference guide 2 for the NHS. NICE clinical guideline 43. London: National Institute for Health and Clinical Excellence, 2006.
- 304. National Institutes of Health, National Heart, Lung and Blood Institute, North American Association for the Study of Obesity. Practical guide to the identification, evaluation, and treatment of overweight and obesity in adults. 2000. Viewed 13 February 2013, www.nhlbi.nih.gov/guidelines/obesity/practgde.htm.

- 305. Galuska DA, Will JC, Serdula MK, Ford ES. Are health care professionals advising obese patients to lose weight? JAMA 1999;282(16):1576-8.
- 306. Bardia A, Holtan SG, Slezak JM, Thompson WG. Diagnosis of obesity by primary care physicians and impact on obesity management. Mayo Clin Proc 2007;82(8):927-32.
- 307. Potter MB, Vu JD, Croughan-Minihane M. Weight management: what patients want from their primary care physicians. J Fam Pract 2001;50(6):513-8.
- 308. Booth AO & Nowson CA. Patient recall of receiving lifestyle advice for overweight and hypertension from their General Practitioner. BMC Fam Pract 2010;11:8.
- 309. Buffart LM, van der Ploeg HP, Smith BJ, Kurko J, King L, Bauman AE. General practitioners' perceptions and practices of physical activity counselling: changes over the past 10 years. Br J Sports Med 2009;43(14):1149-53.
- Leverence RR, Williams RL, Sussman A, Crabtree BF. Obesity counseling and guidelines in primary care: a qualitative study. Am J Prev Med 2007;32(4):334-9.
- 311. Ogden J & Flanagan Z. Beliefs about the causes and solutions to obesity: a comparison of GPs and lay people. Patient Educ Couns 2008;71(1):72-8.
- 312. Winkens RA, Ament AJ, Pop P, Reniers PH, Grol RP, Knottnerus JA. Routine individual feedback on requests for diagnostic tests: an economic evaluation. Med Decis Making 1996;16(4):309-14.
- 313. van Walraven C & Raymond M. Population-based study of repeat laboratory testing. Clin Chem 2003;49(12):1997-2005.
- 314. Grimshaw JM & Russell IT. Effect of clinical guidelines on medical practice: a systematic review of rigorous evaluations. Lancet 1993;342(8883):1317-22.
- 315. Grol R, Dalhuijsen J, Thomas S, Veld C, Rutten G, Mokkink H. Attributes of clinical guidelines that influence use of guidelines in general practice: observational study. BMJ 1998;317(7162):858-61.
- 316. Lugtenberg M, Burgers JS, Besters CF, Han D, Westert GP. Perceived barriers to guideline adherence: a survey among general practitioners. BMC Fam Pract 2011;12:98.
- 317. Lugtenberg M, Zegers-van Schaick JM, Westert GP, Burgers JS. Why don't physicians adhere to guideline recommendations in practice? An analysis of barriers among Dutch general practitioners. Implement Sci 2009;4:54.
- 318. Carlsen B & Norheim OF. "What lies beneath it all?"—an interview study of GPs' attitudes to the use of guidelines. BMC Health Serv Res 2008;8:218.
- 319. Carlsen B & Bringedal B. Attitudes to clinical guidelines--do GPs differ from other medical doctors? BMJ Qual Saf 2011;20(2):158-62.
- 320. Farquhar CM, Kofa EW, Slutsky JR. Clinicians' attitudes to clinical practice guidelines: a systematic review. Med J Aust 2002;177(9):502-6.
- 321. Cabana MD, Rand CS, Powe NR, Wu AW, Wilson MH, Abboud PA et al. Why don't physicians follow clinical practice guidelines? A framework for improvement. JAMA 1999;282(15):1458-65.

- 322. Langley C, Faulkner A, Watkins C, Gray S, Harvey I. Use of guidelines in primary care—practitioners' perspectives. Fam Pract 1998;15(2):105-11.
- 323. Watkins C, Harvey I, Langley C, Gray S, Faulkner A. General practitioners' use of guidelines in the consultation and their attitudes to them. Br J Gen Pract 1999;49(438):11-5.
- 324. Development and validation of an international appraisal instrument for assessing the quality of clinical practice guidelines: the AGREE project. Qual Saf Health Care 2003;12(1):18-23.
- 325. Buchan HA, Currie KC, Lourey EJ, Duggan GR. Australian clinical practice guidelines—a national study. Med J Aust 2010;192(9):490-4.
- 326. National Health and Medical Research Council (NHMRC). Clinical practice guidelines portal. 2012. Viewed 27 February 2013, www.clinicalguidelines.gov.au/.
- 327. van Wijk MA, Bohnen AM, van der Lei J. Analysis of the practice guidelines of the Dutch College of General Practitioners with respect to the use of blood tests. J Am Med Inform Assoc 1999;6(4):322-31.
- 328. Glasziou P, Irwig L, Mant D. Monitoring in chronic disease: a rational approach. BMJ 2005;330(7492):644-8.
- 329. Glasziou P, Aronson JK, Irwig L (eds). Evidence-based medical monitoring: from principles to practice. Oxford: Blackwell Publishing, 2008.
- 330. Britt HC, Harrison CM, Miller GC, Knox SA. Prevalence and patterns of multimorbidity in Australia. Med J Aust 2008;189(2):72-7.
- 331. Fortin M, Bravo G, Hudon C, Vanasse A, Lapointe L. Prevalence of multimorbidity among adults seen in family practice. Ann Fam Med 2005;3(3):223-8.
- 332. Dawes M. Co-morbidity: we need a guideline for each patient not a guideline for each disease. Fam Pract 2010;27(1):1-2.
- 333. Lugtenberg M, Burgers JS, Clancy C, Westert GP, Schneider EC. Current guidelines have limited applicability to patients with comorbid conditions: a systematic analysis of evidence-based guidelines. PLoS One 2011;6(10):e25987.
- 334. Vafiadis P. The dilemma of the unexpected result. Aust Fam Physician 1996;25(6):971-3.
- 335. Houben PH, van der Weijden T, Winkens B, Winkens RA, Grol RP. Pretest expectations strongly influence interpretation of abnormal laboratory results and further management. BMC Fam Pract 2010;11:13.
- 336. Houben PH, van der Weijden T, Winkens RA, Grol RP. Cascade effects of laboratory testing are found to be rare in low disease probability situations: prospective cohort study. J Clin Epidemiol 2010;63(4):452-8.
- 337. Smellie WS, Galloway MJ, Chinn D, Gedling P. Is clinical practice variability the major reason for differences in pathology requesting patterns in general practice? J Clin Pathol 2002;55(4):312-4.
- 338. Houben PH, Winkens RA, van der Weijden T, Vossen RC, Naus AJ, Grol RP. Reasons for ordering laboratory tests and relationship with frequency of abnormal results. Scand J Prim Health Care 2010;28(1):18-23.

- 339. Kravitz RL & Callahan EJ. Patients' perceptions of omitted examinations and tests: A qualitative analysis. J Gen Intern Med 2000;15(1):38-45.
- 340. Boulware LE, Marinopoulos S, Phillips KA, Hwang CW, Maynor K, Merenstein D et al. Systematic review: the value of the periodic health evaluation. Ann Intern Med 2007;146(4):289-300.
- Beaulieu MD, Hudon E, Roberge D, Pineault R, Forte D, Legare J. Practice guidelines for clinical prevention: do patients, physicians and experts share common ground? CMAJ 1999;161(5):519-23.
- 342. Hudon E, Beaulieu MD, Roberge D. Integration of the recommendations of the Canadian Task Force on Preventive Health Care: obstacles perceived by a group of family physicians. Fam Pract 2004;21(1):11-7.
- 343. Dubey V, Mathew R, Iglar K, Moineddin R, Glazier R. Improving preventive service delivery at adult complete health check-ups: the Preventive health Evidence-based Recommendation Form (PERFORM) cluster randomized controlled trial. BMC Fam Pract 2006;7:44.
- 344. Roxon N, Minister for Health and Ageing. Medicare red tape slashed [Media release]. Canberra: Department of Health and Ageing, 2009. Viewed 27 February 2013, www.health.gov.au/internet/ministers/publishing.nsf/Content/mr-yr09-nrnr234.htm?OpenDocument&yr=2009&mth=12.
- 345. Medicare Australia. MBS Group Statistics Reports. Report on Category 6 -Pathology Services, 2008-09 financial year. 2012. Viewed 25 February 2013, https://www.medicareaustralia.gov.au/statistics/mbs_group.shtml.
- 346. Wilson E. Pathology demand and funding. Presented at the Pathology Update 2011, Melbourne 4-6 Mar 2011; The Royal College of Pathologists of Australasia; 2011.
- 347. Keitz SA, Stechuchak KM, Grambow SC, Koropchak CM, Tulsky JA. Behind closed doors: management of patient expectations in primary care practices. Arch Intern Med 2007;167(5):445-52.
- 348. AIHW GP Statistics and Classification Unit. SAND abstract No. 43 from the BEACH program: Initiation and purpose of pathology orders. Pan Y (ed). Sydney: The University of Sydney, 2003. Viewed 25 February 2013, http://nla.gov.au/nla.arc-14007.
- 349. Takahashi O, Glasziou PP, Perera R, Shimbo T, Suwa J, Hiramatsu S et al. Lipid re-screening: what is the best measure and interval? Heart 2010;96(6):448-52.
- Dowling PT, Alfonsi G, Brown MI, Culpepper L. An education program to reduce unnecessary laboratory tests by residents. Acad Med 1989;64(7):410-2.
- 351. Dixon LR. The complete blood count: physiologic basis and clinical usage. J Perinat Neonatal Nurs 1997;11(3):1-18.
- 352. Boland BJ, Wollan PC, Silverstein MD. Yield of laboratory tests for casefinding in the ambulatory general medical examination. Am J Med 1996;101(2):142-52.

- 353. Ruttimann S, Clemencon D, Dubach UC. Usefulness of complete blood counts as a case-finding tool in medical outpatients. Ann Intern Med 1992;116(1):44-50.
- 354. Mates M, Heyd J, Souroujon M, Ben SA, Manny N, Hershko C. The haematologist as watchdog of community health by full blood count. QJM 1995;88(5):333-9.
- 355. Bell KJ, Kirby A, Hayen A, Irwig L, Glasziou P. Monitoring adherence to drug treatment by using change in cholesterol concentration: secondary analysis of trial data. BMJ 2011;342:d12.
- 356. Irwig L, Glasziou P, Wilson A, Macaskill P. Estimating an individual's true cholesterol level and response to intervention. JAMA 1991;266(12):1678-85.
- 357. Wagholikar A, O'Dwyer J, Hansen D, Chu K. Observing effectiveness of pathology ordering controls in emergency departments. Stud Health Technol Inform 2011;168:172-8.
- 358. Sucov A, Bazarian JJ, deLahunta EA, Spillane L. Test ordering guidelines can alter ordering patterns in an academic emergency department. J Emerg Med 1999;17(3):391-7.
- 359. Murtagh J. Fits, faints and funny turns. A general diagnostic approach. Aust Fam Physician 2003;32(4):203-6.
- 360. Metz A & Hebbard G. Nausea and vomiting in adults--a diagnostic approach. Aust Fam Physician 2007;36(9):688-92.
- 361. Doust J. Diagnostic and management strategies for anaemia in adults. Aust Fam Physician 2003;32(11):889-94.
- 362. MedicineNet. MedTerms Medical Dictionary: Differential diagnosis definition. 2008. Viewed 25 February 2013, www.medterms.com.
- Australian Medical Association. Position statement: Pathology 2011. 2011. Viewed 25 February 2013, http://ama.com.au/node/6745.
- 364. Kravitz RL, Bell RA, Franz CE. A taxonomy of requests by patients (TORP): a new system for understanding clinical negotiation in office practice. J Fam Pract 1999;48(11):872-8.
- 365. Kravitz RL, Bell RA, Azari R, Kelly-Reif S, Krupat E, Thom DH. Direct observation of requests for clinical services in office practice: what do patients want and do they get it? Arch Intern Med 2003;163(14):1673-81.
- 366. bpac. Complete blood count in primary care. 2008. Viewed 25 February 2013, www.bpac.org.nz/resources/campaign/cbc/cbc_poem.asp.
- 367. Doll H, Shine B, Kay J, James T, Glasziou P. The rise of cholesterol testing: how much is unnecessary. Br J Gen Pract 2011;61(583):e81-e88.
- 368. Australian Institute of Health and Welfare. Heart, stroke and vascular diseases: Australian facts 2004. Cat. no. CVD 27. Canberra: Australian Institute of Health and Welfare and National Heart Foundation of Australia, 2004.
- 369. Harrison C & Britt H. General practice workforce gaps now and in 2020. Aust Fam Physician 2011;40(1-2):12-5.

- 370. Jackevicius CA, Mamdani M, Tu JV. Adherence with statin therapy in elderly patients with and without acute coronary syndromes. JAMA 2002;288(4):462-7.
- Benner JS, Glynn RJ, Mogun H, Neumann PJ, Weinstein MC, Avorn J. Long-term persistence in use of statin therapy in elderly patients. JAMA 2002;288(4):455-61.
- 372. Simons LA, Levis G, Simons J. Apparent discontinuation rates in patients prescribed lipid-lowering drugs. Med J Aust 1996;164(4):208-11.
- 373. Shroufi A & Powles JW. Adherence and chemoprevention in major cardiovascular disease: a simulation study of the benefits of additional use of statins. J Epidemiol Community Health 2010;64(2):109-13.
- 374. Greving JP, Visseren FL, de Wit GA, Algra A. Statin treatment for primary prevention of vascular disease: whom to treat? Cost-effectiveness analysis. BMJ 2011;342:d1672.
- 375. Mann DM, Woodward M, Muntner P, Falzon L, Kronish I. Predictors of nonadherence to statins: a systematic review and meta-analysis. Ann Pharmacother 2010;44(9):1410-21.
- 376. Lucan SC. Coronary artery disease prevention: as easy as 1, 2, 3? Am Fam Physician 2010;82(10):1167.
- 377. United States National Library of Medicine. Secondary prevention Medical Subject Headings (MeSH) Descriptor Data. 2011. Viewed 1 October 2013, www.nlm.nih.gov/cgi/mesh/2011/MB_cgi?mode=&term=SECONDARY+PR EVENTION.
- 378. Zaat JO & van Eijk JT. General practitioners' uncertainty, risk preference, and use of laboratory tests. Med Care 1992;30(9):846-54.
- 379. DeKay ML & Asch DA. Is the defensive use of diagnostic tests good for patients, or bad? Med Decis Making 1998;18(1):19-28.
- 380. Epstein RM, Franks P, Shields CG, Meldrum SC, Miller KN, Campbell TL et al. Patient-centered communication and diagnostic testing. Ann Fam Med 2005;3(5):415-21.
- 381. Armitage T, Berry G, Mathews J. Statistical methods in medical research. 4th ed. London: Blackwell Science Ltd, 2002.
- 382. Harrison CM, Britt HC, Charles J. Sex of the GP 20 years on. Med J Aust 2011;195(4):192-6.
- 383. Medical Training Review Panel. Medical Training Review Panel fifteenth report. Canberra: Department of Health and Ageing, 2012. Viewed 25 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/work-pubs-mtrp-15.
- 384. Bubner T & Laurence C. Enhancing the quality use of pathology for GP Registrars and International Medical Graduates - assessing the need. 2009. Viewed 25 February 2013, www.health.gov.au/internet/main/Publishing.nsf/Content/9B6B7B4EAAC03 83DCA256F180048CCBD/\$File/Final%20version%20from%20Uni%20Adel aide%20Aug%2009.pdf.

- 385. Australian Institute of Health and Welfare. Health expenditure by remoteness. Health and Welfare Expenditure Series no 50. Cat. no. HWE 50. Canberra: AIHW, 2011.
- 386. Winkens RA, Pop P, Grol RP, Bugter-Maessen AM, Kester AD, Beusmans GH et al. Effects of routine individual feedback over nine years on general practitioners' requests for tests. BMJ 1996;312(7029):490.
- 387. Fitzgerald PD. The ethics of doctors and big business. Med J Aust 2001;175(2):73-5.
- 388. Catchlove BR. The why and the wherefore. Med J Aust 2001;175(2):68-70.
- 389. Fitzgerald PD. General practice corporatisation: the half-time score. Med J Aust 2002;177(2):90-2.
- 390. Australian Medical Association. General practice corporatisation: AMA scoping paper (November 2000). 2002. Viewed 25 February 2013, http://ama.com.au/node/3748.
- Professional Services Review. Professional Services Review: annual report 1999-2000. 2000. Viewed 25 February 2013, www.psr.gov.au/publicationsand-resources/annual-reports.
- 392. Studdert DM, Britt HC, Pan Y, Fahridin S, Bayram CF, Gurrin LC. Are rates of pathology test ordering higher in general practices co-located with pathology collection centres? Med J Aust 2010;193(2):114-9.
- 393. Australian Government. Health Insurance (Eligible Collection Centres) Approval Principles 2010. 2010. Viewed 25 February 2013, www.comlaw.gov.au/Details/F2010L01765.
- 394. Australian Government. Health Insurance Amendment (Inappropriate and Prohibited Practices and Other Measures) Act 2007. 2007. Viewed 3 March 2013, http://www.comlaw.gov.au/Details/C2007A00088.
- 395. Treasury. Australia's Demographic Challenges. Canberra: Commonwealth of Australia, 2004. Viewed 25 February 2013, http://demographics.treasury.gov.au/content/_download/australias_demograp hic_challenges/australias_demographic_challenges.pdf.
- 396. Australian Bureau of Statistics. Australian Social Trends, Mar 2010. Cat. no. 4102.0. Canberra: Australian Bureau of Statistics, 2010. Viewed 25 February 2013, www.abs.gov.au/AUSSTATS/abs@.nsf/Lookup/ 4102.0Main+Features30Mar+2010#10.
- 397. Organization for Economic Co-operation and Development (OECD). Value for money in health spending. Paris. OECD, 2010.
- 398. Organization for Economic Co-operation and Development (OECD). Live Longer, Work Longer: A synthesis report. Ageing and Employment Policies Project. Paris: OECD, 2006.
- 399. Australian Government Department of Health and Ageing (Medical Benefits Reviews Task Group). Review of the funding for pathology services. Final discussion paper. 2011. Viewed 22 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/A7F7F7F94475760 7CA2576A2001C1725/\$File/Pathology%20Review%20Final%20Discussion %20Paper%20-%20final%20-%201%20March%202011%20v2.pdf.

- 400. Australian Government Department of Veterans' Affairs. Annual report of the Department of Veterans' Affairs 2005-06: Outcome 2 Health. Canberra: DVA, 2006. Viewed 3 March 2013, www.dva.gov.au/aboutDVA/publications/corporate/annualreport/2009-2010/Documents/DVA_AR_09-10_access.pdf.
- 401. Australian Government Department of Veterans' Affairs. Annual report of the Department of Veterans' Affairs 2009-10: Outcome 2 - Health. Canberra: DVA, 2010. Viewed 3 March 2013, www.dva.gov.au/aboutDVA/publications/corporate/annualreport/2009-2010/Pages/index.aspx.
- 402. Australian Bureau of Statistics. National Health Survey: Summary of results, 2004-05. Cat. no. 4364.0. Canberra: Australian Bureau of Statistics, 2006.
- 403. Cancer Council Australia. Cervical cancer vaccine. 2011. Viewed 25 February 2013, www.hpvvaccine.org.au/the-hpv-vaccine/vaccinebackground.aspx.
- 404. Cancer Council Australia. Report on Partner reference group consultation, national cervical screening program renewal. 2012. Viewed 21 February 2013, www.cancerscreening.gov.au/internet/screening/publishing.nsf/Content/cvconsult-rpt-15-04-12.
- 405. Lee CM, Colagiuri R, Magliano DJ, Cameron AJ, Shaw J, Zimmet P et al. The cost of diabetes in adults in Australia. Diabetes Res Clin Pract 2013.
- 406. National Vascular Disease Prevention Alliance. New guidelines and treatment procedures for heart attack and stroke [Media release]. 2012. Viewed 16 February 2013, www.diabetesaustralia.com.au/Documents/DA/Media%20Releases/NVDPA %20guidelines-May2012-General%20Release%20FINAL.pdf.
- 407. Harvey K, Clark RB, Visser S. A mechanism for the development, implementation and evaluation of evidence-based, best practice clinical guidelines to facilitate quality use of pathology tests. Quality Use of Pathology Program, Commonwealth Department of Health and Ageing, 2003.
- 408. Gialamas A, Beilby JJ, Pratt NL, Marley JE, Roddick JF, Henning R. A project to examine the utilisation of pathology tests in the investigation of tiredness in general practice. Quality Use of Pathology Program, Commonwealth Department of Health and Aged Care, 2002.
- 409. van Bokhoven MA, Koch H, Dinant GJ, Bindels PJ, Grol RP, van der Weijden T. Exploring the black box of change in improving test-ordering routines. Fam Pract 2008;25(3):139-45.
- 410. Pathology Review Advisory Group. Report of the second phase of the review of NHS pathology services in England (The Carter report 2nd phase). 2008.
- 411. Kennedy PJ, Leathley CM, Hughes CF. Clinical practice variation. Med J Aust 2010;193(8 Suppl):S97-S99.
- 412. Fung V, Schmittdiel JA, Fireman B, Meer A, Thomas S, Smider N et al. Meaningful variation in performance: a systematic literature review. Med Care 2010;48(2):140-8.

- 413. Selby JV, Schmittdiel JA, Lee J, Fung V, Thomas S, Smider N et al. Meaningful variation in performance: what does variation in quality tell us about improving quality? Med Care 2010;48(2):133-9.
- 414. Appleby J, Ragleigh V, Frosini F, Bevan G, Gao H, Lyscom T. Variations in health care: the good, the bad and the inexplicable. London: The King's Fund, 2011. Viewed 25 February 2013, http://www.kingsfund.org.uk/publications/healthcare_variation.html.
- 415. Glasziou P. How much monitoring? Br J Gen Pract 2007;57(538):350-1.

References in alphabetical order (citation order in parentheses)

Aakre KM, Thue G, Subramaniam-Haavik S, Bukve T, Morris H, Muller M et al. Postanalytical external quality assessment of urine albumin in primary health care: an international survey. Clin Chem 2008;54(10):1630-6. (202)

AIHW GP Statistics and Classification Unit. SAND abstract No. 43 from the BEACH program: Initiation and purpose of pathology orders. Pan Y (ed). Sydney: The University of Sydney, 2003. Viewed 25 February 2013, http://nla.gov.au/nla.arc-14007. (348)

American Diabetes Association. Standards of medical care in diabetes-2008. Diabetes Care 2008;31 Suppl 1:S12-S54. (197)

Andriole GL, Crawford ED, Grubb RL, III, Buys SS, Chia D, Church TR et al. Mortality results from a randomized prostate-cancer screening trial. N Engl J Med 2009;360(13):1310-9. (242)

Andriole GL, Crawford ED, Grubb RL, III, Buys SS, Chia D, Church TR et al. Prostate cancer screening in the randomized Prostate, Lung, Colorectal, and Ovarian Cancer Screening Trial: mortality results after 13 years of follow-up. J Natl Cancer Inst 2012;104(2):125-32. (243)

Appleby J, Ragleigh V, Frosini F, Bevan G, Gao H, Lyscom T. Variations in health care: the good, the bad and the inexplicable. London: The King's Fund, 2011. Viewed 25 February 2013,

http://www.kingsfund.org.uk/publications/healthcare_variation.html. (414)

Armitage T, Berry G, Mathews J. Statistical methods in medical research. 4th ed. London: Blackwell Science Ltd, 2002. (381)

Association for Clinical Biochemistry, British Thyroid Association, British Thyroid Foundation. UK Guidelines for the Use of Thyroid Function Tests. 2006. Viewed 25 February 2013, www.british-thyroid-association.org/info-forpatients/Docs/TFT_guideline_final_version_July_2006.pdf. (262)

Austin PC & Hux JE. A brief note on overlapping confidence intervals. J Vasc Surg 2002;36(1):194-5. (132)

Australian Association of Pathology Practices Inc. AAPP submission to Department of Health and Ageing's Review of funding arrangements of pathology services. 2010. Viewed 8 February 2013,

www.aapp.asn.au/images/document/AAPP%20Funding%20Submission%20-%20May%202010.pdf. (19)

Australian Better Health Initiative. The Measure Up campaign. 2008. Viewed 13 February 2013, www.measureup.gov.au/internet/abhi/publishing.nsf/Content/Home. (295) Australian Bureau of Statistics. Australian Social Trends, Mar 2010. Cat. no. 4102.0. Canberra: Australian Bureau of Statistics, 2010. Viewed 25 February 2013, www.abs.gov.au/AUSSTATS/abs@.nsf/Lookup/4102.0Main+Features30Mar+2010# 10. (396)

Australian Bureau of Statistics. Australian Standard Geographical Classification. Cat. no. 1216.0. Canberra: ABS, 2008. (107)

Australian Bureau of Statistics. Health services: patient experiences in Australia, 2009. Cat. no. 4839.0.55.001. Canberra: Australian Bureau of Statistics, 2010. Viewed 27 February 2013,

www.ausstats.abs.gov.au/Ausstats/subscriber.nsf/0/4DE1EBF2E29AAC71CA25777 30022B837/\$File/4839055001_2009.pdf. (141)

Australian Bureau of Statistics. National Health Survey: Summary of results, 2004-05. Cat. no. 4364.0. Canberra: Australian Bureau of Statistics, 2006. (402)

Australian Bureau of Statistics. National Health Survey: Summary of Results, 2007-08. Cat. no. 4364.0. Canberra: Australian Bureau of Statistics, 2009. Viewed 25 February 2013,

www.abs.gov.au/ausstats/abs@.nsf/Latestproducts/4364.0Main%20Features42007-2008%20(Reissue)?opendocument&tabname=Summary&prodno=4364.0&issue=2007-2008%20(Reissue)&num=&view=. (154)

Australian Bureau of Statistics. Population by Age and Sex, Australian States and Territories, Jun 2010. Cat. no. 3201.0. Canberra: Australian Bureau of Statistics, 2010. (15)

Australian Bureau of Statistics. Population Projections, Australia, 2006 to 2101. Cat. no. 3222.0. Canberra: Australian Bureau of Statistics, 2008. (137)

Australian General Practice Statistics and Classification Centre. SAND abstract No. 148 from the BEACH program: Type 2 diabetes, and blood glucose, lipid and blood pressure medication management. 2010. Viewed 27 February 2013, http://sydney.edu.au/medicine/fmrc/publications/sand-abstracts/148-Type_2_diabetes.pdf. (191)

Australian General Practice Statistics and Classification Centre. SAND abstract No. 149 from the BEACH program: Dyslipidaemia and lipid management. 2010. Viewed 27 February 2013, http://sydney.edu.au/medicine/fmrc/publications/sand-abstracts/149-Dyslipidaemia.pdf. (209)

Australian Government Department of Health and Ageing (Medical Benefits Reviews Task Group). First Discussion Paper for Review of Funding of Pathology. 2010. Viewed 14 February 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/A7F7F7F944757607CA257 6A2001C1725/\$File/Pathology%20Review%20Discussion%20Paper.pdf. (20)

Australian Government Department of Health and Ageing (Medical Benefits Reviews Task Group). Review of the funding for pathology services. Final discussion paper. 2011. Viewed 22 February 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/A7F7F7F944757607CA257 6A2001C1725/\$File/Pathology%20Review%20Final%20Discussion%20Paper%20-%20final%20-%201%20March%202011%20v2.pdf. (399) Australian Government Department of Health and Ageing. Australian Health Ministers' Conference. 2008. Viewed 13 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/mr-yr08-deptdept180408.htm. (290)

Australian Government Department of Health and Ageing. Cardiovascular disease. 2012. Viewed 27 February 2013,

http://health.gov.au/internet/main/publishing.nsf/Content/chronic-cardio. (155)

Australian Government Department of Health and Ageing. COAG - reducing the risk of type 2 diabetes. 2007. Viewed 27 February 2013,

www.health.gov.au/internet/budget/publishing.nsf/Content/budget2007-hfact31.htm. (186)

Australian Government Department of Health and Ageing. Health assessment for people aged 45 to 49 years who are at risk of developing chronic disease. 2012. Viewed 14 February 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/mbsprimarycare_mbsitem7 01_703_705_707. (232)

Australian Government Department of Health and Ageing. Health assessment for people aged 75 years and older. 2012. Viewed 14 February 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/mbsprimarycare_mbsitem_ 75andolder. (236)

Australian Government Department of Health and Ageing. MBS primary care items: History of key MBS primary care initiatives 1999-2010. 2011. Viewed 25 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/mbsprimarycare-History. (18)

Australian Government Department of Health and Ageing. Medicare Benefits Schedule Book. Canberra: DoHA, 2007. (23)

Australian Government Department of Health and Ageing. Medicare statistics, March quarter 2012, Group BA tables. 2012. Viewed 25 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/medstat-mar12-tables-ba. (5)

Australian Government Department of Health and Ageing. Medicare statistics, March quarter 2012, Group BB tables. 2012. Viewed 25 February 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/medstat-mar12-tables-bb. (13)

Australian Government Department of Health and Ageing. National chronic disease strategy. 2005. Viewed 27 February 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/pq-ncds-strat. (187)

Australian Government Department of Health and Ageing. Pathology funding agreement between the Australian Government and the Australian Association of Pathology Practices and the Royal College of Pathologists of Australasia and the National Coalition of Public Pathology. 2011. Viewed 25 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/98BD1271A55D8863CA25 7642001C643C/\$File/Pathology%20Funding%20Agreement.pdf. (25) Australian Government Department of Health and Ageing. Pathology Quality and Outlays Memorandum of Understanding (MoU). 2005. Viewed 27 February 2013, http://pandora.nla.gov.au/pan/13326/20080318-

0005/www.health.gov.au/internet/main/publishing.nsf/content/5762a7197be425efca2 56f1800469fa1/\$file/pathmou.pdf. (24)

Australian Government Department of Health and Ageing. PBS-eligibility criteria for cholesterol lowering medicines. 2006. Viewed 27 February 2013,

www.health.gov.au/internet/main/publishing.nsf/content/lipid_eligibilitycriteria.htm. (208)

Australian Government Department of Health and Ageing. Prevention of Type 2 Diabetes Program. 2010. Viewed 27 February 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/chronic-diab-prev-mbs. (189)

Australian Government Department of Health and Ageing. Quality Use of Pathology Program. 2013. Viewed 1 October 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/pathology-qupp-index. (101)

Australian Government Department of Health and Ageing. Swap it. 2011. Viewed 13 February 2013, http://swapit.gov.au/. (296)

Australian Government Department of Health and Ageing. Work of the Australian Government on diabetes. 2011. Viewed 27 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/pq-diabetes-gov. (188)

Australian Government Department of Veterans' Affairs. Annual report of the Department of Veterans' Affairs 2005-06: Outcome 2 - Health. Canberra: DVA, 2006. Viewed 3 March 2013,

www.dva.gov.au/aboutDVA/publications/corporate/annualreport/2009-2010/Documents/DVA_AR_09-10_access.pdf. (400)

Australian Government Department of Veterans' Affairs. Annual report of the Department of Veterans' Affairs 2009-10: Outcome 2 - Health. Canberra: DVA, 2010. Viewed 3 March 2013,

www.dva.gov.au/aboutDVA/publications/corporate/annualreport/2009-2010/Pages/index.aspx. (401)

Australian Government, Office of the Australian Information Commissioner. Privacy in the Private Health Sector. 2001. Viewed 27 February 2013, www.privacy.gov.au/materials/types/download/8675/6517. (140)

Australian Government. Health Insurance (Eligible Collection Centres) Approval Principles 2010. 2010. Viewed 25 February 2013, www.comlaw.gov.au/Details/F2010L01765. (393)

Australian Government. Health Insurance Amendment (Inappropriate and Prohibited Practices and Other Measures) Act 2007. 2007. Viewed 3 March 2013, http://www.comlaw.gov.au/Details/C2007A00088. (394)

Australian Institute of Health and Welfare & Commonwealth Department of Health and Family Services. First report on National Health Priority Areas 1996. Cat. no. PHE 1. Canberra: AIHW & DHFS, 1997. (184) Australian Institute of Health and Welfare. Australian Cancer Incidence and Mortality (ACIM) books: Prostate cancer for Australia. 2012. Viewed 2 March 2013, www.aihw.gov.au/acim-books/. (257)

Australian Institute of Health and Welfare. Australian family of health and related classifications matrix. Canberra: AIHW, 2003. Viewed 27 February 2013, www.aihw.gov.au/WorkArea/DownloadAsset.aspx?id=6442475388&libID=6442475 369. (112)

Australian Institute of Health and Welfare. Diabetes: Australian facts 2008. Cat. no. CVD 40. Canberra: AIHW, 2008. (181)

Australian Institute of Health and Welfare. Health expenditure Australia 2009-10. Health and Welfare Expenditure Series no 46. Cat. no. HWE 55. Canberra: AIHW, 2011. (14)

Australian Institute of Health and Welfare. Health expenditure by remoteness. Health and Welfare Expenditure Series no 50. Cat. no. HWE 50. Canberra: AIHW, 2011. (385)

Australian Institute of Health and Welfare. Heart, stroke and vascular diseases: Australian facts 2004. Cat. no. CVD 27. Canberra: Australian Institute of Health and Welfare and National Heart Foundation of Australia, 2004. (368)

Australian Institute of Health and Welfare. Medical labour force 2008. Cat. no. AUS 131. Canberra: AIHW, 2010. (16)

Australian Institute of Health and Welfare. National Health Priority Areas. 2012. Viewed 27 February 2013, www.aihw.gov.au/health-priority-areas/. (151)

Australian Medical Association. General practice corporatisation: AMA scoping paper (November 2000). 2002. Viewed 25 February 2013, http://ama.com.au/node/3748. (390)

Australian Medical Association. Position statement: Pathology - 2011. 2011. Viewed 25 February 2013, http://ama.com.au/node/6745. (363)

Australian Medical Workforce Advisory Committee. The general practice workforce in Australia: supply and requirements to 2013. Number AMWAC Report 2005.2. Sydney: AMWAC, 2005. (139)

Australian National Audit Office. Administration of the Pathology Quality and Outlays Memorandum of Understanding: Department of Health and Ageing. 2008. Viewed 27 February 2013, www.anao.gov.au/Publications/Audit-Reports/2007-2008/Administration-of-the-Pathology-Quality-and-Outlays-Memorandum-of-Understanding. (26)

Australian National Audit Office. Practice Incentives Program. 2010. Viewed 27 February 2013, www.anao.gov.au/Publications/Audit-Reports/2010-2011/Practice-Incentives-Program. (138)

Australian Primary Care Collaboratives. Australian Primary Care Collaboratives. 2008. Viewed 27 February 2013, www.apcc.org.au/about_the_APCC/. (190)

Axt-Adam P, van der Wouden JC, van der Does E. Influencing behavior of physicians ordering laboratory tests: a literature study. Med Care 1993;31(9):784-94. (93)

Bailey J, Jennings A, Parapia L. Change of pathology request forms can reduce unwanted requests and tests. J Clin Pathol 2005;58(8):853-5. (80)

Bardia A, Holtan SG, Slezak JM, Thompson WG. Diagnosis of obesity by primary care physicians and impact on obesity management. Mayo Clin Proc 2007;82(8):927-32. (306)

Barrie WE, Harrison RV, Khanderia UB, Kiningham RB, Rosenson RS. Screening and Management of Lipids. University of Michigan, 2009. Viewed 27 February 2013, www.med.umich.edu/1info/fhp/practiceguides/lipids/lipidsupdate.pdf. (216)

Bayram C & Valenti L. GP pathology ordering. In: Britt H & Miller GC (eds). General practice in Australia, health priorities and policies 1998-2008. General practice series no. 24. Cat. no. GEP 24. Canberra: Australian Institute of Health and Welfare, 2009;57-86. (1)

Bayram C, Britt H, Kelly Z, Valenti L. Male consultations in general practice in Australia 1999-00. General practice series no. 11. Cat. no. GEP 11. Canberra: Australian Institute of Health and Welfare, 2003. (3)

Bayram C, Britt H, Miller G, Valenti L. Evidence-practice gap in GP pathology test ordering: a comparison of BEACH pathology data and recommended testing. Sydney: The University of Sydney, 2009. Viewed 7 March 2013, www.health.gov.au/internet/main/Publishing.nsf/Content/9B6B7B4EAAC0383DCA 256F180048CCBD/\$File/BEACH_QUPP%20report_FINAL_050609.pdf. (2)

Beaulieu MD, Hudon E, Roberge D, Pineault R, Forte D, Legare J. Practice guidelines for clinical prevention: do patients, physicians and experts share common ground? CMAJ 1999;161(5):519-23. (341)

Begg S, Vos T, Barker B, Stevenson C, Stanley L, Lopez AD. The burden of disease and injury in Australia 2003. Cat. no. PHE 82. Canberra: Australian Institute of Health and Welfare, 2007. (156)

Bell KJ, Irwig L, Craig JC, Macaskill P. Use of randomised trials to decide when to monitor response to new treatment. BMJ 2008;336(7640):361-5. (225)

Bell KJ, Kirby A, Hayen A, Irwig L, Glasziou P. Monitoring adherence to drug treatment by using change in cholesterol concentration: secondary analysis of trial data. BMJ 2011;342:d12. (355)

Benner JS, Glynn RJ, Mogun H, Neumann PJ, Weinstein MC, Avorn J. Long-term persistence in use of statin therapy in elderly patients. JAMA 2002;288(4):455-61. (371)

Bindels R, De Clercq PA, Winkens RA, Hasman A. A test ordering system with automated reminders for primary care based on practice guidelines. Int J Med Inform 2000;58-59:219-33. (59)

Bindels R, Hasman A, van Wersch JW, Talmon J, Winkens RA. Evaluation of an automated test ordering and feedback system for general practitioners in daily practice. Int J Med Inform 2004;73(9-10):705-12. (86)

Boland BJ, Wollan PC, Silverstein MD. Yield of laboratory tests for case-finding in the ambulatory general medical examination. Am J Med 1996;101(2):142-52. (352)

Booth AO & Nowson CA. Patient recall of receiving lifestyle advice for overweight and hypertension from their General Practitioner. BMC Fam Pract 2010;11:8. (308)

Bossuyt PM & McCaffery K. Additional patient outcomes and pathways in evaluations of testing. Med Decis Making 2009;29(5):E30-E38. (63)

Boulware LE, Marinopoulos S, Phillips KA, Hwang CW, Maynor K, Merenstein D et al. Systematic review: the value of the periodic health evaluation. Ann Intern Med 2007;146(4):289-300. (340)

bpac. Complete blood count in primary care. 2008. Viewed 25 February 2013, www.bpac.org.nz/resources/campaign/cbc/cbc_poem.asp. (366)

bpac. The laboratory investigation of tiredness. 2006. Viewed 25 February 2013, www.bpac.org.nz/resources/campaign/tiredness/tiredness_poem.asp. (283)

Bridges-Webb C, Britt H, Miles DA, Neary S, Charles J, Traynor V. Morbidity and treatment in general practice in Australia 1990-1991. Med J Aust 1992;157(19 Oct Spec Sup):S1-S56. (105)

British Columbia Ministry of Health. Thyroid Disease -Thyroid Function Tests in the Diagnosis and Monitoring of Adults. 2010. Viewed 2 March 2013, www.bcguidelines.ca/gpac/pdf/thyroid.pdf. (260)

Britt H, Harris M, Driver B, Bridges-Webb C, O'Toole B, Neary S. Reasons for encounter and diagnosed health problems: convergence between doctors and patients. Fam Pract 1992;9(2):191-4. (117)

Britt H, Knox S, Miller GC. Changes in pathology ordering by general practitioners in Australia 1998-2001. General practice series no. 13. Cat. no. GEP 13. Canberra: Australian Institute of Health and Welfare, 2003. (45)

Britt H, Miller G, Charles J, Henderson J, Bayram C, Pan Y et al. General practice activity in Australia 2008-09. General practice series no. 25. Cat. no. GEP 25. Canberra: Australian Institute of Health and Welfare, 2009. (122)

Britt H, Miller GC, Charles J, Bayram C, Pan Y, Henderson J et al. General practice activity in Australia 2006-07. General practice series no. 21. Cat. no. GEP 21. Canberra: Australian Institute of Health and Welfare, 2008. (120)

Britt H, Miller GC, Charles J, Henderson J, Bayram C, Harrison C et al. General practice activity in Australia 1998-99 to 2007-08: 10 year data tables. General practice series no. 23. Cat. no. GEP 23. Canberra: Australian Institute of Health and Welfare, 2008. (273)

Britt H, Miller GC, Charles J, Henderson J, Bayram C, Harrison C et al. General practice activity in Australia 2007-08. General practice series no. 22. Cat. no. GEP 22. Canberra: Australian Institute of Health and Welfare, 2008. (121)

Britt H, Miller GC, Charles J, Henderson J, Bayram C, Pan Y et al. General practice activity in Australia 2009-10. General practice series no. 27. Cat. no. GEP 27. Canberra: Australian Institute of Health and Welfare, 2010. (17)

Britt H, Miller GC, Charles J, Henderson J, Bayram C, Valenti L et al. General practice activity in Australia 2000-01 to 2009-10: 10 year data tables. General practice series no. 28. Cat. no. GEP 28. Canberra: Australian Institute of Health and Welfare, 2010. (103)

Britt H, Miller GC, Charles J, Pan Y, Valenti L, Henderson J et al. General practice activity in Australia 2005-06. General practice series no. 19. Cat. no. GEP 19. Canberra: Australian Institute of Health and Welfare, 2007. (123)

Britt H, Miller GC, Knox S, Charles J, Pan Y, Henderson J et al. General practice activity in Australia 2004-05. General practice series no. 18. Cat. no. GEP 18. Canberra: Australian Institute of Health and Welfare, 2005. (124)

Britt H, Miller GC, Knox S, Charles J, Valenti L, Henderson J et al. General practice activity in Australia 2000-01. General practice series no. 8. Cat. no. GEP 8. Canberra: Australian Institute of Health and Welfare, 2001. (125)

Britt H, Miller GC, Knox S, Charles J, Valenti L, Henderson J et al. General practice activity in Australia 2001-02. General practice series no. 10. Cat. no. GEP 10. Canberra: Australian Institute of Health and Welfare, 2002. (126)

Britt H, Miller GC, Knox S, Charles J, Valenti L, Henderson J et al. General practice activity in Australia 2002-03. General practice series no. 14. Cat. no. GEP 14. Canberra: Australian Institute of Health and Welfare, 2003. (127)

Britt H, Miller GC, Knox S, Charles J, Valenti L, Pan Y et al. General practice activity in Australia 2003-04. General practice series no. 16. Cat. no. GEP 16. Canberra: Australian Institute of Health and Welfare, 2004. (128)

Britt H, Miller GC, McGeechan K, Sayer GP. Pathology ordering by general practitioners in Australia 1998. Cat. no. GEP 4. Canberra: Department of Health and Aged Care, 1999. Viewed 27 February 2013, http://pandora.nla.gov.au/nph-wb/20001220130000/http://www.health.gov.au/haf/docs/pathorder.htm. (89)

Britt H, Sayer GP, Miller GC, Charles J, Scahill S, Horn F et al. BEACH Bettering the Evaluation and Care of Health: A study of general practice activity, six-month interim report. General practice series no. 1. Cat. no. GEP 1. Canberra: Australian Institute of Health and Welfare, 1999. (104)

Britt H, Sayer GP, Miller GC, Charles J, Scahill S, Horn F et al. General practice activity in Australia 1998-99. General practice series no. 2. Cat. no. GEP 2. Canberra: Australian Institute of Health and Welfare, 1999. (108)

Britt H. A measure of the validity of the ICPC in the classification of reasons for encounter. Journ Informatics in Prim Care 1997;Nov:8-12. (119)

Britt H. A new coding tool for computerised clinical systems in primary care--ICPC plus. Aust Fam Physician 1997;26(Suppl 2):S79-S82. (113)

Britt H. Reliability of central coding of patient reasons for encounter in general practice, using the International Classification of Primary Care. Journ Informatics in Prim Care 1998;May:3-7. (118)

Britt HC, Harrison CM, Miller GC, Knox SA. Prevalence and patterns of multimorbidity in Australia. Med J Aust 2008;189(2):72-7. (330)

Bubner T & Laurence C. Enhancing the quality use of pathology for GP Registrars and International Medical Graduates - assessing the need. 2009. Viewed 25 February 2013,

www.health.gov.au/internet/main/Publishing.nsf/Content/9B6B7B4EAAC0383DCA 256F180048CCBD/\$File/Final%20version%20from%20Uni%20Adelaide%20Aug% 2009.pdf. (384)

Buchan HA, Currie KC, Lourey EJ, Duggan GR. Australian clinical practice guidelines—a national study. Med J Aust 2010;192(9):490-4. (325)

Buffart LM, van der Ploeg HP, Smith BJ, Kurko J, King L, Bauman AE. General practitioners' perceptions and practices of physical activity counselling: changes over the past 10 years. Br J Sports Med 2009;43(14):1149-53. (309)

Bugter-Maessen AM, Winkens RA, Grol RP, Knottnerus JA, Kester AD, Beusmans GH et al. Factors predicting differences among general practitioners in test ordering behaviour and in the response to feedback on test requests. Fam Pract 1996;13(3):254-8. (71)

Bunting PS & van Walraven C. Effect of a controlled feedback intervention on laboratory test ordering by community physicians. Clin Chem 2004;50(2):321-6. (96)

Buntinx F, Winkens R, Grol R, Knottnerus JA. Influencing diagnostic and preventive performance in ambulatory care by feedback and reminders. A review. Fam Pract 1993;10(2):219-28. (92)

Burdon J. Medical Indemnity Insurance 2006. Presented at the Australian College of Legal Medicine's Annual Scientific Meeting, Melbourne 14-15 October 2006; Australian College of Legal Medicine; 2006. (146)

Burnett L & Sugerman DA. Restrictions on 'multiple biochemical analyses' profiling decreases usage of biochemistry test requests. Pathology 1991;23(2):103-6. (98)

Cabana MD, Rand CS, Powe NR, Wu AW, Wilson MH, Abboud PA et al. Why don't physicians follow clinical practice guidelines? A framework for improvement. JAMA 1999;282(15):1458-65. (321)

Cagliero E, Levina EV, Nathan DM. Immediate feedback of HbA1c levels improves glycemic control in type 1 and insulin-treated type 2 diabetic patients. Diabetes Care 1999;22(11):1785-9. (64)

Calcino G. Variability in ordering of pathology by general practitioners and factors related to the variability. Graduate Diploma in Medical Statistics thesis. The University of Newcastle, 1997. (43)

Cameron AJ, Magliano DJ, Zimmet PZ, Welborn T, Shaw JE. The metabolic syndrome in Australia: prevalence using four definitions. Diabetes Res Clin Pract 2007;77(3):471-8. (266)

Canadian Diabetes Association Clinical Practice Guidelines Expert Committee CDA. Canadian Diabetes Association 2008 Clinical Practice Guidelines for the Prevention and Management of Diabetes in Canada. Can J Diabetes 2008;32(Supp 1):s1-s201. (194)

Canadian Task Force on Preventive Health Care. Canadian Task Force on Preventive Health Care recommendations. 2005. Viewed 27 February 2009, http://www.ctfphc.org/. (239)

Cancer Council Australia. Bowel cancer screening. 2012. Viewed 25 February 2013, www.cancer.org.au/cancer-control-policy/early-detection-policy/bowel-cancer-screening.html. (268)

Cancer Council Australia. Cervical cancer vaccine. 2011. Viewed 25 February 2013, www.hpvvaccine.org.au/the-hpv-vaccine/vaccine-background.aspx. (403)

Cancer Council Australia. Position statement: prostate cancer screening. 2010. Viewed 25 February 2013,

www.cancer.org.au/content/pdf/CancerControlPolicy/PositionStatements/PS-Prostate_Cancer_Screening_Joint_key_messages_%20published_May2010.pdf. (248)

Cancer Council Australia. Report on Partner reference group consultation, national cervical screening program renewal. 2012. Viewed 21 February 2013, www.cancerscreening.gov.au/internet/screening/publishing.nsf/Content/cv-consult-rpt-15-04-12. (404)

Carlsen B & Bringedal B. Attitudes to clinical guidelines--do GPs differ from other medical doctors? BMJ Qual Saf 2011;20(2):158-62. (319)

Carlsen B & Norheim OF. "What lies beneath it all?"—an interview study of GPs' attitudes to the use of guidelines. BMC Health Serv Res 2008;8:218. (318)

Carlsen B, Glenton C, Pope C. Thou shalt versus thou shalt not: a meta-synthesis of GPs' attitudes to clinical practice guidelines. Br J Gen Pract 2007;57(545):971-8. (153)

Catchlove BR. The why and the wherefore. Med J Aust 2001;175(2):68-70. (388)

Cathebras PJ, Robbins JM, Kirmayer LJ, Hayton BC. Fatigue in primary care: prevalence, psychiatric comorbidity, illness behavior, and outcome. J Gen Intern Med 1992;7(3):276-86. (272)

Chadban SJ, Briganti EM, Kerr PG, Dunstan DW, Welborn TA, Zimmet PZ et al. Prevalence of kidney damage in Australian adults: The AusDiab kidney study. J Am Soc Nephrol 2003;14(7 Suppl 2):S131-S138. (267)

Chapman S & Barratt A. Has PSA testing truly been a "public health disaster"? Med J Aust 2010;193(7):431. (255)

Chapman S, Barratt A, Stockler M. Let sleeping dogs lie? : what men should know before getting tested for prostate cancer. Sydney: Sydney University Press, 2010. (249)

Charles J, Britt H, Harrison C. General practice workforce and workload. In: Britt H & Miller GC (eds). General practice in Australia, health priorities and policies 1998-2008. General practice series no. 24. Cat. no. GEP 24. Canberra: Australian Institute of Health and Welfare, 2009;22-37. (136)

Classification Committee of the World Organization of Family Doctors (WICC). ICPC-2: International Classification of Primary Care. 2 ed. Oxford: Oxford University Press, 1998. (110)

Colagiuri S, Lee CM, Colagiuri R, Magliano D, Shaw JE, Zimmet PZ et al. The cost of overweight and obesity in Australia. Med J Aust 2010;192(5):260-4. (291)

Commonwealth Department of Health and Aged Care. Medicare Benefits Schedule book. Canberra: DHAC, 2000. (223)

Commonwealth of Australia. Budget measure on cholesterol-lowering drugs. Canberra: Commonwealth of Australia, 2001. Viewed 27 February 2013, http://health.gov.au/internet/main/publishing.nsf/Content/health-budget2001-fact-hfact7a.htm. (222) Cooper A, Nherera L, Calvert N, O'Flynn N, Turnbull N, Robson J et al. Clinical guidelines and evidence review for lipid modification: cardiovascular risk assessment and the primary and secondary prevention of cardiovascular disease. NICE clinical guideline 67. London: National Collaborating Centre for Primary Care and Royal College of General Practitioners [National Institute for Health and Clinical Excellence], 2008. (212)

Cornuz J, Guessous I, Favrat B. Fatigue: a practical approach to diagnosis in primary care. CMAJ 2006;174(6):765-7. (286)

Costello AJ & Murphy DG. Has PSA testing truly been a "public health disaster"? Med J Aust 2010;193(1):4-5. (253)

Coumou HC & Meijman FJ. How do primary care physicians seek answers to clinical questions? A literature review. J Med Libr Assoc 2006;94(1):55-60. (152)

Criqui MH & Golomb BA. Lipid lowering: what and when to monitor. Lancet 2008;372(9638):516-7. (228)

Cullen W, Kearney Y, Bury G. Prevalence of fatigue in general practice. Ir J Med Sci 2002;171(1):10-2. (270)

Cumming G & Finch S. Inference by eye: confidence intervals and how to read pictures of data. Am Psychol 2005;60(2):170-80. (131)

Danese MD, Powe NR, Sawin CT, Ladenson PW. Screening for mild thyroid failure at the periodic health examination: a decision and cost-effectiveness analysis. JAMA 1996;276(4):285-92. (264)

David A, Pelosi A, McDonald E, Stephens D, Ledger D, Rathbone R et al. Tired, weak, or in need of rest: fatigue among general practice attenders. BMJ 1990;301(6762):1199-202. (271)

Davis P, Gribben B, Lay-Yee R, Scott A. How much variation in clinical activity is there between general practitioners? A multi-level analysis of decision-making in primary care. Journal of Health Services Research and Policy 2002;7(4):202-8. (72)

Dawes M. Co-morbidity: we need a guideline for each patient not a guideline for each disease. Fam Pract 2010;27(1):1-2. (332)

Deeble J & Lewis-Hughes P. Directions for pathology. National Health Strategy Unit. National Health Strategy Background Papers 1991;6;1-88. (6)

DeKay ML & Asch DA. Is the defensive use of diagnostic tests good for patients, or bad? Med Decis Making 1998;18(1):19-28. (379)

Development and validation of an international appraisal instrument for assessing the quality of clinical practice guidelines: the AGREE project. Qual Saf Health Care 2003;12(1):18-23. (324)

Diabetes Australia & The Royal Australian College of General Practitioners. Diabetes management in general practice 2008/9. Diabetes Australia, 2008. (192)

Diabetes Australia Guideline Development Consortium. National Evidence Based Guidelines for the Management of Type 2 Diabetes Mellitus [updated 2005]. National Health and Medical Research Council (NHMRC), 2001. Viewed 27 February 2013, www.nhmrc.gov.au/publications/synopses/di7todi13syn.htm. (193) Dick ML & Sundin J. Psychological and psychiatric causes of fatigue. Assessment and management. Aust Fam Physician 2003;32(11):877-81. (279)

Dixon LR. The complete blood count: physiologic basis and clinical usage. J Perinat Neonatal Nurs 1997;11(3):1-18. (351)

Doll H, Shine B, Kay J, James T, Glasziou P. The rise of cholesterol testing: how much is unnecessary. Br J Gen Pract 2011;61(583):e81-e88. (367)

Dorland's illustrated medical dictionary. 32nd Edition. Philadelphia: Elsevier Saunders, 2012. (4)

Doust J. Diagnostic and management strategies for anaemia in adults. Aust Fam Physician 2003;32(11):889-94. (361)

Dowling PT, Alfonsi G, Brown MI, Culpepper L. An education program to reduce unnecessary laboratory tests by residents. Acad Med 1989;64(7):410-2. (350)

Driver B, Britt H, O'Toole B, Harris M, Bridges-Webb C, Neary S. How representative are patients in general practice morbidity surveys? Fam Pract 1991;8(3):261-8. (116)

Dubey V, Mathew R, Iglar K, Moineddin R, Glazier R. Improving preventive service delivery at adult complete health check-ups: the Preventive health Evidence-based Recommendation Form (PERFORM) cluster randomized controlled trial. BMC Fam Pract 2006;7:44. (343)

Dunstan D, Zimmet P, Welborn T, Sicree R, Armstrong T, Atkins R et al. Diabesity and associated disorders in Australia - 2000: The accelerating epidemic. Melbourne: International Diabetes Institute, 2001. (157)

Dunstan DW, Zimmet PZ, Welborn TA, De Court, Cameron AJ, Sicree RA et al. The rising prevalence of diabetes and impaired glucose tolerance: the Australian Diabetes, Obesity and Lifestyle Study. Diabetes Care 2002;25(5):829-34. (180)

Enno A, Foster M, Hines P, Ireland M, McGrath K, O'Connell D et al. Quality use of pathology by general practitioners. Final report by the Hunter Urban Division of General Practice and Hunter Area Pathology Service. Hunter Urban Division of General Practice, 2004. (37)

Epstein RM, Franks P, Shields CG, Meldrum SC, Miller KN, Campbell TL et al. Patient-centered communication and diagnostic testing. Ann Fam Med 2005;3(5):415-21. (380)

Farquhar CM, Kofa EW, Slutsky JR. Clinicians' attitudes to clinical practice guidelines: a systematic review. Med J Aust 2002;177(9):502-6. (320)

Ferrante di Ruffano L, Davenport C, Eisinga A, Hyde C, Deeks JJ. A capturerecapture analysis demonstrated that randomized controlled trials evaluating the impact of diagnostic tests on patient outcomes are rare. J Clin Epidemiol 2012;65(3):282-7. (62)

Ferrante di Ruffano L, Hyde CJ, McCaffery KJ, Bossuyt PM, Deeks JJ. Assessing the value of diagnostic tests: a framework for designing and evaluating trials. BMJ 2012;344:e686. (70)

Ferrier BM, Woodward CA, Cohen M, Goldsmith CH. Laboratory Tests: Which Physicians Order More?: How women's practices differ from men's. Can Fam Physician 1991;37:349-52. (49)

Fisher E. Australian general practice - a celebration. 2008. Viewed 27 February 2013, www.racgp.org.au/historyofthecollege. (233)

Fitzgerald PD. General practice corporatisation: the half-time score. Med J Aust 2002;177(2):90-2. (389)

Fitzgerald PD. The ethics of doctors and big business. Med J Aust 2001;175(2):73-5. (387)

Flynn K. Medical fraud and inappropriate practice in Medibank and Medicare, Australia 1975-1995. PhD thesis. University of Wollongong, 2004. (21)

Fortin M, Bravo G, Hudon C, Vanasse A, Lapointe L. Prevalence of multimorbidity among adults seen in family practice. Ann Fam Med 2005;3(3):223-8. (331)

Fowkes FG. Containing the use of diagnostic tests. Br Med J (Clin Res Ed) 1985;290(6467):488-90. (148)

Fryer AA & Hanna FW. Managing demand for pathology tests: financial imperative or duty of care? Ann Clin Biochem 2009;46(Pt 6):435-7. (29)

Fryer AA & Smellie WS. Managing demand for laboratory tests: a laboratory toolkit. J Clin Pathol 2013;66(1):62-72. (57)

Fung V, Schmittdiel JA, Fireman B, Meer A, Thomas S, Smider N et al. Meaningful variation in performance: a systematic literature review. Med Care 2010;48(2):140-8. (412)

Galuska DA, Will JC, Serdula MK, Ford ES. Are health care professionals advising obese patients to lose weight? JAMA 1999;282(16):1576-8. (305)

Genest J, Frohlich J, Fodor G, McPherson R. Recommendations for the management of dyslipidemia and the prevention of cardiovascular disease: 2003 update. Appendix to CMAJ 2003;168(9):921-4. CMAJ., 2003. Viewed 27 February 2013, www.cmaj.ca/cgi/content/full/169/9/921/DC1. (218)

Genest J, Frohlich J, Fodor G, McPherson R. Recommendations for the management of dyslipidemia and the prevention of cardiovascular disease: summary of the 2003 update. CMAJ 2003;169(9):921-4. (217)

Gialamas A, Beilby JJ, Pratt NL, Henning R, Marley JE, Roddick JF. Investigating tiredness in Australian general practice. Do pathology tests help in diagnosis? Aust Fam Physician 2003;32(8):663-6. (287)

Gialamas A, Beilby JJ, Pratt NL, Marley JE, Roddick JF, Henning R. A project to examine the utilisation of pathology tests in the investigation of tiredness in general practice. Quality Use of Pathology Program, Commonwealth Department of Health and Aged Care, 2002. (408)

Glasziou P, Aronson JK, Irwig L (eds). Evidence-based medical monitoring: from principles to practice. Oxford: Blackwell Publishing, 2008. (329)

Glasziou P, Irwig L, Mant D. Monitoring in chronic disease: a rational approach. BMJ 2005;330(7492):644-8. (328)

Glasziou P. How much monitoring? Br J Gen Pract 2007;57(538):350-1. (415)

Glasziou PP, Irwig L, Heritier S, Simes RJ, Tonkin A. Monitoring cholesterol levels: measurement error or true change? Ann Intern Med 2008;148(9):656-61. (226)

Godwin M, Delva D, Miller K, Molson J, Hobbs N, MacDonald S et al. Investigating fatigue of less than 6 months' duration. Guidelines for family physicians. Can Fam Physician 1999;45:373-9. (276)

Graham I, Atar D, Borch-Johnsen K, Boysen G, Burell G, Cifkova R et al. European guidelines on cardiovascular disease prevention in clinical practice: executive summary. Atherosclerosis 2007;194(1):1-45. (215)

Grenz K, Mortinsen R, Pine D, Solberg L, Wilkinson JM, Harvey L et al. Preventive services for adults. ICSI health care guideline. Institute for Clinical Systems Improvement, 2008. (238)

Greving JP, Visseren FL, de Wit GA, Algra A. Statin treatment for primary prevention of vascular disease: whom to treat? Cost-effectiveness analysis. BMJ 2011;342:d1672. (374)

Grimshaw JM & Russell IT. Effect of clinical guidelines on medical practice: a systematic review of rigorous evaluations. Lancet 1993;342(8883):1317-22. (314)

Grol R, Dalhuijsen J, Thomas S, Veld C, Rutten G, Mokkink H. Attributes of clinical guidelines that influence use of guidelines in general practice: observational study. BMJ 1998;317(7162):858-61. (315)

Grundy SM, Cleeman JI, Merz CN, Brewer HB, Jr., Clark LT, Hunninghake DB et al. Implications of recent clinical trials for the National Cholesterol Education Program Adult Treatment Panel III guidelines. Circulation 2004;110(2):227-39. (211)

Guh DP, Zhang W, Bansback N, Amarsi Z, Birmingham CL, Anis AH. The incidence of co-morbidities related to obesity and overweight: a systematic review and meta-analysis. BMC Public Health 2009;9:88. (292)

Hammett RJ & Harris RD. Halting the growth in diagnostic testing. Med J Aust 2002;177(3):124-5. (149)

Harrison C & Britt H. General practice - workforce gaps now and in 2020. Aust Fam Physician 2011;40(1-2):12-5. (369)

Harrison CM, Britt HC, Charles J. Sex of the GP - 20 years on. Med J Aust 2011;195(4):192-6. (382)

Hartley RM, Charlton JR, Harris CM, Jarman B. Influence of patient characteristics on test ordering in general practice. Br Med J (Clin Res Ed) 1984;289(6447):735-8. (73)

Hartley RM, Charlton JR, Harris CM, Jarman B. Patterns of physicians' use of medical resources in ambulatory settings. Am J Public Health 1987;77(5):565-7. (44)

Harvey K, Clark RB, Visser S. A mechanism for the development, implementation and evaluation of evidence-based, best practice clinical guidelines to facilitate quality use of pathology tests. Quality Use of Pathology Program, Commonwealth Department of Health and Ageing, 2003. (407)

Helfand M. Screening for subclinical thyroid dysfunction in nonpregnant adults: a summary of the evidence for the U.S. Preventive Services Task Force. Ann Intern Med 2004;140(2):128-41. (263)

Henderson J, Miller G, Britt H, Pan Y. Effect of computerisation on Australian general practice: does it improve the quality of care? Qual Prim Care 2010;18(1):33-47. (38)

Houben PH, van der Weijden T, Winkens B, Winkens RA, Grol RP. Pretest expectations strongly influence interpretation of abnormal laboratory results and further management. BMC Fam Pract 2010;11:13. (335)

Houben PH, van der Weijden T, Winkens RA, Grol RP. Cascade effects of laboratory testing are found to be rare in low disease probability situations: prospective cohort study. J Clin Epidemiol 2010;63(4):452-8. (336)

Houben PH, Winkens RA, van der Weijden T, Vossen RC, Naus AJ, Grol RP. Reasons for ordering laboratory tests and relationship with frequency of abnormal results. Scand J Prim Health Care 2010;28(1):18-23. (338)

Hudon E, Beaulieu MD, Roberge D. Integration of the recommendations of the Canadian Task Force on Preventive Health Care: obstacles perceived by a group of family physicians. Fam Pract 2004;21(1):11-7. (342)

International Diabetes Federation Clinical Guidelines Task Force. Global guideline for Type 2 Diabetes. Brussels: International Diabetes Federation, 2005. (200)

Irwig L, Bossuyt P, Glasziou P, Gatsonis C, Lijmer J. Designing studies to ensure that estimates of test accuracy are transferable. BMJ 2002;324(7338):669-71. (55)

Irwig L, Glasziou P, Wilson A, Macaskill P. Estimating an individual's true cholesterol level and response to intervention. JAMA 1991;266(12):1678-85. (356)

Jackevicius CA, Mamdani M, Tu JV. Adherence with statin therapy in elderly patients with and without acute coronary syndromes. JAMA 2002;288(4):462-7. (370)

Jellinger PS, Dickey RA, Ganda OP, Mehta AE, Nguyen TT, Rodbard HW et al. AACE medical guidelines for clinical practice for the diagnosis and treatment of dyslipidemia and prevention of atherogenesis. Endocr Pract 2000;6(2):162-213. (214)

Johnson RK & Mortimer AJ. Routine pre-operative blood testing: is it necessary? Anaesthesia 2002;57(9):914-7. (67)

Jorgensen LG, Brandslund I, Hyltoft PP. Should we maintain the 95 percent reference intervals in the era of wellness testing? A concept paper. Clin Chem Lab Med 2004;42(7):747-51. (142)

Kannel WB, Castelli WP, Gordon T, McNamara PM. Serum cholesterol, lipoproteins, and the risk of coronary heart disease. The Framingham study. Ann Intern Med 1971;74(1):1-12. (205)

Kannel WB, D'Agostino RB, Sullivan L, Wilson PW. Concept and usefulness of cardiovascular risk profiles. Am Heart J 2004;148(1):16-26. (207)

Kannel WB. Blood pressure as a cardiovascular risk factor: prevention and treatment. JAMA 1996;275(20):1571-6. (162)

Keitz SA, Stechuchak KM, Grambow SC, Koropchak CM, Tulsky JA. Behind closed doors: management of patient expectations in primary care practices. Arch Intern Med 2007;167(5):445-52. (347)

Kennedy PJ, Leathley CM, Hughes CF. Clinical practice variation. Med J Aust 2010;193(8 Suppl):S97-S99. (411)

Kerr D, Malcolm L, Schousboe J, Pimm F. Successful implementation of laboratory budget holding by Pegasus Medical Group. N Z Med J 1996;109(1029):334-7. (99)

Khan NA, Hemmelgarn B, Padwal R, Larochelle P, Mahon JL, Lewanczuk RZ et al. The 2007 Canadian Hypertension Education Program recommendations for the management of hypertension: part 2 - therapy. Can J Cardiol 2007;23(7):539-50. (171)

Knox SA, Harrison CM, Britt HC, Henderson JV. Estimating prevalence of common chronic morbidities in Australia. Med J Aust 2008;189(2):66-70. (158)

Koch H, van Bokhoven MA, ter Riet G, Hessels KM, van der Weijden T, Dinant GJ et al. What makes general practitioners order blood tests for patients with unexplained complaints? A cross-sectional study. Eur J Gen Pract 2009;15(1):22-8. (288)

Koch H, van Bokhoven MA, ter Riet G, van Alphen-Jager JT, van der Weijden T, Dinant GJ et al. Ordering blood tests for patients with unexplained fatigue in general practice: what does it yield? Results of the VAMPIRE trial. Br J Gen Pract 2009;59(561):e93-100. (281)

Koch H, van Bokhoven MA, ter Riet G, van der Weijden T, Dinant GJ, Bindels PJ. Demographic characteristics and quality of life of patients with unexplained complaints: a descriptive study in general practice. Qual Life Res 2007;16(9):1483-9. (274)

KPMG. Review of the Intergenerational Report and Pathology Funding Arrangements in Australia. 2011. Viewed 21 February 2013, www.aapp.asn.au/images/document/KPMG%20Econotech%20Pathology%20Fundin g%20Study%20-%20Jan%202011.pdf. (27)

Kravitz RL & Callahan EJ. Patients' perceptions of omitted examinations and tests: A qualitative analysis. J Gen Intern Med 2000;15(1):38-45. (339)

Kravitz RL, Bell RA, Azari R, Kelly-Reif S, Krupat E, Thom DH. Direct observation of requests for clinical services in office practice: what do patients want and do they get it? Arch Intern Med 2003;163(14):1673-81. (365)

Kravitz RL, Bell RA, Franz CE. A taxonomy of requests by patients (TORP): a new system for understanding clinical negotiation in office practice. J Fam Pract 1999;48(11):872-8. (364)

Kristiansen IS & Hjortdahl P. The general practitioner and laboratory utilization: why does it vary? Fam Pract 1992;9(1):22-7. (46)

Lab Tests Online AU partners. C-reactive protein. 2013. Viewed 27 February 2013, www.labtestsonline.org.au/understanding/analytes/crp/glance.html. (175)

Lab Tests Online AU partners. Full blood count. 2009. Viewed 27 February 2013, http://labtestsonline.org.au/understanding/analytes/fbc/tab/test. (221)

Ladenson PW, Singer PA, Ain KB, Bagchi N, Bigos ST, Levy EG et al. American Thyroid Association guidelines for detection of thyroid dysfunction. Arch Intern Med 2000;160(11):1573-5. (265)

Langley C, Faulkner A, Watkins C, Gray S, Harvey I. Use of guidelines in primary care—practitioners' perspectives. Fam Pract 1998;15(2):105-11. (322)

Lau DC, Douketis JD, Morrison KM, Hramiak IM, Sharma AM, Ur E. 2006 Canadian clinical practice guidelines on the management and prevention of obesity in adults and children [summary]. CMAJ 2007;176(8):S1-13. (299)

Laurence C, Gialamas A, Yelland L, Bubner T, Glastonbury B, Beilby J, Shephard M, Gill J on behalf of the Trial Management and Evaluation Group, PoCT Device Group, and RCPA Quality Assurance Programs Pty Ltd. Point of care testing in general practice trial. 2009. Viewed 14 February 2013, www.appn.net.au/Data/Sites/1/SharedFiles/Publications/200901-poctfinalreport27jan09amended5feb09.pdf. (22)

Lee CM, Colagiuri R, Magliano DJ, Cameron AJ, Shaw J, Zimmet P et al. The cost of diabetes in adults in Australia. Diabetes Res Clin Pract 2013. (405)

Leese B. Is there too much laboratory testing? CHE discussion paper 79. York, UK: University of York, 1991. (90)

Leurquin P, van Casteren V, De Maeseneer J. Use of blood tests in general practice: a collaborative study in eight European countries. Eurosentinel Study Group. Br J Gen Pract 1995;45(390):21-5. (51)

Leverence RR, Williams RL, Sussman A, Crabtree BF. Obesity counseling and guidelines in primary care: a qualitative study. Am J Prev Med 2007;32(4):334-9. (310)

Levy D, Larson MG, Vasan RS, Kannel WB, Ho KK. The progression from hypertension to congestive heart failure. JAMA 1996;275(20):1557-62. (160)

Lewington S, Clarke R, Qizilbash N, Peto R, Collins R. Age-specific relevance of usual blood pressure to vascular mortality: a meta-analysis of individual data for one million adults in 61 prospective studies. Lancet 2002;360(9349):1903-13. (159)

Lucan SC. Coronary artery disease prevention: as easy as 1, 2, 3? Am Fam Physician 2010;82(10):1167. (376)

Lugtenberg M, Burgers JS, Besters CF, Han D, Westert GP. Perceived barriers to guideline adherence: a survey among general practitioners. BMC Fam Pract 2011;12:98. (316)

Lugtenberg M, Burgers JS, Clancy C, Westert GP, Schneider EC. Current guidelines have limited applicability to patients with comorbid conditions: a systematic analysis of evidence-based guidelines. PLoS One 2011;6(10):e25987. (333)

Lugtenberg M, Zegers-van Schaick JM, Westert GP, Burgers JS. Why don't physicians adhere to guideline recommendations in practice? An analysis of barriers among Dutch general practitioners. Implement Sci 2009;4:54. (317)

Lundberg GD. The modern clinical laboratory; Justification, scope, and directions. JAMA 1975;232(5):528-9. (150)

Lundberg GD. The need for an outcomes research agenda for clinical laboratory testing. JAMA 1998;280(6):565-6. (42)

MacKenzie R, Chapman S, Barratt A, Holding S. "The news is [not] all good": misrepresentations and inaccuracies in Australian news media reports on prostate cancer screening. Med J Aust 2007;187(9):507-10. (250)

MacKenzie R, Chapman S, Holding S, McGeechan K. 'A matter of faith, not science': analysis of media coverage of prostate cancer screening in Australian news media 2003-2006. J R Soc Med 2007;100(11):513-21. (251)

Mancia G, De Backer G, Dominiczak A, Cifkova R, Fagard R, Germano G et al. 2007 Guidelines for the Management of Arterial Hypertension: The Task Force for the Management of Arterial Hypertension of the European Society of Hypertension (ESH) and of the European Society of Cardiology (ESC). J Hypertens 2007;25(6):1105-87. (168)

Mann DM, Woodward M, Muntner P, Falzon L, Kronish I. Predictors of nonadherence to statins: a systematic review and meta-analysis. Ann Pharmacother 2010;44(9):1410-21. (375)

Mates M, Heyd J, Souroujon M, Ben SA, Manny N, Hershko C. The haematologist as watchdog of community health by full blood count. QJM 1995;88(5):333-9. (354)

McKenney JM, Davidson MH, Jacobson TA, Guyton JR. Final conclusions and recommendations of the National Lipid Association Statin Safety Assessment Task Force. Am J Cardiol 2006;97(8A):89C-94C. (230)

Medical Training Review Panel. Medical Training Review Panel fifteenth report. Canberra: Department of Health and Ageing, 2012. Viewed 25 February 2013, www.health.gov.au/internet/main/publishing.nsf/Content/work-pubs-mtrp-15. (383)

Medicare Australia. MBS Group Statistics Reports. Report on Category 6 - Pathology Services, 2008-09 financial year. 2012. Viewed 25 February 2013, https://www.medicareaustralia.gov.au/statistics/mbs_group.shtml. (345)

MedicineNet. MedTerms Medical Dictionary: Differential diagnosis definition. 2008. Viewed 25 February 2013, www.medterms.com. (362)

Metz A & Hebbard G. Nausea and vomiting in adults--a diagnostic approach. Aust Fam Physician 2007;36(9):688-92. (360)

Meza RA, Angelis M, Britt H, Miles DA, Seneta E, Bridges-Webb C. Development of sample size models for national general practice surveys. Aust J Public Health 1995;19(1):34-40. (106)

Michael Legg & Associates, IRIS Research, University of Wollongong, Cheong I. A study of the impact of the use of general practice computer systems on the ordering of pathology. Quality Use of Pathology Program, Department of Health and Ageing, 2004. (30)

Miller AB. New data on prostate-cancer mortality after PSA screening. N Engl J Med 2012;366(11):1047-8. (246)

Mindemark M & Larsson A. Long-term effects of an education programme on the optimal use of clinical chemistry testing in primary health care. Scand J Clin Lab Invest 2009;69(4):481-6. (84)

Moynihan R, Doust J, Henry D. Preventing overdiagnosis: how to stop harming the healthy. BMJ 2012;344:e3502. (69)

Murtagh J. Fatigue--a general diagnostic approach. Aust Fam Physician 2003;32(11):873-6. (282)

Murtagh J. Fits, faints and funny turns. A general diagnostic approach. Aust Fam Physician 2003;32(4):203-6. (359)

Murtagh J. Murtagh's general practice. Sydney: McGraw-Hill Australia Pty Ltd, 2007. (173)

Nash LM, Walton MM, Daly MG, Kelly PJ, Walter G, van Ekert EH et al. Perceived practice change in Australian doctors as a result of medicolegal concerns. Med J Aust 2010;193(10):579-83. (147)

National Cholesterol Education Program Expert Panel. The Third Report of the Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). National Heart, Lung, and Blood Institute, National Institutes of Health, 2002. (210)

National Collaborating Centre for Chronic Conditions & British Hypertension Society. Hypertension: management of hypertension in adults in primary care (partial update of NICE clinical guideline 18). NICE clinical guideline 34. London: National Institute for Health and Clinical Excellence, 2006. (164)

National Collaborating Centre for Chronic Conditions. Type 2 diabetes: national clinical guideline for management in primary and secondary care (update). NICE clinical guideline 66. London: National Institute for Health and Clinical Excellence, 2008. (195)

National Collaborating Centre for Primary Care & Centre for Public Health Excellence at NICE. Obesity: guidance on the prevention, identification, assessment and management of overweight and obesity in adults and children. NICE clinical guideline 43. London: National Institute for Health and Clinical Excellence, 2006. (302)

National Collaborating Centre for Primary Care & Centre for Public Health Excellence at NICE. Obesity: guidance on the prevention, identification, assessment and management of overweight and obesity in adults and children: Quick reference guide 2 for the NHS. NICE clinical guideline 43. London: National Institute for Health and Clinical Excellence, 2006. (303)

National Health and Medical Research Council (NHMRC). Clinical practice guidelines for the management of overweight and obesity in adults. Canberra: NHMRC, 2004. Viewed 27 February 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/obesityguidelines-guidelines-adults.htm. (297)

National Health and Medical Research Council (NHMRC). Clinical practice guidelines portal. 2012. Viewed 27 February 2013, www.clinicalguidelines.gov.au/. (326)

National Health and Medical Research Council (NHMRC). Diabetes. 2008. Viewed 27 February 2013, www.nhmrc.gov.au/your_health/facts/diabetes.htm. (185)

National Health and Medical Research Council (NHMRC). Overweight and obesity in adults: a guide for general practitioners. Canberra: NHMRC, 2004. Viewed 27 February 2013,

www.health.gov.au/internet/main/publishing.nsf/Content/obesityguidelines-guidelines-gp_guide.htm. (298)

National Heart Foundation of Australia (National blood pressure and vascular disease advisory committee). Guide to management of hypertension 2008: assessing and managing raised blood pressure in adults. National Heart Foundation of Australia, 2008. (163)

National Heart, Lung and Blood Institute. Seventh report of the Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood pressure (JNC 7). National Heart, Lung, and Blood Institute. US Department of Health and Human Services, 2004. (165)

National Institutes of Health, National Heart, Lung and Blood Institute, North American Association for the Study of Obesity. Practical guide to the identification, evaluation, and treatment of overweight and obesity in adults. 2000. Viewed 13 February 2013, www.nhlbi.nih.gov/guidelines/obesity/practgde.htm. (304)

National Obesity Forum. Guidelines on management of adult obesity and overweight in primary care. United Kingdom: National Obesity Forum, 2006. (300)

National Obesity Forum. Training resource for health care professionals: In depth assessment: obesity and overweight. United Kingdom: National Obesity Forum, 2004. (301)

National Prescribing Service Ltd. Are you as healthy as your car. Medicines talk. No. 35. 2010. Viewed 25 February 2013,

www.nps.org.au/__data/assets/pdf_file/0018/116514/MT_Spring_10_press_241210.p df. (269)

National Preventative Health Taskforce. National Preventative Health Strategy. 2009. Viewed 27 February 2013,

www.preventativehealth.org.au/internet/preventativehealth/publishing.nsf/Content/na tional-preventative-health-strategy-1lp. (235)

National Vascular Disease Prevention Alliance. Guidelines for the assessment of absolute cardiovascular disease risk. National Vascular Disease Prevention Alliance, 2009. (176)

National Vascular Disease Prevention Alliance. New guidelines and treatment procedures for heart attack and stroke [Media release]. 2012. Viewed 16 February 2013,

www.diabetesaustralia.com.au/Documents/DA/Media%20Releases/NVDPA%20guid elines-May2012-General%20Release%20FINAL.pdf. (406)

Nijrolder I, van der Windt D, deVries H, van der Horst H. Diagnoses during followup of patients presenting with fatigue in primary care. CMAJ 2009;181(10):683-7. (278)

Ogden J & Flanagan Z. Beliefs about the causes and solutions to obesity: a comparison of GPs and lay people. Patient Educ Couns 2008;71(1):72-8. (311)

Ontario Association of Medical Laboratories. Guideline for the Use of Laboratory Tests to Detect Thyroid Dysfunction. 2007. Viewed 25 February 2013, www.oaml.com/PDF/FINALTSH% 20Guideline% 20July% 2018,% 2007.pdf. (261)

O'Reilly DS. Thyroid function tests-time for a reassessment. BMJ 2000;320(7245):1332-4. (259)

Organization for Economic Co-operation and Development (OECD). Live Longer, Work Longer: A synthesis report. Ageing and Employment Policies Project. Paris: OECD, 2006. (398)

Organization for Economic Co-operation and Development (OECD). Value for money in health spending. Paris. OECD, 2010. (397)

Padwal RS, Hemmelgarn BR, McAlister FA, McKay DW, Grover S, Wilson T et al. The 2007 Canadian Hypertension Education Program recommendations for the management of hypertension: part 1- blood pressure measurement, diagnosis and assessment of risk. Can J Cardiol 2007;23(7):529-38. (170)

Pathology Review Advisory Group. Report of the review of NHS pathology services in England (The Carter report). 2005. (8)

Pathology Review Advisory Group. Report of the second phase of the review of NHS pathology services in England (The Carter report - 2nd phase). 2008. (410)

Peck BM, Ubel PA, Roter DL, Goold SD, Asch DA, Jeffreys AS et al. Do unmet expectations for specific tests, referrals, and new medications reduce patients' satisfaction? J Gen Intern Med 2004;19(11):1080-7. (35)

Phillips P. The trouble with testing: screening. Aust Fam Physician 2003;32(11):925-6. (143)

Phillips PJ & Phillipov G. Cholesterol - frequently asked questions. Aust Fam Physician 2006;35(8):595-6. (229)

Pogson J. PSA study results positive but the debate rolls on. Australian Doctor 2012 Mar 23:2. (247)

Poley MJ, Edelenbos KI, Mosseveld M, van Wijk MA, de Bakker DH, van der Lei J et al. Cost consequences of implementing an electronic decision support system for ordering laboratory tests in primary care: evidence from a controlled prospective study in the Netherlands. Clin Chem 2007;53(2):213-9. (76)

Potter MB, Vu JD, Croughan-Minihane M. Weight management: what patients want from their primary care physicians. J Fam Pract 2001;50(6):513-8. (307)

Price CP. Evidence-based laboratory medicine: linking tests to outcomes. Ned Tijdschr Kin Chem 2001;26:236-42. (61)

Price CP. Evidence-based laboratory medicine: supporting decision-making. Clin Chem 2000;46(8 Pt 1):1041-50. (60)

PriceWaterhouseCoopers. Capital expenditure in the pathology sector. 2010. Viewed 27 February 2013,

 $www.health.gov.au/internet/main/publishing.nsf/Content/20CBB30C457BEB1DCA2\\576D300822677/\$File/PWC\%20Final\%20Report\%2021\%20September\%202010.pdf . (134)$

PriceWaterhouseCoopers. International review of pathology funding arrangements. 2010. Viewed 27 February 2013,

www.hirc.health.gov.au/internet/main/publishing.nsf/Content/20CBB30C457BEB1D CA2576D300822677/\$File/PwC_DoHAPathologyReview_180310_FINAL.pdf. (7)

Professional Services Review. Professional Services Review: annual report 1999-2000. 2000. Viewed 25 February 2013, www.psr.gov.au/publications-and-resources/annual-reports. (391)

Prostate Cancer Foundation of Australia. PCFA Policy Statement: PSA/DRE testing for early detection of Prostate Cancer. 2006. Viewed 2 March 2013, www.prostate.org.au/articleLive/attachments/1/PCFA%20PSA%20%20DRE%20Poli cy%20Statement%20270706.pdf . (252)

Rao GG, Crook M, Tillyer ML. Pathology tests: is the time for demand management ripe at last? J Clin Pathol 2003;56(4):243-8. (28)

Ridsdale L, Evans A, Jerrett W, Mandalia S, Osler K, Vora H. Patients with fatigue in general practice: a prospective study. BMJ 1993;307(6896):103-6. (275)

Ridsdale L, Mandalia S, Evans A, Jerrett W, Osler K. Tiredness as a ticket of entry. The role of patients' beliefs and psychological symptoms in explaining frequent attendance. General Practice Fatigue Group. Scand J Prim Health Care 1999;17(2):72-4. (277)

Rodbard HW, Blonde L, Braithwaite SS, Brett EM, Cobin RH, Handelsman Y et al. American Association of Clinical Endocrinologists medical guidelines for clinical practice for the management of diabetes mellitus. Endocr Pract 2007;13 Suppl 1:1-68. (199)

Rossouw JE, Anderson GL, Prentice RL, LaCroix AZ, Kooperberg C, Stefanick ML et al. Risks and benefits of estrogen plus progestin in healthy postmenopausal women: principal results From the Women's Health Initiative randomized controlled trial. JAMA 2002;288(3):321-33. (135)

Roxon N, Minister for Health and Ageing. Medicare red tape slashed [Media release]. Canberra: Department of Health and Ageing, 2009. Viewed 27 February 2013, www.health.gov.au/internet/ministers/publishing.nsf/Content/mr-yr09-nr-nr234.htm?OpenDocument&yr=2009&mth=12. (344)

Ruttimann S, Clemencon D, Dubach UC. Usefulness of complete blood counts as a case-finding tool in medical outpatients. Ann Intern Med 1992;116(1):44-50. (353)

SAS proprietary software release 9.1. Cary: SAS Institute Inc, 2003. (129)

Schattner A. The unbearable lightness of diagnostic testing: time to contain inappropriate test ordering. Postgrad Med J 2008;84(998):618-21. (53)

Schein OD, Katz J, Bass EB, Tielsch JM, Lubomski LH, Feldman MA et al. The value of routine preoperative medical testing before cataract surgery. Study of Medical Testing for Cataract Surgery. N Engl J Med 2000;342(3):168-75. (68)

Schroder FH, Hugosson J, Roobol MJ, Tammela TL, Ciatto S, Nelen V et al. Prostate-cancer mortality at 11 years of follow-up. N Engl J Med 2012;366(11):981-90. (244) Schroder FH, Hugosson J, Roobol MJ, Tammela TL, Ciatto S, Nelen V et al. Screening and prostate-cancer mortality in a randomized European study. N Engl J Med 2009;360(13):1320-8. (241)

Schwartz G, Canzanello V, Woolley A, Miller T, O'Connor P, Klein D et al. Hypertension diagnosis and treatment. ICSI health care guideline. Institute for Clinical Systems Improvement, 2006. (169)

Scottish Intercollegiate Guidelines Network (SIGN). Hypertension in older people: a national clinical guideline. SIGN publication no. 49. Edinburgh: SIGN, 2001. (166)

Scottish Intercollegiate Guidelines Network (SIGN). Management of diabetes: a national clinical guideline. SIGN publication no. 55. Edinburgh: SIGN, 2001. (196)

Scottish Intercollegiate Guidelines Network (SIGN). Risk estimation and the prevention of cardiovascular disease: a national clinical guideline. SIGN publication no. 97. Edinburgh: SIGN, 2007. (213)

Selby JV, Schmittdiel JA, Lee J, Fung V, Thomas S, Smider N et al. Meaningful variation in performance: what does variation in quality tell us about improving quality? Med Care 2010;48(2):133-9. (413)

Shalev V, Chodick G, Heymann AD. Format change of a laboratory test order form affects physician behavior. Int J Med Inform 2009;78(10):639-44. (81)

Sharpe M & Wilks D. Fatigue. BMJ 2002;325(7362):480-3. (284)

Shaw JE, Sicree RA, Zimmet PZ. Global estimates of the prevalence of diabetes for 2010 and 2030. Diabetes Res Clin Pract 2010;87(1):4-14. (182)

Shroufi A & Powles JW. Adherence and chemoprevention in major cardiovascular disease: a simulation study of the benefits of additional use of statins. J Epidemiol Community Health 2010;64(2):109-13. (373)

Simons LA, Levis G, Simons J. Apparent discontinuation rates in patients prescribed lipid-lowering drugs. Med J Aust 1996;164(4):208-11. (372)

Singapore Health Promotion Board. Health screening. Singapore: Ministry of Health, 2004. (240)

Singapore Ministry of Health. Clinical practice guidelines: diabetes mellitus. Singapore: Ministry of Health, 2006. (198)

Singapore Ministry of Health. Clinical practice guidelines: lipids. Singapore: Ministry of Health, 2006. (219)

Skeie S, Perich C, Ricos C, Araczki A, Horvath AR, Oosterhuis WP et al. Postanalytical external quality assessment of blood glucose and hemoglobin A1c: an international survey. Clin Chem 2005;51(7):1145-53. (201)

Smellie WS, Finnigan DI, Wilson D, Freedman D, McNulty CA, Clark G. Methodology for constructing guidance. J Clin Pathol 2005;58(3):249-53. (56)

Smellie WS, Forth J, Bareford D, Twomey P, Galloway MJ, Logan EC et al. Best practice in primary care pathology: review 3. J Clin Pathol 2006;59(8):781-9. (231)

Smellie WS, Galloway MJ, Chinn D, Gedling P. Is clinical practice variability the major reason for differences in pathology requesting patterns in general practice? J Clin Pathol 2002;55(4):312-4. (337)
Smellie WS, Galloway MJ, Chinn D. Benchmarking general practice use of pathology services: a model for monitoring change. J Clin Pathol 2000;53(6):476-80. (82)

Smellie WS, Lowrie R, Wilkinson E. A laboratory based intervention to improve appropriateness of lipid tests and audit cholesterol lowering in primary care. BMJ 2001;323(7323):1224-7. (83)

Smellie WS, Vanderpump MP, Fraser WD, Bowley R, Shaw N. Best practice in primary care pathology: review 11. J Clin Pathol 2008;61(4):410-8. (258)

Smellie WS, Wilson D, McNulty CA, Galloway MJ, Spickett GA, Finnigan DI et al. Best practice in primary care pathology: review 1. J Clin Pathol 2005;58(10):1016-24. (224)

Smellie WS. Appropriateness of test use in pathology: a new era or reinventing the wheel? Ann Clin Biochem 2003;40(Pt 6):585-92. (54)

Smellie WS. Demand management and test request rationalization. Ann Clin Biochem 2012;49(Pt 4):323-36. (39)

Smellie WS. Testing pitfalls and summary of guidance in lipid management. BMJ 2006;333(7558):83-6. (227)

Smellie WS. What is a significant difference between sequential laboratory results? J Clin Pathol 2008;61(4):419-25. (203)

Smellie WS. When is "abnormal" abnormal? Dealing with the slightly out of range laboratory result. J Clin Pathol 2006;59(10):1005-7. (144)

Smith BJ & McNeely MD. The influence of an expert system for test ordering and interpretation on laboratory investigations. Clin Chem 1999;45(8 Pt 1):1168-75. (87)

Snow V, Aronson MD, Hornbake ER, Mottur-Pilson C, Weiss KB. Lipid control in the management of type 2 diabetes mellitus: a clinical practice guideline from the American College of Physicians. Ann Intern Med 2004;140(8):644-9. (204)

Solomon DH, Hashimoto H, Daltroy L, Liang MH. Techniques to improve physicians' use of diagnostic tests: a new conceptual framework. JAMA 1998;280(23):2020-7. (91)

Sood R, Sood A, Ghosh AK. Non-evidence-based variables affecting physicians' testordering tendencies: a systematic review. Neth J Med 2007;65(5):167-77. (41)

Stamler J, Stamler R, Neaton JD. Blood pressure, systolic and diastolic, and cardiovascular risks. US population data. Arch Intern Med 1993;153(5):598-615. (161)

Stamler J, Wentworth D, Neaton JD. Is relationship between serum cholesterol and risk of premature death from coronary heart disease continuous and graded? Findings in 356,222 primary screenees of the Multiple Risk Factor Intervention Trial (MRFIT). JAMA 1986;256(20):2823-8. (206)

Steven I. Patient presentations in general practice. Sydney: McGraw-Hill Book Company Australia Pty Ltd, 1999. (174)

Stuart PJ, Crooks S, Porton M. An interventional program for diagnostic testing in the emergency department. Med J Aust 2002;177(3):131-4. (66)

Studdert DM, Britt HC, Pan Y, Fahridin S, Bayram CF, Gurrin LC. Are rates of pathology test ordering higher in general practices co-located with pathology collection centres? Med J Aust 2010;193(2):114-9. (392)

Sucov A, Bazarian JJ, deLahunta EA, Spillane L. Test ordering guidelines can alter ordering patterns in an academic emergency department. J Emerg Med 1999;17(3):391-7. (358)

Takahashi O, Glasziou PP, Perera R, Shimbo T, Suwa J, Hiramatsu S et al. Lipid rescreening: what is the best measure and interval? Heart 2010;96(6):448-52. (349)

Taylor PT. Some fundamental influences on the pathology requesting behaviour of Victorian general practitioners--preliminary study results. Pathology 1994;26(1):33-5. (48)

Tham M & Young D. The role of the General Practitioner in weight management in primary care--a cross sectional study in General Practice. BMC Fam Pract 2008;9:66. (293)

The Lewin Group. Laboratory medicine: a national status report. Centers for Disease Control and Prevention's Laboratory Science, Policy and Practice Program Office, 2008. (10)

The Royal Australian College of General Practitioners. Guidelines for preventive activities in general practice. 7th ed. South Melbourne: The Royal Australian College of General Practitioners, 2009. (177)

The Royal College of Pathologists of Australasia & University of Sydney Department of Medical Education. RCPA case scenario - 'There must be something wrong'. 2003. Viewed 25 February 2013,

www.rcpamanual.edu.au/index.php?option=com_content&view=category&layout=bl og&id=35&Itemid=60. (285)

The Royal College of Pathologists of Australasia. Common Sense Pathology. 2012. Viewed 27 February 2013,

www.rcpa.edu.au/Publications/CommonSensePathology.htm. (102)

The Royal College of Pathologists of Australasia. Organised PSA testing could protect Australian men. 2011. Viewed 27 February 2013, www.rcpa.edu.au//static/File/Asset%20library/public%20documents/Media%20Rele ases/2011/Organised_PSA_testing_could_protect_Australian_men.pdf. (256)

The Royal College of Pathologists of Australasia. RCPA announces position on PSA testing for prostate cancer [Media release]. 2011. Viewed 27 February 2013, www.rcpa.edu.au//static/File/Asset% 20library/public% 20documents/Media% 20Rele ases/2011/RCPA_announces_position_on_PSA_testing_for_prostate_cancer.pdf. (254)

The Royal College of Pathologists of Australasia. RCPA Manual. RCPA, 2004. (172)

Therapeutic Guidelines Ltd. Therapeutic Gudelines. eTG complete [CD-ROM]. Melbourne: Therapeutic Guidelines Limited, 2008. (179)

Thomas RE, Croal BL, Ramsay C, Eccles M, Grimshaw J. Effect of enhanced feedback and brief educational reminder messages on laboratory test requesting in primary care: a cluster randomised trial. Lancet 2006;367(9527):1990-6. (58)

Tomlin A, Dovey S, Gauld R, Tilyard M. Better use of primary care laboratory services following interventions to 'market' clinical guidelines in New Zealand: a controlled before-and-after study. BMJ Qual Saf 2011;20(3):282-90. (85)

Tonkin A, Barter P, Best J, Boyden A, Furler J, Hossack K et al. National Heart Foundation of Australia and the Cardiac Society of Australia and New Zealand: position statement on lipid management - 2005. Heart Lung Circ 2005;14(4):275-91. (178)

Treasury. Australia's Demographic Challenges. Canberra: Commonwealth of Australia, 2004. Viewed 25 February 2013,

http://demographics.treasury.gov.au/content/_download/australias_demographic_chal lenges/australias_demographic_challenges.pdf. (395)

Treasury. The Intergenerational report. Australia to 2050: future challenges. Canberra: Commonwealth of Australia, 2010. Viewed 25 February 2013, http://archive.treasury.gov.au/igr/igr2010/report/pdf/IGR_2010.pdf. (234)

Trinick TR, Archbold GPR, Crooks GM, Damani NN, Hughes DFC, Jones FGC et al. Appropriateness of laboratory investigations: report of a CREST working group. Belfast: Clinical Resource Efficiency Support Team (CREST), 2000. (12)

Troughton RW, Frampton CM, Yandle TG, Espiner EA, Nicholls MG, Richards AM. Treatment of heart failure guided by plasma aminoterminal brain natriuretic peptide (N-BNP) concentrations. Lancet 2000;355(9210):1126-30. (65)

U.S. Preventive Services Task Force (USPSTF). Preventive Services. 2008. Viewed 19 May 2009, www.ahrq.gov/clinic/prevenix.htm. (237)

United States National Library of Medicine. Secondary prevention - Medical Subject Headings (MeSH) Descriptor Data. 2011. Viewed 1 October 2013, www.nlm.nih.gov/cgi/mesh/2011/MB_cgi?mode=&term=SECONDARY+PREVEN TION. (377)

Vafiadis P. The dilemma of the unexpected result. Aust Fam Physician 1996;25(6):971-3. (334)

van Bokhoven MA, Koch H, Dinant GJ, Bindels PJ, Grol RP, van der Weijden T. Exploring the black box of change in improving test-ordering routines. Fam Pract 2008;25(3):139-45. (409)

van Bokhoven MA, Koch H, van der Weijden T, Grol RP, Bindels PJ, Dinant GJ. Blood test ordering for unexplained complaints in general practice: the VAMPIRE randomised clinical trial protocol. [ISRCTN55755886]. BMC Fam Pract 2006;7:20. (280)

van Bokhoven MA, Koch H, van der Weijden T, Grol RP, Kester AD, Rinkens PE et al. Influence of watchful waiting on satisfaction and anxiety among patients seeking care for unexplained complaints. Ann Fam Med 2009;7(2):112-20. (289)

van Bokhoven MA, Pleunis-van Empel MC, Koch H, Grol RP, Dinant GJ, van der Weijden T. Why do patients want to have their blood tested? A qualitative study of patient expectations in general practice. BMC Fam Pract 2006;7:75. (33)

van der Weijden T, van Bokhoven MA, Dinant GJ, van Hasselt CM, Grol RP. Understanding laboratory testing in diagnostic uncertainty: a qualitative study in general practice. Br J Gen Pract 2002;52(485):974-80. (36) van der Weijden T, van Velsen M, Dinant GJ, van Hasselt CM, Grol R. Unexplained complaints in general practice: prevalence, patients' expectations, and professionals' test-ordering behavior. Med Decis Making 2003;23(3):226-31. (34)

van Walraven C & Naylor CD. Do we know what inappropriate laboratory utilization is? A systematic review of laboratory clinical audits. JAMA 1998;280(6):550-8. (52)

van Walraven C & Raymond M. Population-based study of repeat laboratory testing. Clin Chem 2003;49(12):1997-2005. (313)

van Walraven C, Goel V, Chan B. Effect of population-based interventions on laboratory utilization: a time-series analysis. JAMA 1998;280(23):2028-33. (11)

van Wijk MA, Bohnen AM, van der Lei J. Analysis of the practice guidelines of the Dutch College of General Practitioners with respect to the use of blood tests. J Am Med Inform Assoc 1999;6(4):322-31. (327)

van Wijk MA, van der Lei J, Mosseveld M, Bohnen AM, van Bemmel JH. Assessment of decision support for blood test ordering in primary care. a randomized trial. Ann Intern Med 2001;134(4):274-81. (79)

van Wijk MA, van der Lei J, Mosseveld M, Bohnen AM, van Bemmel JH. Compliance of general practitioners with a guideline-based decision support system for ordering blood tests. Clin Chem 2002;48(1):55-60. (88)

Vardy DA, Simon T, Limoni Y, Kuperman O, Rabzon I, Cohen A et al. The impact of structured laboratory routines in computerized medical records in a primary care service setting. J Med Syst 2005;29(6):619-26. (100)

Verstappen WH, ter Riet G, Dubois WI, Winkens R, Grol RP, van der Weijden T. Variation in test ordering behaviour of GPs: professional or context-related factors? Fam Pract 2004;21(4):387-95. (50)

Verstappen WH, van der Weijden T, Dubois WI, Smeele I, Hermsen J, Tan FE et al. Improving test ordering in primary care: the added value of a small-group quality improvement strategy compared with classic feedback only. Ann Fam Med 2004;2(6):569-75. (77)

Verstappen WH, van der Weijden T, Sijbrandij J, Smeele I, Hermsen J, Grimshaw J et al. Effect of a practice-based strategy on test ordering performance of primary care physicians: a randomized trial. JAMA 2003;289(18):2407-12. (75)

Verstappen WH, van Merode F, Grimshaw J, Dubois WI, Grol RP, van der Weijden T. Comparing cost effects of two quality strategies to improve test ordering in primary care: a randomized trial. Int J Qual Health Care 2004;16(5):391-8. (78)

Vinker S, Kvint I, Erez R, Elhayany A, Kahan E. Effect of the characteristics of family physicians on their utilisation of laboratory tests. Br J Gen Pract 2007;57(538):377-82. (47)

Wagholikar A, O'Dwyer J, Hansen D, Chu K. Observing effectiveness of pathology ordering controls in emergency departments. Stud Health Technol Inform 2011;168:172-8. (357)

Waring ME, Roberts MB, Parker DR, Eaton CB. Documentation and management of overweight and obesity in primary care. J Am Board Fam Med 2009;22(5):544-52. (294)

Watkins C, Harvey I, Langley C, Gray S, Faulkner A. General practitioners' use of guidelines in the consultation and their attitudes to them. Br J Gen Pract 1999;49(438):11-5. (323)

Wheatland FT. Medical indemnity reform in Australia: "first do no harm". J Law Med Ethics 2005;33(3):429-43. (145)

Whiting P, Toerien M, de Salis I, Sterne JA, Dieppe P, Egger M et al. A review identifies and classifies reasons for ordering diagnostic tests. J Clin Epidemiol 2007;60(10):981-9. (40)

Whitworth JA. 2003 World Health Organization (WHO)/International Society of Hypertension (ISH) statement on management of hypertension. J Hypertens 2003;21(11):1983-92. (167)

Wild S, Roglic G, Green A, Sicree R, King H. Global prevalence of diabetes: estimates for the year 2000 and projections for 2030. Diabetes Care 2004;27(5):1047-53. (183)

Williams S, Weinman J, Dale J, Newman S. Patient expectations: what do primary care patients want from the GP and how far does meeting expectations affect patient satisfaction? Fam Pract 1995;12(2):193-201. (32)

Wilson E. Pathology demand and funding. Presented at the Pathology Update 2011, Melbourne 4-6 Mar 2011; The Royal College of Pathologists of Australasia; 2011. (346)

Winkelman JW. Less utilization of the clinical laboratory produces disproportionately small true cost reductions. Hum Pathol 1984;15(6):499-501. (133)

Winkens R & Dinant GJ. Evidence base of clinical diagnosis: Rational, cost effective use of investigations in clinical practice. BMJ 2002;324(7340):783. (9)

Winkens RA, Ament AJ, Pop P, Reniers PH, Grol RP, Knottnerus JA. Routine individual feedback on requests for diagnostic tests: an economic evaluation. Med Decis Making 1996;16(4):309-14. (312)

Winkens RA, Pop P, Bugter-Maessen AM, Grol RP, Kester AD, Beusmans GH et al. Randomised controlled trial of routine individual feedback to improve rationality and reduce numbers of test requests. Lancet 1995;345(8948):498-502. (74)

Winkens RA, Pop P, Grol RP, Bugter-Maessen AM, Kester AD, Beusmans GH et al. Effects of routine individual feedback over nine years on general practitioners' requests for tests. BMJ 1996;312(7029):490. (386)

Winkens RA, Pop P, Grol RP, Kester AD, Knottnerus JA. Effect of feedback on test ordering behaviour of general practitioners. BMJ 1992;304(6834):1093-6. (95)

Wolfe R & Hanley J. If we're so different, why do we keep overlapping? When 1 plus 1 doesn't make 2. CMAJ 2002;166(1):65-6. (130)

Wong C & Chu S. Optimising clinical laboratory service delivery with electronic decision support: An analysis of market structure and impacts of regional clinical information strategy - Part 1. Health care and informatics review online 2004;8(3). (31)

Woods K. PSA predicament. Medical Observer 2009 May 8:19-20. (245)

Woolley A, Kopecky S, Kottke T, O'Connor P, Hanson S, Conroy W et al. Lipid management in adults. ICSI health care guideline. Institute for Clinical Systems Improvement, 2007. (220)

World Health Organization Collaborating Centre for Drug Statistics Methodology. Anatomical Therapeutic Chemical (ATC) classification index with Defined Daily Doses (DDDs). January 1998 ed. Oslo: WHO, 1997. (114)

World Health Organization Collaborating Centre for Drug Statistics Methodology. International language for drug utilization research. 2012. Viewed 27 February 2013, http://www.whocc.no/. (115)

World Health Organization. Obesity: preventing and managing the global epidemic: report of a WHO consultation. WHO technical report series: 894. 2000. Viewed 27 February 2013, http://whqlibdoc.who.int/trs/WHO_TRS_894.pdf. (109)

World Health Organization. World Health Organization family of international classifications: definition, scope and purpose. Geneva: WHO, 2004. Viewed 27 February 2013, www.who.int/entity/classifications/en/FamilyDocument2007.pdf. (111)

Young DW. Improving laboratory usage: a review. Postgrad Med J 1988;64(750):283-9. (94)

Zaat JO & van Eijk JT. General practitioners' uncertainty, risk preference, and use of laboratory tests. Med Care 1992;30(9):846-54. (378)

Zaat JO, van Eijk JT, Bonte HA. Laboratory test form design influences test ordering by general practitioners in The Netherlands. Med Care 1992;30(3):189-98. (97)

Appendices

Appendix 1: Publications and presentations from this thesis

Reports:

Bayram C, Britt H, Miller G, Valenti L. Evidence-practice gap in GP pathology test ordering: a comparison of BEACH pathology data and recommended testing. Sydney: The University of Sydney. 2009. Viewed 7 March 2013, available at www.health.gov.au/internet/main/Publishing.nsf/Content/9B6B7B4EAAC0383DCA 256F180048CCBD/\$File/BEACH_QUPP%20report_FINAL_050609.pdf

Bayram C & Valenti L. GP pathology ordering. In: Britt H & Miller GC (eds). General practice in Australia, health priorities and policies 1998–2008. General practice series no. 24. Cat. no. GEP 24. Canberra: Australian Institute of Health and Welfare, 2009;57–86.

Presentations:

Bayram C. Quality of pathology ordering in Australian general practice. Presented at the Research Presentation Day: School of Public Health, University of Sydney June 2008 Sydney.

Bayram C, Valenti L, Britt H, Miller G. GP pathology ordering in Australia 2000–02 to 2006–08. Presented at the 2009 General Practice & Primary Health Care Research Conference, Melbourne. Primary Health Care Research & Information Service, July 2009.

Bayram C, Britt H, Miller G, Valenti L. The value of pathology to healthcare -Pathology ordering by general practitioners in Australia. Presented to the Business of Pathology meeting, Melbourne. Australasian Association of Clinical Biochemists & The Dark Report, November 2009.

Bayram C, Britt H, Miller G, Valenti L. Evidence-practice gap in GP pathology test ordering: a comparison of BEACH GP pathology ordering data and recommended testing. Presented at the 2010 Primary Health Care Research Conference, Darwin. Primary Health Care Research & Information Service, June 2010. Bayram C, Britt H, Miller G, Valenti L. Evidence-practice gap in FP pathology ordering: A comparison of BEACH pathology data and recommended testing. Presented at the 38th NAPCRG Annual Meeting, Seattle. North American Primary Care Research Group, November 2010.

Bayram C, Harrison C, Britt H, Miller G. Pathology testing in 2020: the projected growth of family physician pathology test ordering. Presented at the 38th NAPCRG Annual Meeting, Seattle. North American Primary Care Research Group, November 2010.

Bayram C. Evidence-practice gap in GP pathology ordering: a comparison of BEACH pathology data and recommended testing. Presented at the Research Presentation Day: School of Public Health, University of Sydney, November 2010 Sydney.

Britt H, Bayram C, Miller G, Valenti L, Harrison C. GP-initiated pathology testing, data from the BEACH program. Presented at the Pathology Update 2011 conference, Melbourne. Royal College of Pathologists of Australasia, March 2011.

Appendix 2: Example of a 2007–08 BEACH recording form

BEACH (Detternin		aluatio			<u>i</u> eann) -		uity a	inu m	eaune	ente	urvey - r	BEACH General Practice & Sta	tistics Classifica	lion Unit U	University of S	vdney 1996 DC					
Encounter Number	Date of e	ncounte	er Date	of Birt	:h	5	Sex		Pati	ent F	ostcode		Yes /	No	PAT	IENT SE	EN BY	GP			l l
	1	1		1	1			- 			1	New Patient	□		PAT	IENT NO	T SEE	N BY	GP		
			-			· — ['						Health Care/Benefits Car	d		Mec	licare					
START Time	Detient	1										Veterans Affairs Card	🗖		(if ap	oplicable)	N.	/orkers	comp p	aid	Ц
	Reasons	for 1.										NESB	🗖		1.		_ ~	tato Gr	wt/Otho	r naid	
AM / PM	Encounte	er 2.										Aboriginal			2.		3	late Gu	VUOLIIEI	paiu	
(please circle)		3										Torres Strait Islander			3.		N	o charç	je		шЦ
Diagnosis/		0.			Pr	oblem	Statu	IS va			Diagnos	sis/				Pr	oblem	State	IS U		
Problem (1):					Ne	wΠ	Old [elated		Problem	12:				Ne	ew □	DId		ork/	
Drug Name AND Form	n for this prob	lem	Strength of	Dose	Frequency	No. of	OTC	GP	Drug s	status	Drug Name	AND Form for this problem	Streng	gth of	Dose	Frequency	No. of	OTC	GP	Drug	status
			product			Rpts		Supply	New	Cont.			proc	luct			Rpts		Supply	New	Cont.
1.											1.						<u> </u>		\square		
2.											2.										
3											3										
<u>.</u>											0.						<u> </u>	+			
4. Dressedures othe			na allina Ak		a sult fau this						4.										L
Procedures, othe	r treatmer	ns, cou	nselling u	is con	Suit for this	s propi	em	r	Jroo		Procedures, other treatments, counselling this consult for this problem										
1.		N	urse?	2.				1	Nurse'	? 🗋	1.	N	urse?		2.				Ñ	urse'	2
Diagnosis/					Pr	oblem	Statu	IS VA	<i>i</i>		Diagnos	is/				Pr	oblem	Statu		ork	
Problem (3):					Ne	wΠ	Old [elated		Problem	4:				Ne	w	Old [∃ re	lated	
Drug Name AND Form	n for this prob	lem	Strength of	Dose	Frequency	No. of	OTC	GP	Drug s	status	Drug Name	AND Form for this problem	Streng	th of	Dose	Frequency	No. of	OTC	GP	Drug s	status
			product			Rpts		Supply	New	Cont.			prod	uct			Rpts		Supply	New	Cont.
1.											1.		_				\square				
2.											2.										
3											3										
4			1								4		+								
4. Procedures offe	r treatmer	nts cou	nselling th	l lis con	sult for this	s probl	em				4. Procedu	res other treatments cou	Insellin	a th	is con	sult for thi	s probl	em			
	i doudnoi	Pr		10 0011	our for this	5 probi	CITI	F	Prac	_	1 1000uu	Pi	ac -		0 0011	Juit for the	proble	onn	Р	rac	_
1.		Nu	urse? 🛛	2.				١	Jurse	? 📙	1.	N	urse? L		2.				N	urse	2
NEW REFERRALS	S, ADMISS		ablem (a)	IMAGI	NG/Other	tests	م مار ، م ال		we he le m	a (a)	PATHOLO	DGY <u>Pi</u>	oblem(s	<u>s)</u>	PATH	OLOGY (C	ont)		P	robler	<u>n(s)</u>
		<u>P1</u>	<u>(z)maido</u>			B	ody sit	<u>e P</u>	ropien	<u>1(S)</u>	1	1	2 3	4	4				1	2 3	34
1		1	2 3 4	1				1	2 3	34	2.	1	23	4	5.				1	2	34
2.		1	234	2				1	2 3	34	3.	1	23	4							
Detient reported	To t	ho potic	nt if 10+.		To t	ho noti	ont if /	101.			Howman	w 'standard' drinka da yau	Цож	oftor	dovo	u have f	ormor			20	
	Whi	ch best o	lescribes vo	ur smok	tina How	often	do vou	have a	drink		have on a	a typical day when you are	stand	dard	drinks	on one of	ccasio	n?	131 11	ne	
Height:	stat	us?	,.		cont	aining	alcoho	?			drinking?)					_				
	cm Sm	oke dailv	,		□ Nev	/er							Neve	er			<u>ך</u>	11	1	:	1
	0		niopolly			nthly or	less						Less	than	month	ly	Ľ	11			Ч
VVeight:	Smo	JKE OCCE	isionally	ا		ce a we	ek/fort	night				1 1	Mont	hly			<u>C</u>	11	AM		-)
	Prev	vious sm	oker		L 2-3	times a	a week						Wee	kly			<u>C</u>	11	(piease	e circl	э)
	Nev	er smok	ed		∐ 4+1	times a	week						Daily	or a	lmost d	aily	E	1		BA1	0

BEACH (Bettering the Evaluation And Care of Health) - Morbidity and Treatment Survey - National OBEACH (General Practice & Statistics Classification Unit University of Styling 1996 DOC ID

Appendix 3: GP characteristics questionnaire, 2007–08

The University of Sydney at Westmead Hospital	Australian General Practice Statistics and Classification Centre
Doctor Identification Number	a collaborating unit of the Australian Institute of Health and Welfare
Please fill in boxes or circle answers 1. SexMale / Female (please circle) 2. Age	16. Over the past four weeks have you provided any patient care(Circle all that apply) As a locum 1 In a deputising service 2 In a residential aged care facility 3 As a salaried/sessional hospital medical officer4 None of the above 5 17. What are the normal after-hours arrangements for your practice? (Circle all that apply)
f. In which GP Division is this practice	Practice does its own
7. Year of graduation 8. Place of graduation (primary medical degree): Aust	18. Do you bulk bill ALL patients?
 9. Do you conduct any of your consultations in a language other than English? No	19. To what extent are computers used - (i) at your major practice? (ii) by you (at work)? Not at all
 12. Is your major practice accredited ? Yes / No 13. Is there a practice nurse at your major practice address ?	20. Is your major practice site a teaching practice? (Circle all that apply) for undergraduates

Thank you for participating in the **BEACH PROGRAM**.

AGPSCC, Acacia House, Westmead Hospital, WESTMEAD, 2145. Ph: 02 98458151 fax: 02 98458155 email: janc@med.usyd.edu.au Web http://www.fmrc.org.au

Appendix 4: Instructions for participating GPs

BEACH - Bettering the Evaluation And Case of Health NATIONAL MORBIDITY AND TREATMENT STUDY INSTRUCTIONS FOR PARTICIPATING DOCTORS

USING THESE INSTRUCTIONS

Use these instructions as a resource to complete the forms. While they may look daunting, most of the form is self-explanatory. The instructions contain:

- an example consultation scenario
- a completed form for the example scenario
- detailed explanations for each question on the form.

Reading these instructions will:

- show you how to fill out the forms
- ultimately save you time
- decrease the variation among practitioners in their recording techniques.

When to complete the forms

Complete the forms during the course of the consultation as

- some information needs to be asked of the patient
- it will be faster and more accurate than going back to your records at the end of the day. To show the full range of your clinical activity it is vital that you take the pad with you to all hospital, home and nursing home visits.

Informing patients

In your research pack there are two copies of a gloss board notice which tells patients about the study and of their right to refuse to allow inclusion of their unidentified data. Please ask your reception staff to **ensure your patients read the notice**. Patients who consult with you in another language should be made aware of their options regarding the study. For patients not seen, nursing home visits and palliative care, please use your professional discretion in this matter. The Human Research Ethics Committee of the University now requires that a mark be placed in the medical record of each patient who agrees to allow their data to be included in BEACH. Please record B \checkmark or Beach Y (for yes) in the medical record. This action could be performed by any authorised staff member.

Patient information questions at the bottom of the form

These vary and are presented in blocks within the pad, so please read carefully the instructions relating to these questions. When the questions change in the pad, a green instruction sheet gives you instructions for the next block of forms.

EXAMPLE OF ONE TYPE OF RECORDED ENCOUNTER

This is a description of the data recorded on the sample recording form that follows.

On April 30th 2009, Mr A comes to the surgery. He has read the patient information card while in the waiting room and agrees to be included in the study. The consultation starts at 9.10 am. From the medical record you note Mr A's date of birth is 13/3/1947, his postcode is 2145 and that he carries a Health Care Card. You ask if he is from a Non English-Speaking Background or identifies himself as an Aboriginal or Torres Strait Islander and he answers no. You use the tick boxes to show his responses.

He is a regular patient suffering from hypertension and says he has almost run out of Cardizem and requests a script. After examination you feel Mr A is not responding to medication and you refer him to a cardiologist but also provide him with the required script for Cardizem CD 180mg tablets to be taken once a day with two repeats. You also recommend he try to lose weight, advise a low-fat diet and send him for cholesterol screening.

Mr A then complains about his ribs. He says he slipped and bumped himself at his part-time job the day before and his ribs are hurting. You send him for an x-ray and advise him to take Panadol for the pain.

Finally, Mr A asks for his annual flu injection. You tell Mr A that you will arrange for the practice nurse to give him a FluVax injection from your practice supply.

You tell him you have to ask him a couple of extra questions for the study. He says that he is 170 centimetres tall and weighs about 90 kilos. He says he no longer smokes and has a drink most nights but never more than one or two. You show him the 'standard drinks' card and he confirms one or two standard drinks.

This has been a standard surgery consultation in the Item 23 category, with an Item 10993 for the practice murse, which finishes at 9.28 am.

Encounter Date of encounter Date of Binth Sex Patient Postcode Ves / No Patient Nor SEEN BY GP G START Time 9:10 Carbon 13/03/1947 M F 2:45 New Patient M Patient Patient Patient Patient Patient M F 2:45 New Patient M Patient Patient No charge M Patient No charge M Patient No charge No charge <t< th=""></t<>
30 / 04 / 09 13 / 03 / 1947 _ M Z F _ 2145 New Patient M M M Reference _ 2145 New Patient M M M Reference _ 1164 Care/Benefits Card M Reference _ M
START Time Patient Reasons for (A)) PM Script for hypertension.tablets/ 2. Sore r/by Item Nos: (item
9:10 Reasons for Encounter 1. Scrept by hypertension v 2. Scrept by hypertension v 3. NES8
Amy PM (please circle) Encounter 2. Sore riby Aboriginal
Image: conception of the problem Status Problem St
Diagnosis/ Problem ①: Hypertension New Old Problem Status (related Work (related Diagnosis/ Problem ②: Problem ③: Problem Status (related Work (related Image: status (related Image: status (related Work (related Image: status (related
Problem (): Itype (error for this problem (): New () Old () Problem (): Problem (): Problem (): Problem (): New () Old () Problem ():
Did Status Opcoduct Dose Prequency No. of OTC Supply New Cont. 1. Cardizgem CD tablets 180mg 1 tab 1 daily 2 V 1. Panadol tablets 500mg/2 tabls qid V V 2. 3. 1 1 daily 2 V 1. Panadol tablets 500mg/2 tabls qid V V 2. 3. 1 1 daily 2 V 1. Panadol tablets 500mg/2 tablets qid V V V 2. 3. 1 1 1 1 2. 1
1. Cardizem CD tablets 180mg 1 table 1 dadly 2 Image: Processing of tablets 500mg/2 table gid Image: Processing of tablets Image: Processing of tablet
2.
3. .
4. Image: Second se
Procedures, other treatments, course lling this consult for this problem Prac Nurse? 2. Prac Nurse? 1. 1. Prac Nurse? 2. Prac Nurse? 1. 1. Prac Nurse? 2. P
1. Dietary advice Prac Nurse? 2. Prac Nurse? 1. Prac Nurse? 2. Prac Nurse? 1. Diagnosis/ Problem ③: Immunisation Problem Status New Old Ø related Diagnosis/ Problem ④: Problem Status New Old Ø Problem ④: Problem ④: Problem Ø: Problem Ø: New Old Ø related Diagnosis/ Problem Ø: Problem Ø: New Old Ø related Diagnosis/ Problem Ø: Problem Ø Strength of New Dose Frequency No. of OTC GP Drug status New Old Ø Problem Ø: Drug Name AND Form for this problem Strength of product Dose Frequency No. of OTC GP Drug status New Cont. New
Diagnosis/ Problem (3): Immunisation Problem Status New Work related Diagnosis/ Problem (4): Problem (5): Problem Status New Work related Drug Name AND Form for this problem 1. Strength of product Dose Frequency No. of Rpts OTC GP Supply Drug status New Ott Drug Name AND Form for this problem (4): Strength of product Dose Frequency No. of Rpts OTC GP Supply Drug status New Ott Drug Name AND Form for this problem (4): Strength of product Dose Frequency No. of Rpts OTC GP Supply Drug status New Ott Drug status (4): Dose Frequency No. of Rpts OTC GP Supply Drug status (4): Drug status (4): Diagnosis/ (4): Diagnosis/ (4): Drug status (4): Diagnosis/ (4): Diagnosis/
Problem ③: Immunisation New □ Old ☑ related □ Drug Name AND Form for this problem Strength of product Dose Frequency No. of Rpts OTC GP Drug status Problem ④: New □ Old □ related □ 1. FluX ax injection 0.5 ml 1 inj stat 0 ✓ ✓ ✓ ✓ Prug Name AND Form for this problem Strength of product Dose Frequency No. of Supply New □ Ont Prug status Prug Name AND Form for this problem Strength of product Dose Frequency No. of No. of No. of Supply OTC GP Drug status Prug status New □ Ont New □ Ont<
Drug Name AND Form for this problem Strength of product Dose Frequency No. of Rpts OTC GP Drug status Drug status Drug Name AND Form for this problem Strength of product Dose Frequency No. of Rpts OTC GP Drug status 1. FWW av injection 0.5 ml 1 inj stat 0 i
1. FluWax injection 0.5 ml 1 inj stat 0 \checkmark \checkmark \checkmark 1. Image: Constraint of the state
2. Image: Strategy of the streat ments, councer line streat ments, councer line of the streat me
3.
4. 4. <t< td=""></t<>
Procedures, other treatments, counselling this consult for this problem Procedures, other treatments, counselling this consult for this problem 1. Injection given Prac Nurse? 2. Prac Nurse? 1. Prac Nurse? 2. Prac Nurse? Prac Nurse? 2. Prac Nurse? Prac Nurse? 2. Prac Nurse? Problem(s) Problem(
1. Injection given Prac Nurse? Problem(s) PathoLogy (cont) Problem(s) Problem(s)
NEW REFERRALS, ADMISSIONS IMAGING/Other tests Body site Problem(s) PATHOLOGY Problem(s) PATHOLOGY (cont) Problem(s)
$\frac{Problem(s)}{1} + \frac{1}{2} + \frac{1}{$
1. <u>Cardiologist</u> 1 2 3 4 2 1 2 3 4 2 1 2 3 4 5 1 2 3 4
2 1 2 3 4 3 1 2 3 4 3 1 2 3 4
Patient reported To the patient if 18+: To the patient if 18+: How many 'standard' drinks do How often do you have 6 or FINISH Time
Height: Which best describes your How often do you have a drink you have on a typical day when more standard drinks on one
smoking status? containing alcohol? you are drinking? occasion?
Smoke daily
Weight: Once a week/fortnight 2 Monthly Image: Control of the second seco
90 kg Never smoked

BEACH (Bettering the Evaluation An	d <u>C</u> are of <u>H</u> ealth) - Mor	bidity and Tre	eatment Survey - I	National © BEACH The Un	iversity of Sydney 1996		DOC	ID	
Encounter Number Date of encounter	Date of Birth	Sex	Patient Postcode		Yes /	No	PATIENT SEEN	BY GP 🗹	Í
30 / 04 / 09	13/03/1947		2145	New Patient		M	PATIENT NOT S		
DATE: Enter day, month and year of encounter.	BIRTH: Enter day, month and FULL YEAR of patient's	SEX: Tick box for sex of patient.	PATIENT POSTCODE: Postcode of patient's home	Health Care/Bener Veterans Affairs Ca NESB Aboriginal Torres Strait Island	fits Card ard	াইত্র্	Medicare Item Nos: (if applicable) 1. 23 2. 10993 3.	Workers comp paid State Govt/Other paid . No charge	
	birtir.		address.			P	ATIENT SEEN / N	OT SEEN.	
ENCOUNTER NUMBER: Pre-stamped with consecutive encounter number 001-105. A few extra forms allow for error - please complete up to 102. This is not a patient identification number. If you see the same patient more than once during the recording period, complete a new form for each encounter. No linking of forms is required.	sk the patient the following New Patient: If this is the patient has been seen p 'Old' box. Health Care / Benefits Veterans' Affairs Card Patients may hold both V NESB: Non-English Sp English. Aboriginal / Torres Stra Islander origin?" The pa boxes. Otherwise, tick 'Ye or tick 'No' for both optio	g questions and e patient's first reviously at this Card Holder: e holder: Indica /eterans' and H eaking Backgro ait Islander: As tient may answ es' to whicheve ns if that is the	tick either 'Yes' or ' visit to your practice practice by you or g unemployed, pen te whether the patie lealth Care cards. bund i.e. primary lan sk the patient "Are y rer "Yes" to either or r option the patient r patient's response.	No' for each item. e, tick the 'New' box one of your associa sioner, low income nt has a Veterans' (guage spoken at ho ou of Aboriginal or T both. If both, tick be iominates and 'No' t	. If the ites, tick the earner. Card. ome is NOT forres Strait oth 'Yes' o the other,	Ti	ck 'PATIENT SEEI to-face encounter ck 'PATIENT NOT service is provide clinical action resi information into th but the patient is referrals, renewal certificates, case items, where the face-to-face. B - Informed cons obtained from the carer prior to inclu-	N' if this is a face- SEEN' if a ad where your ults in entry of ne patient's record not seen e.g: s for prescription/ conferences/EPC patient is not seen sent must be a patient or their usion in the study,	

** This question was formally adopted in 1995 by the Australian Bureau of Statistics as the standard for measuring membership of the Indigenous population.

McLennan, W. & Madden, R. (1999) The health and welfare of Australia's Aboriginal and Torres Strait Islander Peoples ABS 4704.0 p.149

These items ask about the type of renumeration claimable for the encounter.

Medicare Encounters

Item No:

- Write the item number/s when there is a charge through Medicare, bulk-billed or otherwise.
- When multiple item numbers are involved, record the consultation item, e.g. 23 and up to 2 others.
- If unsure of item number, please provide type and level of consultation e.g. NHV-B (nursing home visit level B) or diabetes care plan - C.

 Include item numbers (when applicable) for services when the patient is not seen e.g. case conferences, enhanced primary care (EPC) items.

Non-Medicare

Workers Compensation paid: For consultations claimable through workers compensation.

State Govt / Other paid: If the encounter is being paid for by a state government (e.g. hospital or other state agency), insurance company or other source. DOES NOT include additional cash payments made by patients charged through Medicare, but would include 'cash only' patients e.g. overseas travellers.

No charge: For services you provide free - with no payment from ANY source.

Encounter Number Date of encounter Date of Dirth	PATIENT REASON FOR ENCOUNTER (RFE):
Date of encounter Date of Birth Sex Patient Postcode	Record at least 1 and up to 3 patient reasons for the encounter.
//////M□F□ └	• The reason for encounter is the patient's view of the reasons he/she is
START Time Script for hypertain i on tablety	consulting you. The patient's own words should be used. May include:
9:10 Patient	 symptoms e.g. "runny nose",
(AMV PM Fincounter 2. Sore ribs	 diagnoses e.g. "diabetes",
(nlease circle)	 requests for service e.g. "script for BP", "referral".
Diamasia (other examples - "Worried about", "follow-up", "check-up
Broblem (1): Hopertension Now Decided V rolated V	circulatory"
Drug Name AND Form for his problem Strength of Dose Frequency No. of DTC GP Drug status	 Specify the body system even when this is not stated by the patient but is understand between user
product Rpts Supply New Cont.	is understood between you.
1. Cardízem CD tarlets 180mg 1 tab 1 daily 2	RK RELATED: Irrespective of the source of payment for the encounter, tick if:
2.	it is likely in your view that the symptom or problem has resulted from work-
	related activity or workplace exposures.
	where there is uncertainty but it is more likely than not that the condition is
Procedures, other treatments, courselling this consult for this problem	work-related
Diatary advise Plac P	If there is a pre-existing condition which is thought to have been significantly
1. Dietur y utavice Nulse? L 2. Nurse? L	exacerbated by work activity or workplace exposures.
START TIME PROBINE Record the time the consultation STARTED in hours and mins STARTED in hours and mins and circle whether the time was AM or PM. 9 : 10 eg. AMY PM	LEM STATUS: lew' if: s is a new problem to the patient, or s is a new episode of a recurrent problem (e.g. URTI), or e patient has not been treated for that problem by any medical practitioner fore. Id' if the patient has been seen before by ANY medical practitioner for this ronic problem or this episode of an acute problem.
PROCEDURES, OTH	IER TREATMENTS, COUNSELLING:
Use one Diagnesis/problem box for each diagnesis/problem For each problem:	
Only record problems actually dealt with at this appointer Record up to two	procedures, other treatments or counselling.
 Only record problems actually dealt with at this encounter Only include those 	e ACTUALLY PROVIDED at the encounter.
 Include III-defined conditions (e.g. cougn), preventive care (e.g. Include in this set in the set in the	ection actions such as NB - If a practice
social problems (e.g. "problems with spouse").	nurse performs the
 Diagnose at the highest level possible with the information available diet and ever 	cise advice procedure, please
(e.g. for diabetes, differentiate between IDDM/NIDDM/Type 1/Type 2	icates tick the box marked
etc.)	in this section: 'Prac Nurse?'
 The order in which you record the problems is not significant - they history 	
do not have to match the order of the RFEs.	cal examinations e.g. blood pressure checks
If more than four problems are managed at the consultation, record	
the four problems that best describe the breadth of the consultation.	ging, or pathology ordered (there are sections for these).

MEDICATIONS: NB - ONLY record medications that were prescribed / advised / supplied at this encounter Record medications when

- a prescription is written at this encounter,
- you recommend that the patient take an "over the counter" (OTC) medication.
- you administer or supply a medication/vaccine. eg. If 'Immunisation' is the problem managed, please enter drugs administered at this encounter, (e.g. CDT, DTP) or any drug samples you provide.



OTHER MEDICATION EXAMPLES

ALWAYS SPECIFY	Syrups	Drug Name AND Form for this problem	Strength of product	Dose	Frequency	No. of Rpts	OTC	GP Supply	Drug s New	status Cont.
NAME and FORM,	Dose may be written in "mg" or "ml" but strength	→ <u>1. Amoxil syrup</u>	250mg/5ml	2.5m	l tds	0			✓ ✓	
STRENGTH,	must be specified.	2. Panadol syrup	120mg/5m	l 10m	l 4 hrly		\checkmark		\checkmark	
DOSE and	Creams Specify the name, form, strength and no. of applications per day. There is no need to specify pack size.	→ Drug Name AND Form for this problem 1. Aristocort cream	Strength of product	Dose	Frequency	No. of Rpts	отс	GP Supply	Drug s New	status Cont.
	Multiple strengths of same drug	Drug Name AND Form for this problem	Strength of product	Dose	Frequency	No. of Rpts	отс	GP Supply	Drug s New	status Cont.
	the same drug to achieve a particular	1. Warfarin tablets	1mg	1 tab	mane	2				\checkmark
	dose, specify both.	2. Warfarín tablets	2mg	1 tab	mane	2				\checkmark
Ir T th N "F if e	njections ick GP supply only if you have provided ne vaccine / medication yourself. IB Please write ' injection given ' in the Procedures, other treatments" section you have given the injection at this ncounter.	 Drug Name AND Form for this problem 1. Fluvax inject 2. Engerix B Adult inject 	Strength of product 0.5 ml 20mcg/m	Dose 1 íný l 1 ínj	Frequency stat stat	No. of Rpts 0	отс	GP Supply	Drug s New ✓	status Cont.
	Inculin	Drug Name AND Form for this problem	Strength of product	Dose	Frequency	No. of Rpts	OTC	GP Supply	Drug s New	status Cont.
	Specify the number of units prescribed.	→ <u>1. Mixtard 30/70 injection</u>	100íu/ml	20un	its mane	- 1				\checkmark
		2. Mixtard 30/70 injection	100íu/ml	12un	its nocte	1				\checkmark
	Inhaled medications Specify the mode of delivery, e.g. inhaler, turbuhaler, nebules, etc. and the strength.	 Drug Name AND Form for this problem 1. Pulmicort turbuhaler 2. Prednisone tablets 3. Ventolin inhaler 	Strength of product 400mcg 25mg 100mca	Dose 2 puffs ½ tab 2 puffs	Frequency bd 1 daily	No. of Rpts 1 0 V	отс	GP Supply	Drug s New ✓	Status Cont.

NEW REFERRALS / ADMISSIONS:

- Specify the type of specialist(s) or allied health professional(s) to whom the referral has been made, e.g. dermatologist, physiotherapist, hospital emergency department etc.
- Record new referrals only. Do not include continuation referrals.
- Indicate the problem or problems for which the referral was made by circling the appropriate problem number.
- Include referrals for clinical measurements such as spirometry and ECG

	PATHOLOGY:	
	IMAGING/Other tests (+ Body site): Imaging Imaging Give details of up to five pathology tests ordered at the encounter. Document one test per line. Imaging Circle the associated Diagnosis/problem number(s).	
	 circle the number(s) of the Diagnosis/problem which is being investigated For single tests, write the test name (e.g. <i>HBA1C</i>, <i>pap smear</i>). If ordering a set of tests such as a FBC or LFT or lipids or thyroid function, record them in this grouped form. You don't need to list each of the individual tests incorporated in the set. Do not record simple urine dip stick tests. 	
	Lateralization is not required.	
NEW REFERRALS, ADMISSION	IMAGING/Other tests Body site Problem(s) PATHOLOGY Problem(s) PATHOLOGY (cont) Problem(s)	2
1. <u>Cardíologíst</u> 2	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	4

Appendix 5: Patient information card



The University of Sydney

School of Public Health

Australian General Practice Statistics and Classification Centre Family Medicine Research Centre

a collaborating unit of the Australian Institute of Health and Welfare



INFORMATION FOR PATIENTS

The BEACH ® Project

Today your doctor is taking part in a National Survey of general practice called *BEACH*[©] (*Bettering the Evaluation and Care of Health*). This study is being done by the Australian General Practice Statistics and Classification Centre, University of Sydney, with the Australian Institute of Health and Welfare.

Your Doctor will be recording information about each patient he/she sees (age, gender etc), the problems that you see the Doctor about and the treatments given to you. **There are no names on the forms so you cannot be identified.** The information about today's visit to the doctor will be one record in a set of 100,000 records collected in general practices across Australia over the year.

This information will be used by researchers to describe what happens in general practice and to look at different aspects of health care; by government departments to help them plan for our future health; and by pharmaceutical companies to gain a picture of the people who use their drugs and of the problems being treated with the drugs they produce.

Remember: your name will not be on the form and no information will ever be released which could possibly let anyone know who you are. However, if you do not wish your doctor to record any unidentified information about you or your visit please tell your Doctor as soon as you go in. Such a decision will not affect the care your doctor is providing in any way.

SEE OVER FOR PROJECT DETAILS

(page 1 / 2)

BEACH [©] **Program Details**

This program has been approved by the Ethics Committees of the University of Sydney and the Australian Institute of Health and Welfare (AIHW). The data are being collected under the AIHW ACT 1987 and in accordance with the Privacy Act 1988 (Amended 2001).

Organisations contributing financially to the conduct of this study in 2009–2010 are:

- ✦ The Australian Institute of Health and Welfare
- ✦ The Australian Government Department of Health and Ageing
- ✦ AstraZeneca Pty Ltd (Australia)
- ✦ Janssen-Cilag Pty Ltd
- ♦ Merck Sharp & Dohme (Aust) Pty Ltd
 ♦ Pfizer Australia Pty Ltd
- ✦ Abbott Australasia Pty Ltd
- ◆ Sanofi-Aventis Australia Pty Ltd
- ♦ Wyeth Australia Pty Ltd

BEACH is endorsed by the Royal Australian College of General Practitioners BEACH is endorsed by the Australian Medical Association





FURTHER INFORMATION:

Australian General Practice Statistics and Classification Centre The University of Sydney Acacia House, Westmead Hospital Westmead 2145 Phone: (02) 9845 8151 Fax: (02) 9845 8155 Email: <u>janc@med.usyd.edu.au</u> Web: http://www.fmrc.org.au

Any person with concerns or complaints about the conduct of this study can contact the Senior Ethics Officer, Ethics Administration, University of Sydney on (02) 9351 4811 (Telephone); (02) 9351 6706 (Facsimile) or gbriody@mail.usyd.edu.au (Email) (page 2/2)

Appendix 6: ICPC-2 and ICPC-2 PLUS code groups used for analysis of problems managed

Table A6.1: Code groups from ICPC-2 and ICPC-2 PLUS

Problem managed	ICPC-2 rubric	ICPC-2 PLUS code	ICPC-2/ICPC-2 PLUS label
Abdominal pain	D01		Pain/cramps; abdominal general
	D06		Pain; abdominal localised; other
Abnormal test results	A91		Abnormal result investigations NOS
	B84		Unexplained abnormal white cells
	U98		Abnormal urine test NOS
	X86		Abnormal cervix smear
Anaemia	B80		Iron deficiency anaemia
	B81		Anaemia; vitamin B12/folate deficiency
	B82		Anaemia; other/unspecified
		B78002	Anaemia; sickle cell
		B78003	Anaemia; hereditary haemolytic
		B79001	Anaemia; congenital
		B79004	Anaemia; hereditary
Arthritis – all	L88		Rheumatoid/seropositive arthritis
	L89		Osteoarthrosis of hip
	L90		Osteoarthrosis of knee
	L91		Osteoarthrosis, other
		L70009	Arthritis; pyogenic
		L70010	Arthritis; viral
		L70021	Arthritis;septic
		L81003	Arthritis; traumatic
		L81015	Haemarthrosis
		L83010	Arthritis; spine cervical
		L83011	Osteoarthritis;spine;cervical
		L84003	Arthritis; spine
		L84004	Osteoarthritis;spine
		L84009	Osteoarthritis;spine;thoracic
		L84010	Osteoarthritis;spine;lumbar
		L84011	Osteoarthritis;lumbosacral
		L84012	Osteoarthritis;sacroiliac
		L92006	Arthritis; shoulder
		L92007	Osteoarthritis;shoulder
		L92011	Humeroscapular periarthritis

Problem managed	ICPC-2 rubric	ICPC-2 PLUS code	ICPC-2/ICPC-2 PLUS label
Arthritis - all (continued)		S91002	Arthritis; psoriatic
		T99063	Arthritis; crystal (excl. gout)
Back complaint	L02		Back symptom/complaint
	L03		Low back symptom/complaint
	L86		Back syndrome with radiating pain
Blood test – all ^(a)	-34		Blood test
Cardiovascular check-up	K30		Medical examination/health evaluation, complete, cardiovascular system
	K31		Medical examination/health evaluation, partial, cardiovascular system
Depression	P03		Feeling depressed
	P76		Depressive disorder
Female genital check-up/Pap smear	X30		Complete medical examination/health evaluation, female genital
	X31		Partial medical examination/health evaluation, female genital
	X37		Histological / exfoliative cytology, female genital
Health check (15+ years) ^(b)		A30001	Health evaluation; complete
		A30002	Exam; complete
		A30011	Check up; complete
		A30010	Exam; complete; physical
		A30017	Medical exam; complete
		A30028	Health assessment
		A30029	Check up; adult health; complete
		A31001	Health evaluation; partial
		A31003	Assessment; normal growth
		A31004	Exam; partial; physical
		A31005	Check up; partial
		A31006	Exam; partial
		A31008	Health screening
		A31012	Check up
		A31013	Medical exam
		A31017	Assessment; aged care
		A31025	Check up; adult health; partial
		A31026	Health surveillance; partial
		A31027	Assessment; physical fitness
		A31030	Check up; height/weight

Table A6.1 (continued): Code groups from ICPC-2 and ICPC-2 PLUS

Problem managed	ICPC-2 rubric	ICPC-2 PLUS code	ICPC-2/ICPC-2 PLUS label
Hypertension	K86		Hypertension; uncomplicated
(non-gestational)	K87		Hypertension; complicated
Ischaemic heart	K74		Ischaemic heart disease with angina
disease	K75		Acute myocardial infarction
	K76		Ischaemic heart disease without angina
Lipid disorders	Т93		Lipid disorders
Menstrual problems	X02		Pain; menstrual
	X03		Pain; intermenstrual
	X05		Menstruation; absent/scanty
	X06		Menstruation; excessive
	X07		Menstruation; irregular/frequent
	X08		Intermenstrual bleeding
	X09		Premenstrual symptom/complaint
	X10		Postponement of menstruation
Overweight/obesity	T82		Obesity
(adults) ^(c)	T83		Overweight
Pregnancy	W01		Question of pregnancy
	W78		Pregnancy
	W79		Unwanted pregnancy
Prescription – $all^{(d)}$	-50		Medication prescription/request/ renewal/injection
Sexually transmitted	B90		HIV infection/AIDS
infection	X70		Syphilis; female
	X71		Gonorrhoea; female
	X73		Genital trichomoniasis; female
	X74		Pelvic inflammatory disease
	X90		Genital herpes; female
	X91		Condylomata acuminata; female
	X92		Chlamydia infection, genital; female
	Y03		Urethral discharge; male
	Y70		Syphilis; male
	Y71		Gonorrhoea; male
	Y72		Genital herpes; male
	Y76		Condylomata acuminata; male
		A78011	Disease; sexually transmitted
		A78053	Disease; venereal
		D72003	Hepatitis B

Table A6.1 (continued): Code groups from ICPC-2 and ICPC-2 PLUS

		ICPC-2 PLUS	
Problem managed	ICPC-2 rubric	code	ICPC-2/ICPC-2 PLUS label
Sexually transmitted		D72008	Hepatitis C
infection (continued)		U72003	Urethral syndrome
		U72005	Urethritis
		U72006	Urethritis; non-gonococcal; female
		U72007	Urethritis; non-gonococcal; male
		U72008	Urethritis; non specific; female
		U72009	Urethritis; non specific; male
		U72010	Urethritis; chlamydial
		Y73002	Prostatitis
		Y99018	Chlamydia; male
		Y99020	Donovanosis; male
Type 2 diabetes	Т90		Diabetes; non-insulin-dependent
Urinary tract infection	U70		Pyelonephritis/pyelitis
	U71		Cystitis/urinary infection other
Weakness/tiredness	A04		Weakness/tiredness, general

Table A6.1 (continued): Code groups from ICPC-2 and ICPC-2 PLUS

(a) The ICPC-2 rubric '-34' signifies that the concept 'blood test – all' includes all of the blood test rubrics across all chapters of ICPC-2 (excluding the social chapter).

(b) Analysis of health check problems was limited to patients aged 15 years and over.

(c) Analysis of overweight/obesity problems was limited to adult patients aged 18 years and over.

(d) The ICPC-2 rubric '-50' signifies that the concept 'prescription – all' includes all of the prescription rubrics across all chapters of ICPC-2 (excluding the social chapter).

Note: NOS - not otherwise specified; HIV - human immunodeficiency virus; AIDS - acquired immune deficiency syndrome.

Appendix 7: ICPC-2 PLUS code groups used for analysis of pathology tests

Table A7.1: Pathology code groups for MBS groups and individual tests/batter	ries

Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Chemistry		
Amylase	D34004	Test; amylase
C reactive protein (CRP)	A34005	Test; C reactive protein
Calcium/phosphate	A34006	Test; calcium
	A34013	Test; phosphate
	A34024	Test; calcium phosphate
Creatine kinase (CK)	K34004	Test; creatine kinase
Drug screen	A34002	Drug assay
	A34026	Blood drug screen
	A34027	Blood screen
	A35003	Drug screen
	A35005	Urine drug screen
	K34005	Test; digoxin
	N34003	Test; phenytoin
	N34004	Test; valproate
	N34005	Test; carbamazepine
	P34002	Test; lithium
Electrolytes, urea and	A34007	Test; chloride
creatinine (EUC)	A34008	Test; electrolytes
	A34010	Test; electrolytes, urea and creatinine (EUC)
	A34014	Test; potassium
	A34017	Test; sodium
	A34029	Test; urea and electrolytes
	A34034	Test; electrolytes and creatinine
	U34002	Test; creatinine
	U34003	Test; urea
HbA1c	T34010	Test; HbA1c
	T34017	Test; fructosamine
	T34022	Test; HBA1
Ferritin	B34016	Test; ferritin
	B34019	Test; iron studies

Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Folic acid	B34017	Test; folic acid
	B34024	Test; folate
Glucose/glucose tolerance	T34005	Test; glucose
	T34009	Test; glucose tolerance
	T34023	Test; glucose (fasting/random)
	T34025	Test; glucose; fasting
	T34026	Test; glucose; random
Hormone assay	A34003	Hormone assay
	D33015	Test; Anti gliadin antibody
	T34007	Test; cortisol
	T34034	Test; adrenocorticotropic hormone
	W34005	Test; HCG
	W34006	Test; B HCG level (titre/quant)
	X34002	Test; luteinising hormone
	X34003	Test; progesterone
	X34004	Test; oestradiol
	X34005	Test; follicle stimulating hormone
	X34006	Test; SHBG; female
	X34007	Test; free androgen index; female
	Y34004	Test; SHBG; male
	Y34005	Test; free androgen index; male
Lactose intolerance	D38002	Test; lactose intolerance
Lipids	T34001	Check up; cholesterol
	T34004	Test; lipids profile
	T34006	Test; cholesterol
	T34011	Test; cholesterol high-density lipoprotein (HDL)
	T34013	Test; cholesterol low-density lipoprotein (LDL)
	T34016	Test; triglycerides
	T34020	Test; free fatty acids
	T34024	Test; chol/trig
Liver function test (LFT)	A34004	Test; albumin
	D34003	Test; alkaline phosphatase
	D34006	Test; bilirubin
	D34007	Test; gamma-glutamyl transferase
	D34008	Test; liver function
	T34012	Test; lactate dehydrogenase

Table A7.1 (continued): Pathology code groups for MBS groups and individual tests/batteries

Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Multi-biochemical analysis (MBA)	A34012	Test; multi-biochemical analysis (MBA)
	A34021	Test; electrolytes & liver function test (E & LFT)
Other chemistry	A33023	Test; alpha fetoprotein
	A33026	Test; cancer antigen 125
	A33027	Test; cancer antigen 15.3
	A33028	Test; cancer antigen 19.9
	A33029	Test; carcinoembryonic antigen
	A33041	Test; cancer antigen
	A34015	Test; protein
	A34018	Vitamin assay
	A34019	Test; lead
	A34020	Test; blood gas analysis
	A34022	Test; mineral
	A34023	Test; zinc
	A34025	Test; dehydroepiandrosterone sulfate
	A34030	Test; biochemistry
	A34031	Test; blood alcohol
	A34032	Test; prolactin
	A34033	Test; testosterone
	A34037	Test; Glutathione S-transferase
	A34038	Test; magnesium
	A34040	Test; renin
	A35004	Test; urine sodium
	A35007	Test; urine; albumin
	A35008	Test; albumin creatine ratio
	B34023	Test; transferrin
	D34002	Test; alanine aminotransferase
	D35002	Test; 5-hydroxyindoleacetic acid
	K34001	Test; blood; digitalis
	K34006	Test; amino acids
	K34007	Test; troponin
	N34001	Test; blood; phenylhydantoin
	P34003	Test; methadone
	T34018	Test; androgens
	T34019	Test; insulin
	T34021	Test; C peptide

Table A7.1 (continued): Pathology code groups for MBS groups and individual tests/batteries

Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Other chemistry (continued)	T34029	Test; aldosterone
	T34030	Test; parathyroid hormone
	T34035	Test; lipase
	T35002	Test; catecholamines
	W34008	Test; pregnancy associated plasma protein A
	W38002	Amniocentesis
Prostate specific antigen	Y34002	Test; acid phosphatase
	Y34003	Test; prostate specific antigen
Thyroid function (TFT)	T34015	Test; thyroid function
	T34027	Test; thyroxine
	T34028	Test; thyroid stimulating hormone
Urate/uric acid	U34004	Test; urate/uric acid
Vitamin B12	B34015	Test; B12
	D34009	Test; Schillings
Cytopathology		
Cytology	A37002	Test; cytology
	B37003	Test; cytology; blood
	D37002	Test; cytology; digestive
	F37002	Test; cytology; eye
	H37002	Test; cytology; ear
	K37002	Test; cytology; cardiovascular
	L37002	Test; cytology; musculoskeletal
	N37002	Test; cytology; neurological
	R37002	Test; cytology; respiratory
	R37003	Test; sputum cytology
	S37002	Test; cytology; skin
	T37002	Test; cytology; endocrine/metabolic
	U37002	Test; cytology; urology
	W37002	Test; cytology; reproduction
	Y37002	Test; cytology; genital; male
Pap smear	X37001	Pap smear
	X37003	Test; cytology; genital; female
	X37004	Vault smear
	X37005	Pap smear; thin prep

Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Haematology		
Blood grouping & typing	B33001	Test; Coombs
	B33002	Test; blood grouping & typing
	B33009	Test; blood group
	B33013	Test; blood; cross match
Coagulation	B34003	Test; coagulation time
	B34006	Test; part thromboplastin time
	B34009	Test; prothrombin time
	B34014	Test; activated partial thromboplastin time
	B34022	Test; thrombin time
	B34025	Test; international normalised ratio (INR)
	B34026	Test; fibrinogen
	B34028	Test; bleeding time
	B34029	Test; coagulation screen
	K34008	Test; D-Dimer
Erythrocyte sedimentation rate (ESR)	A34009	Test; Erythrocyte sedimentation rate (ESR)
Full blood count (FBC)	A34011	Test; full blood count
Haemoglobin	B34018	Test; haemoglobin
Other blood	A33042	Test; lymphocyte type & count
	A34035	Test; blood film
	A34036	Test; blood thick film
	B33003	RH; antibody titer
	B34005	Test; blood; platelets
	B34007	Test; blood; sickle cell
	B34021	Test; reticulocyte count
	B34031	Test; haemoglobin epg
	B34032	Test; packed cell volume
	B34033	Test; blood; blood
	B37001	Exam; bone marrow
Histopathology (Tissue patholo	gy)	
Histology; skin	S37001	Test; histopathology; skin
Other histology	A37001	Test; histopathology
	B37002	Test; histopathology; blood
	D37001	Test; histopathology; digestive
	F37001	Test; histopathology; eye

Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Other histology (continued)	H37001	Test; histopathology; ear
	K37001	Test; histopathology; cardiovascular
	L37001	Test; histopathology; musculoskeletal
	N37001	Test; histopathology; neurological
	R37001	Test; histopathology; respiratory
	T37001	Test; histopathology; endocrine/metabolic
	U37001	Test; histopathology; urology
	W37001	Test; histopathology; reproductive
	X37002	Test; histopathology; genital; female
	Y37001	Test; histopathology; genital; male
Immunology		
Anti-nuclear antibodies	L33004	Test; anti-nuclear antibodies
Other immunology	A32001	Test; sensitivity
	A33005	Test; immunology
	A33011	Test; human leucocyte antigen
	A33024	Test; bone marrow surface mark
	A33025	Test; serum electrophoresis
	A33051	Test; immune status
	A33052	Test; skin patch
	A38004	Test; deoxyribonucleic acid
	B33005	Test; immunology; blood
	B33007	Test; immunoglobulins
	B33011	Test; immunoglobulin E
	B34027	Test; FBC for surface markers
	B34030	Test; intrinsic factor
	D32001	Test; sensitivity; digestive
	D33004	Test; immunology; digestive
	D33014	Test; endomysial antibody
	D33028	Test; mitochondrial antibodies
	D33031	Test; anti-tissue transglutaminase
	D34010	Test; transglutamase
	F33002	Test; immunology; eye
	H33002	Test; immunology; ear
	K33002	Test; immunology; cardiovascular
	K33003	Test; antineutrophil cytoplasmic antibodies
	L33003	Test; immunology; musculoskeletal

Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Other immunology (continued)	L34001	Test; lupus erythematosus; cell prep
	N33002	Test; immunology; neurological
	R32004	Test; sensitivity; respiratory
	R33004	Test; immunology; respiratory
	S32001	Test; sensitivity; skin
	S33002	Test; immunology; skin
	T33002	Test; immunology; endocrine/metabolic
	U33003	Test; immunology; urology
	W33007	Test; immunology; reproductive
	X33002	Test; immunology; genital; female
	Y33002	Test; immunology; genital; male
Radioallergosorbent test (RAST)	A34016	Test; RAST
Rheumatoid factor	L33001	Test; rheumatoid factor
Infertility and pregnancy tests	W33002	Test; pregnancy
	W34002	Test; blood; pregnancy
	W34003	Test; antenatal
	W34007	Test; pregnancy screen
	Y38002	Test; sperm count
	Y38003	Test; semen examination
Microbiology		
Antibody	A33003	Test; antibody
Cervical swab	X33004	Test; cervical swab M,C&S
Chlamydia	A33006	Test; chlamydia
	A33034	Test; chlamydia direct immunofl
	X33006	Test; viral culture; genital; female
Ear swab and M,C&S	H33003	Test; ear swab M,C&S
Faeces M,C&S	D33002	Stool(s); culture
	D33008	Test; faeces M,C&S
	D36001	Test; faeces; cyst/ova/parasite
Fungal ID/sensitivity	A33008	Test; fungal ID/sensitivity
	A33030	Test; skin scraping fungal M,C&S
Hepatitis serology	D33005	Test; hepatitis A serology
	D33006	Test; hepatitis B serology
	D33007	Test; hepatitis C serology
	D33010	Test; hepatitis D serology
	D33011	Test; hepatitis E serology

Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Hepatitis serology (continued)	D33013	Test; hepatitis serology
	D33016	Test; hepatitis C antibody
	D33017	Test; hepatitis B antigen
	D33018	Test; hepatitis A antibody
	D33019	Test; hepatitis B antibody
	D33020	Test; hepatitis D antibody
	D33021	Test; hepatitis E antibody
	D33022	Test; hepatitis A antigen
	D33023	Test; hepatitis C antigen
	D33024	Test; hepatitis D antigen
	D33025	Test; hepatitis E antigen
	D33026	Test; hepatitis antibody
	D33027	Test; hepatitis antigen
HIV	A33021	Test; cytomegalovirus serology
	B33006	Test; HIV
	B33008	Test; AIDS screen
	B33012	Test; HIV viral load
H pylori	D33009	Test; H Pylori
Other microbiology	A33004	Test; microbiology
	A33007	Test; culture and sensitivity
	A33012	Test; mycoplasma serology
	A33013	Test; parvovirus serology
	A33015	Test; Barmah forest virus
	A33016	Test; Antistreptolysin O Titre
	A33017	Test; herpes simplex culture
	A33019	Test; herpes simplex serology
	A33020	Test; toxoplasmosis serology
	A33033	Test; swab M,C&S
	A33035	Test; serology
	A33036	Antibodies screen
	A33038	Test; rapid plasma regain
	A33039	Test; viral swab M,C&S
	A33040	Test; viral serology
	A33043	Test; human papilloma virus (HPV)
	A33044	Test; Brucella
	A33045	Test; fungal M,C&S

 Table A7.1 (continued): Pathology code groups for MBS groups and individual tests/batteries

Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Other microbiology (continued)	A33046	Test; measles virus antibodies
	A33047	Test; Rickettsial serology
	A33053	Test; Bartonella
	A33054	Test; M,C&S
	A34028	Test; blood culture
	A34039	Test; Q fever
	B33004	Test; microbiology; blood
	B33010	Test; serum immunoglobulins
	D33003	Test; microbiology; digestive
	D33012	Test; rotavirus
	F33001	Test; microbiology; eye
	F33003	Test; eye swab M,C&S
	H33001	Test; microbiology; ear
	K33001	Test; microbiology; cardiovascular
	L33002	Test; microbiology; musculoskeletal
	N33001	Test; microbiology; neurological
	R33001	Culture; tuberculosis
	R33002	Culture; throat
	R33003	Test; microbiology; respiratory
	R33009	Test; influenza serology
	R33010	Test; Legionnaires antibodies
	R33011	Test; respiratory syncytial virus
	S33001	Test; microbiology; skin
	S33005	Test; varicella zoster serology
	S33006	Test; varicella zoster culture
	S33007	Test; nail M,C&S
	T33001	Test; microbiology; endocrine/metabolic
	U33002	Test; microbiology; urology
	W34004	Test; antenatal serology
	W33006	Test; microbiology; reproductive
	X33001	Test; microbiology; genital; female
	X33003	Culture; gonococcal; female
	Y33001	Test; microbiology; genital; male
	Y33003	Culture; gonococcal; male
	Y33004	Test; viral culture; genital; male
	Y33005	Test; urethral/penile swab

Table A7.1 (continued): Pathology code groups for MBS groups and individual tests/batteries

Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Monospot	A33002	Test; monospot
	A33014	Test; Paul Bunnell
	A33031	Test; Epstein Barr virus serology
	A33032	Test; Epstein Barr virus
Nose swab M,C&S	R33008	Test; nose swab M,C&S
Pertussis	R33007	Test; pertussis
Ross River fever	A33009	Test; Ross River Fever
Rubella	A33001	Test; rubella
Skin swab M,C&S	S33003	Test; skin swab M,C&S
Sputum M,C&S	R33005	Test; sputum M,C&S
Throat swab M,C&S	R33006	Test; throat swab M,C&S
Urine M,C&S	U33001	Test; culture; urine
	U33004	Test; urine M,C&S
Vaginal swab and M,C&S	X33005	Test; vaginal swab M,C&S
Sexually transmitted infection	A33010	Test; venereal disease
(STI) screen	A33022	Test; syphilis serology
	A33057	STI screen
Simple basic tests		
Occult blood test	D36003	Test; occult blood
Other simple basic tests	B35001	Test; urine; blood
	R32001	Test; Mantoux
	R32002	Test; tuberculin
	W33001	Test; urine; pregnancy
	W35003	Test; urine; HCG
Other tests NEC		
Blood test	A34001	Test; blood
Urine test	A35001	Test; urine
Urinalysis	A35002	Urinalysis
Faeces test	A36001	Test; faeces
Other test NEC	A35006	Test; urine; full ward test
	A38001	Test; other lab
	A38002	Pathology
	A38003	Test; genetic
	A38005	Test; disease screen
	B38001	Test; other lab; blood
	D34001	Test; blood; digestive
	D35001	Test; urine; digestive

 Table A7.1 (continued): Pathology code groups for MBS groups and individual tests/batteries
Pathology test orders	ICPC-2 PLUS code	ICPC-2 PLUS label
Other test NEC (continued)	D36002	Test; faeces; digestive
	D38001	Test; other lab; digestive
	F34001	Test; blood; eye
	F38001	Test; other lab; eye
	H34001	Test; blood; ear
	H38001	Test; other lab; ear
	K34002	Test; blood; cardiovascular
	K38001	Test; other lab; cardiovascular
	L34003	Test; blood; musculoskeletal
	L38001	Test; other lab; musculoskeletal
	N34002	Test; blood; neurological
	N38001	Test; other lab; neurological
	P34001	Test; blood; psychological
	P35001	Test; urine; psychological
	P38001	Test; other lab; psychological
	R34001	Test; blood; respiratory
	R38001	Test; other lab; respiratory
	S34001	Test; blood; skin
	S38001	Test; other lab; skin
	T34002	Test; blood; endocrine/metabolic
	T35001	Test; urine; endocrine/metabolic
	T38001	Test; other lab; endocrine/metabolic
	U34001	Test; blood; urology
	U35002	Test; urine; urology
	U38001	Test; other lab; urology
	W34001	Test; blood; reproductive
	W35001	Test; urine; reproductive
	W38001	Test; other lab; reproductive
	X34001	Test; blood; genital; female
	X35001	Test; urine; genital; female
	X38001	Test; other lab; genital; female
	Y34001	Test; blood; genital; male
	Y35001	Test; urine; genital; male
	Y38001	Test; other lab; genital; male
	Z38001	Test; other lab; social

 Table A7.1 (continued): Pathology code groups for MBS groups and individual tests/batteries

Note: AIDS – acquired immune deficiency syndrome; FBC – full blood count; H Pylori – Helicobacter pylori; HbA1 – haemoglobin alpha 1; HbA1c – haemoglobin A1c; HCG – human chorionic gonadotropin; HIV – human immunodeficiency virus; ID – identification; M,C&S – microscopy, culture and sensitivity; NEC – not elsewhere classified; SHBG – sex hormone binding globulin.